Glen Canyon Dam

Long-Term Experimental and Management Plan Environmental Impact Statement



FINAL Volume 3—Appendices O–Q U.S. Department of the Interior Bureau of Reclamation, Upper Colorado Region

National Park Service, Intermountain Region

October 2016











Cover photo credits:

Title bar: Grand Canyon National Park Grand Canyon: Grand Canyon National Park Glen Canyon Dam: T.R. Reeve High-flow experimental release: T.R. Reeve Fisherman: T. Gunn Humpback chub: Arizona Game and Fish Department Rafters: Grand Canyon National Park

GLEN CANYON DAM LONG-TERM EXPERIMENTAL AND MANAGEMENT PLAN FINAL ENVIRONMENTAL IMPACT STATEMENT

Joint-Lead Agencies

Bureau of Reclamation National Park Service

Cooperating Agencies

Department of the Interior Bureau of Indian Affairs U.S. Fish and Wildlife Service U.S. Department of Energy Western Area Power Administration Arizona Game and Fish Department Colorado River Board of California Colorado River Commission of Nevada Upper Colorado River Commission Havasupai Tribe Hopi Tribe Hualapai Tribe Kaibab Band of Paiute Indians Navajo Nation Pueblo of Zuni Salt River Project Utah Associated Municipal Power Systems

ABSTRACT

The U.S. Department of the Interior (DOI), through the Bureau of Reclamation and National Park Service (NPS), proposes to develop and implement a Long-Term Experimental and Management Plan (LTEMP) for operations of Glen Canyon Dam. The LTEMP would provide a framework for adaptively managing Glen Canyon Dam operations over the next 20 years, consistent with the Grand Canyon Protection Act of 1992 (GCPA) and other provisions of applicable federal law. The LTEMP would determine specific options for dam operations, non-flow actions, and appropriate experimental and management actions that will meet the GCPA's requirements and minimize impacts on resources within the area impacted by dam operations, including those of importance to American Indian Tribes.

The Final Environmental Impact Statement (FEIS) was developed in accordance with the National Environmental Policy Act of 1969, as amended (NEPA), and followed the implementing regulations developed by the President's Council on Environmental Quality in Title 40 *Code of Federal Regulations* (CFR) Parts 1500 to 1508 and DOI regulations implementing NEPA in 43 CFR Part 46. The FEIS analysis draws on the scientific information that has been collected under the Glen Canyon Dam Adaptive Management Program over the last 20 years to identify the potential environmental effects associated with taking no action, as well as a reasonable range of alternatives to no action for implementing the proposed federal action. Seven alternatives were considered and analyzed for the LTEMP EIS—a no action alternative (Alternative A), a hydropower-focused alternative (Alternative B), three condition-dependent alternatives (Alternatives C, D, and E), and two steady flow alternatives (Alternatives F and G). These alternatives incorporated a broad range of operations and experimental actions that together allowed for a full evaluation of possible impacts of the

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

proposed action. Based on the impact analyses conducted, DOI has chosen Alternative D as both the preferred and the environmentally preferred alternative. Alternative D is expected to result in an improvement in conditions for humpback chub, trout, and the aquatic food base; have the least impact on vegetation, wetlands, and terrestrial wildlife; improve sandbar building potential and conserve sediment; sustain or improve conditions for reservoir and river recreation; improve preservation of cultural resources; respect and enhance Tribal resources and values; and have limited impacts on hydropower resources.

For additional information, visit http://ltempeis.anl.gov or contact:

Rob Billerbeck Colorado River Coordinator National Park Service Intermountain Region 12795 West Alameda Parkway Lakewood, CO 80228 Phone: 303-987-6789 Email: rob_p_billerbeck@nps.gov Katrina Grantz Chief, Adaptive Management Group Bureau of Reclamation Upper Colorado Region 125 South State Street Salt Lake City, UT 84138-1147 Phone: 801-524-3635 Email: kgrantz@usbr.gov

CONTENTS

AE	BSTRA	.CT			iii			
AC	CRONY	YMS AN	ND ABBR	EVIATIONS	lxiii			
1	INTR	ODUCT	TION		1-1			
	1.1	Descrip	otion of the	e Proposed Action	1-4			
	1.2	Purpose of and Need for Action						
	1.3	Lead and Cooperating Agencies and Consulting Tribes						
		1.3.1	Lead Ag	encies	1-9			
		1.3.2	Cooperat	ting Agencies and Consulting Tribes	1-9			
	1.4	Objecti	ves and R	esource Goals of the LTEMP	1-10			
	1.5	Scope of	of the EIS		1-13			
		1.5.1	Affected	Region and Resources	1-13			
		1.5.2	Impact T	opics Selected for Detailed Analysis	1-14			
		1.5.3	Impact T	opics Dismissed from Detailed Analysis	1-15			
	1.6	Role of Adaptive Management						
		1.6.1	History of	of the Existing Adaptive Management Program	1-16			
		1.6.2	Relations	ship of Adaptive Management to NEPA and Changes				
			to Operations					
	1.7	1.7 Role of Decision Analysis in the EIS Process						
	1.8	History	tory, Location, and Setting					
		1.8.1	History a	and Purpose of Glen Canyon Dam	1-18			
		1.8.2	Location	of Glen Canyon Dam and LTEMP Affected Area	1-19			
		1.8.3	Operatio	n of the Glen Canyon Dam	1-20			
		1.8.4	History, Units	Purpose, and Significance of the National Park System	1-22			
			1.8.4.1	Grand Canyon National Park	1-22			
			1.8.4.2	Glen Canyon National Recreation Area	1-24			
			1.8.4.3	Lake Mead National Recreation Area	1-24			
		1.8.5	Tribal La	ands	1-25			
			1.8.5.1	Fort Mojave Tribe	1-26			
			1.8.5.2	Havasupai Tribe	1-26			
			1.8.5.3	Hopi Tribe	1-27			
			1.8.5.4	Hualapai Tribe	1-27			
			1.8.5.5	Navajo Nation	1-27			
			1.8.5.6	Pueblo of Zuni	1-28			
			1.8.5.7	Southern Paiute Tribes	1-28			
	1.9	Laws a	nd Regula	tions Related to Operations of Glen Canyon Dam and				
		Park M	anagemen	ıt	1-29			
		1.9.1	Environr	nental Laws and Executive Orders	1-29			
	1.9.2 Cultural/Historical Laws and Executive Orders							

		1.9.3	American Orders	n Indian and Tribal Consultation Laws and Executive	1-31
		1.9.4	Laws Est	tablishing Criteria Related to Power Marketing	1-31
		1.9.5	Law of th	he River	1-31
	1.10	Related	Actions.		1-31
		1.10.1	Biologica	al Opinions	1-33
		1.10.2	Environm	nental Impact Statements and Related Documents	1-33
		1.10.3	Environm	nental Assessments and Related Documents	1-35
		1.10.4	Other Ac	ctions, Programs, Plans, and Projects	1-36
2	DESC	CRIPTIO	N OF AL	TERNATIVES	2-1
	2.1	Develo	pment of A	Alternatives	2-2
	2.2	Descrip	tions of A	Iternatives Considered in Detail	2-4
		2.2.1	Alternati	ve A (No Action Alternative)	2-8
		2.2.2	Alternati	ve B	2-20
		2.2.3	Alternati	ve C	2-23
			2.2.3.1	Base Operations under Alternative C	2-25
			2.2.3.2	Implementation Process for Experiments under	
				Alternative C	2-27
			2.2.3.3	Sediment-Related Experiments To Be Evaluated under	• • •
				Alternative C	2-37
			2.2.3.4	Aquatic Resource-Related Experiments To Be	• • •
		2.2.4	A 1/ /*	Evaluated under Alternative C	2-38
		2.2.4	Alternati	ve D (Preferred Alternative)	2-41
			2.2.4.1	Base Operations under Alternative D.	2-44
			2.2.4.2	Uperational Flexibility under Alternative D	2-46
			2.2.4.3	Alternative D	2 10
			2244	Communication and Consultation Process for	2-48
			2.2.4.4	Alternative D	2 57
			2245	Sediment_Related Experiments To Be Evaluated under	2-37
			2.2.7.3	Alternative D	2-58
			2246	Aquatic Resource-Related Experiments To Be	2 50
			2.2.1.0	Evaluated under Alternative D	2-62
			2247	Conservation Measures under Alternative D	2-72
		2.2.5	Alternati	ve E	2-72
			2.2.5.1	Base Operations under Alternative E	2-72
			2.2.5.2	Experimental Framework for Alternative E	2-75
		2.2.6	Alternati	ve F	2-80
		2.2.7	Alternati	ve G	2-81
	2.3	Alterna	tives Cons	sidered and Eliminated from Detailed Study	2-84
		2.3.1	Modified	Low Fluctuating Flows with Extended Protocols	2-84

		2.3.2	Naturall	y Patterned Flow Alternative	2-85
		2.3.3	Seasonal	Fluctuations with Low Summer Flow Alternative	2-85
		2.3.4	Grand C	anyon First! Alternative	2-86
		2.3.5	Species	Community and Habitat-Based Alternative	2-86
		2.3.6	Stewards	ship Alternative	2-86
		2.3.7	Twelve-	Year Experiment of Two Steady-Flow Alternatives	2-87
		2.3.8	Decomn	nission Glen Canyon Dam Alternative	2-87
		2.3.9	Fill Lake	e Mead First Alternative	2-88
		2.3.10	Full-Pov	verplant Capacity Operations Alternative	2-88
		2.3.11	Run-of-t	he-River Alternative	2-88
	2.4	Alterna	ative Elem	ents Eliminated from Detailed Study	2-89
		2.4.1	New Inf	rastructure	2-89
		2.4.2	Flow and	d Non-Flow Actions	2-90
	2.5	Summa	ary Compa	arison of Alternatives	2-91
3	AFF	ECTED	ENVIRO	MENT	3-1
	2 1	Drojost	Aroo		2 1
	3.1	2 1 1	Colorad	Divor Sotting	2 1
		3.1.1 2 1 2	Colorado	Softing	2 2
		5.1.2	2 1 2 1	Tribal Parspactives on Goologie Setting	2-2
		313	Climatic	Setting	3-3
		3.1.3	Glen Ca	oven Dam Releases and Flow	3-4
		3.1.4	Colorado	River Ecosystem Resource Linkages	3-6
	32	Water	ater Resources		
	5.2	3 2 1	Hydrolo	mv	3_7
		J.2.1	3 2 1 1	I ake Powell Hydrology	3-8
			3212	Hydrology of the Colorado River Downstream of Glen	50
			5.2.1.2	Canyon Dam	3-11
			3213	Lake Mead Hydrology	3-13
			3214	Seens and Springs	3-14
		322	Water O	nality	3-16
		5.2.2	3221	Lake Powell Water Quality	3-16
			3222	Colorado River Water Quality	3-19
			3223	Lake Mead Water Quality	3-27
		3.2.3	Tribal Pe	erspectives on Water Resources.	3-29
		3.2.4	Hydrolo	gy and Climate Change	3-32
			3.2.4.1	Basis for Runoff Estimates	3-32
			3.2.4.2	Water Variability and Availability	3-33
			3.2.4.3	Seasonal Timing Shifts	3-34
			3.2.4.4	Water Quality	3-35
	3.3	Sedime	ent Resour	ces	3-35
		3.3.1	Backgro	und: Geomorphology of the Colorado River	3-35

		3.3.1.1	Geomorphic Features of the Colorado River	3-37
		3.3.1.2	Glen Canyon Geomorphology	3-40
		3.3.1.3	Marble and Grand Canyon Geomorphology	3-41
	3.3.2	Sediment	Characteristics and Transport Mechanisms	3-43
		3.3.2.1	Particle Size and Sediment Supply	3-45
		3.3.2.2	Sediment Transport Capacity	3-46
		3.3.2.3	River Stage	3-46
	3.3.3	Sediment	Sources	3-46
		3.3.3.1	Gaged Tributaries	3-48
		3.3.3.2	Ungaged Tributaries	3-48
	3.3.4	Sediment	Transport and Storage	3-50
		3 3 4 1	Sediment Transport	3-50
		3 3 4 2	Sediment Storage	3-51
		3 3 4 3	High-Flow Experiments	3-51
		3 3 4 4	Sediment Supply Limitation	3-53
		3345	Sandbar Deposition and Retention	3-53
	335	Lake Del	tas	3-56
34	Natural	Processes		3-57
3 5	Aquatio	E Ecology		3-59
5.5	3 5 1	Aquatic F	Food Base	3-59
	5.5.1	3511	Perinhyton and Rooted Aquatic Plants	3-61
		3512	Plankton	3-63
		3513	Macroinvertebrates	3-63
		3514	Nonnative Invasive Species	3-66
		3515	Food Web Dynamics	3-69
	352	Native Fi	sh	3-70
	5.5.2	3 5 2 1	Snecial Status Fish Snecies	3-72
		3522	Other Native Species	3-86
	353	Nonnativ	e Fish	3-90
	5.5.5	3 5 3 1	Coldwater Nonnative Species	3-94
		3532	Warmwater Nonnative Species	3-100
		3533	Interactions with Native Species	3-101
		3534	Nonnative Fish Control Activities and Effects of Flow	5 101
		5.5.5.4	Conditions	3-104
36	Vegeta	tion		3-109
5.0	3.6.1	Historic 2	and Remnant Rinarian Plant Communities	3-110
	3.6.2	Existing	Rinarian Vegetation Downstream from Glen Canyon Dam	3-111
	5.0.2	3621	Tribal Perspectives on Vegetation	3-119
	363	Special S	tatus Plant Species	3-121
37	Wildlif	e Special D		3-124
5.1	371	Invertebr	ates	3-124
	372	Amphibia	ans and Rentiles	3_125
	373	Birds		3-120
	5.7.5	Dirus		$J^{-1} \Delta I$

	3.7.4	Mammals 3-			
	3.7.5	Special Status Wildlife Species	3-131		
		3.7.5.1 Invertebrates	3-132		
		3.7.5.2 Amphibians and Reptiles	3-136		
		3.7.5.3 Birds	3-136		
		3.7.5.4 Mammals	3-142		
	3.7.6	Tribal Perspectives on Wildlife	3-142		
3.8	Cultura	I Resources	3-144		
	3.8.1	Area of Potential Effect	3-146		
	3.8.2	Description of Cultural Resources and Site Types	3-149		
		3.8.2.1 Archaeological Resources	3-149		
		3.8.2.2 Historic Resources	3-153		
		3.8.2.3 Cultural Landscapes	3-155		
		3.8.2.4 Traditional Cultural Properties and Ethnographic			
		Resources	3-156		
3.9	Tribal l	Resources	3-156		
	3.9.1	Fort Mojave Indian Tribe	3-157		
	3.9.2	Havasupai	3-158		
	3.9.3	Норі	3-158		
	3.9.4	Hualapai	3-159		
	3.9.5	Navajo Nation	3-162		
	3.9.6	Pueblo of Zuni	3-164		
	3.9.7	Southern Paiute Tribes	3-167		
	3.9.8	Indian Trust Assets and Trust Responsibility	3-168		
3.10	Recreat	tion, Visitor Use, and Experience	3-168		
	3.10.1	Glen Canyon Reach of the Colorado River in Glen Canyon			
		National Recreation Area	3-169		
		3.10.1.1 Lees Ferry Recreational Fishery	3-169		
		3.10.1.2 Day-Rafting, Boating, and Camping in the Glen			
		Canyon Reach	3-174		
	3.10.2	The Colorado River in Grand Canyon National Park	3-178		
		3.10.2.1 Campsites in Grand Canyon National Park	3-179		
		3.10.2.2 River Flow and Fluctuation	3-183		
		3.10.2.3 Hualapai Tribe Recreation Program	3-186		
	3.10.3	Recreation Use on Lake Mead and Lake Powell	3-187		
		3.10.3.1 Lake Mead National Recreation Area	3-187		
	2 10 1	3.10.3.2 Lake Powell, Glen Canyon National Recreation Area	3-187		
0.11	3.10.4	Park Operations and Management	3-187		
3.11	Wilder	ness	3-188		
	3.11.1	Law and Policy	3-189		
0.10	3.11.2	Defining Wilderness Character	3-190		
3.12	Visual	Resources	3-192		
	3.12.1	Glen Canyon National Recreation Area	3-193		

		3.12.2	Grand Ca	nyon and the Colorado River	3-195	
		3.12.3	Lake Mea	ad National Recreation Area	3-198	
	3.13	Hydropower				
		3.13.1	Power Op	perations	3-199	
			3.13.1.1	Hydropower Generation	3-199	
			3.13.1.2	Basin Fund	3-200	
			3.13.1.3	Operational Flexibility	3-201	
			3.13.1.4	Scheduling	3-202	
			3.13.1.5	Load/Generation Following and Regulation	3-203	
			3.13.1.6	Capacity Reserves	3-205	
			3.13.1.7	Disturbances and Emergencies and Outage Assistance	3-205	
			3.13.1.8	Transmission System	3-206	
		3.13.2	Power Ma	arketing	3-206	
			3.13.2.1	Wholesale Rates	3-209	
			3.13.2.2	Retail Rates	3-209	
	3.14	Socioed	conomics a	nd Environmental Justice	3-209	
		3.14.1	The Six-C	County Region of Influence	3-210	
			3.14.1.1	Population	3-210	
			3.14.1.2	Income	3-211	
			3.14.1.3	Employment	3-211	
			3.14.1.4	Unemployment	3-213	
			3.14.1.5	Environmental Justice	3-213	
		3.14.2	The Sever	n-State Region of Influence	3-221	
			3.14.2.1	Population	3-221	
			3.14.2.2	Income	3-222	
			3.14.2.3	Employment	3-224	
			3.14.2.4	Unemployment	3-226	
			3.14.2.5	Environmental Justice	3-226	
		3.14.3	Non-Use	Value	3-228	
	3.15	Air Qua	ality		3-229	
		3.15.1	Local Air	Quality	3-229	
		3.15.2	Regional	Air Quality	3-231	
		3.15.3	Regional	Air Emissions	3-234	
	3.16	Climate	e Change		3-236	
4	ENV	IRONMI	ENTAL CO	ONSEQUENCES	4-1	
	4.1	Overall	Analysis a	and Assessment Approach	4-1	
	4.2	Water F	Resources.	11	4-9	
		4.2.1	Analysis	Methods	4-10	
			4.2.1.1	Hydrology	4-10	
			4.2.1.2	Water Quality	4-13	
		4.2.2	Summary	of Impacts	4-14	

		4.2.2.1	Hydrology	4-14
		4.2.2.2	Water Quality	4-37
	4.2.3	Alternativ	ve-Specific Impacts	4-47
		4.2.3.1	Alternative A (No Action Alternative)	4-48
		4.2.3.2	Alternative B	4-49
		4.2.3.3	Alternative C	4-53
		4.2.3.4	Alternative D (Preferred Alternative)	4-55
		4.2.3.5	Alternative E	4-58
		4.2.3.6	Alternative F	4-60
		4.2.3.7	Alternative G	4-62
4.3	Sedime	nt Resourc	es	4-65
	4.3.1	Analysis	Methods	4-66
	4.3.2	Summary	of Impacts	4-70
	4.3.3	Alternativ	/e-Specific Impacts	4-72
		4.3.3.1	Alternative A (No Action Alternative)	4-75
		4.3.3.2	Alternative B	4-77
		4.3.3.3	Alternative C	4-80
		4.3.3.4	Alternative D (Preferred Alternative)	4-81
		4.3.3.5	Alternative E	4-84
		4.3.3.6	Alternative F	4-85
		4.3.3.7	Alternative G	4-86
4.4	Natural	Processes		4-87
	4.4.1	Analysis	Methods	4-88
	4.4.2	Summary	of Impacts	4-88
	4.4.3	Alternativ	ve-Specific Impacts	4-97
		4.4.3.1	Alternative A (No Action Alternative)	4-97
		4.4.3.2	Alternative B	4-97
		4.4.3.3	Alternative C	4-98
		4.4.3.4	Alternative D (Preferred Alternative)	4-99
		4.4.3.5	Alternative E	4-99
		4.4.3.6	Alternative F	4-100
		4.4.3.7	Alternative G	4-101
4.5	Aquatic	Ecology.		4-102
	4.5.1	Analysis	Methods	4-102
		4.5.1.1	Aquatic Food Base	4-102
		4.5.1.2	Nonnative Fish	4-103
		4.5.1.3	Native Fish	4-106
		4.5.1.4	Aquatic Parasites	4-107
	4.5.2	Summary	of Impacts	4-107
		4.5.2.1	Aquatic Food Base	4-107
		4.5.2.2	Nonnative Fish	4-119
		4.5.2.3	Native Fish	4-127
		4.5.2.4	Aquatic Parasites	4-135

	4.5.3	Alternativ	ve-Specific Impacts on Aquatic Resources	4-137
		4.5.3.1	Alternative A (No Action Alternative)	4-137
		4.5.3.2	Alternative B	4-140
		4.5.3.3	Alternative C	4-143
		4.5.3.4	Alternative D (Preferred Alternative)	4-148
		4.5.3.5	Alternative E	4-154
		4.5.3.6	Alternative F	4-157
		4.5.3.7	Alternative G	4-160
4.6	Vegetat	tion		4-163
	4.6.1	Analysis	Methods	4-164
	4.6.2	Summary	of Impacts	4-168
		4.6.2.1	Impacts on Old High Water Zone Vegetation	4-177
		4.6.2.2	Impacts on New High Water Zone	4-179
		4.6.2.3	Wetlands	4-185
		4.6.2.4	Special Status Plant Species	4-187
	4.6.3	Alternativ	ve-Specific Impacts	4-190
		4.6.3.1	Alternative A (No Action Alternative)	4-191
		4.6.3.2	Alternative B	4-192
		4.6.3.3	Alternative C	4-194
		4.6.3.4	Alternative D (Preferred Alternative)	4-196
		4.6.3.5	Alternative E	4-197
		4.6.3.6	Alternative F	4-199
		4.6.3.7	Alternative G	4-201
4.7	Wildlife	ə		4-203
	4.7.1	Analysis	Methods	4-203
	4.7.2	Summary	of Impacts	4-204
		4.7.2.1	Terrestrial Invertebrates	4-204
		4.7.2.2	Amphibians and Reptiles	4-210
		4.7.2.3	Birds	4-212
		4.7.2.4	Mammals	4-214
		4.7.2.5	Special Status Species	4-216
	4.7.3	Alternativ	ve-Specific Impacts on Wildlife	4-227
		4.7.3.1	Alternative A (No Action Alternative)	4-227
		4.7.3.2	Alternative B	4-229
		4.7.3.3	Alternative C	4-230
		4.7.3.4	Alternative D (Preferred Alternative)	4-231
		4.7.3.5	Alternative E	4-232
		4.7.3.6	Alternative F	4-233
		4.7.3.7	Alternative G	4-234
4.8	Cultura	l Resource	2S	4-235
	4.8.1	Compliar	nce with Federal Regulations	4-235
	4.8.2	Analysis	Methods	4-235
	4.8.3	Summary	of Impacts	4-238

	4.8.4	Alternativ	ve-Specific Impacts	4-246
		4.8.4.1	Alternative A (No Action Alternative)	4-246
		4.8.4.2	Alternative B	4-246
		4.8.4.3	Alternative C	4-247
		4.8.4.4	Alternative D (Preferred Alternative)	4-248
		4.8.4.5	Alternative E.	4-248
		4.8.4.6	Alternative F	4-249
		4.8.4.7	Alternative G	4-250
4.9	Tribal I	Resources.		4-251
	4.9.1	Tribal Re	source Goals	4-251
		4.9.1.1	Increase the Health of the Ecosystem in Glen, Marble,	
			and Grand Canyons	4-252
		4.9.1.2	Protect and Preserve Sites of Cultural Importance	4-254
		4.9.1.3	Preserve and Enhance Respect for Canyon Life	4-256
		4.9.1.4	Preserve and Enhance the Sacred Integrity of Glen,	
			Marble, and Grand Canyons	4-258
		4.9.1.5	Maintain and Enhance Healthy Stewardship	
			Opportunities and Maintain and Enhance Tribal	
			Connections to the Canyons	4-259
		4.9.1.6	Economic Opportunity	4-259
		4.9.1.7	Maintain Tribal Water Rights and Supply	4-260
		4.9.1.8	LTEMP Process	4-261
	4.9.2	Analysis	Methods	4-261
	4.9.3	Summary	of Impacts	4-262
	4.9.4	Alternativ	ve-Specific Impacts	4-268
		4.9.4.1	Alternative A (No Action Alternative)	4-268
		4.9.4.2	Alternative B	4-270
		4.9.4.3	Alternative C	4-271
		4.9.4.4	Alternative D (Preferred Alternative)	4-273
		4.9.4.5	Alternative E	4-274
		4.9.4.6	Alternative F	4-276
		4.9.4.7	Alternative G	4-277
4.10	Recreat	tion, Visito	or Use, and Experience	4-279
	4.10.1	Analysis	Methods	4-279
	4.10.2	Summary	of Impacts	4-281
		4.10.2.1	Glen Canyon Fishing	4-281
		4.10.2.2	Glen Canyon Day Rafting	4-290
		4.10.2.3	Glen Canyon Recreational Facilities	4-291
		4.10.2.4	Whitewater Boating	4-291
		4.10.2.5	Reservoir Activities and Facilities	4-292
		4.10.2.6	Tribal Recreation Operations	4-293
		4.10.2.7	Pearce Ferry	4-296
		4.10.2.8	Park Operations and Management	4-296

	4.10.3	Alternativ	ve-Specific Impacts	4-297
		4.10.3.1	Alternative A (No Action Alternative)	4-297
		4.10.3.2	Alternative B	4-298
		4.10.3.3	Alternative C	4-300
		4.10.3.4	Alternative D (Preferred Alternative)	4-301
		4.10.3.5	Alternative E	4-303
		4.10.3.6	Alternative F	4-304
		4.10.3.7	Alternative G	4-306
4.11	Wilden	ness		4-307
	4.11.1	Analysis	Methods	4-307
	4.11.2	Summary	of Impacts	4-308
	4.11.3	Alternativ	ve-Specific Impacts	4-313
		4.11.3.1	Alternative A (No Action Alternative)	4-314
		4.11.3.2	Alternative B	4-314
		4.11.3.3	Alternative C	4-315
		4.11.3.4	Alternative D (Preferred Alternative)	4-316
		4.11.3.5	Alternative E	4-317
		4.11.3.6	Alternative F	4-317
		4.11.3.7	Alternative G	4-318
4.12	Visual	Resources		4-319
4.13	Hydropower			
	4.13.1	Analysis	Methods	4-322
		4.13.1.1	Hydropower Resource and Capacity Expansion Impacts	4-323
		4.13.1.2	Wholesale Rate Impacts	4-330
		4.13.1.3	Retail Rate Impacts	4-332
		4.13.1.4	Hoover Dam Impacts	4-333
	4.13.2	Summary	of Hydropower Impacts	4-334
		4.13.2.1	Monthly Water Release Impacts	4-334
		4.13.2.2	Hydropower Power Generation and Capacity Impacts	4-340
		4.13.2.3	Economic Impacts	4-343
		4.13.2.4	Change in FES Wholesale Rates	4-349
		4.13.2.5	Retail Rate and Bills Impacts	4-350
		4.13.2.6	Impacts of LTEMP Alternatives on Hoover Dam Power	
			Economics	4-351
	4.13.3	Alternativ	ve-Specific Impacts	4-353
		4.13.3.1	Alternative A (No Action Alternative)	4-353
		4.13.3.2	Alternative B	4-353
		4.13.3.3	Alternative C	4-354
		4.13.3.4	Alternative D (Preferred Alternative)	4-355
		4.13.3.5	Alternative E	4-358
		4.13.3.6	Alternative F	4-359
		4.13.3.7	Alternative G	4-360

4.14	Socioed	conomics a	nd Environmental Justice	4-362
	4.14.1	Analysis	Methods	4-362
		4.14.1.1	Recreational Use and Environmental Non-Use Values	4-362
		4.14.1.2	Recreational Economic Impacts	4-364
		4.14.1.3	Electricity Bill Increase and Generation Capacity	
			Expansion Impacts	4-364
		4.14.1.4	Environmental Justice	4-365
	4.14.2	Summary	of Impacts on Socioeconomics and Environmental	
		Justice	-	4-365
		4.14.2.1	Recreational Use Values	4-365
		4.14.2.2	Environmental Non-Use Values	4-378
		4.14.2.3	Recreational Economic Impacts	4-379
		4.14.2.4	Customer Utility Electricity Generation Capacity and	
			Residential Rate Increase Impacts	4-382
		4.14.2.5	Environmental Justice Impacts	4-384
	4.14.3	Alternativ	/e-Specific Impacts	4-390
		4.14.3.1	Alternative A (No Action Alternative)	4-390
		4.14.3.2	Alternative B	4-391
		4.14.3.3	Alternative C	4-393
		4.14.3.4	Alternative D (Preferred Alternative)	4-395
		4.14.3.5	Alternative E	4-396
		4.14.3.6	Alternative F	4-398
		4.14.3.7	Alternative G	4-400
4.15	Air Qua	ality		4-402
	4.15.1	Analysis	Methods	4-402
	4.15.2	Summary	of Impacts	4-403
	4.15.3	Alternativ	/e-Specific Impacts	4-409
		4.15.3.1	Alternative A (No Action Alternative)	4-409
		4.15.3.2	Alternative B	4-409
		4.15.3.3	Alternative C	4-410
		4.15.3.4	Alternative D (Preferred Alternative)	4-410
		4.15.3.5	Alternative E	4-410
		4.15.3.6	Alternative F	4-411
		4.15.3.7	Alternative G	4-411
4.16	Climate	e Change		4-411
	4.16.1	Analysis	Methods	4-412
		4.16.1.1	Effects of LTEMP Alternatives on Climate Change	4-412
		4.16.1.2	Effects of Climate Change on Hydrology and	
		_	Downstream Resources	4-413
	4.16.2	Summary	of Impacts	4-414
		4.16.2.1	Effects of LTEMP Alternatives on Climate Change	4-414
		4.16.2.2	Effects of Climate Change on Hydrology and	
			Downstream Resources	4-419

		4.16.3	Alternative	e-Specific Impacts	4-421
			4.16.3.1	Alternative A (No Action Alternative)	4-421
			4.16.3.2	Alternative B	4-424
			4.16.3.3	Alternative C	4-424
			4.16.3.4	Alternative D (Preferred Alternative)	4-424
			4.16.3.5	Alternative E	4-425
			4.16.3.6	Alternative F	4-425
			4.16.3.7	Alternative G	4-426
	4.17	Cumula	tive Impact	ts	4-426
		4.17.1	Past, Prese	ent, and Reasonably Foreseeable Future Actions	
			Affecting	Cumulative Impacts	4-426
			4.17.1.1	Past and Present (Ongoing) Actions	4-427
			4.17.1.2	Reasonably Foreseeable Future Actions	4-442
		4.17.2	Climate-R	elated Changes	4-445
		4.17.3	Cumulativ	e Impacts Summary by Resource	4-446
			4.17.3.1	Water Resources	4-446
			4.17.3.2	Sediment Resources	4-457
			4.17.3.3	Natural Processes	4-458
			4.17.3.4	Aquatic Ecology	4-458
			4.17.3.5	Vegetation	4-461
			4.17.3.6	Wildlife	4-463
			4.17.3.7	Cultural Resources	4-466
			4.17.3.8	Tribal Resources	4-467
			4.17.3.9	Recreation, Visitor Use, and Experience.	4-468
			4.17.3.10	Wilderness	4-469
			4.17.3.11	Visual Resources	4-470
			4 17 3 12	Hydropower	4-471
			4 17 3 13	Socioeconomics and Environmental Justice	4-472
			4 17 3 14	Air Quality and Climate Change	4-472
	4 18	Unavoi	dable Adve	rse Impacts	4-474
	4 19	Relatio	nshin hetwe	een Short-Term Use and Long-Term Productivity	4-475
	4 20	Irrevers	wible and Irr	retrievable Commitments of Resources	4-475
		1110 010			,.
5	CON	SULTAT	FION AND	COORDINATION	5-1
	51	Consult	tation and C	Coordination with Other Agencies and Programs	5-1
	0.1	5 1 1	US Depa	rtment of the Interior	5-1
		512	Cooperatir	ng Agencies	5-1
		513	American	Indian Tribes	5-2
		514	Other Con	sultations	5-6
		v.1.1	5141	National Historic Preservation Act	5-6
			5142	State and Local Water and Power Agency Coordination	5-7
			5143	U S Fish and Wildlife Service	5-7

	5.2	Public	Involvement	5-8
		5.2.1	Public Scoping Process and Comments Received	5-8
		5.2.2	Public Meetings on Alternatives	5-10
		5.2.3	Glen Canyon Dam Adaptive Management Working Group	5-11
	5.2.4	Public	Involvement on the LTEMP DEIS	5-11
6	REFE	ERENCI	ES	6-1
7	LIST	OF PRI	EPARERS	7-1
8	GLO	SSARY		8-1
Ał	PPEND	DIX A:	Adaptive Management Working Group Desired Future Conditions	A-1
	A.1	Desire	d Future Conditions: Colorado River Ecosystem	A-5
		A.1.1	DFC Description	A-5
		A.1.2	DFC Background and Legislation	A-6
		A.1.3	Why the Colorado River Ecosystem DFCs Are Important	A-6
		A.1.4	Colorado River Ecosystem DFCs	A-6
			A.1.4.1 Sediment-Related Resources DFCs	A-6
			A.1.4.2 Water Quality DFCs	A-7
			A.1.4.3 Colorado River Ecosystem Aquatic Resource DFCs	A-7
			A.1.4.4 Colorado River Ecosystem Riparian Resource DFCs	A-9
		A.1.5	Colorado River Ecosystem DFCs Additional Information	A-9
			A.1.5.1 Colorado River Ecosystem Linkages	A-9
		_	A.1.5.2 Colorado River Ecosystem Metrics	A-10
	A.2	Power	Desired Future Conditions	A-10
		A.2.1	Power DFC Description	A-10
		A.2.2	Power DFC Background and Legislation	A-10
		A.2.3	Why the Power DFC Is Important	A-11
		A.2.4	Power DFCs	A-11
		A.2.5	Power DFC Additional Information	A-12
			A.2.5.1 Power Linkages	A-12
	• 2	0.1	A.2.5.2 Power Metrics	A-12
	A.3		al Resources Desired Future Conditions	A-13
		A.3.1	DEC Declarger d en d Legislation	A-13
		A.3.2	DFC Background and Legislation	A-13
		A.3.3	NDUD Elizible Historic Promotion DECa	A-13
		A.3.4	NKHIT Eligible Historic Properties DFUS A 2 4 1 Drahistoria Arabasalagiaal Sitag and Uistaria Sitag	A-14
			A.3.4.1 Preditional Cultural Dranartica	A-14
		٨ 2 5	NDUD Eligible Historic Droportion DEC Additional Information	A-14
		A.3.3	A 2.5.1 NDUD Eligible Historia Properties Linkages	A-13
			A.S.S.1 NKHP Eligible Historic Properties Linkages	A-13

		A.3.5.2	NRHP Eligible Historic Properties Metrics	A-16	
	A.3.6	Resource	es of Traditional Cultural Significance but Not NRHP		
		Eligible.	~	A-16	
	A.3.7	Resource	es of Traditional Cultural Significance DFCs	A-16	
	A.3.8	Resource	es of Traditional Cultural Significance Linkages	A-17	
	A.3.9	Resource	es of Traditional Cultural Significance Metrics	A-17	
A.4	Recreation Desired Future Conditions				
	A.4.1	Recreation	on DFC Description	A-17	
	A.4.2	DFC Bac	kground and Legislation	A-17	
	A.4.3	Why the	Recreation DFC Is Important	A-18	
		A.4.3.1	Grand Canyon National Park	A-18	
		A.4.3.2	Glen Canyon National Recreation Area	A-18	
	A.4.4	Recreation	on DFCs	A-18	
		A.4.4.1	River Recreation in Grand Canyon National Park	A-18	
		A.4.4.2	River Recreation in Glen Canyon National Recreation		
			Area	A-19	
		A.4.4.3	Blue Ribbon Trout Fishery in Glen Canyon National		
			Recreation Area	A-19	
		A.4.4.4	River Corridor Stewardship	A-19	
	A.4.5	Recreation	on DFC Additional Information	A-20	
	A.4.6	Recreation	on Linkages	A-20	
	A.4.7	Recreation	on Metrics	A-20	
A.5	Referen	nce		A-21	
APPEND	IX B: P	erformand	ce Metrics Used to Evaluate Alternatives	B-1	
B.1	Aquation	e Ecology		B-3	
	B.1.1	Humpba	ck Chub	B-3	
	B.1.2	Other Na	tive Fish	B-5	
	B.1.3	Trout Fis	shery	B-5	
	B.1.4	Nonnativ	ve Aquatic Species	B-6	
B.2	Archae	ological a	nd Cultural Resources	B-7	
B.3	Hydrop	ower and	Energy	B-9	
B.4	Natural	Processes	5	B-10	
B.5	Recreat	tional Exp	erience	B-11	
	B.5.1	Grand Ca	anyon Metrics	B-11	
	B.5.2	Glen Car	nyon Metrics	B-15	
B.6	Riparia	n Vegetat	ion	B-16	
B.7	Sedime	nt		B-18	
B.8	Tribal I	Resources		B-18	
B.9	Water I	Delivery		B-24	
B.10	Referen	nces		B-25	

APPEND	DIX C: I	Decision A	nalysis to Support Development of the Glen Canyon Dam	C 1	
Long	- I CIIII E	хрепшени		C-1	
APPENE	DIX D: H	Iydrology	Technical Information and Analysis	D-1	
D.1	Analys	is Methods	5	D-3	
	D.1.1	Backgrou	ınd	D-3	
	D.1.2	Initial Co	onditions	D-4	
	D.1.3	Reservoi	r Operations	D-4	
		D.1.3.1	Upper Basin Reservoirs above Lake Powell	D-4	
		D.1.3.2	Lake Powell and Lake Mead	D-5	
		D.1.3.3	Lake Mohave and Lake Havasu	D-9	
	D.1.4	Represen	tation of the Different Alternatives in CRSS	D-9	
		D.1.4.1	Experimental Components Modeled in CRSS	D-22	
	D.1.5	Input Hy	drology	D-23	
	D.1.6	Input De	mands	D-26	
	D.1.7	Other Ke	v Assumptions	D-26	
D.2	Supple	mental Inf	ormation on Impact Modeling	D-27	
	D.2.1	Low Sun	nmer Flows	D-27	
	D.2.2	Modeled	Annual Releases Extending beyond the End of the Water		
		Year		D-28	
	D.2.3	Lake Ele	vation	D-29	
D.3	Referen	nces		D-36	
APPENI	DIX E · S	ediment R	esources Technical Information and Analysis	F-1	
			esources reenheur information une rinarysis	LI	
E.1	Introdu	ction		E-3	
	E.1.1	Analysis	Period	E-3	
	E.1.2	General S	Scope	E-3	
E.2	Methods				
	E.2.1	Sand Buc	lget Model	E-4	
		E.2.1.1	Model Description	E-4	
		E.2.1.2	Sand Budget Model Modifications	E-5	
		E.2.1.3	Modified Sand Budget Model Inputs	E-6	
	E.2.2	Sediment	t Metrics	E-8	
		E.2.2.1	Sand Load Index	E-9	
		E.2.2.2	Standard Deviation of High Flows	E-9	
		E.2.2.3	Sand Mass Balance Index	E-10	
E.3	Results			E-10	
	E.3.1	HFEs De	termined by Alternative	E-10	
	E.3.2	Metrics		E-11	
		E.3.2.1	Sand Load Index	E-12	
		E.3.2.2	Standard Deviation of High Flows	E-12	

	E.3.3	Sand Ma	ass Balance Index	E-12		
	E.3.4	Alternat	ive Performance under Climate Change Scenarios	E-13		
	E.3.5	Relative	Impacts of Dam Operations and Hydrology on	F 10		
Γ 4	T 1 F	Performa	ance	E-13		
E.4	Lake L	Deltas	·····	E-15		
E.5	Limita	tions and I	Known Issues	E-17		
	E.5.1	Geograp	hic Scope	E-17		
	E.5.2	Modelin	g Improvements	E-17		
E.6	Refere	nces		E-18		
APPENI	DIX F: A	Aquatic Re	esources Technical Information and Analysis	F-1		
F.1	Introdu	action		F-3		
F.2	Aquati	c Food Ba	se Assessment	F-4		
	F.2.1	Descript	ion of the Aquatic Food Base Downstream from Glen			
		Canyon	Dam	F-4		
		F.2.1.1	The Aquatic Food Base Prior to Construction of Glen			
			Canyon Dam	F-5		
		F.2.1.2	The Aquatic Food Base of the Colorado River			
			Downstream from Glen Canyon Dam	F-5		
		F.2.1.3	Influence of New Zealand Mudsnail on the Aquatic			
			Food Base	F-13		
	F.2.2	Impacts	of LTEMP Alternatives on the Aquatic Food Base	F-14		
		F.2.2.1	Flow Effects on the Aquatic Food Base	F-15		
		F.2.2.2	Temperature Effects on the Aquatic Food Base	F-23		
	F.2.3	Conclusi	ion	F-26		
F.3	Modeling Effects of LTEMP Alternatives on Rainbow Trout and					
	Humpback Chub F-2					
	F.3.1	Model C	Overview	F-28		
		F.3.1.1	Glen Canyon Trout Submodel	F-29		
		F.3.1.2	Trout Movement Submodel	F-36		
		F.3.1.3	Humpback Chub Population Submodel	F-40		
	F.3.2	Results f	for LTEMP Alternatives	F-52		
		F.3.2.1	Rainbow Trout Performance Measures	F-52		
		F.3.2.2	Humpback Chub Performance Measures	F-60		
F.4	Model	ing the Eff	fects of LTEMP Alternatives on Temperature Suitability	F-62		
	F.4.1	Model C	Overview	F-64		
	F.4.2	Humpba	ck Chub Aggregations	F-67		
		F.4.2.1	Historic Temperature Suitability for Humpback Chub	F-69		
		F.4.2.2	Results for LTEMP Alternatives	F-70		
	F.4.3	Other Na	ative Fish	F-71		
		F.4.3.1	Historic Temperature Suitability for Native Fish	F-73		
		F.4.3.2	Results for LTEMP Alternatives	F-73		

	F.4.4	Nonnative Fish	F-78
		F.4.4.1 Historic Temperature Suitability for Nonnative Fish	F-79
		F.4.4.2 Results for LTEMP Alternatives	F-79
	F.4.5	Aquatic Parasites	F - 81
		F.4.5.1 Historic Temperature Suitability for Aquatic Parasites	F-84
		F.4.5.2 Results for LTEMP Alternatives	F-86
F.5	Referen	nces	F-90
APPEN	DIX G: V	Vegetation Technical Information and Analysis	G-1
G.1	Analys	is Methods	G-3
	G.1.1	Old High-Water Zone Analysis	G-4
	G.1.2	New High-Water Zone	G-5
		G.1.2.1 Native Cover Metric	G-6
		G.1.2.2 Native Diversity Metric	G-7
		G.1.2.3 Native/Nonnative Ratio Metric	G-7
		G.1.2.4 Arrowweed Metric	G-8
		G.1.2.5 Overall Score	G-8
	G.1.3	Wetlands	G-9
G.2	Alterna	ative-Specific Impacts	G-10
	G.2.1	Alternative A (No Action Alternative)	G-10
	G.2.2	Alternative B	G-10
	G.2.3	Alternative C	G-10
	G.2.4	Alternative D (Preferred Alternative)	G-11
	G.2.5	Alternative E	G-11
	G.2.6	Alternative F	G-12
	G.2.7	Alternative G	G-12
G.3	Summa	ary	G-12
G.4	Referen	nces	G-13
APPEN	DIX H: (Cultural Resources Technical Information and Analysis	H-1
H.1	Wind Tr	ansport of Sediment	H-3
	H.1.1	Wind Transport of Sediment—Methods	H-4
	H.1.2	Wind Transport of Sediment—Results	H-5
H.2	Flow Eff	fects on Cultural Resources in Glen Canyon	H-8
	H.2.1	Flow Effects on Cultural Resources in Glen Canyon—Methods	H-9
	H.2.2	Flow Effects on Cultural Resources in Glen Canyon—Results	H-10
H.3	Time Of	f River	H-13
	H.3.1	Time Off River—Methods	H-13
	H.3.2	Time Off River—Results	H-14
H.4	Reference	es	H-16

APPENI	DIX I: T	ribal Reso	urces Technical Information and Analysis	I-1		
I.1	Quanti I.1.1	fiable Mea Riparian	usures Used to Assess Impacts on Tribal Resources Diversity	I-4 I-4		
	I.1.2	Wetland	Abundance	I-8		
	I.1.3	Trout Ma	anagement Flows	I-10		
	I.1.4	Mechani	cal Removal of Trout	I-12		
	I.1.5	Water Le	evels at Lake Powell	I-14		
	I.1.6	Access to	o Resources	I-17		
I.2	Refere	nces		I-18		
APPENI	DIX J: R	ecreation.	Visitor Use, and Experience Technical Information and			
Anal	ysis			J-1		
I 1	Recrea	tional Exp	erience Metrics	I_3		
J .1	I 1 1	Grand C	anyon Metrics	J-J I-4		
	J112	Glen Ca	nyon Metrics	J-4		
12	Metric Definitions Analysis Methods and Results					
0.2	J.2.1	Camping	Area Index	J-5		
		J.2.1.1	Camping Area Index—Methods	J-6		
		J.2.1.2	Camping Area Index—Results	J-7		
	J.2.2	Navigati	onal Risk Index	J-8		
		J.2.2.1	Navigational Risk Index—Methods	J-9		
		J.2.2.2	Navigational Risk Index—Results	J-10		
	J.2.3	Fluctuati	on Index	J-11		
		J.2.3.1	Fluctuation Index—Methods	J-12		
		J.2.3.2	Fluctuation Index—Results	J-13		
	J.2.4	Time Of	f River	J-14		
		J.2.4.1	Time Off River Index—Methods	J-14		
		J.2.4.2	Time Off River Index—Results	J-15		
	J.2.5	Glen Car	nyon Rafting Use	J-15		
		J.2.5.1	Glen Canyon Rafting Use Metric—Methods	J-17		
		J.2.5.2	Glen Canyon Rafting Use Metric—Results	J-18		
	J.2.6	Glen Car	nyon Inundation Metric	J-18		
		J.2.6.1	Glen Canyon Inundation Metric—Methods	J-19		
		J.2.6.2	Glen Canyon Inundation Metric—Results	J-21		
J.3	Lake P	owell and	Lake Mead Dock Access	J-21		
J.4	Summa	ary		J-26		
J.5	Refere	nces		J-26		

K.1 Ec	onomic Value	of Glen Canyon Dam Powerplant Capacity and Energy
Pre	oduction	
К.	1.1 Power S	ystems Background
К.	1.2 Glen Ca	nyon Dam, Reservoir, and Powerplant Background
К.	1.3 Power S	ystems Geographic Scope
	K.1.3.1	Top Tier: General Western Interconnection Perspective Modeling
	K.1.3.2	Middle Tier: LTF Customer Utility Systems
	K.1.3.3	Bottom Tier: WAPA SLCA/IP Hydropower Resources
K.	1.4 Overvie	w of Power Systems Methods
K.	1.5 Descript	ion of Individual Power System Models
	K.1.5.1	Colorado River Simulation System Model
	K.1.5.2	Representative Trace Tool.
	K.1.5.3	Hydropower Outage Model
	K.1.5.4	Generation and Transmission Maximization-Lite
	K.1.5.5	Sand Budget Model
	K.1.5.6	Large SLCA/IP Powerplant Spreadsheets
	K.1.5.7	Small SLCA/IP Powerplant Spreadsheet
	K.1.5.8	Loads Shaping Algorithm
	K.1.5.9	AURORA
	K.1.5.10	LMP Calibration Spreadsheet
	K.1.5.11	Firm Capacity Spreadsheet
K.	1.6 SLCA/I	P Market System, Data Sources, and Model
	K.1.6.1	Historical Data Sources
	K.1.6.2	AURORA Model Dispatch Results for 2013
	K.1.6.3	SLCA/IP Market System Projections
K.	1.7 Glen Ca	nyon Dam Powerplant Capacity Cost and Benefit
	Methode	ology
	K.1.7.1	Treatment of Glen Canyon Dam Capital and Fixed
	K.1.7.2	WAPA's SLCA/IP LTF Obligations and Glen Canyon
		Dam Replacement Capacity
	K.1.7.3	WAPA SLCA/IP Firm Hydropower Capacity
	K.1.7.4	Firm Capacity Curves for LTEMP Power Systems
		Analyses for the Peak Month of August
	K.1.7.5	AURORA Capacity Expansion Reserve Margin Targets
		and Capacity Additions
	K.1.7.6	Dispatch Performed by AURORA Model Capacity Expansion Runs
	K.1.7.7	Rationale for the Selection of Hydrology Conditions
		Used for Capacity Expansion Runs

	K.1.8	Glen Can	yon Dam Energy Economic Benefits Methodology	K-90		
	K.1.9	Net Prese	ent Value Calculations and Study Period Adjustments	K-94		
	K.1.10	Power Sy	vstems Results	K-96		
		K.1.10.1	Main Drivers of Differences among Alternatives	K-97		
		K.1.10.2	Capacity Expansion Modeling	K-101		
		K.1.10.3	Economic Impacts	K-108		
		K.1.10.4	Sensitivity of Results to Exceedance Level	K-116		
		K.1.10.5	Sensitivity of Results to Discount Rate	K-121		
		K.1.10.6	Sensitivity of Results to the Base Capacity Expansion			
			Path	K-121		
		K.1.10.7	Sensitivity of Results to the Assumed Future			
			Hydrological Conditions	K-123		
		K.1.10.8	Sensitivity of Results to Changes in Ancillary Services	K-125		
		K.1.10.9	Sensitivity of Results to Customer Capacity Expansion			
			Assumptions	K-129		
		K.1.10.10	Summary of Economic Ranking	K-131		
K.2	WAPA	's SLCA/I	P Firm Electric Service Rate Impacts	K-131		
	K.2.1	Relations	hip between the Economic Impacts of LTEMP			
		Alternativ	ves and Impacts on SLCA/IP FES Rates	K-133		
	K.2.2	Temporal	l Scope of the Analysis and Input Data	K-133		
	K.2.3	SLCA/IP Rate Setting				
	K.2.4	Calculati	on of Net Electrical Energy Expense	K-135		
		K.2.4.1	SLCA/IP Electrical Production	K-135		
		K.2.4.2	Sustainable Hydropower and Available Hydropower			
			Capacity and Energy	K-135		
	K.2.5	Calculation	on of Capacity Expenses and Total Net Costs	K-139		
	K.2.6	WAPA R	Leplacement Resources	K-143		
	K.2.7	Post-2024	4 Marketing Period	K-144		
	K.2.8	Power Re	epayment Studies to Determine Rate Impacts	K-146		
		K.2.8.1	PRS Expenses	K-146		
		K.2.8.2	PRS Revenue Distribution	K-149		
		K.2.8.3	The SLCA/IP PRS	K-149		
		K.2.8.4	Standard PRS Rate-Setting Method Versus the Method			
			Used in This Analysis	K-150		
	K.2.9	Results		K-150		
		K.2.9.1	Pinch-Point Year	K-152		
	K.2.10	Definition	ns Used in Section K.2	K-153		
K.3	Impacts	s on Retail	Electricity Rates	K-154		
	K.3.1	Analysis	Approach	K-154		
		K.3.1.1	Database of Sales, Rates, and SLCA/IP Allocation for			
			Retail Utility Systems	K-156		
		K.3.1.2	Incorporation of Power Systems Analysis and Capital			
			Recovery Factors	K-158		

		K.3.1.3	Inflation Rates, Sales Growth, and Interest Rates from EIA	K-161
	V 2 D	K.3.1.4	Calculation Process for Computing Rate and Bill Impacts	K-162
	N .3.2	K esuits	Grid Cost Changes Palative to WAPA Wholesale	K- 10/
		K .J.2.1	Revenues	K-168
		к 3 2 2	Retail Rate Changes for Individual Systems and	K -100
		11.J.2.2	SLCA/IP Power Relative to Total Resources	K-169
		К 3 2 3	Using Regression Equations to Approximate Retail	IX 109
		11.5.2.5	Rate Changes for Systems Not Included in the Database	K-171
	K.3.3	Summary	v of Impacts	K-172
		K.3.3.1	Average Rate Impacts under LTEMP Alternatives over	
		K 2 2 2	the 20-Year LTEMP Period	K-1/2
		K.3.3.2	Average Year-by-Year Rate Impacts	K- 1/4
		K.3.3.3	af LTEMP Alternatives	V 170
	K 2 /	Importa	of LTEMP Alternatives	K-1/9 V 195
	K.3.4 V 2.5	Altornatio	vo Spacific Impacts	K-103 V 197
	K.3.3	K 3 5 1	Alternative A	K-10/ K 197
		K_{352}	Alternative B	K-107 K-187
		K 3 5 3	Alternative C	K-107 K-188
		K 3 5 4	Alternative D	K-189
		K 3 5 5	Alternative E	K-189
		K 3 5 6	Alternative F	K-190
		K 3 5 7	Alternative G	K-190
K.4	Financi	ial Impacts	of LTEMP Alternatives on American Indian Tribes	K-190
	K.4.1	Contractu	al Requirements for Calculating and Delivering Benefits	
		to Tribes		K-192
	K.4.2	Calculation	on of Tribal Benefit Baseline under Alternative A	
		(No Actio	on Alternative)	K-193
	K.4.3	Calculation	on of Change in Tribal Benefit as a Result of LTEMP EIS	
		Alternativ	ves	K-193
	K.4.4	Impacts c	on Tribes through a Change in the Retail Rate of the	
		Electrical	Supplier to Tribal Lands	K-194
	K.4.5	Calculation	on of Tribal Impacts for Tribes That Are Direct SLCA/IP	
		Recipient	S	K-195
	K.4.6	Total Imp Rate Effe	bact of LTEMP EIS Alternatives: Benefit Change and et	K-196
	K.4.7	Total Imp	pact on Tribes and Tribal Members Versus Retail Rate	
		Changes	to Households	K-202
	K.4.8	Conclusio	ons	K-202
K.5	Impact	s of LTEM	P Alternatives on Lake Mead and the Hoover Dam	
	Powerp	olant		K-206

	K.5.1	Hoover Analysis Methods, Model, and Supporting Data	K-209
	K.5.2	Hoover Monthly Energy Production and Water-to-Power	
		Conversion	K-210
	K.5.3	Hoover Maximum Operational Capacity and Firm Capacity	K-213
	K.5.4	Economic Value of Hoover Powerplant Energy	K-213
	K.5.5	Sensitivity of Model Results	K-214
K.6	Referen	ices	K-215
Attachme	ent K.1:	Geographic Scope of the Analysis	K-219
Attachme	ent K.2:	AURORA Western Interconnection Spot Market Energy Prices	
		Adjustments	K-223
Attachme	ent K.3:	Selection of Representative Trace	K-226
Attachme	ent K.4:	Discounting Procedures	K-244
Attachme	ent K.5:	Forced Outage Scenario Generation for Hydroelectric Power	
		Facilities	K-247
Attachme	ent K.6:	Forecast of Monthly Peak Loads and Energy by SLCA/IP	
		Long-Term Firm Customer	K-250
Attachme	ent K.7:	Analysis of the Timing of the Peak Load	K-252
Attachme	ent K.8:	Analysis of Capacity Determinations — Comparing Results Using	
		a Range of Exceedence Levels and Two Summer Peak Months	K-258
Attachment K.9:		Results of a Survey of Electric Utility Integrated Resource Plans	K-260
Attachm	ent K.10:	Indices Used for Converting Dollars from One Year to Another	K-270
Attachm	ent K.11:	Summary IRP of Capacity Additions in the Joint System and	
		Surrounding Utility Systems through the End of CY2034	K-275
Attachme	ent K.12:	Annual SLCA/IP Allocations to American Indian Tribes	
		and Benefit Information	K-277
APPENI	DIX L · S	ocioeconomic Technical Information and Analysis	L-1
			21
L.1	Recreat	ion Economic Analyses	L-3
	L.1.1	Recreation Use Values	L-3
		L.1.1.1 Lake_Full Utility Model	L-4
		L.1.1.2 GCRec_Full Utility Model	L-5
	L.1.2	Recreation Non-Use Values	L-6
	L.1.3	Regional Recreation Economic Impacts	L-10
	L.1.4	Estimates of Recreational Economic Impacts	L-12
L.2	Regiona	al Electricity Analyses	L-13
	L.2.1	Regional Electricity Price Impacts	L-13
	L.2.2	Regional Electricity Generating Capacity Expansion Impacts	L-15
L.3	Additio	nal Socioeconomic Data	L-15
	L.3.1	Urban Population in the Six-County Region	L-15
	L.3.2	Urban Income in the Six-County Region	L-20
L.4	Referen	ices	L-21

APPENI	DIX M: Air Quality and Climate Change Technical Information and Analysis	M-1
M.1	Analysis Methods	M-4
	M.1.1 System Power Generation	M-4
	M.1.2 Spot Market	M-5
	M.1.3 Generation Type	M-6
M.2	Results	M-6
	M.2.1 SO ₂ and NO _X	M-6
	M.2.2 GHG Emissions	M-8
M.3	References	M-9
APPENI	DIX N: Government-to-Government and National Historic Preservation	NT 1
Act (Consultation	N-I
N.1	Government-to-Government Consultation	N-3
N.2	Consultation with the State Historic Preservation Office	N-31
N.3	Reference	N-31
Attachm	ent N.1	N-33
Attachm	ent N.2	N-75
Attachm	ent N.3	N-137
APPENI Expe	DIX O: Biological Assessment for the Glen Canyon Dam Long-Term rimental and Management Plan	O-1
APPENI	DIX P: High-Flow Experiment Protocol for the Preferred Alternative	P-1
P.1	Decision-Making Process	P-3
	P.1.1 Planning	P-4
	P.1.2 Modeling	P-5
	P.1.3 Decision and Implementation	P-8
P.2	Operation of Glen Canyon Dam under the HFE Protocol	P-9
	P.2.1 Potential Operation of Glen Canyon Dam during the Fall	
	HFE Implementation Window	P-10
	P.2.2 Potential Operation of Glen Canyon Dam during the Spring	
	HFE Implementation Window	P-10
P.3	High Flow Experiments to be Evaluated under the Preferred Alternative	P-11
	P.3.1 Sediment-Triggered Spring HFEs	P-11
	P.3.2 Proactive Spring HFEs.	P-12
	P.3.3 Sediment-Triggered Fall HFEs	P-13
	P.3.4 Extended-Duration Fall HFEs	P-13
REF	ERENCES	P-14
APPENI	DIX O: Responses to Public Comments on the Glen Canyon Dam Long-Term	

FIGURES

1-1	Generalized Locations of Glen Canyon Dam, Lake Powell, the Colorado River between Lake Powell and Lake Mead, and Adjacent Lands	1-5
1-2	Glen Canyon Dam	1-6
1-3	Map of the Colorado River between Lake Powell and Lake Mead	1-20
1-4	Indian Reservations within or Adjacent to the LTEMP EIS Project Area	1-25
2-1	Mean, Minimum, and Maximum Daily Flows under Alternative A in an 8.23-maf Year Based on Values Presented in Table 2-3	2-18
2-2	Simulated Hourly Flows under Alternative A in an 8.23-maf Year	2-18
2-3	Simulated Hourly Flows under Alternative A for a Week in July in an 8.23-maf Year Showing Typically Lower Weekend Flows	2-19
2-4	Mean, Minimum, and Maximum Daily Flows under Alternative B in an 8.23-maf Year Based on Values Presented in Table 2-4	2-21
2-5	Simulated Hourly Flows under Alternative B in an 8.23-maf Year	2-22
2-6	Simulated Hourly Flows under Alternative B for a Week in July in an 8.23-maf Year Showing Typically Lower Weekend Flows	2-22
2-7	Example Mean, Minimum, and Maximum Daily Flows for a Hydropower Improvement Experiment under Alternative B in an 8.23-maf Year	2-24
2-8	Simulated Hourly Flows for a Hydropower Improvement Experiment under Alternative B in an 8.23-maf Year	2-24
2-9	Simulated Hourly Flows for a Hydropower Improvement Experiment under Alternative B for a Week in July in an 8.23-maf Year	2-25
2-10	Mean, Minimum, and Maximum Daily Flows under Base Operations of Alternative C in an 8.23-maf Year Based on the Values Presented in Table 2-5	2-26
2-11	Simulated Hourly Flows under Alternative C in an 8.23-maf Year	2-28
2-12	Simulated Hourly Flows under Alternative C for a Week in July in an 8.23-maf Year Showing Typically Lower Weekend Flows	2-28
2-13	Decision Tree for Sediment-Related Actions under Alternative C	2-29

2-14	Decision Tree for Humpback Chub-Related Actions under Alternative C	2-30
2-15	Example Implementation of a Two-Cycle TMF in June and July with Resumption of Normal Fluctuations between Cycles and Afterward	2-40
2-16	Mean, Minimum, and Maximum Daily Flows under Triggered Low Summer Flows of Alternative C in an 8.23-maf Year Based on the Values Presented in Table 2-6	2-42
2-17	Mean, Minimum, and Maximum Daily Flows under Alternative D in an 8.23-maf Year Based on Values Presented in Table 2-8	2-45
2-18	Simulated Hourly Flows under Alternative D in an 8.23-maf Year	2-47
2-19	Simulated Hourly Flows under Alternative D for a Week in July in an 8.23-maf Year Showing Typically Lower Weekend Flows	2-47
2-20	Decision Tree for Implementation of Sediment-Related Experimental Treatments under Alternative D	2-55
2-21	Decision Tree for Implementation of Aquatic Resource-Related Experimental Treatments under Alternative D	2-56
2-22	Mean, Minimum, and Maximum Daily Flows under Triggered Low Summer Flows of Alternative D in an 8.23-maf Year Based on the Values Presented in Table 2-10	2-69
2-23	Mean, Minimum, and Maximum Daily Flows under Alternative E in an 8.23-maf Year Based on the Values Presented in Table 2-11	2-74
2-24	Simulated Hourly Flows under Alternative E in an 8.23-maf Year	2-74
2-25	Simulated Hourly Flows under Alternative E for a Week in July in an 8.23-maf Year Showing Typically Lower Weekend Flows	2-75
2-26	Decision Tree for Sediment-Related Actions under Alternative E	2-76
2-27	Decision Tree for Trout-Related Actions under Alternative E	2-77
2-28	Mean, Minimum, and Maximum Daily Flows under Base Operations of Alternative F in an 8.23-maf Year Based on the Values Presented in Table 2-12	2-82

2-29	Mean, Minimum, and Maximum Daily Flows under Alternative G in an 8.23-maf Year Based on Values Presented in Table 2-13	2-83
3.1-1	LTEMP Project Area and Surrounding Lands	3-2
3.2-1	Map of Lake Powell and Associated Major Tributaries	3-8
3.2-2	Pattern of Annual Historic Flows at Lees Ferry	3-12
3.2-3	Map of Lake Mead and Associated Major Tributaries	3-13
3.2-4	Profile of Lake Powell from Glen Canyon Dam to the Inflow of the Colorado River	3-16
3.2-5	Water Temperature at Lees Ferry	3-18
3.2-6	Water Temperatures at Four Stations along the Colorado River from Lees Ferry to Diamond Creek, 1995–2014	3-21
3.2-7	Mid-June Warming above Release Temperatures Measured at Diamond Creek, 1994–2004, as a Function of Mean Weekly Discharge	3-22
3.3-1	Geomorphic Features of the Colorado River	3-36
3.3-2	Schematic Diagram of the Fan-Eddy Complex on the Colorado River	3-38
3.3-3	River Cross Section Depicting Sediment Entrapment and Sandbar Building	3-38
3.3-4	Aeolian and Fluvial Sand Deposits along the Colorado River	3-40
3.3-5	Debris Fans and Variation in Water-Surface Elevation and Channel Width for Colorado River Flows below Glen Canyon Dam	3-42
3.3-6	Comparison of Sandbars Used as Campsites, based on Inventories Conducted in 1973, 1983, and 1991	3-42
3.3-7	Repeated Photography Illustrating Sediment Losses and Sandbar Changes along the Colorado River	3-44
3.3-8	Annual Sediment Contributions from the Paria and Little Colorado Rivers	3-49
3.3-9	Matched Photographs of RM 172 Illustrating Positive Depositional Response to the 2008 HFE	3-52

3.3-10	Conceptual Diagram of the Dependency between Net Sandbar Size, Duration and Frequency of HFEs, and Post-HFE Erosion Rates	3-54
3.3-11	Average Campsite Area above the References Stage: before, after, and 6 Months following the 2008 HFE	3-55
3.3-12	Longitudinal Profiles of the Mainstem Colorado Riverbed Upstream of the Hoover Dam in 1935, 1948, 1963, and 2001	3-57
3.5-1	Temperature Ranges for Spawning, Egg Incubation, and Growth by Native and Nonnative Fishes of the Colorado River System below Glen Canyon Dam	3-71
3.5-2	Humpback Chub Aggregation Areas along the Colorado River between Glen Canyon Dam and Lake Mead	3-74
3.5-3	Estimated Adult Humpback Chub Abundance from Age-Structured Mark- Recapture Model Incorporating Uncertainty in Assignment of Age	3-75
3.5-4	Estimated Total Adult Abundance of Humpback Chub in the Lower 8 mi of the Little Colorado River and a 2-mi Portion of the Colorado River Just Downriver of the Confluence of the Little Colorado and Colorado Rivers, for September, 2009 through 2012	3-76
3.5-5	Water Temperatures at Lees Ferry and the Little Colorado River Confluence, 1995 to Present	3-81
3.5-6	Mean Electrofishing Catch Rates of Rainbow Trout in the Glen Canyon Reach, 1991–2013	3-96
3.5-7	Mean Electrofishing Catch Rates of Brown Trout in the Colorado River between Lees Ferry and Lake Mead, 2000–2009	3-99
3.6-1	Riparian Vegetation Zones along the Colorado River below Glen Canyon Dam	3-114
3.7-1	Riparian Zones Used by Nesting Birds	3-124
3.7-2	Threatened, Endangered, and Sensitive Species Observed along the Colorado River Corridor	3-135
3.8-1	Spencer Steamboat	3-148
3.8-2	A Roasting Pit Feature in a Grand Canyon Dune	3-148

3.8-3	An Archaic Period Site on the Colorado River in GCNP	3-151
3.8-4	Glen Canyon Linear Style Petroglyph in GCNRA	3-151
3.8-5	Puebloan Era Architecture along the Colorado River in GCNP	3-152
3.8-6	Lees Ferry and Lonely Dell Ranch Historic District Located in GCNRA	3-155
3.10-1	Glen Canyon Reach Rainbow Trout	3-170
3.10-2	Mean Rainbow Trout Catch Per Unit Effort of Both Boat Anglers and Shore- Line Anglers from Creel Surveys at Lees Ferry	3-171
3.10-3	Fishing User Days by Month in the Glen Canyon Reach for 2006 and 2009	3-172
3.10-4	Angler Days in the Glen Canyon Reach from 1965 through 2011	3-173
3.10-5	Designated Campsite Areas in the Glen Canyon Reach	3-175
3.10-6	Shoreline Environment with Steep Erosion Banks at Glen Canyon Reach Ferry Swale Campsite	3-176
3.10-7	Pontoon Raft Operated by Colorado River Discovery	3-176
3.10-8	Boating in Grand Canyon, Anticipated Annual Use by Month	3-179
3.10-9	Change in Camp Size over Time in the Lees Ferry to Diamond Creek Reach of GCNP	3-181
3.10-10	Total High-Elevation Campsite Area for Each Survey between 1998 and 2006	3-182
3.10-11	High-Elevation Campsite Area in Critical and Noncritical Reaches between 1998 and 2006.	3-182
3.12-1	Glen Canyon Viewed from the Colorado River	3-194
3.12-2	Horseshoe Bend	3-194
3.12-3	Typical View of the Colorado River and Grand Canyon Afforded Recreationists on a River Trip	3-196
3.12-4	Colorado River and Granaries at Nankoweap	3-196

3.12-5	Entrance to Havasu Canyon	3-197
3.12-6	Vasey's Paradise	3-197
3.14-1	Minority Population Groups in the 11-County Area	3-219
3.14-2	Low-Income Population Groups in the 11-County Area	3-220
3.15-1	Nonattainment Areas for SO ₂ , 8-Hour O ₃ , PM _{2.5} , PM ₁₀ , and Pb in the 11-State Area	3-233
3.15-2	PSD Class I Areas in the 11-State Affected Area	3-235
3.16-1	Historical Supply and Use and Projected Future Colorado River Basin Water Supply and Demand	3-238
4-1	Integrated Multiple-Resource Modeling Framework Showing Inputs, Intermediate Calculations, and Output	4-3
4-2	Example Box-and-Whisker Plot for Alternatives and Their Resource Metric Values	4-6
4.2-1	Monthly Releases under Each Alternative in Years with Different Annual Release Volumes	4-22
4.2-2	Mean Monthly Volume under the LTEMP Alternatives Showing the Mean, Median, 75th Percentile, 25th Percentile, Minimum, and Maximum Values for 21 Hydrology Scenarios and Three Sediment Scenarios	4-24
4.2-3	Mean Daily Flows by Month under the LTEMP Alternatives Showing the Mean, Median, 75th Percentile, 25th Percentile, Minimum, and Maximum Values for 21 Hydrology Scenarios and Three Sediment Scenarios	4-27
4.2-4	Mean Daily Change in Flows by Month under the LTEMP Alternatives Showing the Mean, Median, 75th Percentile, 25th Percentile, Minimum, and Maximum Values for 21 Hydrology Scenarios and Three Sediment Scenarios	4-29
4.2-5	Lake Powell and Lake Mead End of Calendar Year Pool Elevation for 21 Hydrology Traces and Seven Alternatives	4-32
4.2-6	Percentage of Traces below Lake Powell's Minimum Power Pool and Percentage of Traces with a Lower Basin Shortage for 21 Hydrology Traces and Seven Alternatives	4-33

4.2-7	Percentage of Time in Different Operating Tier than Alternative	4-34
4.2-8	Frequency of Lake Powell Operating Tiers from 2014 to 2026 under Each of the Alternatives for 21 Hydrologic Traces	4-35
4.2-9	Frequency of Lake Powell Operating Tiers from 2027 to 2033 under Each of the Alternatives for 21 Hydrologic Traces	4-36
4.2-10	Frequency of Occurrence of Modeled Annual Releases Extending Beyond the Water Year per 20-Year Trace for Each of the Alternatives	4-37
4.2-11	Comparison of Mean Water Temperatures for Representative Wetter, Moderate, and Drier Hydrology Traces for Glen Canyon Dam Releases	4-39
4.2-12	Seasonal Glen Canyon Dam Release Temperatures for LTEMP Alternatives	4-40
4.2-13	Seasonal Temperature Trends under the Seven LTEMP Alternatives	4-42
4.2-14	Projected Mean Salinity Concentrations under the LTEMP Alternatives at Lees Ferry	4-46
4.3-1	Conceptual Depiction of the Sand Load Index	4-67
4.3-2	Conceptual Depiction of the Sand Mass Balance Model	4-68
4.3-3	Variation in Lake Powell Pool Elevation Relative to Full for 21 Hydrology Traces and Seven Alternatives	4-75
4.3-4	Variation in Lake Mead Pool Elevation Relative to Full for 21 Hydrology Traces and Seven Alternatives	4-76
4.3-5	Number and Type of HFEs Expected to Occur during the 20-Year LTEMP Period under the Seven Alternatives	4-77
4.3-6	Sand Load Index Values for the 20-Year LTEMP Period under the Seven Alternatives	4-78
4.3-7	Sand Mass Balance Index Values for the 20-Year LTEMP Period under the Seven Alternatives	4-79
4.4-1	Anticipated Relationships among Dam Releases, Physical Conditions, Habitats, and Ecological Resources in the Colorado River Ecosystem	4-89

4.5-1	Modeled Average Population Size of Age-1 and Older Rainbow Trout in the Glen Canyon Reach during the 20-Year LTEMP Period under the LTEMP Alternatives Showing the Mean, Median, 75th Percentile, 25th Percentile, Minimum, and Maximum Values for 21 Hydrology Scenarios	4-121
4.5-2	Modeled Annual Average Number of Rainbow Trout Emigrating into the Marble Canyon Reach from the Glen Canyon Reach during the 20-Year LTEMP Period under the LTEMP Alternatives	4-123
4.5-3	Modeled Mean Annual Number of Rainbow Trout in the Glen Canyon Reach Exceeding 16 in. Total Length during 20-Year Simulation Periods under the LTEMP Alternatives	4-124
4.5-4	Modeled Mean Annual Temperature Suitability for Rainbow and Brown Trout under LTEMP Alternatives at Four Locations Downstream of Glen Canyon Dam	4-125
4.5-5	Modeled Mean Annual Temperature Suitability for Warmwater Nonnative Fish under LTEMP Alternatives at Four Locations Downstream of Glen Canyon Dam	4-127
4.5-6	Mean Annual Mainstem Temperature Suitability for Humpback Chub under LTEMP Alternatives at Reported Aggregation Locations and Combined Temperature Suitability for RM 157 and RM 213 Locations	4-129
4.5-7	Mean Modeled Total Length Attained by December 31 for YOY Humpback Chub Based on Predicted Mainstem Water Temperatures at the Little Colorado River Confluence and at Pumpkin Spring under Each Alternative	4-130
4.5-8	Modeled Minimum Population Size for Humpback Chub during the 20-Year LTEMP Period under LTEMP Alternatives	4-131
4.5-9	Modeled Mean Annual Temperature Suitability for Native Fish under LTEMP Alternatives at Four Locations Downstream of Glen Canyon Dam	4-135
4.5-10	Overall Modeled Mean Annual Temperature Suitability under LTEMP Alternatives for Aquatic Fish Parasites at Four Locations Downstream of Glen Canyon Dam	4-136
4.6-1	Dominant Factors Affecting Riparian Plant Communities below Glen Canyon Dam	4-177

4.6-2	Comparison among Alternatives for Four Riparian Vegetation Metrics as Predicted by a Vegetation Model	4-178
4.6-3	Comparison among Alternatives for Combined Riparian Vegetation Metrics as Predicted by a Vegetation Model	4-179
4.6-4	Comparison among Alternatives for Wetland Cover as Predicted by a Vegetation Model	4-187
4.8-1	Number of Days per Year Flows Would Be >23,200 cfs under LTEMP Alternatives	4-243
4.8-2	Wind Transport of Sediment Index Values for LTEMP Alternatives	4-245
4.10-1	Recreation, Visitor Use, and Experience Metric Results for LTEMP Alternatives	4-288
4.13-1	Flow Diagram of the Power Systems Methodology Used in the LTEMP EIS	4-326
4.13-2	Average Daily Glen Canyon Dam Generation under Representative Hydrological Conditions under LTEMP Alternatives	4-341
4.13-3	Total NPV of Hoover Powerplant Energy over a 20-Year Period under LTEMP Alternatives	4-352
4.15-1	Annual Power Generation by Alternative over the 20-Year LTEMP Period	4-407
4.16-1	Weights Used To Reflect the Expected Frequency of Hydrologic Conditions under Climate Change	4-415
4.16-2	Mean Annual Inflow Showing the Mean, Median, 75th Percentile, 25th Percentile, Minimum, and Maximum Values for 112 Climate-Change Inflow Traces and 21 Historic Inflow Traces	4-416
4.16-3	Differences between Historic and Climate-Change-Weighted Values of Mean Daily Flow and Mean Daily Change in Flow by Month for LTEMP Alternatives	4-420
D-1	Operating Tiers as Specified by the 2007 Interim Guidelines for the Operations of Lake Powell and Lake Mead	D-7
D-2	Monthly Releases for Each Alternative in an 8.23-maf Release Year	D-21
D-3	Locations of CRSS 29 Natural Flow Nodes	D-24
------	--	------
D-4	Comparison of CRSS Results Generated Using 105 Traces and 21 Traces for Lake Powell Annual Inflow, Lake Powell Water-Year Release Volume, and Lake Powell Monthly Release Volume	D-25
D-5	Comparison of CRSS Results Generated Using 105 Traces and 21 Traces for Lake Powell End-of-December Water Elevations at the 10th, 50th, and 90th Percentiles	D-25
D-6	Occurrences of Low Summer Flows in Applicable Alternatives	D-28
D-7	Frequency of Occurrence of Modeled Annual Releases Extending beyond the End of the Water Year per 20-yr Trace for Each of the Alternatives	D-29
D-8	Median Volume of Modeled Annual Releases Extending beyond the End of the Water-Year Releases by Trace for Each of the Alternatives	D-30
D-9	Lake Powell and Lake Mead End-of-December Pool Elevation for 21 Hydrology Traces under Alternatives A and B	D-30
D-10	Lake Powell and Lake Mead End-of-December Pool Elevation for 21 Hydrology Traces under Alternatives A and C	D-31
D-11	Lake Powell and Lake Mead End-of-December Pool Elevation for 21 Hydrology Traces under Alternatives A and D	D-31
D-12	Lake Powell and Lake Mead End-of-December Pool Elevation for 21 Hydrology Traces under Alternatives A and E	D-32
D-13	Lake Powell and Lake Mead End-of-December Pool Elevation for 21 Hydrology Traces under Alternatives A and F	D-32
D-14	Lake Powell and Lake Mead End-of-December Pool Elevation for 21 Hydrology Traces under Alternatives A and G	D-33
D-15	Percentage of Traces below Lake Powell's Minimum Power Pool and Percentage of Traces with a Lower Basin Shortage for 21 Hydrology Traces under Alternatives A and B.	D-33
D-16	Percentage of Traces below Lake Powell's Minimum Power Pool and Percentage of Traces with a Lower Basin Shortage for 21 Hydrology Traces under Alternatives A and C	D-34

D-17	Percentage of Traces below Lake Powell's Minimum Power and Percentage of Traces with a Lower Basin Shortage for 21 Hydrology Traces under Alternatives A and D	D-34
D-18	Percentage of Traces below Lake Powell's Minimum Power Pool and Percentage of Traces with a Lower Basin Shortage for 21 Hydrology Traces under Alternatives A and E	D-35
D-19	Percentage of Traces below Lake Powell's Minimum Power Pool and Percentage of Traces with a Lower Basin Shortage for 21 Hydrology Traces under Alternatives A and F	D-35
D-20	Percentage of Traces below Lake Powell's Minimum Power Pool and Percentage of Traces with a Lower Basin Shortage for 21 Hydrology Traces under Alternatives A and G	D-36
E-1	Comparison of Calendar, Water, and Sediment Years	E-20
E-2	Model Flow Diagram for Analyses Showing Inputs, Intermediate Calculations, and Output	E-21
E-3	Conceptual Schematic of the Sand Budget Model	E-22
E-4	Historical Paria Sediment Load per Accounting Period and the 20.5-year Load for the Trace That Begins in Each Fall Accounting Period	E-22
E-5	Sediment Traces s1, s2, and s3 for the Paria River Used in the Modeling to Account for Uncertainty in Future Delivery	E-23
E-6	Little Colorado River Sediment Traces for s1, s2, and s3 Used in the Modeling to Account for Uncertainty in Future Delivery	E-23
E-7	Conceptual Representation of the Sand Mass Balance Index	E-24
E-8	Average Sediment and Hydrology Triggered HFE Count by Type for Each Long-Term Strategy	E-24
E-9	Average HFE Count for Sediment Traces s1, s2, and s3 for Each Long-Term Strategy	E-25
E-10	Definition of the Statistics Represented by the Box and Whisker Plots Used in This Analysis	E-25

E-11	Sand Load Index Statistics from 63 Simulations for Each Long-Term Strategy	E-26
E-12	Standard Deviation of High Flows Statistics from 63 Simulations for Each Long-Term Strategy	E-26
E-13	Correlation between SDHF and SLI	E-27
E-14	Sand Mass Balance Index Statistics from 63 Simulations for Each Long-Term Strategy	E-27
E-15	Correlation between SMBI and SLI	E-28
E-16	Sand Load Index for Long-Term Strategies Using Climate Change Weights	E-28
E-17	Comparison of the Sand Load Index between Climate Change and Historical Weights	E-29
E-18	Standard Deviation of High Flows Using Climate Change Weights	E-29
E-19	Comparison of the Standard Deviation of High Flows between Climate Change and Historical Weights	E-30
E-20	Sand Mass Balance Index Using Climate Change Weights	E-30
E-21	Comparison of the Sand Mass Balance Index between Climate Change and Historical Weights	E-31
E-22	Comparison of Long-Term Strategies C1 and C2 for Hydrology Trace 1, Sediment Trace 3	E-31
E-23	SMBI for Alternative E Plotted against Alternative C	E-32
E-24	SLI for Alternative E Plotted against Alternative C	E-32
E-25	Load-Following Curtailment Effects on SLI and SMBI	E-33
E-26	Low Summer Flows for WY 2014, Hydrology Trace 1, Sediment Trace 1	E-34
E-27	Hydrology Impacts on the Sand Load Index	E-35
E-28	Hydrology Impacts on the Sand Mass Balance Index	E-35
E-29	Conceptual Diagram of Water Surface Elevation Affecting Delta Location	E-36

E-30	Historical Cumulative Sand Load Leaving Marble Canyon and Reaching the Gage above Diamond Creek	E-36
E-31	Hydrology Impacts of Lake Powell Pool Elevations by Month across Alternatives	E-37
E-32	Hydrology Impacts of Lake Mead Pool Elevations by Month across Alternatives	E-37
F-1	Fit of Regressions Predicting the Log of Recruitment of Rainbow Trout in the Glen Canyon Reach Estimated by the Korman et al. Stock Synthesis Model as a Function of the Annual Release Volume from Glen Canyon Dam, the Range of Mean Daily Flows during May–August, and the Maximum Flow Each Year	F-31
F-2	Relationship between Annual Estimates of the Asymptotic Length of Rainbow Trout in Lees Ferry Predicted by the Stock Synthesis Model as a Function of the Estimated Abundance for Fish >150 mm Each Year	F-32
F-3	The Relationship between Annual Recruitment of Rainbow Trout in Lees Ferry Estimated by the Korman et al. Stock Synthesis Model and the Number of Trout That Emigrate from Lees Ferry into Marble Canyon the Following Year	F-35
F-4	Fit of the Glen Canyon Rainbow Trout Simulation Model to Predictions of Recruitment, Asymptotic Length, and the Number of Out-migrants Predicted by the Korman et al. Stock Synthesis Model	F-37
F-5	Predicted and Observed Annual Catch of Rainbow Trout by Year and River Reach	F - 41
F-6	Best-Fitting Distributions Describing Monthly Movement of Rainbow Trout in Marble Canyon Assuming Either a Normal or Cauchy Distribution	F-43
F-7	Visual Summary of Humpback Chub Population Model Structure	F-44
F-8	Modeled Effects of Trout Abundance and Temperature on Humpback Chub Survival and Growth	F-48
F-9	Simulated Adult Abundances from Backcasted Model Compared to Patterns Reported in Coggins and Walters	F-51
F-10	Modeled Average Population Size of Age-1 and Older Rainbow Trout in the Glen Canyon Reach during the 20-year LTEMP Period under LTEMP Alternatives and Long-Term Strategies	F-54

F-11	Modeled Mean Annual Number of Rainbow Trout in the Glen Canyon Reach Exceeding 16 in. Total Length during the 20-year LTEMP Period under the LTEMP Alternatives and Long-Term Strategies	F-55
F-12	Relationship between Modeled Mean Rainbow Trout Abundance in the Glen Canyon Reach and the Mean Number of Rainbow Trout Exceeding 16 in. Total Length during the 20-year LTEMP Period under the LTEMP Alternatives and Long-Term Strategies	F-56
F-13	Modeled Mean Annual Angler Catch Rate for Rainbow Trout in the Glen Canyon Reach during the 20-year LTEMP Period under the LTEMP Alternatives and Long-Term Strategies	F-57
F-14	Relationship between Modeled Mean Rainbow Trout Abundance in the Glen Canyon Reach and Mean Angler Catch Rates during the 20-year LTEMP Period under the LTEMP Alternatives and Long-Term Strategies	F-58
F-15	Modeled Annual Average Number of Rainbow Trout Emigrating into the Marble Canyon Reach from the Glen Canyon Reach during the 20-year LTEMP Period under the LTEMP Alternatives and Long-Term Strategies	F-59
F-16	Modeled Frequency of Triggered Mechanical Removal for Rainbow Trout in the Little Colorado River Reach during the 20-year LTEMP Period under the LTEMP Alternatives and Long-Term Strategies	F-61
F-17	Modeled Minimum Population Size for Humpback Chub during the 20-year LTEMP Period under the LTEMP Alternatives and Long-Term Strategies	F-62
F-18	Relationship between Modeled Mean Numbers of Rainbow Trout Out-migrants from the Glen Canyon Reach and the Modeled Mean Minimum Abundance of Adult Humpback Chub during the 20-year LTEMP Period under the LTEMP Alternatives and Long-Term Strategies	F-63
F-19	Suitability for Spawning, Egg Incubation, and Growth of Humpback Chub as a Function of Water Temperature	F -6 8
F-20	Months for Which Annual Temperature Suitability for Specific Life History Aspects of Humpback Chub Were Calculated	F-68
F-21	Modeled Historic Water Temperatures in the Colorado River at Humpback Chub Aggregation Locations, Water Years 1990–2012	F-70

F-22	Output from the Temperature Suitability Model for Humpback Chub Aggregation Locations Based on Modeled Water Temperatures for Water Years 1990–2012	F-71
F-23	Mainstem Temperature Suitability for Humpback Chub Aggregation Locations under LTEMP Alternatives and Long-Term Strategies	F-72
F-24	Suitability of Water Temperatures for Spawning, Egg Incubation, and Growth of Native Fish Species	F-74
F-25	Months for Which Temperature Suitability for Specific Life History Aspects Were Considered for Native Fish Species	F-74
F-26	Modeled Historic Water Temperatures in the Colorado River Downstream of Glen Canyon Dam, Water Years 1990–2012	F-75
F-27	Annual Temperature Suitability Scores for Growth, Spawning, and Egg Incubation of Native Fish Species at RM 225 Based on Modeled Water Temperatures for Water Years 1990–2012	F-76
F-28	Annual Temperature Suitability Scores for Native Fish by Assessment Location Based on Modeled Water Temperatures for Water Years 1990–2012	F-77
F-29	Mean Annual Overall Temperature Suitability for Native Fish by Assessment Location Based on Modeled Water Temperatures for Water Years 1990–2012	F-77
F-30	Mean Annual Mainstem Temperature Suitability for Native Fish under LTEMP Alternatives and Long-Term Strategies at RM 61, RM 157, and RM 225, and Overall Mean for RM 61–225	F-78
F-31	Suitability of Water Temperatures for Spawning, Egg Incubation, and Growth of Nonnative Fish Species	F-80
F-32	Months during Which Temperature Suitability for Specific Life History Aspects Were Calculated for Nonnative Fish Species	F-81
F-33	Annual Temperature Suitability Scores for Spawning, Incubation, and Growth of Nonnative Fish Species at RM 225 Based on Modeled Temperatures for Water Years 1990 to 2012	F-82
F-34	Mean Annual Temperature Suitability Scores for Nonnative Fish Species and for Temperature Groups by River Location Based on Modeled Water Temperatures for Water Years 1990–2012	F-83

F-35	Mean Annual Overall Temperature Suitability Scores for Coldwater and Warmwater Nonnative Fish Species Groups Based on Modeled Historic Temperatures for Water Years 1990–2012	F-83
F-36	Mean Annual Mainstem Temperature Suitability for Coldwater Nonnative Fish under LTEMP Alternatives and Long-Term Strategies at RM 15, RM 61, RS 157, and RM 225	F-84
F-37	Mean Annual Mainstem Temperature Suitability for Warmwater Nonnative Fish under LTEMP Alternatives and Long-Term Strategies at RM 0, RM 61, RM 157, and RM 225	F-85
F-38	Suitability of Various Water Temperatures for Host Activity and Infestation Rates of Parasite Species	F-86
F-39	Annual Temperature Suitability Scores for Parasite Species at RM 225 Based on Modeled Water Temperatures for Water Years 1990–2012	F-87
F-40	Mean Annual Temperature Suitability Scores for Parasite Species by River Location Based on Modeled Water Temperatures for Water Years 1990–2012	F-87
F-41	Overall Means of Annual Suitability Scores for Parasite Species across All River Locations during the 1990–2012 Water Years	F-88
F-42	Overall Modeled Mean Annual Temperature Suitability under LTEMP Alternatives and Long-Term Strategies for Aquatic Fish Parasites at Four Locations Downstream of Glen Canyon Dam	F-89
G-1	Native Cover Metric for the LTEMP Alternatives and Associated Long-Term Strategies	G-46
G-2	Native Cover Metric under Climate Change for the LTEMP Alternatives and Associated Long-Term Strategies	G-46
G-3	Native Diversity Metric for the LTEMP Alternatives and Associated Long- Term Strategies	G-47
G-4	Native Diversity Metric under Climate Change for the LTEMP Alternatives and Associated Long-Term Strategies	G-47
G-5	Native/Nonnative Ratio Metric for the LTEMP Alternatives and Associated Long-Term Strategies	G-48

G-6	Native/Nonnative Ratio Metric under Climate Change for the LTEMP Alternatives and Associated Long-Term	G-48
G-7	Arrowweed Metric for the LTEMP Alternatives and Associated Long-Term Strategies; Higher Values Indicate Less	G-49
G-8	Arrowweed Metric under Climate Change for the LTEMP Alternatives and Associated Long-Term Strategies; Higher Values Indicate Less Arrowweed	G-49
G-9	Overall Combined Score for the LTEMP Alternatives and Associated Long- Term Strategies	G-50
G-10	Overall Combined Score under Climate Change for the LTEMP Alternatives and Associated Long-Term Strategies	G-50
G-11	Relative Change in Wetland Cover for the LTEMP Alternatives and Associated Long-Term Strategies	G-51
G-12	Relative Change in Wetland Cover under Climate Change for the LTEMP Alternatives and Associated Long-Term Strategies	G-51
H-1	Wind Transport of Sediment Index Values for the LTEMP Alternatives and Associated Long-Term Strategies	Н-6
H-2	Average Number of HFEs in the 20-Year LTEMP Period	H - 7
Н-3	Daily Maximum Discharge in a Typical 8.23-maf Water Volume Release Year from the Glen Canyon Dam during the Windy Season of March–June	H-8
H-4	Number of Days per Year Flows Would Be >23,200 cfs under LTEMP Alternatives and Long-Term Strategies	H-10
H-5	Average Number of Days of an HFE Event per Year	H-11
H-6	Modeled Glen Canyon Dam Discharge for the Same Year	H-12
H-7	Time Off River Index Values for All LTEMP Alternatives and Associated Long-Term Strategies	H-14
H-8	Daily Average Discharge for Representative Long-Term LTEMP Strategies	H-15
I-1	Riparian Diversity for the LTEMP Alternatives and Associated Long-Term Strategies	I-6

I-2	Riparian Diversity under Climate Change Assumptions for the LTEMP Alternatives and Associated Long-Term Strategies	I-7
I-3	Wetland Abundance for the LTEMP Alternatives and Associated Long-Term Strategies	I-9
I-4	Wetland Abundance under Climate Change Assumptions for the LTEMP Alternatives and Associated Long-Term Strategies	I-10
I-5	Frequency of Trout Management Flows for the LTEMP Alternatives and Associated Long-Term Strategies	I-12
I-6	Frequency of Trout Management Flows under Climate Change Assumptions for the LTEMP Alternatives and Associated Long-Term Strategies	I-13
I-7	Frequency of Mechanical Removal for the LTEMP Alternatives and Associated Long-Term Strategies	I-14
I-8	Frequency of Mechanical Removal under Climate Change Assumptions for the LTEMP Alternatives and Associated Long-Term Strategies	I-15
I-9	Lake Powell Water Levels for the LTEMP Alternatives and Associated Long- Term Strategies	I-16
I-10	Lake Powell Water Levels under Climate Change Assumptions for the LTEMP Alternatives and Associated Long-Term Strategies	I-17
J- 1	Camping Area Index for LTEMP Long-Term Strategies	J-8
J-2	Sand Load Index for LTEMP Long-Term Strategies	J-9
J-3	Navigational Risk Index Values for the LTEMP Long-Term Strategies	J-11
J-4	Fluctuation Index for LTEMP Long-Term Strategies	J-14
J-5	Time Off River Index for LTEMP Long-Term Strategies	J-16
J-6	Average Daily Discharge for All Modeled Traces and Years under LTEMP Alternatives	J-16
J-7	Glen Canyon Rafting Metric for All LTEMP Long-Term Strategies	J-19

J-8	Average Number of HFEs in the 20-Year LTEMP Period for LTEMP Long- Term Strategies	J-20
J-9	Glen Canyon Inundation Metric for All LTEMP Long-Term Strategies	J-22
J-10	Percentage of Traces Lake Powell Elevation Equal to or below 3,580 ft AMSL for the Summer Season	J-24
J-11	Percentage of Traces Lake Powell Elevation Equal to or below 3,580 ft AMSL for the Fall and Spring Seasons	J-24
J-12	Percentage of Traces Lake Mead Elevation Equal to or below 1,050 ft AMSL for the Summer Season	J-25
J-13	Percentage of Traces Lake Mead Elevation Equal to or below 1,050 ft AMSL for the Fall and Spring Seasons	J-25
K.1-1	Network Topology Used by the AURORA Western Interconnection Model	K-10
K.1-2	Percentage of Total Generation Resources Owned by Individual Customers	K-12
K.1-3	Simplified Network Topology of the SLCA/IP Market System	K-13
K.1-4	Percentage of SLCA/IP Federal Hydropower Nameplate Capacity by Facility	K-16
K.1-5	Flow Diagram of the Power Systems Methodology Used in the LTEMP EIS	K-18
K.1-6	Illustration of a Typical GTMax-Lite Result for a 24-Hour Period	K-24
K.1-7	GTMax-Lite Network Topology for the Large SLCA/IP Hydropower Resources Other Than Glen Canyon Dam	K-26
K.1-8	Illustration of Temporal Modeling Method Used in the GTMax-Lite Five Large SLCA/IP Plant Configuration	K-28
K.1-9	Example of the Load Scaling Algorithm LDC	K-31
K.1-10	Example of the Load Scaling Algorithm Chronological Hourly Loads	K-32
K.1-11	Projected Annual Average Calibrated AURORA LMPs at the Palo Verde Market Hub	K-36

K.1-12	Historical Palo Verde On- and Off-Peak Electricity Prices Compared to Natural Gas Prices	K-36
K.1-13	2014 AEO Projected Delivered Utility Natural Gas Prices	K-37
K.1-14	Comparison of Modeled and Actual Annual Aggregated Generation Levels for Natural Gas, Coal, and Nuclear Powerplants under the No Action Alternative	K-43
K.1-15	Total Non-coincidental Peak Loads for the SLCA/IP Market System	K-49
K.1 - 16	Total Monthly Loads for the SLCA/IP Market System	K-49
K.1-17	Projected 2014 AEO Natural Gas Prices at the Henry Hub	K-56
K.1-18	2014 AEO Electricity Market Module Regions and the Mapping of States to Regions	K-57
K.1-19	Projected AEO 2014 Coal Prices by Electricity Market Module Region	K-59
K.1-20	WAPA SLCA/IP Hydropower Powerplants	K-63
K.1-21	SLCA/IP Federal Hydropower Capacity Uses and Variability Factors	K-64
K.1-22	Reductions in Operating Range when Providing Ancillary Services	K-66
K.1-23	Illustration of the Trapezoidal Method Used to Compute Capacity at Flaming Gorge Dam	K-72
K.1-24	Illustration of Ramping Time Increase as a Function of Increasing Water Volumes at Flaming Gorge Dam	K-73
K.1-25	Capacity Values at Flaming Gorge Dam Calculated over a Wide Range of Daily Water Releases Using the Trapezoidal Method	K-74
K.1-26	Average Historical Non-coincidental Annual Peak Loads for the Eight Large Customers and Percentages Relative to the Total	K-75
K.1-27	Historical Coincidental Annual and Average Peak Loads for the Eight Large Customers during the CY 2006 through CY 2009 Time Period	K-76
K.1-28	Historical Coincidental Annual and Average Peak Loads for Six Large Customers during the CY 2010 through CY 2012 Time Period	K-77

K.1-29	SLCA/IP Models Used for Estimating SHP Capacity	K-78
K.1-30	Historical SHP Capacity Obligation and WAPA Estimated Risk Level in July	K-80
K.1-31	Historical SHP Capacity Obligation and WAPA Estimated Risk Level in August	K-80
K.1-32	Comparison of Firm Marketable Capacity Determinations across Alternatives, Exceedance Levels, and Summer Peak Months	K-81
K.1-33	Timing of Capacity Additions in AURORA	K-83
K.1-34	Annual Average Hydropower Generation for the Glen Canyon Dam Powerplant and the Aggregate Generation for All Other SLCA/IP Resources	K-88
K.1-35	Average Daily Glen Canyon Dam Powerplant Hydropower Generations by Month Based on the Average of All 21 CRSS/SBM Hydrology Traces	K-89
K.1-36	Average Annual Glen Canyon Dam Powerplant Generation for the Representative CRSS/SBM Hydrology Trace	K-92
K.1-37	Representative Trace Average Daily Glen Canyon Dam Powerplant Generations by Month	K-93
K.1-38	Average Daily Generation by Month for Each Alternative	K-97
K.1-39	Palo Verde Average Daily Electricity Market Price Statistics by Month during the 20-Year LTEMP Period	K-98
K.1-40	Typical Hourly Winter/Summer Price Patterns in the Western United States	K-99
K.1-41	Average Number of HFEs by Alternative for All 21 Hydrology Traces of Sediment Trace 2	K-100
K.1-42	Average Amount of Non-Power Water Releases by Alternative for All 21 Hydrology Traces of Sediment Trace 2	K-100
K.1-43	Average Annual Number of Hours the Lake Powell Elevation Is below the Penstock Intake by Alternative for All 21 Hydrology Traces of Sediment Trace 2	K-101
K.1-44	2014 Annual Energy Outlook Projections of Capacity Additions in the Western Interconnection over the LTEMP Period	K-104

K.1-45	Cumulative Capacity Additions for Alternative A	K-106
K.1-46	Cost Difference of Alternatives Compared to Alternative A at 50% Exceedance Level and 3.375% Discount Rate	K-119
K.1-47	Cost Difference of Alternatives Compared to Alternative A at 90% Exceedance Level and 3.375% Discount Rate	K-119
K.1-48	Cost Difference of Alternatives Compared to Alternative A at 99% Exceedance Level and 3.375% Discount Rate	K-120
K.1-49	Cost Difference of Alternatives Compared to Alternative A at 90% Exceedance Level and 1.4% Discount Rate	K-122
K.1-50	Cost Difference of Alternatives Compared to Alternative A at 90% Exceedance Level and 3.375% Discount Rate, Assuming All Alternative A New Capacity Additions Are Advanced Combustion Turbines	K-123
K.1-51	Cost Difference of Alternatives Compared to Alternative A at 90% Exceedance Level and 3.375% Discount Rate, Assuming All Alternative A New Capacity Additions Are Advanced Combined Cycle	K-124
K.1-52	Cost Difference of Alternatives Compared to Alternative A at 90% Exceedance Level and 3.375% Discount Rate Assuming Average Hydropower Conditions	K-125
K.2-1	Determination of Energy Hourly Expense or Revenue	K-140
K.2-2	Determination of Total Net Expense or Revenue by Year for Each Alternative	K-142
K.3-1	Flowchart Diagram of Rate Impact Analysis Process	K-156
K.3-2	Projected Carrying Charge Rates Used in Rate Impact Analysis	K-161
K.3-3	Flowchart Diagram of Allocation Process	K-162
K.3-4	Load Factors of SLCA/IP Power	K-165
K.3-5	Scatter Plot of Preference Ratio and Annual Retail Sales	K-169
K.3-6	Scatter Plots of Percent Retail Rate Change and the Preference Ratio	K-171

K.3-8	Monthly Residential Bill Distribution under Alternative A	K-173
K.3-9	Average Retail Rate Impacts under LTEMP Alternatives Relative to Alternative A	K-175
K.3-10	Average Residential Bill Impacts under LTEMP Alternatives Relative to Alternative A	K-175
K.3-11	Average Retail Percent Revenue Increase Relative to Alternative A	K-176
K.3-12	Average Monthly Residential Bill Changes Relative to Alternative A	K-177
K.3-13	Retail Rates under Alternative A	K-188
K.4-1	Calculation of Change in Tribal Benefit Resulting from LTEMP EIS Alternatives	K-194
K.4-2	Financial Impacts under Alternative B Relative to Alternative A for Tribal and Non-Tribal Entities	K-203
K.4-3	Financial Impacts under Alternative C Relative to Alternative A for Tribal and Non-Tribal Entities	K-203
K.4-4	Financial Impacts under Alternative D Relative to Alternative A for Tribal and Non-Tribal Entities	K-204
K.4-5	Financial Impacts under Alternative E Relative to Alternative A for Tribal and Non-Tribal Entities	K-204
K.4-6	Financial Impacts under Alternative F Relative to Alternative A for Tribal and Non-Tribal Entities	K-205
K.4-7	Financial Impacts under Alternative G Relative to Alternative A for Tribal and Non-Tribal Entities	K-205
K.5-1	Change in Lake Powell Monthly Water Release Volumes under LTEMP Alternatives	K-207
K.5-2	Average Lake Mead EOM Pool Elevations under LTEMP Alternatives	K-207
K.5-3	Average Monthly Difference in Turbine Water Releases Relative to Alternative A	K-212

K.5-4	Average Monthly Hoover Powerplant Generation for All Power Systems under Primary LTEMP EIS Alternatives	K-212
K.5-5	NPV of Hoover Powerplant Benefits Relative to Alternative A Resulting from LTEMP Alternatives	K-214
L-1	Elements of the Lake_Full Utility Model	L-5
L-2	Elements of the GCRec_Full Utility Model	L-6
L-3	Elements of the Regional Recreation Impacts Analysis	L-11
L-4	Elements of the Regional Electricity Price Impacts Analysis	L-16
L-5	Elements of the Regional Electricity Generating Capacity Expansion Impacts Analysis	L-17
P-1	Average Monthly Sand Load from the Paria River and Little Colorado River Showing the Fall and Spring HFE Accounting Periods and Implementation Windows	P-4

TABLES

1-1	Glen Canyon Dam Release Constraints under Modified Low Fluctuating Flows	1-21
1-2	Selected Documents Included in the Law of the River	1-32
2-1	Operational Characteristics of LTEMP Alternatives	2-9
2-2	Condition-Dependent and Experimental Elements of LTEMP Alternatives	2-11
2-3	Flow Parameters under Alternative A in an 8.23-maf Year	2-17
2-4	Flow Parameters under Alternative B in an 8.23-maf Year	2-21
2-5	Flow Parameters under Alternative C in an 8.23-maf Year	2-26
2-6	Implementation Criteria for Experimental Treatments of Alternative C	2-31
2-7	Flow Parameters for a Year with Low Summer Flows under Alternative C in an 8.23-maf Year	2-42
2-8	Flow Parameters under Alternative D in an 8.23-maf Year	2-45
2-9	Implementation Criteria for Experimental Treatments of Alternative D	2-50
2-10	Flow Parameters for a Year with Low Summer Flows under Alternative D in an 8.23-maf Year	2-68
2-11	Flow Parameters under Alternative E in an 8.23-maf Year	2-73
2-12	Flow Parameters under Alternative F in an 8.23-maf Year	2-82
2-13	Flow Parameters under Alternative G in an 8.23-maf Year	2-83
2-14	Summary of Impacts of LTEMP Alternatives on Resources	2-93
3.5-1	Native Fish of the Colorado River through Glen and Grand Canyons	3-72
3.5-2	Nonnative Fish Found in the Colorado River through Glen and Grand Canyons	3-91
3.5-3	Mean Water Temperature and Turbidity for Selected Sites in the Colorado River Mainstem from 2006 to 2009	3-94
3.5-4	Nonnative Warmwater Fish Species Reported from the Little Colorado River Watershed	3-102

3.6-1	Plant Communities Occurring on Reattachment Bars, Separation Bars, and Channel Margins	3-113
3.6-2	Special Status Plant Species Known to Occur along the Colorado River from Glen Canyon Dam to Lake Mead	3-122
3.7-1	Habitat and Distribution of Threatened, Endangered, and Sensitive Wildlife Species along the Colorado River Corridor between Glen Canyon Dam and Lake Mead	3-133
3.10-1	Colorado River Discovery Commercial Rafting Passengers 2009–2013	3-177
3.10-2	Tolerable Daily Flow Fluctuations Reported by Commercial and Private Trip Leaders	3-184
3.13-1	Energy and Capacity Characteristics of the Eight Largest WAPA Customers, 2013.	3-208
3.14-1	Population in the Six-County Region	3-210
3.14-2	Income in the Six-County Region	3-212
3.14-3	Employment in the Six-County Region	3-213
3.14-4	Employment by Sector in 2011	3-214
3.14-5	Unemployment Rates in the Six-County Region	3-215
3.14-6	Minority and Low-Income Populations in the 11-County Area	3-217
3.14-7	Population in the Seven-State Region of Influence	3-222
3.14-8	Income in the Seven-State Region of Influence	3-223
3.14-9	Employment in the Seven-State Region of Influence	3-224
3.14-10	Employment by Sector in 2011 in the Seven-State Region of Influence	3-225
3.14-11	Unemployment in the Seven-State Region of Influence	3-226
3.14-12	State Minority and Low-Income Populations, 2010	3-227
3.15-1	Clean Air Act Prevention of Significant Deterioration Designations	3-230

3.15-2	Criteria Pollutant and VOC Emissions in Counties Encompassing Grand Canyon National Park and for the Navajo Generating Station, 2011	3-231
3.15-3	Criteria Pollutant and VOC Emissions for 2011, and GHG Emissions for 2010, over the 11-State Affected Area within the Western Interconnection	3-236
4.1-1	Experimental Elements Included in Long-Term Strategies Associated with Each LTEMP Alternative	4-4
4.2-1	Summary of the Impacts of LTEMP Alternatives on Hydrology and Water Quality	4-15
4.2-2	Summary of Seasonal Temperature Data for LTEMP Alternatives from Lees Ferry to Diamond Creek	4-50
4.3-1	Summary of Impacts of LTEMP Alternatives on Sediment Resources	4-73
4.4-1	Indicators Used To Examine Natural Processes under Each LTEMP Alternative	4-90
4.4-2	Summary of Impacts of LTEMP Alternatives on Natural Processes Associated with Flow, Water Temperature, Water Quality, and Sediment Resources	4-92
4.5-1	Summary of Impacts of LTEMP Alternatives on Aquatic Ecology	4-109
4.5-2	Impact of High-Flow Experiments from Glen Canyon Dam on the Aquatic Food Base	4-116
4.6-1	Vegetation States, Plant Associations, and Corresponding Submodels	4-165
4.6-2	Vegetation States and Corresponding Mapped Vegetation Types	4-169
4.6-3	Summary of Impacts of LTEMP Alternatives on Vegetation	4-170
4.6-4	Transitions between Riparian Community Types and the Flows That Initiate Transitions	4-181
4.6-5	Priority Nonnative Species Identified for Control within the Colorado River Corridor	4-185
4.6-6	Summary of Impacts of LTEMP Alternatives on Special Status Plant Species	4-188
4.7-1	Summary of Impacts of LTEMP Alternatives on Wildlife	4-205

4.7-2	Summary of Impacts of LTEMP Alternatives on Special Status Wildlife Species	4-217
4.8-1	Summary of Impacts of LTEMP Alternatives on Cultural Resources in Glen and Grand Canyons	4-239
4.9-1	Vegetation Community Diversity and Change in Tamarisk Cover	4-253
4.9-2	Summary of Impacts of LTEMP Alternatives on Tribal Resources	4-263
4.10-1	Summary of Impacts of LTEMP Alternatives on Recreation, Visitor Use, and Experience.	4-282
4.10-2	Amount of Sediment Transported Out of Marble Canyon under the LTEMP Alternatives over the 20-Year LTEMP Period	4-295
4.11-1	Summary of Impacts of LTEMP Alternatives on Wilderness Experience	4-310
4.13-1	Summary of Impacts of LTEMP Alternatives on Hydropower Resources	4-335
4.14-1	Summary of Impacts of LTEMP Alternative on Socioeconomics and Environmental Justice	4-366
4.14-2	Mean Annual Net Economic Value of Recreation Associated with LTEMP Alternatives	4-376
4.14-3	Recreational Visitation by Activity in Lake Powell, Upper and Lower Grand Canyon, and Lake Mead, 2012	4-380
4.14-4	Mean Annual Employment Associated with Recreational Expenditures under LTEMP Alternatives	4-380
4.14-5	Mean Annual Income Associated with Recreational Expenditures under LTEMP Alternatives	4-381
4.14-6	Seven-State Economic Impacts under LTEMP Alternatives of Additional Generating Capacity for the Eight Largest Customer Utilities, 2015–2033	4-383
4.14-7	Average Annual Impacts on Economic Activity from Changes to Residential Electricity Bills of Largest Eight Customer Utilities, 2015–2033, Relative to	1 201
	Alternative A	4-384
4.14-8	Financial Impacts on Tribal and Non-Tribal Electricity Customers	4-389

4.15-1	Summary of Impacts of LTEMP Alternatives on Visibility and Regional Air Quality	4-404
4.15-2	Distributions of SO_2 and NO_x Emissions Averaged over the 20-Year LTEMP Period by Alternative	4-408
4.16-1	Summary of Impacts of LTEMP Alternatives on GHG Emissions	4-417
4.16-2	Expected Impact of LTEMP Alternatives on Downstream Resources under Climate Change Compared to Those under Historic Conditions	4-422
4.17-1	Impacting Factors Associated with Past, Present, and Reasonably Foreseeable Future Actions and Basin-Wide Trends in the LTEMP Project Area	4-428
4.17-2	Summary of Cumulative Impacts and Incremental Contributions under LTEMP Alternatives	4-447
5.1-1	Summary of Cooperating Agency Involvement	5-3
B-1	Tolerable Flow Fluctuations for Recreational River Use	B-13
D-1	Initial Reservoir Conditions	D-5
D-2	Monthly Release Volumes by Water-Year Release for Alternative A	D-11
D-3	Monthly Release Volumes by Water-Year Release for Alternative B	D-12
D-4	Monthly Release Volumes by Water-Year Release for Alternative C	D-13
D-5	Monthly Release Volumes by Water-Year Release for Alternative C with Low Summer Flows	D-14
D-6	Monthly Release Volumes by Water-Year Release for Alternative D	D-15
D-7	Monthly Release Volumes by Water-Year Release for Alternative D with Low Summer Flows	D-16
D-8	Monthly Release Volumes by Water-Year Release for Alternative E	D-17
D-9	Monthly Release Volumes by Water-Year Release for Alternative E with Low Summer Flows	D-18
D-10	Monthly Release Volumes by Water-Year Release for Alternative F	D-19

D-11	Monthly Release Volumes by Water-Year Release for Alternative G	D-20
D-12	Minimum Release Constraints Used for Each Alternative	D-21
D-13	Input Demands by State	D-27
E-1	Sources for Historical Tributary Sediment Load Data	E-38
E-2	Historical Periods Used for Paria Sediment Traces s1, s2, and s3	E-38
E-3	Historical Periods Used for Little Colorado River Sediment Traces s1, s2, and s3	E-38
E-4	List of HFEs Available for Sediment-Triggered Experiments	E-39
F-1	Average Mean Habitat-Weighted Invertebrate Biomass at Select Sites in the Colorado River, July 2006–June 2009	F-8
F-2	Average Mean Habitat-Weighted Invertebrate Production at Select Sites in the Colorado River, July 2006–June 2009	F-9
F-3	Average Mean Habitat-Weighted Invertebrate Abundance at Select Sites in the Colorado River, July 2006–June 2009	F-10
F-4	Distribution, Ecological Importance, and Favorable Temperature Range for Select Primary Producers	F-11
F-5	Temperature Requirements for Common Zooplankton	F-24
F-6	Distribution, Importance to Higher Trophic Levels, and Temperature Range for Common Benthic Macroinvertebrates Downstream of Glen Canyon Dam	F-25
F-7	Temperature Requirements for the Asian Tapeworm, Anchor Worm, and Trout Nematode	F-26
F-8	Summary of Metrics Values from the Rainbow Trout-Humpback Chub Model	F-53
F-9	Description of Input Parameters for the LTEMP Temperature Suitability Model	F-66
F-10	Humpback Chub Aggregation Locations	F-67
F-11	Locations Used for Temperature Suitability Modeling of Native Fish, Nonnative Fish, and Parasites	F-73

G-1	Vegetation States, Plant Associations, and Corresponding Submodels	G-15
G-2	Hydrologic Events Considered in the Riparian Vegetation Model	G-16
G-3	Riparian Vegetation Model Transition Rules	G-17
G-4	New High-Water Zone and Old High-Water Zone Vegetation Classes Mapped from Lees Ferry to Diamond Creek	G-22
G-5	Vegetation States and Corresponding Mapped Vegetation Types	G-24
G-6	Example Results for the Native Cover Metric	G-25
G-7	Example Results for the Arrowweed Metric	G-25
G-8	Results for Alternative A	G-26
G-9	Results for Alternative B, Long-Term Strategy B1	G-27
G-10	Results for Alternative B, Long-Term Strategy B2	G-28
G-11	Results for Alternative C, Long-Term Strategy C1	G-29
G-12	Results for Alternative C, Long-Term Strategy C2	G-30
G-13	Results for Alternative C, Long-Term Strategy C3	G-31
G-14	Results for Alternative C, Long-Term Strategy C4	G-32
G-15	Results for Alternative D, Long-Term Strategy D1	G-33
G-16	Results for Alternative D, Long-Term Strategy D2	G-34
G-17	Results for Alternative D, Long-Term Strategy D3	G-35
G-18	Results for Alternative D, Long-Term Strategy D4	G-36
G-19	Results for Alternative E, Long-Term Strategy E1	G-37
G-20	Results for Alternative E, Long-Term Strategy E2	G-38
G-21	Results for Alternative E, Long-Term Strategy E3	G-39

G-22	Results for Alternative E, Long-Term Strategy E4	G-40
G-23	Results for Alternative E, Long-Term Strategy E5	G-41
G-24	Results for Alternative E, Long-Term Strategy E6	G-42
G-25	Results for Alternative F	G-43
G-26	Results for Alternative G	G-44
G-27	The Effects of LTEMP Long-Term Strategies on Wetlands	G-45
I-1	Federal Regulations and Executive Orders Pertaining to Consultation with Tribes	I-3
I-2	Vegetation States	I-5
J-1	Reported Mean Tolerable Daily Changes in Flow Levels for Commercial Motor Guides, Commercial Oar Guides, and Private Trip Leaders Who Have Experienced Daily Fluctuations of at Least 15,000 cfs in the Grand Canyon	J-12
J-2	Recreation Response to Daily Maximum Flow	J-20
J-3	Summary of Recreation, Visitor Use, and Experience Metrics	J-26
K.1-1	Cost and Performance Characteristics of Capacity Expansion Candidates	K-51
K.1-2	Cumulative Annual Percentage Increase in Natural Gas Prices at the Henry Hub	K-55
K.1-3	Cumulative Annual Percent Increase in Regional Coal Prices	K-58
K.1-4	Cumulative Annual Percentage Increase in Regional Distillate Fuel Prices	K-60
K.1-5	Assumed Ancillary Services Provided by SLCA/IP Hydropower Facilities from 2013 to 2030	K-65
K.1-6	Summary of Utility IRPs for Four Large SLCA/IP Customers and Other Large Utilities in Areas Neighboring the SLCA/IP System	K-87
K.1-7	SLCA/IP Marketable Capacity at the 90% Exceedance Level	K-102
K.1-8	Difference in Cumulative Capacity Additions of Each Alternative Relative to Alternative A.	K-105

K.1-9	Comparison of the Amount and Timing of New Capacity Additions for Alternatives A and F	K-107
K.1-10	Total Economic Impacts by Alternative at the 90% Exceedance Level	K-109
K.1-11	Estimated Cost of LTEMP Experiments	K-116
K.1-12	Comparison of Marketable Capacity, Replacement Capacity, and Capacity Additions by Exceedance Level	K-117
K.1-13	Firm SLCA/IP Federal Hydropower Capacity at Various Exceedance Levels at Glen Canyon Dam by Alternative and Ancillary Service Scenarios	K-127
K.1-14	Comparison of Capacity and Energy Values at Glen Canyon Dam by Alternative and Ancillary Services Scenarios at the 90% Exceedance Level	K-128
K.1-15	Difference Relative to Alternative A	K-128
K.1-16	Alternative Cost Increase Estimates for Mixed NGCC/CT Technology Capacity Replacement Relative to CT Only Replacement as Estimated Using GTMax-Lite	K-130
K.1-17	Summary of Economic Rankings for Baseline and All Sensitivity Scenarios	K-132
K.2-1	SHP Contractual Obligations and Project Use Required Deliveries by Month	K-136
K.2-2	SHP Hourly Load Shapes by Month and Type of Day	K-137
K.2-3	Total Levelized Capital and Fixed O&M Cost for System Capacity Expansion by Alternative and Year	K-141
K.2-4	Total Levelized Capital and Fixed O&M Cost for System Capacity Expansion by Alternative and Year Expressed as the Difference from Alternative A	K-143
K.2-5	Total Net Purchase Power Expenses by Year and Alternative for the Continuous Current Obligations Scenario	K-147
K.2-6	Total Net Purchase Power Expenses by Year and Alternative for the Reduced Obligations to Match Resource Scenario	K-148
K.2-7	SLCA/IP Rate Impact by Alternative	K-151
K.3-1	Energy Allocations for Systems Receiving Indirect SLCA/IP Allocations	K-157

K.3-2	Coverage of Retail Information from EIA Database Relative to SLCA/IP Preference Power Allocation	K-159
K.3-3	Average Annual Grid Cost Relative to SLCA/IP Wholesale Revenues Relative to Alternative A	K-168
K.3-4	Summary Table of Comparative Values	K-178
K.3-5	Individual Systems with the Largest Percent Retail Rate Impacts Relative to Alternative A	K-180
K.3-6	Individual Systems with the Smallest Percent Retail Rate Impacts Relative to Alternative A	K-182
K.3-7	Individual Systems with the Largest Monthly Residential Bill Impacts Relative to Alternative A	K-183
K.3-8	Individual Systems with the Smallest Monthly Residential Bill Impacts Relative to Alternative A	K-184
K.3-9	Size and Preference Ratio for Utility Systems with Large Rate Impacts	K-186
K.3-10	Rate Impacts for Selected Groups in Maximum Impact Year	K-186
K.4-1	Monthly Change in Residential Electric Utility Bill for Tribes by Alternative	K-197
K.4-2	Total Dollar Annual Impact on Tribes under LTEMP Alternatives Relative to Alternative A	K-199
L-1	Total Regional Economic Impacts of Non-Local Recreation Expenditures, 2013.	L-13
L-2	Urban Population, Income, and Poverty in the Six-County Region	L-18
M-1	Emission Factors by Plant for System Power Generation and Spot Market	M-10
M-2	Power Generation Averaged over the 20-Year LTEMP Period by Alternative	M-12
M-3	Summary of Potential Impacts of LTEMP Alternatives on Visibility and Regional Air Quality	M-15
M-4	Annual SO ₂ Emissions Averaged over the 20-Year LTEMP Period by Alternative.	M-16

M-5	Annual NO _x Emissions Averaged over the 20-Year LTEMP Period by Alternative.	M-19
M-6	Summary of Impacts of LTEMP Alternatives on SO ₂ and NO _x Emissions	M-22
M-7	Annual Greenhouse Gas Emissions under LTEMP Alternatives	M-23
M-8	Summary of Impacts of LTEMP Alternatives on CO2e Emissions	M-26
N-1	Summary of Tribal Participation Status	N-5
N-2	Summary of Tribal Correspondence	N-6
N-3	Summary of Major Face-to-Face Meetings, Webinars, and Conference Calls Involving Tribes	N-25
N-4	Summary of Individual Tribal Consultation Meetings to Date	N-27
N-5	Index of Project Tribal Government Consultation and Coordination Correspondence	N-29
P-1	List of HFEs Available for Sediment-Triggered Experiments under the Preferred Alternative	P-7
Q-1	Comment Issues and Location in the Report	Q-6
Q-2	Commenter Distribution by State	Q-10
Q-3	Commenter, Affiliation, and Location Where Responses Are Found	Q-88

ACRONYMS AND ABBREVIATIONS

ac	acre(s)
ac-ft	acre-foot (feet)
ACHP	Advisory Council on Historic Preservation
AML	abandoned mine land
AMSL	above mean sea level
AMWG	Adaptive Management Work Group
AOP	Annual Operation Plan for Colorado River Reservoirs
APE	Area of Potential Effect
Argonne	Argonne National Laboratory
ASMR	Age-Structured Mark Recapture Model
AZGFD	Arizona Game and Fish Department
AZ-SGCN	Arizona Species of Greatest Conservation Need
BA	Balancing Authority (in Chapter 3 only)
	Biological Assessment (in all other sections)
BGEPA	Bald and Golden Eagle Protection Act of 1940
BIA	Bureau of Indian Affairs
BO	Biological Opinion
С	Celsius
CAA	Clean Air Act
CAAA	Clean Air Act Amendments
CAEDYM	Computational Aquatic Ecosystem Dynamics Model
CCC	Civilian Conservation Corps
CEQ	Council on Environmental Quality
CFMP	Comprehensive Fisheries Management Plan
CFR	Code of Federal Regulations
cfs	cubic feet per second
CH ₄	methane
CO	carbon monoxide
CO_2	carbon dioxide
CO ₂ e	carbon dioxide equivalent
CPUE	catch per unit effort
CRBC	Colorado River Board of California
CRCN	Colorado River Commission of Nevada
CRD	Colorado River Discovery
CREDA	Colorado River Energy Distributors Association
CRMP	Colorado River Management Plan
CRSP	Colorado River Storage Project
CRSPA	Colorado River Storage Project Act of 1956
CRSS	Colorado River Simulation System
CSU	Colorado Springs Utilities

DEIS	Draft Environmental Impact Statement
Deseret	Deseret Generation and Transmission Cooperative
DFC	desired future condition
DO	dissolved oxygen
DOE	U.S. Department of Energy
DOI	U.S. Department of the Interior
DPS	Distinct Population Segment
EA	Environmental Assessment
eGRID	Emissions & Generation Resource Integrated Database
EIA	Energy Information Administration
EIR	Environmental Impact Report
EIS	Environmental Impact Statement
ELCOM	Estuary, Lake and Coastal Ocean Model
EMS	emergency medical services
E.O.	Executive Order
EPA	U.S. Environmental Protection Agency
EPT	Ephemeroptera, Plecoptera, and Trichoptera
ESA	Endangered Species Act of 1973, as amended
F	Fahrenheit
FCPP	Four Corners Power Plant
FES	Firm Electric Service
FONSI	Finding of No Significant Impact
FR	Federal Register
ft	foot (feet)
FWS	U.S. Fish and Wildlife Service
FY	fiscal year
GCDAMP	Glen Canyon Dam Adaptive Management Program
GCM	general circulation model
GCMRC	Grand Canyon Monitoring and Research Center
GCNP	Grand Canyon National Park
GCNRA	Glen Canyon National Recreation Area
GCPA	Grand Canyon Protection Act of 1992
GHG	greenhouse gas
GMP	General Management Plan
GW	gigawatt(s)
GWh	gigawatt-hour(s)
GWP	global warming potential
H_2S	hydrogen sulfide
HBC	humpback chub
HFC	hydrofluorocarbon
HFE	high-flow experiment
hr	hour(s)

HRR	Hualapai River Runners
in.	inch(es)
IPM	Integrated Pest Management
IRP	integrated resource plan
ISM	Indexed Sequential Method
kaf	thousand acre-feet
kWh	kilowatt-hour(s)
lb	pound(s)
LCRMSCP	Lower Colorado River Multi-Species Conservation Program
LMM	Lake Mead Model
LMNRA	Lake Mead National Recreation Area
LROC	Long-Range Operating Criteria
LTEMP	Long-Term Experimental and Management Plan
LTEP	Long Term Experimental Plan
LTF	long-term firm
maf	million acre-feet
MAMB	miscellaneous algae, macrophytes, and bryophytes
MBTA	Migratory Bird Treaty Act
MCL	maximum contaminant level
mi	mile(s)
MLFF	Modified Low Fluctuating Flow
MMt	million metric tons
MOA	Memorandum of Agreement
MT	metric ton(s)
MW	megawatt(s)
MWh	megawatt-hour(s)
N ₂ O	nitrous oxide
NAAQS	National Ambient Air Quality Standards
NAU	Northern Arizona University
NC	no change
NEPA	National Environmental Policy Act of 1969, as amended
NERC	North American Electric Reliability Corporation
NEV	net economic use value
NGO	nongovernmental organization
NHPA	National Historic Preservation Act
NM	national monument
NO ₂	nitrogen dioxide
NO ₃	nitrate-nitrogen
NOI	Notice of Intent
NO _x	nitrogen oxides
NPS	National Park Service

NPV	net present value
NRHP	National Register of Historic Places
NTUA	Navajo Tribal Unit Authority
O&M	operation and maintenance
O ₃	ozone
OPAC	Office of Planning and Compliance
OSMRE	Office of Surface Mining Reclamation and Enforcement
	-
PA	Programmatic Agreement
Pb	lead
PEPC	Planning, Environment, and Public Comment
PFC	perfluorocarbon
P.L.	Public Law
РМ	particulate matter
PM2 5	particulate matter ≤ 2.5 µm in aerodynamic diameter
PM10	particulate matter <10 um in aerodynamic diameter
POM	particulate organic matter
PSAR	preventative search and rescue
PSD	Prevention of Significant Deterioration
I SD	Trevention of Significant Deterioration
RA	resource available
Reclamation	Bureau of Reclamation
RM	river mile
RMP	Resource Management Plan
ROD	Record of Decision
RSG	Reserve Sharing Group
Roo	Reserve blaring Group
SAAQS	State Ambient Air Quality Standards
SBM	Sand Budget Model
SCP	Salinity Control Project
SD	standard deviation
SDA	Structured Design Analysis
SE	standard error
Secretary the	Secretary of the Interior
SECONDARY, INC	sulfur hexaflouride
SHPO	State Historic Preservation Officer
SI CA/IP	Salt Lake City Area Integrated Projects
SLCA/II	San Lake City Area Integrated 110 jetts
30 SO:	
SU2	Suntu utoxide
SPU	Soumern Palute Consortium
SKP	Salt River Project
SRSG	Southwest Reserve Sharing Group
TCD	temperature control device
	traditional aultural property
ICF	inautional cultural property

TDS	total dissolved solids
THPO	Tribal Historic Preservation Officer
TL	total length
TMF	trout management flow
Tri-State	Tri-State Generation and Transmission Association
TWG	Technical Working Group
UAMPS	Utah Associated Municipal Power Systems
UBWR	Utah Board of Water Resources
UCRC	Upper Colorado River Commission
UMPA	Utah Municipal Power Agency
USC	United States Code
USFS	U.S. Forest Service
USGS	U.S. Geological Survey
VOC	volatile organic compound
WACM	Western Area Colorado-Missouri Region
WALC	Western Area Lower Colorado Region
WAUW	Western Area Upper Great Plains West Region
WECC	Western Electricity Coordinating Council
Western	Western Area Power Administration
YOY	young-of-year
yr	year(s)
ZHHPO	Zuni Heritage and Historic Preservation Office

This page intentionally left blank

APPENDIX O:

BIOLOGICAL ASSESSMENT FOR THE GLEN CANYON DAM LONG-TERM EXPERIMENTAL AND MANAGEMENT PLAN

This page intentionally left blank



Biological Assessment for the Glen Canyon Dam Long-Term Experimental and Management Plan (LTEMP)


CONTENTS

A	CRON	YMS A	ND ABBREVIATIONS	ix
1	INTR	ODUCI	TON AND BACKGROUND	1
	1.1	Synops	is of the Proposed Action	3
	1.2	Descrip	otion of the Federal Action	4
		1.2.1	Regulatory Context	4
		1.2.2	Action Area	5
		1.2.3	Project Description	6
		1.2.4	Conservation Measures	10
			1.2.4.1 Humpback Chub	10
			1.2.4.2 Razorback Sucker	12
			1.2.4.3 Nonnative Species – Removal and Control Actions	13
			1.2.4.4 Southwestern Willow Flycatcher and Yuma Ridgway's Rail	14
		1.2.5	Base Operations	15
		1.2.6	Operational Flexibility	15
		1.2.7	Implementation of Experimental Elements	18
		1.2.8	Overall Implementation Process for Experiments	18
		1.2.9	Sediment-Related Experimental Treatments	24
			1.2.9.1 Sediment-Triggered Spring HFEs	26
			1.2.9.2 Proactive Spring HFEs	27
			1.2.9.3 Sediment-Triggered Fall HFEs	28
			1.2.9.4 Extended-Duration Fall HFEs	28
		1.2.10	Aquatic Resource-Related Experimental Treatments	30
			1.2.10.1 Trout Management Flows	31
			1.2.10.2 Tier 1 Conservation Actions for Humpback Chub	34
			1.2.10.3 Tier 2 Mechanical Removal of Nonnative Fish	35
			1.2.10.4 Low Summer Flows	36
			1.2.10.5 Low Steady Weekend Flows for Macroinvertebrate	
			Production	40
		1.2.11	Native and Nonnative Plant Management and Experimental Actions	41
2	ENVI	RONM	ENTAL BASELINE	43
	2.1	Descrip	otion of Species Identified for Analysis	43
		2.1.1	Kanab Ambersnail	43
			2.1.1.1 Legal Status	43
			2.1.1.2 Recovery Goals and Status	43
			2.1.1.3 Historic and Current Range	44
			2.1.1.4 Habitat	44
			2.1.1.5 Life History	44
		2.1.2	Humpback Chub	45
			2.1.2.1 Legal Status	45
			2.1.2.2 Recovery Goals and Status	45

CONTENTS (Cont.)

			2.1.2.3	Historic and Current Range	47
			2.1.2.4	Population within the Action Area	48
			2.1.2.5	Habitat	52
			2.1.2.6	Life History	55
			2.1.2.7	Factors Affecting Distribution and Abundance in the	
				Grand Canyon	57
		2.1.3	Razorba	ck Sucker	60
			2.1.3.1	Legal Status	60
			2.1.3.2	Recovery Goals and Status	61
			2.1.3.3	Historic and Current Range	61
			2.1.3.4	Population within the Action Area	63
			2.1.3.5	Habitat	64
			2.1.3.6	Life History	64
			2.1.3.7	Factors Affecting Distribution and Abundance in the	
				Grand Canyon	65
		2.1.4	Southwe	estern Willow Flycatcher	68
			2.1.4.1	Legal Status	68
			2.1.4.2	Recovery Goals and Status	68
			2.1.4.3	Historic and Current Range	68
			2.1.4.4	Population within the Action Area	69
			2.1.4.5	Lees Ferry to Diamond Creek	70
			2.1.4.6	Lower Grand Canyon	70
			2.1.4.7	Life History and Habitat	71
			2.1.4.8	Factors Affecting Distribution and Abundance in the	
				Grand Canyon	72
		2.1.5	Yuma R	idgway's Rail	72
			2.1.5.1	Legal Status	72
			2.1.5.2	Recovery Goals and Status	73
			2.1.5.3	Historic and Current Range	73
			2.1.5.4	Population within the Action Area	73
			2.1.5.5	Habitat and Life History	74
			2.1.5.6	Factors Affecting Distribution and Abundance in the	
				Grand Canyon	75
	2.2	Baselin	ne Take R	esulting from Handling and GCDAMP Research	75
		2.2.1	Humpba	ick Chub	75
		2.2.2	Razorba	ck Sucker	76
r	PPPP	OTO AN		OF PROPOSED ACTION	77
3	EFFE	UIS AP	NALYSIS	OF PROPOSED ACTION	//
	3.1	Effects	of the Pr	oposed Action on Kanab Ambersnail	77
		3.1.1	Direct a	nd Indirect Effects	77
			3.1.1.1	Effects of Sediment-Related Experiments	77
			3.1.1.2	Effects of Vegetation Management	78

CONTENTS (Cont.)

		3.1.1.3	Effects of Climate Change	78
3.2	Effects	of the Pro	oposed Action on Humpback Chub	78
	3.2.1	Direct ar	nd Indirect Effects	79
		3.2.1.1	Effects of Base Operations	79
		3.2.1.2	Effects of Dam Operational Flexibility	82
		3.2.1.3	Overall Implementation Process for Experiments	84
		3.2.1.4	Effects of Sediment-Related Experiments	84
		3.2.1.5	Effects of Trout Management Flow Experiments	88
		3.2.1.6	Effects of Triggered Conservation Actions and Mechanical	
			Removal of Nonnative Fish Experiments	90
		3.2.1.7	Effects of Low Summer Flow Experiments	91
		3.2.1.8	Effects of Low Steady Weekend Flows for Macroinvertebrate	
			Production Experiments	92
		3.2.1.9	Effects of Ongoing Conservation Measures	93
		3.2.1.10	Effects of New Conservation Measures	98
		3.2.1.11	Effects of Native and Nonnative Plant Management	
			Experimental Actions	98
		3.2.1.12	Effects of Activities to Preserve and Protect Historic	
			Properties and Cultural Sites	98
		3.2.1.13	Effects of Climate Change	98
3.3	Effects	of the Pro	oposed Action on Razorback Sucker	99
	3.3.1	Direct ar	nd Indirect Effects	100
		3.3.1.1	Effects of Base Operations	100
		3.3.1.2	Effects of Sediment-Related Experiments and Low Summer	
			Flows	102
		3.3.1.3	Effects of Low Steady Weekend Flows for Macroinvertebrate	
			Production Experiments	103
		3.3.1.4	Effects of Trout Management Flow Experiments	103
		3.3.1.5	Effects of Triggered Conservation Actions and Mechanical	
			Removal of Nonnative Fish Experiments	104
		3.3.1.6	Effects of Conservation Measures	105
		3.3.1.7	Effects of Climate Change	108
3.4	Effects	of the Pro	pposed Action on Southwestern Willow Flycatcher	108
3.5	Effects	of the Pro	oposed Action on Yuma Ridgway's Rail	109
3.6	Cumula	tive Effe	cts	110
	3.6.1	Grand C	anyon Escalade	110
	3.6.2	Uranium	Mining on State and Private Lands	111
	3.6.3	Increased	d Municipal and Agricultural Water Demand	112
	3.6.4	Urban ar	nd Agricultural Runoff	112
	3.6.5	Nonnativ	ve Vegetation and Defoliating Nonnative Insect Expansion	112
	3.6.6	Cumulat	ive Effects Summary	113
		3.6.6.1	Aquatic Species – Humpback Chub, Razorback Sucker, and	
			Kanab Ambersnail	113

CONTENTS (Cont.)

		267	3.6.6.2 Wildlife	113
		5.0.7	Summary of Chinate-Related Changes	115
4	CON	CLUSI	ONS AND DETERMINATIONS OF EFFECT	117
	4.1	Listed	Species Determinations	117
	4.2	Hump	back Chub	119
	4.3	Razorl	back Sucker	120
	4.4	Kanab	Ambersnail	121
	4.5	Southv	vestern Willow Flycatcher	121
	4.6	Yuma	Ridgway's Rail	121
5	LITE	RATUF	RE CITED	123
A	PPENI	DIX A:	Southwestern Willow Flycatcher Detections and Nest Sites,	
			Grand Canyon National Park, Arizona, 1982–2012	A-1
A	PPENI	DIX B:	Southwestern Willow Flycatcher Historic and Recent Territories and Nesting Sites, Grand Canyon National Park, Arizona	B-1
A	PPENI	DIX C:	Ridgway's Rail Historic Records at Grand Canyon National Park, Arizona	C-1
A	PPENI	DIX D:	Proposed Action Triggers for the Management of Humpback Chub Colorado River, Grand Canyon	D-1

FIGURES

1	Locations of Glen Canyon Dam, Lake Powell, and the Colorado River between Lake Powell and Lake Mead, and Adjacent Lands	5
2	Mean, Minimum, and Maximum Daily Flows under the Proposed Action in an 8.23-maf Year	16
3	Simulated Hourly Flows under the Proposed Action in an 8.23-maf Year	17
4	Decision Tree for Implementation of Sediment-Related Experimental Treatments under the Proposed Action	25
5	Example Implementation of a Two-Cycle Trout Management Flow in June and July with Resumption of Normal Fluctuations between Cycles and Afterward	31

6	Mean, Minimum, and Maximum Daily Flows under Triggered Low Summer Flows of the Proposed Action in an 8.23-maf Year Based on the Values Presented in Table 4	38
7	Closed Chapman Petersen Abundance Estimates of Humpback Chub \geq 150 mm and \geq 200 mm in the Little Colorado River during Spring and Fall Seasons	49
8	Estimated Adult Humpback Chub Abundance from Age-Structured Mark- Recapture Model Incorporating Uncertainty in Assignment of Age	51
9	Estimated Total Adult Abundance of Humpback Chub in the Lower 8 mi of the Little Colorado River and a 2-mi Portion of the Colorado River Just Downriver of the Confluence of the Little Colorado and Colorado Rivers, for September 2009 through 2012	51
10	Total Abundance of Humpback Chub in Havasu Creek, Based on Annual Population Estimates between 2012 and 2015	53
11	Humpback Chub Aggregation Areas along the Colorado River between Glen Canyon Dam and Lake Mead	53
12	Water Temperatures at the Little Colorado River Confluence, 1995 to Present	60
13	Historic and Present Range of Razorback Sucker in the Colorado River Basin	62
C.1	Threatened, Endangered, and Sensitive Species Observed along the Colorado River Corridor	C-3

TABLES

1	Federally Listed Species Occurring within the Project Area	1
2	Flow Parameters under the Proposed Action in an 8.23-maf Year	16
3	Implementation Criteria for Experimental Treatments under the Proposed Action	20
4	Flow Parameters for a Year with Low Summer Flows under the Proposed Action in an 8.23-maf Year	37
5	Temperature Requirements for Spawning, Incubation, Growth, and Lethality of 36 Native and Nonnative Fish Species of the Colorado River and Its Tributaries in and near the Project Area	83
6	Summary of Effects Determinations for the Five Listed Species	118
C.1	Ridgway's Rail Historic Records at Grand Canyon National Park, Arizona	C-3

This page intentionally left blank

ACRONYMS AND ABBREVIATIONS

ac	acre(s)
ac-ft	acre-foot (feet)
AMWG	Adaptive Management Work Group
ASMR	Age-Structured Mark Recapture Model
AZGFD	Arizona Game and Fish Department
	1
BA	biological assessment
BIA	Bureau of Indian Affairs
BO	Biological Opinion
_	
C	Celsius
CFMP	Comprehensive Fisheries Management Plan
CFR	Code of Federal Regulations
cfs	cubic feet per second
cm	centimeter(s)
CRSP	Colorado River Storage Project
CRSPA	Colorado River Storage Project Act of 1956
DOI	U.S. Department of the Interior
EA	environmental assessment
FIS	Environmental Impact Statement
LID	Environmental impact Statement
EPA	U.S. Environmental Protection Agency
EPT	Ephemeroptera Plecoptera and Trichoptera
ESA	Endangered Species Act of 1973 as amended
2011	Endungered Species rice of 1979, us uniended
F	Fahrenheit
FR	Federal Register
ft	foot (feet)
ft ³	cubic foot (feet)
FWS	U.S. Fish and Wildlife Service
GCDAMP	Glen Canyon Dam Adaptive Management Program
GCMRC	Grand Canyon Monitoring and Research Center
GCNP	Grand Canyon National Park
GCNRA	Glen Canyon National Recreation Area
GCPA	Grand Canyon Protection Act of 1992
HBC	humpback chub
HEE	high_flow experiment
	mgn-now experiment
11g hr	hour(a)
111	nour(s)

IKAMPT in.	Interagency Kanab Ambersnail Monitoring Team inch(es)
JCM	juvenile chub monitoring
kaf	thousand acre-feet
km	kilometer(s)
lb	pound(s)
LCRMSCP	Lower Colorado River Multi-Species Conservation Program
LMNRA	Lake Mead National Recreation Area
LTEMP	Glen Canyon Dam Long-Term Experimental and Management Plan
maf	million acre-feet
mi	mile(s)
MLFF	Modified Low Fluctuating Flow
mm	millimeter(s)
MVP	Minimum Viable Population
NEPA	National Environmental Policy Act of 1969, as amended
NHPA	National Historic Preservation Act
NPS	National Park Service
NRHP	<i>National Register of Historic Places</i>
OSMRE	Office of Surface Mining Reclamation and Enforcement
ppb	part(s) per billion
Reclamation	Bureau of Reclamation
RM	river mile
ROD	Record of Decision
Se	selenium
SNARRC	Southwestern Native Aquatic Resources and Recovery Center
TL	total length
TMF	trout management flow
TWG	Technical Working Group
UCRC	Upper Colorado River Commission
USC	United States Code
USFS	U.S. Forest Service
USGS	U.S. Geological Survey

- WAPA Western Area Power Administration
- YOY young-of-the-year

This page intentionally left blank

1 INTRODUCTION AND BACKGROUND

The Bureau of Reclamation (Reclamation) has prepared this biological assessment (BA) to analyze the potential effects of the proposed Glen Canyon Dam Long-Term Experimental and Management Plan (LTEMP) as described in the associated Environmental Impact Statement (EIS) on federally listed species. The BA analyzes the effects of implementation of flow and non-flow actions over a 20-year period at Glen Canyon Dam and in the Colorado River downstream of Glen Canyon Dam within Glen Canyon National Recreation Area (GCNRA) and Grand Canyon National Park (GCNP), Coconino County, Arizona, to Lake Mead National Recreation Area (LMNRA) (Figure 1). This BA was prepared by Reclamation as part of its compliance with the Endangered Species Act of 1973, as amended (ESA; 87 Stat. 884; 16 U.S.C. §1531 et seq.). A BA evaluates the potential effects of the action on listed and proposed species and designated and proposed critical habitat and determines whether any such species or habitat are likely to be adversely affected by the action (50 CFR 402.12).

A total of nine species currently listed under the ESA occur within the project area (Table 1). Five species are addressed in this BA due to potential effects from the proposed action: humpback chub (*Gila cypha*), razorback sucker (*Xyrauchen texanus*), southwestern willow flycatcher (*Empidonax traillii extimus*), Ridgway's Rail (formerly Yuma Clapper Rail) (*Rallus obsoletus*), and the Kanab ambersnail (*Oxyloma kanabensis*). Reclamation made "No effect" determinations for the Mexican spotted owl (*Strix occidentalis lucida*), the western yellow-billed cuckoo (*Coccyzus americanus*), and sentry milk-vetch (*Astragalus cremnophylax cremnophyla*). None of these species, or their habitat, occur in the area where activities would be implemented, and thus they are not considered further in this analysis. Reclamation has previously consulted on the bald eagle (*Haliaeetus leucocephalus*), American peregrine falcon (*Falco peregrinus anatum*), and California condor (*Gymnogyps californianus*). The California condor is an endangered species that would not be affected by this action, and the condors released in Arizona are designated as a non-essential, experimental population under the 10(j) rule of the ESA. The peregrine falcon and bald eagle have been removed from the list of threatened and endangered species and are not addressed in this BA.

Name	Species	Status
Mexican Spotted Owl	Strix occidentalis lucida	Threatened
Southwestern Willow Flycatcher	Empidonax traillii extimus	Endangered
California Condor	Gymnogyps californianus	Considered Threatened in National Parks; Experimental Non-Essential population in Northern Arizona
Western Yellow-billed Cuckoo	Coccyzus americanus	Threatened
Yuma Ridgway's (formerly Yuma Clapper) Rail	Rallus obsoletus yumanensis	Endangered
Kanab Ambersnail	Oxyloma haydeni kanabensis	Endangered
Humpback Chub	Gila cypha	Endangered
Razorback Sucker	Xyrauchen texanus	Endangered
Sentry Milk-vetch	Astragalus cremnophylax cremnophylax	Endangered

TABLE 1 Federally Listed Species Occurring within the Project Area

The BA provides the information and analysis to support Reclamation's conclusions concerning possible effects on the species and habitats listed in Table 1. Based upon the information contained in this BA, Reclamation has determined that the proposed action may affect, but is not likely to adversely affect, the southwestern willow flycatcher and Ridgway's Rail, and requests concurrence from the U.S. Fish and Wildlife Service (FWS). Reclamation has also determined that the proposed action may affect and is likely to adversely affect the humpback chub and its critical habitat, the razorback sucker and its critical habitat, and the Kanab ambersnail, and provides this information to the FWS to initiate formal consultation.

Reclamation is the lead agency within the U.S. Department of the Interior (DOI) that operates Glen Canyon Dam of the Colorado River Storage Project as a multipurpose storage facility in northern Arizona. Construction and operation of the dam were authorized by the 1956 Colorado River Storage Project Act. Operation of the dam is governed by a complex set of compacts, federal statutes and regulations, court decrees, and an international treaty collectively and commonly referred to as the Law of the River.

Development of the proposed action included the evaluation of specific alternatives that could be implemented to meet the Grand Canyon Protection Act's (GCPA's; 1992) requirements and to minimize, consistent with law, adverse impacts on the downstream natural, recreational, and cultural resources in the two park units, including resources of importance to American Indian Tribes. The need for the proposed action stems from the need to use scientific information developed since the 1996 Record of Decision (ROD) and the Operation of Glen Canyon Dam Final Environmental Impact Statement (1995 EIS) to better inform DOI decisions on dam operations and other management and experimental actions so that the Secretary of the Interior (Secretary) may continue to meet statutory responsibilities for protecting downstream resources for future generations, conserving federally listed species, avoiding or mitigating impacts on National Register-eligible properties, and protecting Native American interests, while meeting obligations for water delivery and the generation of hydroelectric power. The action will provide a framework for adaptively managing Glen Canyon Dam over the next 20 years consistent with the GCPA and other provisions of applicable federal law. Several key issues related to natural resources downstream of Glen Canyon Dam and new scientific information related to them are summarized below.

The Colorado River downstream from Glen Canyon Dam is depleted of its natural sediment load due to the presence of the dam, and many types of ongoing dam releases further deplete sediment delivered to the main channel by causing erosion. However, high-flow releases, between approximately 30,000 and 45,000 cfs that are triggered when there is sufficient sediment from the Paria River, mobilize sand stored in the river channel and redeposit it as sandbars and beaches and associated backwater and riparian habitats (Melis et al. 2011). Because sandbars are one of the natural resources addressed by the GCPA that are to be protected, mitigated, and, where possible, improved, the LTEMP EIS uses current comprehensive scientific data and modeling to consider possible improvements related to the use of high-flow experiments (HFEs), as well as possible intervening flow operations, that may help better achieve the goal of retaining sandbars.

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

Since the 1995 EIS, the status of the endangered humpback chub has continued to be an issue of concern since the population in Grand Canyon, the largest in existence, declined during the late 1990s coincident with cooler water temperatures and high nonnative trout abundance and other factors. The cause of the decline was probably the result of a combination of factors, and based on Age-Structured Mark Recapture (ASMR) and multi-state models (see Figure 8), the species has since rebounded over the last decade, when water temperatures were warmer and trout abundance lower (Yackulic et al. 2014; Yard et al. 2011). Until recently, Rainbow trout have been the nonnative species of concern, but brown trout and warmwater nonnatives are becoming or are likely to become an additional concern. Uncertainty remains as to future humpback chub population response to changes in flows, nonnative species, and water temperatures that are largely driven by reservoir elevation.

There is concern among managers and scientists that nonnative fish compete with or prey upon the native or endangered fish to varying degrees (Yard et al. 2011; Whiting et al. 2014; Gloss and Coggins 2005). The effects of dam operations were examined in the 1995 EIS, and much additional information has been accumulated about the effects of dam operations on native and nonnative fish since that time. The LTEMP EIS applies the best available science and modeling methods to further consider the impacts of a variety of dam operations on native and nonnative fish and to determine what future experimentation is needed regarding these flow regimes to reduce the negative interactions of nonnative fish with native fish.

In addition to humpback chub, other important fish fauna found in the Colorado River below Glen Canyon Dam include razorback sucker (also listed as endangered), and three other native fish, flannelmouth sucker (*Catostomus latippinis*), bluehead sucker (*Catostomus discobolus*), and speckled dace (*Rhinichthys osculus*). Razorback sucker were thought to be extirpated from the Grand Canyon, but adults (Arizona Game and Fish Department [AZGFD], 2012 Data) and larval razorback sucker (Albrecht et al. 2014) have recently been found in the western Grand Canyon. Populations of bluehead and flannelmouth suckers have fluctuated in the Colorado River in the Grand Canyon since the 1995 EIS (Rogowski et al. 2015).

1.1 SYNOPSIS OF THE PROPOSED ACTION

The LTEMP EIS proposed action (Alternative D) affects monthly, daily, and hourly releases from Glen Canyon Dam but does not affect annual water release determinations. Under the proposed action (Alternative D of the EIS), the total monthly release volume of October, November, and December would be equal to that under Modified Low Fluctuating Flows (MLFF), the current operations, to avoid the possibility of the operational tier differing from that of MLFF, as established in the Interim Guidelines (Reclamation 2007a). The August volume was set to a moderate volume level (800 kaf in an 8.23-maf release year) to balance sediment conservation prior to a potential fall HFE, and power production and capacity concerns. January through July monthly volumes were set at levels that roughly track Western Area Power Administration's (WAPA's) contract rate of power delivery. This produced a redistribution of flows. Under the proposed action, the allowable within-day fluctuation range from Glen Canyon Dam would be proportional to the volume of water scheduled to be released during the month

 $(10 \times \text{monthly volume in kaf in the high-demand months of June, July, and August and 9 \times \text{monthly volume in kaf in other months}), with a maximum daily fluctuation of 8,000 cfs. The down-ramp rate would be increased to 2,500 cfs/hr, but the up-ramp rate would remain unchanged at 4,000 cfs/hr.$

Experimentation under the proposed action includes testing the effects of the following actions:

- 1. Sediment-triggered spring and fall HFEs through the entire 20-year LTEMP period;
- 2. 24-hr proactive spring HFEs in high volume years (≥ 10 maf release volume);
- 3. Extension of the duration of up to 45,000 cfs fall HFEs for as many as 250 hr depending on sediment availability;
- 4. Proactive conservation actions for humpback chub if triggered by declines in humpback chub abundance and mechanical removal of nonnative fish near the Little Colorado River confluence, if proactive conservation actions fail to arrest humpback chub population decline;
- 5. Trout management flows;
- 6. A test of low summer flows in the second 10 years of the LTEMP period to allow greater warming; and
- 7. Sustained low flows to improve the aquatic food base.

A wide range of possible hydrologic conditions will occur over the LTEMP implementation time frame in response to intra-annual and inter-annual variability in basin-wide precipitation cycles. Within a year, monthly operations are typically adjusted (increased or decreased) based on changing annual runoff forecasts, and, since 2007, application of the Interim Guidelines for Lower Basin Shortages and Coordinated Operations for Lake Powell and Lake Mead (Reclamation 2007a).

1.2 DESCRIPTION OF THE FEDERAL ACTION

1.2.1 Regulatory Context

Past consultations have evaluated the impact of proposed actions on the threatened and endangered species that live in the Colorado River and its floodplain between Glen Canyon Dam and at the time, Separation Canyon, near the inflow area of Lake Mead, Coconino and Mohave Counties, northern Arizona. This BA focuses on the area of the Colorado River from Glen Canyon Dam to Lake Mead below Pierce Ferry Rapid (river mile [RM] 280, [451 km]), although the impacts of some elements of the proposed action could extend downstream and upstream of this area, and into tributary streams. The anticipated area of effect lies within the State of Arizona and in LMNRA, GCNRA, and GCNP. The area is bordered by, or is in proximity to, the Navajo Nation, Havasupai, and Hualapai tribal lands.

1.2.2 Action Area

The action area for this proposed federal action is the Colorado River corridor from Glen Canyon Dam in Coconino County, Arizona, downstream to the Colorado River Inflow in Lake Mead. The action area includes the area potentially affected by implementation of the LTEMP (normal and experimental operations of Glen Canyon Dam and non-flow actions). This area includes Lake Powell, Glen Canyon Dam, and the river downstream to Lake Mead (see Figure 1 for the locations of Lakes Powell and Mead, Glen Canyon Dam, the Colorado River, and adjacent lands). More specifically, the scope primarily encompasses the Colorado River Ecosystem, which includes the Colorado River mainstream corridor and interacting resources in associated riparian and terrace zones, located primarily from the forebay of Glen Canyon Dam to the western boundary of GCNP. It includes the area where dam operations impact physical,



FIGURE 1 Locations of Glen Canyon Dam, Lake Powell, and the Colorado River between Lake Powell and Lake Mead, and Adjacent Lands

biological, recreational, cultural, and other resources. Portions of GCNRA, GCNP, and LMNRA are included within this area.

1.2.3 Project Description

The DOI has identified several primary objectives of operating Glen Canyon Dam under the LTEMP, as well as more specific goals to improve resources within the Colorado River ecosystem (primarily from Glen Canyon Dam downstream to the headwaters of Lake Mead) through experimental and management actions. These objectives and resource goals were considered in the formulation and development of alternatives in this EIS.

The following is a list of the objectives of the LTEMP:

- Develop an operating plan for Glen Canyon Dam in accordance with the GCPA to protect, mitigate adverse impacts on, and improve the values for which GCNP and GCNRA were established, including, but not limited to, natural and cultural resources and visitor use, and to do so in such a manner as is fully consistent with and subject to the Colorado River Compact, the Upper Colorado River Basin Compact, the Water Treaty of 1944 with Mexico, the decree of the U.S. Supreme Court in Arizona v. California, and the provisions of the Colorado River Storage Project Act (CRSPA) of 1956 and the Colorado River Basin Project Act of 1968 that govern the allocation, appropriation, development, and exportation of the waters of the Colorado River Basin and in conformance with the Criteria for Coordinated Long-Range Operations of Colorado River Basin Shortages and Coordinated Operations for Lake Powell and Lake Mead.
- Ensure water delivery to the communities and agriculture that depend on Colorado River water consistent with applicable determinations of annual water release volumes from Glen Canyon Dam made pursuant to the Long-Range Operating Criteria for Colorado River Basin Reservoirs, which are currently implemented through the 2007 Interim Guidelines for Lower Basin Shortages and Coordinated Operations for Lake Powell and Lake Mead.
- Consider potential future modifications to Glen Canyon Dam operations and other flow and non-flow actions to protect and improve downstream resources.
- Maintain or increase Glen Canyon Dam electric energy generation, load following capability, and ramp rate capability, and minimize emissions and costs to the greatest extent practicable, consistent with improvement and long-term sustainability of downstream resources.
- Respect the interests and perspectives of American Indian Tribes.

- Make use of the latest relevant scientific studies, especially those conducted since 1996.
- Determine the appropriate experimental framework that allows for a range of programs and actions, including ongoing and necessary research, monitoring, studies, and management actions in keeping with the adaptive management process.
- Identify the need for a Recovery Implementation Program for endangered fish species below Glen Canyon Dam.
- Ensure Glen Canyon Dam operations are consistent with the GCPA, ESA, National Historic Preservation Act (NHPA), CRSPA, and other applicable federal laws.

Reclamation and the National Park Service (NPS) developed resource goals considering public input and desired future conditions (DFCs) previously adopted by the Glen Canyon Dam Adaptive Management Work Group (AMWG). The following resource goals were identified:

- 1. *Archaeological and Cultural Resources*. Maintain the integrity of potentially affected *National Register of Historic Places* (NRHP)-eligible or listed historic properties in place, where possible, with preservation methods employed on a site-specific basis.
- 2. *Natural Processes*. Restore, to the extent practicable, ecological patterns and processes within their range of natural variability, including the natural abundance, diversity, and genetic and ecological integrity of the plant and animal species native to those ecosystems.
- 3. *Humpback Chub*. Meet humpback chub recovery goals, including maintaining a self-sustaining population, spawning habitat, and aggregations in the humpback chub's natural range in the Colorado River and its tributaries below the Glen Canyon Dam.
- 4. *Hydropower and Energy*. Maintain or increase Glen Canyon Dam electric energy generation, load following capability, and ramp rate capability, and minimize emissions and costs to the greatest extent practicable, consistent with improvement and long-term sustainability of downstream resources.
- 5. *Other Native Fish.* Maintain self-sustaining native fish species populations and their habitats in their natural ranges on the Colorado River and its tributaries.

- 6. *Recreational Experience*. Maintain and improve the quality of recreational experiences for the users of the Colorado River ecosystem. Recreation includes, but is not limited to, flatwater and whitewater boating, river corridor camping, and angling in Glen Canyon.
- 7. *Sediment*. Increase and retain fine sediment volume, area, and distribution in the Glen, Marble, and Grand Canyon reaches above the elevation of the average base flow for ecological, cultural, and recreational purposes.
- 8. *Tribal Resources*. Maintain the diverse values and resources of traditionally associated Tribes along the Colorado River corridor through Glen, Marble, and Grand Canyons.
- 9. *Rainbow Trout Fishery*. Achieve a healthy high-quality recreational rainbow trout fishery in GCNRA and reduce or eliminate downstream trout migration consistent with NPS fish management and ESA compliance.
- 10. *Nonnative Invasive Species*. Minimize or reduce the presence and expansion of aquatic nonnative invasive species.
- 11. *Riparian Vegetation*. Maintain native vegetation and wildlife habitat, in various stages of maturity, such that they are diverse, healthy, productive, self-sustaining, and ecologically appropriate.

The preferred alternative in the LTEMP EIS is Alternative D, which was developed by the DOI after a full analysis of the other six LTEMP alternatives had been completed. Alternative D adopted operational and experimental characteristics from other alternatives, after the effects of operations under other alternatives were modeled, and the results of that modeling suggested ways in which characteristics of each could be combined and modified to improve performance and reduce impacts, while meeting the purpose, need, and objectives of the LTEMP EIS. Alternative D features condition-dependent flow actions (operations at Glen Canyon Dam) and non-flow actions that would be triggered by resource conditions. Operations under the preferred alternative would use only existing dam infrastructure. There are a number of experimental and management actions that are a part of the preferred alternative; these are described briefly in the list below.

- Spring and fall HFEs would be implemented when triggered to rebuild sandbars. These HFEs include sediment-triggered HFEs in spring or fall, proactive spring HFEs as triggered by high annual release volume (> 10 maf), and extended duration (>96 hr) fall HFEs (Section 1.2.9).
- Nonnative fish control actions would be implemented if the Little Colorado River humpback chub population declined, and proactive conservation actions had failed to reverse declining populations. Two different tiers of population metrics would be used to trigger responses such as actions to increase growth and survival of humpback chub (Tier 1), or mechanical nonnative fish control

(Tier 2), which would only be implemented when Tier 1 actions fail to slow or reverse the decline in the humpback chub population (Section 1.2.10; Appendix D).

- Experimental trout management flows (TMFs) could be used to control annual trout production in the Glen Canyon reach for the purposes of managing the trout fishery and for limiting emigration to Marble Canyon and the Little Colorado River reach. Trout management flows will be tested early in the experimental period, preferably in the first 5 years (Section 1.2.10).
- Low summer flows may be tested in the second 10 years of the LTEMP period, for the purpose of achieving warmer river temperatures (> 14°C) to benefit humpback chub and other native species. Under low summer flows, daily fluctuations would be less than under base operations (e.g., approximately 2,000 cfs) (Section 1.2.10).
- Low steady weekend flows for macroinvertebrate production, "bug flows." The primary objective of bug flows is to test whether these flows will increase insect abundance. On an experimental basis, for example, flows would be held low and steady for 2 days per week (weekends) from May through August to attempt to improve the productivity of the aquatic food base, and increase the diversity and abundance of mayflies (Ephemeroptera), stoneflies (Plecoptera), and caddisflies (Trichoptera), referred to as EPT. While seeing EPT in the canyon is a worthwhile goal, it is a very long-term goal and unlikely to occur during initial experimentation (Section 1.2.11).
- Preservation of historic properties through a program of research, monitoring, and mitigation to address erosion and preservation of archeological and ethnographic sites and minimize loss of integrity at National Register historic properties.
- Continued adaptive management under the Glen Canyon Dam Adaptive Management Program (GCDAMP), including a research and monitoring component. Research projects have been developed on a 3-year funding cycle under the direction of the GCDAMP, which would continue under the proposed action. The level of fisheries monitoring will be expected to remain the same, or increase slightly, in order to address uncertainties and monitor responses to experiments and changed operations.

In addition, the joint-lead agencies have established "sideboards" that constrain the breadth and nature of the implementation of flow and non-flow actions of the proposed action.

1.2.4 Conservation Measures

Conservation measures identified in the 2011 Biological Opinion (BO) on operations of Glen Canyon Dam (FWS 2011a) included the establishment of a humpback chub refuge, evaluation of the suitability of habitat in the lower Grand Canyon for the razorback sucker, and establishment of an augmentation program for the razorback sucker, if appropriate. Other measures include humpback chub translocation; Bright Angel Creek brown trout control; Kanab ambersnail monitoring; determination of the feasibility of flow options to control trout, including increasing daily down-ramp rates to strand or displace age-0 trout, and high flow followed by low flow to strand or displace age-0 trout; assessments of the effects of actions on humpback chub populations; sediment research to determine effects of equalization flows; and Asian tapeworm (Bothriocephalus acheilognathi) monitoring. Most of these conservation measures are ongoing and are elements of existing management practices (e.g., brown trout control, humpback chub translocation, and sediment research to determine the effects of equalization flows), while new conservation measures or adjustments to the existing ones have been developed for the proposed action. These include experimental and management actions at specific sites such as nonnative plant removal, revegetation with native species, and mitigation at specific and appropriate cultural sites.

Five species listed under the ESA are addressed in the BA: humpback chub (*Gila cypha*), razorback sucker (*Xyrauchen texanus*), Kanab ambersnail (*Oxyloma kanabensis*), southwestern willow flycatcher (*Empidonax traillii extimus*), and Yuma Ridgway's rail (*Rallus longirostris yumanensis*). Conservation measures to minimize or reduce the effects of the proposed action or benefit or improve the status of listed species as part of the LTEMP are organized by species below.

1.2.4.1 Humpback Chub

Ongoing actions for the humpback chub include the following:

- Reclamation will continue to support the NPS, FWS, Grand Canyon Monitoring and Research Center (GCMRC), and the GCDAMP in funding and implementing translocations of humpback chub into tributaries of the Colorado River in Marble and Grand Canyons, and in monitoring the results of these translocations, consistent with agencies' plans and guidance (e.g., NPS Comprehensive Fisheries Management Plan [CFMP], FWS Humpback Chub Genetics Management Plan, and Translocation Framework, and GCMRC Triennial Work Plan). Specifically, the following will occur:
 - Humpback chub will be translocated from the lower reaches of the Little Colorado River to areas above Chute Falls in an effort to increase growth rates and survivorship.
 - Monitoring will be conducted annually, or as needed, depending on the data required, to determine survivability, population status, or genetic integrity of the Havasu humpback chub population. Intermittent translocations of additional humpback chub in Havasu Creek will be

conducted if the FWS and NPS determine it is necessary to maintain genetic integrity of the population.

- Reclamation will continue to fund a spring and fall population estimate annually, using a mark recapture based model for the Little Colorado River or the most appropriate model developed for the current collecting techniques and data.
- Reclamation will continue to fund control or removal of nonnative fish in tributaries prior to chub translocations depending on the existing fish community in each tributary. Reclamation, NPS, and FWS will lead any investigation into the possibility of using a chemical piscicide, or other tools, as appropriate. Tributaries and the appropriate control methods will be identified by the FWS, NPS, Reclamation, and GCMRC, in consultation with the AZGFD. Depending on the removal methods identified, additional planning and compliance (e.g., National Environmental Policy Act [NEPA], NHPA) may be necessary.
- Reclamation will continue to fund the FWS in maintenance of a humpback chub refuge population at a federal hatchery (Reclamation has assisted the FWS in creating a humpback chub refuge at the Southwestern Native Aquatic Resources and Recovery Center [SNARRC]) or other appropriate facility by providing funding to assist in annual maintenance (including the collection of additional humpback chub from the Little Colorado River for this purpose). In the unlikely event of a catastrophic loss of the Grand Canyon population of humpback chub, a humpback chub refuge will provide a permanent source of sufficient numbers of genetically representative stock for repatriating the species.
- Reclamation will continue to assist the FWS, NPS and the GCDAMP to ensure that a stable or upward trend of humpback chub mainstem aggregations can be achieved by:
 - Continuing to conduct annual monitoring of the Little Colorado River humpback chub aggregation (e.g., juvenile chub monitoring [JCM] parameters). Periodically, an open or multistate model should be run to estimate abundance of the entire Little Colorado River aggregation inclusive of mainstem fish.
 - Supporting annual monitoring in the mainstem Colorado River to determine status and trends of humpback chub and continuing to investigate sampling and analytical methods to estimate abundance of chub in the mainstem.
 - Conducting periodic surveys to identify additional aggregations and individual humpback chub.
 - Evaluating existing aggregations and determining drivers of these aggregations, for example, recruitment, natal origins, spawning locations,

and spawning habitat (e.g., consider new and innovative methods such as telemetry or the Judas-fish approach) (Keggeries et al. 2015).

 Exploring means of expanding humpback chub populations outside of the Little Colorado River Inflow aggregation. Evaluate the feasibility of mainstem augmentation of humpback chub that would include larval collection, rearing, and release into the mainstem at suitable areas outside of or within existing aggregations.

An ongoing action for the humpback chub that needs enhancement is the following:

• Reclamation will, through the GCDAMP, conduct disease and parasite monitoring in humpback chub and other fishes in the mainstem Colorado. The U.S. Geological Survey (USGS) and GCMRC are currently conducting parasite monitoring in the Little Colorado River. However, in order to better understand how/if disease and parasites (primarily Asian tapeworm) are affecting chub and how temperature differences may affect parasite occurrence, this work will be expanded to include investigations of parasites in humpback chub (and surrogate fish if necessary) in the mainstem.

New actions for the humpback chub include the following:

- Reclamation will collaborate with the FWS, GCMRC, NPS, and the Havasupai Tribe to conduct preliminary surveys and a feasibility study for translocation of humpback chub into Upper Havasu Creek (above Beaver Falls). The implementation of surveys and translocations, following the feasibility study, will be dependent on interagency discussions, planning and compliance, and resulting outcomes of Tribal consultation under Section 106 of the NHPA.
- Reclamation will, in cooperation with the FWS, NPS, GCMRC, and AZGFD, explore and evaluate other tributaries for potential translocations.

1.2.4.2 Razorback Sucker

Ongoing actions for the razorback sucker include the following:

- Reclamation will continue to assist the NPS, FWS, and the GCDAMP in funding larval and small-bodied fish monitoring in order to:
 - Determine the extent of hybridization in flannelmouth and razorback sucker larvae collected in the western Grand Canyon through genetic analysis.

- Determine habitat use and distribution of different life stages of razorback sucker to assist in future management of flows that may help conserve the species. Sensitive habitats to flow fluctuations could be identified and prioritized for monitoring.
- Assess the effects of TMFs and other dam operations on razorback sucker.

1.2.4.3 Nonnative Species – Removal and Control Actions (for all native aquatic species)

Nonnative fish are important stressors contributing to the decline of native and endangered fish species in the Colorado River. While many of these species are ubiquitous and likely impossible to remove or control, actions targeted on particular species and locations can decrease nonnative effects on endangered species. Reclamation maintains, through the GCDAMP, the experimental fund that retains funding for emergency needs such as removal of nonnative fish.

Reclamation will continue, in cooperation with, the NPS, FWS, AZGFD, GCMRC, and GCDAMP, funding and implementation of the following ongoing actions:

- Reclamation, in collaboration with the NPS and FWS, and in consultation
 with the AZGFD, will investigate the possibility of renovating Bright Angel
 and Shinumo Creeks with a chemical piscicide, or other tools, as appropriate.
 Additional planning and compliance, and Tribal consultation under Section
 106 of the NHPA, would be required. This feasibility study is outlined in the
 CFMP (see "Feasibility Study for Use of Chemical Fish Control Methods").
- Reclamation will continue to fund efforts of the GCMRC and NPS to remove brown trout (and other nonnative species) from Bright Angel Creek and the Bright Angel Creek Inflow reach of the Colorado River, and from other areas where new or expanded spawning populations develop, consistent with the NPS CFMP. After 5 years of removal efforts are completed (in 2017), an analysis of success will be conducted. Piscicides may be considered for removal of nonnative species if determined to be appropriate and following completion of the necessary planning and compliance actions.

Reclamation will continue, in cooperation with, the NPS, FWS, AZGFD, GCMRC, and the GCDAMP, funding and implementation of the following new actions:

• Reclamation will explore the efficacy of a temperature control device at the dam to respond to potential extremes in hydrological conditions due to climate change and could result in nonnative fish establishment. Evaluations would be ongoing for all current and evolving technological advances that could provide for warming and cooling the river in both high- and low-flow discharge scenarios, high and low reservoir levels. These studies should

include evaluating and pursuing new technologies, an analysis of the feasibility, and a risk assessment and cost analysis for any potential solutions.

- Reclamation will pursue means of preventing the passage of deleterious invasive nonnative fish through Glen Canyon Dam. Because Glen Canyon Dam release temperatures are expected to be warmer under low reservoir elevations that may occur through the LTEMP period, options to hinder expansion of warmwater nonnative fishes into Glen and Grand Canyons will be evaluated. Potential options to minimize or eliminate passage through the turbine or bypass intakes, or minimize survival of nonnative fish that pass through the dam will be assessed (flows, provide cold water, other). While feasible options may not currently exist, technology may be developed in the LTEMP period that could help achieve this goal.
- Reclamation will, in consultation with the FWS and AZGFD, fund the NPS and GCMRC on the completion of planning and compliance to alter the backwater slough at RM-12, making it unsuitable or inaccessible to warmwater nonnative species. Depending on the outcome of NPS planning and compliance, Reclamation will implement the plan in coordination with the FWS, AZGFD, NPS and GCMRC. Additional coordination will be conducted to determine and access any habitats that may support warmwater nonnatives.
- Reclamation will support the GCMRC and NPS in consultation with the FWS and AZGFD on the completion of planning and compliance of a plan for implementing rapid response control efforts for newly establishing or existing deleterious invasive nonnative species within and contiguous to the action area. Control efforts may include chemical, mechanical, or physical methods. While feasible options may not currently exist, new technology or innovative methods may be developed in the LTEMP period that could help achieve this goal. Rapid response to new warmwater fish invasions may become a more frequent need in the future with lower reservoir elevations and warmer dam releases.
- Reclamation, in cooperation with the GCDAMP, will explore the use of flow (e.g., TMFs) to inhibit brown trout spawning and recruitment in Glen Canyon, or other mainstem locations.

1.2.4.4 Southwestern Willow Flycatcher and Yuma Ridgway's Rail

Reclamation will fund the NPS to conduct southwestern willow flycatcher and Yuma Ridgway's rail presence/absence, nest surveys, and on-the-ground monitoring of habitat throughout the action area throughout the life of the LTEMP. Specifically, the following will occur:

- Reclamation will partially assist in funding NPS to conduct Yuma Ridgway's rail surveys once every 3 years for the life of the LTEMP.
- Reclamation will partially assist in funding NPS to conduct southwestern willow-flycatcher surveys once every 2 years for the life of the LTEMP.

1.2.5 Base Operations

Under the proposed action, monthly water release volumes correspond closely to the monthly hydropower demand as shown in Table 2. The total monthly release volume of October, November, and December is equal to that under current operations (MLFF) to minimize the possibility of the operational tier differing from that of current operations, as established in the Interim Guidelines (Reclamation 2007a). The distribution of monthly release volumes under the proposed action results in a more even distribution of flows over the course of a water year relative to existing, or baseline, MLFF.

Under base operations of the proposed action, the allowable daily fluctuation range from Glen Canyon Dam (i.e., the difference between the minimum and maximum flows within a day) is proportional to the volume of water scheduled to be released during the month ($10 \times$ monthly volume in thousand acre-feet (kaf) in the high-demand months of June, July, and August and $9 \times$ monthly volume in kaf in other months; Table 2; Figure 2). For example, the daily fluctuation range in July with a scheduled release volume of 800 kaf would be 8,000 cfs, and the daily fluctuation range in December with the same scheduled release volume would be 7,200 cfs. The maximum allowable daily fluctuation range in flows in any month would be 8,000 cfs, which is the same as under current operations. However, the down-ramp rate limit under the proposed action would be increased by 1,000 cfs/hr, to 2,500 cfs/hr. The up-ramp rate would remain the same at 4,000 cfs/hr. Figure 2 shows minimum, mean, and maximum daily flows in an 8.23-milion acre-foot (maf) year, assuming all days in a month adhere to the same mean daily flow within a month. Figure 3 shows the hourly flows in a simulated 8.23-maf year within the constraints of the proposed action.

1.2.6 Operational Flexibility

Reclamation retains the authority to utilize operational flexibility at Glen Canyon Dam because hydrologic conditions of the Colorado River Basin (or the operational conditions of Colorado River reservoirs) cannot be completely known in advance. Consistent with current operations, Reclamation, in consultation with WAPA, will make specific adjustments to daily and monthly release volumes during the water year. Monthly release volumes may be rounded for practical implementation or for maintenance needs. In addition, when releases are actually implemented, minor variations may occur regularly for a number of operational reasons that cannot be projected in advance.

Month	Monthly Release Volume (kaf)	Proportion of Total Annual Volume	Mean Daily Flow (cfs)	Daily Fluctuation Range (cfs)
				<u> </u>
October	643	0.0781	10,451	5,783
November	642	0.0780	10,781	5,774
December	716	0.0870	11,643	6,443
January	763	0.0928	12,415	6,871
February	675	0.0820	12,157	6,076
March	713	0.0866	11,596	6,417
April	635	0.0772	10,672	5,715
May	632	0.0768	10,278	5,688
June	663	0.0806	11,142	6,630
July	749	0.0910	12,181	7,490
August	800	0.0972	13,011	8,000
September	600	0.0729	10,083	5,400

TABLE 2 Flow Parameters under the Proposed Action in an 8.23-mafYear^a

^a Within a year, monthly operations may be increased or decreased based on changing annual runoff forecasts and other factors, such as application of the Long-Range Operating Criteria for Colorado River Basin Reservoirs, which are currently implemented through the 2007 Interim Guidelines for Lower Basin Shortages and Coordinated Operations for Lake Powell and Lake Mead (Reclamation 2007a).



FIGURE 2 Mean, Minimum, and Maximum Daily Flows under the Proposed Action in an 8.23-maf Year



FIGURE 3 Simulated Hourly Flows under the Proposed Action in an 8.23-maf Year (Note that there are differences in the mean, maximum, and minimum flows shown here and in Figure 2. These differences reflect flexibility in operational patterns allowed within the constraints of the proposed action.)

Reclamation also will make specific adjustments to daily and monthly release volumes, in consultation with other entities as appropriate, for a number of reasons, including operational, resource-related, and hydropower-related issues. Examples of these adjustments may include, but are not limited to, the following:

- For water distribution purposes, volumes may be adjusted to allocate water between the Upper and Lower Basins, consistent with the Law of the River as a result of changing hydrology;
- For resource-related issues that may occur uniquely in a given year, release adjustments may be made to accommodate nonnative species removal, to assist with aerial photography, or to accommodate other resource considerations separate from experimental treatments under the LTEMP; and
- For hydropower-related issues, adjustments may occur to address issues such as electrical grid reliability, actual or forecasted prices for purchased power, transmission outages, and experimental releases from other Colorado River Storage Project dams.

In addition, Reclamation may make modifications under circumstances that may include operations that are prudent or necessary for the safety of dams, public health and safety, other emergency situations, or other unanticipated or unforeseen activities arising from actual operating experience (including, in coordination with the Basin States, actions to respond to low reservoir conditions as a result of drought in the Colorado River Basin). Also, the Emergency Exception Criteria established for Glen Canyon Dam will continue under the proposed action. (See, e.g., Section 3 of the Glen Canyon Operating Criteria at 62 FR 9448, March 3, 1997.)

1.2.7 Implementation of Experimental Elements

The proposed action identifies condition-dependent flow and non-flow treatments intended to safeguard against unforeseen adverse changes in resource impacts, and to prevent irreversible changes to those resources. These condition-dependent treatments will be implemented experimentally during the LTEMP period unless they prove ineffective or result in unacceptable adverse impacts on other resources. Experimental treatments include sediment treatments, triggered humpback chub conservation actions, TMFs, mechanical removal of nonnative fish, low summer flows, macroinvertebrate production flows, and non-flow-related vegetation treatments.

1.2.8 Overall Implementation Process for Experiments

Prior to implementation of any experiment, the relative effects of the experiment on the following resource areas will be evaluated and considered: (1) water quality and water delivery, (2) humpback chub, (3) sediment, (4) riparian ecosystems, (5) historic properties and traditional cultural properties, (6) Tribal concerns, (7) hydropower production and WAPA's assessment of the status of the Basin Fund, (8) the rainbow trout fishery, (9) recreation, and (10) other resources. Although these key resources are listed for consideration on a regular basis, the DOI intends to retain sufficient flexibility in implementation of experiments to allow for response to unforeseen circumstances or events that involve any other resources not listed here. The recent discovery of nonnative green sunfish in the Glen Canyon reach illustrates the need to be responsive to unforeseen conditions.

The proposed approach differs fundamentally from a more formal experimental design (e.g., before-after control-impact design, factorial design) that attempts to resolve uncertainties by controlling for or treating potentially influential or confounding factors. There are several reasons to avoid such a formal design and instead focus on the condition-dependent approach described here. Among these are (1) the difficulties in controlling for specific conditions in a system as complex as the Colorado River; (2) wide variability in temperature and flow conditions that are important drivers in ecological processes; (3) inherent risk of some experimentation to protected sensitive resources, in particular, endangered humpback chub; (4) conflicting multiple-use values and objectives; and (5) low expected value-of-information for the uncertainties that could be articulated, and around which a formal experimental design would be established. For these reasons, the proposed action includes a condition-dependent adaptive approach.

O-32

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

Table 3 provides the implementation criteria for condition-dependent experimental treatments of the proposed action, and triggers for treatments are discussed under the description below. Triggers for experimental changes in operations, implementation considerations for determining if an experimental treatment should proceed, conditions that would cause the treatment to be terminated prior to completion (i.e., off-ramps), and the number of replicates that are initially considered needed are discussed below. In many cases, two to three replicates of an experimental treatment are considered necessary. The results of these tests will be used to determine if these condition-dependent treatments should be retained as part of the suite of longterm actions implemented under LTEMP. In other cases, implementation of experimental treatments would continue throughout the LTEMP period if triggered (e.g., spring and fall HFEs), except in years when it was determined that the proposed experiment could result in unacceptable adverse impacts on resource conditions. For these experiments, effectiveness would be monitored and the experiments would be terminated or modified only if sufficient evidence suggested the treatment was ineffective or had unacceptable adverse impacts on other resources. All experimental treatments will be closely monitored for adverse side effects on important resources. At a minimum, an unacceptable adverse impact would include significant negative impacts on resources as a result of experimental treatments that have not been analyzed for the proposed action in the LTEMP EIS.

In implementing the processes described here and the associated decision process shown in Table 3, the DOI will exercise a formal process of stakeholder engagement to ensure decisions are made with sufficient information regarding the condition and potential effects on important resources. As an initial platform to discuss potential future experimental actions, the DOI will hold GCDAMP annual reporting meetings for all interested stakeholders; these meetings will present the best available scientific information and learning from previously implemented experiments and ongoing monitoring of resources. As a follow-up to this process, the DOI will meet with the Technical Working Group (TWG) to discuss the experimental actions being contemplated for the year.

The DOI also will conduct monthly Glen Canyon Dam operational coordination meetings or calls with the DOI bureaus (USGS, NPS, FWS, Bureau of Indian Affairs [BIA], and Reclamation), WAPA, AZGFD, and representatives from the Basin States and the Upper Colorado River Commission (UCRC). Each DOI bureau will provide updates on the status of resources and dam operations. In addition, WAPA will provide updates on the status of the Basin Fund, projected purchase power prices, and its financial and operational considerations. These meetings or calls are intended to provide an opportunity for participants to share and obtain the most up-to-date information on dam operational considerations and the status of resources (including ecological, cultural, Tribal, recreation, and the Basin Fund). One liaison from each Basin State and from the UCRC will be allowed to participate in the monthly operational coordination meetings or calls.

Experimental Treatment	Trigger and Primary Objective	Replicates	Duration	Annual Implementation Considerations ^a	Long-Term Off-Ramp Conditions	Action, If Successful
Sediment Treatme	ents					
Spring HFE up to 45,000 ft ³ /s (cfs) (in Mar. or Apr.)	Trigger: Sufficient Paria River sediment input in spring accounting period (Dec.–Jun.) to achieve a positive sand mass balance in Marble Canyon with implementation of an HFE Objective: Rebuild sandbars	Not conducted during first 2 years of LTEMP, otherwise, implement in each year triggered, dependent on resource condition and response	≤96 hr	Potential short-term unacceptable impacts on the resources listed in footnote (a); unacceptable cumulative effects of sequential HFEs; sediment-triggered spring HFEs will not occur in the same water year as an extended-duration (>96 hr) fall HFE	Sediment-triggered spring HFEs are not effective in building sandbars, or long- term unacceptable adverse impacts on the resources listed in footnote (a) are observed	Implement as adaptive treatment when triggered and resource conditions allow
Proactive spring HFE up to 45,000 cfs (Apr., May, or Jun.)	Trigger: High-volume year with planned equalization releases (≥10 maf) Objective: Protect sand supply from equalization releases	Not conducted during first 2 years of LTEMP, otherwise, implement in each year triggered, dependent on resource condition and response	First test 24 hr; subsequent tests could be shorter, but not longer, depending on results of first tests	Potential short-term unacceptable impacts on the resources listed in footnote (a); unacceptable cumulative effects of sequential HFEs; would not be implemented in same water year as a sediment-triggered spring HFE or extended duration fall HFE	Proactive spring HFEs are not effective in building sandbars; or long-term unacceptable adverse impacts on the resources listed in footnote (a) are observed	Implement as adaptive treatment when triggered and existing resource conditions allow
Fall HFEs \leq 96 up to 45,000 cfs in Oct. or Nov.	Trigger: Sufficient Paria River sediment input in fall accounting period (Jul.– Nov.) to achieve a positive sand mass balance in Marble Canyon with implementation of an HFE	Implement in each year triggered dependent on resource condition and response	≤96 hr	Potential unacceptable impacts on resources listed in footnote (a)	This type of HFE was not effective in building sandbars; or long-term unacceptable adverse impacts on the resources listed in footnote (a) are observed	Implement as adaptive treatment when triggered and existing resource conditions allow

TABLE 3 Implementation Criteria for Experimental Treatments under the Proposed Action

TABLE 3 (Cont.)

Experimental Treatment	Trigger and Primary Objective	Replicates	Duration	Annual Implementation Considerations ^a	Long-Term Off-Ramp Conditions	Action, If Successful		
Sediment Treatments (Cont.)								
Fall HFEs longer than 96-hr duration up to 45,000 cfs in Oct. or Nov.	Trigger: Sufficient Paria River sediment input in fall accounting period (Jul.– Nov.) to achieve a positive sand mass balance in Marble Canyon with implementation of an HFE longer than a 96-hr, up to 45,000-cfs flow	Implement in each year triggered; limited to total of four tests in LTEMP period	Up to 250 hr depending on availability of sand; duration of first test not to exceed 192 hr	Potential short-term unacceptable impacts on the resources listed in footnote (a); unacceptable cumulative effects of sequential HFEs.	Extended duration fall HFEs are not effective in building sandbars; resulting sandbars are no bigger than those created by shorter-duration HFEs; or long-term unacceptable adverse impacts on the resources listed in footnote (a) are observed	Implement as adaptive treatment when triggered and existing resource conditions allow		
	Objective: Rebuild sandbars							
Aquatic Resource	Treatments							
Trout management flows	Trigger: Predicted high trout recruitment in the Glen Canyon reach Objective: Test efficacy of flow regime on trout numbers and survival of chub	Implement as needed when triggered; after consultation with Tribes; test may be conducted early in the 20-year period even if not triggered by high trout recruitment ^b	Implemented in as many as 4 months (May–Aug.)	Potential unacceptable impacts on resources listed in footnote (a)	Trout management flows have little or no effect on trout recruitment after at least three tests, or long- term unacceptable adverse impacts on the resources listed in footnote (a) are observed	Implement as adaptive treatment triggered by predicted high trout recruitment in Glen Canyon, taking into consideration Tribal concerns		
Tier 1: Expanded translocation of humpback chub in the Little Colorado River	Trigger: Number of adult or subadult humpback chub in the Little Colorado River reach below tier 1 triggers Objective: Increase number of adult and subadult humpback chub	Implement in each year triggered unless determined ineffective.	As needed	Potential short-term unacceptable impacts on resources listed in footnote (a)	Expanded translocation has little or no effect on increasing the number of adult or subadult humpback chub; or long-term unacceptable adverse impacts on the resources listed in footnote (a) are observed	Implement as adaptive treatment when triggered		

TABLE 3 (Cont.)

Experimental Treatment	Trigger and Primary Objective	Replicates	Duration	Annual Implementation Considerations ^a	Long-Term Off-Ramp Conditions	Action, If Successful		
Aquatic Resource Treatments (Cont.)								
Tier 1: Implement headstart program for larval humpback chub	Trigger: Number of adult or subadult humpback chub in Little Colorado River reach below Tier 1 triggers Objective: Increase number of adult and subadult humpback chub	Implement in each year triggered unless determined ineffective	As needed	Potential short-term unacceptable impacts on resources listed in footnote (a)	Head-start program has little or no effect on increasing the number of adult or subadult humpback chub; or long-term unacceptable adverse impacts on the resources listed in footnote (a) are observed	Implement as adaptive treatment when triggered		
Tier 2: Mechanical removal of nonnative fish in Little Colorado River reach	Trigger: Tier 1 actions ineffective; humpback chub number in Little Colorado River below Tier 2 triggers Objective: Increase number of adult and sub-adult humpback chub	Implement in each year triggered unless determined ineffective, after consultation with Tribes	Monthly removal trips (Feb.–Jul.) until "predator index" or adult humpback chub reach acceptable levels (Appendix D)	Potential short-term unacceptable impacts on resources listed in footnote (a)	Mechanical removal has little or no reduction in predator index in the Little Colorado River reach; no population-level benefit on humpback chub or unacceptable adverse impacts on the resources listed in footnote (a) are observed	Implement as adaptive treatment when triggered, taking into consideration Tribal concerns		
Low summer flows (minimum daily mean 5,000 to 8,000 cfs) to target \geq 14°C at Little Colorado River confluence	Trigger: Initial experiment in the second 10 years of the LTEMP period, when target temperature of ≥14°C can only be achieved with low summer flow Objective: Increase humpback chub growth	Subsequent experimental use if (1) initial test was successful, (2) humpback chub population concerns warrant their use, (3) water temperature appears to be limiting recruitment and (4) target temperature of $\geq 14^{\circ}$ C could be achieved only with low summer flow	3 months (Jul.– Sep.)	Potential unacceptable impacts on resources listed in listed in footnote (a)	Low summer flows do not increase growth and recruitment of humpback chub; increase in warmwater nonnative species or trout at the Little Colorado River; or long- term unacceptable impacts on resources listed in footnote (a) are observed, or sufficient warming does not occur as predicted	Implement as adaptive treatment when conditions allow		

22 0-36

TABLE 3 (Cont.)

0-37

23

Experimental Treatment	Trigger and Primary Objective	Replicates	Duration	Annual Implementation Considerations ^a	Long-Term Off-Ramp Conditions	Action, If Successful		
Aquatic Resource Treatments (Cont.)								
Low steady weekend flows for macroinvertebrate production	Trigger: None Objective: Improve food base productivity and abundance or diversity of mayflies, stoneflies, and caddisflies	Target two to three replicates	Up to 4 months (May–Aug.)	Potential short-term unacceptable impacts on resources listed in footnote (a); coordinate planning with other experiments to avoid confounding conditions or results	Steady weekend flows have little or no benefit on food base, trout fishery, or native fish; increase in warmwater nonnative species or trout in the Little Colorado River reach; or long-term unacceptable adverse impacts on the impacts on resources listed in footnote (a) are observed	Implement as adaptive treatment in target months when conditions allow		
Riparian Vegetation Treatments								
Non-flow vegetation treatments	Trigger: None Objective: Improve vegetation conditions at key sites	Not applicable	20 years if successful pilot phase	Potential short-term unacceptable impacts on resources listed in footnote (a)	Control and replanting techniques are not effective or practical; or long-term unacceptable adverse impacts on the resources listed in footnote (a) are observed	Implement as adaptive treatment if invasive species can be reduced and native species increased		

^a Prior to implementation of any experiment, the relative effects of the experiment on the following resource areas will be evaluated and considered: (1) water quality and water delivery, (2) humpback chub, (3) sediment, (4) riparian ecosystems, (5) historic properties and traditional cultural properties, (6) Tribal concerns, (7) hydropower production and WAPA's assessment of the status of the Basin Fund, (8) the rainbow trout fishery, (9) recreation, and (10) other resources.

^b The decision to conduct TMFs in a given year would consider the resource conditions listed in footnote (a) and would also involve considerations regarding the efficacy of the test based on those resource conditions.

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

To determine whether conditions are suitable for implementing or discontinuing experimental treatments or management actions, the DOI will schedule implementation/planning meetings or calls with the DOI bureaus (USGS, NPS, FWS, BIA and Reclamation), WAPA, AZGFD, and one liaison from each Basin State and from the UCRC, as needed or requested by the participants. The implementation/planning group will strive to develop a consensus recommendation to bring forth to the DOI regarding resource issues as detailed at the beginning of this section, as well as including WAPA's assessment of the status of the Basin Fund. The Secretary of the Interior will consider the consensus recommendations of the implementation/planning group; however, it retains sole discretion to decide how best to accomplish operations and experiments in any given year pursuant to the LTEMP ROD and other binding obligations.

The DOI will also continue separate consultation meetings with the Tribes, AZGFD, the Basin States, and UCRC upon request, or as required under existing RODs.

Sections 1.2.9 and 1.2.10 describe specific processes for development and implementation of experiments related to sediment, aquatic resources, and riparian vegetation. The overall approach attempts to strike a balance between identifying specific experiments and providing flexibility to implement those experiments when resource conditions are appropriate. As discussed above, rather than proposing a prescriptive approach to experimentation, an adaptive management-based approach that is responsive and flexible would be used to adapt to changing environmental and resource conditions and new information. The potential for confounding interactions among individual experimental treatments is discussed when relevant for each of the proposed treatments. Given the size of the project area, and the variability inherent in the system, this pragmatic approach to experimentation is warranted. Although confounding treatments are possible given the complexity of the experimental plan, they are not expected to limit learning over the life of the LTEMP.

1.2.9 Sediment-Related Experimental Treatments

Figure 4 shows the decision tree for implementation of sediment experiments during the LTEMP period. Under the proposed action, the existing HFE protocol was updated and incorporated into the LTEMP process as specified in Appendix P of the LTEMP EIS. Changes to the existing protocol were related to implementation of the new HFEs that are included under the proposed action and an extension of the protocol to the end of the LTEMP period. This new protocol would replace the existing protocol when the LTEMP ROD is issued. Spring and fall HFEs would be implemented when triggered during the 20-year LTEMP period based on the estimated sand mass balance resulting from Paria River sediment inputs during the spring and fall accounting periods, and the dam release pattern during the accounting period. HFE releases would be 1 to 250 hr long and between 31,500 and 45,000 cfs. Depending on the cumulative amount of sediment input from the Paria River during the spring (December 1 through June 30) or fall (July 1 through November 30) accounting periods, and the expected accumulation of sand, the maximum possible magnitude and duration of HFEs that would achieve a positive sand mass balance in Marble Canyon, as determined by modeling, would be implemented.



HFE = High-Flow Experiment NEPA = National Environmental Policy Act

FIGURE 4 Decision Tree for Implementation of Sediment-Related Experimental Treatments under the Proposed Action

Sand mass balance modeling would be used to ensure that the duration and magnitude of an HFE are best matched with the mass of sand present in the system during a particular release window. The magnitude and duration of HFEs will not affect the total annual release from Glen Canyon Dam. Reclamation will consider the total water to be released in the water year when determining the magnitude and duration of an HFE.

Sediment-related experiments under the proposed action include (1) sediment-triggered spring and fall HFEs up to 96-hr duration; (2) short-duration (24-hr) proactive spring HFEs in high-volume equalization years prior to equalization releases; and (3) implementation of up to four extended-duration (>96 hr) HFEs, up to 250 hr long, depending on sediment conditions. The pattern of transferring water volumes from other months to make up the HFE volume would be discussed in the monthly Glen Canyon Dam operational coordination meetings described in Section 1.2.8.

If sediment resources are stable or improving, the combination of base operations, HFE protocols, and other treatments will continue as prescribed. If sediment resource conditions decrease to unacceptable levels during the LTEMP period, operations may be modified to the extent allowable under the LTEMP ROD or would be evaluated and considered under a separate NEPA process, potentially including additional studies of sediment augmentation or other actions.

For all sediment experiments, testing will be modified or temporarily or permanently suspended if (1) experimental treatments are ineffective at accomplishing their objectives, or (2) there were unacceptable adverse impacts on resources (Table 3). Monitoring results will be evaluated to determine whether additional tests, modification of experimental treatments, or discontinuation of experimental treatments are warranted.

Implementation of HFEs would consider resource condition assessments and resource concerns using the annual processes described in Section 1.2.8. HFEs may not be tested when there appears to be the potential for unacceptable impacts on the resources listed in Section 1.2.8. In addition, there is uncertainty associated with cumulative impacts from sequential HFEs. These cumulative impacts would be considered before implementing an HFE.

1.2.9.1 Sediment-Triggered Spring HFEs

Under the proposed action, sediment-triggered spring HFEs will be implemented after an initial 2-year delay in order to enable testing of the effectiveness of TMFs, if warranted, and address concerns raised by the apparent positive response of trout to the 2008 spring HFE (Korman et al. 2011; Melis et al. 2011). Modeling the trout response to spring HFEs for the EIS was based on relationships developed from the observed response to the 2008 spring HFE. That modeling also evaluated uncertainty related to the effectiveness of TMFs to control excess trout produced by HFEs. Modeling indicated that even at a relatively low level of effectiveness (10% reduction in trout recruitment), TMFs could effectively reduce the number of trout out-migrants from Glen Canyon to the Little Colorado River reach (RM 61) where humpback chub occur.
After the first 2 years of the LTEMP period, spring HFEs will be implemented when triggered in accordance with the HFE protocol unless an extended-duration fall HFE occurred in the same water year. Modeling indicates spring HFEs will be triggered in about 26% of the years in the LTEMP period. Sediment-triggered spring HFEs would be implemented when triggered during the entire LTEMP period unless new information indicated they were not effective in building sandbars, or there were unacceptable adverse effects on resources (Section 1.2.8).

Implementation of a spring HFE may provide important replication of the 2008 spring HFE and aid in understanding the effect of spring HFEs on the trout population. It is possible that the strong 2008 response was a result of the specific conditions present in 2008 (e.g., condition of the food base, trout population size). It is unclear whether implementation under current conditions would produce the same result, and there is a good deal of learning that could result from implementation early in the LTEMP period. Implementing a spring HFE early in the LTEMP period when chub numbers are relatively high may also be a relatively low-risk option. To provide a means of controlling trout recruitment following tests of spring HFEs, TMFs will be experimentally implemented and tested for efficacy early in the LTEMP period (see discussion of TMFs below).

Implementation of sediment-triggered spring HFEs would consider resource condition assessments and resource concerns using the processes described in Section 1.2.8. Spring HFEs may not be tested when there appears to be the potential for unacceptable adverse impacts on the resources listed in Section 1.2.8. In addition, there is uncertainty associated with the cumulative impacts of sequential HFEs on sediment, aquatic, and potentially other resources. These cumulative impacts would be considered before implementing a spring HFE particularly if a fall HFE had been implemented in the same water year.

1.2.9.2 Proactive Spring HFEs

GCMRC scientists identified proactive spring HFEs as a potential experimental treatment to transport and deposit in-channel sand at elevations above those of equalization flows. These HFEs will be tested only in years with high annual water volume (i.e., ≥ 10 maf), and modeling suggests that this would be a relatively rare treatment. A first test would be a 24-hr 45,000-cfs release conducted in April, May, or June. Duration in subsequent tests could be shortened depending on the observed response during the first tests. It would be preferable to test proactive spring HFEs at least two to three times in the 20-year LTEMP period, but being able to do so will be dependent upon annual hydrology. Modeling indicates that proactive spring HFEs would be triggered in about 10% of the years in the LTEMP period.

Proactive spring HFEs would not be tested in the first 2 years of the LTEMP period. In addition, proactive spring HFEs will not be tested in years when there had been a spring HFE or extended-duration fall HFE earlier in the same water year. Proactive spring HFEs could be performed in the same water year as a 96-hr or shorter sediment-triggered fall HFE, although prior to implementation, the potential effects of these HFEs would be carefully evaluated using the processes described in Section 1.2.8. The first test would be carefully evaluated to determine whether additional tests were warranted based on the efficacy of building and maintaining

sandbars. If initial tests show positive results without unacceptable adverse effects on the resources listed in Section 2.2.4.3, proactive spring HFEs would be implemented when triggered during the entire LTEMP period..

Implementation of proactive spring HFEs would consider resource condition assessments and resource concerns using the processes described in Section 1.2.8. Proactive spring HFEs may not be tested when there appears to be the potential for unacceptable impacts on the resources identified in Section 1.2.8. The cumulative impacts of sequential HFEs would be considered before implementing a proactive spring HFE.

1.2.9.3 Sediment-Triggered Fall HFEs

The effects of sediment-triggered fall HFEs on trout recruitment are uncertain, but fall HFEs are expected to have less effect on trout production than spring HFEs. HFEs in November 2012, 2013, and 2014 resulted in little or no increase in the number of young-of-the-year (YOY) trout (VanderKooi 2015; Winters et al. 2016), and this observation may be based on the observed resilience of the food base to disturbance in the fall (Kennedy et al. 2015). However, factors affecting trout response to fall HFEs are not well understood. Modeling for the EIS considered the effect of fall HFEs on trout and modeled fall HFEs in two ways: in one, the effect of fall HFEs was half as long as that of a spring HFE (i.e., it affected trout production only in the water year in which it occurred); in the other, fall HFEs had no effect on trout production. Modeling the effect of fall HFEs in these two ways had an effect on the overall predicted number of trout produced, the number of out-migrants, and ultimately their effect on humpback chub. The number of trout produced and out-migrants were higher, and the number of humpback chub was lower if fall HFEs were assumed to have an effect on trout production.

Modeling indicates fall HFEs would be triggered in about 77% of the years in the LTEMP period. Testing fall HFEs is considered to be a relatively low-risk treatment due to the lack of observed or documented trout response from previous fall HFEs, and testing would be implemented when triggered during the entire LTEMP period unless new information indicated fall HFEs were not effective in building sandbars, or there were unacceptable adverse effects.

Implementation of sediment-triggered fall HFEs would consider resource condition assessments and resource concerns using the processes described in Section 1.2.8. Fall HFEs may not be tested when there appears to be the potential for unacceptable impacts on the resources listed in Section 1.2.8. The cumulative impacts of sequential HFEs would be considered before implementing a sediment-triggered fall HFE.

1.2.9.4 Extended-Duration Fall HFEs

Sediment-triggered fall HFEs with durations longer than 96 hr (up to 250 hr) would be tested under the proposed action. The duration of these extended-duration fall HFEs would be based on the amount of sediment delivered from the Paria River during the fall accounting period and would be no more than the maximum magnitude and duration of an HFE that would achieve

a positive sand mass balance in Marble Canyon, as determined by modeling. Based on examination of the observed historical sediment input from the Paria River, it was determined that HFEs up to 10.4 days in length (250 hr) could be supported before exhausting seasonal sediment inputs and affecting water delivery requirements. GCMRC scientists have suggested that increasing the duration of HFEs when sediment supply can support a longer duration may lead to more sand being deposited at higher elevations, resulting in bigger sandbars. Modeling indicates that this treatment will be triggered in 25% of the years in the LTEMP period. There will be no more than four extended-duration fall HFEs over the 20-year LTEMP period.

The duration of the first implementation of an extended-duration HFE will be limited to no more than 192 hr (twice as long as the current limit of 96 hr). This duration is considered long enough to produce a measurable result if the treatment represents an effective approach to building sandbars under enriched sediment conditions. The duration of all tests will be based on available sediment, current hydrology, reviews of available information, the expert opinion of GCMRC and other Grand Canyon scientists, and consideration of potential effects on the resources listed in Section 1.2.8. If feasible, monitoring will include real-time observations of sediment concentrations to determine if sediment deposition continues throughout the duration of the extended HFEs.

Implementation of extended-duration fall HFEs would consider resource condition assessments and resource concerns using the processes described in Section 1.2.8. Extendedduration fall HFEs may not be tested when there appears to be potential unacceptable impacts on the resources listed in Section 1.2.8. Because the effects of extended-duration HFEs on Lake Mead water quality are a concern, DOI will coordinate with relevant water quality monitoring programs or affected agencies prior to implementing any test of extended-duration HFEs. The cumulative impacts of sequential HFEs would be considered before implementing an extended-duration fall HFE.

Another important concern that results from the large volume of water bypassed is water delivery. Water delivery issues would be considered before deciding to implement an extended-duration fall HFE. An extended-duration HFE would not be implemented if annual release volume would be affected. It is possible that in lower volume years there would not be sufficient water available to support an extended-duration HFE. A 250-hr extended-duration HFE would result in a monthly total release of approximately 1.2 maf. In lower volume release years (e.g., 7.0 maf or 7.48 maf), the maximum duration may be less than 250 hr. In addition, a sediment-triggered spring HFE or proactive spring HFE would not be conducted in the same water year as an extended-duration fall HFE. If an extended-duration fall HFE was triggered but not implemented for any of the reasons described above, a fall HFE 96 hr or less in duration could be implemented instead. Implementation would necessitate reducing water volume in other months of the same water year.

In order to fully test the efficacy of these longer HFEs, several replicates would be desirable in the 20 year LTEMP period. Extended-duration HFEs would be considered successful and would be continued up to a total of four times in the 20-year LTEMP period as part of an adaptive experimental treatment if there was a widespread increase in bar size relative to \leq 96-hr HFEs, and if sand mass balance was not significantly compromised relative to the

ability to maintain a long-term equilibrium. Extended-duration HFEs would not continue to be tested if they were not effective in building sandbars, if resulting total sandbar volumes were no bigger than those created by shorter-duration HFEs, or if unacceptable adverse impacts on the resources listed in Section 1.2.8 were observed.

1.2.10 Aquatic Resource-Related Experimental Treatments

Under the proposed action, most experimental flow and non-flow actions will be triggered or adjusted by considering the estimated number of humpback chub, number of rainbow trout, number of other nonnative fish, water release temperature, or a combination of these variables, depending on the action under consideration. Humpback chub triggers and trout triggers were developed in consultation with the FWS, NPS, and AZGFD. These triggers may be modified based on experimentation conducted early in the LTEMP period.

Aquatic resource experiments that will be tested include (1) TMFs, (2) Tier 1 conservation actions for humpback chub; (3) Tier 2 mechanical removal of nonnative fish, (3) low summer flows in the second 10 years of the LTEMP, and (4) macroinvertebrate production flows. Aquatic resource experiments would seek to refine our understanding of the impacts of water releases, HFEs, and TMFs on these resources. The primary uncertainty surrounding HFEs revolves around the extent to which the seasonality of HFEs or the number of adult rainbow trout determines the strength of rainbow trout recruitment.

Experimental nonnative fish control actions would be implemented if the humpback chub population declined, and proactive conservation actions had failed to reverse declining populations. Two different tiers of population metrics would be used to trigger responses including actions to increase growth and survival of humpback chub (Tier 1) and mechanical removal of nonnative fish (Tier 2); these would only be implemented when Tier 1 actions fail to slow or reverse the decline in the humpback chub population. This tiered approach and the triggers that would be used to implement it are described below.

For all aquatic resource experiments, testing will be modified or temporarily or permanently suspended if (1) experimental treatments are ineffective at accomplishing their objectives, or (2) there are potential unacceptable adverse impacts on the resources listed in Section 1.2.8. Monitoring results will be evaluated to determine whether additional tests, modification of experimental treatments, or discontinuation of experimental treatments are warranted.

Implementation of aquatic resource experiments would consider resource condition assessments and resource concerns using the processes described in Section 1.2.8. Aquatic resource experiments may not be tested when there appears to be the potential for unacceptable impacts on the resources listed in Section 1.2.8.

1.2.10.1 Trout Management Flows

Mechanical removal of nonnative fish is a controversial issue in the Colorado River through Glen and Grand Canyons. A spring 2015 meeting of Grand Canyon biologists (NPS, FWS, AZGFD, and GCMRC) to assess current removal triggers resulted in a concept of early conservation measure intervention to maximize conservation benefit to humpback chub and minimize the likelihood of mechanical predator removal. Trout management flows are a potential tool that could be used to control annual trout production in the Glen Canyon reach for the purposes of managing the trout fishery and for limiting emigration from the Glen Canyon reach to Marble Canyon and the Little Colorado River reach.

Trout management flows are a special type of fluctuating flow designed to reduce the recruitment of trout by disadvantaging YOY trout (Figure 5). Trout management flows have been proposed and developed on the basis of research described in Korman et al. (2005). The underlying premise of TMFs is based on observations that YOY trout tend to occupy nearshore shallow-water habitats to avoid predation by larger fish. A potential scenario for TMFs features repeated fluctuation cycles that consist of relatively high flows (e.g., 20,000 cfs) sustained for a period of time (potentially ranging from 2 days to 1 week) followed by a rapid drop to a low flow (e.g., 5,000 to 8,000 cfs). This low flow will be maintained for a period of less than a day (e.g., 12 hr) to prevent adverse effects on the food base. Low flows would be timed to start in the morning, after sunrise, to expose stranded fish to direct sunlight and heat. Up-ramp rates to the



FIGURE 5 Example Implementation of a Two-Cycle Trout Management Flow in June and July with Resumption of Normal Fluctuations between Cycles and Afterward (Monitoring for effectiveness would occur before and after each cycle. The horizontal line below the graph shows periods of normal fluctuation [blue] and TMFs [orange].)

peak TMF will be the same as the limit for the proposed action overall (i.e., 4,000 cfs/hr). The down-ramp from peak to base would be over a single hour (e.g., 15,000 cfs/hr for a drop from 20,000 cfs to 5,000 cfs). These flow levels and rates are not fixed in order to allow flexibility in their design. In a TMF cycle, YOY trout are expected to occupy nearshore habitat when flows are highest, and would be subsequently stranded by the sudden drop to low flow. Because older age classes of trout tend to occupy deeper habitats toward the middle of the river channel, they are less susceptible to stranding and are less likely to be directly affected by TMFs. Trout management flows will be used to control trout recruitment in the Glen Canyon reach to manage the rainbow trout fishery, and to limit emigration of juvenile trout to downstream reaches, particularly to habitat occupied by humpback chub near the confluence with the Little Colorado River. Triggers for implementation of TMFs will be determined in consultation with the AZGFD, GCMRC, NPS, and Reclamation.

If resource conditions are appropriate, TMFs may be tested early in the experimental period, preferably in the first 5 years. These first tests could be triggered by modeled trout recruitment levels or otherwise implemented to test the effectiveness of TMFs. The intent of early tests is to determine the effectiveness of TMFs and a best approach to trout management. If TMFs are determined to be effective for controlling trout numbers while minimizing impacts on other resources, they may be deployed as an adaptive experimental treatment triggered by estimated trout recruitment.

It should be noted that several Tribes have expressed concerns about TMFs as a taking of life within the canyon without a beneficial use. The Pueblo of Zuni has expressed concern that the taking of life by trout stranding has an adverse effect on the Zuni value system. The joint-lead agencies will continue to work with the Tribes regarding options for trout management and to determine the most appropriate means of mitigating impacts on Tribal values when TMFs are implemented.

As many as three TMF cycles per month in a period of up to 4 months during May through August could be tested, depending on the results of early tests. Aspects of TMF design that would be investigated include:

- Duration of high flows needed to lure YOY rainbow trout into nearshore habitats,
- Magnitude of the high flow that would be more effective in luring YOY trout to nearshore habitats,
- Whether or not moving to high flows first is needed to reduce YOY trout numbers (as opposed to simply dropping rapidly from normal flows to minimum flows),
- Timing of TMF cycles during the May–August period of trout emergence, and
- Number of cycles necessary to effectively limit trout recruitment.

If TMFs prove to be effective in controlling trout production and emigration to the Little Colorado River reach and they become an integral part of the LTEMP, regular implementation of TMFs may need to include variable timing to prevent adaptation of the population to specific timing (e.g., increase in recruitment by fall-spawning rainbow trout).

Certain aspects of TMF effectiveness can be addressed through observational studies (e.g., the number of YOY rainbow trout observed in the nearshore environment in daily increments after the high flow is initiated)¹; others may be addressed through consideration of the physical environment in Glen Canyon (i.e., what areas are inundated or exposed at different flows). Ultimately, however, effectiveness will be judged based on comparison of fall trout recruitment estimates to expectations based on prior years. It may take several years to make this determination, depending on the strength of the response and the type of TMFs tested. Ultimately, however, effectiveness will be based on the ability of TMFs to reduce recruitment in and emigration from the Glen Canyon reach. The driving forces behind emigration are not fully understood, but they are expected to be related to population size and food base in the Glen Canyon reach.

For the EIS modeling, a trigger of 200,000 YOY trout was used to determine when TMFs would be implemented. A regression equation based on annual volume, the variability in flows from May through August, and the occurrence of a spring HFE was used to predict the number of YOY. The actual trigger used could be higher or lower depending on the results of experiments that will be conducted on the effectiveness of TMFs. In addition, the predictive regression equation could be modified based on new information. The trigger and predictive equation used would be modified as needed in an adaptive management context utilizing the process described in Section 1.2.8. Triggers for implementation of TMFs would also be developed in consultation with the AZGFD and other entities as appropriate.

Monitoring of other resources, particularly the food base and the physiologic condition of adult rainbow trout, will also be considered. In addition, the number of YOY trout at the end of the summer would be estimated to determine if it equals or exceeds the estimated number of recruits needed to sustain the desired number of adult trout. If the estimated number of recruits is less than the recruitment target, TMFs will be re-evaluated for modification before implementation in subsequent years. It is anticipated that the trout population could rebound from a 1-year drop below this target level.

As discussed in relation to sediment experiments above, there is concern among scientists and stakeholders with regard to the risk associated with the implementation of spring HFEs as related to trout response and subsequent effects on the humpback chub population. For this reason, TMFs will be implemented and tested for effectiveness early in the LTEMP period if possible; preferably before the first spring HFEs are triggered, even if not triggered by high trout recruitment. Trout management flows could be implemented in years that featured a spring HFE and in the water year that follows an equalization flow because of the expected positive effects of

¹ Because older age classes of trout tend to occupy deeper habitats toward the middle of the river channel, they are less susceptible to stranding and are less likely to be directly affected by TMFs.

equalization on rainbow trout recruitment. Modeling indicates TMFs will be triggered by trout recruitment numbers in 32% of the years in the LTEMP period.

There is potential for confounding effects when coupling TMFs with HFEs. If trout recruitment is still high after implementation of TMFs that follow HFEs, this would suggest that TMFs were not as effective as designed for that trial. If recruitment is lower than expected after TMF implementation, however, uncertainty will remain about whether an HFE failed to stimulate trout recruitment or whether TMFs were effective in suppressing otherwise strong recruitment. It may not be necessary to determine the underlying effect on trout numbers unless TMFs have undesirable side effects on other resources or the trout population.

If TMFs are found to be highly effective in controlling trout recruitment and emigration of trout, and emigration only occurs or primarily occurs immediately following high recruitment years, it may be possible to limit TMF implementation and achieve multiple resource goals, particularly if unintended impacts of TMFs on other resources such as native fish become evident. If adverse impacts of TMFs become evident, this may also suggest revisiting whether or not TMFs are necessary in response to spring HFEs. Lastly, if there is an observed increase in trout recruitment due to fall HFEs, then application of TMFs in the spring following a fall HFE would be considered.

Implementation of TMFs would consider resource condition assessments and resource concerns using the processes described in Section 1.2.8. TMFs may not be tested when there appears to be the potential for unacceptable impacts on the resources listed in Section 1.2.8.

1.2.10.2 Tier 1 Conservation Actions for Humpback Chub

Tier 1 conservation actions designed to improve rearing and recruitment of juvenile humpback chub would be implemented if the combined point estimate for adult (\geq 200 mm) humpback chub in the Colorado River mainstem Little Colorado River aggregation (RM 57–RM 65.9) and in the Little Colorado River falls below 9,000 (2,000 in the mainstem and 7,000 in the Little Colorado River) as estimated by the currently accepted humpback chub population model, or if recruitment of subadult (150–199 mm) humpback chub does not meet or exceed estimated adult mortality. Tier 1 actions would include expanded translocations of YOY humpback chub within the Little Colorado River to areas within the river that have relatively few predators (i.e., above Chute Falls, Big Canyon), or larval fish would be taken to a rearing facility and released in the Little Colorado River inflow area once they reach 150 to 200 mm. In addition to these translocation activities, 300 to 750 larval or YOY humpback chub would be collected from the Little Colorado River and reared in a fish hatchery to less vulnerable sizes before releasing them. Once these fish reach 150 to 200 mm, they would be translocated to the Little Colorado River in the following year.

1.2.10.3 Tier 2 Mechanical Removal of Nonnative Fish

Experimental implementation of mechanical removal of nonnative fish will incorporate aspects of the protocol outlined in Reclamation's Non-Native Fish Control Environmental Assessment (NNFC EA) (Reclamation 2011b). However, modifications to the trigger criteria (Appendix D) were made to address Tribal concerns related to "taking of life" in the canyon, which prioritize proactive humpback chub conservation actions over mechanical removal of nonnative fish at the Little Colorado River when humpback chub abundance begins to decline. These modifications attempt to minimize the likelihood of mechanical removal, and mechanical removal of nonnative fish at the Little Colorado River would only occur if proactive conservation actions failed to reverse declines in adult humpback chub, and the adult abundance falls below 7,000. Modeling conducted for the EIS indicated that mechanical removal at this level was effective unless immigration rates into the Little Colorado River reach were high.

Two tiers of sequential actions were identified; the first would emphasize conservation actions that would take place early during an adult or subadult humpback chub population decline (Section 1.2.10.2). The second tier would serve as a backstop prescribing predator removal (Threat Reduction) if conservation measures did not mitigate a decline in humpback chub abundance.

Many factors affect humpback chub population dynamics such as water temperature and turbidity in the Colorado River, and potentially the hydrology of the Little Colorado River. These factors may constrain the effectiveness of available conservation actions that can be implemented in the event of a declining population of humpback chub. Juveniles and YOY can be translocated to other areas within and outside the Little Colorado River system; YOY or juvenile humpback chub can be head-started (temporarily held until sufficient growth is achieved) at a hatchery, and predator control can be implemented. Under current constraints, these are the only conservation tools available for humpback chub within the Colorado River reach and associated tributaries between Glen Canyon Dam and Lake Mead.

While healthy wildlife populations are rarely static, trigger objectives include prescribing actions to reverse/ameliorate impacts in order to maintain the Little Colorado River humpback chub population within an acceptable range, and, secondarily to reduce reliance on mechanical removal of nonnative fish. For the purposes of these triggers, it is assumed that the primary drivers of humpback chub population dynamics are interspecific interactions with nonnative species, especially rainbow trout (or brown trout), and low water temperature in the mainstem of the Colorado River (Kaeding and Zimmerman 1983; Douglas and Marsh 1996; FWS 2002a; Coggins et al. 2011; Yard et al. 2011). It is suspected that coldwater temperatures suppress growth and thus subject young humpback chub to predation for extended periods of time. The triggers focus on management of predator impacts only, if management actions are found to be ineffective or other factors are found to negatively control the humpback chub population, adaptation of triggers and conservation responses will be necessary.

The DOI recognizes that lethal mechanical removal is a concern for Tribes, particularly the Hopi Tribe and Pueblo of Zuni, who view it as a taking of life in the canyon without a beneficial use. Reclamation had committed in agreements with Tribes in 2012 to consider live removal when feasible (Reclamation 2012a); however, the presence of whirling disease prohibits live removal of trout due to the risk of spreading the disease to other waters. Reclamation and NPS have worked with the Tribes to determine a beneficial use of the removed fish on other projects and understand that what is considered beneficial use may not be the same for all Tribes. Reclamation and NPS are committed to consult further with the Tribes to determine acceptable mitigation for nonnative fish control.

1.2.10.4 Low Summer Flows

Low summer flows could be considered a potential tool for improving the growth and recruitment of young humpback chub if temperature had been limiting these processes for a period of years. Low summer flows may lead to warmer water temperatures in the Little Colorado River reach and farther downstream, as well as contributing to enhanced growth rates of young humpback chub. There are also potential negative effects from low summer flows to several resources such as hydropower, sediment, water quality, vegetation, and recreation. Low summer flows may also negatively affect humpback chub due to an increase in warmwater nonnative fish or a decrease in the aquatic food base. One test of low steady summer flows was conducted below Glen Canyon Dam in 2000; the results, however, relative to humpback chub were not conclusive (Ralston et al 2012).

Because of the uncertainty related to the effects of low summer flows on humpback chub, other native fish, warmwater nonnative fish, water quality, and potentially other resources, DOI will ensure that the appropriate baseline data are collected throughout the implementation of LTEMP. In addition, DOI will convene a scientific panel that includes independent experts prior to the first potential use of low summer flows to synthesize the best available scientific information related to low summer flows. The panel may meet periodically to update the information, as needed. This information will be shared as part of the AMWG annual reporting process.

It is thought that the potential benefit of an increase in temperature will be greatest if a water temperature of at least 14°C could be achieved at the Little Colorado River, because these warmer temperatures could favor higher growth rates of humpback chub (nearly 60% higher). For comparison, the July through September growth increments of YOY humpback chub are estimated to be 4, 7, 11, 14, and 17 mm at temperatures of 12, 13, 14, 15, and 16°C, respectively, based on a growth-temperature regression in Robinson and Childs (2001). Note that reduction in summer flows would necessitate increasing flows in other months relative to base operations (Table 4; Figure 6).

Month	Monthly Release Volume (kaf)	Proportion of Total Annual Volume	Mean Daily Flow (cfs)	Daily Fluctuation Range (cfs)
October	643	0.0781	10,451	5,783
November	642	0.0780	10,781	5,774
December	716	0.0870	11,643	6,443
January	764	0.0928	12,423	6,874
February	675	0.0820	12,153	6,074
March	691	0.0840	11,245	6,223
April	859	0.1044	14,433	7,730
May	851	0.1034	13,841	7,659
June	930	0.1130	15,631	8,000
July	492	0.0598	8,000	2,000
August	492	0.0598	8,000	2,000
September	476	0.0578	8,000	2,000

TABLE 4 Flow Parameters for a Year with Low Summer Flows underthe Proposed Action in an 8.23-maf Year^a

^a Within a year, monthly operations may be increased or decreased based on changing annual runoff forecasts and other factors, such as application of the Long-Range Operating Criteria for Colorado River Basin Reservoirs, which are currently implemented through the 2007 Interim Guidelines for Lower Basin Shortages and Coordinated Operations for Lake Powell and Lake Mead (Reclamation 2007a).

If tested, low summer flows would occur for 3 months (July, August, and September), and only in the second 10 years of the LTEMP period. The duration of low summer flows could be shortened to less than 3 months in successive experiments if supported by the scientific panel described above or based on the scientific data and observed effects. The probability of triggering a low summer flow experiment is considered low (about 7% of years) because the water temperature conditions that would allow such a test occur infrequently.

Low summer flows would only be implemented in years when the projected annual release was less than 10 maf, and if the temperature at the Little Colorado River confluence would be below 14°C without low summer flows, and release temperature was sufficiently high that 14°C could be achieved at the Little Colorado River with the use of low summer flows.

The ability to achieve target temperatures at the Little Colorado River confluence by providing lower flows is dependent on release temperatures, which are in turn dependent on reservoir elevation. For example, using the temperature model of Wright et al. (2008), in a 8.23-maf year, release temperatures of 10.8°C, 11.0°C, and 11.7°C would be needed in July, August, and September, respectively, to achieve a target temperature of 14°C at the Little Colorado River confluence at flows of 8,000 cfs.



FIGURE 6 Mean, Minimum, and Maximum Daily Flows under Triggered Low Summer Flows of the Proposed Action in an 8.23-maf Year Based on the Values Presented in Table 4

Release temperatures fall into three categories for any temperature target: (1) too low to achieve the target temperature at the Little Colorado River even at low flow, (2) high enough to achieve the target temperature at the Little Colorado River only if low flows (5,000 to 8,000 cfs) are provided, and (3) high enough to achieve target temperature at the Little Colorado River regardless of the flow level. Low summer flows would only be triggered in years that fell into the second category.

Implementation of a low-flow experiment is complicated by two factors: the earliest date at which it could be determined that a target temperature of at least 14°C could be achieved in all 3 months, and the ability to release the remaining annual volume once that determination is made. The earliest time a determination could be made would be in early April of each year, and it would be based on the April 1 forecast of reservoir elevation. Because low summer flows could be implemented in the 3 months at the end of the water year, it is possible that by the time a determination was made to conduct a low summer flow experiment, it may not be possible to release enough water in the remainder of the spring to compensate for the low-flow period. A low-flow experiment would only be tested in years when performing the experiment would not result in a deviation from the annual Glen Canyon Dam release volumes made pursuant to the Long-Range Operating Criteria for Colorado River Basin Reservoirs, which are currently implemented through the 2007 Interim Guidelines for Lower Basin Shortages and Coordinated Operations for Lake Powell and Lake Mead (Reclamation 2007a).

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

It is possible that a low summer flow could negatively impact humpback chub through impacts on the food base. In moderate or higher water years, the April–June monthly volumes would have to be quite high and would likely be at or near 25,000 cfs. Following these high flows, there would be an abrupt drop to 8,000 cfs or lower (e.g., 5,000 cfs), which would likely leave much or all of the food base in Glen Canyon and downriver up above the new lower water line (Kennedy and Ralston 2011). These values represent declines in midges and gammarus in the portions of the channel that have been dewatered, and only a small portion of the channel is actually dewatered. Low summer flows will also provide ideal egg laying conditions for aquatic insects, which should facilitate a rapid recovery of the invertebrate prey base to the reduced habitat area associated with this type of experiment. Nonetheless, the potential for low summer flow experiments to negatively impact the food base exist, and measures that seek to minimize any negative impacts will be considered if a low summer flow is implemented.

Finch et al. (2015) found that humpback chub grew more slowly during steady flows than they did during fluctuating flows in the same year, but that turbidity was potentially confounding in this analysis. Recent evidence suggests that not only is the system food limited in most years (Cross et. al. 2013), but that in 2014 and 2015, a system-wide reduction in fish numbers and condition resulting from food limitation was observed. This was a strong enough response that spring Little Colorado River adult estimates were extremely low, and the leading hypothesis was that the adults were skipping spawning and remaining in the mainstem due to lack of sufficient conditions to spawn, yet temperatures were quite warm in 2014 and 2015 (reaching 16°C and 15°C, respectively). Thus, it would seem that under the right conditions, low summer flow could benefit chub, but under conditions that already show high numbers of fish and relatively low food availability, the impacts could be adverse, the rate of skip spawning could increase, and the net impact on chub could be low reproduction and recruitment for that year. However, in 3 years of monitoring the Little Colorado River in June and July, YOY numbers were highest despite the lowest number of spawners (Van Haverbeke 2015). Also key to this experiment being successful is the number of YOY humpback chub being produced in the Little Colorado River, which was pointed out as a confounding factor in the 2000 low summer flow experiment. Production in the Little Colorado River was low (poor habitat conditions), and thus few fish were in the mainstem to be exposed to the treatment. If similar conditions occur, low summer flow would be expected to not have a beneficial impact on the population.

A first test of a low summer flow would feature low flow of 8,000 cfs and relatively little fluctuation (\pm 1,000 cfs per day). Depending on the results of the first test with regard to warming and humpback chub response, the magnitude of the low flow could be adjusted up or down (as low as 5,000 cfs), and the level of fluctuation also modified up to the full range allowed under the proposed action (i.e., 10 × monthly volume [in kaf] in July and August, and 9 × monthly volume [in kaf] in September).

The first test of low summer flows will be determined to be successful or unsuccessful for humpback chub based on input from an independent scientific panel review. If the first test was determined to be unsuccessful (and it was determined to have been implemented properly without major confounding factors), then additional tests would not be performed. Low summer flows would be considered successful if it can be determined that they produced sufficient growth of YOY humpback chub and that growth resulted in an increase in recruitment, but

avoided unacceptable increases in warmwater nonnative fishes, trout, or aquatic parasites, or resulted in unacceptable adverse impacts on other aquatic resources. If it was determined to be successful, then additional low summer flows would occur only when humpback chub population concerns warranted them and water temperature has been colder for a period of years and the desired warming could be achieved only with low summer flows. The temperature target could be adjusted 1°C higher based on the results of the first test or the limitations between predicted and measured temperatures.

Implementation of low summer flows would consider resource condition assessments and resource concerns using the processes described in Section 1.2.8. Low summer flows may not be conducted in years when there appears to be the potential for unacceptable impacts on the resources listed in Section 1.2.8.

The effects of low summer flows on Lake Mead water quality are an identified concern. DOI will coordinate with relevant water quality monitoring programs or affected agencies prior to implementing any test of low summer flows. There are concerns related to the risk of warmwater nonnative fish expansion or invasion (e.g., the elevation of Lake Mead was high or the number of warmwater nonnative fish was high). These issues are potential off-ramps as described in Section 1.2.8.

1.2.10.5 Low Steady Weekend Flows for Macroinvertebrate Production

A more diverse and productive aquatic food base could benefit a variety of priority resources including native fish (including the endangered humpback chub), the rainbow trout fishery, and other riparian species that occur in Glen, Marble, and Grand Canyons. Mayflies (Ephemeroptera), stoneflies (Plecoptera), and caddisflies (Trichoptera), collectively referred to as EPT, are notably absent from the Glen and Marble Canyon reaches and very low in abundance and diversity in the Colorado River in the Grand Canyon. Furthermore, it has been demonstrated that EPT taxa are recruitment limited, because daily flow fluctuations to meet hydropower demand cause high egg mortality and which is the limiting factor. The rationale behind this hypothesis is that fluctuations from Glen Canyon Dam create a varial (intermittently wetted) zone along shorelines that is large enough to dewater eggs. The varial zone is primarily determined by the shape of the canyon, and in narrow stretches of the canyon the varial zone can be high, while in wider stretches of the river the varial zone is much smaller. Because the Colorado River in Glen, Marble, and Grand Canyons is canyon-bound and the tributaries that join the river all have comparatively low flow, the size of the varial zone does not appreciably decrease with distance downstream. Thus, although water temperature regimes become more naturalized with distance downstream, the effect that daily flow fluctuations to meet hydropower demand have on the stability of shoreline habitat does not attenuate with distance from the dam.

This hypothesis attributes the absence of EPT and the poor health of the invertebrate assemblage to the width of the varial zone, similar to earlier investigations (Blinn et al. 1995), but focuses on the effects unstable shorelines have on the eggs of these species. This hypothesis assumes that egg-laying by EPT occurs principally along shorelines. According to the hypothesis, EPT taxa downstream of Glen Canyon Dam are recruitment limited, because daily

flow fluctuations to meet hydropower demand negatively affect habitat quality along the shorelines where egg-laying is assumed to occur. This could have been tested during the 2000 Low Steady Summer Flow experiment or in the fall steady-flow experiment from 2008 to 2012, but food base field research was not adequate at the time to detect any changes in EPT related to the steady flows. However, even though substantial steady flow periods were implemented, especially in 2000, increases in EPT species and production were not observed (Kennedy and Ralston 2011).

To test this hypothesis, steady flows would be provided every weekend from May through August (34 days total).² The flow on weekends would be held to the minimum flow for that month, which would ensure that the insect eggs laid during weekends would remain submerged throughout larval development. If the hypothesis is true, there would be an increase in insect production, and possibly diversity, due to the reproductive success of insects that laid eggs during weekends. No changes in monthly volumes, ramping rates, or the maximum daily range in flow during weekdays would be required for this experiment. To offset the smaller water releases that would occur during weekends within a given month, larger releases would need to occur during the weekdays within a given month.

Implementation of macroinvertebrate production flows would consider resource condition assessments and resource concerns using the processes described in Section 1.2.8. These flows may not be tested when there appears to be the potential for unacceptable impacts on the resources listed in Section 1.2.8.

Effects of the tests would be evaluated using observations to determine the locations where insect eggs are deposited and the emergence rates of species. Depending on the outcome of the tests, the experiment could be discontinued if there were unacceptable effects on other resources. There is also the possibility that implementation would result in confounding interactions with TMF experiments, and this would be discussed during the process described in Section 1.2.8.

1.2.11 Native and Nonnative Plant Management and Experimental Actions

Experimental riparian vegetation treatment activities would be implemented by NPS under the proposed action and would modify the cover and distribution of riparian plant communities along the Colorado River. All activities would be consistent with NPS Management Policies (NPS 2006) and would occur only within the Colorado River Ecosystem in areas that are influenced by dam operations. NPS will work with Tribal partners and GCMRC to experimentally implement and evaluate a number of vegetation control and native replanting activities on the riparian vegetation within the Colorado River Ecosystem in GCNP and GCNRA. These activities would include ongoing monitoring and removal of selected nonnative plants, species in the corridor, systematic removal of nonnative vegetation at targeted sites, and native replanting at targeted sites and subreaches, which may include complete removal of

² The duration and other characteristics of experimental macroinvertebrate production flows could be adjusted within the range of the analysis based on the results of initial experiments.

tamarisk (both live and dead) and revegetation with native vegetation. Treatments would fall into two broad categories, including the control of nonnative plant species and revegetation with native plant species. Principal elements of this experimental riparian vegetation proposal include:

- Control nonnative plant species affected by dam operations, including tamarisk and other highly invasive species;
- Develop native plant materials for replanting through partnerships and use of regional greenhouses;
- Replant native plant species to priority sites along the river corridor, including native species of interest to Tribes;
- Remove vegetation encroaching on campsites; and
- Manage vegetation to assist with cultural site protection.

2 ENVIRONMENTAL BASELINE

2.1 DESCRIPTION OF SPECIES IDENTIFIED FOR ANALYSIS

2.1.1 Kanab Ambersnail

2.1.1.1 Legal Status

The Kanab ambersnail, *Oxyloma haydeniensis*, was listed as endangered in 1992 (FWS 1995). The Kanab ambersnail is the only threatened, endangered, or sensitive invertebrate species that may occur along the Colorado River in the Grand Canyon. However, recent evidence from anatomical and molecular genetics studies indicates that this is a geographically widespread taxon whose listing under the ESA may have been incorrect (Littlefield 2007). In a study of *Oxyloma* specimens collected from 12 locations throughout the western United States, including Kanab ambersnail from the Grand Canyon, morphometric and genetic results indicated that the Kanab ambersnail can be regarded as a member of the same species as the other *Oxyloma* populations analyzed (Culver et al. 2013). However, until this taxonomic change occurs, the Kanab ambersnail remains a listed species (FWS 2011b). No critical habitat is designated for this species.

2.1.1.2 Recovery Goals and Status

The recovery plan for Kanab ambersnail was completed in 1995 (FWS 1995). The Kanab ambersnail may be considered for downlisting to threatened when the following criteria have been attained:

- The location and/or establishment of additional populations. Maintain 10 separate populations which have been demonstrated to have population numbers large enough to allow for the long-term viability of the population. This criterion is provisional. It is probable that this criterion will be modified as additional information is acquired concerning the species distribution, abundance, and stability of its separate populations.
- 2. The establishment of formal land management designations and/or implementation of land management plans which provide long-term, undisturbed habitat for the Kanab ambersnail for the above 10 populations.

The 5-year status review for the Kanab ambersnail was completed in 2011 (FWS 2011d). The FWS found that no change in the status of the species was warranted due to the ongoing, existing threats due to private land development, controlled flooding in the Colorado River, climate change, and inadequate existing regulatory mechanisms. However, Arizona and Utah ambersnail populations identified as "Kanab ambersnail" and "Niobrara ambersnail" are based

primarily on morphological distinctions. Recent genetic analysis and morphological evaluation by Culver et al. (2013) on ambersnail specimens suggests that the Arizona and Utah populations, including Vaseys Paradise, are genetically and morphologically similar to other *Oxyloma* populations in the study, and their taxonomic identity may be revised in the future. The consensus appears to be that this snail is part of a much larger population that has higher numbers and distribution. The FWS did recognize that genetic, anatomical, and morphological information resulted in conflicting views on the taxonomy of the species.

2.1.1.3 Historic and Current Range

Globally, the Kanab ambersnail is only found in three locations. Two of these are within the Grand Canyon: the riparian vegetation at Vasey's Paradise and Elves Chasm. Vasey's Paradise is at RM 31.5, and Upper Elves Chasm is at RM 116.6. The latter population was created from snails translocated from Vasey's Paradise (FWS 2008). The third location for the Kanab ambersnail is Three Lakes, Utah (Reclamation 2007b).

2.1.1.4 Habitat

The Kanab ambersnail lives in association with watercress (*Nasturtium officinale*), cardinal monkeyflower (*Mimulus cardinalis*), cattails (*Typha*), sedges (*Carex*), and rushes (*Juncus*). Populations within the Grand Canyon occur in areas with water sources originating from limestone or sandstone geologic strata (Meretsky and Wegner 2000; Sorensen 2009). The increase in cover, reduction in beach-scouring flows, and introduction of the nonnative watercress led to a >40% increase in suitable Kanab ambersnail habitat area at Vasey's Paradise compared to pre-dam conditions (Stevens et al. 1997a).

Climate change has the potential to affect the Kanab ambersnail habitat. The water source at Vasey's Paradise consists of waterfalls emanating from groundwater and emerging from the cliff face. In 2014 and 2015, the flow was noticeably reduced, likely as a result of basinwide drought. Consequently, the usually dense vegetation at Vasey's Paradise is notably diminished.

2.1.1.5 Life History

Kanab ambersnails live 12 to 15 months and are capable of self-fertilization. Mating and reproduction occur from May to August. Subadults dominate the overwinter population. Snails enter dormancy in October–November and become active in March–April. Overwinter mortality ranges between 25 and 80% (Stevens et al. 1997a; IKAMT 1998). During mild winters, they can continue their life cycle without dormancy or may go in and out of dormancy several times throughout the winter (Sorensen and Nelson 2002).

Based on annual survey data, live counts of Kanab ambersnails at Vasey's Paradise declined in 2011 from previous years, although the ambersnail habitat at Vasey's Paradise was in overall good condition in 2011. At Elves Chasm, live counts of Kanab ambersnails remained

higher in 2011 than previous years, and habitat at this location was in good condition in 2011 (Sorensen 2012). The population at Vasey's Paradise generally occurs at elevations above 33,000-cfs flows. However, as much as 7.3% of the Vasey's Paradise population occurs below the elevation of 33,000-cfs flow, and as much as 16.4% of the population occurs below the elevation of 45,000-cfs flow. The Elves Chasm population is located above the elevation of 45,000-cfs flow (Reclamation 2011b).

2.1.2 Humpback Chub

2.1.2.1 Legal Status

The humpback chub (*Gila cypha*) is currently listed as "endangered" under the ESA. It was first included in the List of Endangered Species issued by the Office of Endangered Species on March 11, 1967 (32 FR 4001) and was considered endangered under provisions of the Endangered Species Conservation Act of 1969 (16 U.S.C. 668aa). The humpback chub was included in the United States List of Endangered Native Fish and Wildlife issued on June 4, 1973 (38 FR 33085), and received protection as endangered under Section 4(c)(3) of the original ESA of 1973. Critical habitat includes 280 km of the Colorado River through Marble and Grand Canyons from Nautiloid Canyon (RM 34) to Granite Park (RM 208), and the lower 13 km of the Little Colorado River. Primary threats to the species include streamflow regulation and habitat modification (including coldwater dam releases and habitat loss), competition with and predation by nonnative fish species, parasitism, hybridization with other native Gila, and pesticides and pollutants (FWS 1990, 2002a).

2.1.2.2 Recovery Goals and Status

Recovery for the humpback chub was defined by the FWS Humpback Chub Recovery Goals (Recovery Goals) (67 FR 55270) (FWS 2002a). In 2006, a U.S. District Court ruling set aside the Recovery Goals, because they lacked time and cost estimates for recovery. Nevertheless, the recovery programs and the GCDAMP continue to utilize the underlying science in the Recovery Goals. The FWS's 2011 Humpback Chub 5-Year Review relies on the information provided in the recovery goals and provides supplemental information on the species' distribution and status.

The court did not fault the Recovery Goals as deficient in any other respect, thus the FWS, the GCDAMP, and the Upper Colorado River Endangered Fish Recovery Program (UCRRP), the program that addresses conservation of all of the upper Colorado River basin populations of humpback chub, continue to utilize the underlying science in the Recovery Goals. In the 2009 Supplemental Opinion, the FWS referenced the draft 2009 revisions to the Recovery Goals document because that document provided updates on species biology and distribution and represented the best available scientific information at that time. The draft 2009 revisions to the Recovery Goals. Thus, we are using the demographic criteria found in both the 2002 Recovery Goals and

2009 draft Recovery Goals. The FWS's 2011 Humpback Chub 5-Year Review relies on the information provided in the Recovery Goals and provides supplemental information on the species' distribution and status.

That supplemental information, as well as the demographic criteria found in the Recovery Goals, has been considered in this BA and is summarized here.

The Recovery Goals consist of actions to improve habitat and minimize threats.

The Recovery Goal demographic criteria for downlisting (endangered to threatened) are as follows:

Upper Basin Recovery Unit

- 1. Each of the five self-sustaining populations is maintained over a 5-year period, starting with the first point estimate acceptable to the FWS, such that:
 - a. The trend in adult (age 4+; $\geq 200 \text{ mm}$ [7.9 in.] total length [TL]) point estimates does not decline significantly, and
 - b. Mean estimated recruitment of age-3 (150–199 mm [5.9–7.8 in.] TL) naturally produced fish equals or exceeds mean annual adult mortality, and
- 2. One of the five populations (e.g., Black Rocks/Westwater Canyon or Desolation/Grey Canyons) is maintained as a core population such that each point estimate exceeds 2,100 adults (Note: 2,100 is the estimated Minimum Viable Population (MVP) number; see Section 3.3.2 of the Recovery Goals).

Lower Basin Recovery Unit

- 1. The Grand Canyon population is maintained as a core over a 5-year period, starting with the first point estimate acceptable to the FWS, such that:
 - a. The trend in adult (age 4+; $\geq 200 \text{ mm}$ [7.9 in.] TL) point estimates does not decline significantly,
 - b. Mean estimated recruitment of age-3 (150–199 mm [5.9–7.8 in.] TL) naturally produced fish equals or exceeds mean annual adult mortality, and
 - c. Each core population point estimate exceeds 2,100 adults (MVP).

The Recovery Goal demographic criteria for delisting are as follows:

Upper Basin Recovery Unit

- 1. Each of the five self-sustaining populations is maintained over a 3-year period beyond downlisting, starting with the first point estimate acceptable to the FWS, such that:
 - a. The trend in adult (age 4+; $\geq 200 \text{ mm}$ [7.9 in.] TL) point estimates does not decline significantly, and
 - b. Mean estimated recruitment of age-3 (150–199 mm [5.9–7.8 in.] TL) naturally produced fish equals or exceeds mean annual adult mortality, and
- 2. Two of the five populations (e.g., Black Rocks/Westwater Canyon and Desolation/Grey Canyons) are maintained as core populations such that each point estimate exceeds 2,100 adults (MVP).

Lower Basin Recovery Unit

- a. The Grand Canyon population is maintained as a core over a 3-year period beyond downlisting, starting with the first point estimate acceptable to the FWS, such that:
- b. The trend in adult (age 4+; ≥ 200 mm [7.9 in.] TL) point estimates does not decline significantly,
- c. Mean estimated recruitment of age-3 (150–199 mm [5.9–7.8 in.] TL) naturally produced fish equals or exceeds mean annual adult mortality, and
- d. Each core population point estimate exceeds 2,100 adults (MVP).

The 5-year status review for the humpback chub was completed in 2011 (FWS 2011c). A change in the status of the humpback chub was not recommended because 5 of 6 demographic recovery criteria and 4 of 22 downlisting criteria had not been met. Factors continuing to threaten the Grand Canyon humpback chub population are described below.

2.1.2.3 Historic and Current Range

The humpback chub is a large, long-lived species endemic to the Colorado River system. This member of the minnow family may attain a length of 20 in., weigh 2 lb or more, and live as long as 40 years (Andersen 2009). Historically, this species occurred throughout much of the Colorado River and its larger tributaries from below Hoover Dam upstream into Arizona, Utah, Colorado, and Wyoming (FWS 2002b and refs. therein). Although historic abundance levels are unknown, the humpback chub is currently thought to occupy 24% of its historic range, restricted to six population centers, five in the upper Colorado River basin and one in the lower basin (FWS 2002b). The upper basin populations occur in (1) the Colorado River in Cataract Canyon, Utah; (2) the Colorado River in Black Rocks, Colorado; (3) the Colorado River in Westwater Canyon, Utah; (4) the Green River in Desolation and Gray Canyons, Utah; and (5) the Yampa River in Yampa Canyon, Colorado. The only population in the lower basin occurs in the Colorado River in Marble Canyon, the Grand Canyon, and Little Colorado River (FWS 2011a).

The Colorado River/Little Colorado River population is the largest of the six population centers of the humpback chub (FWS 2011c). Within the Grand Canyon, this species is most abundant in the vicinity of the confluence of the Colorado River and Little Colorado River (Kaeding and Zimmerman 1983; Douglas and Marsh 1996; Valdez and Ryel 1995). This population is specifically referred to as the Little Colorado River aggregation of humpback chub and includes those fish residing in the Little Colorado River and in the mainstem within proximity of a few miles to the Little Colorado River mouth. In addition, there are eight other areas (aggregation areas) where humpback chub are, or have been, regularly collected; these aggregation areas are located in the mainstem at 30 Mile, Lava Chuar-Hance, Bright Angel Creek inflow, Shinumo Creek inflow, Stephen Aisle, Middle Granite Gorge, Havasu Creek inflow, and Pumpkin Spring (Figure 7; Valdez and Ryel 1995; Ackerman 2008; Persons et al. 2016). In addition, since 2009, translocations of humpback chub have been made by the FWS to introduce juvenile fish upstream of Chute Falls in the Little Colorado River, and by the NPS, with assistance provided by Reclamation, to introduce juvenile fish into Shinumo and Havasu Creeks, with the goal of establishing additional spawning populations within the Grand Canyon (NPS 2013). Survey data collected in 2013, 2014, and 2015 suggest that translocated humpback chub have successfully spawned in Havasu Creek (NPS 2013). Humpback chub occupy approximately 3.5 mi (5.6 km) of lower Havasu Creek, from the mouth to Beaver Falls, which is a barrier to upstream movement of fish.

Sampling conducted between October 2013 and September 2014 in western Grand Canyon between Lava Falls (RM 180) and Pearce Ferry (RM 280) captured 144 juvenile humpback chub during sampling of the small-bodied fish community. In addition, 209 humpback chub larvae were collected during sampling of the larval fish community in randomly selected sites (Albrecht et al. 2014). Results were similar in larval and small-bodied fish sampling in 2015 (Kegerries et al. 2015): 285 juvenile and 67 age-0 humpback chub were captured during smallbodied and larval fish sampling, respectively, from throughout the study area. These results suggest that young humpback chub are using widespread nursery and rearing habitats between RM 180 and RM 280 in the western Grand Canyon.

2.1.2.4 Population within the Action Area

The Little Colorado River population (aggregation) of humpback chub is measured with closed and open population models. Closed models estimate the annual spring and the annual fall abundance of various size classes of chub within the Little Colorado River (Van Haverbeke et al. 2013, 2016). As such, the closed models do not account for chub that are not residing in the

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement



FIGURE 7 Closed Chapman Petersen Abundance Estimates (±95% CI) of Humpback Chub ≥150 mm and ≥200 mm in the Little Colorado River during (A) Spring (2001–2015) and (B) Fall Seasons (2000–2015) (Closed spring and fall abundance estimates of humpback chub >150 mm in the Little Colorado River during 1991 and 1992 are from Douglas and Marsh [1996].)

O-63

Little Colorado River during any particular year (i.e., there is always a portion of the Little Colorado River aggregation that is residing in the nearby mainstem each year). Initial closed mark-recapture population efforts in the Little Colorado River were conducted in the early 1990s (Douglas and Marsh 1996), after which there was a hiatus until they were resumed again in 2000 (Van Haverbeke et al. 2013, 2016). Results from both of these studies indicate that sometime in the mid- to late-1990s, humpback chub underwent a significant decline in the Little Colorado River (Figures 7 and 8). This was followed by a period of relatively low, but stable abundance between 2000 and 2006, and by a period (2007–2014) of significantly increased abundance levels (Van Haverbeke et al. 2013). The post-2006 increase in humpback chub \geq 150 mm and \geq 200 mm was visible during both spring and fall seasons, but it was more apparent during spring months (Figure 7). Spring 2015 saw a significant lowering of abundance of humpback chub \geq 150 mm and \geq 200 mm compared to the previous several years. The cause of this decline is unknown, but there is evidence from sampling in the mainstem during 2015 that many chub may have simply remained or emigrated into the mainstem during 2015 (i.e., the portion of the Little Colorado River aggregation of chub residing in the nearby mainstem was higher than usual).

In addition, open population models are conducted to estimate the abundance of the Little Colorado River aggregation of humpback chub (Coggins and Walters 2009; Yackulic et al. 2014), both those fish residing in the Little Colorado River and those residing in the nearby mainstem. Because the open models use capture histories of humpback chub that span across years and because most humpback chub will eventually be captured either in the Little Colorado River or in the mainstem (and thus accounted for in the open models), the open models estimate the abundance of humpback chub in the Little Colorado River as well as those residing in the nearby mainstem. As such, the open models provide annual abundance estimates that are higher than the closed model estimates described above. Open models show that from 1989 through 2001, there was a decline of adult humpback chub within the Grand Canyon, with estimated numbers of the Little Colorado River aggregation declining from approximately 11,000 adults (age 4+) in 1989 to about 5,000 adults in 2001 (Coggins and Walters 2009) (Figure 8). However, since about 2001, the downward population trend reversed, with the estimated number of adult fish increasing to approximately 8,000 fish by 2008 (Figure 8) (Coggins and Walters 2009). More recently, abundance estimates for 2009 to 2012 suggest the population has continued to increase to approximately 11,000 adults (Figure 9) (Yackulic et al. 2014). Unlike the closed models, Coggins and Walters (2009) open models suggested that the increasing trend in adult humpback chub abundance began earlier than 2007. They explain that this may be an effect of aging error, with the least biased estimates for recruitment and adult abundance trends being those most proximal to the end of the dataset being analyzed. As such, one might expect to see the adult increases in abundances beginning earlier in their open models (Figure 8). Importantly, both the closed and open population estimators provide the same trend, and that is that the humpback chub experienced a period of significant decline followed by a period of significant increase.

Aside from the Little Colorado River aggregation, based on sampling within and outside the other known aggregations of humpback chub by Persons et al. (2016), catch rates of humpback chub generally increased as well. Factors suggested as being responsible for this estimated increase are discussed later in this section. In addition, recent preliminary population estimates for humpback chub aggregations suggest that humpback chub in several aggregations



FIGURE 8 Estimated Adult Humpback Chub Abundance (Age 4+) from Age-Structured Mark-Recapture Model Incorporating Uncertainty in Assignment of Age (Error bars represent minimum 95% confidence intervals and do not consider uncertainty in growth or mortality.) (Source: Coggins and Walters 2009)



FIGURE 9 Estimated Total Adult Abundance of Humpback Chub in the Lower 8 mi of the Little Colorado River and a 2-mi Portion of the Colorado River Just Downriver of the Confluence of the Little Colorado and Colorado Rivers, for September 2009 through 2012 (Error bars represent the 95% confidence intervals.) (Source: Yackulic et al. 2014)

may have increased as a result of (1) translocations to Shinumo and Havasu Creeks; (2) good production in the Little Colorado River; (3) water temperatures that were about 1°C warmer since the early 2000s; (4) significantly warmer than normal water temperatures in 2004, 2005, and 2011; and (5) trout control implemented at the Little Colorado River inflow (NPS 2013; Yackulic et al. 2014).

The most recent humpback chub population estimate in Havasu Creek was approximately 280 individuals as of May, 2015 (Figure 10). While reproduction has been documented, the population has increased primarily as a result of continued translocations, and less attributed to recruitment.

2.1.2.5 Habitat

Adult humpback chub occupy swift, deep, canyon reaches of the river (Valdez and Clemmer 1982; Valdez and Ryel 1995), with microhabitat use varying among age-groups (Valdez 1990; Stone and Gorman 2006; Gerig et al 2014; Dodrill et al 2015). Within the Grand Canyon, the largest number of humpback chub and their primary spawning are in the vicinity of the Little Colorado River and its inflow reach (RM 57–RM 77; Persons et al. 2016; (Figure 11), with adults being associated with large eddy complexes. Mark-recapture studies in the Grand Canyon reported that most captures of humpback chub occurred in and around the Little Colorado River, with more than 80% of recaptured fish being collected in the same mainstem river reach or tributary where they were originally tagged (Paukert et al. 2006; Persons et al. 2016). However, some of the marked fish were determined to have moved as much as 154.5 km (96 mi) throughout the Grand Canyon (Paukert et al. 2006).

Valdez and Ryel (1995, 1997) reported on adult humpback chub habitat use in the Colorado River in GCNP. They found adults used primarily large recirculating eddies, occupying areas of low velocity adjacent to high-velocity currents that deliver food items. Within GCNP, adults demonstrate high microsite fidelity and occupy main channel eddies, while subadults use nearshore habitats (Valdez and Ryel 1995; Robinson et al. 1998, Stone and Gorman 2006). Adults also congregated at tributary mouths and flooded side canyons during high flows.

Recently, a study conducted in 2010 examined the habitat use and movement of 30 radiotagged adult humpback chub in the Colorado River during 2 months of fluctuating flow followed by 2 months of steady flow (Gerig et al. 2014). The radio-tagged fish were found to use eddies extensively while avoiding runs. During both flow treatments, the tagged fish exhibited only small daily movements of about 33 ft/day, and no effect of flow was observed on either habitat selection or movement.

The main spawning area for the humpback chub within the Grand Canyon is the Little Colorado River, which provides warm temperatures suitable for spawning and shallow low-velocity pools for larvae (Gorman 1994). The species spawns primarily in the lower 13.6 km of the Little Colorado River, but occasional spawning is suspected in other areas of the Colorado River as well (Valdez and Masslich 1999; Anderson et al. 2010). Gorman and Stone (1999) found ripe adults aggregated in areas of complex habitat structure associated with clean gravel



FIGURE 10 Total Abundance of Humpback Chub in Havasu Creek, Based on Annual Population Estimates between 2012 and 2015 (Error bars represent 95% confidence intervals.)



FIGURE 11 Humpback Chub Aggregation Areas along the Colorado River between Glen Canyon Dam and Lake Mead (Sources: NPS 2013)

deposits among large boulders mixed with travertine masses in or near runs and eddies. While mainstem spawning is suspected near 30-Mile Spring, or in other areas in the western Grand Canyon following the detection of larval humpback chub in recent years (Albrecht et al. 2014; Kegerries et al. 2015), studies have not been completed to identify spawning areas or habitat in the Colorado River in the Grand Canyon.

Young humpback chub seek areas that provide physical cover and contain some velocity refuges, including shoreline talus, vegetation, and backwaters typically formed by eddy return current channels (AZGFD 1996; Converse et al. 1998; Dodrill et al. 2015). Backwaters can have warmer water temperatures than other habitats, and native fish, including the humpback chub, are frequently observed in backwaters, leading to a common perception that this habitat is critical for juvenile native fish conservation. However, backwaters are rare and ephemeral habitats, so they contain only a small portion of the overall population. For example, Dodrill et al. (2015)) showed the total abundance of juvenile humpback chub was much higher in talus than in backwater habitats. Moreover, when extrapolated to relative densities based on estimates of backwater prevalence directly after a controlled flood, the majority of juvenile humpback chub were still found outside of backwaters. This suggests that the role of controlled floods in influencing native fish population trends may be limited. As young humpback chub grow, they exhibit an ontogenic shift toward deeper and swifter offshore habitats. Valdez and Ryel (1995, 1997) found that young humpback chub (21-74 mm [0.83-2.9 in.] TL) remain along shallow shoreline habitats throughout their first summer, at low water velocities and depths less than 1 m (3.3 ft). They shift as they grow larger (75–259 mm [2.95–10.20 in] TL), and by fall and winter move into deeper habitat with higher water velocities and depths up to 1.5 m (4.9 ft). Stone and Gorman (2006) found similar results in the Little Colorado River, discovering that humpback chub undergo an ontogenesis from diurnally active, vulnerable, nearshore-reliant YOY (30-90 mm [1.81-3.54 in.] TL) to nocturnally active, large-bodied adults (180 mm [7.09 in.] TL), which primarily reside in deep mid-channel pools during the day and move inshore at night. This ontogenetic habitat shift may be important for smaller streams to which humpback chub have been translocated. Spurgeon (2012) found that larger humpback chub translocated to Shinumo Creek were more likely to emigrate, and in addition, most movements were nocturnal.

Young humpback chub prefer shallow, low-velocity nearshore pools and backwaters, vegetated undercut banks often harbor high densities of juvenile chub in the Little Colorado River; they move to deeper and faster areas with increasing size and age, within the Little Colorado River (AZGFD 2001). In the mainstem of the Colorado River, juvenile fish may be found in backwater and other nearshore, slow-velocity areas that may serve as nursery habitats (Valdez and Ryel 1995; Robinson et al. 1998; AZGFD 2001; Stone and Gorman 2006); however, detection probability is also higher in these areas. Juvenile humpback chub (<3 years old) have been collected in all types of nearshore habitats by the Nearshore Ecology Study, with the highest density found in backwaters in both years of the study. However, the greatest number of humpback chub were collected from talus slopes (Dodrill et al. 2015). Backwater habitats are perceived to be important to juvenile native fish in the Grand Canyon; however, they are relatively rare habitats in the Little Colorado River inflow reach, and thus support only a small proportion of the native fish there (Dodrill et al. 2015). Thus the significance of these backwater habitats to the overall population is unclear.

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

The low summer flow experiment conducted in 2000 studied four nearshore geomorphic unit types between the confluence of the Colorado and Little Colorado Rivers and Lava Canyon in the summer and fall of 2010, for study periods of 10 to 27 days. Five to seven sites were studied during each interval. Persistent thermal gradients greater than the 0.2°C accuracy of the instruments were not observed in any of the sampled shoreline environments. Temperature gradients between the shoreline and mainstem on the order of 4°C, believed to be important to the habitat-seeking behavior of native or nonnative fishes, were not detected (Ross and Grams 2013). Temperature differences between main channel and nearshore habitats that are isolated from the main channel can be pronounced in backwaters and other low-velocity areas. The extent of warming is variable and depends on the timing of the daily minimum and maximum flows, the difference between air and water temperatures, and the topography and orientation of the backwater relative to solar insolation (Korman et al. 2006). For example, summertime water temperatures in backwaters have been reported to be as high as 25°C (77°F), while main channel temperatures are near 10°C (50°F) (Maddux et al. 1987). The amount of warming that occurs in backwaters is affected by daily fluctuations that drain and fill backwater habitats with cold main channel waters (Valdez 1991; Angradi et al. 1992; AZGFD 1996; Behn et al. 2010). During the low summer flow experiment, temperatures in one backwater were as much as 13°C (23°F) warmer than in the adjacent main channel during some portions of the day; temperature differences were much less at night (Vernieu and Anderson 2013). Backwater temperatures in summer have been reported to be as much as 2 to 4°C (3.6 to 7.2°F) warmer under steady flows than under fluctuating flows (Hoffnagle 1996; Trammell et al. 2002; Korman et al. 2006; Anderson and Wright 2007).

Although the use of thermal refugia such as backwaters has been documented in a variety of systems (e.g., Tyus and Haines 1991; Bodensteiner and Lewis 1992; Torgersen et al. 1999; Ebersole et al. 2001; Westhoff et al. 2014), the overall importance of backwater habitats in the Colorado River relative to humpback chub survival and recruitment is uncertain (Reclamation 2011b). While juvenile humpback chub have been reported to show positive selection for backwater habitats, the spatial extent of such habitats in the Colorado River inflow reach is small compared to other nearshore habitats such as talus slopes (Dodrill et al. 2015). Nevertheless, the presence of backwaters and other types of nearshore habitats that may be important to native fish varies by reach in the Grand Canyon and is related to sediment deposition that is dependent on the reach geomorphology (i.e., reach width and depth, etc.; reviewed in Stevens et al. 1997b).

2.1.2.6 Life History

The humpback chub is primarily an insectivore, with larvae, juveniles, and adults all feeding on a variety of aquatic insect larvae and adults, including dipterans (primarily chironomids and simuliids), Thysanoptera (thrips), Hymenoptera (ants, wasps, bees), and amphipods (such as *Gammarus lacustris*) (AZGFD 2001). Donner (2011) found that 65% of humpback chub production in the Grand Canyon was attributed to chironomids and simuliids, and that the potential for competition between humpback chub and nonnative fish was high when nonnative fish abundance was high. Feeding by all life stages may occur throughout the water column as well as at the water surface and on the river bottom. Spurgeon et al. (2015) also found

that humpback chub consumed and assimilated native fish, and that they occupied a high trophic position in the food web in a Grand Canyon tributary, similar to rainbow trout.

Adult humpback chub move into the Little Colorado River from the Colorado River to spawn from March to May (Gorman and Stone 1999). Relatively little spawning and juvenile rearing occur in the mainstem of the Colorado River, primarily because of the cold mainstem water temperatures (Kaeding and Zimmerman 1983; Minckley et al. 1991); however, in recent years, some evidence of rearing has been observed in the western Grand Canyon (Albrecht et al. 2014; Kegerries et al. 2015). Optimal spawning temperature for the species is 16°C, but mainstem water temperatures typically have ranged from 7 to 12°C (45 to 54°F) near the Little Colorado, due to coldwater releases from Glen Canyon Dam (Wright et al. 2008). Drought-induced warming has resulted in mainstem water temperatures since 2003 consistently exceeding 12°C (54°F) in the summer and fall months, which may have played a role in the estimated increase in the humpback chub population in the system since that time (Andersen 2009; Coggins and Walters 2009; Yackulic et al. 2014). Water temperatures in Havasu Creek typically exceed 16 degrees by April (GCMRC 2016), and humpback chub in spawning condition have been captured in mid-May during monitoring trips (Healy and Nelson 2013; Nelson et al. 2015).

Following spawning, larvae have been reported to drift in the Little Colorado River from April through June, and many drift out into nearshore habitats of the Colorado River (FWS 2008). Robinson et al. (1998) estimated about 38,000 larval humpback chub drifted from the Little Colorado River into the mainstem in May and June 1993. In addition, larval and juvenile humpback chub have been relatively common in the first 2 years of a study that includes sampling of nearshore habitats from Lava Falls (RM 180) to Pearce Ferry (RM 208; Albrecht et al. 2014; Kegerries et al. 2015). The natal source of these fish is unknown; however, data on the timing, age, and location of these larval humpback chub captures may suggest spawning in the mainstem or tributaries (Albrecht et al. 2015; Kegerries et al. 2015). Juveniles generally have lower monthly rates of movement than adults, with the exception of a high probability of juveniles being transported from the Little Colorado River to the Colorado River during high flows of the monsoon season, when numbers of juvenile humpback chub in the mainstem have been documented to increase by as much as 4,000 fish (Yackulic et al. 2014).

Although survival of larval and juvenile fish in the mainstem was once thought to be very rare because of seasonally constant, low water temperatures (Clarkson and Childs 2000), more recent information suggests that juveniles can successfully rear to adulthood in the Colorado River mainstem, at least under recent environmental conditions that include warmer water (Yackulic et al. 2014; Albrecht et al. 2014; Kegerries et al. 2015). Increasing water temperatures have been shown in the laboratory to increase hatching success and larval survival (Hamman 1982), larval and juvenile growth (Clarkson and Childs 2000), and improve swimming ability and reduce predation vulnerability (Ward 2011). Yackulic et al. (2014) postulated that, with warmer water, growth and survival of juveniles in the mainstem will be greater and result in increased mainstem recruitment, and thus contribute to the overall adult population.

2.1.2.7 Factors Affecting Distribution and Abundance in the Grand Canyon

Primary threats to humpback chub include habitat alterations associated with dams and reservoirs and the introduction of nonnative fishes (FWS 2011c), which act as competitors and/or predators of the humpback chub (Andersen 2009; Yard et al. 2011; Kennedy et al. 2013). In addition, the Colorado River now includes nonnative fish parasites, such as the Asian tapeworm and anchor worm, which may infect some humpback chub and affect survival (Clarkson et al. 1997; Andersen 2009). While coldwater releases from Glen Canyon Dam have been implicated in affecting reproduction and recruitment of humpback chub (and other native fishes) in the mainstem Colorado River, warmer water temperatures in the mainstem over the last decade may be providing some temporary benefit and contributing to the improving status of the humpback chub (Reclamation 2011a). Recent studies also indicated that toxic mercury (Hg) and selenium (Se) concentrations in native fish were elevated in the Grand Canyon (Walters et al. 2015). While humpback chub were not tested in the study, elevated levels of Hg in the food web, and in particular, primary prey items, including blackfly larvae (*Simuliidae*), may result in impacts on the species (Walters et al. 2015).

Population estimates indicate that the number of adult humpback chub in the Grand Canyon has been increasing since 2000 or 2001 and has been stable for about the last 5 years (Figure 8). A number of factors have been suggested as being responsible for the observed increases, including experimental water releases, trout removal, and drought-induced warming (Andersen 2009; Coggins and Walters 2009). In addition, translocations of juvenile humpback chub to Shinumo and Havasu Creeks have resulted in increased numbers of adult humpback chub captured in the mainstem aggregations (Persons et al. 2016). Translocations to tributaries have been shown to provide an adequate mechanism for rearing juvenile humpback chub that may later disperse to the Colorado River and augment aggregations (Spurgeon et al. 2015).

Dam discharge and river flow regimes have mixed results. They both may impact the shoreline rearing habitat, and thus survival of juvenile humpback chub (Converse et al. 1998). Releases such as HFEs can create shallow backwater habitats associated with sandbars and are thought to provide rearing habitat for native fish, because they can be warmer than the mainstem river during the summer months due to solar radiation (Behn et al. 2010, reviewed in Dodrill et al. 2015). Flow regimes include, for example, fluctuating flows, which destabilize backwater habitats and may impact warming and primary production (Behn et al. 2010). Backwater habitat is relatively rare in the Little Colorado River inflow area, and thus may be less important to maintaining the Little Colorado River aggregation of humpback chub than previously thought (Dodrill et al. 2015). Even though HFE water releases from Glen Canyon Dam between 2000 and 2008 may have improved some habitat characteristics (e.g., backwaters for humpback chub and other native fish), the limited availability of suitably warmwater temperatures in the mainstem may have constrained the potential for positive population responses (Kennedy and Ralston 2011). Some experimental releases, such as the November HFE in 2004, may have affected nonnative fish (possible humpback chub predators or competitors) and improved humpback chub habitat along the main channel (Korman et al. 2010). However, the March 2008 HFE may have improved the quality of spawning habitat for rainbow trout in the Lees Ferry reach, and the abundance of rainbow trout (using catch-per-unit-effort as a surrogate for abundance) in this reach was reported to be about 300% larger in 2009 than in 2007 (about

3.9 fish per minute versus 1.3 fish per minute, respectively) (Makinster et al. 2011). A similar increase in rainbow trout abundance between 2007 and 2009 was observed at the Little Colorado River confluence (RM 56–RM 69) (Kennedy and Ralston 2010). The effects of HFEs on trout abundance are discussed in more detail in Section 1.2.9. Complete evaluations of more recent fall HFEs, conducted in 2013 and 2014, are incomplete.

Predation by rainbow and brown trout at the Little Colorado River confluence has been identified as an additional mortality source affecting humpback chub survival, reproduction, and recruitment (Valdez and Ryel 1995; Marsh and Douglas 1997; Yard et al. 2011). The incidence of piscivory by brown trout has been found to be much higher than for rainbow trout in the Grand Canyon (Yard et al. 2011; Whiting et al. 2014), but rainbow trout have been much more abundant in the Colorado River, and thus may impact native fish at a similar magnitude (Yard et al. 2011). Predation by channel catfish and black bullhead are also thought to impact humpback chub in the Grand Canyon, particularly if warmer water conditions occur (NPS 2013). Because of their size, adult humpback chub are less likely to be preyed on by trout; however, emergent fry, YOY, and juvenile humpback chub are susceptible to predation in the Little Colorado River and mainstem Colorado River (Yard et al. 2011).

Experimental removal of nonnative brown and rainbow trout was conducted in the Colorado River in the Grand Canyon between 2003 and 2006. Twenty-three trips to remove trout from the vicinity of the confluence of the Little Colorado River (RM 56–RM 66) resulted in the removal of more than 23,000 fish (mostly rainbow trout). During this time, the rainbow trout population in the Colorado River in the vicinity of the Little Colorado River was reduced by more than 80% (Andersen 2009), while estimated humpback chub abundance increased during this time (Figure 9). However, this increase may be attributable to a variety of other factors, including warmer water temperatures that occurred during this time and the HFE experimental flows (Andersen 2009; Coggins et al. 2011).

As previously discussed, the coldwater temperatures in most places of the main channel are below the temperature needed for spawning, egg incubation, and growth of the humpback chub. Survival of humpback chub young in the mainstem near the Little Colorado River is thought to be low because of cold mainstem water temperatures (Clarkson and Childs 2000; Robinson and Childs 2001), which may limit hatching success, reduce larval survival and larval and juvenile growth, reduce swimming ability, and increase predation vulnerability (Ward and Bonar 2003; Ward 2011). Water temperatures in the mainstem Colorado River have generally been warmer over the last decade, and warming over the summer increases downstream, due to solar radiation. For example, maximum daily temperatures exceeded 20°C (68°F) in the lower river (RM 180-RM 280), and daily average temperature was 18.3°C (65°F) in early July (Kegerries et al. 2015). There is some evidence of recruitment at the 30-mi aggregation possibly due to the presence of warm springs. Adult chub captured near RM 35 suggests the possibility of a new aggregation or expansion of the 30-mi aggregation, and during 2013 and 2014, three female humpback chub were captured near the 30-mi aggregation that expressed eggs. Ultrasonic images of several hundred adult humpback chub from many locations in the mainstem, as well as in the Little Colorado River and Havasu and Shinumo Creeks, indicated that adult female humpback chub are able to produce eggs in the mainstem Colorado River. In 2013, approximately 33% of humpback chub examined from the mainstem Colorado River, 52% of

chub examined from the Little Colorado River, and 23% of chub examined from Havasu Creek were females with eggs (GCMRC 2015).

Temperatures, particularly in the upper reaches of the Colorado River, even in warmer years, are not optimal for humpback chub spawning and growth. However, juveniles can now successfully rear to adulthood in the Colorado River mainstem, and mainstem recruitment is likely contributing to the overall adult population that increased from about 5,000 adults in 2000 to about 11,000 adults in 2012 (Yackulic et al. 2014; Figures 8 and 9). Water temperatures below Glen Canyon Dam began increasing in 2003 as a result of drought conditions that lowered the level of Lake Powell and resulted in the release of warmer water from the dam (Andersen 2009; Andersen et al. 2010); temperatures have remained elevated relative to operations during the 1980s and 1990s due to continued drought-induced lower Lake Powell reservoir levels and somewhat due to relatively high inflow in 2008, 2009, and 2011. In 2005, maximum mainstem water temperature exceeded 15°C (59°F) at Lees Ferry and approached 18°C (64°F) in the vicinity of the Little Colorado River (RM 61), the warmest temperature at those locations since the reservoir was filled in 1980. Maximum water temperature in the mainstem at Lees Ferry reached about 14°C (57°F) in 2008 (USGS 2014), similar to temperatures in 2003 when drought effects from low Lake Powell levels began to raise Glen Canyon Dam release temperatures. In 2011, maximum mainstem water temperatures at the Little Colorado River confluence (RM 61) reached about 15°C (59°F) and 16°C (61°F), respectively (Figure 12). This warmer water appears to have benefitted the humpback chub and other native fish, but they may have benefitted nonnative warmwater species as well (Andersen 2009; Coggins and Walters 2009; Kennedy and Ralston 2011). Low reservoir levels can be attributed to drought, as well as consumptive water use in the Colorado River basin.

Increased water temperatures may also affect predation of YOY humpback chub by rainbow and brown trout (Ward 2011; Yard et al. 2011; Ward and Morton-Starner 2015). Ward (2011) reported that in the laboratory, the level of attempted predation by brown trout was positively correlated with increasing water temperature, but predation success of rainbow trout on YOY humpback chub decreased as water temperature increased from 10°C to 20°C (50°F to 68°F); predation success by brown trout did not change significantly over the same temperature range (Ward 2011; Ward and Morton-Starner 2015). Yard et al. (2011) examined the effects of temperature on trout piscivory in the Colorado River and reported no relationship between water temperature and the incidence of piscivory by rainbow trout, but a significant positive correlation was found between water temperature and the incidence of piscivory for the brown trout.

Climate change and drought have direct influences on hydrologic patterns and water temperature, thus impacting humpback chub. In the Colorado River Basin, hydrologic impacts include lower precipitation and decreased inflow to the reservoir system, resulting in more frequent lower reservoir release volumes, potentially impacting shoreline habitats and riparian areas. More frequent droughts and warmer atmospheric temperatures have the potential to result in warmer temperature of water being released from the dam. Although this may improve thermal suitability for humpback chub, any subsequent benefits may be offset by increased abundance and expansion of warmwater nonnative fish and aquatic fish parasites.



FIGURE 12 Water Temperatures at the Little Colorado River Confluence (RM 61), 1995 to Present (Source: USGS 2014)

Although the Little Colorado River stretches almost 340 m (550 km), only the headwaters and the lowermost reaches flow year-round. The lower 13 mi (21 km) of the Little Colorado River is fed by groundwater springs. This reach is occupied by the largest self-sustaining population for the species, and the lower 8 mi (13 km) is designated critical habitat. These water sources may also be vulnerable to basinwide drought and climate change impacting overall habitat availability and the population.

The development of a second population of reproducing humpback chub in the Grand Canyon was identified as an important conservation action in the 1995 EIS on the operations of Glen Canyon Dam, in case of catastrophic loss of the Little Colorado River population (Valdez et al. 2000). While some reproduction of humpback chub has been documented in Havasu Creek (Healy and Nelson 2013) near 30-mile spring (RM 30; Valdez and Masslich 1999) and may occur in the western Grand Canyon (Albrecht et al. 2015; Kegerries et al. 2015), it is likely limited in the Colorado River by water temperature, rearing habitat, or other factors.

2.1.3 Razorback Sucker

2.1.3.1 Legal Status

The razorback sucker (*Xyrauchen texanus*) was listed as endangered under the ESA on October 23, 1991 (56 FR 54957). The final rule for determination of critical habitat was

published on March 21, 1994 (59 FR 13374), and the final designation became effective on April 20, 1994. Designated critical habitat within the project area includes the Colorado River and its 100-year floodplain from the Paria River downstream through Marble and Grand Canyons to Hoover Dam, including the full pool elevation of Lake Mead.

2.1.3.2 Recovery Goals and Status

Recovery goals for razorback sucker were established in 2002 (FWS 2002b). Demographic criteria that describe numbers of populations and individuals (adults and juveniles) for downlisting and delisting are presented for upper and lower basin recovery units in the FWS's razorback sucker Recovery Goals document:

These criteria require four genetically and demographically viable, self-sustaining populations (two in each recovery unit), based on requirements of no significant decline in numbers of adults for each population and recruitment equal to or exceeding adult mortality. In addition, a genetic refuge needs to be maintained in Lake Mohave of the lower basin recovery unit (based on the majority opinion of lower basin biologists, the number of adults for maintaining this refuge is 50,000).

Monitoring is also necessary to determine if the recovery criteria are met. Adequate habitat and sufficient range are required to support recovered populations, in addition to demographic criteria.

The 5-year status review for the razorback sucker was completed by the FWS in 2012 (FWS 2012). The recovery of the species is based on whether the reduction or removal of threats has occurred, and on whether improvement in the demographic criteria has been achieved. Based on the review, only 1 of the 10 demographic criteria had been met, 2 had been partially met, and 7 were unmet. In addition, the majority of the most meaningful threats to the species were not mitigated, as only 9 of the 29 recovery factor criteria were met. As a result, the FWS decided that a change in the species' endangered status was not warranted (FWS 2012).

2.1.3.3 Historic and Current Range

The razorback sucker is a large river sucker (Catostomidae) endemic to the Colorado River system. It is a large fish, with adults reaching lengths up to 3 ft and weighing as much as 13 lb, and it may live 40 years or more (FWS 2002b). The species is endemic to large rivers of the Colorado River Basin from Wyoming to Mexico; however, the specie's range has been substantially reduced (Figure 13; Marsh et al. 2015). Currently, it occurs in the Green River, upper Colorado River, and San Juan River subbasins; the lower Colorado River between Lake Havasu and Davis Dam; Lake Mead and Lake Mohave; and tributaries of the Gila River subbasin (FWS 2002b) and Lake Powell (Francis et al. 2015).



FIGURE 13 Historic (light gray) and Present Range (dark) of Razorback Sucker in the Colorado River Basin (The lower Grand Canyon/Colorado River inflow area of Lake Mead, where razorback sucker larvae have been detected is shown in red [Albrecht et al. 2014; Kegerries et al. 2015].) Source: Marsh et al. (2015)

Recent estimates of razorback sucker populations were summarized by Marsh et al. (2015), but the authors noted that the precision of many estimates was questionable given low recapture rates of marked individuals. Despite this, collectively, the number of remaining individuals is likely only a fraction of historic populations. Many populations in the upper Colorado River Basin are maintained by stocking, and in the lower basin, with the exception of Lake Mead, razorback sucker are maintained through stocking, including populations in Lakes Mohave and Havasu (Marsh et al. 2015). Recruitment has been occurring since the 1970s, sustaining the small population remaining in Lake Mead (Albrecht et al. 2010; FWS 2012; Mohn et al. 2015), which was estimated to consist of approximately 596 individuals through 2012 (95% Confidence Interval: 468–786; Albrecht et al. 2013). Low numbers of recaptures likely precluded a population estimate for the sampling period between 2013 and 2015 in
Lake Mead (Mohn et al. 2015). Lake Mead contains one of the smaller populations; rangewide, however, recruitment is rare or nonexistent in other populations (Marsh et al. 2015).

2.1.3.4 Population within the Action Area

Inventories of fish populations in the pre-dam Grand Canyon were extremely rare, and thus describing historic fish distribution and abundance is difficult (Webb et al. 2002). Within the Grand Canyon, it is likely that razorback sucker historically occurred throughout the Colorado River to Lake Mead (after Hoover Dam construction), with several documented captures in the mainstem (near Bright Angel and Shinumo Creeks), at the Little Colorado River inflow in 1989 and 1990, and from the Paria River mouth (in 1963 and 1978, as reported in NPS 2013). Until recently, the last razorback sucker collected from the Grand Canyon (RM 39.3) was caught in 1993, and the species was considered extirpated from the Grand Canyon.

Adult razorback suckers have recently been captured from the western Grand Canyon (summarized in NPS 2013). Four fish that were sonic-tagged in Lake Mead in 2010 and 2011 were detected in the spring and summer of 2012 in GCNP up to Quartermaster Canyon (RM 260) (Kegerries and Albrecht 2012, as cited in NPS 2013). An additional untagged adult razorback sucker was captured in GCNP near Spencer Creek (RM 246) in October 2012 (as reported in NPS 2013), and another adult was captured in late 2013 (GCMRC 2014). Sampling of channel margin habitats has also documented 462 and 81 razorback sucker larvae as far upstream as RM 179 (just upstream of Lava Falls) in 2014 (Albrecht et al. 2014) and 2015 (Kegerries et al. 2015), respectively, indicating that spawning is occurring in the mainstem river in the western Grand Canyon (Albrecht et al. 2014; Kegerries et al. 2015). However, smallbodied fish sampling designed to detect juvenile razorback sucker in western Grand Canyon has failed to detect any older larval fish or those that have transformed to juveniles.

Tagged adult razorback suckers have also been located as far upstream as RM 184.4 near Lava Falls, and along with the collection of larvae, indicates that the species utilizes the Colorado River above the Lake Mead inflow area more than previously thought (Albrecht et al. 2014). In 2015, Submersible ultrasonic receivers (SURs), which are devices that detect sonic-tagged razorback suckers, were installed upstream of Lava Falls, to an area below Bright Angel Creek. No detections of razorback sucker were recorded above Lava Falls through September 2015; however, the continued collection of larval fish upstream of Lava Falls (Kegerries et al. 2015) indicates spawning is occurring in an unknown location in the mainstem or tributaries.

Occurrences since 2013 of adult and larval razorback sucker in Lake Mead and the lower Grand Canyon downstream of RM 180 indicate that the connectivity of the lake to the riverine reaches may be important to maintenance of razorback sucker in the project area.

2.1.3.5 Habitat

Razorback sucker use a wide variety of habitats, including rivers and streams and their associated floodplains, as well as reservoirs (Valdez et al. 2012a). Habitat use varies by life stage and by habitat type (i.e., lentic vs. lotic habitats). Valdez et al. (2012a) completed a comprehensive review of razorback sucker habitat use studies conducted throughout the species' range. In rivers, habitat requirements of adults in spring include low-velocity runs, eddies, backwaters, and areas of inundated vegetation (e.g., flooded off-channel areas); in summer, runs and mid-channel bars; and in winter, low-velocity runs, pools, and eddies (reviewed in Valdez et al. 2012a). Razorback suckers spawn on cobble and gravel bars in rivers, and on gravel and cobble shorelines in reservoirs. In reservoirs, adults prefer areas with water depths of 3 ft or more over sand, mud, or gravel substrates. Young require nursery areas with quiet, warm, shallow water such as tributary mouths, backwaters, and inundated floodplains along rivers and coves or shorelines in reservoirs (FWS 2002a). Recent captures of larval razorback sucker in the western Grand Canyon found the highest density of larvae in isolated pools, which comprised less than 2% of all habitat sampled (Albrecht et al. 2014). Similar results were found in 2015, when the highest catch of larval razorback sucker was found in isolated pools, followed by backwaters, which composed 2.1% and 9.1% of habitats sampled, respectively (Kegerries et al. 2015). Larval razorback sucker may drift along the shoreline adjacent to the main channel until settling into warmer, shallow backwaters, or floodplain wetlands (Valdez et al. 2012a).

Critical habitat was designated for this species in 1994, and within the project area, including the Colorado River and its 100-year floodplain from the confluence of the Paria River downstream to Hoover Dam (a distance of about 500 mi), including Lake Mead to full pool elevation (59 FR 13374).

2.1.3.6 Life History

Razorback suckers are long-lived (40+ years) and exhibit relatively fast growth the first 5 to 7 years of life in warm, food-rich environments, until adulthood, and after which growth slows and possibly stops (FWS 2002b). Both sexes are sexually mature by age 4 and can reach a maximum size of about 1 meter. As described above, spawning in rivers occurs over bars of cobble, gravel, and sand substrates during spring runoff at widely ranging flows and at water temperatures as low as 6°C (reviewed in Marsh 1985), but typically greater than 14°C (57°F) (FWS 2002b). High flow and temperature cues during spring were thought to be important to the initiation and success of spawning (reviewed in FWS 2002b). For example, vegetated floodplain wetlands and other off-channel habitats inundated by spring high flows provide warmer, food-rich rearing habitats for drifting larval razorback sucker (reviewed in Modde et al. 2001). High flows may also have flushed fine sediment from cobble bars, maintaining interstitial space for egg deposition and incubation. In reservoirs, spawning occurs over wave-swept rocky shoals and shorelines.

Historically, this species exhibited upstream migrations in spring for spawning, although current populations include groups that are sedentary and others that move extensively (Minckley et al. 1991). Adults in the Green River subbasin have been reported to move as much as 62 mi to specific areas to spawn (Tyus and Karp 1990). In Lake Mohave, individuals have been reported to move 12 to 19 mi between spring spawning and summer use areas (Mueller et al. 2000). Sonic-tagged razorback sucker, either stayed near spawning areas, or have moved great distances, up to 361 km (224 mi) within the western Grand Canyon, the Colorado River inflow to Lake Mead, and throughout the lake (Kegerries et al. 2015).

Razorback sucker have high fecundity, with the average number of eggs per female ranging from 27,614 to 103,000 in various studies (reviewed in FWS 2002b). Eggs incubate for 6 to 7 days before hatching, prior to emerging from cobbles and being transported downstream to floodplains or backwaters. Hatching success is temperature dependent, with the potential for complete mortality occurring at temperatures less than 10°C (50°F; Marsh 1985; Bozek et al. 1990). Marsh (1985) found the highest survival of embryos and percentage hatched was at 20°C in laboratory studies, while Bozek et al. (1990) showed that hatching success was between 32 and 65% in 15°C water.

In the Grand Canyon, the estimated onset of spawning was in mid-February, based on back-calculation from the dates of larval collection, when average daily water temperatures were between 10 and 12°C (50 and 53.6°F; Kegerries et al. (2015). Spawning appeared to peak toward the end of March when water temperatures ranged from 12 to 14°C, but the duration of spawning was protracted—occurring between mid-February and July (Kegerries et al. 2015). A faster larval razorback sucker growth rate is thought to benefit survival by minimizing the time larvae are susceptible to predation by small-bodied fish. Growth was 50% greater at 19.5°C compared to growth in 16.5°C water, with the highest growth rates at 25.5°C (Bestgen 2008). As described above, following hatching, larval razorback sucker may drift into low-velocity, warm, backwaters or floodplain wetlands to rear. Maturity is reached in 2 to 5 years.

Both adults and immature fish are omnivorous, feeding on algae, zooplankton, and aquatic insect larvae; however, diet varies by life stage. In Lake Mohave, their diet has been reported to be dominated by zooplankton, diatoms, filamentous algae, and detritus (Marsh 1987). Larval razorback sucker feed on planktonic organisms, switching to benthic organisms as their sub-terminal mouth develops, with chironomid larvae being a primary food item (FWS 2002b).

2.1.3.7 Factors Affecting Distribution and Abundance in the Grand Canyon

The decline of the razorback sucker in the Grand Canyon and throughout its range has been attributed primarily to habitat modification due to dam construction (including coldwater dam releases, habitat loss, and migration impediments), streamflow regulation, and predation by nonnative fish species, which have resulted in a lack of recruitment (FWS 2002b; Gloss and Coggins 2005). For example, the estimated 80% rangewide reduction in distribution of this species has been attributed to the construction of Hoover, Parker, Davis, and Glen Canyon Dams on the Colorado River and Flaming Gorge Dam on the Green River (Valdez et al. 2012a). Similar to the humpback chub, as described above, cold hypolimnetic releases from dams,

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

including Glen Canyon Dam, have also contributed to reproductive failure in razorback sucker (Gloss and Coggins 2005). Flow regulation and water diversions have decreased the magnitude of spring peak runoff, which is closely linked to reproduction of the razorback sucker. The loss or drastic reduction in peak flows, along with channelization or disconnection of floodplain nursery habitats with the main channel (as a result of loss of peak flows), have resulted in the elimination of reproduction and recruitment (FWS 2002b). In 2012, the FWS (2012) determined that flow regimes necessary to establish and maintain razorback sucker populations in the lower basin, including flows that provided adequate spawning cues and spawning and nursery habitat, had not been established. Similar to impacts on humpback chub, elevated Hg and Se described by Walters et al. (2015) may be another factor that may impact razorback sucker. While not tested, other native suckers with similar diets were found to have high levels of Hg and Se in the Grand Canyon (Walters et al. 2015).

The large reservoirs formed by dams also provide habitat for nonnative sport fishes, which were stocked with nonnative sport fish predators. Competition with and predation by nonnative fishes have also been identified as important factors in the decline of this razorback sucker (Minckley et al. 1991; FWS 2002a). The reduced sediment supply and resulting clear water due to dam operations also is thought to favor sight-feeding nonnative predators, over razorback sucker and other native fish that evolved in highly turbid conditions (reviewed in Gloss and Coggins 2005). Large-scale nonnative fish control and management programs have been implemented to meet recovery goals for razorback sucker in the upper basin, including control of small- and large-bodied predators in the San Juan and Upper Colorado River Basins, and revising nonnative fish stocking procedures (FWS 2012). Marsh et al. (2015) professed that only complete elimination of nonnative fish from razorback sucker spawning and nursery areas, such as off-channel ponds, would lead to successful conservation of the species in Lake Mohave. The success of stocking programs has been limited by the presence of nonnative predators. For example, Schooley and Marsh (2007) found that few individuals survived out of millions of small razorback suckers that had been stocked in the lower basin, likely because of predation by nonnative fish, and larger stocked fish had a much higher likelihood of survival.

In the Grand Canyon, the decline of native fish, including razorback sucker, has been attributed in large part to an increased diversity and abundance of nonnative fishes, along with the effects of Glen Canyon Dam on water temperatures, flow, and sediment (Gloss and Coggins 2005). As described above, recent efforts to better understand the use of the western Grand Canyon by razorback sucker has revealed that the species is more widespread there than previously thought. Spawning has occurred in unknown locations in the river or tributaries from Lake Mead to at least to areas upstream of Lava Falls in the Grand Canyon, or in the inflow areas of tributaries to Lake Mead, which has led to the maintenance of a small, but recruiting population in the project area (Albrecht et al. 2014). It is unclear whether razorback sucker spawning had been occurring previous to larval fish studies initiated in the western Grand Canyon in 2014.

Through the first 2 years of the first phase of a study to determine the habitat suitability and status of razorback sucker in the Grand Canyon (see NPS 2013), it was found that adult razorback sucker moved throughout the study area, and spawned, but no habitat preference could be inferred. Larval razorback sucker catches were higher in isolated pools and backwaters than

other nearshore habitats that were sampled (Kegerries et al. 2015). Despite the presence of adults and larval fish, no juvenile razorback sucker had been captured (Kegerries et al. 2015). Plausible explanations for this, assuming appropriate habitats are sampled for juvenile suckers, could be that all larval suckers are consumed by nonnative fish prior to their transformation to juveniles, or suitable low-velocity nursery habitats with sufficient food and water temperatures may not be available in the western Grand Canyon. However, relatively few nonnative fish have been sampled in the past 2 years in nearshore habitats from Lava Falls to Pearce Ferry, compared to native fish (Albrecht et al. 2014; Kegerries et al. 2015), and Valdez et al. (2012a) observed that there were backwaters similar to those on the San Juan River that are used by razorback sucker larvae, downstream of Lava Falls. Since the flow and temperature regime of the Colorado River in the Grand Canyon has been altered (i.e., reduced sediment supply, lack of high annual peaks, daily fluctuating flows, and cold dam releases), leading to less backwater stability and more frequent flushing (Behn et al. 2010), these backwaters may be less suitable for rearing than under stable summer flows (see Trammell et al. 2002; Vernieu and Anderson 2013). Alternatively, larval razorback sucker may drift out of the Grand Canyon and rear in the Colorado River Inflow of Lake Mead, which may provide suitable cover (turbidity) and temperatures (Valdez et al. 2012b; Kegerries et al. 2015). Captures of an age-2 and an age-3 immature razorback sucker, and an age-0 juvenile in or near the Colorado River Inflow of Lake Mead, suggests that some recruitment is occurring associated with the Colorado River in the Grand Canyon (summarized in Kegerries et al. 2015), despite altered habitats.

The presence of razorback suckers and their use of habitats in the Grand Canyon have only recently been investigated, and it is unclear how long spawning has been occurring, since larval fish studies were not conducted prior to 2014. Recent warming river temperatures due to lower Lake Powell elevations, attributed to drought and consumptive water use, may have resulted in more suitable habitats in the western Grand Canyon. For example, in 2015, river temperatures were within the acceptable range needed for razorback sucker spawning and successful hatching, particularly farther downstream (Kegerries et al. 2015). In addition, fish community composition in the lower river below Diamond Creek has changed dramatically from one dominated by nonnative species, to native species (reviewed in Kegerries et al. 2015). However, the cause of the change in fish community composition is unknown. The drop in nonnative predator abundance, combined with periodically warmer water temperatures, may have allowed for the expansion of razorback sucker into the western Grand Canyon. Additional research and monitoring are needed to better understand the management implications for recovery of razorback sucker in this reach of its range (Albrecht et al. 2014).

More frequent droughts and warmer atmospheric temperatures as a result of climate change and warmer temperature of water being released from the dam are likely to impact razorback suckers in the Grand Canyon. Razorback suckers are currently located in the lower reaches of the canyon where backwater habitat is more available. Warmer water temperatures may create more suitable habitat for larvae and juveniles. However, these effects may be offset by increased abundance and expansion of nonnative fish and aquatic fish parasites. Also, lower reservoir releases may result in fewer backwater habitats available for spawning and rearing of razorback sucker.

2.1.4 Southwestern Willow Flycatcher

2.1.4.1 Legal Status

The southwestern willow flycatcher (*Empidonax traillii extimus*) is one of four currently recognized willow flycatcher subspecies (Phillips 1948; Unitt 1987; Browning 1993). On March 29, 1995, the southwestern willow flycatcher was listed as endangered (60 FR 10694) in its entire range, which is known to include Arizona, California, Colorado, New Mexico, Texas, Utah, and Mexico. In August 2002, the FWS released the "Final Recovery Plan for the Southwestern Willow Flycatcher." The Recovery Plan establishes six recovery units that are further subdivided into management units. These Recovery and Management Units are based on watershed and hydrologic units within the breeding range of the flycatcher (FWS 2002c). The Grand Canyon is within the Lower Colorado Recovery Unit. This Recovery Unit encompasses the Colorado River and its tributaries from Glen Canyon Dam downstream to the Mexican border. Despite the large size of this Recovery Unit, the unit contains only 146 known territories (15% of the rangewide total) (FWS 2002c).

2.1.4.2 Recovery Goals and Status

The Recovery Plan and goals for the southwestern willow flycatcher were completed in 2002 (FWS 2002c). The recovery objective for the species is to attain a population size and distribution of habitat sufficient to maintain metapopulations over the long term. Threats to the species must be addressed to achieve this goal. In general, the minimum number of territories needed and maintained over a 5-year period is 1,950 (approximately 3,900 individuals), distributed geographically to allow for functioning metapopulations. In addition, the total known population would need to be increased to 1,500 territories (3,000 individuals) distributed geographically among the Management Units and Recovery Units, so that the species is no longer in danger of extinction.

The 5-year status review for the southwestern willow flycatcher was completed by the FWS in 2014 (FWS 2014). It was determined that the status of the species had improved due to an increase in the number of known territories since it was listed in 1995. However, ongoing threats related to the spread of tamarisk leaf beetle and land management remained concerns. Due to these ongoing threats, a decline in the rangewide distribution of the species, and anticipated future adverse effects of the tamarisk leaf beetle to its habitat, the FWS determined that no change in status was warranted.

2.1.4.3 Historic and Current Range

The southwestern willow flycatcher is a neotropical migrant that breeds in the southwestern United States and migrates to Mexico, Central America, and possibly northern South America during the non-breeding season (Phillips 1948; Stiles and Skutch 1989; Peterson 1990; Ridgely and Tudor 1994; Howell and Webb 1995). The Pacific lowlands of

Costa Rica appear to be a key winter location for the southwestern willow flycatcher, although other countries in Central America may also be important (Paxton et al. 2011). Historically, the range of the flycatcher in Arizona included portions of all major watersheds (FWS 2002c); however, these watersheds have changed in many cases. As a result, most of the areas where flycatchers were locally abundant now support few or no individuals (FWS 2002c). Habitat and population numbers of southwestern willow flycatchers have declined in recent decades due to several factors, including loss, degradation, and fragmentation of riparian habitat; invasion by nonnative plants; brood parasitism by brown-headed cowbirds; and loss of wintering habitat (Stroud-Settles et al. 2013).

2.1.4.4 Population within the Action Area

Seventeen flycatcher sites were identified in the 2002 Recovery Plan (FWS 2002c) within the Grand Canyon. Flycatcher territories in the Grand Canyon are generally located in the tamarisk-dominated riparian vegetation along the river corridor but not in the mesquite-acacia and hackberry-dominated habitats higher on the slopes (Sogge et al. 1997a). The flycatcher's nesting habitat is dynamic in that it varies in occupancy, suitability, and location over time. In GCNRA, southwestern willow flycatchers are uncommon restricted migrants in riparian areas, rare summer residents, and probable breeders (Spence et al. 2011). Historic and recent nesting site locations in the Grand Canyon have been documented below Lees Ferry in Marble Canyon and in lower Grand Canyon below Diamond Creek (RM 225.5-RM 277) (Appendix B). No southwestern willow flycatcher nests or nesting behavior have been identified in the inner gorge (RM 77.9–RM 116.5); however, migrant birds have been documented. Because river channels, river flows, and floodplains are varied and can change over time, the location and quality of nesting habitat may also change over time. This is especially noticeable in the lower Grand Canyon where dropping lake levels in Lake Mead have resulted in high walls (approximately 10 to 20 ft high in many areas) of sediment topped with tamarisk bordering the Colorado River. The backwaters and saturated soils preferred by southwestern willow flycatchers have become rare.

Numbers of southwestern willow flycatcher detections in the Grand Canyon have declined since the 1980s, which is based on studies referenced below. There is little information on the number of willow flycatchers along the river before the construction of the Glen Canyon Dam. However, what data are available suggest that southwestern willow flycatchers were not common breeders along the Colorado River in the Grand Canyon (Brown 1988a,b; Brown 1991; Sogge et al. 1997b). Studies conducted along the river from 1982 to 1991 between Lees Ferry and Phantom Ranch found a total of 47 adult southwestern willow flycatchers—14 pairs, and 15 nests (Brown 1988a; Sogge et al. 1997a). From 1992 to 2001, the breeding population fluctuated between 1 and 4 breeding pairs per year with a total over the 10 years of 66 adult southwestern willow flycatchers, 14 pairs, and 20 nests (Brown 1988a,b; Brown 1991; Sogge et al. 1997a; Sogge and Tibbitts 1992; Sogge et al. 1993; Sogge and Tibbitts 1994; Sogge et al. 1995; Petterson and Sogge 1996; Sogge 1998; Yard 2001, 2002, 2003). Appendix B summarizes southwestern willow flycatcher observation data, territories, and nesting sites located in the Grand Canyon from 1909 to 2011. Although surveys were conducted in 2012, southwestern willow flycatchers were not detected. Survey methods followed the most recent

USGS survey protocol, at the time of survey, and involved the use of broadcast calling to elicit southwestern willow flycatcher responses (Sogge et al. 2010). Nest searches were not conducted during the 2010–2012 surveys at the Grand Canyon; however, due to behavior and timing of southwestern willow flycatcher detections at RM 275 in 2010, two breeding pairs were suspected to be present. Appendix A and Appendix B provide information on the locations of birds and nests from 1982 to 2011. Data are based on the following monitoring reports: Sogge et al.1997a; McKernan and Braden 1997, 1998, 1999, 2000, 2001, 2002, 2004; Paradzick et al. 2001; Smith et al. 2002, 2003, 2004; Koronkiewicz 2004, 2005, 2006, 2007, 2008; Yard et al. 2004; Albert 2005; Laczek-Johnson and Ward 2006; Ward and Haynes 2007; Northrip et al. 2008; Slayton et al. 2010; Stroud-Settles and Lawrence 2011.

After the 2004 survey season, the USGS and GCMRC elected to discontinue their monitoring of known southwestern willow flycatcher nesting habitat in the Grand Canyon. Beginning in 2005, the Grand Canyon conducted annual surveys from Lees Ferry to Phantom Ranch, but funding prevented surveying the isolated habitat patches between Phantom Ranch and Diamond Creek. From 2004 to 2008, only two southwestern willow flycatchers were detected between Lees Ferry and Phantom Ranch.

2.1.4.5 Lees Ferry to Diamond Creek

Suitable habitat is located disjunctly through the river corridor from approximately RM 28.3 to RM 275. Surveys conducted between 1992 and 2004 indicated a small resident breeding population between Lees Ferry and Cardenas Marsh (RM 71), but no territories from RM 71 through RM 246 have been located. Recent (2007–2009) surveys have only detected non-resident/migratory flycatchers between Lees Ferry and Phantom Ranch (Palarino et al. 2010). Formal nest searches have not been conducted above Diamond Creek since 2004 (Stroud-Settles et al. 2013).

From 1993 to 2004, flycatchers were consistently present during the breeding season at RM 50.5–RM 51.5, but have not been present since 2004 (Ward and Haynes 2007; Northrip et al. 2008). In 2003, 2004, and 2010, the area around RM 28–RM 29 was occupied. Another area of importance in the mid-1990s was RM 71–RM 71.5; however, this area does not appear to have been occupied for the last 17 years. In 2004, the Grand Canyon instituted an emergency closure at two sites. This closure was in effect between May 1 and July 15 and included closure of visitor use, including hiking, camping, and river landings at RM 28.1–RM 28.5, river left, and RM 50.2–RM 50.6, river left. Closures at RM 28 and RM 50 have been put in place intermittently in the past; closure at Cardenas (RM 71) was instituted in the early and mid-1990s.

2.1.4.6 Lower Grand Canyon

Koronkiewicz et al. (2004) reported that the Colorado River in the Grand Canyon downstream of Separation Canyon (RM 234) is strongly influenced by water levels in Lake Mead. Potential southwestern willow flycatcher habitat in this area has changed dramatically in

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

the last several years as the result of a 105-ft drop in the level of Lake Mead since 2000. Areas that were inundated in the late 1990s are now well above the current water level, and the existing riparian vegetation in many of these areas is dead or dying. A 5-year boost in detections along this stretch of river that occurred from 1997 to 2001 is likely due to favorable water levels of Lake Mead in combination with increased survey effort (Stroud-Settles et al. 2013). The river corridor continues to provide essential habitat for migrating southwestern willow flycatcher (Stroud-Settles et al. 2013).

Southwestern willow flycatchers have been detected in the lower Grand Canyon below RM 234 since 1995 with the exception of 2002, 2003, 2011, and 2012 (Appendix B). In 2004, Koronkiewicz et al. identified approximately 76 hectares of suitable habitat at several sites between RM 239 and RM 275 within the Grand Canyon. These disjunct habitat patches have been inconsistently monitored during the past 11 years for both flycatcher presence and habitat suitability. Suitability ranking of these sites has proven to be largely dependent on current hydrological conditions of the Colorado River. As a result, a habitat assessment survey conducted during one year may result in a habitat ranking that is deemed suitable, but a revisit to the same site during a different year may rank the site as only potential habitat.

2.1.4.7 Life History and Habitat

The southwestern willow flycatcher eats insects and needs riparian habitats to complete its life cycle. It breeds and forages in dense, multi-storied riparian vegetation near saturated soils, slow-moving water, or surface water (Sogge et al. 1995). The southwestern willow flycatcher breeds across the lower Southwest from May through August (Reclamation 2007b). The southwestern willow flycatcher arrives on the breeding grounds throughout May and early June, eggs are generally laid beginning in May, and fledging occurs between June and August (Sogge et al. 1997a, 2010). Occupied sites most often have a patchy interior of dense vegetation or dense patches of vegetation intermingled with openings. Most often, this dense vegetation occurs within the first 3 to 4 m above the ground (FWS 2002c). The structures of occupied patches vary, with a scattering of small openings, shorter vegetation, and open water. Occupied patches can be as small as 2 acres and as large as several hundred acres, but are typically >10 m wide (Reclamation 2007b).

The southwestern willow flycatcher breeds in dense riparian habitats from sea level in California to approximately 8,500 ft in Arizona and southwestern Colorado. Throughout its range, the southwestern willow flycatcher arrives on breeding grounds in late April and May (Sogge and Tibbitts 1992; Sogge et al. 1993; Sogge and Tibbitts 1994; Muiznieks et al. 1994; Maynard 1995; Sferra et al. 1995, 1997). Nesting begins in late May and early June, and young fledge from late June through mid-August (Willard 1912; Ligon 1961; Brown 1988a,b; Whitfield 1990; Sogge and Tibbitts 1992; Sogge et al. 1993; Muiznieks et al. 1994; Whitfield 1994; Maynard 1995). The entire breeding cycle, from egg laying to fledging, is approximately 28 days. Nesting occurs during the spring and early summer months (May 1–August 31) in the Grand Canyon.

Historical egg/nest collections and species descriptions throughout its range identify the southwestern willow flycatcher's widespread use of willow (*Salix spp.*) for nesting (Phillips 1948; Phillips et al. 1964; Hubbard 1987; Unitt 1987; San Diego Natural History Museum 1995). Other habitats are also used, including nonnative species such as tamarisk (*Tamarix spp.*) and Russian olive (*Eleagnus angustifolia*). Throughout the southwestern willow flycatcher's current range, suitable riparian habitats tend to be rare, widely separated small and/or linear locales separated by vast expanses of arid lands.

2.1.4.8 Factors Affecting Distribution and Abundance in the Grand Canyon

The southwestern willow flycatcher has experienced extensive loss and modification of habitat and is also endangered by other factors, including brood parasitism by the brown-headed cowbird (*Molothrus ater*). The southwestern willow flycatcher was listed primarily due to riparian habitat reduction, degradation, and elimination as a result of agricultural and urban development. Other reasons for the decline/vulnerability of the flycatcher include the fragmented distribution and low numbers of the current population; predation; and other events such as fires and floods that are naturally occurring, but have become more frequent and intense as a result of the proliferation of exotic vegetation and degraded watersheds, respectively.

The Grand Canyon does not provide extensive stands of dense riparian habitat suited for breeding southwestern willow flycatchers. The majority of habitat patches in the Grand Canyon lack a consistent, dependable source of water for maintaining moist/saturated soil conditions and/or slow-moving or standing surface water (Stroud-Settles et al. 2013). As a result, the majority of flycatcher habitats in the Grand Canyon are marginal and, unless current hydrological conditions change, these patches will likely continue to decline. Furthermore, the recent arrival of the tamarisk leaf beetle has transformed and will continue to transform the patches of dense tamarisk into unpredictable, diminished patches (Stroud-Settles et al. 2013). Effects of climate change and continued dry basin hydrology may further diminish habitat for the flycatcher due primarily to lower water levels within riparian areas.

2.1.5 Yuma Ridgway's Rail

2.1.5.1 Legal Status

The Yuma Ridgway's rail (*Rallus obsoletus yumaensis*) was listed as endangered on March 11, 1967 (32 FR 4001). A 5-year review of the species was completed in 2006, and currently the 1983 Recovery Plan is in the revision process; a draft was released in 2010 for public review (FWS 2009b), and finalization is expected in 2016. It is categorized as a subspecies with a high degree of threat and low recovery potential due to habitat loss.

2.1.5.2 Recovery Goals and Status

Recovery goals and objectives for Yuma Ridgway's rail are based on the need to achieve and maintain a viable population level and to protect and maintain a sufficient amount of core and other habitats to support the population. The primary threat to the species is that water sources supporting its habitat are not managed and protected, and as a result, habitat may be lost. The draft recovery goal is "To achieve population stability and habitat protection sufficient to downlist and/or delist the Yuma Ridgway's rail" (FWS 2009b).

The 5-year status review for Yuma Ridgway's rail was completed in 2006 (FWS 2006). Based on the review, it was determined that the population in the United States is small, and limited information is available to determine the species' demographic status. Habitat loss in a Mexican population's range and accumulation of selenium in the environment remain significant threats, and thus no change in status was warranted.

2.1.5.3 Historic and Current Range

The Yuma Ridgway's rail occurs along the lower Colorado River and tributaries (Muddy, Virgin, Bill Williams, Lower Gila and Salt Rivers) in Arizona, California, Nevada, and Utah; the Salton Sea in California; and the Cienega de Santa Clara and Colorado River Delta in Mexico (FWS 2009b). Significant breeding areas in the United States include Mittry Lake (Arizona), Imperial Reservoir, Imperial National Wildlife Refuge, Bill Williams River National Wildlife Refuge, Topock Gorge and Topock Marsh in Havasu National Wildlife Refuge, Cibola National Wildlife Refuge along the lower Colorado River and Imperial Wildlife Area, Sonny Bono Salton Sea National Wildlife Refuge at the Salton Sea in California, and the Cienega de Santa Clara in Mexico. There is a small population in southern Nevada/northwest Arizona on the Muddy and Virgin Rivers.

2.1.5.4 Population within the Action Area

McKernan and Braden (1999) reported the presence of Yuma Ridgway's rails between Spencer (RM 246) and the boundary of the Grand Canyon (RM 277); these observations were made while conducting southwestern willow flycatcher surveys in the area. Specifically, McKernan and Braden (1999) report at least one rail observed between May 26, 1996, and June 30, 1996, and they indicate that nesting was confirmed. They report at least one rail observed between May 14, 1997, and June 17, 1997, but indicate that nesting was not confirmed (McKernan and Braden 1999). Surveys for the southwestern willow flycatcher in the lower Grand Canyon from 2003 to 2008 under the Lower Colorado River Multi-Species Conservation Program (LCRMSCP) did not record any incidental detections of Yuma Ridgway's rails (LCRMSCP 2008). GCNP surveys for southwestern willow flycatchers in the lower Grand Canyon in 2010 to 2012 did not have any incidental detection; however, the method of surveying used would be unlikely to pick up rail vocalizations. The LCRMSCP mapped marsh habitat in the lower Grand Canyon/upper Lake Mead from aerial imagery taken in 2004; that mapping extended from RM 249 (401 km) to RM 277 (446 km), all within the boundary of GCNP. Koronkiewicz et al. (2004) and LCRMSCP (2004) report the presence of live cattails at Spencer Canyon (RM 246) and Burnt Springs (RM 259.5). The persistence of the cattail patches identified in the 2004 imagery and as documented in 2004 and 2005 is unknown. The continuing fall in Lake Mead elevation (60 ft to 18 m from 2004– 2015) results in a longer reach of the river, and shallow bays or coves along the channel have become dewatered. Further, within the full-pool of Lake Mead, sediments have built up over time during higher lake elevations forming a delta. The retreating lake flows allows the erosive action of the river to cut through those sediments and move them downstream. The result can be steep eroding banks with riparian and marsh vegetation on top beginning to dry out as the water recedes with the lowered channel as more and more sediment is removed. With these changes, it is not known if cattail marsh habitat is present in sufficient quantity to allow for nesting.

Because of the limited information about the Yuma Ridgway's rail and its habitat in the Grand Canyon, managers must rely heavily upon the limited information available. Given that Yuma Ridgway's rails have been recorded historically at the Grand Canyon but have not been surveyed consistently or recently, managers presume that the Yuma Ridgway's rail may be present in the lower Grand Canyon during the lifetime of the LTEMP.

2.1.5.5 Habitat and Life History

The Yuma Ridgway's rail is a secretive species and is not often seen in the wild; however, it does have a series of distinctive calls and is most often identified by those. This bird inhabits freshwater or brackish stream sides and marshes under 4,500 ft (1,372 m) in elevation. It is associated with dense riparian and marsh vegetation, dominated by cattails (Typha sp.) and bulrush (Scirpus ssp.) with a mix of riparian tree and shrub species. Yuma Ridgway's rails may climb into a shrub or tree, but overall, they do not perch above the ground (FWS 2009b). Rails are capable of swimming and are also known to dive underwater, and may hold onto submerged vegetation to avoid threats or use its wings to "swim" (Todd 1986; Ripley 1977 cited in Eddleman and Conway 1998). The rail requires a wet substrate such as a mudflat, sandbar, or slough bottom that supports cattail stands of moderate to high density adjacent to shorelines. Other important factors are the presence of vegetated edges between marshes and shrubby riparian habitat (tamarisk or willow thickets) and the amount and rate of water-level fluctuations. Nests are built 3 to 6 in. (8 to 16 cm) above the surface in sloughs and backwaters that support dense stands of bulrush and cattails, and breeding occurs from March to early July. Along the lower Colorado River, males begin calling in February and pair bonding occurs shortly after. Nonnative cravfish provide the primary food base for the rail today; prior to the introduction of crayfish, isopods, aquatic and terrestrial insects, clams, plant seeds, and small fish likely dominated their diet (LCRMSCP 2008).

Eddleman (1989) determined that vocalizations are significantly reduced in winter, and telemetry data indicated that the majority of rails do not migrate. There is evidence that some populations may be more migratory than others, and this could be based on habitat and a stable food source (Eddleman 1989; Corman and Wise-Gervais 2005). Very little is known about the

dispersal of adult or juvenile birds, but the expansion of the population northward along the lower Colorado River (to Lake Mead, the Salton Sea, and up the Gila and Salt Rivers in central Arizona over the last 80 years) indicates that dispersal does occur (FWS 2009b).

2.1.5.6 Factors Affecting Distribution and Abundance in the Grand Canyon

Historically, the primary concentrations of Yuma Ridgway's rails were likely to be found in cattail/bulrush marshes in the Colorado River Delta. Unfortunately, due to diversions from the river for agriculture and municipal uses, the freshwater flows down the lower Colorado River, necessary to maintain marsh habitat, have virtually been eliminated (FWS 2009b). The majority of Yuma Ridgway's rail habitats that exists today are mostly human-made, such as the managed ponds at the Salton Sea (FWS 2009b). Without active management and protection of water resources to address land use changes in floodplains, human activities, environmental contaminants, and reductions in connectivity between core habitat areas, these habitats will be permanently lost to the Yuma Ridgway's rail (FWS 2009b). Effects of climate change and continued dry basin hydrology may further diminish habitat for the Yuma Ridgway's rail due primarily to lower water levels within riparian areas.

Another specific threat to the Yuma Ridgway's rail includes selenium in crayfish, the major prey item of the species. Selenium levels in crayfish collected in Yuma Ridgway's rail habitat were high enough to cause concern for the rail's reproductive success (FWS 2009b). No adverse effects from selenium have been observed; however, due to the rail's secretive nature, nests are very difficult to find and young birds difficult to observe (FWS 2009b).

2.2 BASELINE TAKE RESULTING FROM HANDLING AND GCDAMP RESEARCH

2.2.1 Humpback Chub

Fisheries monitoring and research are important aspects of ongoing research in the Grand Canyon that assesses the impacts of adaptive management and experimental treatments upon humpback chub and other resources. An extensive fisheries monitoring and research program is being implemented by the GCDAMP through the USGS and GCRMC throughout the project area, which will continue under the LTEMP proposed action. While stress upon individual humpback chub may result in occasional fatality, an objective of monitoring is to gather information in order to ensure that LTEMP goals for humpback chub are met.

Stress upon individual humpback chub related to sampling using netting and electrofishing gears, as well as handling by researchers, is also a potential source of injury or mortality to individuals. No immediate mortality of humpback chub has been documented during tempering prior to release, hoop-netting, or electro-fishing during monitoring associated with NPS translocation projects in Shinumo and Havasu Creeks. Humpback chub were captured and handled 1,068 times between June 2010 and September 2012 in Shinumo and Havasu Creeks (Healy et al. 2014), for example. However, some minimal mortality has occurred during juvenile collection efforts at the Little Colorado River for translocations and refuge development, although it is generally less than 5% (Van Haverbeke 2012). Five to 10 fish out of 500 to 800 generally die during transport in route to hatchery rearing facilities and during parasite and disease treatments; however, approximately 60 (10%) were lost at a hatchery facility during disease treatments and tagging prior to translocations in 2011. In 2014, approximately 30 out of 600 died during transport to the hatchery facility due to excessive cooling. Mortality of humpback chub captured during monitoring by all management agencies and researchers is reported to the FWS annually.

Incidental fatality was reported to the FWS by the USGS and GCMRC biology program manager (S. VanderKooi) and is summarized here. In 2012, 22 incidental fatalities out of 7,755 total humpback chub captures (0.28% mortality rate from capture) were associated with fish monitoring in the Colorado River using a variety of gear types. Another 4 humpback chub were lost out of 725 captures in the lower sections of the Little Colorado River (0.55%). Incidental mortalities of subadult or juvenile humpback chub (< 200 mm TL) included 88 in 2013 and 36 in 2014. The highest mortality rate was associated with experimental collections and rearing of larval humpback chub to be used in translocations (close to 30% of fish captured). Larval humpback chub collections may continue, or potentially increase slightly, to support translocations and Tier 1-triggered conservation actions included in the proposed action; however, the number collected would be similar to baseline conditions due to hatchery capacity constraints. Larval fish have an extremely low survival rate, and the risk to the humpback chub population of collecting several hundred larval fish is low (Pine. 2013). Nevertheless, collections for translocations or refuge development would be guided by FWS guidance documents (Translocation Framework, Van Haverbeke et al. 2016) and population viability modelling (Pine 2013). While inter-annual variability in incidental fatality due to handling occurs, the research program is not expected to change substantially through the LTEMP, and thus no change in impacts relative to the baseline condition is expected.

2.2.2 Razorback Sucker

Research designed to determine the status of razorback sucker in the lower Grand Canyon has resulted in mortality of larval razorback sucker and other native fish. While the mortality of up to several hundred individual razorback sucker larvae would be expected under the proposed action through collection and sacrifice of larval razorback suckers, with continued monitoring, the population-level impact would be minimal. For example, a single female razorback sucker could produce more than 100,000 larvae. While a final science plan for monitoring razorback sucker has not been developed, given the interest in determining and monitoring the impacts of TMFs and other actions included in the preferred alternative, mortality of larval fish and handling of juvenile fish (if present) may be expected to increase. However, these impacts would not likely impact the population, and monitoring would assist managers in determining the extent of impacts of flows on the species, potentially leading to adjustments in operations to benefit the species, or minimize large-scale impacts on habitat.

3 EFFECTS ANALYSIS OF PROPOSED ACTION

3.1 EFFECTS OF THE PROPOSED ACTION ON KANAB AMBERSNAIL

3.1.1 Direct and Indirect Effects

Aspects of the proposed action that have the potential to affect Kanab ambersnail individuals and habitat include sediment-related experiments (HFEs) and nonnative vegetation management. Other experiments or changes in dam operations included in the proposed action such as changes in base operations, trout management flows, nonnative fish control, low summer flows, and sustained low flows for invertebrate production—would have no impact on the Kanab ambersnail, because these activities would not occur in locations known to contain populations or habitat. The impacts of changes in base operations, low summer flows, and sustained low flows for invertebrates would be limited to areas within the river channel, where no habitat or individuals occur. Nonnative fish control would occur in the river, in the vicinity of the Little Colorado River inflow, which does not contain habitat for Kanab ambersnail. No fish control would occur in off-channel springs.

3.1.1.1 Effects of Sediment-Related Experiments

Within the Grand Canyon, populations of the Kanab ambersnail occur at Vasey's Paradise and Elves Chasm. Because the Elves Chasm population is located above the 100,000 cfs stage (FWS 2008), this population would not be affected by any of the proposed actions. At Vasey's Paradise, the proposed action will have no effect on the water flow from the side canyon spring that maintains wetland and aquatic habitat at Vasey's paradise. Some Kanab ambersnail habitat could be adversely affected by scouring at Colorado River flows exceeding 17,000 cfs (Kennedy and Ralston 2011). The HFE may reach flows of up to 45,000 cfs. These flows will inundate Kanab ambersnail habitat up to that stage and likely scour the vegetation and carry some snails downstream.

Very little Kanab ambersnail habitat and only a few individuals occur below the 25,000 cfs stage (Meretsky and Wegner 2000; Sorensen 2009). Most Kanab ambersnail habitat is located above the 33,000 cfs stage (Reclamation 2011b). HFEs may scour or inundate portions of Kanab ambersnail habitat. Surveys conducted after HFEs revealed no substantial declines in the Kanab ambersnail population (Kennedy and Ralston 2011). Kanab ambersnails can survive up to 32 hours underwater in cold, well-oxygenated water (FWS 2011a), so as long as they are not washed away, they could survive inundation from short-term HFEs.

Recovery of Kanab ambersnail habitat scoured by HFEs can take up to 2.5 years (Sorensen 2009). Therefore, frequent HFEs may result in long-term loss of Kanab ambersnail habitat (FWS 2011a). Under the proposed action, HFEs would occur more frequently than they would under baseline conditions. Based on modeling results, only five HFEs would occur under the baseline condition, due to the expiration of the HFE protocol (Reclamation 2011b), but an

HFE would be expected to occur almost every year under the proposed action (19.3 HFEs over the LTEMP period). Loss of habitat due to scouring could extend several years beyond the LTEMP period. In addition, impacts on individuals, including fatality, could occur during HFEs, if the individuals are dislodged from habitat and swept into the river. Depending on the elevation of maximum dam discharge during trout management flows, temporary scouring of habitat may occur, but it is unlikely to reach Kanab ambersnail habitat that is above the 30,000 cfs elevation. However, the snail population at Vasey's Paradise survived and persisted through natural predam floods and the 1983 flood, which were much larger in magnitude and longer in duration than the HFEs proposed under the LTEMP (Reclamation 1995; Kennedy and Ralston 2011).The amount of habitat and number of snails that would be unaffected by the proposed action are sufficient to maintain the population.

3.1.1.2 Effects of Vegetation Management

It is possible that vegetation management may occur, on rare occasions, near or within habitat at Vasey's Paradise, which may disturb individuals in small areas. Some nonnative vegetation may encroach on the spring, in which case careful manual removal of the vegetation would occur. To mitigate potential impacts on individual Kanab ambersnails, vegetation biologists would complete surveys and relocate any individuals detected on nonnative vegetation within the treatment areas. It is possible that some individuals could be killed during the treatment, but pre-treatment surveys would minimize this risk.

3.1.1.3 Effects of Climate Change

Climate change has the potential to affect the Kanab ambersnail habitat. The water source at Vasey's Paradise consists of waterfalls emanating from groundwater and emerging from the cliff face. In 2014 and 2015 the flow was noticeably reduced, likely as a result of basin-wide drought. Consequently, the usually dense vegetation at Vasey's Paradise is notably diminished. Climate change and drought will likely diminish the water source supporting Vasey's Paradise and consequently the vegetation and habitat available to the Kanab ambersnail.

Consequently, Reclamation has concluded the proposed action may affect and is likely to adversely affect the Kanab ambersnail.

3.2 EFFECTS OF THE PROPOSED ACTION ON HUMPBACK CHUB

The evaluation of potential impacts of the proposed action on humpback chub included consideration of the results of previous investigations conducted below Glen Canyon Dam that examined the status and abundance of native fish (e.g., Coggins and Walters 2009; Albrecht et al. 2014; Gerig et al. 2014; Kegerries et al. 2015), as well as studies of the effects of nonnative fish (Coggins et al. 2011; Yard et al. 2011; Whiting et al. 2014), experimental flows (such as HFEs and other flows), and water temperature on native fish (e.g., Grams et al. 2010;

Makinster et al. 2011; Trammell et al. 2002; Ward 2011; Ward and Morton-Starner 2015; Valdez and Speas 2007).

Several aspects of the proposed action may affect humpback chub directly or indirectly through direct impacts on individuals or habitat; or indirectly by influencing the abundance and distribution of nonnative predators and competitors, or by influencing macroinvertebrate production. Dam operations have the potential to influence river temperatures, the quantity and quality of nearshore rearing habitats (e.g., backwaters), aquatic insects that are food for humpback chub, and nonnative species abundance. These operations may also result in direct fatality to juvenile humpback chub through stranding. Flow-related actions that are analyzed below, which may impact humpback chub, include base operations, operational flexibility incorporated into base operations, sediment-related experiments, and aquatic resource experiments including trout management flows, low summer flows, and sustained low flows for benthic invertebrate production. Other non-flow actions that may impact humpback chub directly or indirectly, and that are analyzed below, include triggered humpback chub conservation actions (Appendix D), mechanical removal of nonnative fish, conservation measures, and continued fisheries research and monitoring under the GCDAMP. Some actions may have negative impacts on individuals, but beneficial effects at a population level.

Sediment monitoring and research and efforts to manage riparian vegetation are not expected to have impacts on humpback chub individuals or habitat. Riparian vegetation management would occur in only small, localized areas within the project area. Although riparian vegetation can play an important role in small stream ecosystems, particularly in providing cover, shade, or habitat for terrestrial invertebrates for fish to feed upon, riparian vegetation is less important for providing cover, shade, or energy inputs in larger rivers (Vannote and Sweeney 1980). Thus, riparian vegetation management would have large impacts on humpback chub and the species' habitat in the Colorado River.

3.2.1 Direct and Indirect Effects

3.2.1.1 Effects of Base Operations

Base dam operations are expected to change with the implementation of the proposed action. Changes will include more consistency across months, and thus more stability throughout the year, in monthly dam release volumes. The daily fluctuation in discharge would be proportional to the volume (i.e., lower fluctuations during lower flows), with a maximum discharge fluctuation range of 8,000 cubic feet/second, which would be the same as the current maximum fluctuation range. The daily downramp rate would be increased, compared to the current condition, by 67% (from 1,500 cfs/hr to 2,500 cfs/hr). Year-round, fluctuating flows throughout the life of the LTEMP period may continue to have detrimental, long-term direct impacts on humpback chub, humpback chub habitat, and aquatic invertebrate prey (food base).

Stranding of fish is a potential outcome of daily hydropeaking (Bunn and Arthington 2002; Nagrodski et al. 2012). Increasing downramp rates under the proposed action

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

may increase the risk of stranding humpback chub. Stranding could include fish being temporarily (i.e., until flows come back up hours later) restricted to isolated habitats away from the main channel, which may or may not become desiccated, as a result of dropping water levels. Desiccation of these isolated habitats would result in fatality of any juvenile humpback chub present. Factors that may influence the probability of stranding fish in a river with altered flow regimes—such as below hydroelectric plants—include the rate of flow reduction, water temperature, channel geomorphology, and substrate composition, as well as biotic factors including fish life stage and size (as reviewed in Nagrodski et al. 2012). Increasing drawdown rates under the proposed actions will increase the risk of negative impacts; however, humpback chub may be less reliant upon river margin habitats that are sensitive to flow fluctuations (e.g., backwaters; Dodrill et al. 2015) than other native fish, at least in the Little Colorado River inflow reach. The potential for, and the effect of, stranding on individual humpback chub survival has not been directly investigated; however, a summary of stranding literature indicated that fish stranding as a result of hydroelectric and irrigation projects is well documented (Nagrodski et al. 2012).

Daily fluctuating flows will continue to degrade nearshore habitats under the life of the LTEMP, particularly cobble shoals, riffles, and backwaters in wider, sections of the canyon, and where low-angle shorelines susceptible to flow fluctuations are prevalent. For example, cobble riffles and shoals, which are important spawning areas for fish, but also invertebrate production areas, would be dewatered daily. Generally, low catch rates of larval and small-bodied juvenile native fish have been observed in these types of habitats below Lava Falls (Albrecht et al. 2014; Kegerries et al. 2015). Although backwaters were not found to be common in the Little Colorado River inflow aggregation reach during a low steady flow experiment, humpback chub were found at higher densities in these habitats than in others (Dodrill et al. 2015). Backwaters may be more prevalent in other areas of the canyon, outside of the Little Colorado River inflow.

Base operations have been, and will continue to be, characterized as hydropeaking or load-following, which is an operational regime that has been demonstrated to have long-term detrimental impacts on aquatic invertebrates below hydroelectric dams in the western United States (Kennedy et al. 2016). Kennedy et al. (2016) found that hydropeaking was a primary factor implicated in reduced aquatic insect diversity; the Grand Canyon was found to have among the lowest insect diversity and the highest hydropeaking index among 16 dammed rivers. A diverse source of invertebrates colonize the Colorado River among the Grand Canyon's tributaries (Oberlin et al. 1999; Whiting et al. 2014), but hydropeaking and an altered temperature and flow regime will continue to negatively impact the food base for humpback chub over the life of the LTEMP.

Under base operations, through the life of the implementation of the LTEMP proposed action, river temperatures are expected to continue to be more suitable for coldwater nonnative species than for warmwater nonnative fish, particularly closer to the dam. In general, the estimated average main channel temperatures modeled for the LTEMP EIS indicated that temperature conditions would be most suitable for warmwater nonnative species at locations farther downstream from Glen Canyon Dam (e.g., RM 157 and RM 225) compared to upstream locations (e.g., RM 0 and RM 61), where temperatures would be more suitable to cold- or

coolwater nonnative fish (e.g., brown and rainbow trout); this is consistent with past surveys that have found more warmwater nonnative fish species in those areas than upstream.

Colder water temperatures restrict growth of YOY humpback chub, which prolongs the time during which individuals are vulnerable to trout predation (Yackulic et al. 2014). Essentially, juvenile humpback chub do not grow at temperatures under 12°C. Larger, older juvenile humpback chub may be able to avoid, withstand, or escape predation by rainbow trout under warmer and turbid conditions; however, temperature and size did not afford the same advantage in escaping brown trout predation (Ward and Morton-Starner 2015).

Unless the water can be warmed to above 12°C, it should be expected that the reduced water temperatures will result in increased egg incubation time (Hamman 1982), greatly reduced egg and fry survival (Hamman 1982), reduced larval-to-juvenile transition time (Clarkson and Childs 2000), and reduced growth rates of all life stages (Clarkson and Childs 2000; Coggins et al. 2011). From Glen Canyon Dam to the Little Colorado River inflow, water temperatures are generally <12°C. Consequently, there will be little or no spawning of humpback chub in this reach of the mainstem, unless it occurs at isolated locations such as 30-mile springs (RM 30) or springs at RM 34. There is little opportunity for larval fish to grow in Marble Canyon, unless the limited backwaters provide some thermal opportunities. In the Little Colorado River to Diamond reach, backwaters are more prevalent due to the more open geomorphology. Recently, temperatures have reached or exceeded 16°C at about RM 160 during September. This may provide opportunities for mainstem spawning activity and growth.

HFEs build sandbars, which can cause additional—but temporary—backwater habitats. Following the 2014 HFE, 22 sandbars and resultant backwaters throughout the Canyon became larger, 11 sites showed no change, and 5 sites became smaller. Although these sandbars erode within 6 to 12 months, they do not return to the pre-flood stage (Grams 2015). Three fall HFEs have occurred (2012, 2013 and 2014) in November; temperatures are unlikely to significantly increase in backwaters at this time, which is outside the spawning period for humpback chub. Spring HFEs may provide more thermal refuge in backwaters, particularly in the lower Canyon. However, relative densities of humpback chub, based on estimates of backwater prevalence directly after a controlled flood, showed the majority of juvenile humpback chub were found outside of backwaters. This suggests that the role of controlled floods in influencing native fish population trends may be limited (Dodrill et al. 2015).

It is possible that within the LTEMP period uncontrolled warm water releases may occur as a result of the elevation and subsequent water temperature of withdrawals from Lake Powell. Increased temperatures may benefit humpback chub spawning and growth, but would also allow numbers and species of warmwater nonnative species to expand in the system. A new conservation measure for LTEMP will explore new technologies for controlling the temperature of water discharged into the Colorado River and technologies to prevent the passage of fish through the dam.

Reclamation developed a risk assessment model to evaluate effects of a selective withdrawal structure on aquatic resources in the Grand Canyon area. It indicates there may be benefits for all native fishes from warmer water, as well as benefits for many nonnative fish species that may compete with or prey upon native species. Results also indicate that there may be more suitable conditions for warmwater fish parasites (Valdez and Speas 2007). Species that occur in the Colorado River are listed in Table 5, along with their spawning, incubation, growth, and lethal temperature tolerances.

The impacts of the proposed action on temperature suitability for coldwater nonnative trout were assessed to determine the potential for increased trout populations, and thus increased likelihood of competition or predation upon humpback chub. Using the temperature suitability model, generally, temperature regimes would be suitable—although not optimal—for brown and rainbow trout under the proposed action. Temperature suitability for brown and rainbow trout would be similar to the baseline at most locations downstream of Glen Canyon Dam. However, main channel temperatures at and downstream of RM 61 would be more suitable for trout than at locations closer to the dam. The abundance of trout is lower in those locations because other habitat characteristics (e.g., substrate composition and water clarity) are less suitable at these downstream locations.

Previous studies have shown that rainbow trout recruitment and population size within the Glen Canyon reach appear to be largely driven by dam operations (Avery et al. 2015; AZGFD 1996; McKinney and Persons 1999; McKinney et al. 2001a,b; Makinster et al. 2011; Wright and Kennedy 2011; Korman et al. 2011, 2012). Increases in abundance have been attributed to the changes in flows beginning with interim flows in 1991 and later the implementation of MLFF in 1996. These changes both increased minimum flows and reduced fluctuations in daily flows, which created more stable and productive nursery habitats for rainbow trout in Glen Canyon (McKinney and Persons 1999). Declines in abundance (such as those observed from 2001 to 2007) have been attributed to the combined influence of warmer water releases from Glen Canyon Dam, high abundance, increased intraspecific competition, and periodic dissolved oxygen deficiencies, along with possible limitations in the food base (Makinster et al. 2011). Episodic emigration from the Lees Ferry reach toward the Little Colorado River was likely related to trout density, trout condition, and turbidity during the rainbow trout natal origins study (Korman et al. 2015).

In summary, under base operations, continued hydropeaking flows with increased downramp rates compared to the existing condition under coldwater conditions will continue to degrade nearshore rearing habitats, prevent the establishment of aquatic invertebrates (food base), and increase the risk of stranding juvenile humpback chub over the long term (throughout the life of the LTEMP).

3.2.1.2 Effects of Dam Operational Flexibility

Operational flexibility in dam operations is necessary to account for unforeseen changes in basin-wide hydrologic conditions, dam maintenance needs, water distribution, resource-related issues, or hydropower-related issues. Flexibility in dam operations would not change from the current conditions. These minor operational changes are not expected to measurably impact humpback chub, because no known impacts have been identified related to the current regime.

	Spawning			Incubation				Growth			Lethal	
Species	Minimum	Maximum	Optimum	Minimum	Maximum	Optimum	Days	Minimum	Maximum	Optimum	High	Low
Black bullhead	19	22	20	21	24	23	6	20	22	21	3	35
Black crappie	14	18	16	15	22	17	4	24	30	27	0	34
Bluehead sucker	15	25	18	17	23	20	7	15	21	18	0	29
Bluegill	19	27	20	20	26	20	3	15	25	23	0	35
Brown trout	7	14	10	8	20	10	41	12	20	15	0	27
Bonytail	18	22	20	18	28	21	4	18	24	20	0	35
Common carp	18	30	23	20	30	24	4	15	30	27	0	38
Channel catfish	21	29	27	20	30	27	7	21	30	28	3	38
Colorado pikeminnow	16	24	22	19	25	22	5	18	23	25	0	35
Flathead catfish	22	29	26	22	29	27	8	24	30	26	5	40
Fathead minnow	16	30	25	16	29	25	5	18	27	25	2	33
Flannelmouth sucker	14	25	19	14	23	18	6	16	22	20	0	35
Grass carp	18	24	19	18	24	19	1	18	22	20	0	35
Green sunfish	19	31	22	19	24	23	5	26	31	30	0	35
Gizzard shad	15	27	20	17	27	24	4	16	26	21	5	35
Humpback chub	16	22	18	16	27	19	3	16	22	18	0	35
Little Colorado spinedace	16	20	18	16	20	18	6	16	30	27	3	35
Largemouth bass	16	20	18	16	20	18	6	16	30	27	3	35
Mosquitofish	18	30	22	20	24	22	25	14	28	22	4	43
Plains killifish	20	30	28	20	30	28	21	20	30	25	5	40
Razorback sucker	14	22	18	14	25	19	7	18	24	20	0	30
Rainbow trout	8	13	10	7	15	10	31	12	21	16	0	25
Redside shiner	10	18	15	12	20	18	8	15	22	18	0	24
Red shiner	15	30	23	15	25	24	4	18	28	24	0	40
Roundtail chub	14	22	16	16	22	18	5	18	24	18	0	35
Smallmouth bass	13	18	17	14	18	15	9	20	26	23	0	35
Speckled dace	18	25	20	18	23	20	6	18	22	20	0	35
Sand shiner	16	30	24	18	26	24	4	18	28	24	0	38
Striped bass	14	24	18	16	26	18	3	23	30	24	5	33
Threadfin shad	20	25	21	20	27	20	5	22	35	25	5	37
Blue tilapia	21	28	23	21	28	26	5	25	32	27	8	46
Utah chub	12	20	16	14	20	18	6	16	30	23	0	35
Walleye	6	13	7	6	14	13	7	18	23	21	0	33
Yellow bullhead	19	24	20	21	24	23	7	21	23	22	3	35
Zuni bluehead sucker	10	15	13	10	15	13	6	15	21	20	0	29

 TABLE 5 Temperature Requirements for Spawning, Incubation, Growth, and Lethality of 36 Native and Nonnative Fish Species of the Colorado River and Its Tributaries in and near the Project Area^a

^a All temperatures are in °C. Days are provided as average time for incubation of eggs (Valdez and Speas 2007).

3.2.1.3 Overall Implementation Process for Experiments

As discussed in the preferred action section, an adaptive management approach is being taken to implement all experimental elements of the LTEMP. Prior to the implementation of any experiment, potential impacts on humpback chub (and other resources) would be considered by the DOI, and deliberations would include the FWS and other DOI subject experts. This would help to identify and minimize the risk of impacts on humpback chub as a result of the implementation of an experimental treatment. Each experiment's specific effects on humpback chub habitat are discussed for each set of experiments below.

3.2.1.4 Effects of Sediment-Related Experiments

Experimental flows under the proposed action include: (1) sediment-triggered spring and fall HFEs through 3.2.1.3 the entire 20-year LTEMP period; (2) 24-hr proactive spring HFEs in high volume years (\geq 10 maf release volume); and (3) extension of the duration of up to 45,000 cfs fall HFEs for as many as 250 hr depending on sediment availability. These experimental flows have the potential to affect humpback chub indirectly by increasing sandbar area and shoreline habitats; however, the importance of these habitats to humpback chub rearing and recruitment is uncertain. The aquatic invertebrate community may also be impacted by sediment experiments (Cross et al. 2011). Recruitment of rainbow trout in Glen Canyon may be affected by spring HFEs, which may potentially indirectly affect humpback chub by increasing competition with, and predation upon, humpback chub.

HFEs under the proposed action were designed to increase and retain fine sediment for ecological purposes in Grand Canyon. Sediment is a fundamental component of the riverine ecosystem, and the sediment regime has been drastically altered by Glen Canyon Dam. Sand deposits were redistributed during pre-dam annual floods, creating substrate for riparian vegetation, and sandbars and backwater habitats for native fish (reviewed in Grams et al. 2010). Slow-water habitats that may be important to native fish, including humpback chub that are dependent upon sand bar development, include backwaters and embayments. For example, relatively high densities of humpback chub were found in backwaters in the Little Colorado River inflow reach (Dodrill et al. 2015), and when 10 habitats were sampled in 2015, more larval humpback chub were caught in embayments than in most other habitat types (Kegerries et al. 2015). However, in 2014, catch rates for larval humpback chub did not differ significantly among habitats sampled in western Grand Canyon (Albrecht et al. 2014). It is uncertain how the availability of these habitats influences juvenile humpback chub survival and recruitment to adult life stages.

HFEs have been shown to build sandbars that substantially increased the total area and volume of backwater habitat (Grams et al. 2010). The increased elevation of backwaters also increased the degree of isolation from the main channel, which would likely result in a higher degree of warming in backwater habitats (Trammell et al. 2002; Vernieu and Anderson 2013), which may be important to native fish in the summer (Albrecht et al. 2014; Kegerries et al. 2015). However, studies of habitat used by juvenile humpback chub have been mainly limited to the Little Colorado River inflow reach, and it is unclear how important

backwater or embayment habitats are to humpback chub throughout other reaches of the Grand Canyon. Beaches and backwaters are not common in the Little Colorado River reach and, although young chub occurred in higher densities in backwaters, their overall abundance in the Little Colorado River reach was higher in talus (Dodrill et al. 2015). Although talus slopes may be important to humpback chub rearing, and are not maintained by sediment deposition expected to occur as a result of HFEs, predation risk to juvenile humpback chub may be higher in talus compared to backwaters under low- and intermediate-turbidity conditions (Dodrill et al. 2016).

Modeling of sediment resources and the impacts of the proposed action focused on areas upstream of RM 87, but literature suggested that the impacts in this area would be indicative of impacts throughout the project area. Sandbars and backwaters cannot be built or maintained without high flows, and sediment-triggered HFEs result in the largest impacts under the proposed action. Substantial improvements in the ability to build and maintain sandbars over the baseline condition are expected for the LTEMP preferred alternative, because no HFEs would occur under the baseline condition after 2020 (existing HFE experimental protocol expires; Reclamation 2011a). The maintenance of backwater, under the proposed action may therefore benefit juvenile humpback chub rearing habitat. However, depending on the frequency and intensity of impacts of TMFs, as described below, and because daily downramp rates will increase under the preferred alternative, the potential for backwater habitats to be dewatered—stranding juvenile humpback chub—remains. In addition, the Near Shore Ecology project concluded that backwaters are likely not important to the Little Colorado River chub aggregation because they are not a significant habitat component in that area (Pine 2011).

HFEs have the potential to affect benthic aquatic invertebrates, which comprises the primary food base for humpback chub, but spring and fall HFE effects may differ. Much of what is known related to the effect of HFEs on the aquatic food base was determined through research in Glen Canyon, but some food base studies have been initiated recently in the Little Colorado River inflow reach. Cross et al. (2011) found that a spring HFE resulted in an overall reduction in invertebrate biomass and production, mainly reflected by declines in two nonnative invertebrates (New Zealand mudsnail, *Gammaris* sp.), but an increase in rainbow trout was also observed; this corresponded with an increase in blackflies and midges, which are prey items for fish. The effects of fall HFEs are less certain, but the aquatic invertebrate community appeared to have responded differently to fall HFEs that were implemented in 2012–2014, compared to effects described in Cross et al. (2011).

Under the proposed action, we could expect there to be 19.3 to 21 HFEs (maximum of 38 HFEs) during the 20-year LTEMP period. The more frequent HFEs are expected to favor blackfly and midge production; however, spring and fall HFEs may differ in their effects, so it is unclear what long-term impacts of annual (i.e., 19.3–21 out of 20 years) HFEs may have on the food base. Under the preferred alternative, up to four of the fall HFEs could be long-duration HFEs (lasting up to 250 hr). These extended-duration HFEs would be of higher magnitude and could increase benthic scouring, compared to short-duration HFEs. Drift from an extended-duration fall HFE may be elevated due to increased biomass of benthic invertebrates that may develop over the summer months The 4 to 5 months between a fall and spring HFE following a fall HFE, particularly a long-duration fall HFE, could scour the remaining primary producers and

susceptible invertebrates and further delay the recovery of the aquatic food base. For this reason, implementation of a spring HFE in years that follow an extended-duration fall HFE would be carefully considered.

Due to increased turbidity and other factors at the Little Colorado River inflow that differ from those at Glen Canyon, food base dynamics may not be comparable to those observed in Glen Canyon, and it is unclear what impacts HFEs have had upon the food base at the Little Colorado River. In terms of food base impacts upon humpback chub, recent preliminary USGS unpublished data indicate that as invertebrate drift declines (2014–2015), adult humpback chub body condition may also decline. Cross et al. (2013) found that native fish may be limited by food production, because all of the produced midges and blackflies are consumed. However, it is unclear how declining condition may be reflected in adult humpback chub survival, reproduction, and future recruitment of juveniles. It is possible that skipped spawning may be more prevalent in adults when food is limited, for example.

No current published research is available to show what factors may have correlated to the decline in invertebrate drift at the Little Colorado River in 2014 and 2015, compared to 2012–2013, but flow, temperature, and turbidity are likely factors. However, annual variation in drift since 2012 (potentially related to these factors) is likely much less detrimental to the food base for humpback chub than long-term loss of taxa that has likely resulted from hydropeaking or load-following (Kennedy et al. 2016; see base operations above). Despite the apparent impact on adult humpback chub body condition that may be correlated to food availability, humpback chub population dynamics will likely not respond to variations in food availability in a similar manner to the response shown for rainbow trout (see Cross et al. 2011). Humpback chub evolved in a system with limited autochthonus production due to high turbidity in the pre-dam Colorado River, and thus may have relied more heavily on terrestrial invertebrates or other allochthonous food items washed into the river during floods. In addition, other aspects of the species' life history, including high adult longevity, skip spawning, and higher relative fecundity compared to species in more stable environments, may moderate the effects of highly variable environment, including variation in food availability, on long-term population dynamics of the species. Many fish species evolved in variable environments, with a "boom and bust" life history strategy (see periodic life history strategy: Winemiller and Rose 1992). However, the more stable postdam environment may also favor nonnative species (Olden et al. 2006; see discussion below). Nevertheless, spring and fall HFEs may not be tested in years when there would appear to be unacceptable risks to key resources including the aquatic food base, and the results of experimental flows will be discussed with FWS prior to implementation to minimize indirect effects on humpback chub.

Proactive spring HFEs would occur in high volume years with planned equalization releases (≥ 10 maf) in order to protect the sand supply from equalization releases. Fall HFEs longer than 96 hr in duration will be implemented when there is sufficient Paria River sediment input in the fall accounting period (July–October) to achieve a positive sand mass balance in Marble Canyon. Spring HFEs and high and steady flows (Avery et al. 2015), have both demonstrated an increase in rainbow trout production in the Colorado River, especially in the 16-mi reach below the dam (Lees Ferry reach). This increase in survival of juvenile rainbow trout was attributable to an improvement in habitat conditions and food availability in

Glen Canyon for recently emerged trout (Korman et al. 2011). Flows and temperature were modeled over the LTEMP period to assess how rainbow trout recruitment and the potential for emigration downstream toward the Little Colorado River may be impacted by flow experiments. Under the preferred alternative, emigration of trout is expected to increase by 11%, which could increase the likelihood of negative impacts on humpback chub through competition and predation, particularly in cold water release years. To provide a means of controlling trout recruitment following tests of spring HFEs, TMFs would be experimentally implemented and tested for efficacy as early in the LTEMP period as possible. In addition, if a tiered trigger approach to reversing declines in possible humpback chub recruitment fails, mechanical trout control are considered experimental actions, they are expected to offset risks of trout predation upon juvenile chub at the Little Colorado River. In addition, basin-wide drought and the low Lake Powell levels expected to occur occasionally over the next 20 years would result in warmer dam releases, which also mitigate effects of rainbow trout predation upon juvenile humpback chub (Ward and Morton-Starner 2015).

The impacts of spring and fall HFEs on humpback chub are incorporated into the modeling described in the LTEMP EIS. In general, HFE impacts incorporated into the model are focused on influences of the flow and temperature regime changes upon rainbow trout population dynamics and indirect effects on humpback chub as a result of competition and predation. The output of the model for the preferred alternative represents a range of potential outcomes that may occur over the next 20 years for humpback chub, including a "worst-case scenario" that may be defined as colder/coldest dam discharge and high rainbow trout abundance. However, although humpback chub declines may occur during years with cold temperatures and high trout abundance, TMFs and mechanical control of trout are expected to mitigate these negative impacts; therefore, it is unlikely that actual declines in humpback chub would cause the population to reach the low levels displayed by the model (around 1500 individuals) over the life of the LTEMP.

Nonnative brown trout, or warmwater species occurrence or abundance, may also be influenced by the HFEs. Brown trout spawning occurs mostly in tributaries, primarily in Bright Angel Creek (Reclamation 2011a,b); consequently, reproduction would not be significantly affected by the flow operations of the dam. However, recent increases in brown trout recruitment in 2014–2015 have occurred in the Lees Ferry reach of the Colorado River in Glen Canyon (Stewart 2016). Brown trout were observed to be spawning near the 4-mile bar in Glen Canyon during the fall of 2014, and an increase in age-1 brown trout, likely as a result of spawning and recruitment in 2014, was observed in 2015 (Korman et al. 2015). Spawning of brown trout was also observed during October and November of 2015 near the 4-mile bar in Glen Canyon (Korman et al. 2015). It is unclear if flow operations, including recent fall HFEs, caused an increase in brown trout in recent years. The YOY of fall-spawning brown trout would benefit from increased food availability, similar to rainbow trout (Cross et al. 2011), if the fall HFEs' effects on benthic invertebrates are similar to those of the spring HFE. As discussed above, an increase in blackfly and midge production following fall HFEs has not been observed; therefore, it does not appear that recent fall HFEs have resulted in conditions that facilitated increased brown trout reproduction in Glen Canyon.

Although increases in brown trout recruitment do not appear to be related to HFEs, it is possible that the establishment of green sunfish in 2015 in a warmwater slough downstream of Glen Canyon Dam may be related to dam operations. Based on this assumption, and combined with expected occasional warmwater flow release periods during drought years, an increase in the likelihood of warmwater nonnative fish occurrence is expected below Glen Canyon Dam. Rapid increases in dam discharge from baseflow to up to 45,000 cfs during HFE implementation may increase the chance that abundant warmwater nonnative fish near the intakes become entrained in the bypass tubes. The risk would increase during periods of low Lake Powell levels because when this occurs intakes are closer to the surface and lake littoral zones, where smallmouth bass or green sunfish are present. During basin-wide drought years, warmer discharge would increase the risk further, because warm water could facilitate reproduction of these species in the Grand Canyon, potentially near the Little Colorado River inflow. To offset this risk, the conservation measures discussed below were developed. These include the investigation of means to prevent warmwater species' passage through the dam, completion of planning and compliance to make a warmwater slough less hospitable to warmwater nonnative fish, and the development of a rapid response protocol that includes chemical control methods and others not included in the NPS CFMP.

In summary, sediment experiments have the potential to negatively affect humpback chub individuals and habitat by increasing rainbow trout emigration to the Little Colorado River reach. This may increase predation or competition effects, alter the food base, or increase the risk of establishment of warmwater nonnative fish. However, these negative impacts may be offset by conservation actions and triggered mechanical removal of nonnative fish (trout or warmwater nonnative fish) to benefit juvenile humpback chub rearing, the development of a rapid response protocol, and warmwater species' habitat modification. Further, backwaters and embayments created following HFEs may provide warm rearing habitat for juvenile humpback chub, but the benefit may be minimal for the Little Colorado River inflow aggregation. Finally, these sediment treatments will not continue to be tested if they are not effective in achieving their purpose or have unacceptable adverse impacts on the trout fishery, humpback chub population, or other resources.

3.2.1.5 Effects of Trout Management Flow Experiments

TMFs will be implemented when there is predicted high trout recruitment in the Glen Canyon reach, but tests may be implemented when trout are at low densities, early in the LTEMP period. TMFs are designed to cause fatality in YOY rainbow trout by inundating low-angle, nearshore habitats for several days, and then quickly reducing dam discharge, which would strand YOY fish; however, there is also potential for stranding and increased fatality of juvenile humpback chub. Although the Lees Ferry reach trout population is the target of TMFs, an examination of USGS hydrograph data from Lees Ferry (RM 0) and National Canyon (approximately RM 166) indicated there may be little attenuation of stage changes due to flows downstream in the Marble and Grand Canyons. YOY humpback chub are found primarily in the Little Colorado River inflow reach, as well as further downstream, in unknown numbers. Stranding may occur, including stranding of endangered humpback chub and razorback sucker. TMFs are proposed to occur from May through August, in years when the experimental treatment would be tested or implemented, including up to three cycles per month of high steady flows to draw juvenile fish into shallow nearshore waters, followed by rapid drops in discharge to strand fish.

Modeling conducted for the LTEMP EIS evaluated the number of trout recruits in the Glen Canyon reach, and the numbers of trout and humpback chub in the Little Colorado River reach to determine when TMFs would be triggered under the proposed action. It was estimated that triggers for TMFs would be met approximately every 3 years (32% of years).

Although indirect benefits of TMFs for humpback chub at the Little Colorado River are expected under the proposed action as a result of reduced competition and predation by rainbow trout, an unknown number of YOY humpback chub may also suffer fatality as a result of being stranded during TMFs, including those at the Little Colorado River inflow and downstream in the Grand Canyon. Stranding impacts may be difficult to observe, particularly lesser studied, sublethal impacts upon growth (Nagrodski et al. 2012). However, due to the geomorphological features of the Little Colorado inflow reach, fewer shoreline habitats such as backwaters, which are susceptible to fluctuations, are present, and may support only a small portion of juvenile humpback chub in that reach (Dodrill et al. 2015). It is less likely that TMFs would affect adult or juvenile chub, because they have a greater ability to swim out of confined spaces as flows dropped. Small YOY or larval fish, which already have a high mortality rate, could suffer increased mortality as a result of TMFs. TMFs could occur throughout the summer, overlapping with periods when larval humpback chub are found in nearshore habitats (May–August; Albrecht et al. 2014; Kegerries et al. 2015) and would be susceptible to extreme flow fluctuations under TMFs.

The extent of humpback chub mortality due to stranding and the benefits of reduced trout recruitment in a given year as a result of repeated (every third year) TMFs in Marble and Grand Canyons over the life of the LTEMP are uncertain. These impacts may depend on the quantity of channel margin habitats and their sensitivity to flow changes, the distribution and abundance of humpback chub in sensitive habitats, the timing and number of TMFs, and the degree of attenuation of flows downstream. The beneficial effects of TMFs for humpback chub may depend on the effectiveness of flows in reducing recruitment of YOY trout and emigration to the Little Colorado reach, as well as on the strength of the interaction between humpback chub and rainbow trout, which is mediated by water temperature (see Ward and Morton-Starner 2015). Stranding has not been observed during previous flow experiments, but no studies have been completed, so the degree of stranding is uncertain. The impacts of TMFs throughout GCNP will be monitored to assess effectiveness of the action and the detrimental impacts on native fish and other resources. TMFs will not continue to be tested if there is little or no reduction in trout recruitment after at least three tests, or if there are unacceptable adverse impacts on the humpback chub population or other resources. Consultation with the FWS would be conducted prior to implementation of the experiment. Thus, with "offramps" in place, long-term negative impacts on humpback chub would not be expected.

3.2.1.6 Effects of Triggered Conservation Actions and Mechanical Removal of Nonnative Fish Experiments

Under the proposed action, mechanical removal of nonnative rainbow and brown trout (and other nonnative predators) would be implemented through a triggered, tiered approach (see Appendix D) near the confluence of the Little Colorado River and the Colorado River if conservation actions designed to reverse declines in the Little Colorado River humpback chub aggregation are ineffective. This approach was developed by biologists with the FWS, USGS-GCMRC, AZGFD, NPS, and Reclamation, using the best available science and professional judgement. Conservation actions designed to improve rearing and recruitment of juvenile humpback chub would be implemented when adult humpback chub abundance declines to a threshold, or if recruitment of subadult humpback chub does not meet or exceed estimated adult mortality. Conservation actions would include expanded translocations of YOY humpback chub within the Little Colorado River to grow-out areas (i.e., above Chute Falls, Big Canyon), or larval fish would be taken to a rearing facility and released in the mainstem Little Colorado River inflow area once they reach 150-200 mm. If these conservation actions fail to reverse the declining trends and adult abundance dropped below 7000, mechanical removal of nonnative fish would be implemented in the Little Colorado River inflow area to lessen the effects of competition and predation upon humpback chub by nonnative fish.

Based on a humpback chub population model, which is primarily driven by temperature and rainbow trout population dynamics, it is likely that humpback chub populations may fluctuate over the life of the implementation of LTEMP (20 years). If model assumptions are correct, adult humpback chub populations could decline as low as 1800 individuals, coinciding with cold temperatures and high rainbow trout abundance. For example, expected emigration of rainbow trout from Glen Canyon was estimated to be 11% higher than under the current dam operations. This would increase predation upon juvenile humpback chub, potentially leading to reduced recruitment.

Tier 1 conservation actions would involve collecting 300–750 larvae or YOY—age classes that are vulnerable to predation by nonnative fish and that would naturally have relatively high mortality rates (see Pine 2013)—and rearing them to less vulnerable sizes before releasing them. Alternatively, YOY would immediately be translocated for rearing to areas with few predators, such as Big Spring or above Chute Falls. In the past, fish were successfully translocated within the Little Colorado River and to tributaries; the translocated fish in these locations experienced high survival and/or growth rates (Healy et al. 2014a; Spurgeon et al. 2015; Van Haverbeke et al. 2016), and there is a high likelihood for humpback chub to benefit through augmentation of the adult population as a result of Tier 1 actions. Detrimental effects on humpback chub, including fatality, could occur during handling, transport, or tempering; however, these occurrences are generally low, see Section 2.2.

Mechanical removal of nonnative fish would be implemented at the Little Colorado River inflow if declines in adult humpback chub continued, which would help minimize mortality of juvenile humpback chub. Past studies have indicated that trout predation upon humpback chub could be significant (Yard et al. 2011), and humpback chub are particularly vulnerable to predation at cold water temperatures (Ward and Morton-Starner 2015). It is somewhat uncertain

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

whether mechanical removal in the vicinity of the Little Colorado River would be successful in reversing declines in humpback chub, because (a) at high immigration rates into the Little Colorado River reach, mechanical removal may not be successful in reducing trout abundance (Coggins et al. 2011), and (b) other factors beyond trout predation and competition may be negatively impacting humpback chub survival and abundance. Past removal efforts appeared to be effective in controlling rainbow trout, and humpback chub recruitment increased; however, the removal effort coincided with a system-wide decline in trout abundance and warming dam discharge, which confounded results (Coggins et al. 2011). Nevertheless, TMFs designed to reduce Glen Canyon rainbow trout recruitment, and thus emigration downstream, would likely result in low immigration rates of trout to the Little Colorado River reach, thereby increasing the effectiveness of mechanical removal and benefitting humpback chub recruitment. The adaptive management approach taken to adjust trout removal targets and thresholds for action would also help ensure the effectiveness of this strategy in stemming humpback chub declines.

As designed, the tiered, adaptive approach to responding to humpback chub declines described here is expected to offset potentially negative impacts of management actions and experiments under the LTEMP. If conservation actions are ultimately successful in improving rearing success, these tools could be implemented proactively, and could potentially lead to long-term beneficial impacts on humpback chub in the Little Colorado River inflow aggregation.

3.2.1.7 Effects of Low Summer Flow Experiments

Low summer flow experiments involve holding flows low (approximately 8,000 cfs) and relatively steady, compared to base operations during warm summer months. Flows may continue to fluctuate 1000 cfs in a day, which would be less fluctuation than under base operations. The goal of the experiment is to achieve warmer river temperatures to benefit humpback chub; however, more stable shoreline habitats may also occur, which could also improve juvenile humpback chub rearing habitat. Lower flows have the potential for more warming via heat transfer from the air compared to higher flows. Low summer flows will be tested when the water temperature has been $<12^{\circ}$ C for 2 consecutive years, and a target temperature of $\ge 14^{\circ}$ C can only be achieved if flows are dropped with the objective of testing the efficacy of low summer flows on warming and humpback chub growth. Low summer flows will only be tested after the first 10 years, and only under specific, rarely occurring, conditions. This test will not continue if there is no increase in growth and recruitment of humpback chub, if there is an increase in warmwater nonnative species or trout at the Little Colorado River; or if there are unacceptable adverse impacts on the trout fishery, humpback chub population, or other resources.

Under this design (1 to 2 years of testing), the effect of the low flow experiment upon humpback chub growth, rearing, and recruitment will be difficult to determine, because results are likely to be confounded by other factors (variation in annual precipitation, turbidity, etc.). In addition, the tests would be conducted only when humpback chub populations are healthy and stable; this would decrease the test's effectiveness in determining how low flows may improve humpback chub rearing during periods when the population is unhealthy or in decline. Nevertheless, in addition to benefits associated with warming waters described in previous sections, stable nearshore habitats may benefit rearing, depending on the use of habitats that are sensitive to flow fluctuations (backwaters, embayments, cobble bars, riffles, etc.). The role of backwaters or similar habitats in humpback rearing is discussed in more detail in Section 3.2.1.4. However, reduced fluctuations under the low summer flow experiment would reduce stranding potential for humpback chub, and benefit the food base.

Low summer flows may allow for a prolonged test of the effect of steady flows on the diversity and abundance of aquatic invertebrates, similar to the intent of the low steady weekend flows for macroinvertebrate production experiment (Section 1.2.12). Base operations have been, and will continue to be, characterized as hydropeaking or load-following, which is an operational regime that has been demonstrated to have long-term detrimental impacts on aquatic invertebrates below hydroelectric dams in the Western United States (Kennedy et al. 2016). A diverse source of invertebrates to colonize the Colorado River is found among Grand Canyon's tributaries (Oberlin et al.1999; Whiting et al. 2014), but hydropeaking and an altered temperature and flow regime will continue to negatively impact the food base for humpback chub over the life of the LTEMP. Consistently more stable flows over the summer could result in successful egg-laying and reproduction by sensitive macroinvertebrate taxa (e.g., mayflies), which would also be important prey for humpback chub. However, the experiment may only be implemented 1–3 times in the second year of LTEMP, which would limit the ability of this experiment to benefit humpback chub food base over the long term. Only short-term (within the year of the test) beneficial effects on the food base would be expected.

It is possible that warming as a result of the implementation of low summer flows may increase the risk of warmwater nonnative fish establishment, threatening humpback chub populations. However, these tests are unlikely to increase the risk significantly, because they would occur infrequently (1–3 times out of 20 years), and due to low Lake Powell levels, water temperatures are already within the spawning range of warmwater nonnative species during some part of the year in the lower river (see Kegerries et al. 2015). Basin-wide hydrology that influences the Lake Powell level and dam discharge temperatures would likely overwhelm the effects of dam operations on river temperatures.

Although benefits to humpback chub may be realized as a result of this experiment on habitats and food base, the effect may be of short duration and may be insignificant over the life of the LTEMP (1–3 years out of 20). However, if the experiment has beneficial effects, it may be implemented more frequently beyond the life of the LTEMP period, if additional planning and compliance is completed.

3.2.1.8 Effects of Low Steady Weekend Flows for Macroinvertebrate Production Experiments

Experimental steady flows on weekends, in an attempt to increase invertebrate production and diversity, may benefit nearshore habitats by providing stability and some warming in habitats that are more isolated from the mainstem (low flows would be held steadier plus/minus 1000 cfs fluctuation); however, this would be limited to approximately 34 days over the summer. This experiment will not be conducted during first 2 years of LTEMP. The goal is to replicate it two to three times to determine its effectiveness.

Kennedy et al. (2016) demonstrated that hydropeaking flows in rivers below hydroelectric facilities may prevent the maintenance of viable populations of many aquatic invertebrates (e.g., mayflies, stoneflies, caddisflies) that would constitute important components of the food base for humpback chub. Many shoreline egg-laying invertebrates were absent from western rivers with higher degrees of hydropeaking fluctuations. This experiment may allow some shoreline egg-laying invertebrates to avoid desiccation, which results in a high degree of egg mortality (Kennedy et al. 2016), and potentially will allow for increased production and possibly diversity of aquatic invertebrates.

In addition to potential benefits of increased aquatic food base production, steady flows on weekends may help to improve the quality of nearshore habitats for juvenile humpback chub. However, it is unclear whether 2 days per week of nearshore habitat stability will result in a measureable impact on juvenile growth, rearing, and recruitment.

Sustained low flows for benthic invertebrate production may benefit humpback chub by providing additional aquatic insect prey. It may also provide stable nearshore habitats for rearing, for 2 days per week. This treatment will be discontinued if there is no observed benefit to the food base, the trout fishery, or native fish; if there is an increase in warmwater nonnative species or trout at the Little Colorado River; or if there is an unacceptable adverse impact on the trout fishery, humpback chub population, or other resources.

3.2.1.9 Effects of Ongoing Conservation Measures

Conservation measures from the 2011 Biological Opinion were evaluated and adapted for the LTEMP, where necessary, and included in the preferred alternative. Some additional conservation measures were developed specifically for the LTEMP and were incorporated into the preferred alternative. The intention of conservation measures is to mitigate or offset potential impacts on ESA-listed species that may result due to the implementation of the LTEMP preferred alternative. These conservation measures were developed in consultation with the FWS and biologists from the NPS. Effects on humpback chub are discussed for each conservation measure below.

Humpback Chub Translocations

Humpback chub translocations were included as conservation measures in the 2008 and 2011 Biological Opinions on the operation of Glen Canyon Dam and on the Nonnative Fish Control and HFE plans, respectively. The preferred alternative includes the continued implementation of humpback chub translocations from the Little Colorado River to other areas in the Little Colorado River or to other tributaries of the Colorado River in the project area in order to (a) improve rearing and growth and provide additional rearing opportunities, or (b) establish additional spawning populations. As discussed in previous sections of this document,

translocations have been conducted above Chute Falls in the Little Colorado River, to Shinumo Creek, and to Havasu Creek. Translocations may result in increased growth or improved rearing and survival of juvenile humpback chub.

Translocations of juvenile humpback chub were first conducted in 2003. They involved moving fish from the lower reaches of the Little Colorado River to areas above Chute Falls (Van Haverbeke et al. 2013). These translocations have continued through recent years. Compared to juvenile humpback chub in the lower Little Colorado River, fish above Chute Falls grew much faster.

Prior to 2014, beginning in 2009, approximately 300 juvenile humpback chub per year were translocated to Shinumo Creek. Estimated growth and survival of translocated humpback chub generally was consistent with, or exceeded, growth and survival estimates of juvenile humpback chub in the Little Colorado River and Colorado River, although emigration rates exceeded 50% (Spurgeon et al. 2015; Healy et al. 2014b). In May of 2014, an NPS-managed fire started by lightning, burned approximately 10% of the Shinumo Creek watershed. Flooding and ash appeared to have extirpated the remaining humpback chub from Shinumo Creek in July and August 2014.

Havasu Creek translocations were initiated in June 2011 and have continued annually through May 2016. Approximately 300 fish were released in Havasu Creek annually. Biannual monitoring of translocated fish indicated that survival and growth objectives were met. In addition, evidence of reproduction, including ripe adult fish and untagged juveniles (age-1+), was observed annually beginning in 2013. Data collected from the May 2016 monitoring trip indicated that multiple year classes of untagged (non-translocated) humpback chub were observed (Nelson et al. 2016).

Through 2016, translocations to Havasu Creek have been limited to the lower 3.5 mi from the Havasupai Reservation/NPS boundary to the confluence with the Colorado River—on NPS lands. A new effort will be pursued during the LTEMP to conduct initial fish surveys and complete a feasibility study to translocate humpback chub to Havasupai Reservation lands above Beaver Falls. Although the activities are subject to permission and support from the Havasupai Tribe, if habitat is suitable, translocations to this area could lead to an expanded spawning population at Havasu Creek. Larger populations would be more likely to persist in tributaries (Pine 2013).

Genetics management principles outlined in FWS guidance documents (FWS Genetics Management Plan 2010; Van Haverbeke et al. 2016) and the NPS CFMP (NPS 2013) would be followed to maintain genetic diversity and minimize detrimental genetic effects on translocated populations. For example, genetic monitoring would be conducted in tributary spawning populations established through translocations, and new migrants would be introduced occasionally (NPS 2013). As a result, no loss of genetic diversity is expected.

As part of the ongoing humpback chub translocation effort, Reclamation has committed to funding nonnative fish control in tributaries, where necessary, to maximize survival of translocated humpback chub. Control methods would vary depending on the tributary and the existing nonnative fish species. Complete eradication of a nonnative species would likely provide the most beneficial impact on humpback chub translocated to the tributary; however, no specific plans for chemical piscicides have been identified, and piscicide use would require additional planning and compliance activities.

No detrimental population-scale impacts are expected as a result of translocations. In general, as demonstrated by previous or ongoing translocations, successful rearing and potential additional spawning populations can be developed or augmented through translocation of juvenile humpback chub, thereby benefitting the population. In addition to the continuation of the Chute Falls translocations under the preferred alternative, Havasu, Shinumo, and Bright Angel (pending successful trout control) creeks would continue to be the focus of translocations efforts outside the Little Colorado River. Although translocations can result in mortality of individual juveniles during collection from the Little Colorado River, transport to rearing facilities, disease treatment, or tempering and release (thus far, Chute Falls only), no detrimental population-level impacts on the adult population in the Little Colorado River has been observed. In addition, based on population viability modeling (Pine 2013), collection methods have been further refined to mitigate this risk by targeting life stages (e.g., larval) that would have a high mortality rate in the wild.

Humpback Chub Refuge

A refuge population of humpback chub was established at the FWS SNARRC facility in Dexter, New Mexico. The goal of the refuge is to maintain a genetically diverse population of 1000 humpback chub to be used for propagation in the case of a catastrophic loss of the Grand Canyon population. Impacts would be similar to those described for translocations, in that a small number of larval chub may be lost during collection, transport, and disease treatment; however, no detrimental impacts on the population would be expected.

Mainstem Humpback Chub Aggregations

Conserving mainstem humpback chub aggregations was a focus of past conservation measures related to dam operations. In the LTMEP, Reclamation will continue to assist the FWS, NPS, and GCDAMP to ensure that a stable or upward trend of humpback chub mainstem aggregations can be achieved. Actions include continued monitoring of the Little Colorado River humpback chub aggregation using the juvenile chub monitoring protocol developed following the natal origins project (see Pine et al. 2011); monitoring the mainstem aggregations annually; developing a methodology for locating additional aggregations and individual humpback chub; evaluating "drivers" of recruitment, potential spawning, and natal origins; and exploring a means of expanding humpback chub populations outside the Little Colorado River inflow aggregation.

Mainstem monitoring is critical to understanding how base operations and other experiments under the LTEMP impact humpback chub. Understanding more about the ecology and population dynamics of mainstem aggregations outside the Little Colorado River, including the impacts of dam operations, may lead to improved management and conservation of those aggregations.

The impacts on individual humpback chub that may occur as a result of monitoring are covered below in Section 2.2. Monitoring under this action would be designed to assess trends in humpback chub in the mainstem Colorado River, and to assess impacts of experimental and management actions on humpback chub. Results of monitoring may allow managers to adapt actions to ensure that projects goals for humpback chub are met.

Disease and Parasite Monitoring

Reclamation intends to fund the GCDAMP to conduct disease and parasite monitoring in humpback chub and other fish in the mainstem Colorado River, in addition to ongoing disease and parasite monitoring conducted by USGS/GCMRC in the Little Colorado River.

It is unclear how this monitoring will be accomplished or how it may benefit humpback chub over the long term; however, understanding the impacts of parasites may inform future conservation actions.

Razorback Sucker Conservation Measures

Studies designed to help understand the status of the razorback sucker population in the Grand Canyon and Lake Mead would continue with the implementation of the preferred alternative. Although the focus of these studies is on razorback sucker, seining for larval and small-bodied YOY and juvenile fish may lead to an increased understanding of juvenile and larval humpback chub habitat use and spawning areas (at a course scale) in the Grand Canyon.

Fatality of individual larval humpback chub would occur as a result of this sampling regime; however, high larval mortality rates are likely in the wild, and thus, this effort would not be expected to impact the adult population. In the first 2 years of the larval and small-bodied fish study conducted from Lava Falls (RM 180) to Pearce Ferry above Lake Mead (RM 280), indicated that reproduction may be occurring (larval humpback chub with yolk sac attached).

No population-scale impacts on humpback chub would be expected, and the study may provide important information for conservation of the species in the western Grand Canyon.

Nonnative Species Conservation Measures

Nonnative fish will continue to threaten humpback chub in the Grand Canyon under the LTEMP preferred alternative. Under the continued colder temperature regimes and clear water that result from dam construction and operations, particularly in Glen and Marble Canyons, nonnative cool-/coldwater species are expected to proliferate and continue to prey upon and compete with humpback chub. Brown trout are a particular threat due to their piscivorous nature

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

(Yard et al. 2011; Whiting et al. 2014). However, with continued basin-wide drought, and lower Lake Powell levels, warmer water releases from Glen Canyon dam are expected occasionally. For example, in late fall 2015 (November) water released from Glen Canyon dam approached 15°C, which is within the range of suitable temperatures for smallmouth bass spawning. In addition, temperatures in western/lower Grand Canyon reached or exceeded 20°C in 2015. Warmer temperatures may allow for highly piscivorous warmwater nonnative species such as smallmouth bass to proliferate in the project area. Smallmouth bass were identified as a major threat to native fish in the Yampa River, and have caused major declines and extirpation of humpback chub there (Johnson et al. 2008).

Conservation measures that were developed to specifically address nonnative species threats to humpback chub include (a) continued nonnative trout (brown and rainbow) removal efforts at Bright Angel Creek and its inflow, or other areas where expanded populations have been found; (b) feasibility study, and planning, and compliance for chemical piscicide renovation at Bright Angel and Shinumo creeks; (c) study of efficacy of a temperature control device on the dam to address future conditions under climate change; (d) potential research and development of a means to prevent passage of warmwater nonnative fish through Glen Canyon Dam; (e) planning, compliance, and implementation of a plan to address (make less suitable) the suitability of warmwater nonnative fish habitat in a slough below Glen Canyon dam; and (f) completion of a plan for rapid response to nonnative fish invasions, using tools not approved in the NPS CFMP (e.g., chemicals). Finally, the use of flows (e.g., TMFs) to inhibit brown trout spawning and recruitment in Glen Canyon, or other mainstem locations, will be explored over the life of the LTEMP. This measure is necessary to prevent recently detected spawning of brown trout in Glen Canyon from further expanding or increasing in abundance.

Control of brown and rainbow trout in Bright Angel Creek and the Bright Angel Creek inflow reach of the Colorado River would be continued under the proposed action. In 2017, results of control efforts implemented between 2012 and 2017, including trends in abundance of both native and nonnative fish, would be reviewed, and methodology for continued control efforts may be adapted to ensure trout control objectives are met. Brown trout catch-per-unit effort in the Colorado River outside of Bright Angel Creek appears to have declined in recent years (Stewart 2016), concurrent with trout control activities, and declines of brown trout abundance and biomass in Bright Angel Creek (Nelson 2016). Brown trout are defined as high-risk nonnative predators (NPS 2013), and continued control in and around Bright Angel Creek—and in other areas where they expand—would minimize a major threat to humpback chub by reducing predation risk. Long-term beneficial impacts for humpback chub would be expected as a result of continued control of brown trout and reduced predation risk.

The other nonnative fish conservation measures could all lead to beneficial impacts on humpback chub as a result of reduced the risk of invasion and establishment of warmwater nonnative species. The outcome will depend on the effectiveness of the actions, including whether feasible options are developed to manage slough habitat and prevent the passage of warmwater nonnative fish through the dam; no technology currently exists for this.

3.2.1.10 Effects of New Conservation Measures

New conservation measures for humpback chub will explore additional tributaries for potential translocations including consulting with the Havasupai Tribe for potential translocations above Beaver Falls in Havasu Creek. Razorback sucker measures will include continued monitoring of larval and small-bodied fish; these measures will determine the extent of hybridization of flannelmouth and razorback sucker larvae from the western Grand Canyon through genetic analysis. Additional measures to address nonnative fish include evaluation of new technologies for temperature-controlled releases from the dam and methods to prevent fish passage through the dam. Reclamation will support planning and compliance for the development and implementation of a rapid response control effort for new and established nonnative invasive species. This includes modifying the backwater slough at RM 12 to prevent establishment of warmwater invasive species. Reclamation will also explore the use of flow such as TMFs to inhibit brown trout spawning and recruitment in mainstem locations.

3.2.1.11 Effects of Native and Nonnative Plant Management Experimental Actions

Under the proposed action, exotic plant species that are spread or favored by dam operations would be mechanically removed, and potentially treated with herbicides, in selected areas within the project area.

Herbicides may be poisonous to fish and aquatic species; however, standard practices for herbicide application would be followed. This would minimize any risk of impacts on humpback chub or their habitat as a result of contact with the chemicals.

3.2.1.12 Effects of Activities to Preserve and Protect Historic Properties and Cultural Sites

Under the preferred alternative, various mitigation and erosion control measures would be implanted at cultural sites along the river corridor. Sites that would be treated would generally be located above the Colorado River high water mark and outside of humpback chub habitat. Thus, these activities would be expected to have no impacts on humpback chub individuals or habitat.

3.2.1.13 Effects of Climate Change

Climate change and drought have direct influences on hydrologic patterns and water temperature, thus impacting humpback chub. In the Colorado River Basin, hydrologic impacts include lower precipitation and decreased inflow to the reservoir system, resulting in more frequent lower reservoir release volumes, which could potentially impact shoreline habitats and riparian areas. More frequent droughts and warmer atmospheric temperatures have the potential to result in warmer water being released from the dam. Although this may improve thermal suitability for humpback chub, any subsequent benefits may be offset by increased abundance
and expansion of warmwater nonnative fish and aquatic fish parasites (see Section 3.6). Uncontrolled warmer water releases may occur as a result of the elevation and subsequent water temperature of withdrawals from Lake Powell. Increased temperatures may allow numbers and species of warmwater nonnative species to increase in the system.

Springs that supply the Little Colorado River within critical habitat may be vulnerable to basin-wide drought and climate change impacting overall habitat availability and the largest self-sustaining population for the species.

3.3 EFFECTS OF THE PROPOSED ACTION ON RAZORBACK SUCKER

The evaluation of potential impacts of the proposed action on razorback sucker included consideration of the results of previous investigations of the species' ecology (Albrecht et al. 2014; Kegerries et al. 2015), as well as studies of the effects of nonnative fish (Coggins et al. 2011; Yard et al. 2011; Whiting et al. 2014), experimental flows (such as HFEs and other flows) and water temperature on native fish and native fish habitat (e.g., Grams et al. 2010; Makinster et al. 2011; Trammell et al. 2002; Ward 2011; Ward and Morton-Starner 2015).

Several aspects of the proposed action may affect razorback sucker directly or indirectly through direct impacts on individuals or habitat; or indirectly by influencing the abundance and distribution of nonnative predators and competitors, or by influencing macroinvertebrate production. Dam operations have the potential to influence river temperatures, the quantity and quality of nearshore rearing habitats (e.g., backwaters), aquatic insects that are food for razorback sucker, and nonnative species abundance; operations may also result in direct mortality to larval or juvenile razorback sucker through stranding. Flow-related actions are analyzed below that may impact razorback sucker; these include base operations, sedimentrelated experiments, and aquatic resource experiments including trout management flows, low summer flows, and sustained low flows for benthic invertebrate production. Because identified impacts of operational flexibility under current dam operations have been identified for native fish, no effects are expected that are not considered for other dam operations. Other non-flow actions, which may impact razorback sucker directly or indirectly and are analyzed below, include mechanical removal of nonnative fish, conservation measures, and continued fisheries research and monitoring under the GCDAMP. Some actions may have negative impacts on individuals, but beneficial effects at a population level.

Efforts to manage riparian vegetation, humpback chub translocations, and sediment monitoring and research are not expected to have impacts on razorback sucker individuals or habitat. Riparian vegetation management would occur in only small, localized areas within the project area. Although riparian vegetation can play an important role in small stream ecosystems, particularly in providing cover, shade, or habitat for terrestrial invertebrates for fish to feed upon, riparian vegetation is less important in larger rivers for providing cover, shade, or energy inputs (Vannote and Sweeney 1980). Thus, riparian vegetation management would have large impacts on the razorback sucker and the species' habitat in the Colorado River.

3.3.1 Direct and Indirect Effects

3.3.1.1 Effects of Base Operations

Base dam operations are expected to change with the implementation of the proposed action. Changes will include more consistency across months, and thus more stability throughout the year, in monthly dam release volumes. The daily fluctuation in discharge would be proportional to the volume (i.e., lower fluctuations during lower flows), with a maximum discharge fluctuation range of 8,000 cfs; this would be the same as the current maximum fluctuation range. The daily downramp rate would be increased, compared to the current condition, by 67% (from 1,500 cfs/hr to 2,500 cfs/hr). Under base operations, year-round, fluctuating flows throughout the life of the LTEMP period may have detrimental, long-term direct impacts on juvenile razorback sucker and rearing habitat, razorback back sucker spawning habitat, and aquatic invertebrate prey (food base).

Stranding of fish is a potential outcome of daily hydropeaking (reviewed in Bunn and Arthington 2002; Nagrodski et al. 2012). Increasing downramp rates under the preferred alternative may increase the risk of stranding larval razorback sucker. Thus far, juvenile (older than larval) life stages of razorback sucker have not been sampled in 2 years of study in the Grand Canyon (Albrecht et al. 2014, Kegerries et al. 2015). It is unclear whether larval razorback sucker experience near 100% mortality prior to transformation to juveniles, or if larvae drift to Lake Mead and rear in the lake. Thus, effects are mainly discussed for larval age classes. Stranding could include fish being temporarily (i.e., until flows came back up hours later) restricted to isolated habitats away from the main channel, which may or may not become desiccated, as a result of dropping water levels. Desiccation of these isolated habitats would result in fatality of any larval razorback sucker present. Factors that may influence the probability of stranding fish in a river with altered flow regimes, such as below hydroelectric plants, include the rate of flow reduction, water temperature, channel geomorphology, and substrate composition, as well as biotic factors including fish life stage and size (reviewed in Nagrodski et al. 2012). Larval razorback sucker have limited swimming ability, and thus are at risk of stranding. Increasing drawdown rates under the preferred alternative will increase the risk of fatality. Larval sampling in the lower Grand Canyon found razorback sucker larvae to be distributed throughout most shoreline habitats from Lava Falls to Pearce Ferry from May to July. The highest densities of larvae were found in isolated pools, which composed less than 2% of all habitats sampled. These pools have a high likelihood of desiccation with daily flow fluctuations; however, survival rates among different habitats have not been studied. The potential for, and the effect of stranding on, individual razorback sucker survival has not been directly investigated; however, a summary of stranding literature indicated that fish stranding as a result of hydroelectric and irrigation projects is well documented (Nagrodski et al. 2012).

Razorback sucker spawn on cobble bars and other gravel substrates near or associated with riffles. Riffles and spawning bars may occur in wider, shallower areas of the channel. Therefore, they are susceptible to flow fluctuations. Continued daily flow fluctuations combined with increased downramp rates will likely increase the risk of desiccation of spawning areas and

incubating razorback sucker eggs, leading to increased mortality rates of eggs. Adult razorbacks can swim off of bars as flows decline, and therefore are at less risk.

Under base operations, through the life of the implementation of the LTEMP preferred alternative, river temperatures are expected to continue to be more suitable for coldwater nonnative species than for warmwater nonnative fish, particularly closer to the dam. In general, the estimated average main channel temperatures modeled for the LTEMP EIS indicated that temperature conditions would be most suitable for warmwater nonnative species, as well as native fish such as razorback sucker at locations farther downstream from Glen Canyon Dam (e.g., RM 157 and RM 225) compared to upstream locations (e.g., RM 0 and RM 61), where temperatures would be more suitable to cold or coolwater nonnative fish (e.g., brown and rainbow trout); this is consistent with past surveys that have found more warmwater fish species in those areas than upstream.

Habitat requirements for the endangered razorback sucker, explained in detail in Section 2.1.3, are summarized here. Suitable water temperatures for spawning, egg incubation, and growth range from 14 to 25°C (FWS 2002b), with estimated optimal temperatures of 18°C for spawning, 19°C for egg incubation, and 20°C for growth (Valdez and Speas 2007). Hatching success is temperature dependent, with the potential for complete mortality occurring at temperatures less than 10°C (FWS 2002b). Young razorback suckers require nursery areas with quiet, warm, shallow water such as tributary mouths, backwaters, and inundated floodplains along rivers, and coves or shorelines in reservoirs (FWS 2002b). Razorback sucker are distributed within the project area, from the Colorado River inflow of Lake Mead and at least upstream as far as an area above Lava Falls in Grand Canyon. The upstream distribution of adult razorback sucker is unknown. During May of 2014, razorback sucker larvae were found in the Colorado River as far upstream as RM 173 (upstream of Lava Falls), which is the farthest upstream razorback sucker spawning has been documented in the Grand Canyon (Albrecht et al. 2014). Warm temperatures may be even more critical for larval razorback suckers than other native suckers to transform into juveniles (see Bestgen 2008). Therefore, continued cold temperatures under base operations during the LTEMP implementation period will continue to negatively impact razorback sucker over the long term. However, experimental stable flows may provide some benefit to offset these impacts (see the following sections).

As described for humpback chub, under base operations and fluctuating flows the food base will continue to be degraded throughout the life of the LTEMP.

In summary, under base operations and coldwater conditions, continued hydropeaking flows with increased downramp rates compared to the existing conditions will continue to degrade nearshore rearing habitats, prevent the establishment of aquatic invertebrates (food base), and increase the risk of stranding larval or juvenile razorback sucker in the long term (throughout the life of the LTEMP). In addition, daily fluctuating flows will degrade razorback sucker spawning habitat, river-wide, throughout the life of the plan.

3.3.1.2 Effects of Sediment-Related Experiments and Low Summer Flows

Base flow fluctuations may impact rearing habitat for larval or juvenile razorback sucker. Juvenile fishes in rivers generally rely on low-velocity nearshore environments for rearing. Backwaters potentially created by both HFEs and low summer flows are one such nearshore habitat frequently used by native fishes in the Grand Canyon, and these habitats and their use by native fishes have been a major focus of several investigations. For example, during and after the 2008 HFE physical scientists described the number and area of backwaters that were created during this flood disturbance, and documented how normal operations under MLFF eroded these features to pre-HFE levels within 6 months (Grams et al. 2010). These backwaters, although short lived, may provide ideal rearing conditions for native fishes because water temperatures are often much greater than in the mainstem, particularly when flows are steady. During the 2000 Low Steady Summer Flow experiment, which included steady flows of 8,000 cfs from June through September 2000, mainstem temperatures were 1.4–3°C higher than under previous dam operations, and backwaters were 0.3-5.3°C warmer (Trammell et al. 2002). Similarly, NPS sampling of 47 backwaters distributed from RM 115 to 220 under steady flows in September 2009 showed that at least 21 backwaters were at least 2°C warmer than the adjacent mainstem (Speas and Trammell 2009). The longer residence time of water in nearshore and backwater habitats with steady flows results in warmer temperatures than in the mainstem (Behn et al. 2010), potentially affecting rearing habitat for larval or juvenile razorback sucker.

Although very few studies in the Grand Canyon have specifically investigated razorback sucker habitat, biologists have demonstrated that densities of juvenile native fishes were greatest in backwater habitats compared to other habitats (Dodrill et al. 2015). However, in the Little Colorado River confluence reach where these investigations occurred, backwaters represent a small percentage of available nearshore habitat. Therefore, investigators concluded that maintenance of backwater habitats was not critical to the persistence of the Little Colorado River confluence aggregation of humpback chub. Nonetheless, no studies have evaluated habitat use by larval or juvenile native fishes in downstream reaches where backwaters are larger and more persistent than in the Little Colorado River confluence aggregation (Grams et al. 2010). Few studies have investigated the use of backwaters or other shoreline habitats that may be important to native fish rearing in the western Grand Canyon, where water temperatures are not ideal, but are more suitable for native fish rearing than closer to Glen Canyon Dam. However, shoreline habitats have the potential to provide warmer rearing habitat than the mainstem under certain conditions (Grams et al. 2010; Trammell et al. 2002). It is likely that prolonging the "turnover" of water in nearshore and backwater habitats with steady flows resulted in warmer temperatures than in the mainstem (Behn et al. 2010). Given the need for warm, productive floodplain or backwater habitats for rearing larval and juvenile native fishes, and the lack or low abundance of nonnative fish found in recent backwater sampling (Albrecht et al. 2014; Kegerries et al. 2015), reduced fluctuations, lower flows, or low summer flows may be beneficial for razorback sucker by providing warm and persistent backwater habitats. Thus, low summer flows included under the proposed action as an experiment after the first 10 years of the LTEMP period would likely increase warming and overall stability in these nearshore habitats, thereby benefitting razorback sucker in Grand Canyon. However, the implementation of the experiment may occur only under rare conditions. Benefits to razorback sucker would be minimal if the experiment was implemented only once or a few times in the LTEMP period.

Due to the impact that HFEs and low summer flows have on the creation and maintenance of backwater habitats important to larval or juvenile razorback sucker, these experiments would be expected to benefit razorback sucker in a similar manner to humpback chub. Low summer flows could potentially create or maintain warm backwater habitat beneficial to razorback sucker rearing. Spring HFEs may also create backwater habitat during a time period that may coincide with spawning and emergence of larval fish.

3.3.1.3 Effects of Low Steady Weekend Flows for Macroinvertebrate Production Experiments

Experimental low steady flows on weekends to attempt to increase macroinvertebrate production and diversity may benefit nearshore habitats by providing stability and some warming in habitats that are more isolated from the mainstem (low flows would be held steadier plus/minus 1,000 cfs fluctuation); however, this would be limited to approximately 34 days over the summer. This experiment will not be conducted during first 2 years of LTEMP. The goal is to replicate it two to three times to determine its effectiveness.

The scientific background for this experimental action is discussed in Section 1.2.12 (and see Kennedy et al. 2016). In addition to potential benefits of increased aquatic food base production, steady flows on weekends may help improve the quality of nearshore habitats for larval or juvenile razorback sucker. However, it is unclear whether 2 days per week of nearshore habitat stability will result in a measureable impact on juvenile growth, rearing, and recruitment. Nevertheless, it does not currently appear that razorback sucker are successfully recruiting from the larval to juvenile life stages, possibly due to cold temperatures and lack of stable rearing habitats. Because larval razorback sucker need stable, low-velocity, warm, productive habitats for rearing, a beneficial effect is expected under this experiment, combined with HFEs to build bars and backwaters.

Sustained low flows for benthic invertebrate production may benefit razorback sucker by providing additional aquatic insect prey and stable nearshore habitats for rearing 2 days per week. This treatment will be discontinued if there is no observed benefit to the food base, trout fishery, or native fish; if there is an increase in warmwater nonnative species or trout at the Little Colorado River; or if there are unacceptable adverse impacts on the trout fishery, humpback chub population, or other resources.

3.3.1.4 Effects of Trout Management Flow Experiments

The potential for TMFs to strand and cause mortality in razorback sucker would increase under the proposed action. The extent of mortality due to stranding of razorback sucker in a given year in Marble and Grand Canyons as a result of TMFs is unknown; it may depend on the quantity of channel margin habitats and their sensitivity to flow changes, the distribution and abundance of juvenile fish in sensitive habitats, the timing and number of TMFs, and the degree of flow attenuation downstream. TMFs could be implemented from May through August, which would overlap with the presence of larval razorback sucker. Given that razorback sucker spawning was documented for the first time in the study area in 2014 (and continued in 2015; Kegerries et al. 2015) and studies are ongoing, potential impacts on the species are particularly difficult to predict. Larval or juvenile razorback sucker may be particularly sensitive to the drastic and rapid fluctuations associated with TMFs, because those life stages may be more reliant than other native species on low-velocity backwater habitats. For example, in both 2014 and 2015, catch-per-unit-effort of razorback sucker larvae was higher in isolated pools and/or backwaters (Albrecht et al. 2014; Kegerries et al. 2015).

In addition, although the location or habitat preference for spawning adult razorback suckers has not been identified or defined in the Grand Canyon, the species is known to spawn on clean cobble bars in other systems (reviewed in FWS 2002b), and Valdez et al. (2012b) identified potential spawning bars in the Grand Canyon at tributary inflows and canyon mouths including Diamond Creek (RM 226), Spencer Canyon (RM 246), and Surprise Canyon (RM 248.3). These and other shallow cobble bars important for spawning and egg incubation may be sensitive to flow variation under the proposed action. Large fluctuations during the spawning period (February or March to July; Kegerries et al. 2015) associated with TMFs implemented between May and August may impact spawning and incubation habitat by dewatering.

In summary, TMFs will have negative, wide-ranging impacts on razorback sucker juveniles or larvae that would continue to occur for the duration of the action. Similarly to humpback chub, risk to razorback sucker would likely vary by location depending upon the level of stage changes experienced and the steepness of shallow nearshore areas. Monitoring of the impacts of trout management flows throughout GCNP would be implemented to assess the effectiveness of the action, as well as the detrimental impacts on native fish and other resources.

3.3.1.5 Effects of Triggered Conservation Actions and Mechanical Removal of Nonnative Fish Experiments

Under the preferred alternative, mechanical removal of nonnative rainbow and brown trout (and other nonnative predators) would be implemented through a triggered, tiered approach (see Appendix D) near the confluence of the Little Colorado River and the Colorado River if conservation actions designed to reverse declines in the Little Colorado River humpback chub aggregation were ineffective.

Razorback sucker have not been sampled near the Little Colorado River inflow since the 1980s or early 1990s. Although removal of nonnative fish would benefit razorback sucker, this action would likely have limited beneficial impacts on the razorback sucker. Warmer and more turbid waters characterize the lower Grand Canyon where evidence of razorback sucker spawning has been found. These areas are less suitable to trout, so removal of nonnative trout near the Little Colorado River would likely have negligible impacts on razorback sucker.

104 0-118 Similar, Tier 1 conservation actions developed to conserve humpback chub, including actions to enhance rearing and translocations, would have no impact on razorback sucker because actions would occur within the Little Colorado River and its inflow, and no negative impacts of humpback chub upon razorback sucker have been noted.

3.3.1.6 Effects of Conservation Measures

Conservation measures from the 2011 Biological Opinion were evaluated and adapted for the LTEMP, where necessary, and included in the preferred alternative. Some additional conservation measures were developed specifically for the LTEMP, and were incorporated into the preferred alternative. The intention of conservation measures is to mitigate or offset potential impacts on ESA-listed species that may result with the implementation of the LTEMP preferred alternative. These measures were developed in consultation with the FWS and biologists with the NPS. Effects on razorback sucker are discussed for each conservation measure below.

Humpback Chub Translocations, Disease, and Parasite Monitoring (in humpback chub), Humpback Chub Refuge, Mainstem Humpback Chub Aggregations

Humpback chub translocations may result in increased growth or improved rearing and survival of juvenile humpback chub, but would likely have minimal impacts on the razorback sucker. Although adult humpback chub may prey upon larval or juvenile native fish, it is unlikely that a population-level impact would be expected on suckers, including the razorback sucker. For example, as the translocated humpback chub population has grown in Havasu Creek, bluehead sucker have continued to recruit and possibly increase in number (Nelson et al. 2016).

Reclamation intends to fund the GCDAMP to conduct disease and parasite monitoring in humpback chub and other fish in the mainstem Colorado River, in addition to ongoing disease and parasite monitoring conducted by USGS/GCMRC in the Little Colorado River. No impacts on razorback sucker would be expected because no razorback sucker would be targeted for parasite monitoring.

A refuge population of humpback chub was established at the FWS SNARRC facility in Dexter, New Mexico. The goal of the refuge is to maintain a genetically diverse population of 1000 humpback chub to be used for propagation in the case of a catastrophic loss of the Grand Canyon population. As for disease and parasite monitoring, no effects on razorback sucker are expected because they would not be targeted for refuge development.

Conserving mainstem humpback chub aggregations was a focus of past conservation measures related to dam operations. In the LTMEP, Reclamation will continue to assist the FWS, NPS, and GDAMP to ensure that a stable or upward trend of humpback chub mainstem aggregations can be achieved. The impacts on individual razorback sucker that may occur as a result of monitoring are covered in Section 2.2.

Razorback Sucker Conservation Measures

Studies designed to understand the status of the razorback sucker population in the Grand Canyon and Lake Mead would continue with the implementation of the preferred alternative. Actions include (a) studies to determine the extent of hybridization in flannelmouth and razorback sucker larvae collected; (b) studies to determine habitat use and distribution of different life stages of razorback sucker to assist in future management of flows that may help to conserve the species (including identification of habitats sensitive to flow fluctuations), and (c) studies to assess the effects of trout management flows and other dam operations on the species.

Although some fatality of individual razorback sucker may occur as a result of these studies, including fatality of larval fish, as has occurred in 2014 and 2015 related to larval studies (Albrecht et al. 2014; Kegerries et al. 2015), the information gained will be critical to understanding population-level impacts on the species due to dam operations and experimental actions under the preferred alternative. The adaptive management structure of the LTEMP will allow for adjustments in management actions throughout the life of the LTEMP. Thus, negative effects on individuals would likely be offset by population-level benefits that may be expected if adjustments in operations are made as a result of these studies.

Nonnative Species Conservation Measures

Nonnative fish will continue to threaten razorback sucker in the Grand Canyon under the LTEMP preferred alternative. As a result of continued colder temperature regimes and clear water from of dam construction and operations, particularly in Glen and Marble Canyons, nonnative cool-/coldwater species are expected to proliferate and continue to prey upon and compete with native fish. However, razorback sucker may be more at risk from warmwater nonnative fish that may become established because they have been primarily found in the lower river. Warmer temperatures, which have been occurring in the lower river recently, may allow for highly piscivorous warmwater nonnative species such as smallmouth bass to proliferate. Smallmouth bass were identified as a major threat to native fish in the Yampa River, and have caused major declines in native fish (Johnson et al. 2008).

Conservation measures that were developed to specifically address nonnative species threats to humpback chub and, secondarily, to benefit razorback sucker, include (a) continued nonnative trout (brown and rainbow) removal efforts at Bright Angel Creek and its inflow, or other areas where expanded populations have been found; (b) feasibility study, and planning and compliance for chemical piscicide renovation at Bright Angel and Shinumo creeks; (c) study of efficacy of a temperature control device on the dam to address future conditions under climate change; (d) potential research and development of a means to prevent passage of warmwater nonnative fish through Glen Canyon Dam; (e) planning, compliance, and implementation of a plan to address (make less suitable) the suitability of warmwater nonnative fish habitat in a slough below Glen Canyon dam; and (f) completion of a plan for rapid response to nonnative fish invasions, using tools not approved in the NPS CFMP (e.g., chemicals). Finally, the use of flows (e.g., TMFs) to inhibit brown trout spawning and recruitment in Glen Canyon or other

mainstem locations will be explored over the life of the LTEMP. This measure is necessary to prevent recently detected spawning of brown trout in Glen Canyon from further expanding or increasing in abundance.

Among nonnatives present in the project area, rainbow trout recruitment and abundance would be the species most influenced by changes in dam operations. In the context of impacts on the razorback sucker, interactions between rainbow trout would be expected to be less frequent, because habitat for rainbow trout is less suitable where razorback sucker were found recently (Lava Falls to Lake Mead). Nevertheless, critical habitat for razorback sucker extends into Marble Canyon and the Paria River, where rainbow trout are abundant, and where razorback sucker were detected in the 1990s. As described above, trout emigration through Marble Canyon to the Little Colorado River reach may be higher under the proposed alternative. The magnitude of the impacts of rainbow trout on razorback sucker may depend on the extent of the current distribution of razorback sucker in upstream areas of the project area (currently relatively unknown, and habitat is less suitable), the effects of TMFs on trout recruitment, and the effectiveness of triggered mechanical removal in the vicinity of the Little Colorado River. However, TMFs may have a greater negative impact on larval or juvenile razorback sucker that prefer still, warmer, nearshore habitats in areas near Lava Falls and downstream, compared to the potential benefits of reducing rainbow trout through TMFs.

As discussed above for impacts on humpback chub, it is unclear what factors (e.g., potential flow regime changes) have influenced populations of brown trout within the project area and their expansion to Glen Canyon. Brown trout have the potential to impact razorback sucker more than rainbow trout; however, the species is primarily found near Bright Angel Creek, near Tapeats Creek, and recently expanding in Glen Canyon. The continuation of brown trout control efforts, as described under the NPS CFMP, would likely maintain low levels of brown trout near areas where their distribution overlaps with razorback sucker. In addition, because brown trout tend to be far-ranging, it is possible the reductions of brown trout at their source (Bright Angel Creek) could benefit razorback sucker in other areas of the canyon by reducing predation risk slightly. The mainstem area near Tapeats Creek is less suitable for trout due to increased turbidity, and it is unlikely that flow regime changes under the proposed action would influence brown trout in that area.

Warmwater nonnative fish are a primary threat to razorback sucker, range-wide (FWS 2002b). Since the 1990s, a shift in species composition from nonnative to native species for unknown reasons has been noted below Diamond Creek (reviewed in Kegerries et al. 2015). Nevertheless, modeled temperature suitability for warmwater nonnative species in the lower river (RM 157 or 225), where razorback sucker may be spawning, would be expected to change little compared to the baseline condition. Regardless, a response to new or expanded populations of warmwater nonnative fish may become necessary in the lower river in the future. The rapid response plan may allow for additional tools to suppress or remove warmwater nonnative fish, reducing negative impacts due to predation upon razorback sucker.

In summary, the nonnative fish conservation measures could all lead to beneficial impacts on razorback sucker by reducing the risk of invasion and establishment of warmwater nonnative species, or allowing for control of new invasions. The outcome will depend on the

effectiveness of the actions, including whether feasible options are developed for management of slough habitat, and prevention of passage of warmwater nonnative fish through the dam, for which no technology currently exists.

3.3.1.7 Effects of Climate Change

More frequent droughts and warmer atmospheric temperatures as a result of climate change and warmer water being released from the dam are likely to impact razorback suckers in the Grand Canyon. Razorback suckers are currently located in the lower reaches of the canyon where backwater habitat is more available. Warmer water temperatures may create more suitable habitat for larvae and juveniles. However, these effects may be offset by increased abundance and expansion of nonnative fish and aquatic fish parasites. In addition, lower reservoir releases may result in fewer backwater habitats available for spawning and rearing of razorback sucker.

3.4 EFFECTS OF THE PROPOSED ACTION ON SOUTHWESTERN WILLOW FLYCATCHER

The southwestern willow flycatcher nests and forages in habitats ranging from dense, multi-storied riparian vegetation (such as cottonwood/willow stands with a mix of trees and shrubs) to dense tamarisk stands with little layering of vegetation. However, changes in the availability of suitable habitat may not necessarily translate into changes in the southwestern willow flycatcher populations. Despite the abundance of woody riparian vegetation (e.g., tamarisk) since construction of the Glen Canyon Dam, numbers of nesting southwestern willow flycatchers in the Grand Canyon have declined since the 1980s and no nests have been confirmed in the Grand Canyon since 2007 (Stroud-Settles et al. 2013). Consequently, the effect of the proposed action will have no effect on willow flycatchers because they do not occur in the Grand Canyon; however, the proposed action may have effects on habitat the flycatchers could potentially occupy.

The proposed action features several types of HFEs: (a) proactive spring HFEs, (b) sediment-triggered spring HFEs, and (c) sediment-triggered fall HFEs. Proactive spring HFEs in April, May, or June coincide with the nesting period of the southwestern willow flycatcher. However, these flows are only anticipated to occur approximately two times (10%) during the 20-year life of the LTEMP. These HFEs could coincide with the southwestern willow flycatcher nesting period (May and June). However, southwestern willow flycatcher nests in the Grand Canyon have typically been located above the elevation of 45,000 cfs (Gloss et al. 2005), the maximum HFE flow. The action proposes approximately <5 years (<25%) of spring HFEs triggered by sediment loading. However, these HFEs occur in March or April, prior to nest initiation for the southwestern willow flycatcher and would have no direct impact on the species. Both types of HFEs that occur in the fall (sediment triggered and extended duration) occur long after nesting and fledging dates of the southwestern willow flycatcher and would have no direct impact on the species. This action is not expected to result in important structural changes in riparian habitat or vegetation productivity that could affect the southwestern willow flycatcher. In addition, invertebrates with only terrestrial life stages are not expected to be affected. The action should result in no impacts on the southwestern willow flycatcher's ability to forage effectively and sustain a nest.

Low summer flows, if tested, would occur for 3 months (July, August, and September). This test could overlap with the nesting period (May to August) during the months of July and August. Low summer flows have monthly average low flows of 10,000 cfs or more during the nesting period, which could sustain some inundated riparian nesting habitat, depending on the topography. Low summer flow experiments under this alternative are not expected to have long-term effects on nesting habitat.

The effects of climate change and continued dry basin hydrology may further diminish habitat for the flycatcher, due primarily to lower water levels within riparian areas.

3.5 EFFECTS OF THE PROPOSED ACTION ON YUMA RIDGWAY'S RAIL

Yuma Ridgway's rail nests can be found on small raised mud hummocks, or in the crotches of small shrubs, just above water in dense cattail and tamarisk habitat (Abbott 1940). The Grand Canyon has very little true cattail/marsh habitat, but it does contain considerable stands of dense tamarisk, both near and in the water. However, most potential habitat in the lowest portion of the Grand Canyon where Yuma Ridgway's rails have been detected is now 10 feet or more above the flowing river level; this is a result of Lake Mead water levels dropping in recent years.

Although they are only anticipated to occur approximately two times (10%) during the 20-year life of the plan, this alternative features proactive spring HFEs in April, May, or June that coincide with the nesting period of the Yuma Ridgway's rail. This alternative also proposes approximately <5 years (<25%) of spring HFEs triggered by sediment loading. There is a possibility that these HFEs, when they coincide with the Yuma Ridgway's rail nesting period (April and May), could impact nests and suitable habitat if they are along the river edge. However, it is unlikely that the nests or habitat would be close enough to the river to be impacted by HFEs because HFEs have minimal stage change in this broad floodplain habitat for the rail. Both types of HFEs that occur in the fall (sediment triggered and extended duration) occur long after nesting and fledging dates of the Yuma Ridgway's rail and would have no direct impact on the species. Low summer flows would occur under this alternative. Lower flows during the May to August nesting and fledging period could impact Yuma Ridgway's rail habitat by decreasing the density of vegetation in the marsh habitat within the rail's preferred nesting and predator avoidance areas. Low summer flow experiments under this alternative are not expected to have long-term effects on potential Yuma Ridgway's rail habitat. Designated critical habitat for the Yuma Ridgway rail does not occur in the area of the proposed action. Effects of climate change and continued dry basin hydrology may further diminish habitat for the rail, due primarily to lower water levels within riparian areas.

3.6 CUMULATIVE EFFECTS

Cumulative effects include the effects of future state, local, private, or tribal management actions that may occur in the project area during the duration of the plan (20 years). Future federal actions that have not been previously approved are not included in this section because additional ESA Section 7 consultation would be required. For example, several federal actions are being planned that would require Section 7 consultation including continued permitting of the Navajo Generating Station and the Red Gap Ranch Pipeline. Existing or previously implemented actions are discussed in Section 2. The following projects were considered when analyzing cumulative effects of the proposed action, and other non-federal actions, upon endangered species.

3.6.1 Grand Canyon Escalade

The Navajo Nation has proposed a 420-ac development project, known as the Grand Canyon Escalade, on the Grand Canyon's eastern rim on the western edge of the Navajo reservation at the confluence of the Little Colorado and Colorado rivers. The development would include a 1.4-mi-long, eight-person tramway (gondola) to transport visitors 3,200 ft from the rim to the canyon floor. On the rim, the development would include retail shops, restaurants, a museum, a cultural/visitor center, a hotel, multiple motels, a lodge with patio, roads, and parking for cars and recreational vehicles. It would also include a restaurant, gift shops, an amphitheater, and a riverwalk (with an elevated walkway) along the canyon floor. Analysis for this project has not been conducted, so impacts have not been fully determined; however, the construction and operation of the Escalade project could result in adverse impacts on natural and cultural resources in the areas of the Little Colorado River confluence, wilderness, visual resources, and resources of importance to multiple Tribes. It could also result in beneficial impacts on the local economy through increased tourism and job creation. The Little Colorado River contains critical spawning habitat for humpback chub.

The Grand Canyon Escalade Project and its associated facilities near the confluence of the Little Colorado River could cause both a localized loss of wildlife habitat and source of wildlife disturbance due to human presence. Wildlife species in the Grand Canyon are currently exposed to various sources of man-made noise ranging from human conversation to aircraft flyovers. The potential effects of noise on wildlife include acute or chronic physiological damage to the auditory system, increased energy expenditures, physical injury incurred during panicked responses, interference with normal activities (e.g., feeding), and impaired communication (AMEC Americas Limited 2005). The response of wildlife to noise would vary by species; physiological or reproductive condition; distance; and the type, intensity, and duration of the disturbance. Regular or periodic noise could cause adjacent areas to be less attractive to wildlife and result in a long-term reduction in use by wildlife in those areas. Responses of wildlife to disturbance often involve activities that are energetically costly (e.g., flying or running), altering their behavior in a way that might reduce food intake, communication, and nesting (Hockin et al. 1992; Brattstrom and Bondello 1983; Cunnington and Fahrig 2010; Francis et al. 2009; Maxell 2000).

3.6.2 Uranium Mining on State and Private Lands

Uranium mining peaked in the 1980s in the Grand Canyon region, but there is now a renewed interest due to increases in uranium prices. Increased uranium mining (on state and private lands) could increase the amount of uranium, arsenic, and other trace elements in local surface water and groundwater flowing into the Colorado River (Alpine 2010). Uranium, other radionuclides, and metals associated with uranium mines can affect the survival, growth, and reproduction of aquatic biota.

Aquatic biota and habitats most likely to be affected during mine development and operation are those associated with small, ephemeral, or intermittent drainages. Impacts on aquatic biota and habitats from the accidental release of regulated or hazardous materials into ephemeral drainages would be localized and small, especially if a rapid response to a release is undertaken. The accidental spill of uranium ore into a permanent stream or river such as Kanab Creek would potentially pose a localized short-term impact on the aquatic resources. However, the potential for such an event is extremely low. Most ore solids would settle in the waterbody within a short distance from a spill site (Edge Environmental, Inc. 2009). It is expected that expedient and comprehensive cleanup actions would be required under U.S. Department of Transportation regulations and that an emergency response plan would be in place for responding to accidents and cargo spills (Edge Environmental, Inc. 2009). Overall, the potential for impacts on aquatic biota from an accidental spill would be small to negligible. Spencer and Wenrich (2011) estimated that if an ore load is washed into the Colorado River and is pulverized and dissolved (a scenario that is extremely unlikely to impossible), the uranium concentration in the river would increase from the current 4.0 ppb to only 4.02 ppb (undetectable against natural variations). Predicted no-chemical-effect concentrations for aquatic vascular plants, aquatic invertebrates, and fish are \geq 5.0 ppb; the lowest chronic concentrations are well above that concentration (see Hinck et al. 2010). For these reasons, the impacts from uranium mining on aquatic biota in the Colorado River or its major tributaries would be localized and would not reduce the viability of affected resources.

In the past, uranium mining led to localized peregrine falcon nest failures in areas such as Kanab Canyon and its multiple side canyons, where numerous mining claims existed (Payne et al. 2010). Although 684,449 ac of federal land administered by BLM north of GCNP (North and East Parcels) and 322,096 ac of federal land administered by the USFS south of GCNP (South Parcel) was withdrawn from locatable mineral exploration and development (i.e., uranium mining), increased uranium mining on non-federal (state and private) lands remaining open to mining could locally affect wildlife habitat (e.g., habitat loss and fragmentation) and increase the amount of uranium, arsenic, and other trace elements in local surface water and groundwater flowing into the Colorado River (Alpine 2010). Edge habitat associated with uranium mines and associated access roads may provide habitat for brownheaded cowbirds (Payne et al. 2010), which are brood parasites of songbirds. In general, any impacts on wildlife from uranium mining would be localized and should not affect the viability of affected resources, especially with the use of best management practices to control mine discharges and proper mine reclamation.

3.6.3 Increased Municipal and Agricultural Water Demand

As the population in the Basin States grows and expands, municipal, industrial, and agricultural water demand continues to increase. A Reclamation study in 2012 showed that the demand for Colorado River Basin water may exceed demand before 2060 (Reclamation 2012b), which may result in lower Lake Powell levels and changes in flow, sediment, and water temperature regimes in the Grand Canyon.

Population and industrial growth, coupled with climate change, will act in concert to increase water demand in the region (Schindler 2001) and lower flows downstream of Glen Canyon Dam. This could stress existing riparian and wetland vegetation, leading to plant community alterations that would affect both wildlife habitats and the wildlife prey base. Climate change would not affect all wildlife species uniformly. Some species would experience distribution contractions and likely shrinking populations, while other species would increase in suitable areas and thus possibly experience increases in population numbers. Generally, the warmer the current range is for a species, the greater the projected distributional increase (or lower the projected loss) will be for that species due to climate change (van Riper et al. 2014).

3.6.4 Urban and Agricultural Runoff

Urban runoff, industrial releases, and municipal discharges are considered some of the leading nonpoint sources of contaminants to surface waters (EPA 2004). Areas of intensive agriculture can have an adverse effect on the water quality as a result of the salinity, nutrients, pesticides, selenium, and other trace elements that are common constituents in agricultural runoff. For example, elevated selenium found in aquatic organisms in Colorado River in Grand Canyon is thought to be partly due to agricultural runoff from areas with soils containing selenium (Walters et al. 2015). It is unclear how contamination due to agricultural and urban discharge may change in the future.

3.6.5 Nonnative Vegetation and Defoliating Nonnative Insect Expansion

The highly flammable tamarisk has created a fire hazard previously absent along the river. This threatens breeding bird populations, as well as other wildlife. In addition, if native or mixed habitat stands burn, monotypic tamarisk will likely recolonize, eliminating the crucial structure necessary for southwestern willow flycatchers and other nesting birds (e.g., thermal buffering through shading becomes insufficient and will be further exacerbated by warming climate trends) (Schell 2005). Additional factors that could affect riparian wildlife habitat include the tamarisk leaf beetle and splendid tamarisk weevil, which occur along much of the Colorado River below Glen Canyon Dam and result in defoliation and mortality of tamarisk. Widespread tamarisk mortality would likely result in a net loss in riparian habitat for at least a decade or more (Paxton et al. 2011). It seems unlikely that the effects of large-scale defoliation in areas dominated by tamarisk will be compensated for by use of tamarisk beetles as a food resource by birds (Puckett and van Riper 2014).

3.6.6 Cumulative Effects Summary

3.6.6.1 Aquatic Species – Humpback Chub, Razorback Sucker, and Kanab Ambersnail

The incremental effects of the LTEMP proposed action on razorback sucker and humpback chub are not expected to contribute significantly to cumulative impacts along the Colorado River corridor or within the basin at large. Examination of the various hydrologic traces used to model effects of the proposed action on aquatic resources indicated that hydrology (i.e., whether a 20-year trace was drier or wetter on average) had a greater influence on the model results than did the operational differences among alternatives. Similarly, climate change has the potential to have greater effects on fish resources than any of the alternatives because of its direct influences on hydrologic patterns. For example, more frequent droughts and warmer atmospheric temperatures have the potential to result in greater increases in the temperature of water being released from the dam than the operational actions being considered, and this in turn may improve thermal suitability for humpback chub, humpback chub aggregations, and native fish. However, any subsequent benefits may be offset by increased abundance and expansion of nonnative fish and aquatic fish parasites. There are a number of other actions being taken within the Colorado River Basin that could also contribute to significant cumulative effects on fish populations or fish communities. For example, actions to increase the number of self-sustaining populations of humpback chub within the basin (e.g., translocation of humpback chub from the Little Colorado River to other tributaries within the Grand Canyon) have the potential to increase overall numbers of humpback chub and could provide some level of protection against catastrophic events in the Little Colorado River that could greatly reduce or eliminate the population of humpback chub in the Grand Canyon.

3.6.6.2 Wildlife

The effects of the LTEMP proposed action on wildlife are relatively small compared to the effects of other factors, especially future hydrology, and they are not expected to contribute significantly to cumulative impacts along the Colorado River corridor or within the basin at large. The proposed action would have little effect on most wildlife species.

3.6.7 Summary of Climate-Related Changes

The general picture for climate change as it relates to Colorado River Basin hydrology includes decreased inflow to the reservoir system (e.g., lower precipitation) and greater losses (e.g., evapotranspiration associated with higher temperatures and increased demand from the growing population). Climate change is expected to result in more frequent and severe drought conditions in the Southwest. Meeting increasing water needs (e.g., the Lake Powell Pipeline project and the Page-LeChee water supply project) will likely lead to lower reservoir levels in Lake Powell, which may already be affected by increased evaporation associated with higher air temperatures. As discussed in the baseline, the decreasing the elevation of Lake Powell can lead

to warmer water discharges from Glen Canyon Dam and increased water temperatures downstream.

Water discharge from springs supports the vegetative habitat upon which Kanab ambersnail relies. Changes in spring discharge due to reduced snowpack, in addition to impacts of HFEs, may further reduce habitat at Vaseys Paradise. The Southwest is already experiencing the effects of climate change (Garfin et al. 2014). The decade from 2001 to 2010 was the warmest on record, with temperatures almost 1.1°C higher than historic averages (Garfin et al. 2014; World Meteorological Organization 2014). Higher temperatures in the Colorado River Basin have resulted in less precipitation falling and being stored as snow at high elevations in the Upper Basin (the main source of runoff to the river), increased evaporative losses, and a shift in the timing of peak spring snowmelt (and high streamflow) to earlier in the year (NAS 2007; Christensen et al. 2004; Jacobs 2011). Such effects are likely to continue well into the foreseeable future (NAS 2007).

Warmer water temperatures would likely increase production rates of algae and invertebrates (Woodward et al. 2010; FWS 2011c). Lower water levels in Lake Powell may also result in increases in the composition and density of zooplankton downstream of Glen Canyon Dam, because waters would be withdrawn closer to the surface (Reclamation 1995). However, warmer temperatures, particularly in winter, may allow many invertebrate species to complete their life cycles more quickly (Schindler 2001). For example, if stream temperatures are raised by only a few degrees in winter, many aquatic insects that normally emerge in May or June may emerge in February or March and face death by freezing or be prevented from mating because they are inactivated by low air temperatures. In addition, increases in stream temperatures may cause an exaggeration in the separation of the emergence of males and females (e.g., males may emerge and die before females emerge) (Nebeker 1971). Temperatures above the optimum can lead to the production of small adults and lower fecundity (Vannote and Sweeney 1980).

Warmer water temperatures can expand the distribution of nonnative species adapted to warmer temperatures. This includes fish parasites such as the Asian tapeworm, anchor worm, and nonnative crayfish. Increased zooplankton due to climate change may increase abundance of cyclopoid copepods. All cyclopoid copepod species appear to be susceptible to infection by, and therefore serve as intermediate hosts for, the Asian tapeworm (Marcogliese and Esch 1989). Crayfish can prey on fish eggs and larvae and can diminish the abundance and structure of aquatic vegetation such as filamentous algae through grazing (FWS 2011c). Nonnative crayfish are present in Lake Powell (northern or virile crayfish [*Orconectes virilis*]) and Lake Mead (red swamp crayfish [*Procambarus clarkii*]). Warmer temperatures may allow the crayfish to expand into the mainstem of the Colorado River either downstream of Lake Powell or upstream of Lake Mead.

Higher temperatures in the Colorado River Basin have resulted in less precipitation falling and being stored as snow at high elevations in the Upper Basin (the main source of runoff to the river), increased evaporative losses, and a shift in the timing of peak spring snowmelt (and high streamflow) to earlier in the year (NAS 2007; Christensen et al. 2004; Jacobs 2011). These effects in turn have exacerbated competition among users (farmers, energy producers, urban dwellers), as well as effects on ecological systems, during a time when due to a rapidly

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

rising population water demand has never been higher (Garfin et al. 2014). The combination of decreasing supply and increasing demand will present a challenge in meeting the water delivery commitments outlined in the Colorado River Compact of 1922 (apportioning water between the Upper and Lower Basins) and the United States–Mexico Treaty of 1944 (which guarantees an annual flow of at least 1.5 million ac-ft to Mexico). In 2007, DOI adopted interim guidelines (Reclamation 2007a) to specify modifications to the apportionments to the Lower Basin states in the event of water shortage conditions.

Changes in temperature and precipitation patterns attributed to climate change could also take a toll on the region's rich diversity of plant and animal species (e.g., widespread loss of trees due to wildfires).

As described above, depending on the extent of drought in the Colorado River basin, lower Lake Powell levels may result in the discharge of warmer than anticipated water into Glen and Grand Canyons. This could increase the likelihood that warmwater nonnative predators will become established. This page intentionally left blank

4 CONCLUSIONS AND DETERMINATIONS OF EFFECT

A summary of effects determinations for the five listed species is presented in Table 6. Analysis of effects determination are based 50 CFR 402.02, in which "[e]ffects of the action refers to the direct and indirect effects of an action on the species or critical habitat, together with the effects of other activities that are interrelated or interdependent with that action, that will be added to the environmental baseline. The environmental baseline includes the past and present impacts of all Federal, State, or private actions and other human activities in the action area, the anticipated impacts of all proposed Federal projects in the action area that have already undergone formal or early section 7 consultation, and the impact of State or private actions which are contemporaneous with the consultation in process."

4.1 LISTED SPECIES DETERMINATIONS

For the purposes of Section 7 consultation under the ESA, the following definitions are used when determining the level of anticipated effect for each listed species and its habitat (FWS and NMFS 1998a).

No effect: The proposed action will not affect listed species or designated critical habitat.

May affect, but is not likely to adversely affect: Effects of the proposed action on listed species is expected to be discountable, insignificant, or completely beneficial.

- Discountable effects are those extremely unlikely to occur. Based on best judgment, a person would not (1) be able to meaningfully measure, detect, or evaluate insignificant effects; or (2) expect discountable effects to occur.
- Insignificant effects relate to the size of the impact and should never reach the level where incidental take may occur.
- Beneficial effects are contemporaneous positive effects without any adverse effects on the species. In the event that the overall effect of the proposed action is beneficial to the listed species, but also likely to cause some adverse effects, then the proposed action "is likely to adversely affect" the listed species.

May affect, and is likely to adversely affect: Effects of the proposed action on listed species may occur as a direct or indirect result of the proposed action or its interrelated or interdependent actions, and the effect is not discountable, insignificant, or beneficial.

Jeopardize the continued existence of a species: Effects of the proposed action would be expected, directly or indirectly, to appreciably reduce the likelihood of both the survival and recovery of a listed species in the wild by reducing the reproduction, numbers, or distribution of that species (50 CFR 402.02).

Basis for Determination Determination Humpback chub (Gila cypha) May affect, is likely to The stranding of young chub could occur during TMFs and downramp rates, but adversely affect longer-term beneficial impacts on older age classes may result from reduced trout. Impacts on habitat include direct, minor, short-term reductions in nearshore habitat could occur in the vicinity of the Little Colorado River with changes in flow stage, but long-term benefit is expected from sand re-deposition that rebuilds and maintains nearshore and backwater nursery habitats. Base operations will continue to degrade the food base. Increased predation from expanded population of rainbow trout is expected under cold water discharge, especially with spring or multiple HFEs. Translocations, brown trout control, triggered mechanical removal, triggered proactive conservation actions, and other conservation measures may offset impacts at the population level. Razorback sucker (Xyrauchen texanus) May affect, is likely to Potential for creation of warm, productive nursery habitats from increased adversely affect reshaping of nearshore deposits and backwater development. Potential short-term dewatering of spawning areas during TMFs. Stranding of larval and young-of-year razorback sucker in nearshore habitats may occur as a result of TMFs and base operations (daily fluctuating flows, including increased downramp rates). Kanab ambersnail (Oxyloma kanabensis) May affect, is likely to Potential habitat would be frequently inundated during HFEs that would occur adversely affect almost annually. Proportionally less habitat area would be scoured and fewer snails would be displaced by lower magnitude HFEs. Sequential HFEs could re-inundate and scour primary habitat prior to full recovery from previous HFE. Southwestern willow flycatcher (*Empidonax traillii extimus*) May affect, not likely to Not expected to be in the action area during the spring HFE release window adversely affect (March-April) and high flows of 45,000 cfs or less are not likely to adversely affect their nesting and feeding sites. Yuma Ridgway's rail (*Rallus obsoletus yumaensis*) Spring HFEs may flood nesting habitat; however, nesting habitat is rare, and May affect, not likely to adversely affect likely above the level HFE effects due to dropping Lake Mead levels.

TABLE 6 Summary of Effects Determinations for the Five Listed Species

4.2 HUMPBACK CHUB

Based on the evaluation contained in this BA, Reclamation has determined that the proposed action may affect and is likely to adversely affect the humpback chub, and may adversely affect its designated critical habitat. This determination is based on adverse effects on habitat, related to fluctuating flows (increased downramp rates), the potential for increases in negative interactions with rainbow trout, and the potential for TMFs to impact larval and juvenile individuals occupying shallow stream margin habitats. Fluctuating flows under base operations would also continue to limit the diversity and abundance of aquatic invertebrate prey. These combined effects could result in lower survival of young fish and less recruitment to the adult population. The unintended consequence of an increased rainbow trout population that could result from HFEs, especially those in spring, would likely increase downstream dispersal of trout into the vicinity of the Little Colorado River where they could prey on and compete with young humpback chub. This effect would also reduce recruitment of humpback chub and possibly the overall population size. However, TMFs may also reduce negative effects on humpback chub related to downstream emigration and predation or competition by rainbow trout, but their effectiveness is uncertain. Mechanical control of nonnative fish would be initiated if adult or subadult humpback chub abundance declined, and proactive conservation actions to improve rearing fail to reverse population declines downstream of Glen Canyon Dam; this could reduce nonnative trout abundance and their effects on competition with and predation on humpback chub. Trout control in Bright Angel Creek and the Bright Angel inflow is also expected to reduce brown trout in the project area; however, the potential to control brown trout spawning in other areas is less certain.

The HFEs are also expected to have long-term beneficial effects on the humpback chub population. Although periodic high flows would likely temporarily affect habitat and reduce the food base, multiple HFEs would be expected to rebuild and maintain backwater habitats—so long as sufficient fine sediment was available—and stimulate productivity in backwaters and nearshore habitats. Experimental sustained low flows for benthic invertebrate production may improve the food base; however, the success of this experiment is highly uncertain. The number of consecutive HFEs that would benefit the ecosystem (e.g., rebuilding and maintenance of habitat, stimulate food base productivity) or adversely modify or alter the ecosystem from periodic scouring is unknown. In addition to the effects of low flows to benefit the food base, this will be investigated as part of the LTEMP.

Low summer flows could lead to stable, more productive nearshore habitats, increased likelihood of the establishment of a more diverse and productive food base, and warmer water temperatures in the Little Colorado River reach and further downstream, as well as contributing to enhanced growth rates of young humpback chub. However, this experimental treatment would be used only rarely, if a single test in the second 10 years of the LTEMP period proves it to be a useful tool for improving humpback chub habitat. It is thought that the potential benefit of an increase in temperature would be greatest if a water temperature of at least 14°C could be achieved, because these warmer temperatures could favor higher growth rates of humpback chub.

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

Finally, as described in Section 3.6.7 the potential for increased occurrence of new warmwater nonnative species' introductions may increase in the future as the likelihood of low Lake Powell levels increases with drought and expanded water use, or during LSF experiments. For example, it is likely that the establishment of green sunfish in a backwater slough in Glen Canyon was the result of fish passing through the Glen Canyon Dam (Ward 2015). There is a chance that over the life of the LTEMP, under extreme drought and low Lake Powell levels, dam release temperatures could reach temperatures suitable for reproduction of high-risk nonnative predatory species such as smallmouth bass (15°C). However, current temperatures are already suitable for many warmwater species in the lower river. The river temperature at the Lees Ferry gauge approached 15°C during the fall of 2015 (GCMRC 2016), and reached 20°C below Diamond Creek. Increases in smallmouth bass in the Yampa River have coincided with the decline of the humpback chub population; the threat smallmouth bass pose to humpback chub is thought to be 10 times greater than the threats posed by channel catfish and northern pike (Johnson et al. 2008).

4.3 RAZORBACK SUCKER

Reclamation has also determined that the proposed action **may affect and is likely to adversely** affect the razorback sucker, and may adversely affect its designated critical habitat. This determination is based on potential adverse effects of TMFs stranding juvenile or larval razorback sucker in nearshore habitats, and long-term, river-wide effects as the result of fluctuating flows under base operations, which will continue to compromise the diversity and abundance of aquatic insect prey and degrade nearshore habitats that are important for rearing. Spawning habitat (cobble bars and riffles) is also expected to be negatively affected by base flow fluctuations (including increased downramp rates) and TMFs. TMFs will be less likely to beneficially impact razorback sucker in the western Grand Canyon by reducing trout abundance in Glen and Marble Canyons, and may result in stranding of larval or young-of-year razorback sucker during the summer months. The impacts of TMFs on razorback sucker will be investigated in the LTEMP, and ongoing razorback sucker monitoring in the western Grand Canyon will assist in this investigation. In addition, TMFs could impact spawning habitat and incubating eggs in shallow cobble areas that are most susceptible to changes in flow. Increased downramp rates could also impact razorback sucker young-of-year.

The increased frequency of HFEs could lead to improved sediment conservation and backwater habitat maintenance, benefitting razorback sucker habitat. Increases in sediment load will temporarily increase turbidity, which larvae use as cover from predators (Kegerries et al. 2009). In addition, modeling the hydrology and temperature regime expected during the life of the LTEMP indicated an improvement in temperature suitability compared to the baseline condition at areas inhabited by razorback sucker (i.e., Diamond Creek). Improved backwater habitats, in addition to stability during low summer flows and low steady weekend flows for macroinvertebrate production, could benefit young-of-year as well; however, low summer flows would be implemented only rarely, and invertebrate flows would only be in place for a total of 34 days per summer.

4.4 KANAB AMBERSNAIL

Reclamation has also determined that the proposed action **may affect and is likely to adversely affect the Kanab ambersnail**. Although impacts from the proposed action related to habitat scouring during HFEs will be limited to the habitat near the river, the impact would occur almost annually, which would exceed the rate of habitat recovery (2.5 years). This impact will be limited to one of two populations of Kanab ambersnail. Nevertheless, habitat quality or quantity may decline as a result of the proposed action, in the context of climate change.

There is no designated critical habitat for this species, and this analysis did not evaluate primary constituent elements. This determination is based on short-term adverse effects on habitat and snails located in the inundation zone at Vasey's Paradise. Habitat and snails below the high-water line are expected to be scoured and transported downstream with little or no survival of snails. The proportion of habitat and the number of snails affected would vary with the magnitude of the high release. For the past HFEs, Reclamation has removed habitat and snails from the projected inundation zone. When the habitat was relocated, the vegetation recovered within about 6 months, but when the habitat was not relocated, recovery was delayed about 2.5 years.

4.5 SOUTHWESTERN WILLOW FLYCATCHER

Reclamation has determines the proposed LTEMP **may affect, but is not likely to adversely affect** the southwestern willow flycatcher and southwestern willow flycatcher habitat. This determination is based on the fact that nesting activity, nests, or young are not expected to be in the action area during any of the HFEs except the two proactive spring HFEs. In addition, high flows of 45,000 cfs are not likely to adversely affect the species nesting and foraging habitat. Designated critical habitat for the southwestern willow flycatcher does not occur in the area of the proposed action.

4.6 YUMA RIDGWAY'S RAIL

Reclamation has determined the proposed LTEMP **may affect**, **but is not likely to adversely affect** the Yuma Ridgway's rail and habitat. Designated critical habitat for the Yuma Ridgway's rail does not occur in the area of the proposed action.

The volume and duration of HFEs is determined by the amount of Paria River sediment that is available. Modeling indicates that the amount of sediment required to trigger spring HFEs will occur in about 26% (5 years) of the 20 years of the LTEMP period. Spring HFEs would occur in April, May, or June and may coincide with the nesting period of the Yuma Ridgway's rail. The Grand Canyon has very little true cattail/marsh habitat and it is unlikely that the nests or habitat would be close enough to the river to be impacted by HFEs because HFEs have minimal stage change in this broad floodplain habitat. In addition, due to dropping levels of Lake Mead, most potential habitat in the lowest portion of Grand Canyon where Yuma Ridgway's rails have been detected is now raised 10 ft or more above the flowing river level, and therefore is out of

reach of fluctuating flows, including HFEs. Any implementation of sediment-triggered spring HFEs will consider resource condition assessments and resource concerns using the interagency process described in Section 1.2.8.

5 LITERATURE CITED

Abbott, C.C., 1940, "Notes from the Salton Sea," California. Condor 42(5):264.

Ackerman, M.W., 2008, 2006 Native Fish Monitoring Activities in the Colorado River, Grand Canyon, Annual Report, Grand Canyon Monitoring and Research Center, Flagstaff, Ariz.

Albert, S., 2005, *Surveys for Southwestern Willow Flycatcher Survey in Grand Canyon National Park*, final report submitted to Grand Canyon National Park, Division of Science and Resource Management, Wildlife Team.

Albrecht, B., Z. Shattuck, and R. Rogers, 2013, *Razorback Sucker Studies on Lake Mead*, *Nevada and Arizona, 2011–2012 Annual Report*, PR-1475-01, prepared by BIO-WEST, Inc., Logan, Utah, for U.S. Bureau of Reclamation and the Department of Resources, Southern Nevada Water Authority.

Albrecht, B., R. Kegerries, J.M. Barkstedt, W.H. Brandenburg, A.L. Barkalow, S.P. Platania, M. McKinstry, B. Healy, J. Stolberg, and Z. Shattuck, 2014, *Razorback Sucker* Xyrauchen texanus *Research and Monitoring in the Colorado River Inflow Area of Lake Mead and the Lower Grand Canyon, Arizona and Nevada*, final report prepared by BIO-WEST, Inc., for U.S. Bureau of Reclamation, Upper Colorado Region, Salt Lake City, Utah.

Albrecht, B.A., P.B. Holden, R.B. Kegerries, and M.E. Golden, 2010, "Razorback Sucker Recruitment in Lake Mead, Nevada-Arizona, Why Here?" *Lake and Reservoir Management* 26:4.

Alpine, A.E. (ed.), 2010, *Hydrological, Geological, and Biological Site Characterization of Breccia Pipe Uranium Deposits in Northern Arizona*, Scientific Investigations Report 2010-5025, U.S. Geological Survey.

AMEC Americas Limited, 2005, *Mackenzie Gas Project Effects of Noise on Wildlife*, prepared for Imperial Oil Resources Ventures Limited, July. Available at http://ulpeis.anl.gov/documents/depies/references/pdfs/AMEC_Americas_2005.pdf. Accessed April 1, 2015.

Andersen, M.E., 2009, *Status and Trends of the Grand Canyon Population of Humpback Chub*, U.S. Geological Survey Fact Sheet 2009-3035, April.

Andersen, M.E., M.W. Ackerman, K.D. Hilwig, A.E. Fuller, and P.D. Alley, 2010, "Evidence of Young Humpback Chub Overwintering in the Mainstem Colorado River, Marble Canyon, Arizona, USA," *The Open Fish Science Journal* 2010(3):42–50.

Anderson, C.R., and S.A. Wright, 2007, "Development and Application of a Water Temperature Model for the Colorado River Ecosystem below Glen Canyon Dam, Arizona," pp. 13–26 in *The American Institute of Hydrology and Technology, 2007 Annual Meeting and International Conference—Integrated Watershed Management—Partnerships in Science, Technology and Planning*, T. Hromadka (ed.), Reno, Nev., April 22–25.

Angradi, T.R., R.W. Clarkson, D.A. Kubly, and S.A. Morgensen, 1992, *Glen Canyon Dam and the Colorado River: Responses of the Aquatic Biota to Dam Operations, Glen Canyon Environmental Studies Report*, Arizona Game and Fish Department, Phoenix, Ariz.

Avery, L.A., J. Korman, and W.R. Persons, 2015, "Effects of Increased Discharge on Spawning and Age-0 Recruitment of Rainbow Trout in the Colorado River at Lees Ferry, Arizona," *North American Journal of Fisheries Management* 35:671–680.

AZGFD (Arizona Game and Fish Department), 1996, *The Ecology of Grand Canyon Backwaters, Cooperative Agreement Report (9-FC-40-07940) to Glen Canyon Environmental Studies*, Flagstaff, Ariz.

AZGFD, 2001, "Gila cypha. Humpback Chub," Heritage Data Management System, Phoenix, Ariz.

Behn, K.E., T.A. Kennedy, and R.O. Hall, Jr., 2010, *Basal Resources in Backwaters of the Colorado River Below Glen Canyon Dam—Effects of Discharge Regimes and Comparison with Mainstem Depositional Environments*, U.S. Geological Survey Open-File Report 2010-1075. Available at http://pubs.usgs.gov/of/2010/1075/.

Bestgen, K.R., 2008, "Effects of Water Temperature on Growth of Razorback Sucker Larvae," *Western North American Naturalist* 68(1):15–30.

Blinn, D.W., J.P. Shannon, L.E. Stevens, and J.P. Carder, 1995, "Consequences of Fluctuating Discharge for Lotic Communities," *Journal of the American Benthological Society* 14:233–248.

Bodensteiner, L.R., and W.M. Lewis, 1992, "Role of Temperature, Dissolved Oxygen, and Backwaters in the Winter Survival of Freshwater Drum (*Aplodinotus grunniens*) in the Mississippi River," *Canadian Journal of Fisheries and Aquatic Sciences* 49:173–184.

Bozek, M.A., L.J. Paulson, and G.R. Wilde, 1990, "Effects of Ambient Lake Mohave Temperatures on Development, Oxygen Consumption, and Hatching Success of the Razorback Sucker," *Environmental Biology of Fishes* 27:255–263.

Brattstrom, B.H., and M.C. Bondello, 1983, "Effects of Off-Road Vehicle Noise on Desert Vertebrates," pp. 167–206 in *Environmental Effects of Off-Road Vehicles, Impacts and Management in Arid Region*, R.H. Webb and H.G. Wilshire (eds.), Springer-Verlag, New York, N.Y.

Brown, B.T., 1988a, "Breeding Ecology of a Willow Flycatcher Population in Grand Canyon, Arizona," *Western Birds* 19:25–33.

Brown, B.T., 1988b, *Monitoring Bird Population Densities along the Colorado River in Grand Canyon: 1987 Breeding Season, Final Report to the Glen Canyon Environmental Studies*, Bureau of Reclamation, Salt Lake City, Utah.

Brown, B.T., 1991, *Status of Nesting Willow Flycatchers along the Colorado River from Glen Canyon Dam to Cardenas Creek, Arizona,* Endangered Species Report No. 20 to the U.S. Fish and Wildlife Service, Phoenix, Ariz.

Browning, M.R., 1993, "Comments on the Taxonomy of *Empidonax traillii* (willow flycatcher)," *Western Birds* 24:241–257.

Bunn, S.E., and A.J. Arthington, 2002, "Basic Principles and Ecological Consequences of Altered Flow Regimes for Aquatic Biodiversity," *Environmental Management* 30:493–507.

Christensen, N.S., A.W. Wood, N. Voisin, D.P. Lettenmaier, and R.N. Palmer, 2004, "The Effects of Climate Change on the Hydrology and Water Resources of the Colorado River Basin," *Climate Change* 62:337–363.

Clarkson, R.W., and M.R. Childs, 2000, "Temperature Effects of Hypolimnial-Release Dams on Early Life Stages of Colorado River Basin Big-River Fishes," *Copeia* 2002:402–412.

Clarkson, R.W., A.T. Robinson, and T.L. Hoffnagle, 1997, "Asian Tapeworm (*Bothriocephalus acheilognathi*) in Native Fishes from the Little Colorado River, Grand Canyon, Arizona," *Great Basin Naturalist* 57:66–69.

Coggins, L.G., Jr., and Walters, C.J., 2009, *Abundance Trends and Status of the Little Colorado River Population of Humpback Chub: An Update Considering Data from 1989–2008*, Open-File Report 2009-1075, U.S. Geological Survey.

Coggins, L.G., Jr., M.D. Yard, and W.E. Pine, 2011, "Nonnative Fish Control in the Colorado River in Grand Canyon, Arizona – An Effective Program or Serendipitous Timing?" *Transactions of the American Fisheries Society* 140(2):456–470.

Converse, Y.K., C.P. Hawkins, and R.A. Valdez, 1998, "Habitat Relationships of Subadult Humpback Chub in the Colorado River through Grand Canyon: Spatial Variability and Implications of Flow Regulation," *Regulated Rivers: Research and Management* 14:267–284.

Corrman, T.E., and C. Wise-Gervais, 2005, *Arizona Breeding Bird Atlas*. University of New Mexico Press, Albuquerque, N.M.

Cross, W.F., C.V. Baxter, K.C. Donner, E.J. Rosi-Marshall, T.A. Kennedy, R.O. Hall, Jr., H.A. Wellard Kelly, and R.S. Rogers, 2011, "Ecosystem Ecology Meets Adaptive Management: Food Web Response to a Controlled Flood on the Colorado River, Glen Canyon," *Ecological Applications* 21(6):2016–2033.

Cross, W.F., C.V. Baxter, E.J. Rosi-Marshall, R.O. Hall, Jr., T.A. Kennedy, K.C. Donner, H.Q. Wellard Kelly, S.E. Seegert, K.E. Behn, and M.D. Yard, 2013, "Foodweb Dynamics in a Large River Discontinuum," *Ecological Monographs* 83(3):311–337.

Culver, M., H.-W. Hermann, M. Miller, B. Roth, and J. Sorensen, 2013, *Anatomical and Genetic Variation of Western Oxyloma* (Pulmonata: Succineidae) *Concerning the Endangered Kanab Ambersnail* (Oxyloma haydeni kanabense) *in Arizona and Utah*, Scientific Investigations Report 2013-5164, U.S. Geological Survey, Reston, Va.

Cunnington, G.M., and L. Fahrig, 2010, "Plasticity in the Vocalizations of Anurans in Response to Traffic Noise," *Acta Oecologia* 36:463–470.

Dodrill, M.J., C.B. Yackulic, B. Gerig, W.E. Pine, J. Korman, and C. Finch, 2015, "Do Management Actions to Restore Rare Habitat Benefit Native Fish Conservation? Distribution of Juvenile Native Fish among Shoreline Habitats of the Colorado River," *River Research and Applications*. doi:10.1002/rra.2842.

Dodrill, M.J., C.B. Yackulic and T.A. Kennedy, 2016, "Prey Size and Availability Limits Maximum Size of Rainbow Trout in a Large Tailwater: Insights from a Drift-foraging Bioenergetics Model," *Can. J. Fish. Aquat. Sci.* 73:1–14.

Donner, K.S., 2011, "Secondary Production Rates, Consumption Rates, and Trophic Basis of Production of Fishes in the Colorado River, Grand Canyon, AZ: An Assessment of Potential Competition for Food," M.S. thesis, Idaho State University, Pocatello, Idaho, April.

Douglas, M.E., and P.C. Marsh, 1996, "Population Estimates/Population Movements of *Gilacypha*, an Endangered Cyprinid Fish in the Grand Canyon Region of Arizona," *Copeia* 1:15–28. Available at http://www.jstor.org/stable/pdfplus/1446938.pdf. Accessed Feb.10, 2010.

Ebersole, J.L., W.J. Liss, and C.A. Frissell, 2001, "Relationship between Stream Temperature, Thermal Refugia and Rainbow Trout *Oncorhynchus mykiss* Abundance in Arid-Land Streams in the Northwestern United States," *Ecology of Freshwater Fishes* 10:1–10.

Eddleman, W.R., 1989, *Biology of the Yuma Clapper Rail in the Southwestern U.S. and Northwestern Mexico*, final report to Bureau of Reclamation, Yuma Projects Office and Fish and Wildlife Service, Region 2, Wyoming Cooperative Fish and Wildlife Research Unit, University of Wyoming.

Eddleman, W.R., and C.J. Conway, 1998, "Clapper Rail (*Rallus longirostris*)," in *The Birds of North America*, No. 340, A. Poole and F. Gill (eds), The Birds of North America, Inc., Philadelphia, Pa.

Edge Environmental, Inc., 2009, *Piñon Ridge Project Environmental Report Montrose County, Colorado*, prepared by Edge Environmental, Inc., Lakewood, Colo., for Energy Fuels Resources Corporation, Lakewood, Colo., Nov.

EPA (U.S. Environmental Protection Agency), 2004, *National Water Quality Inventory: Report to Congress, 2004 Reporting Cycle*, Office of Water, Washington, D.C. Available at http://water.epa.gov/lawsregs/guidance/cwa/305b/2004report index.cfm.

Finch, C., W.E. Pine III, C.B. Yackulic, M.J. Dodrill, M. Yard, B.S. Gerig, L.G. Coggins, Jr., and J. Korman, 2015, "Assessing Juvenile Native Fish Demographic Responses to a Steady Flow Experiment in a Large Regulated River," *River Research and Applications*. doi:10.1002/rra.2893.

Francis, C.D., C. Ortega, and A. Cruz, 2009, "Noise Pollution Changes Avian Communities and Species Interactions," *Current Biology* 19:1415–1419.

Francis, T., D.S. Elverud, B.J. Schleicher, D.W. Ryden, and B. Gerig, 2015, *San Juan River Arm of Lake Powell Razorback Sucker* (Xyrauchen texanus) *Survey: 2012*, draft interim progress report to the San Juan River Endangered Fish Recovery Program.

FWS (U.S. Fish and Wildlife Service), 1990, *Humpback Chub Recovery Plan*, 2nd revision, prepared for U.S. Fish and Wildlife Service, Region 6, Denver, Colo., by Colorado River Fishes Recovery Team.

FWS, 1995, *Kanab Ambersnail* (Oxyloma haydeni kanabensis) *Recovery Plan*, FWS Region 6 and Region 2.

FWS, 2002a, *Humpback Chub Recovery Goals: Amendment and Supplement to the Humpback Chub Recovery Plan*, Mountain-Prairie Region (6), Denver, Colo.

FWS, 2002b, Razorback Sucker Recovery Goals: Amendment and Supplement to the Razorback Sucker Recovery Plan, Mountain-Prairie Region (6), Denver, Colo.

FWS, 2002c, Southwestern Willow Flycatcher Recovery Plan, Albuquerque, N.M.

FWS, 2002d, "Biological Opinion of Section 7 Consultation of Proposed Experimental Releases from Glen Canyon Dam and Removal of Non-Native Fish," memorandum to Regional Director, Bureau of Reclamation, Salt Lake City, Utah; Superintendent, Grand Canyon National Park, Arizona, and Chief, Grand Canyon Monitoring and Research Center, U.S. Geological Survey, Flagstaff, Ariz., Dec. 6.

FWS, 2006, *Yuma Clapper Rail* (Rallus longirostris yumanensis) *5-Year Status Review*, Southwest Region, Phoenix, Ariz.

FWS, 2008, *Final Biological Opinion for the Operation of Glen Canyon Dam*, U.S. Department of the Interior, U.S. Fish and Wildlife Service, Phoenix, Ariz.

FWS, 2009a, Kanab Ambersnail (Oxyloma haydeni kanabensis) Recovery Plan, Denver, Colo.

FWS, 2009b, *Yuma Clapper Rail* (Rallus longirostris) *Recovery Plan*, draft, Southwest Region, Albuquerque, N.M.

FWS, 2011a, Final Biological Opinion on the Operation of Glen Canyon Dam Including High-Flow Experiments and Non-Native Fish Control, Phoenix, Ariz.

FWS, 2011b, "An Overview of Humpback Mark-Recapture Trips in the Little Colorado River During 2010," presentation at Glen Canyon Dam Adaptive Management Technical Work Group Annual Reporting Meeting, Jan. 19.

FWS, 2011c, *5-Year Review: Humpback Chub* (Gila cypha) *Summary and Evaluation*, Upper Colorado River Endangered Fish Recovery Program, Denver, Colo.

FWS, 2012, *5-Year Review: Razorback sucker* (Xyrauchen texanus) *Summary and Evaluation*, Upper Colorado River Endangered Fish Recovery Program Denver, Colo.

FWS, 2014, 5-Year Review: Southwestern Willow Flycatcher (Empidonax traillii extimus), Arizona Ecological Services, Phoenix, Ariz.

FWS and NMFS (National Marine Fisheries Service), 1998, Endangered Species Consultation Handbook: Procedures for Conducting Consultation and Conference Activities under Section 7 of the Endangered Species Act.

Garfin, G., G. Franco, H. Blanco, A. Comrie, P. Gonzalex, T. Piechota, R. Smyth, and R. Waskom, 2014, "Chapter 20: Southwest," pp. 462–486 in *Climate Change Impacts in the United States: The Third National Climate Assessment*, J.M. Melillo, T.C. Richmond, and G.W. Yohe (eds.), U.S. Global Change Research Program. doi:10.7930/JO8G8HMN.

GCMRC (Grand Canyon Monitoring and Research Center), 2014, Grand Canyon Monitoring and Research Center Fiscal Year 2014 Annual Project Report for the Glen Canyon Dam Adaptive Management Program, annual report prepared for Upper Colorado Region, Bureau of Reclamation.

GCMRC, 2015, Grand Canyon Monitoring and Research Center Fiscal Year 2015 Annual Project Report for the Glen Canyon Dam Adaptive Management Program, annual report prepared for Upper Colorado Region, Bureau of Reclamation.

GCMRC, 2016, "Maps and Data Portal, Havasu Creek above the Mouth near Supai, AZ 09404115." Available at http://www.gcmrc.gov/discharge_qw_sediment/station/ GCDAMP/09404115. Accessed July 21, 2016.

Gerig, B., M.J. Dodrill, and W.E. Pine, III, 2014, "Habitat Selection and Movement of Adult Humpback Chub in the Colorado River in Grand Canyon, Arizona, during an Experimental Steady Flow Release," *North American Journal of Fisheries Management* 34(1):39–48. Gloss, S.P., and L.G. Coggins, 2005, "Fishes of Grand Canyon," Chapter 2 in *The State of the Colorado River Ecosystem in Grand Canyon*, U.S. Geological Survey Circular 1282, S.P. Gloss et al. (eds.), U.S. Geological Survey, Reston, Va.

Gloss, S.P., J.E. Lovich, and T.S. Melis (eds.), 2005, *The State of the Colorado River Ecosystem in Grand Canyon*, *A Report of the Grand Canyon Monitoring and Research Center 1991–2004*, U.S. Geological Survey Circular 1282.

Gorman, O.T., 1994, *Habitat Use by Humpback Chub*, Gila cypha, *in the Little Colorado River and Other Tributaries of the Colorado River*, prepared by U.S. Fish and Wildlife Service, Arizona Fisheries Resources Office, Flagstaff, Ariz., for U.S. Bureau of Reclamation, Glen Canyon Environmental Studies.

Gorman, O.T., and D.M. Stone, 1999, "Ecology of Spawning Humpback Chub, *Gila cypha*, in the Little Colorado River near Grand Canyon, Arizona," *Environ. Biol. Fish* 55:115–113.

Grams, P.E., J C. Schmidt, and M.E. Andersen, 2010, 2008 High-flow Experiment at Glen Canyon Dam – Morphologic Response of Eddy-deposited Sandbars and Associated Aquatic Backwater Habitats along the Colorado River in Grand Canyon National Park, Open-File Report 2010-1032, U.S. Geological Survey.

Hamman, R.L., 1982, "Induced Spawning and Culture of Bonytail Chub," *Progressive Fish Culturist* 44:201–203.

Healy, B., E. Omana Smith, C. Nelson, and M. Trammell, 2014, *Translocation of Humpback Chub to Grand Canyon Tributaries and Related Nonnative Fish Control Activities: 2011–2013 Report*, prepared for Upper Colorado Region, Bureau of Reclamation, Interagency Agreement Number: 09-AA-40-2890/Interagency Acquisition No. R10PG40063, Aug. 28.

Healy, B.D., and C. Nelson, 2013, *Havasu Creek Humpback Chub Translocation and Monitoring: May 7–15, 2013, Trip Report*, prepared for Bureau of Reclamation, Upper Colorado Region, May 20.

Healy, B.D., C. Nelson, and E.O. Smith, 2014a, *Havasu Creek Fish Population Monitoring*. *October 7–15, 2014, Trip Report*, prepared for Bureau of Reclamation, Upper Colorado Region, Nov. 19.

Healy, B.D., C. Nelson, and E.O. Smith, 2014b, *Merlin Abyss, Upper Shinumo Creek Nonnative Trout Control October 29–November 3, 2014, Trip Report*, prepared for Bureau of Reclamation, Upper Colorado Region, Dec. 23.

Hinck, J.E., G. Linder, S. Finger, E. Little, D. Tillitt, and W. Kuhne, 2010, "Biological Pathways of Exposure and Ecotoxicity Values for Uranium and Associated Radionuclides," Chapter D in *Hydrological, Geological, and Biological Site Characterization of Breccia Pipe Uranium Deposits in Northern Arizona*, Alpine, A.E. (ed.), Scientific Investigations Report 2010-5025, U.S. Department of the Interior, U.S. Geological Survey.

Hockin, D., et al., 1992, "Examination of the Effects of Disturbance on Birds with Reference to Its Importance in Ecological Assessments," *Journal of Environmental Management* 36:253–286.

Hoffnagle, T.L., 1996, *Changes in Water Quality Parameters and Fish Usage of Backwaters during Fluctuating vs. Short-Term Steady Flows in the Colorado River, Grand Canyon*, prepared by Arizona Game and Fish Department for Glen Canyon Environmental Studies, U.S. Bureau of Reclamation.

Howell, S.N.G., and S. Webb, 1995, *A Guide to the Birds of Mexico and Northern Central America*, Oxford University Press, New York, N.Y.

Hubbard, J.P., 1987, *The Status of the Willow Flycatcher in New Mexico*, New Mexico Department of Game and Fish, Endangered Species Program, Santa Fe, N.M.

IKAMT (Interagency Kanab Ambersnail Monitoring Team), 1998, *The Endangered Kanab Ambersnail at Vaseys Paradise, Grand Canyon, Arizona: 1997 Final Report*, prepared for Grand Canyon Monitoring and Research Center, Flagstaff, Ariz., April 29.

Jacobs, J., 2011, "The Sustainability of Water Resources in the Colorado River Basin," *The Bridge: Linking Engineering and Society*, National Academy of Engineering, Winter: 6–12.

Johnson, B.M., P.J. Martinez, J.A. Hawkins, and K.R. Bestgen, 2008, "Ranking Predatory Threats by Nonnative Fishes in the Yampa River, Colorado, via Bioenergetics Modeling," *North American Journal of Fisheries Management* 28:1941–1953.

Kaeding, L.R., and M.A. Zimmerman, 1983, "Life History and Ecology of the Humpback Chub in the Little Colorado and Colorado Rivers in Grand Canyon," *Transactions of the American Fisheries Society* 112:577–594.

Kegerries, R., and B. Albrecht, 2012, *Razorback Sucker Studies at the Colorado River Inflow of Lake Mead, Nevada and Arizona – 2012*, presentation to the Lake Mead Razorback Sucker Workgroup, Nev.

Kegerries, R., B. Albrecht, and P. Holden, 2009, *Razorback Sucker Studies on Lake Mead*, *Nevada and Arizona*, 2008–2009, final annual report to the U.S. Bureau of Reclamation and Southern Nevada Water Authority, Bio/West, Inc., Logan, Utah.

Kegerries, R., B. Albrecht, R. Rogers, E. Gilbert, W.H. Brandenburg, A.L. Barkalow, S.P. Platania, M. McKinstry, B. Healy, J. Stolberg, E. Omana Smith, C. Nelson, and H. Mohn, 2015, *Razorback Sucker* Xyrauchen texanus *Research and Monitoring in the Colorado River Inflow Area of Lake Mead and the Lower Grand Canyon, Arizona and Nevada*, final report, prepared by BIO-WEST, Inc., for the U.S. Bureau of Reclamation, Upper Colorado Region, Salt Lake City, Utah.

130

Kennedy, T.A., and B.E. Ralston, 2011, "Biological Responses to High-Flow Experiments at Glen Canyon Dam," pp. 93–125 in *Effects of Three High Flow Experiments on the Colorado River Ecosystem Downstream from Glen Canyon Dam, Arizona*, T.S. Melis (ed.), Circular 1366, U.S. Geological Survey, Reston, Va.

Kennedy, T.A., Cross, W.F., Hall, R.O., Jr., Baxter, C.V., and Rosi-Marshall, E.J., 2013, *Native and Nonnative Fish Populations of the Colorado River Are Food Limited—Evidence from New Food Web Analyses*, U.S. Geological Survey Fact Sheet 2013–3039. Available at http://pubs.usgs.gov/fs/2013/3039/. Accessed Jan, 21, 2015.

Kennedy, T.A., M. Dodrill, J. Muehlbauer, C. Yackulic, and R. Payn, 2015, *Big Flood, Small Flood, Spring Flood, Fall Flood: HFE Timing Affects Foodbase Response in Lees Ferry*, Glen Canyon Dam Adaptive Management Program, High Flow Experiment Workshop, February 25–26. Available at https://www.usbr.gov/uc/rm/amp/amwg/mtgs/15feb25. Accessed June 30, 2016.

Kennedy, T.A., J.D. Muehlbauer, C.B. Yackulic, D.A. Lytle, S.W. Miller, K.L. Dibble, E.W. Kortenhoeven, A.N. Metcalfe, and C.V. Baxter, 2016, "Flow Management for Hydropower Extirpates Aquatic Insects, Undermining River Food Webs," *BioScience* biw059, Advanced Access, May 2.

Korman, J., M. Kaplinski, J.E. Hazel III, and T.S. Melis, 2005, *Effects of the Experimental Fluctuating Flows from Glen Canyon Dam in 2003 and 2004 on the Early Life History Stages of Rainbow Trout in the Colorado River*, final report to Grand Canyon Monitoring and Research Center, Flagstaff, Ariz.

Korman, J., M. Kaplinski, and J. Buszowski, 2006, *Effects of Air and Mainstem Water Temperatures, Hydraulic Isolation, and Fluctuating Flows from Glen Canyon Dam on Water Temperatures in Shoreline Environments of the Colorado River in Grand Canyon*, final report prepared for the U.S. Geological Survey, Grand Canyon Monitoring and Research Center, Flagstaff, Ariz.

Korman, J., M. Kaplinski, and T.S. Melis, 2010, *Effects of High-Flow Experiments from Glen Canyon Dam on Abundance, Growth, and Survival Rates of Early Life Stages of Rainbow Trout in the Lees Ferry Reach of the Colorado River*, Open-File Report 2010–1034, U.S. Geological Survey.

Korman, J., M. Kaplinski, and T.S. Melis, 2011, "Effects of Fluctuating Flows and a Controlled Flood on Incubation Success and Early Survival Rates and Growth of Age-0 Rainbow Trout in a Large Regulated River," *Transactions of the American Fisheries Society* 140:487–505.

Korman, J., S.J.D. Martell, C.J. Walters, A.S. Makinster, L.G. Coggins, M.D. Yard, and W.R. Persons, 2012, "Estimating Recruitment Dynamics and Movement of Rainbow Trout (*Oncorhynchus mykiss*) in the Colorado River in Grand Canyon Using an Integrated Assessment Model," *Canadian Journal of Fisheries and Aquatic Sciences* 69(11):1827–1849.

Korman, J., M.D. Yard, and C.B. Yackulic, 2015, "Factors Controlling the Abundance of Rainbow Trout in the Colorado River in Grand Canyon in a Reach Utilized by Endangered Humpback Chub," *Canadian Journal of Fisheries and Aquatic Science* 73:105–124.

Koronkiewicz, T., 2004, "SW willow flycatcher studies-Virgin/Colorado Rivers 2004-Update #4," personal communication from Koronkiewicz (SWCA Environmental Consultants) to E. Leslie (Grand Canyon National Park), June 29.

Koronkiewicz, T., 2005, "SW willow flycatcher studies-Virgin/Colorado Rivers 2005-Update #4," personal communication from Koronkiewicz (SWCA Environmental Consultants) to R.V. Ward (Grand Canyon National Park), June.

Koronkiewicz, T., 2006, "SW willow flycatcher studies-Virgin/Colorado Rivers 2006-Update #4," personal communication from Koronkiewicz (SWCA Environmental Consultants) to R.V. Ward (Grand Canyon National Park), June.

Koronkiewicz, T., 2007, "SW willow flycatcher studies-Virgin/Colorado Rivers 2007-Update #4," personal communication from Koronkiewicz (SWCA Environmental Consultants) to R.V. Ward (Grand Canyon National Park), June.

Koronkiewicz, T., 2008, "SW willow flycatcher studies-Virgin/Colorado Rivers 2008-Update #4," personal communication from Koronkiewicz (SWCA Environmental Consultants) to R.V. Ward (Grand Canyon National Park), June.

Laczek-Johnson, C., and R.V. Ward, 2006, *Southwestern Willow Flycatcher Survey Results in Grand Canyon National Park*, final report, submitted to Grand Canyon National Park, Division of Science and Resource Management, Wildlife Team.

LCRMSCP (Lower Colorado River Multi-Species Conservation Program), 2004, *Lower Colorado River Multi-Species Conservation Program, Volume III: Biological Assessment*, Final, (J&S 00450.00.) Sacramento, Calif., Dec. 17.

LCRMSCP, 2008, Species Accounts for the Lower Colorado River Multi-Species Conservation *Program*, Bureau of Reclamation, Lower Colorado Region, Boulder City, Nev., Sept.

Ligon, J.S., 1961, *New Mexico Birds and Where to Find Them*, University of New Mexico Press, Albuquerque, N.M.

Littlefield, J., 2007, Endangered or Not? Taxonomy of the Kanab Ambersnail, Arizona Agricultural Experiment Station Research Report for 2007.

Maddux, H.R., D.M. Kubly, J.C. DeVos, Jr., W.R. Pearsons, R. Staedicke, and R.L. Wright, 1987, *Effects of Varied Flow Regimes on Aquatic Resources of Glen and Grand Canyons*, Glen Canyon Environmental Studies Technical Report, Arizona Game and Fish Department, Phoenix, Ariz.

Makinster, A.S., W.R. Persons, and L.A. Avery, 2011, *Status and Trends of the Rainbow Trout Population in the Lees Ferry Reach of the Colorado River Downstream from Glen Canyon Dam, Arizona, 1991–2009*, Scientific Investigations Report 2011–5015, U.S. Geological Survey, Reston, Va.

Marcogliese, D.J., and G.W. Esch, 1989, "Experimental and Natural Infection of Planktonic and Benthic Copepods by the Asian Tapeworm, *Bothriocephalus acheilognathi*," *Proceedings of the Helminthological Society of Washington* 56(2):151–155.

Marsh, P.C., 1985, "Effect of Incubation Temperature on Survival of Embryos of Native Colorado River Fishes," *The Southwestern Naturalist* 30(1):129–140.

Marsh, P.C., 1987, "Food of Adult Razorback Sucker in Lake Mohave, Arizona-Nevada," *Transactions of the American Fisheries Society* 116:117–119.

Marsh, P.C., and M.E. Douglas, 1997, "Predation by Introduced Fishes on Endangered Humpback Chub and Other Native Species in the Little Colorado River, Arizona," *Transactions of the American Fisheries Society* 126:343–346.

Marsh, P.C., T.E. Dowling, B.R. Kesner, T.F. Turner, and W.L. Minckley, 2015, "Conservation to Stem Imminent Extinction: The Fight to Save Razorback Sucker *Xyrauchen texanus* in Lake Mohave and Its Implications for Species Recovery," *Copeia* 103(1):141–156.

Maynard, W.R., 1995, Summary of 1994 Survey Efforts in New Mexico for Southwestern Willow Flycatcher (Empidonax traillii extimus)," New Mexico Department of Game and Fish, Santa Fe, N.M.

Maxell, B.A., 2000, Management of Montana's Amphibians: A Review of Factors That May Present a Risk to Population Viability and Accounts on the Identification, Distribution Taxonomy, Habitat Use, Natural History, and the Status and Conservation of Individual Species, prepared for U.S. Department of Agriculture Forest Service, Northern Regional Office (Region 1), Missoula, Mont., Sept. 20. Available at http://www.isu.edu/~petechar/iparc/ Maxell Mgmnt.pdf. Accessed Aug. 10, 2009.

McKernan, R.L., and G. Braden, 1997, *Status, Distribution, and Habitat Affinities of the Southwestern Willow Flycatcher along the Lower Colorado River: Year 1 – 1996*, prepared for U.S. Bureau of Reclamation, Lower Colorado River Region, Boulder City, Nev., and U.S. Fish and Wildlife Service, Carlsbad Field Office, Carlsbad, Calif.

McKernan, R.L., and G. Braden, 1998, *Status, Distribution, and Habitat Affinities of the Southwestern Willow Flycatcher along the Lower Colorado River: Year 2 – 1997,* prepared for U.S. Bureau of Reclamation, Lower Colorado River Region, Boulder City, Nev., and U.S. Fish and Wildlife Service, Carlsbad Field Office, Carlsbad, Calif.

McKernan, R.L., and G. Braden, 1999, *Status, Distribution, and Habitat Affinities of the Southwestern Willow Flycatcher along the Lower Colorado River: Year 3 – 1998*, prepared for U.S. Bureau of Reclamation, Lower Colorado River Region, Boulder City, Nev., and U.S. Fish and Wildlife Service, Carlsbad Field Office, Carlsbad, Calif.

McKernan, R.L., and G. Braden, 2000, *Status, Distribution, and Habitat Affinities of the Southwestern Willow Flycatcher along the Colorado River, Year 4 – 1999*, San Bernardino County Museum, Biological Science Section, Redlands, Calif.

McKernan, R.L., and G. Braden, 2001, *Status, Distribution, and Habitat Affinities of the Southwestern Willow Flycatcher along the Colorado River, Year 5 – 2000*, San Bernardino County Museum, Biological Science Section, Redlands, Calif.

McKernan, R.L., and G. Braden, 2002, *Status, Distribution, and Habitat Affinities of the Southwestern Willow Flycatcher along the Colorado River, Year 6 – 2001*, San Bernardino County Museum, Biological Science Section, Redlands, Calif.

McKernan, R.L., and G. Braden, 2004, *Status, Distribution, and Habitat Affinities of the Southwestern Willow Flycatcher along the Colorado River, Year 7 – 2002: Final Report, San Bernardino County Museum, Biological Science Section, Redlands, Calif.*

McKinney, T., and W. R. Persons, 1999, *Rainbow Trout and Lower Trophic Levels in the Lees Ferry Tailwater below Glen Canyon Dam, Arizona – A Review.*

McKinney, T., D.W. Speas, R.S. Rodgers, and W.R. Persons, 2001a, "Rainbow Trout in a Regulated River below Glen Canyon Dam, Arizona, Following Increased Minimum Flows and Reduced Discharge Variability," *North American Journal of Fisheries Management* 21(1):216–222.

McKinney, T., A.T. Robinson, D.W. Speas, and R.S. Rodgers, 2001b, "Health Assessment, Associated Metrics, and Nematode Parasitism of Rainbow Trout in the Colorado River below Glen Canyon Dam, Arizona," *North American Journal of Fisheries Management* 21:62–69.

Melis, T.S., P.E. Grams, T.A. Kennedy, B.E. Ralston, C.T. Robinson, J.C. Schmidt, L.M. Schmit, R.A. Valdez, and S.A. Wright, 2011, "Three Experimental High-Flow Releases from Glen Canyon Dam, Arizona—Effects on the Downstream Colorado River Ecosystem," Fact Sheet 2011–301, U.S. Geological Survey, Southwest Biological Science Center, Grand Canyon Monitoring and Research Center, Feb. Available at http://pubs.usgs.gov/fs/2011/ 3012/fs2011-3012.pdf. Accessed Feb. 19, 2015.

Meretsky, V., and D. Wegner, 2000, *Kanab Ambersnail at Vasey's Paradise, Grand Canyon National Park, 1998–1999 Monitoring and Research*, final report, Grand Canyon Monitoring and Research Center, Flagstaff, Ariz.
Minckley, W.L., P.C. Marsh, J.E. Brooks, J.E. Johnson, and B.L. Jensen, 1991, "Management toward Recovery of the Razorback Sucker," Ch. 17 in *Battle Against Extinction: Native Fish Management in the American West*, University of Arizona Press, Tucson, Ariz.

Modde, T., R.T. Muth, and G.B. Haines, 2001, "Floodplain Wetland Suitability, Access, and Potential Use by Juvenile Razorback Suckers in the Middle Green River, Utah," *Transactions of the American Fisheries Society* 130:1095–1105.

Mohn, H., B. Albrecht, R.J. Rogers, and R. Kegerries, 2015, *Razorback Sucker Studies on Lake Mead, Nevada and Arizona: 2014–2015 Final Annual Report*, prepared for U.S. Bureau of Reclamation, Lower Colorado River Multi-Species Conservation Program, Boulder City, Nev.

Mueller, G., P.C. Marsh, G. Knowles, and T. Wolters, 2000, "Distribution, Movements, and Habitat Use of Razorback Suckers (*Xyrauchen texanus*) in a Lower Colorado Reservoir, Arizona-Nevada," *Western North American Naturalist* 60:180–187.

Muiznieks, B.D., S.J. Sferra, T.E. Corman, M.K. Sogge, and T.J. Tibbitts, 1994, *Arizona Partners in Flight Southwestern Willow Flycatcher Survey, 1993*, draft, Arizona Game and Fish Department, Nongame and Endangered Wildlife Program, Phoenix, Ariz.

Nagrodski, A., G.D. Raby, C.T. Hasler, M.K. Taylor, and S.J. Cooke, 2012, "Fish Stranding in Freshwater Systems: Sources, Consequences, and Mitigation," *Journal of Environmental Management* 103:133–141.

NAS (National Academies of Science), 2007, Colorado River Basin Water Management: Evaluating and Adjusting to Hydro Climatic Variability, Feb.

Nebeker, A.V., 1971, "Effect of High Winter Water Temperatures on Adult Emergence of Aquatic Insects," *Water Research* 5:777–783.

Nelson, C.B., D. Ward, B. Healy, S. Blackburn, and E. Omana Smith, 2015, *Bright Angel Creek Comprehensive Brown Trout Control Project: October 1–December 1, 2014*, prepared for Bureau of Reclamation, Upper Colorado Region, Feb 4.

Nelson, C., 2016, "Bright Angel Creek Trout Reduction, Grand Canyon National Park," presentation at Glen Canyon Dam Adaptive Management Technical Work Group Annual Reporting Meeting, January 26–27.

Nelson, C., E. Omana Smith, and B. Healy, 2016, *Havasu Creek Humpback Chub Monitoring, October 6–12, 2016 Trip Report*, prepared for the Upper Colorado Region, Bureau of Reclamation, Interagency Agreement Number: 09-AA-40-2890, Feb. 3,

Northrip, Q., R. Palarino, R. Palarino, and R.V. Ward, 2008, *Southwestern Willow Flycatcher at Grand Canyon National Park, 2008 Annual Report*, Grand Canyon National Park, Ariz.

NPS (National Park Service), 2006, *Management Policies 2006*, U.S. Department of Interior, Washington, D.C. Available at http://www.nps.gov/policy/mp2006.pdf. Accessed April 30, 2014.

NPS, 2013, Comprehensive Fisheries Management Plan, Environmental Assessment, Grand Canyon National Park and Glen Canyon National Recreation Area, Coconino County, Arizona, U.S. Department of the Interior, May.

Oberlin, G.E., J.P. Shannon, and D.W. Blinn, 1999, "Watershed Influence on the Macroinvertebrate Fauna of Ten Major Tributaries of the Colorado River through Grand Canyon, Arizona," *The Southwest Naturalist* 44:17–30.

Olden, J.D., N.L. Poff, and K.R. Bestgen, 2006, "Life-History Strategies Predict Fish Invasions and Extirpations in the Colorado River Basin," *Ecological Monographs* 76:25–40. doi:10.1890/05-0330.

Palarino, R., E. Slayton, T. Bowden, J. White, and R.V. Ward, 2010, *Surveying for Southwestern Willow Flycatchers in Grand Canyon National Park, 2010 Annual Report*, Grand Canyon National Park, Ariz.

Paradzick, C.E., T.D. McCarthey, R.F. Davidson, J.W. Rourke, M.W. Sumner, and A.B. Smith, 2001, *Southwestern Willow Flycatcher 200 Survey and Nest Monitoring Report*, Arizona Game and Fish Department, Phoenix, Ariz.

Paukert, C.P., L.G. Coggins, Jr., and C.E. Flaccus, 2006, "Distribution and Movement of Humpback Chub in the Colorado River, Grand Canyon, Based on Recaptures," *Transactions of the American Fisheries Society* 135:539–544.

Paxton, E.H., P. Unitt, M.K. Sogge, M.J. Whitfield, and P. Keim, 2011, "Winter Distribution of Willow Flycatcher Subspecies," *Condor* 113:608–618.

Payne, K., J. White, and R.V. Ward, 2010, *Potential Impacts of Uranium Mining on the Wildlife Resource of Grand Canyon National Park*, U.S. Department of the Interior, National Park Service, Natural Resource Program Center, Natural Sounds Program, Jan.

Persons, W.R., D.R. Van Haverbeke, and M.J. Dodrill, 2016, *Colorado River Fish Monitoring in Grand Canyon, Arizona: 2002–2014 Humpback Chub Aggregations*, U.S. Geological Survey, in review.

Peterson, R.T., 1990, *A Feld Guide to Western Birds*, 3rd ed., Houghton Mifflin Company, Boston, Mass.

Petterson, J.R., and M.K. Sogge, 1996, *Distribution and Breeding Productivity of the Southwestern Willow Flycatcher along the Colorado River in the Grand Canyon – 1996*, summary report, Grand Canyon National Park, Grand Canyon, Ariz., and Northern Arizona University, National Biological Service Colorado Plateau Research Station.

Phillips, A.R., 1948, "Geographic Variation in Empidonax traillii," The Auk 65:507-514.

Phillips, A., J. Marshall, and G. Monson, 1964, *The Birds of Arizona*, University of Arizona Press, Tucson, Ariz.

Pine, W., M. Dodrill, C. Finch, B. Gerig, and K. Kunburg, 2011, *Nearshore Ecology of Grand Canyon Fish 2010 Progress Report*, prepared by University of Florida for Bureau of Reclamation, Upper Colorado Region, April.

Pine W., 2013, *Nearshore Ecology of Grand Canyon Fish Final Report*, prepared for the Upper Colorado Region, Bureau of Reclamation, Interagency Agreement: 08-AA-40-2808, University of Florida, Aug.

Puckett, S.L., and C. van Riper, III, 2014, *Influences of the Tamarisk Leaf Beetle* (Diorhabda carinulata) *on the Diet of Insectivorous Birds along the Dolores River in Southwestern Colorado*, Open File Report 2014-1100, U.S. Geological Survey. Available at http://pubs.usgs.gov/of/2014/1100. Accessed Nov. 25, 2014.

Reclamation (U.S. Bureau of Reclamation), 1995, *Operation of Glen Canyon Dam: Colorado River Storage Project, Arizona, Final Environmental Impact Statement*, U.S. Department of the Interior, Bureau of Reclamation, Salt lake City, Utah, March.

Reclamation, 2002, Proposed Experimental Releases from Glen Canyon Dam and Removal of Non-Native Fish: Environmental Assessment, Salt Lake City, Utah.

Reclamation, 2007a, *Environmental Impact Statement—Colorado River Interim Guidelines for Lower Basin Shortages and Coordinated Operations for Lake Powell and Lake Mead*, U.S. Department of the Interior, Bureau of Reclamation, Upper and Lower Colorado Region, Oct. Available at http://www.usbr.gov/lc/region/programs/strategies.html. Accessed May 2013.

Reclamation, 2007b, *Biological Assessment on the Operation of Glen Canyon Dam and Proposed Experimental Flows for the Colorado River Below Glen Canyon Dam during the Years* 2008–2012, U.S. Department of the Interior, Bureau of Reclamation, Upper Colorado Region, Salt Lake City, Utah, Dec. 1.

Reclamation, 2011a, *Environmental Assessment—Development and Implementation of a Protocol for High-Flow Experimental Releases from Glen Canyon Dam 2011–2020*, U.S. Department of the Interior, Bureau of Reclamation, Upper Colorado River Region, Salt Lake City, Utah.

Reclamation, 2011b, *Environmental Assessment—Non-native Fish Control Downstream from Glen Canyon Dam*, U.S. Department of the Interior, Bureau of Reclamation, Upper Colorado River Region, Salt Lake City, Utah.

Reclamation, 2012a, *Finding of No Significant Impact for the Environmental Assessment for Non-Native Fish Control Downstream from Glen Canyon Dam*, Bureau of Reclamation, Upper Colorado Region, May 22. Available at http://www.usbr.gov/uc/envdocs/ea/gc/nnfc/FINAL-FONSI.pdf. Accessed May 2013.

Reclamation, 2012b, *Colorado River Basin Water Supply and Demand Study*, U.S. Department of the Interior, Bureau of Reclamation.

Reclamation, 2013, *Annual Operating Plan for Colorado River Reservoirs 2013*, U.S. Department of the Interior, Bureau of Reclamation, Upper Colorado River Region, Salt Lake City, Utah.

Ridgely, R.S., and G. Tudor, 1994, *The Birds of South America: Suboscine Passerines*, University of Texas Press, Austin, Tex.

Ripley, S.D., 1977, Rails of the World, David R. Godine Publishers, Boston, Mass.

Robinson, A.T., and M.R. Childs, 2001, "Juvenile Growth of Native Fishes in the Little Colorado River and in a Thermally Modified Portion of the Colorado River," *North American Journal of Fisheries Management* 21:809–815.

Robinson, A.T., R.W. Clarkson, and R.E. Forrest, 1998, "Dispersal of Larval Fishes in a Regulated River Tributary," *Transactions of the American Fisheries Society* 127:722–786.

Rogowski, D.L., P.N. Wolters, and L.K. Winters, 2015, *Colorado River Fish Monitoring in Grand Canyon, Arizona – 2014 Annual Report*, prepared for Grand Canyon Monitoring and Research Center, Flagstaff, Ariz., Feb.

Ross, R.P., and P.E. Grams, 2013, *Nearshore Thermal Gradients of the Colorado River near the Little Colorado River Confluence, Grand Canyon National Park, Arizona, 2010*, Open-File Report 2013-1013, U.S. Geological Survey.

San Diego Natural History Museum, 1995, "*Empidonax traillii extimus* in California," presented at the Willow Flycatcher Workshop, November 17, 1995.

Schell, R.A., 2005, *Effects of Glen Canyon Dam on the Avifauna of the Grand Canyon, Arizona, Ecogeomorphology of the Grand Canyon and Its Tributary Streams*, J. Mount et al. (eds.), University of California, Center for Watershed Sciences, Davis, Calif., March 15. Available at https://watershed.ucdavis.edu/education/classes/ecogeomorphology-grand-canyon.

Schindler, D.W., 2001, "The Cumulative Effects of Climate Warming and Other Human Stresses on Canadian Freshwaters in the New Millennium," *Canadian Journal of Fisheries and Aquatic Sciences* 58:18–29.

Schooley, J.D., and P.C. Marsh, 2007, "Razorback Suckers in the Lower Colorado River Basin over Three Decades: 1974–2004," *North American Journal of Fisheries Management* 27:43–51.

Sferra, S.J., R.A. Meyer, and T.E. Corman, 1995, *Arizona Partners in Flight 1994 Southwestern Willow Flycatcher Survey*, Final Technical Report 69, Arizona Game and Fish Department, Nongame and Endangered Wildlife Program, Phoenix, Ariz.

Sferra, S.J., T.E. Corman, C.E. Paradzick, J.W. Rourke, J.A. Spencer, and M.W. Sumner, 1997, *Arizona Partners in Flight Southwestern Willow Flycatcher Survey: 1993–1996 Summary Report*, Technical Report 113, Arizona Game and Fish Department, Phoenix, Ariz.

Slayton, E., R. Palarino, and R.V. Ward, 2009, *Southwestern Willow Flycatchers at Grand Canyon National Park, 2009 Annual Report*, Grand Canyon National Park, Ariz.

Smith, A.B., C.E. Paradzick, A.A. Woodward, P.E.T. Dockens, and T.D. McCarthey, 2002, *Southwestern Willow Flycatcher 2001 Survey and Nest Monitoring Report*, Nongame and Endangered Wildlife Program Technical Report 191, Arizona Game and Fish Department, Phoenix, Ariz.

Smith, A.B., A.A. Woodward, P.E.T. Dockens, J.S. Martin, and T.D. McCarthey, 2003, *Southwestern Willow Flycatcher 2002 Survey and Nest Monitoring Report*, Nongame and Endangered Wildlife Program Technical Report 210, Arizona Game and Fish Department, Phoenix, Ariz.

Smith, A.B., P.E.T. Dockens, A.A. Tudor, H.C. English, and B.L. Allen, 2004, *Southwestern Willow Flycatcher 2003 Survey and Nest Monitoring Report*, Nongame and Endangered Wildlife Program Technical Report 233, Arizona Game and Fish Department, Phoenix, Ariz.

Sogge, M.K., 1998, Status and Distribution of the Southwestern Willow Flycatcher along the Colorado River in the Grand Canyon – 1997, final report, U.S. Geological Survey, Colorado Plateau Field Station.

Sogge, M.K., and T.J. Tibbitts, 1992, *Southwestern Willow Flycatcher (*Empidonax traillii extimus) *Surveys along the Colorado River in Grand Canyon National Park and Glen Canyon National Recreation Area*, National Park Service, Colorado Park Studies Unit, and Northern Arizona University, Flagstaff, Ariz.

Sogge, M.K., and T.J. Tibbitts, 1994, *Distribution and Status of the Southwestern Willow Flycatcher along the Colorado River in the Grand Canyon – 1994*, summary report, National Biological Service, Colorado Plateau Research Station and Northern Arizona University, Flagstaff, Ariz.

Sogge, M.K., T.J. Tibbitts, and S.J. Sferra, 1993, *Status of the Southwestern Willow Flycatcher along the Colorado River between Glen Canyon Dam and Lake Mead – 1993*, summary report, National Park Service Colorado Park Studies Unit, Northern Arizona University, U.S. Fish and Wildlife Service, and Arizona Game and Fish Department, Flagstaff, Ariz.

Sogge, M.K., T.J. Tibbitts, C. Van Riper III, and T. May, 1995, *Status of the Southwestern Willow Flycatcher along the Colorado River in Grand Canyon National Park – 1995*, summary report, National Biological Service, Colorado Plateau Research Station and Northern Arizona University, Flagstaff, Ariz.

Sogge, M.K., T.J. Tibbitts, and J. Petterson, 1997a, "Status and Ecology of the Southwestern Willow Flycatcher in the Grand Canyon," *Western Birds* 28:142–157.

Sogge, M.K., R.M. Marshall, S.J. Sferra, and T.J. Tibbitts, 1997b, *A Southwestern Willow Flycatcher Natural History Summary and Survey Protocol*, Technical Report NPS/NAUCPRS/NRTR-97/12, National Park Service.

Sogge, M.K., D. Ahlers, and S.J. Sferra, 2010, *A Natural History Summary and Survey Protocol for the Southwestern Willow Flycatcher*, Techniques and Methods 2A-10, U.S. Geological Survey.

Sorensen, J.A., 2009, *Kanab Ambersnail Habitat Mitigation for the 2008 High Flow Experiment*, Technical Report 257, Arizona Game and Fish Department, Phoenix, Ariz., Aug.

Sorensen, J.A., 2012, *Kanab Ambersnail 2011 Status Report*, Technical Report 268, Arizona Game and Fish Department, Phoenix, Ariz., Jan.

Sorensen, J.A., and C.B. Nelson, 2002, *Interim Conservation Plan for* Oxyloma (haydeni) kanabensis *Complex and Related Ambersnails in Arizona and Utah*, Nongame and Endangered Wildlife Program Technical Report 192, Arizona Game and Fish Department, Phoenix, Ariz.

Speas, D., and M. Trammell, 2009, *Aquatic Habitat Characteristics by River Mile in Lower Grand Canyon; Preliminary Results of 2009 Field Survey*, Bureau of Reclamation and National Park Service, Salt Lake City, Utah.

Spence, J.R., C.T. LaRue, and J.D. Grahame, 2011, "Birds of Glen Canyon National Recreation Area, Utah and Arizona," *Monographs of the Western North American Naturalist* 5:20–70.

Spencer, J.E., and K. Wenrich, 2011, *Brecicia-Pipe Uranium Mining in the Grand Canyon Region and Implications for Uranium Levels in Colorado River Water*, OFR-11-04, V1.0, Arizona Geological Survey.

Spurgeon, J.J., 2012, "Translocation of Humpback Chub (*Gila cypha*) and Food-web Dynamics in Grand Canyon National Park Tributary Streams," M.S. thesis, University of Missouri, Columbia.

Spurgeon, J.J., C.P. Paukert, B.D. Healy, M.T. Trammell, D.W. Speas, and E. Omana-Smith, 2015, "Translocation of Humpback Chub into Tributary Streams of the Colorado River: Implications for Conservation of Large-River Fishes," *Transactions of the American Fisheries Society* 144(3):502–514.

Stevens, L.E., F.R. Protiva, D.M. Kubly, V.J. Meretsky, and J. Petterson, 1997a, *The Ecology of Kanab Ambersnail (Succineidae:* Oxyloma haydeni kanabensis *Pilsbry, 1948) at Vaseys Paradise, Grand Canyon, Arizona: 1995 Final Report*, prepared for Grand Canyon Monitoring and Research Center, Flagstaff, Ariz., July 15.

Stevens, L.E., J.P. Shannon, and D.W. Blinn, 1997b, "Colorado River Benthic Ecology in Grand Canyon, Arizona, USA: Dam, Tributary, and Geomorphological Influences," *Regulated Rivers: Research and Management* 13:129–149.

Stewart, B., 2016, "Brown Trout Update Lees Ferry," presentation at Glen Canyon Dam Adaptive Management Technical Work Group Annual Reporting Meeting, January 26–27.

Stiles, F.G., and A.F. Skutch, 1989, *A Guide to the Birds of Costa Rica*, Comstock, Ithaca, New York, N.Y.

Stone, D.M., and O.T. Gorman, 2006, "Ontogenesis of Endangered Humpback Chub (*Gila cypha*) in the Little Colorado River, Arizona," *American Midland Naturalist* 155:123–135.

Stroud-Settles, J., and J. Lawrence, 2011, *Surveying for Southwestern Willow Flycatchers in Grand Canyon National Park, 2011 Annual Report*, Grand Canyon National Park, Ariz.

Stroud-Settles, J., G. Holm, and R. Palarino, 2013, *Surveying for Southwestern Willow Flycatchers in Grand Canyon National Park, 2010–2012, Final Project Report*, U.S. Department of the Interior, National Park Service, Grand Canyon National Park, Grand Canyon, Ariz., Aug.

Todd, R.L., 1986, *A Saltwater Marsh Hen in Arizona. A History of the Yuma Clapper Rail* (Rallus longirostris yumanensis), Completion Report Federal Aid Project W-95-R, Arizona Game and Fish Department, Phoenix, Ariz.

Torgersen, C.E., D.M. Price, H.W. Li, and B.A. McIntosh, 1999, "Multiscale Thermal Refugia and Stream Habitat Associations of Chinook Salmon in Northeastern Oregon," *Ecological Applications* 9(1):301–319.

Trammell, M., R. Valdez, S. Carothers, and R. Ryel, 2002, *Effects of a Low Steady Summer Flow Experiment on Native Fishes of the Colorado River in Grand Canyon, Arizona*, prepared by SWCA Environmental Consultants, for Grand Canyon Monitoring and Research Center, Flagstaff, Ariz.

Tyus, H.M., and C.A. Karp, 1990, "Spawning and Movements of Razorback Sucker, *Xyrauchen texanus*, in the Green River Basin of Colorado and Utah," *The Southwestern Naturalist* 35(4):427–433.

Tyus, H.M., and G.B. Haines, 1991, "Distribution, Habitat Use, and Growth of Age-0 Colorado Squawfish in the Green River Basin, Colorado and Utah," *Transactions of the American Fisheries Society* 120:78–89.

Unitt, P., 1987, "*Empidonax traillii extimus*: An Endangered Subspecies," Western Birds 18:137–162.

USGS (U.S. Geological Survey), 2014, USGS Water Data for the Nation. Available at http://waterdata.usgs.gov/nwis. Accessed Jan. 21, 2015.

Valdez, R.A., 1990, *The Endangered Fish of Cataract Canyon*, final report, prepared for U.S. Bureau of Reclamation, Salt Lake City, Utah, by Bio/West, Inc., Logan, Utah.

Valdez, R.A., 1991, Evaluation of Alternatives for the Glen Canyon Dam Environmental Impact Statement, Report No. TR-250-06, Bio/West, Inc., Logan, Utah.

Valdez, R.A., and G.H. Clemmer, 1982, "Life History and Prospects for Recovery of the Humpback Chub and Bonytail Chub," pp. 109–119 in *Fishes of the Upper Colorado River System: Present and Future*, W.H. Miller et al. (eds.), American Fisheries Society, Western Division, Bethesda, Md.

Valdez, R.A., and W.J. Masslich, 1999, "Evidence of Reproduction of Humpback Chub in a Warm Spring of the Colorado River in Grand Canyon, Arizona," *The Southwestern Naturalist* 44(3):384–387.

Valdez, R.A., and R.J. Ryel, 1995, *Life History and Ecology of the Humpback Chub* (*Gilacypha*), *in the Colorado River, Grand Canyon, Arizona: Final Report*. Available at http://www.gcmrc.gov/library/reports/biological/Fish studies/gces/valdez1995f.pdf.

Valdez, R.A., and R.J. Ryel, 1997, "Life History and Ecology of the Humpback Chub in the Colorado River in Grand Canyon, Arizona," pp. 3–31 in *Proceedings of the Third Biennial Conference of Research on the Colorado Plateau*, C. VanRiper III and E. T. Deshler (eds.).

Valdez, R.A., and D.W. Speas, 2007, *A Risk Assessment Model to Evaluate Risks and Benefits to Aquatic Resources from a Selective Withdrawal Structure on Glen Canyon Dam*, Bureau of Reclamation, Salt Lake City, Utah.

Valdez, R.A., S.W. Carothers, M.E. Douglas, M. Douglas, R.J. Ryel, K. Bestgen, and D.L. Wegner, 2000, *Final Research and Implementation Plan for Establishing a Second Population of Humpback Chub in Grand Canyon*, U.S. Department of the Interior, Grand Canyon Monitoring and Research Center, Flagstaff, Ariz.

Valdez, R.A., D.A. House, M.A. McLeod, and S.W. Carothers, 2012a, *Review and Summary of Razorback Sucker Habitat in the Colorado River System, Report Number 1*, draft, prepared by SWCA Environmental Consultants for U.S. Bureau of Reclamation, Upper Colorado Region, Salt Lake City, Utah,

Valdez, R.A., D.A. House, M.A. McLeod, and S.W. Carothers, 2012b, *Augmentation Strategy for the Razorback Sucker in the Lower Grand Canyon, Report Number 3*, prepared by SWCA Environmental Consultants for U.S. Bureau of Reclamation, Upper Colorado Region, Salt Lake City, Utah. Available at http://www.usbr.gov/uc/rm/crsp/gc/Eq-IntGuide/Eq-IntGuidelines-Fact.pdf.

VanderKooi, S., 2015, *Native and Nonnative Fishes in Glen, Marble, and Grand Canyons*, Glen Canyon Dam Adaptive Management Program, High Flow Experiment Workshop, February 25–26. Available at https://www.usbr.gov/uc/rm/amp/amwg/mtgs/15feb25/Attach_HFE08.pdf. Accessed July 1, 2016.

Van Haverbeke, D.R., D.M. Stone, and M.J. Pillow, 2011, *Mark-Recapture and Fish Monitoring Activities in the Little Colorado River in Grand Canyon during 2010*, prepared by U.S. Fish and Wildlife Service, Flag Staff, Ariz., for U.S. Geological Survey, Grand Canyon Monitoring and Research Center (GCMRC), Flagstaff, Ariz.

Van Haverbeke, R., 2012, personal communication from Van Haverbeke (U.S. Fish and Wildlife Service) to B. Healy (National Park Service), Nov. 28.

Van Haverbeke, D.R., D.M. Stone, and M.J. Pillow, 2013, "Long-Term Monitoring of an Endangered Desert Fish and Factors Influencing Population Dynamics," *Journal of Fish and Wildlife Management* 4(1):163–177.

Van Haverbeke, D.R., K.L Young, and B. Healy, 2016, *Translocation and Refuge Framework for Humpback Chub* (Gila cypha) *in Grand Canyon*, USFWS-AZFWCO-FL-16-03, U.S. Fish and Wildlife Service, Flagstaff, Ariz.

Vannote, R.L., and B.W. Sweeney, 1980, "Geographic Analysis of Thermal Equilibria: A Conceptual Model for Evaluating the Effect of Natural and Modified Thermal Regimes in Aquatic Insect Communities," *The American Naturalist* 115(5):667–695.

van Riper, C., III., J.R. Hatten, J.T. Giermakowski, D. Mattson, J.A. Holmes, M.J. Johnson, E.M. Nowak, K. Ironside, M. Peters, P. Heinrich, K.L. Cole, C. Truettner, and C.R. Schwalbe, 2014, *Projecting Climate Effects on Birds and Reptiles of the Southwestern United States*, Open-File Report 2014–1050, U.S. Geological Survey. Available at http://dx.doi.org/10.3133/ ofr20141050.

Vernieu, W.S., and Anderson, C.R., 2013, *Water Temperatures in Select Nearshore Environments of the Colorado River in Grand Canyon, Arizona, during the Low Steady Summer Flow Experiment of 2000*, Open-File Report 2013–1066, U.S. Geological Survey.

Walters, D.M., E. Rosi-Marshall, T.A. Kennedy, W.F. Cross, and C.V. Baxter, 2015, "Mercury and Selenium Accumulation in the Colorado River Food Web, Grand Canyon, USA," *Environmental Toxicology and Chemistry* 34(10):2385–2394.

Ward, D.L., 2011, "How Does Temperature Affect Fish?" presented at Knowledge Assessment II: 2nd Synthesis Workshop with the Grand Canyon Technical Workgroup—Aquatic Resources, October 18–19, Grand Canyon Monitoring and Research Center, Flagstaff, Ariz.

Ward, D.L., 2015, "Green Sunfish *Lepomis cyanellus*; Risk Assessment for the Colorado River Ecosystem (CRe)," technical memo to K. Grantz, Bureau of Reclamation (High Flow Experiment Technical Team Lead), from U.S. Geological Survey, Southwest Biological Science Center, Sept 30.

Ward, D., and S.A. Bonar, 2003, "Effects of Cold Water on Susceptibility of Age-0 Flannelmouth Sucker to Predation by Rainbow Trout," *The Southwestern Naturalist* 48(1):43–46.

Ward, R.V., and D. Haynes, 2007, *Southwestern Willow Flycatcher Survey Results in Grand Canyon National Park*, final report, prepared for Grand Canyon National Park, Division of Science and Resource Management, Wildlife Team.

Ward, D.L., and R. Morton-Starner, 2015, "Effect of Water Temperature and Fish Size on Predation Vulnerability of Juvenile Humpback Chub to Rainbow Trout and Brown Trout," *Transactions of the American Fisheries Society* 144(6):1184–1191.

Webb, R.H., T.S. Melis, and R.A. Valdez, 2002, *Observations of Environmental Change in Grand Canyon, Arizona*, Water-Resource Investigations Report 02-4080, U.S. Geological Survey, Tucson, Ariz.

Westhoff, J.T., C. Paukert, S. Ettinger-Dietzel, H. Dodd, and M. Siepker, 2014, "Behavioural Thermoregulation and Bioenergetics of Riverine Smallmouth Bass Associated with Ambient Cold-Period Thermal Refuge," *Ecology of Freshwater Fish* 25(1):72–85. doi:10.1111/eff.12192.

Whitfield, M.J., 1990, "Willow Flycatcher Reproductive Response to Brown-headed Cowbird Parasitism," Master's Thesis, California State University, Chico, Calif.

Whitfield, M.J., 1994, *A Brown-headed Cowbird Control Program and Monitoring for the Southwestern Willow Flycatcher, South Fork Kern River, California*, prepared for the California Department of Fish and Game, Kern River Research Center, Weldon, Calif.

Whiting, D.P., C.P. Paukert, B.D. Healy, and J.J. Spurgeon, 2014, "Macroinvertebrate Prey Availability and Food Web Dynamics of Nonnative Trout in a Colorado River Tributary, Grand Canyon," *Freshwater Science* 33(3):872–884.

Willard, F.C., 1912, "A Week Afield in Southern Arizona," The Condor 14:53-63.

Winemiller, K.O., and K.A. Rose, 1992, "Patterns of Life-history Diversification in North American Fishes: Implications for Population Regulation," *Canadian Journal of Fisheries and Aquatic Sciences* 49:2196–2218.

O-158

Winters, L., B. Stewart, D. Rogowski, R. Osterhouldt, P. Wolters, and K. Manuell, 2016, *Long-Term Monitoring of the Lees Ferry Fishery: Update*, Glen Canyon Dam Adaptive Management Program, Technical Work Group Public Meeting, January 27. Available at http://www.usbr.gov/uc/rm/amp/twg/mtgs/16jan26/documents/AR14_Winters.pdf. Accessed July 1, 2016.

Woodward, G., J.B. Dybkjer. J.S. Ólafsson, G.M. Gíslason, E.R. Hannesdóttir, and N. Friberg, 2010, "Sentinel Systems on the Razor's Edge: Effects of Warming on Arctic Geothermal Stream Ecosystems," *Global Change Biology* 16:1979–1991.

World Meteorological Organization, 2014, 2001–2010: A Decade of Climate Extremes, WMO-No. 1103.

Wright, S.A., and T.A. Kennedy, 2011, "Science-Based Strategies for Future High-Flow Experiments at Glen Canyon Dam," in *Effects of Three High-Flow Experiments on the Colorado River Ecosystem Downstream from Glen Canyon Dam, Arizona*, T.S. Melis (ed.)., U.S. Geological Survey Circular 1366.

Wright, S.A., C.R. Anderson, and N. Voichick, 2008, "A Simplified Water Temperature Model for the Colorado River below Glen Canyon Dam," *River Research and Applications* 25(6):675–686.

Yackulic, C.B., M.D. Yard, J. Korman, and D.R. Van Haverbeke, 2014, "A Quantitative Life History of Endangered Humpback Chub That Spawn in the Little Colorado River: Variation in Movement Growth and Survival," *Ecology and Evolution* 4(7):1006–1018. doi:10.1002/ece3.990.

Yard, H.K., 2001, *Southwestern Willow Flycatcher Surveys and Nest Searches 2001*, prepared for Grand Canyon Research and Monitoring Center.

Yard, H.K., 2002, *Southwestern Willow Flycatcher Surveys and Nest Searches 2002*, prepared for Grand Canyon Research and Monitoring Center.

Yard, H.K., 2003, *Southwestern Willow Flycatcher Surveys and Nest Searches 2003*, prepared for Grand Canyon Research and Monitoring Center.

Yard, H.K., C. van Riper III, B.T. Brown, and M.J. Kearsley, 2004, "Diets of Insectivorous Birds along the Colorado River in Grand Canyon, Arizona," *Condor* 106:106–115.

Yard, M.D., L.G. Coggins, C.V. Baxter, G.E. Bennett, and J. Korman, 2011, "Trout Piscivory in the Colorado River, Grand Canyon: Effects of Turbidity, Temperature, and Fish Prey Availability," *Transactions of the American Fisheries Society* 140:471–486.

APPENDIX A:

SOUTHWESTERN WILLOW FLYCATCHER DETECTIONS AND NEST SITES, GRAND CANYON NATIONAL PARK, ARIZONA, 1982–2012

APPENDIX A:

SOUTHWESTERN WILLOW FLYCATCHER DETECTIONS AND NEST SITES, GRAND CANYON NATIONAL PARK, ARIZONA, 1982–2012



APPENDIX B:

SOUTHWESTERN WILLOW FLYCATCHER HISTORIC AND RECENT TERRITORIES AND NESTING SITES, GRAND CANYON NATIONAL PARK, ARIZONA

APPENDIX B:

SOUTHWESTERN WILLOW FLYCATCHER HISTORIC AND RECENT TERRITORIES AND NESTING SITES, GRAND CANYON NATIONAL PARK, ARIZONA

River Mile/Location	Year	Notes		
0 (Lees Ferry)	1909	Single male collected		
0	1933	Specimen collected		
0	1935	Used nest collected		
0	1961	2 male 1 female 1 unknown sex collected		
Lava Canvon	1931			
Little Colorado	1953			
RM 0	1987			
RM 46	1987			
RM 51.5–50.5	1993	1 territory		
RM 71.3–71	1993	1 territory		
RM 277–274	1993	1 territory		
RM 51.5–50.5	1994	4 territories		
RM 65.3	1994	1 territory		
RM 71	1994			
RM 51.5–50.5	1995	1 territory		
RM 65.3	1995	1 territory		
Lake Mead Delta	1995	1 territory		
RM 51.5–50.5	1996	3 territories		
Lake Mead Delta	1996	6 territories		
RM 51.5–50.5	1997	2 territories		
RM 270–168	1997	2 territories (1 presumed nesting at RM 252.9)		
Lake Mead Delta	1997	6 territories/3 nesting pairs		
RM 51.5–50.5	1998	1 territory		
RM 246	1998	2 territories		
RM 254 (Spencer Cany.)	1998	2 territories		
RM 265–263.5	1998	1 territory		
RM 268–264	1998	1 territory		
RM 268–265	1998	5 territories		
RM 270–268	1998	1 territory		
RM 272–268	1998	2 territories		
RM 273–270	1998	2 territories		
RM 277–273	1998	1 territory		
RM 51.5–50.5	1999	1 territory		
RM 246	1999	3 territories		
RM 254 (Spencer Cany.)	1999	2 territories, 3 yg. fledged		
RM 259.5	1999	1 territory (McKernan and Braden report 2)		
RM 266–262.5	1999	1 territory		
RM 268–265	1999	5 territories		
KM 272–268	1999	1 territory		
KM 276	1999	1 territory		
KM 51.5-50.5	2000	1 territory		
KM 246	2000	2 territories		
RM 257.5–257	2000	l territory		

River Mile/Location	Year	Notes	
RM 259.5	2000	1 territory	
RM 266–262.5	2000	1 territory	
RM 268–265	2000	1 territory	
RM 51.5–50.5	2001	1 territory	
RM 246	2001	3 territories	
RM 254	2001	3 pairs	
RM 257.5	2001	1 pair	
RM 259.5	2001	2 territories	
RM 262.5–259.5	2001	1 territory/2 nests	
RM 263.5–262.5	2001	1 territory	
RM 268–265	2001	3 territories	
RM 272–268	2001	2 territories	
RM 276	2001	2 individuals, no nest confirmed	
RM 51.5–50.5	2002	1 territory	
RM 28–29	2003	1 pair	
RM 50.5-51.5	2003	1 pair	
RM 28.3	2004	1 pair	
RM 50.4	2004	1 female (June)	
RM 274.5	2004	1 pair nesting (+1 lone male)	
RM 274.5	2005	1 unpaired male	
RM 274.5	2006	1 pair nesting (+1 lone male)	
RM 259.5	2007	1 pair nesting	
RM 274.5	2007	3 unpaired males	
RM 47	2009	1 individual	
RM 49	2009	1 individual	
RM 50.4–50.7	2009	1 individual	
RM 28.5	2010	1 individual	
RM 196.4	2010	1 individual	
RM 218	2010	1 pair	
RM 275	2010	2 territories	
RM 51.8	2011	1 individual	
RM 183.5	2011	1 individual	

APPENDIX C:

RIDGWAY'S RAIL HISTORIC RECORDS AT GRAND CANYON NATIONAL PARK, ARIZONA

C-1 O-169

APPENDIX C:

RIDGWAY'S RAIL HISTORIC RECORDS AT GRAND CANYON NATIONAL PARK, ARIZONA

TABLE C.1 Ridgway's Rail Historic Records at GrandCanyon National Park, Arizona

River Mile/Location	Year	Notes	
DM 246 DM 277	1006	1 rail: pasting confirmed	
RM 246–RM 277	1990	1 rail; nesting not confirmed	
RM 275	1999	1 male observed	
RM 276	2000	1 female observed	
Burnt Springs (~RM 259.8)	2001	3 individuals detected	



FIGURE C.1 Threatened, Endangered, and Sensitive Species Observed along the Colorado River Corridor (Sources: Drost et al. 2011; FWS 2011b; Johnson et al. 2008; NPS 2013; Stroud-Settles 2012)

APPENDIX D:

PROPOSED ACTION TRIGGERS FOR THE MANAGEMENT OF HUMPBACK CHUB COLORADO RIVER, GRAND CANYON

APPENDIX D

DRAFT FINAL

Proposed Action Triggers for the Management of Humpback Chub

Colorado River, Grand Canyon

November 2015

Developed by an Ad Hoc group of Grand Canyon Aquatic Biologists from USFWS, USGS-GCMRC, AZGFD, NPS, USBR (Kirk Young, David Van Haverbeke, Scott VanderKooi, David Ward, Charles Yackulic, Mike Yard, Brian Healy, Melissa Trammel, David Rogowski, Marianne Crawford)

PURPOSE

Mechanical removal of nonnative species is a controversial issue in the Colorado River through Glen and Grand Canyons. A spring 2015 meeting of Grand Canyon biologists (NPS, USFWS, AZGFD, GCMRC) to assess current trout removal triggers resulted in a concept of early conservation measure intervention to maximize conservation benefit to humpback chub (HBC) and minimize the likelihood of mechanical predator removal.

Many factors affect HBC population dynamics such as water temperature, turbidity, and water volume in the Little Colorado River (LCR). This restrains available conservation actions that can be implemented in the event of a declining population of HBC. We can translocate juveniles and young of the year to other areas within and outside the LCR system, juvenile HBC can be head-started at a hatchery, and we can attempt to remove predators. Other conservation tools may include parasite control (although this is unlikely from a population standpoint), non-native fish control in the LCR, and protect from over-utilization for scientific purposes.

Methods to actively manage temperature releases from Glen Canyon Dam sediment augmentation below the Paria River are not included in the Long-Term Experimental Management Program (LTEMP), for Glen Canyon Dam. Inclusion of infrastructure options including these were eliminated from detailed study in the LTEMP alternatives for a variety of reasons. We mention them here because these methods may still represent the most important potential conservation tools that could be used for the long term conservation of HBC in Grand Canyon and the concepts should not be lost.

While healthy wildlife populations are rarely static, trigger objectives include prescribing actions to reverse/ameliorate impacts in order to maintain the LCR HBC population within an acceptable range; and, secondarily to reduce reliance on mechanical removal of predators. For the purposes of these triggers, it is assumed that the primary drivers of HBC population dynamics are interspecific interactions with non-native species, especially rainbow trout, and low water temperature in the mainstem of the Colorado River (Kaeding and Zimmerman 1983; Douglas and Marsh 1996; USFWS 2002; Coggins et al. 2011; Yard et al. 2011). It is suspected that cold water temperatures suppress growth and thus subject young HBC to predation for extended periods of time. The approach described here puts the emphasis on managing humpback chub as opposed to managing predators. Predator removal will only occur if other conservation measures do not appear to be effective in maintaining targeted HBC population levels.

Two Tier Approach

Two tiers of sequential actions were identified; the first would emphasize conservation actions that would take place early during an adult or sub-adult HBC population decline. The second tier would serve as a backstop prescribing predator removal (Threat Reduction) if conservation measures did not mitigate a decline in HBC abundance.

ACTION TRIGGERS

Tier 1 Trigger – Early Intervention Through Conservation Actions:

 If the combined point estimate for adult HBC (adults defined ≥200 mm) in the Colorado River mainstem LCR aggregation; RM 57-65.9) and Little Colorado River (LCR) falls below 9,000 as estimated by the currently accepted HBC population model (e.g., ASMR, multi-state).

-OR-

- 1b. If recruitment of sub-adult HBC (150-199mm) does not equal or exceed estimated adult mortality such that:
 - 1) Sub-adult abundance falls below a three-year running average of 1,250 fish in the spring LCR population estimates.

-OR-

2) Sub-adult abundance falls below a three-year running average of 810 fish in the mainstem Juvenile Chub Monitoring reach (JCM annual fall population estimate; RM 63.45-65.2).

<u>Tier 1 Trigger Response:</u> Tier 1 conservation actions listed below will be immediately implemented either in the LCR or in the adjacent mainstem. Conservation actions will focus on increasing growth, survival and distribution of HBC in the LCR & LCR mainstem aggregation area.

<u>Tier 2 Trigger - Reduce threat using mechanical removal if conservation actions in Tier 1</u> <u>are insufficient to arrest a population decline:</u>

Mechanical removal of nonnative aquatic predator will ensue: If the point abundance estimate of adult HBC decline to <7,000, as estimated by the currently accepted HBC population model.

Mechanical removal will terminate if:

Predator index (described below) is depleted to less than 60 RBT/km for at least two years in the JCM reach and immigration rate is low (the long term feasibility of using immigration rates as a metric still needs to be assessed), or

Adult HBC population estimates exceed 7,500 and recruitment of sub-adult chub exceed adult mortality for at least two years.

If immigration rate of predators into JCM reach is high, mechanical removal may need to continue. These triggers are intended to be adaptive based on ongoing and future research (e.g.,

Lees Ferry recruitment and emigration dynamics, effects of trout suppression flows, effects of Paria River turbidity inputs on predator survival and immigration rates, interactions between humpback chub and rainbow trout, other predation studies).

ACTION TRIGGERS BACKGROUND AND RATIONALE

Tier 1 Trigger Target

Adult Humpback Chub population target: 9,000

Using an age-structured mark-recapture (ASMR) model, Coggins and Walters (2009) estimated the adult population of the LCR aggregation of HBC in 2008 was approximately 7,650 fish (6,000-10,000 fish considering a range of assumed mortality rates and ageing error). Using a multi-state model, Yackulic et al. (2014) obtained point abundance estimates of adult HBC between ~11,000-13,000 from 2009 through 2012. This increase in adult abundances roughly coincides with the significant increase of adult HBC that first appeared in the LCR post-2006 (Van Haverbeke et al. 2013). We suggest a population estimate of 9,000 adult fish as a desired future conditions target. Estimates falling below 9,000 would trigger additional conservation actions to increase recruitment until the HBC population recovered to 9,000 adult fish. A 9,000 adult chub target is below the most recent estimate of ~11,000-13,000 individuals and would preclude conservation measures from being initiated immediately, but also provides a "buffer zone" above 7,000 adult fish, at which point mechanical removal is warranted, as prescribed in the 2011 high flow Biological Opinion (USFWS 2011).

LCR and mainstem (LCR aggregation) population targets: 2,000 and 7,000 adult HBC, respectively

We separate the 9,000 total adult target number into an LCR component (2,000 adults), and a mainstem Colorado River component (7,000 adults). It is estimated that ~82% of adult HBC reside in the mainstem Colorado River during the non-spawning season (Yackulic et al. 2014, p. 1015). This proportion was based on estimates obtained during September/October 2011, so this proportion would be expected to vary, possibly considerably, on an annual basis. Nevertheless, objectives to maintain 2,000 adults in the LCR and 7,000 adults in the mainstem during the non-spawning season (i.e., September/October) are proposed. A desired target of 2,000 adults in the LCR is reasonable because the average fall population estimate for adults was 2,380 (SE = 518) from 2007-2014, compared to the average level of 789 adults (SE = 281) from 2000-2006 (Van Haverbeke et al. 2015).

LCR Humpback Chub recruitment target

To maintain a population of 2,000 adult HBC in the LCR during the non-spawning season, there must be sufficient recruitment of sub-adult chub (150-199 mm size class). We estimate that a sub-adult chub population of 1,250 fish annually, as measured during the annual spring spawning season in the LCR is sufficient to maintain the adult HBC target population. This number is derived from an assumption that the annual adult mortality rate in the LCR is estimated at 0.35 (Yackulic et al. 2014, updates Yackulic pers. com). Hence 2,000 x 0.35 = 700 new adults needed annually to replace adult mortality. To annually recruit 700 adults, we estimate that 1,250 sub-adults are annually needed (i.e., not all sub-adults will survive into adulthood). If annual mortality for sub-adults in the LCR is 0.44 (Yackulic et al. 2014, updates Yackulic pers. com.), then 700/(1-0.44) = 1,250 sub-adults needed to offset adult mortality. Hence, if the three-year

running average point population size of sub-adult chub measured during the spring season in the LCR drops below 1,250 fish, additional conservation measures would be triggered (Figure 1).

A three-year running average is used for sub-adults because production of younger life stages of HBC can be highly variable (Van Haverbeke et al. 2013). For long-lived species such as HBC, reduced recruitment of sub-adults in any one year can be compensated in subsequent years with increased recruitment. Three years is considered a reasonable timeframe from which to trigger actions to minimize large changes in adult HBC numbers.



Figure 1. Running three year averages (\pm 95% CI) of sub-adult humpback chub abundances based on closed spring markrecapture studies in the Little Colorado River (Van Haverbeke et al. 2013, 2015). For example, the bar for 2003 represents the average abundance of the 150-199 mm size class of humpback chub for 2001, 2002 and 2003 obtained in the Little Colorado River during spring monitoring (note: error bars are large because of typically large annual variability in the abundance of this size class). Additional conservation measures would have been triggered during 2003-2006. The red line represents a trigger value of 1,250 sub-adults, below which conservation measures would be initiated.

Mainstem LCR aggregation recruitment target

To maintain a population of 7,000 adults in the mainstem LCR aggregation reach outside of the spawning season, there must be sufficient recruitment of sub-adult fish (150-199 mm size class). The boundaries of the LCR aggregation in the mainstem traditionally extend from RM 57 (Malagosa Crest) to 65.9 (Lava-Chuar Rapid)(Valdez and Ryel 1995). Since 2009, most mainstem monitoring efforts in the LCR aggregation reach have focused in the JCM (Juvenile Chub Monitoring) reach (RM 63.45-65.2), which is below the LCR and contains ~18% of the adult HBC population found in the mainstem LCR aggregation reach (Yackulic et al. 2014). If ~18% of the population is in the JCM reach, then the desired number of adult chub to maintain in the JCM reach is 7,000 x 0.18 = 1,260 adults. Annual adult mortality in the mainstem LCR aggregation is estimated at 0.15 (Yackulic et al. 2014, updates Yackulic pers. com.). To replace the adults in the JCM reach each year would require 1,250 x 0.15 = 189 adults. Annual mortality of sub-adult chub in the mainstem is estimated at 0.3. Replacing 189 adults annually would require 189/(1-0.3) = 270 sub-adults. Approximately 1/3 of sub-adult chub grow to adult size each year, and accordingly it may take ~3 years1 for a chub in the mainstem to transition from

the sub-adult to the adult size class (Yackulic et al. 2014). Therefore an acceptable target population of sub-adults in the JCM reach each year would be $810 (270 \times 3 = 810)$. As with the LCR component, a running three year average of <810 sub-adults in the JCM reach would trigger conservation actions (Figure 2).

The above scenario assumes that population recruitment dynamics are operating more or less equally throughout the LCR aggregation reach in the mainstem, which is likely not true. Most juvenile chub exiting the LCR are displaced downriver from the confluence (Valdez and Ryel 1995). As such, we might expect that recruitment into adulthood might be more prevalent downstream of the confluence. As such, the proportional number of sub-adults measured in the JCM reach may not reflect the number actually needed to annually replace a total 7,000 adults. In other words, the JCM reach proportional calculation of 810 sub-adults could be low. For example, consider that if the JCM reach harbors a higher than average percent of the mainstem sub-adult chub that recruit into adulthood, then even more than 810 sub-adults in this reach may be needed to maintain a population of 7,000 adults.



Figure 2. Running three year average abundances (\pm 95% CI) of sub-adult humpback chub (150-200 mm) abundances based on multi-state model in the mainstem Colorado River in the Juvenile Chub Monitoring reach (data from Yackulic pers. com.). For example, the bar for 2014 represents the average abundance of sub-adult humpback chub in the Juvenile Chub Monitoring reach during 2012, 2013, and 2014. The red line represents the approximate value of a 3-year running average of 810, below which conservation actions would be enacted.

¹The mainstem LCR population recruitment scenario assumes temperature in the LCR mainstem reach is suitable for growth. If LCR mainstem temperatures are cold (do not exceed 11 oC during the year in the JCM reach), HBC will take longer to reach adulthood, experience greater mortality, and therefore require a larger number of sub-adults targeted to maintain the adult population objective. Target number adjustments will be made prior to implementation of LCR mainstem trigger actions if thought necessary.

Tier 1 Trigger response – HBC Conservation

It is expected that the conservation actions proposed below will assist in ameliorating HBC adult losses or recruitment failures from predation. First, ongoing translocations in the LCR above Chute Falls (~300 fish/year) as well as outside the LCR population (e.g., to Havasu Creek, etc.) will continue, regardless of Tier 1 triggers are met or not. New conservation actions will include expansion of existing activities coupled with experimental actions:

- LCR Expand translocation actions in the LCR by collecting an additional 300-600 young of the year (YOY) HBC and move to above Chute Falls in October.
- LCR Assess efficacy of transporting larval HBC (April/May) into Big Canyon and above Blue Springs in the LCR system. Evaluate growth and survival of these transplants;
- Mainstem LCR Aggregation Larval fish will be removed from LCR (April/May) and head-started at Southwest Native Aquatic Resources and Recovery Center (SNARRC). Once fish reach 150-200 mm they will be translocated to the mainstem LCR reach the following year (currently grow-out space at SNARRC is limited to 750 HBC, use of fish for this purpose would reduce numbers available for other actions, e.g. Havasu, Shinumo.);
- Additional conservation actions as identified and evaluated.

Tier 2 Trigger Targets

Aquatic Predator index

A trout or aquatic predator index is proposed as a means to terminate mechanical removal should it become initiated. Essentially, this is the level (60 predator index fish/km in the JCM reach) at which mechanical removal becomes a futile exercise (i.e., very small return for a high amount of effort). The predator index concept was originally intended to serve as an index whereby mechanical removal would be initiated (e.g., mechanical removal would be initiated once trout levels reached a certain density (~760 index fish/km in the JCM reach). However, because of uncertainty of the actual predation rates of trout on HBC (at differing temperatures, densities, turbidities, etc.), and on its population level effects on HBC, determining an appropriate density of trout at which to initiate mechanical removal is highly uncertain.

A predator index will be developed in the JCM reach to weigh each probable predator by its ability to prey on HBC. The index calculates predator densities by incorporating additional species besides rainbow trout and makes assumptions about their relative predation rates compared to rainbow trout. For example, brown trout are estimated to be about 17 times more predacious on HBC than rainbow trout (Ward and Morton-Starner 2015). Additional predators (e.g., smallmouth bass) could be included through an assignment of their piscivory level relative to rainbow trout. Thus, relative piscivory can be captured in a rainbow trout equivalent predator index (Table 1). For species for which population estimates cannot be estimated with mark/recapture methods, capture probabilities or relative abundance (e.g. catch per unit effort) will be used to estimate the population and incorporate into the density matrix. Also, for certain

species regarded as potentially very piscivorous and dangerous (e.g., small mouth bass, green sunfish), targeted removal efforts for these species may be initiated immediately, regardless of meeting any type of threshold. If initiated, mechanical removal would be terminated once the relative predator index declines to 60 in the JCM reach for two years or HBC recover to a target level. A predator index of 60 in the JCM reach likely represents a point at which there is very diminished return for effort expended, and is roughly equivalent to densities at which mechanical removal was deemed to be not worthwhile as an effective tool to pursue in the past (i.e., mechanical was terminated).

Table 1. Hypothetical predator index. The predator index assigns a relative piscivory rate of 17 to brown trout (Ward and Morton-Starner 2015) and to smallmouth bass (assumed at brown trout rate) and sums the hypothetical numbers of fish. If initiated, mechanical removal would be terminated once the relative predator index declines to 60 in the JCM reach for two years or HBC recover to a target level.

Species	Number	Relative predation factor	RBT equivalent
Brown Trout	21	17	357
Rainbow trout	400	1	400
Smallmouth Bass	1	17	17
Predator index total			774

HBC population level triggers

Continue to use the existing adult HBC population estimate of 7,000, as the trigger for predator removal actions, as stated in the 2011 Biological Opinion (USFWS 2011). Population estimates of sub-adults are not incorporated in Tier 2 triggers, as in Tier 1 triggers.

Tier 2 Trigger response – Threat Reduction.

Mechanical removal of predators from the LCR aggregation reach (& immediate vicinity) will be conducted.

TRIGGER CAVEATS

- If HBC decline and the identified actions are not working, USFWS, in coordination with action agencies and traditionally associated Tribes, will identify future appropriate actions;
- Triggers will be reviewed and modified as necessary (evaluated; new information considered and included; etc.), but no less than every five years;
- Actions and triggers will need to adapt if HBC are found to be impacted by other factors;
- If estimating abundances of small size classes of chub becomes problematic because of population decline (i.e., if numbers get so low capture probability cannot be estimated for

each trip), catch divided by the best estimate of capture probability will be used to estimate abundance.

LITERATURE CITED

- Coggins LG Jr, Walters CJ. 2009. Abundance trends and status of the Little Colorado River population of humpback chub; an update from 1989–2008. U.S. Geological Survey Open-File Report 2009-1075.
- Coggins LG Jr, Yard MD, Pine WE III. 2011. Nonnative fish control in the Colorado River in Grand Canyon, Arizona: an effective program or serendipitous timing? Transactions of the American Fisheries Society 140: 456–470.
- Douglas M.E., and Marsh P.C. 1996. Population estimates/population movements of Gila cypha, an endangered Cyprinid fish in the Grand Canyon region of Arizona. Copeia 1996:15–28.
- Kaeding LR, Zimmerman MA. 1983. Life history and ecology of the humpback chub in the Little Colorado and Colorado Rivers in Grand Canyon. Transactions of the American Fisheries Society 112:577–594.
- U.S. Fish and Wildlife Service. 2002. Humpback chub (Gila cypha) Recovery Goals: amendment andNative Fish Monitoring in Little Colorado River, Grand Canyon supplement to the humpback chub Recovery Plan. Denver, Colorado.
- USFWS. 2011. Final Biological Opinion for the Operation of Glen Canyon Dam including High Flow Experiments Non-native Fish Control. U.S. Fish and Wildlife Service, Phoenix, Arizona, December 23, 2011. AESO/SE 22410-2011-F-0100 and 22410-2011-F-0112. Available: http://www.fws.gov/southwest/es/arizona/Documents/Biol Opin/110112 HFE NNR.pdf
- Valdez RA, Ryel RJ. 1995. Life history and ecology of the humpback chub (Gila cypha) in the Colorado River, Grand Canyon, Arizona. Final Report to Bureau of Reclamation, Salt Lake City, Utah. Contract No. 0-CS-40-09110. Salt Lake City, Utah. Available: <u>http://www.gcmrc.gov/library/reports/biological/Fish_studies/Biowest/Valdez1995f.pdf</u>
- Van Haverbeke DR, Stone DM, Coggins LG Jr., Pillow MJ. 2013. Long-term monitoring of an endangered desert fish and factors influencing population dynamics. Journal of Fish and Wildlife Management 4(1):163–177; e1944-687X.
- Van Haverbeke, D.R., K. Young, D.M. Stone, and M.J. Pillow. 2015. Mark-Recapture and Fish Monitoring Activities in the Little Colorado River in Grand Canyon from 2000 to 2014. Submitted to USGS Grand Canyon Monitoring and Research Center, Flagstaff, Arizona. U.S. Fish and Wildlife Service, Flagstaff, Arizona. 55 pp.
- Ward, D. W. and Morton-Starner, R. 2015. Effects of water temperature and fish size on predation vulnerability of juvenile humpback chub to rainbow trout and brown trout. Transactions of the American Fisheries Society 144(6): 1184-1191.
- Yackulic, C.B., Yard, M.D., Korman, J., Van Haverbeke, D.R., 2014. A quantitative life history of endangered humpback chub that spawn in the Little Colorado River: variation in movement, growth, and survival Ecology and Evolution 4(7):1006–1018.
- Yard MD, Coggins LG, Baxter CV, Bennett GE, Korman J. 2011. Trout piscivory in the Colorado River, Grand Canyon: effects of turbidity, temperature, and fish prey availability. Transactions of the American Fisheries Society 140:471–486.

This page intentionally left blank

D-14 O-186

APPENDIX P:

HIGH-FLOW EXPERIMENT PROTOCOL FOR THE PREFERRED ALTERNATIVE

This page intentionally left blank

APPENDIX P:

HIGH-FLOW EXPERIMENT PROTOCOL FOR THE PREFERRED ALTERNATIVE

High-volume dam releases for sediment conservation are an experimental action that would be implemented under the preferred alternative (Alternative D) of the Glen Canyon Dam Long-Term Experimental and Management Plan (LTEMP). Implementation of high-flow experiments (HFEs) under the preferred alternative would follow the HFE protocol described below for the overall process of implementation of HFEs, including implementation considerations and conditions that would result in discontinuing specific experiments.

HFE releases are restricted to limited periods of the year when the highest volumes of sediment are most likely available for building sandbars. Water year releases would follow the pattern identified for the preferred alternative as adopted by the Secretary of the Interior in the LTEMP Record of Decision (ROD) and the Long Range Operating Criteria (LROC) as currently implemented through the 2007 Colorado River Interim Guidelines for Lower Basin Shortages and Coordinated Operations for Lake Powell and Lake Mead (Reclamation 2007a). Sediment-triggered HFEs may be made in spring (March or April) or fall (October or November) (Figure P-1). Fall Extended HFEs duration would range from less than 1 hr to 250 hr. Spring and fall HFEs which are not Extended HFEs, would have a duration range from less than 1 hr to 96 hr. Proactive HFEs may be made in spring or early summer (April, May or June), and would have a duration range up to 24 hr. HFE magnitude would range from 31,500 cfs to 45,000 cfs. Frequency of HFEs would be determined by tributary sediment inputs, annual release volumes, resource conditions, and the decision process carried out by the Department of the Interior (DOI) (Sections 2.2.4.3 and 2.2.4.4 of the EIS). Extended-duration fall HFEs are limited to a frequency of 4 times total in a 20 year period.

The HFE protocol uses a "store and release" approach for sediment-triggered HFEs, in which sediment inputs are tracked over two accounting periods, one for each seasonal HFE: spring (December through June) and fall (July through November) (Figure P-1). In addition, the HFE protocol can trigger proactive spring HFEs that would be tested only in years with high annual water volume (i.e., ≥ 10 maf) when no sediment-triggered HFE occurs. Implementation of an HFE may require reallocating water from other months in order to maintain flows above the required minimum (i.e., 5,000 to 8,000 cfs). The protocol would implement the maximum possible magnitude and duration of HFE that would achieve a positive sand mass balance in Marble Canyon, as determined by modeling.

P.1 DECISION-MAKING PROCESS

The HFE protocol is a decision-making process that consists of three components: (1) planning, (2) modeling, and (3) decision and implementation. The following three subsections describe each of these components.



FIGURE P-1 Average Monthly Sand Load from the Paria River and Little Colorado River Showing the Fall and Spring HFE Accounting Periods and Implementation Windows

P.1.1 Planning

The first component of the HFE protocol is planning. An important aspect of planning is the development and implementation of research and monitoring activities appropriate to monitor the effects of the HFEs. The Bureau of Reclamation (Reclamation) would be prepared to conduct an HFE if resource conditions are suitable, there is sufficient sediment input or projected annual release to trigger an HFE, and DOI determines conditions are suitable for proceeding. An annual process prior to a decision on conducting experiments, including an HFE, would evaluate the information on the status and trends of the following resources: (1) water quality and water delivery, (2) humpback chub, (3) sediment, (4) riparian ecosystems, (5) historic properties and traditional cultural properties, (6) Tribal concerns, (7) hydropower production and Western Area Power Administration's (WAPA's) assessment of the status of the Basin Fund, (8) the rainbow trout fishery, (9) recreation, and (10) other resources. Although these resources are listed for consideration on a regular basis, DOI intends to retain sufficient flexibility in implementation of HFEs to allow for response to unforeseen circumstances or events that involve any other resources not listed here. The recent discovery of nonnative green sunfish in the Glen Canyon reach illustrates the need to be responsive to unforeseen conditions.

In implementing HFEs and other experiments, the DOI will exercise a formal process of stakeholder engagement to ensure decisions are made with sufficient information regarding the condition and potential effects on important resources. As an initial platform to discuss potential future experimental actions, the DOI will hold Glen Canyon Dam Adaptive Management Program (GCDAMP) annual reporting meetings for all interested stakeholders; these meetings

will present the best available scientific information and learning from previously implemented experiments and ongoing monitoring of resources. As a follow up to this process, the DOI will meet with the GCDAMP Technical Work Group (TWG) to discuss the experimental actions being contemplated for the year.

The DOI also will conduct monthly Glen Canyon Dam operational coordination meetings or calls with the DOI bureaus (U.S. Geological Survey [USGS], National Park Service [NPS], Fish and Wildlife Service [FWS], Bureau of Indian Affairs [BIA], and Reclamation), Western Area Power Administration (WAPA), Arizona Game and Fish Department (AZGFD), and representatives from the Basin States and the Upper Colorado River Commission (UCRC). Each DOI bureau will provide updates on the status of resources and dam operations. In addition, WAPA will provide updates on the status of the Basin Fund, projected purchase power prices, and its financial and operational considerations. These meetings or calls are intended to provide an opportunity for participants to share and obtain the most up-to-date information on dam operational considerations and the status of resources (including ecological, cultural, Tribal, recreation, and the Basin Fund). One liaison from each Basin State and from the UCRC will be allowed to participate in the monthly operational coordination meetings or calls.

P.1.2 Modeling

Mathematical models are used to make recommendations for future sediment-triggered HFEs using contemporary sediment data and forecasted hydrologic data to determine whether suitable sediment and hydrology conditions exist for a sediment-triggered HFE.

The two basic inputs for the modeling are the water input or hydrology, which is taken from the Colorado River Simulation System (CRSS) (Reclamation 1988, 2007b) and the sediment input, which in this case is restricted to inputs from the Paria River. A flow routing model (Wiele and Smith 1996) is used to simulate water passing downstream. A sediment budget model (Wright et al. 2010) is used to integrate the flow routing with the sediment inputs and outputs to determine whether or not a sediment mass balance is achieved for sediment-triggered HFEs (Russell and Huang 2010).

The sand budget is the net amount of sand in metric tons that has accumulated in the river channel over some period of time. In the Paria River, the two primary sand input periods are July through October and January through March (Figure P-1). During these two periods, sand is being accumulated at a higher rate than in other months. In order to accommodate the decision process and to address other resource needs or concerns, the sediment-triggered HFE windows are two-months long (October–November and March–April).

The sand budget model would use the sediment inputs and estimates the outputs for three river reaches where sand is tracked: (1) Lees Ferry/Paria River (RM 0) to RM 30, (2) RM 30 to Little Colorado River (RM 61), and (3) Little Colorado River to RM 87. The first two reaches would be used to estimate the maximum possible magnitude and duration of an HFE that will not create a negative sand mass balance in Marble Canyon (RM 0 to RM 61).

Hydrologic data for implementation would be based on forecasted monthly inflow volumes from the National Weather Service's Colorado Basin River Forecast Center and Reclamation's 24-month study projected storage conditions. The 24-month study computer model projects future reservoir conditions and potential dam operations for the system reservoirs given existing reservoir conditions; inflow forecasts and projections; and a variety of operational policies and guidelines. Monthly volumes would be apportioned to daily dam releases by WAPA. Water supply forecasts and models would be needed to make these projections and uncertainty associated with these projects would need to be considered in the decision-making process (Grantz and Patno 2010). The sediment data used would be real-time accumulated inputs estimated from the Paria River streamflow gages.

Sand availability at the onset of each release window is determined by the amount of sand received from the Paria River during the accumulation period less the amount transported downstream to the Little Colorado River as estimated by the sand routing model. Sand in Grand Canyon received from the Little Colorado River is viewed as an added benefit to the amount received from the Paria River. The Little Colorado River input cycle largely follows the same accrual periods as the Paria River; however, only sand inputs from the Paria River would be used in HFE modeling recommendations.

Each run is evaluated against 16 different HFE magnitudes and durations to determine their possible occurrence in the sediment-triggered HFE window months (Table P-1). The magnitude and duration of an HFE would be determined from the stored sand mass available on October 1 and March 1 of each water year, and the forecasted hydrology. The model evaluates each of the 16 sediment-triggered HFE types sequentially starting with the highest magnitude and duration of release. For example, the initial run determines if there is enough sediment available to achieve a positive sand mass balance in Marble Canyon for a release of 45,000 cfs for 250 hours. A positive sand mass balance is defined as a condition in which the amount of sediment being delivered by tributaries into the system exceeds the amount being exported from the system by ongoing dam operations and HFEs in the accounting period under consideration.

If the model run concludes that enough sediment is not available to achieve a positive sand mass balance, the next lower magnitude or duration sediment-triggered HFE is evaluated by the model. This is repeated until a sediment-triggered HFE scenario is reached that can be implemented with the available sediment or it is determined that a sediment-triggered HFE cannot be implemented. If it is determined that a sediment-triggered HFE cannot be implemented and the projected annual volume is greater than or equal to 10 maf, a proactive spring HFE would be triggered.

	Peak Discharge	Duration at Peak
HFE ID	(cfs)	(hours)
1	45,000	250
2	45,000	192
3	45,000	144
4	45,000	96
5	45,000	72
6	45,000	60
7	45,000	48
8	45,000	36
9	45,000	24
10	45,000	12
11	45,000	1
12	41,500	1
13	39,000	1
14	36,500	1
15	34,000	1
16	31,500	1

TABLE P-1 List of HFEs Available for
Sediment-Triggered Experiments (fall,
extended-duration fall and spring) under the
Preferred Alternative

The modeling component is based on four key analysis phases associated with the two sand budget accounting periods and the two sediment-triggered HFE windows:

- Phase 1: Fall Accounting Period. The fall accounting period is from July 1 to November 30. Beginning on July 1 of each year, monitoring data would be used to track the sand storage from Paria River inputs in Marble Canyon.
- Phase 2: October-November HFE Window. Beginning October 1, sand storage and forecast hydrology would be evaluated using the sediment budget model to determine whether conditions are suitable for an HFE. The model determines what magnitude and duration of the HFE, if any, would produce a positive sand mass balance at the end of the accounting period. If the model produces a positive result, the largest HFE that would result in a positive sand mass balance is forwarded to the decision and implementation component (see below), which also allows for other factors to be considered in the planning process (see Section P.1.1). During the decision process, sediment input would continue to be measured, the model would continue to be run and results or output would be forwarded to decision-makers to allow for refinement of the previously recommended magnitude and duration of the HFE. If the model produces a negative result, the model would be rerun using more recent sediment input to determine whether a positive sand mass balance would be reached in time to have an HFE in the release window.

- Phase 3: Spring Accounting Period. The spring accounting period is December 1 to June 30. As with the fall accounting period, monitoring data would be used to track the sand storage conditions in Marble Canyon during this time period. This accounting would be conducted regardless of whether or not a previous October or November HFE was conducted such that two HFEs could theoretically occur in the same year. The exception to this would be following an extended-duration fall HFE, as there would be no spring HFE following an extended-duration fall HFE. The accounting would continue to consider sand storage conditions present at the end of Phase 2, whether or not an HFE has occurred.
- Phase 4: March-April sediment-triggered HFE Window. The evaluation in this phase is the same for the October-November HFE window (see Phase 2) with the model output being forwarded to the decision and implementation component. The model output would be used in the same way as for the October-November determination. Whether or not a spring HFE is scheduled, sediment inputs would continue to be monitored through the end of the spring accounting period for use in the next accounting period. Note that proactive spring HFEs (see Section P.3.2), which are triggered by water volume and not sediment inputs, could occur in April, May, or June. In addition, spring HFEs would not be tested in years when there had been an extended-duration fall HFE earlier in the same water year.

P.1.3 Decision and Implementation

The third component of the HFE protocol is decision and implementation. This component could span a portion or most of the HFE window, depending on when conditions are deemed suitable for an HFE. The output from the model runs described above is used to determine if sediment and hydrology conditions are suitable for an HFE of a given magnitude and duration. For example, if the scenario that is identified by the model cannot be implemented because of facility limitation (e.g., if one or more turbines are out of service), managers would assess the need to modify the range of magnitude and duration of the HFE. Because this EIS has considered the effects of 45,000 cfs HFEs for 1 to 250 hours, it also serves to analyze the effects of HFEs at lower magnitudes and equivalent durations.

Because the model only considers water and sediment, an added purpose of the decision and implementation component is to consider potential effects on other resources. To determine whether conditions are suitable for implementing or discontinuing HFEs, the DOI will schedule implementation meetings or calls with the DOI bureaus (USGS, NPS, FWS, BIA, and Reclamation), WAPA, AZGFD, and one liaison from each Basin State and from the UCRC, as needed or requested by the participants. The implementation group will strive to develop a consensus recommendation to bring forth to the DOI regarding resource issues as detailed at the beginning of this section as well as including WAPA's assessment of the status of the Basin Fund. The Secretary of the Interior will consider the consensus recommendations of the implementation/planning group, but retains sole discretion to decide how best to accomplish operations and experiments in any given year pursuant to the ROD and other binding obligations.

DOI also will continue separate consultation meetings with the Tribes, AZGFD, the Basin States, and UCRC upon request, or as required under existing RODs.

If the decision is made to conduct an HFE, staff of the USGS's Grand Canyon Monitoring and Research Center (GCMRC) would prepare to conduct monitoring and research in cooperation with other agencies. If not, the process would be repeated during the next accounting window. For each HFE, GCMRC staff would analyze results and integrate information from other HFEs for use in future HFE decisions.

The decision process could result in a sediment triggered HFE being considered whether or not a positive sand mass balance is projected for that release, since the decision must be made in advance of the actual sediment-triggered HFE release and there is an admitted uncertainty in the modeled forecast for both sediment input and dam releases. Caution would be exercised; however, because the sand mass balance only accounts for the difference between inputs and outputs, and does not adequately portray the erosion of sand in the river channel. Successive HFEs or intervening periods of erosion without HFEs could negatively affect the ability of future HFEs to form sandbars and beaches. Furthermore, this erosion could impact other resources and it is advisable to ensure that the net amount of sand in the river channel not be depleted so as to compromise other ecosystem components and functions. The output of the model would be integrated with an assessment of the status and trend of other resources, as an acknowledgement that the decision cannot be focused solely on the condition of the sediment to ensure that the decision encompasses the impacts on the resources identified above in P-1.1.

P.2 OPERATION OF GLEN CANYON DAM UNDER THE HFE PROTOCOL

The scenarios considered below describe how Reclamation would modify the operation of Glen Canyon Dam to reallocate monthly volumes when necessary to achieve HFEs as called for under this protocol. Implementation of the protocol will be done in concert with coordinated river operations. Since 1970, the annual volume of water released from Glen Canyon Dam has been made according to the provision of the Criteria for Coordinated Long-Range Operation of Colorado River Reservoirs (LROC) that includes a minimum objective release of 8.23 maf.

The 2007 Colorado River Interim Guidelines for Lower Basin Shortages and Coordinated Operations for Lake Power and Lake Mead (Reclamation 2007a) for lower basin shortages and the coordinated reservoir operations (Reclamation 2007b) implements relevant provisions of the LROC for an interim period through 2026. This allows Reclamation to modify these operations by allowing for potential annual releases both greater than and less than the minimum objective release under certain conditions. A more thorough description of Reclamation's process for determining and implementing annual release volumes is available in the 2007 EIS and Record of Decision (Reclamation 2007b).

Pursuant to the 2007 Colorado River Interim Guidelines, the annual release volume from Lake Powell is projected and updated each month in response to the monthly 24-month study model run. This projected annual release volume is allocated to produce projected monthly release volumes and becomes the basis for scheduled monthly releases from Glen Canyon Dam. It is important to note that, regardless of the timing of releases, implementation of the HFE protocol would not affect annual release volumes.

HFEs could require more water than what is scheduled for release through the coordinated operation process described above. In order to perform these HFEs, reallocation of monthly releases from Glen Canyon Dam may be necessary. Monthly reallocations for an HFE would not affect annual release volumes.

P.2.1 Potential Operation of Glen Canyon Dam during the Fall HFE Implementation Window

Reclamation would attempt to implement fall HFEs by lowering the remaining days within the fall HFE period to the degree practicable up to as low as allowed under the LROC and LTEMP ROD in order to release the projected October and November volumes in the 24-month study. If a fall high-HFE could be achieved within the release volumes projected for October and November, no reallocation of the monthly volumes from other months would need to be performed.

If, however, a fall HFE could not be achieved within the release volumes projected for October and November, Reclamation would reduce the projected monthly release volumes as necessary through the remainder of the water year. The reallocation would be determined in consultation with WAPA to minimize adverse impacts on hydropower. Reallocation would only be conducted up to the amount necessary to result in the projected monthly volume for October and November being sufficient to conduct the HFE.

P.2.2 Potential Operation of Glen Canyon Dam during the Spring HFE Implementation Window

Reclamation would attempt to achieve spring HFEs by lowering the remaining days within the spring HFE period to the degree practicable up to as low as allowed under the LROC and LTEMP ROD in order to release the projected March and April volumes in the 24-month study. If the sediment-triggered spring HFE could be achieved within the release volumes projected for March and April, no reallocation of the monthly volumes from other months would need to be performed. Note that proactive spring HFEs (see Section P.3.2) would not require reallocation of water outside of the month in which they occurred because they would be 24 hours or less in duration.

If, however, Reclamation determined that it would not be possible to achieve the sediment-triggered HFE within the monthly release volumes projected for March and April, Reclamation would reduce the projected monthly release volumes as necessary through the

remainder of the water year. The reallocation would be determined in consultation with WAPA to minimize adverse impacts on hydropower. The reallocation process would only be conducted up to the amount necessary to result in the projected monthly volume for March and April being sufficient to conduct the sediment-triggered HFE. If additional reallocation of the monthly volumes is required to achieve the sediment-triggered HFE, Reclamation would attempt to do so while maintaining the projected July and August release volumes in the 24-month study.

P.3 HIGH FLOW EXPERIMENTS TO BE EVALUATED UNDER THE PREFERRED ALTERNATIVE

Sediment-related experiments under the preferred alternative include (1) sedimenttriggered spring and fall HFEs up to 96-hr duration; (2) short-duration (24-hr) proactive spring HFEs in high-volume equalization years (>=10 maf) prior to equalization releases; and (3) implementation of up to four extended-duration (>96 hr) HFEs, up to 250 hr long, depending on sediment conditions. The pattern of transferring water volumes from other months to make up the HFE volume would be discussed in the monthly Glen Canyon Dam operational coordination meetings described in Section P.1.

If sediment resources are stable or improving, the combination of base operations, HFEs, and other treatments would continue as prescribed for the preferred alternative. If sediment resource conditions decrease to unacceptable levels during the LTEMP period, operations may be modified to the extent allowable under the LTEMP ROD or would be evaluated and considered under a separate NEPA process, potentially including additional studies of sediment augmentation or other actions.

For all sediment experiments, testing would be modified or temporarily or permanently suspended if (1) experimental treatments were ineffective at accomplishing their objectives, or (2) there were unacceptable adverse impacts on resources. Monitoring results would be evaluated to determine whether additional tests, modification of experimental treatments, or discontinuation of experimental treatments were warranted.

Implementation of HFEs would consider resource condition assessments and resource concerns using the annual processes described in Sections P.1. HFEs may not be tested when there appears to be the potential for unacceptable impacts on the resources listed in Section P.1. In addition, there is uncertainty associated with cumulative impacts from sequential HFEs. These cumulative impacts would be considered before implementing an HFE.

P.3.1 Sediment-Triggered Spring HFEs

Under the preferred alternative, sediment-triggered spring HFEs would be implemented after an initial 2-year delay in order to enable testing of the effectiveness of trout management flows and address concerns raised by the apparent positive response of trout to the 2008 spring HFE (Korman et al. 2011; Melis et al. 2011). After the first 2 years of the LTEMP period, spring HFEs would be implemented when triggered by sediment conditions, except in water years when

an extended-duration fall HFE (see description in Section P.3.4) was conducted. Modeling indicates that there may be sufficient sediment input for spring HFEs in about 26% of the years in the LTEMP period. Sediment-triggered spring HFEs would be implemented when triggered during the entire LTEMP period unless new information indicated they were not effective in building sandbars, or there were unacceptable adverse effects on resources (Section P.1).

Implementation of sediment-triggered spring HFEs would consider resource condition assessments and resource concerns using the processes described in Section P.1. Spring HFEs may not be tested when there appears to be the potential for unacceptable adverse impacts on the resources listed in Section P.1. In addition, there is uncertainty associated with the cumulative impacts of sequential HFEs on sediment, aquatic, and potentially other resources. These cumulative impacts would be considered before implementing a spring HFE particularly if a fall HFE had been implemented in the same water year.

P.3.2 Proactive Spring HFEs

GCMRC scientists identified proactive spring HFEs as a potential experimental treatment to transport and deposit in-channel sand at elevations above those of equalization flows. Proactive spring HFEs would be tested only in years with high annual release volume (i.e., ≥ 10 maf). A first test would be a 24-hr 45,000-cfs release conducted in April, May, or June. Duration in subsequent tests could be shortened depending on the observed effects during the first tests. It would be preferable to test proactive spring HFEs at least two to three times in the 20 year LTEMP period, but being able to do so will be dependent upon annual hydrology. Modeling indicates that proactive spring HFEs would be triggered in about 10% of the years in the LTEMP period.

Proactive spring HFEs would not be tested in the first 2 years of the LTEMP. In addition, proactive spring HFEs would not be tested in years when there had been a sediment-triggered spring HFE or an extended-duration fall HFE earlier in the same water year. Proactive spring HFEs could be performed in the same water year as a 96-hr or shorter sediment-triggered fall HFE, although prior to implementation, the potential effects of these HFEs would be carefully evaluated using the processes described in Section P.1. The first test would be carefully evaluated to determine whether additional tests were warranted based on the efficacy of building and maintaining sandbars. If initial tests show positive results without unacceptable adverse effects on the resources listed in Section P.1, proactive spring HFEs would be implemented when triggered during the entire LTEMP period.

Implementation of proactive spring HFEs would consider resource condition assessments and resource concerns using the processes described in Section P.1. Proactive spring HFEs may not be tested when there appears to be the potential for unacceptable impacts on the resources identified in Section P.1. The cumulative impacts of sequential HFEs would be considered before implementing a proactive spring HFE.

P.3.3 Sediment-Triggered Fall HFEs

Under the preferred alternative, sediment-triggered fall HFEs could be implemented throughout the 20-year LTEMP period unless new information indicated fall HFEs were not effective in building sandbars, or there were unacceptable adverse effects. Modeling indicates fall HFEs would be triggered in about 77% of the years in the LTEMP period.

Implementation of sediment-triggered fall HFEs would consider resource condition assessments and resource concerns using the processes described in Section P.1. Fall HFEs may not be tested when there appears to be the potential for unacceptable impacts on the resources listed in Section P.1. The cumulative impacts of sequential HFEs would be considered before implementing a sediment-triggered fall HFE.

P.3.4 Extended-Duration Fall HFEs

The HFE EA (Reclamation 2011) had a limit of 96-hr duration HFEs at various release levels. Under the preferred alternative, sediment-triggered fall HFEs with durations longer than 96 hr (up to 250 hr) would be tested. The duration of these extended-duration fall HFEs would be based on the amount of sediment delivered from the Paria River during the fall accounting period and would be no more than the maximum magnitude and duration of HFE that would achieve a positive sand mass balance in Marble Canyon, as determined by modeling. Based on examination of the observed historical sediment input from the Paria River, it was determined that HFEs up to 10.4 days in length (250 hr) could be supported before exhausting seasonal sediment inputs and affecting water delivery requirements. GCMRC scientists have suggested that increasing the duration of HFEs when sediment supply can support a longer duration may lead to more sand being deposited at higher elevations, resulting in bigger sandbars. Modeling indicates the sediment trigger for this treatment may be reached in 25% of the years in the LTEMP period. There would be no more than four extended-duration fall HFEs allowed over the 20-year LTEMP period.

The duration of the first implementation of an extended-duration HFE would be limited to no more than 192 hr (twice as long as the 96-hr limit). This duration is considered long enough to produce a measurable result if the treatment represents an effective approach to building sandbars under enriched sediment conditions. The duration of all tests would be based on available sediment, current hydrology, reviews of available information, the expert opinion of GCMRC and other Grand Canyon scientists, and consideration of potential effects on other resources listed in Section P.1. If feasible, monitoring would include real-time observations of sediment concentrations to determine if sediment deposition continues throughout the duration of the extended HFEs.

Implementation of extended-duration fall HFEs would consider resource condition assessments and resource concerns using the processes described in Section P.1. Extendedduration fall HFEs may not be tested when there appears to be the potential for unacceptable impacts on the resources listed in Section P.1. Because the effects of extended-duration HFEs on Lake Mead water quality are a concern, DOI will coordinate with relevant water quality monitoring programs or affected agencies prior to implementing any test of extended-duration HFEs. The cumulative impacts of sequential HFEs would be considered before implementing an extended-duration fall HFE.

Another important concern that results from the large volume of water bypassed during an extended-duration HFE is water delivery. Water delivery issues would be considered before deciding to implement an extended-duration fall HFE. An extended-duration HFE would not be implemented if annual release volume would be affected. It is possible that in lower volume years there would not be sufficient water available to support an extended-duration HFE. A 250-hr extended-duration HFE would result in a monthly total release of approximately 1.2 maf. In lower volume release years (e.g., 7.0 maf or 7.48 maf), the maximum duration may be less than 250 hr. In addition, a sediment-triggered spring HFE or proactive spring HFE would not be conducted in the same water year as an extended-duration fall HFE. If an extended-duration fall HFE was triggered but not implemented for any of the reasons described above, a fall HFE 96 hr or less in duration could be implemented instead. Implementation would necessitate reducing water volume in other months of the same water year.

In order to fully test the efficacy of these longer HFEs, several replicates would be desirable in the 20 year LTEMP period. Extended-duration HFEs would be considered successful and would be continued up to a total of four times in the 20-year LTEMP period as part of an adaptive experimental treatment if there was a widespread increase in bar size relative to \leq 96-hr HFEs, and if sand mass balance was not significantly compromised relative to the ability to maintain a long-term equilibrium. Extended-duration HFEs would not continue to be tested if they were not effective in building sandbars, if resulting total sandbar volumes were no bigger than those created by shorter-duration HFEs, or if unacceptable adverse impacts on resources listed in Section P.1 were observed.

REFERENCES

Grantz, K., and H. Patno, 2010, *Glen Canyon Dam High Flow Protocol Hydrologic Trace Selection and Disaggregation to Hourly Flows*, U.S. Bureau of Reclamation, Upper Colorado Region, Salt Lake City, Utah.

Korman, J., M. Kaplinski, and T.S. Melis, 2011, "Effects of Fluctuating Flows and a Controlled Flood on Incubation Success and Early Survival Rates and Growth of Age-0 Rainbow Trout in a Large Regulated River," *Transactions of the American Fisheries Society* 140:487–505.

Melis, T.S., P.E. Grams, T.A. Kennedy, B.E. Ralston, C.T. Robinson, J.C. Schmidt, L.M. Schmit, R.A. Valdez, and S.A. Wright, 2011, "Three Experimental High-Flow Releases from Glen Canyon Dam, Arizona—Effects on the Downstream Colorado River Ecosystem," Fact Sheet 2011–301, U.S. Geological Survey, Southwest Biological Science Center, Grand Canyon Monitoring and Research Center, Feb. Available at http://pubs.usgs.gov/fs/2011/3012/fs2011-3012.pdf. Accessed Feb. 19, 2015.

Reclamation (Bureau of Reclamation), 1988, *Colorado River Simulation System User's Manual*, U.S. Bureau of Reclamation, Denver, Colorado.

Reclamation, 2007a, *Environmental Impact Statement—Colorado River Interim Guidelines for Lower Basin Shortages and Coordinated Operations for Lake Powell and Lake Mead*, Bureau of Reclamation, Upper and Lower Colorado Region, Oct. Available at http://www.usbr.gov/lc/region/programs/strategies.html. Accessed May 2013.

Reclamation, 2007b, *Record of Decision, Colorado River Interim Guidelines for Lower Basin Shortages and the Coordinated Operations for Lake Powell and Lake Mead*, Bureau of Reclamation, Upper and Lower Colorado Region, Dec. Available at http://www.usbr.gov/lc/region/programs/strategies.html. Accessed May 2013.

Reclamation, 2011, *Environmental Assessment Development and Implementation of a Protocol for High-flow Experimental Releases from Glen Canyon Dam, Arizona, 2011–2020*, Upper Colorado Region, Salt Lake City, Utah, Dec. Available at http://www.usbr.gov/uc/envdocs/ea/gc/HFEProtocol/index.html. Accessed May 2013.

Russell, K., and V. Huang, 2010, *Sediment Analysis for Glen Canyon Dam Environmental Assessment, Upper Colorado Region, AZ*, prepared for Bureau of Reclamation, Salt Lake City, Utah.

Wiele, S.M., and Smith, J.D., 1996, "A Reach-Averaged Model of Diurnal Discharge Wave Propagation down the Colorado River through the Grand Canyon," *Water Resources Research* 32(5):1375–1386.

Wright, S.A., D.J. Topping, D.M. Rubin, and T.S. Melis, 2010, "An Approach for Modeling Sediment Budgets in Supply-Limited Rivers," *Water Resources Res*earch 46(10):W10538. DOI:10.1029/2009WR008600.

This page intentionally left blank

APPENDIX Q:

RESPONSES TO PUBLIC COMMENTS ON THE GLEN CANYON DAM LONG-TERM EXPERIMENTAL AND MANAGEMENT PLAN DRAFT ENVIRONMENTAL IMPACT STATEMENT

This page intentionally left blank

APPENDIX Q:

RESPONSES TO PUBLIC COMMENTS ON THE GLEN CANYON DAM LONG-TERM EXPERIMENTAL AND MANAGEMENT PLAN DRAFT ENVIRONMENTAL IMPACT STATEMENT

The U.S. Department of the Interior (DOI), through the Bureau of Reclamation (Reclamation) and National Park Service (NPS), proposes to develop and implement a Long-Term Experimental and Management Plan (LTEMP) for operations of Glen Canyon Dam. The LTEMP would provide a framework for adaptively managing Glen Canyon Dam operations over the next 20 years, consistent with the Grand Canyon Protection Act of 1992 (GCPA) and other provisions of applicable federal law. The LTEMP would determine specific options for dam operations, non-flow actions, and appropriate experimental and management actions that will meet the GCPA's requirements and minimize impacts on resources within the area impacted by dam operations, including those of importance to American Indian Tribes.

On January 8, 2016, the LTEMP DEIS was filed with Region 9 of the U.S. Environmental Protection Agency (EPA); a Notice of Availability and Notice of Public Meetings were published in the *Federal Register* (81 FR 963); and an email notification of the availability of the DEIS for download from the project website (www.ltempeis.gov) was sent to approximately 600 members of the public who had signed up for notification during the scoping period. Prior to this date, the DEIS was sent to each of the Governors, Senators, and Representatives from relevant congressional districts of the seven Colorado River Basin States (Arizona, California, Colorado, Utah, Nevada, New Mexico, and Wyoming).

In addition to making the DEIS available on the public website, 84 compact disc copies of the DEIS were mailed to individuals at their request; 46 copies were picked up at public meetings held for the DEIS; and copies were made available for public review after the DEIS was published at the following locations:

- J. Willard Marriott Library, University of Utah, 295 South 1500 East, Salt Lake City, Utah 84112.
- Cline Library, Northern Arizona University, 1001 S. Knoles Drive, Flagstaff, Arizona 86011-6022.
- Burton Barr Central Library, 1221 North Central Avenue, Phoenix, Arizona 85004.
- Page Public Library, 479 South Lake Powell Boulevard, Page, Arizona 86040.
- Grand County Library, Moab Branch, 257 East Center Street, Moab, Utah 84532.
- Sunrise Library, 5400 East Harris Avenue, Las Vegas, Nevada 89110.

- Denver Public Library, 10 West 14th Avenue Parkway, Denver, Colorado 80204.
- Natural Resources Library, U.S. Department of the Interior, 1849 C Street NW, Main Interior Building, Washington, D.C. 20240-0001.

The original 90-day public comment period was extended an additional 32 days (122-day total) to May 9, 2016, after several requests were received from the public and Cooperating Agencies. During the comment period, two in-person meetings and two Internet-based webinars were held to provide the public with information about the content and findings of the DEIS and to receive written comments on the DEIS. The meetings and webinars were held on the following dates:

- Webinar—Tuesday, February 16, 2016, at 6:30 p.m. Mountain Standard Time (MST);
- Meeting—Monday, February 22, 2016, at 6:00 p.m. MST, Flagstaff, Arizona;
- Meeting—Thursday, February 25, 2016, at 6:00 p.m. MST, Phoenix, Arizona; and
- Webinar—Tuesday, March 1, 2016, at 1:00 p.m. MST.

At these meetings, LTEMP staff were available to take comments and answer questions before and after presentations were made on the DEIS.

During the public comment period, the public was encouraged to submit comments electronically through the NPS Public, Environment, and Public Comment (PEPC) website. Comments were also received, however, through the mail or using a public comment form provided at the public meetings. More than 3,000 individual comment documents were received on the DEIS.

Comments were determined to be substantive or nonsubstantive in nature. NEPA regulations require that responses be provided to substantive comments. Comments are considered substantive if they:

- Challenge accuracy of analysis
- Dispute information accuracy
- Suggest different viable alternatives
- Provide new information that makes a change in the proposal In other words, they raise, debate, or question a point of fact or policy.

Comments in favor of or against the proposed action or alternatives, or comments that only agree or disagree with co-lead agency policies, are not considered substantive. Substantive comments were summarized and are presented, along with a response, per issue or impact topic in this appendix. Substantive comments were used to make changes to the DEIS when deemed appropriate and justified. Comment issues and the pages on which responses are located are presented in Table Q-1.Commenters, their affiliation (if any), and the section(s) where responses to their substantive comments are located are presented in Table Q-3, in Section 15.

Nonsubstantive comments are comments that offer opinions or provide information not directly related to issues or impact analyses. Nonsubstantive comments have been considered by the joint leads, but do not require a formal response.

PUBLIC COMMENT SUBMITTAL METRICS

Comment documents were received from 3,035 individuals; organizations (including environmental groups and other special interest groups); Tribes, private businesses (including recreational companies) and industry; and local, state, and federal agencies. Submissions were received from all 50 states plus the District of Columbia. About 12% of the submissions were from Arizona. Most submissions were received from California, at 19%. Table Q-2 shows the percentages for the five states with the most comments. Additional states that submitted more than 100 comments included Illinois, Pennsylvania, Texas, and Washington. Over 2,900 submissions were received via the LTEMP public website, and about 30 were received via postal mail. Another 20 were submitted using a comment form distributed at the Public Meetings. One petition included approximately 400 signatures.

FORM LETTERS

Over 1,400 of the 3,035 comment letters received were submitted as form letters that included the same comment language. Ten form letters were received, with submittals numbering: 15, 17, 21, 21, 24, 24, 29, 100, 166 and >1,000. Also, as noted above, one letter, a petition to enhance the Lees Ferry rainbow trout fishery, was signed by 404 individuals. The vast majority of the form letters expressed support for the preferred alternative. The form letter with over 1,000 submittals read as follows:

"Thank you for the opportunity to provide comments on the future of the Colorado River below Glen Canyon Dam. I support the preferred management plan (Alternative D) for Glen Canyon Dam, including experimental high flow releases, for the next 20 years. The Grand Canyon's native fish, recreational experiences, and other natural and cultural resources deserve this best option. It is especially important to keep daily water release fluctuations capped at 8,000 cfs. This will help preserve natural resources that provide habitat for endangered fish and the re-building of beaches along the Colorado River. Thank you for including efforts to restore riparian vegetation, which would otherwise degrade under any of the other alternatives."

TABLE Q-1 Comment Issues and Location in the Report

1	AIR Q	UALITY AND CLIMATE CHANGE	Q-13
	1.1 1.2	Delete "negligible" because it is Subjective Minimize Potential Impacts by Comparing Project-Related GHG Emissions with	Q-13
		Regional Total and US Total	Q-13
	1.3	Consider Effects of HFE on GHG Emissions	Q-13
	1.4	Display Greenhouse Gas (GHG) Emissions Increase in Everyday Terms	Q-13
	1.5	Alternative D increases GHG Emissions, Nearly Four Times the Level Associated with Alternative A	0-14
	1.6	Methane Emissions from Lake Powell that Contribute to Climate Change	Q-14
2			0.15
2	ALIE	KNATIVES	Q-15
	2.1	EIS Does Not Comply with NEPA	Q-15
		2.1.1 The DEIS Did Not Evaluate All Reasonable Alternatives	Q-15
		2.1.2 No Alternative Featured Greatly Increased Hydropower	Q-16
		2.1.3 Experiments are not Adequately Defined or Analyzed	Q-16
	2.2	EIS Does Not Comply with ESA	Q-17
	2.3	EIS Does Not Comply with CRSPA	Q-20
	2.4	Support Alternative D	Q-22
	2.5	Support Alternative D with Caveats	Q-22
	2.6	Support Alternative B	Q-25
	2.7	Support Alternative A	Q-27
	2.8	Support Alternatives Considered but Rejected (Fill Mead First, Decommission Dam)	Q-28
	2.9	EIS Should Have Considered Infrastructure Additions/Modifications	Q-30
	2.10	Alternative D ignores (or does not meet) Hydropower Objective	Q-31
	2.11	EIS Did Not Consider a Full Range of Alternatives	Q-32
	2.12	Flows Should Not Drop Below 8,000 cfs	Q-33
	2.13	8,000 CFS Fluctuation Limit	Q-34
	2.14	EIS Did Not Adequately Consider the Effects of Climate Change	Q-35
	2.15	Geographic Scope of the EIS	Q-36
	2.16	Monitoring to Support Adaptive Management	Q-37
	2.17	Include a Summary of Experimental and Management Actions for the Rainbow Trout	0-37
	2 18	Glen Canyon Dam Safety and Security	Q=37
	2.10	Experimental Uncertainty	Q=37
	2.17	Impacts on Basin Fund	Q_{-38}
	2.20	Due Consideration of Hydronower	Q 30 Q-39
	2.21	Support Alternative F	0-39
-	<u> </u>		× 57
3	AQUA	TIC ECOLOGY	Q-40
	3.1	Alternatives Adequate Will Not Protect Colorado River Ecosystem	Q-40
	3.2	Impacts of Fall HFEs on Fish and Aquatic Food Base	Q-40

	3.1	None of the Alternatives Adequate to Change and Protect Colorado River Ecosystem	Q-40
	3.2	Impacts of Fall HFEs on Fish and Aquatic Food Base	Q-40
	3.3	Reduce Number of HFEs to Improve the Aquatic Food Base	Q-41
	3.4	Impacts of Minimum Flows Before and After HFEs	Q-41
	3.5	Provide Additional Information on Most Recent Fall HFEs	Q-42
	3.6	Support Careful Testing of TMFs	Q-42
	3.7	Quagga Mussels	Q-43
	3.8	Provide Additional/Updated Information on Aquatic Resources and Species	Q-43
	3.9	AZGFD Management Plan	Q-44
4	CULT	URAL RESOURCES	Q-45
	4.1	Applicability and Composition of Cultural Resource Metrics	Q-45
		4.1.1 Wind Transport	Q-45
		4.1.2 Time Off River	Q-45
		4.1.3 Glen Canyon Flow Effect Metric	Q-46
	4.2	Area of Potential Effects	Q-46
	4.3	Landmark Status for Grand Canyon and its Resources	Q-46
5	CUMU	ILATIVE IMPACTS	Q-47
	5.1	Additional Foreseeable Actions (Including Management Plans) to be Included	Q-47
	5.2	Refinement of Description and/or Impact Assessment for Foreseeable Actions	
		Considered	Q-47
6	HYDR	OPOWER	Q-48
	6.1	Scope of Analysis and Study Elements	Q-48
		6.1.1 Hydropower Goals and Objectives	Q-48
		6.1.2 Generation at Hoover Dam	Q-48
		6.1.3 Cumulative Impacts on Power Systems	Q-48
		6.1.4 Analysis of Impacts on Basin Fund	Q-48
		6.1.5 Consideration of Impacts on Basin Fund	Q-49
		6.1.6 Power System Impacts on Climate Change	Q-49
		6.1.7 Sensitivity Analysis of Fuel Prices	Q-49
		6.1.8 Clean Powerplant Legislation	Q-49
	6.2	General Overarching Comments on the Power Systems Analysis	Q-50
		6.2.1 Adequacy of Power Systems Analysis	Q-50
		6.2.2 Ratepayer Analysis Flawed	Q-50
	6.0	6.2.3 Inputs to Capital Costs Analysis	Q-51
	6.3	Legal Issues	Q-52
		6.3.1 HFEs as Part of Long-Term Operations	Q-52
	<i>с</i> н	6.3.2 Irretrievable Loss of Hydropower	Q-52
	6.4	Technical Power System Issues	Q-52
		6.4.1 Discount Rate	Q-52
		6.4.2 Capital Costs	Q-52
		6.4.3 Exceedance Level	Q-53
		6.4.4 Ancillary Service Assumptions	Q-53
		6.4.5 Modeling Assumptions	Q-54

		6.4.6 Small Customer Representation	Q-54
		6.4.7 Small Customer Loads and Load Growth	Q-54
		6.4.8 System Expansion Plan	Q-54
		6.4.9 Transmission Constraints	Q-55
		6.4.10 Renewable Portfolio Standards	Q-55
		6.4.11 Costs of HFEs	Q-56
		6.4.12 Costs of Experiments	Q-56
		6.4.13 Impacts of Macroinvertebrate Production Flows on Hydropower	Q-56
		6.4.14 Effects at Hoover Dam and Glen Canyon Dam	Q-56
	6.5	Retail Rate Analysis	Q-57
		6.5.1 Consideration of Wholesale Rate Input	Q-57
	6.6	Wholesale Rate Analysis	Q-57
		6.6.1 Assumptions	Q-57
	6.7	Editorial Comments	Q-57
7	NATU	RAL PROCESSES	Q-58
	7.1	Naturally Patterned Flow and Run-of-the-River Flow Alternatives	Q-58
	7.2	Remove Natural Processes from the EIS	Q-58
	7.3	Natural Resource Performance Goal	Q-58
		7.3.1 Quantitative Performance Metric	Q-58
		7.3.2 Goal to Restore Is Not Achievable	Q-58
		7.3.3 National Park Service Natural Processes Management Policy and Goals	Q-59
	7.4	Opportunity to Expand Non-Flow Management Options to Improve Aquatic and	
		Riparian Habitat	Q-59
	7.5	HFEs Should be Included in the Evaluation of Effects to Natural Processes	Q-60
8	PROCE	ESS AND POLICY	Q-61
	8.1	Include THPOs in Contact List	Q-61
	8.2	Tribes Involvement in Alternatives Development	Q-61
	8.3	Sierra Club's Scoping Comments	Q-61
	8.4	Involvement of Cooperating Agencies and SME Teams	Q-68
	8.5	LTEMP Effects on Reservoir Operations	Q-69
	8.6	List Relevant Laws and Management Documents	Q-69
	8.7	Application of Structured Decision Analysis (SDA)	Q-69
	8.8	DOI Disclaimer	Q-71
	8.9	DOI Response to Comments	Q-71
	8.10	Comment Period Deadline	Q-71
	8.11	Language Regarding LROC	Q-/1
	8.12	Description of HFE Protocol and Nonnative Fish Control Protocol for the Preferred	
		A ltown active	0.72
	8.13	Alternative	Q-72 Q-72
9	8.13 RECRE	Alternative Conservation measures to Be Applied Under Alternative D	Q-72 Q-72 Q-73
9	8.13 RECRE	Alternative Conservation measures to Be Applied Under Alternative D EATION	Q-72 Q-72 Q-73
9	8.13 RECRE 9.1	Alternative Conservation measures to Be Applied Under Alternative D EATION Note Economic Benefits of the Recreation Industry	Q-72 Q-72 Q-73 Q-73
9	8.13 RECRI 9.1 9.2	Alternative Conservation measures to Be Applied Under Alternative D EATION Note Economic Benefits of the Recreation Industry Recreation Resource Goals Incomplete	Q-72 Q-72 Q-73 Q-73 Q-73 Q-73

	9.4	Adoption of MLFF Flows Corresponded with Degradation of the Lees Ferry Fishery Drop in Visitations	Q-74
10	TRIBA	L RESOURCES	Q-75
	10.1 10.2	Opposition to Mechanical Removal of Trout and Trout Management Flows Lack of Tribal Representation in Development of EIS	Q-75 Q-75
	10.3 10.4 10.5	Cultural Landscapes and Traditional Cultural Properties not Thoroughly Discussed Tribal Lands and Resources on Tribal Lands Not Taken Into Consideration	Q-76 Q-77
	10.6 10.7 10.8	Impacts to Hualapai Tourism Operations Not Thoroughly Analyzed Tribal Perspectives on Components of the Colorado River Ecosystem Editorial Comments	Q-77 Q-77 Q-78
11	SEDIM	IENT.	Q-79
	11.1 11.2 11.3	Concern Over Erosion of Sandbars and Beaches following High Flows (i.e., HFEs) Spring HFEs—Uncertainty and Concern Over Impacts; More Testing Is Needed Equalization Flows—Concern Over Increased Erosion and Higher Rate of seDiment Transport	Q-79 Q-79 Q-79
	11.4 11.5	Sediment Should Be Considered a "secondary" Resource Concern Over Sediment Accumulation in Lake Powell and Behind the Dam	Q-80 Q-80
12	SOCIO	ECONOMICS AND ENVIRONMENTAL JUSTICE	Q-81
	12.1 12.2	Regional Economic Impacts Associated with Boating Visitor Expenditures Regional Economic Impacts Associated with Angler Visitor Expenditures, in	Q-81
	12.3 12.4	Use Values Associated with Angling and Boating Trips Non-Use Values Associated with Fishing Resources	Q-81 Q-81 Q-81
	12.5 12.6	Economic Impacts on Tribes Impact on Water Provision Pagional Electric Power Economic Impacts	Q-82 Q-82
	12.7 12.8 12.9	Economic Impacts of Activities of Confluence Partners Out of Scope of EIS Depth of Literature review on Use Valuation in Appendix I	Q-82 Q-82 Q-83
	12.10	Mechanical Removal and Trout Management Flows	Q-83
13	WATE	R	Q-84
	13.1	Consideration of Climate Change Effects Related to Water Supply Reductions, Evaporation, and Operational Levels of Lake Powell	Q-84
	13.3	and Greater Depletion	Q-84
	13.4	Levels Editorial Comments	Q-84 Q-85
14	WILDI	LIFE	Q-86
	14.1 14.2	Averting the Responsibility to Protect Native Species HFE IMPACTS on Kanab Ambersnail	Q-86 Q-86

14.3	Operate Glen Canyon Dam in a Way that Protects All Wildlife (Including Listed	
	Species)	Q-87

by State			
State	Percentage	-	
AZ	12		
CA	19		
CO	5		
FL	5		
NY	5		

53

TABLE Q-2 Commenter Distribution

The form letter with 166 submittals and the petition with 404 signatures had nearly identical language, which recommended that the LTEMP should:

All others

- Explicitly recognize that the Lees Ferry tailwater trout fishery as priority resource "value" for which Glen Canyon Dam should be operated as provided for in the Grand Canyon Protection Act.
- Incorporate the goals and objectives of the Arizona Game and Fish Department's Fishery Management Plan for the Lee Ferry Trout Fishery as stated goals and objectives of the LTEMP.
- Fully consider the impacts of repeated fall high flow experiments on the aquatic food base, rainbow trout, and invasive species in Glen and Marble Canyons.
- Place more emphasis on conducting high flow experiments in the spring to benefit a variety of resources besides sediment/sandbars including the aquatic food base, native fish, rainbow trout and riparian vegetation.
- Test the use of sustained low steady flows (also known as "macroinvertebrate production flows") to increase the production and diversity of the aquatic insects in the Colorado River below Glen Canyon Dam.
- Carefully test the use of Trout Management Flows (TMF), but only when there is an identified need to reduce the number of young of the year rainbow trout in Lee Ferry.

The form letter with 100 submittals simply endorsed "DOI's plan to operate the Glen Canyon Dam in a way that will best protect Grand Canyon National Park." Conversely, the form letter with 29 submittals stated that the "DEIS fails to adequately analyze the impacts of climate change on the entire Colorado River water supply system" and "should have considered a wider range of reasonable alternatives that fully take climate change into account." The remaining form letters were variations on the themes of protecting the Lees Ferry trout fishery or endorsing DOI's plan. Substantive issues raised in the form letters and in individual letters submitted are addressed in the following sections.

SUMMARY OF ISSUES RAISED IN COMMENTS ON THE DEIS AND DOI'S RESPONSES

Comment letters received on the public draft of the LTEMP EIS raised substantive issues in a number of technical areas related to the analyses presented in the EIS and in non-technical areas related to the requirements of preparing the EIS under NEPA. Comments ranged from support of the preferred alternative as is, to support with qualifications, to support of other alternatives. Some commenters disagreed with the exclusion of some proposed alternatives as not meeting the purpose and need of the proposed action, including decommissioning of the dam. Other comments questioned the adequacy of the EIS under NEPA and ESA and in addressing the effects of climate change. Many comments addressed the proposed experiments that are a part of the preferred alternative and other alternatives, their triggers, their potential effectiveness, and their potential effects on other resources not targeted by the experiments. Other comments addressed the modeling conducted to analyze effects of dam operations on resources and on the metrics produced by the models to assess impacts. A number of comments concerned the effects of daily fluctuations and minimum river flows downstream of the dam resulting from dam operations under the various alternatives. Many comments addressed the use of HFEs, their timing, potential benefits and effects on hydropower generation. These and other substantive issues are outlined in the following section. Following the outline is a section that presents a summary of each issue as raised in comments and DOI's response. Finally, an index of all comment letters submitted, the names of the commenters, and their affiliation is presented following the issues and responses section.

Comment Categories, Comment Issues, and Location Where Responses are Found

Comments issues were categorized according to a number of technical and non-technical areas based on the substantive content of the letters submitted. In addition, a large number of editorial comments were received, and we have updated the text where appropriate in response to these comments. These comments refer to specific page numbers and line numbers in the DEIS. Because the overall document pagination has changed from the DEIS to the FEIS, including the specific location of each change would have limited utility and be cumbersome. Accordingly, we have not included the specific locations of each change in this Appendix. The EIS was revised appropriately to address all substantive editorial comments, to correct typographical errors, and to add specific language submitted by Tribes.

The following outline presents the comment categories, a brief description of the issues raised, and the location in the following section where each issue is summarized and DOI's response is presented.

LTEMP CONSOLIDATED ISSUES AND RESPONSES

1 AIR QUALITY AND CLIMATE CHANGE

1.1 DELETE "NEGLIGIBLE" BECAUSE IT IS SUBJECTIVE.

<u>Summary Comment</u>: Commenters expressed that the term "negligible" is a biased, subjective term.

Response: For emissions of SO2 and NOx presented in Section 4.15, the term "negligible" was used to describe impacts, but the actual projected emissions (in tons/yr) and percentage change relative to Alternative A also were provided. No change was made to the text. In Section 4.16, the text was edited to remove the term negligible and only the actual projected emissions of greenhouse gases (in tons/yr) and percentage change relative to Alternative A, and current conditions in the 11-state region and the United States were provided.

1.2 MINIMIZE POTENTIAL IMPACTS BY COMPARING PROJECT-RELATED GHG EMISSIONS WITH REGIONAL TOTAL AND US TOTAL

<u>Summary Comment</u>: Commenters expressed that the effects of increases in GHG emissions for the alternatives should be presented on a local or regional basis, and not at a national level.

<u>Response</u>: Text was updated to show project-related GHG emissions compared to total regional (11 states) while still retaining U.S. GHG emissions as that level is a conventional way of assessing their impacts.

1.3 CONSIDER EFFECTS OF HFE ON GHG EMISSIONS

<u>Summary Comment</u>: Commenters suggested that strong consideration should be given to the air emission impacts resulting from HFEs.

<u>Response</u>: The analysis considered the effects of HFEs on air quality in Section 4.15 and on greenhouse gas emissions in Section 4.16. The modeling results presented in these sections included HFEs in all of the alternatives.

1.4 DISPLAY GREENHOUSE GAS (GHG) EMISSIONS INCREASE IN EVERYDAY TERMS

<u>Summary Comment</u>: Commenters recommended that increases in GHG emissions be expressed as an equivalent mass of coal burned using the EPA's "Greenhouse Gas Equivalencies Calculator," rather than just presenting GHG emissions in metric tons. <u>Response</u>: In Table 4.16-1, we added a footnote which states, "Using an online tool from the EPA (https://www.epa.gov/energy/greenhouse-gas-equivalencies-calculator), one can express a given amount of GHG emissions in MT in everyday terms. For example, 1 million MT/yr is estimated to be equivalent to the amount of CO_2 that is emitted as a result of the electricity use of 148,000 households. However, because the EPA cautions that these estimates are approximate and should not be used for emission inventory or formal carbon footprinting exercises" the authors felt it was not appropriate to report the results in this document using that conversion.

1.5 ALTERNATIVE D INCREASES GHG EMISSIONS, NEARLY FOUR TIMES THE LEVEL ASSOCIATED WITH ALTERNATIVE A

<u>Summary Comment</u>: Commenters noted that Alternative D, which averages more than 1 HFE per year for the 20-year DEIS period, increases GHG emissions by 22,908 metric tons per year, nearly four times the level associated with Alternative A, which is the lowest GHG-producing Alternative due to the lowest frequency of HFEs.

<u>Response</u>: While the commenter is correct that Alternative D increases GHG by 22,908 metric tons per year (MT/yr), the EIS analysis found that the level of emissions for Alternative A was 55,177,668 MT/yr and the level of emissions for Alternative D was 55,200,576 Mt/yr and so the difference from Alternative A would be only 0.042%.

1.6 METHANE EMISSIONS FROM LAKE POWELL THAT CONTRIBUTE TO CLIMATE CHANGE

<u>Summary Comment</u>: Commenters state that preliminary estimates of methane emissions from the operation of Glen Canyon Dam and Lake Powell indicate that the combined life cycle GHG equivalents may be as high as one-third of a natural gas power plant.

<u>Response</u>: Because no studies on GHG emissions from Lake Powell and Grand Canyon Dam have been made, GHG emissions, such as CO₂, CH₄, and N₂O, from the reservoirs were discussed qualitatively in Section 4.16.2.1. Reservoirs such as Lake Powell would be expected to produce some amount of GHG emissions consistent with levels reported for reservoirs in the semiarid Western U.S. (Tremblay et al. 2004 and http://www.ipccnggip.iges.or.jp/public/2006gl/pdf/4_Volume4/V4_p_Ap3_WetlandsCH4.pdf). However, GHG emissions from these sources and associated climate change are not anticipated to be different among the alternatives.

2 ALTERNATIVES

2.1 EIS DOES NOT COMPLY WITH NEPA

2.1.1 The DEIS Did Not Evaluate All Reasonable Alternatives

<u>Summary Comment</u>: Citing 40 CFR 1502.14 requiring agencies to "[r]igorously explore and objectively evaluate all reasonable alternatives, and for alternatives which were eliminated from detailed study briefly discuss the reasons for their having been eliminated." Commenters stated that the LTEMP EIS process does not appear to comply with NEPA because Reclamation and NPS (i) did not objectively evaluate all reasonable alternatives, (ii) failed to assess mitigation measures regarding sediment issues, and (iii) appeared to utilize a partial decision making process.

<u>Response</u>: The LTEMP EIS carried forward a range of reasonable alternatives that were developed based on the input of DOI experts, Cooperating Agencies, stakeholders, Tribes and the public. Early alternative concepts were provided to cooperators and stakeholders for input and a public meeting was held on in April of 2012 to further explore and characterize those concepts. Mitigation measures within the scope of this EIS that fit the purpose and need were considered. The decision making process for this EIS conforms with the established NEPA process and considered input from the public, cooperators, stakeholders and Tribes. This NEPA document provides detailed analysis for a broad range of alternatives ranging from those proposed for ecological purposes, and those proposed by hydropower interests and focused on hydropower generation to, and those proposed to strike different a balances among all resource areas.

Alternatives F and G, the seasonally adjusted steady flow and year round steady flow alternatives were proposed by members of the public and interest groups during scoping. These alternatives were similar to ones considered in the 1996 EIS as well. These alternatives considered as part of a range of reasonable alternatives and these showed many positive and negative impacts on a variety of resources as were disclosed in the draft impact statement. The preferred alternative showed a better balance between resources to achieve the purpose and need.

Full powerplant capacity operations were considered and eliminated as an alternative as discussed in Section 2.3.10 because fluctuation levels to the degree proposed would have adverse impacts on many downstream resources and would not meet the purpose and need for this EIS. Alternative B was originally crafted and submitted by CREDA, a power utility stakeholder. It has a greatly increased daily range compared to the No Action alternative and it includes an experimental implementation of maximum powerplant capacity releases. This alternative showed a number of positive and negative impacts on a variety of resources as were disclosed in the DEIS. The preferred alternative showed a better balance among resources to achieve the purpose and need. [519951]

DOI believes the EIS is in full compliance with NEPA regulations. The USEPA reviewed the DEIS and determined that, based on their review, rated the preferred alternative as "Lack of Objections-Adequate," the highest rating possible.

2.1.2 No Alternative Featured Greatly Increased Hydropower

<u>Summary Comment</u>: Commenters stated that, none of the alternatives feature greatly increased (hydropower) flexibility and that the Agencies should provide more explanation regarding why an alternative with even more daily flexibility was not considered. Others stated that the LTEMP EIS preferred alternative does not meet the renewable resource goal of NEPA Section 101 (b) 6 or the hydropower resource goal established by the DOI for the LTEMP EIS process due to restrictions on the utilization of Glen Canyon Dam for hydropower production under the preferred alternative

<u>Response</u>: Full powerplant capacity operations were considered and eliminated as an alternative as discussed in Section 2.3.10 because fluctuation levels to the degree proposed would have adverse impacts on many downstream resources and would not meet the purpose and need for this EIS. Alternative B was originally crafted and submitted by CREDA, a power utility stakeholder. It has a greatly increased daily range compared to the No Action alternative and it includes an experimental implementation of maximum powerplant capacity releases. This alternative showed a number of positive and negative impacts on a variety of resources as were disclosed in the DEIS. The preferred alternative showed a better balance between resources to achieve the purpose and need.

The analysis performed in this EIS is fully consistent with both the hydropower resource goal in Section 1.4 and the goal set forth in the "policies and goals" subchapter of NEPA, including the cited goal in Section 101(b)(6). Hydropower impacts and consistency with the hydropower resource goal are discussed in Section 4.13. Regarding Section 101(b)(6) of NEPA, this is a general statutory statement that Congress that cannot be read in isolation. Instead, it must be read in connection with the Law of the River, the GCPA, and other statutory provisions specific to operations of Glen Canyon Dam. When these authorities are read together, Congress has established specific requirements for the operation of Glen Canyon Dam. This EIS considers those specific requirements and identifies a preferred alternative that meets the purpose and need and achieves an appropriate balance for meeting the objectives

2.1.3 Experiments are not Adequately Defined or Analyzed

<u>Summary Comment</u>: One commenter stated that Alternative D violates the requirements of NEPA because it contains experiments or other proposed actions that are not adequately defined or analyzed. Another commenter stated that "poorly defined experimental conditions, coupled with the unquantified discretion of the DOI to determine whether there are "adverse effects" on "other resources" renders the EIS legally inadequate because it does not contain the required "hard look" at the environmental and other impacts of the proposed action."

<u>Response</u>: The co-lead agencies feel that this NEPA process takes a comprehensive "hard look" at the proposed action and alternatives to determine whether there are "adverse effects" on "other resources" and is fully consistent with NEPA. As stated in Section 2.2.4.3, This EIS analyzed a wide range of conditions that could occur and states the potential effects in the document.

DOI disagrees with the statement that experiments or other proposed actions are not adequately defined or analyzed. Extensive detail on the costs of experiments appears in Appendix K and text was added to Section 4.13.2.3 to more fully describe effects to hydropower from all experiments. The LTEMP EIS describes experiments to be conducted under different alternatives in sufficient detail to allow for a full analysis of their effects. Extensive modeling of the effects of these alternatives on a full set of resources was conducted and formed the basis of the assessment presented in the EIS. These modeling results were supplemented with qualitative assessments of effects based on existing literature and observations. An assessment of the cumulative impacts of LTEMP alternatives in combination with other past, present, and reasonably foreseeable future impacts is presented in Section 4.17 of the EIS. GCMRC has developed a science plan that describes the information that will be gathered during implementation of the LTEMP, and how that information would be used in an adaptive management context.

The adaptive management process and triggers are adequately addressed in this NEPA document. Section 4.1 and Appendices C and K of the LTEMP DEIS have detailed information on the extensive modeling that was performed with a variety of long term strategies and referring to uncertainties explicitly. The range of adaptations are also well defined and analyzed in the EIS. Text was added between the DEIS and the FEIS to further define this and state specifically under each experiment how magnitudes, duration or frequencies of components of the experiments may change. The long term strategies were variations of the alternatives that turned various experiments "off" or "on" to model different combinations and frequencies of experiments under a variety of hydrologic inflow conditions. This extensive modeling was coupled with extensive literature review and subject matter expert input to provide thorough analysis. This is sufficient for a NEPA analysis. Additional details of experiments and monitoring will be developed through the AMP annual work plans.

DOI believes the EIS is in full compliance with NEPA regulations. The USEPA reviewed the DEIS and determined that, based on their review, rated the preferred alternative as "Lack of Objections-Adequate," the highest rating possible

2.2 EIS DOES NOT COMPLY WITH ESA

<u>Summary Comment</u>: Commenters stated that the LTEMP EIS does not comply with the Endangered Species Act because considered experiments, operations or management actions are quite likely to result in negative impacts to an endangered species.

(1) In addition, DOI should seek input from AMWG and TWG representatives on the draft Biological Assessment under the ongoing Species Status Assessment for HBC, which could improve and refine experimental treatments in the Preferred Alternative and clarify triggers, off ramps, monitoring and definitions of success.

- (2) Further, Alternative D is incompatible with the ESA, in part because there are no binding obligations or criteria or required mitigation that protect the endangered HBC.
- (3) Because the DEIS fails to define "annual implementation considerations" and "off-ramps," so nearly every experimental treatment will be undertaken unless there are "potential unacceptable impacts" on certain defined and undefined resources.
- (4) A commenter stated that the failure of the DEIS to include and consider the cumulative impacts all of the HFEs, including the effects on rainbow trout recruitment and humpback chub from the 2013 and 2014 HFEs, represents a significant analytical flaw, and that it is inappropriate to consider 250 hour long HFEs due to erosion concerns.
- (5) A number of people commented that the EIS did not consider or should have considered more the restoration of extirpated species and recovery implementation plans for all species known to be native to Grand Canyon prior to the operation of Glen Canyon Dam.
- (6) With respect to proposed low summer flows, one commenter supports convening a panel of experts to analyze existing science on low summer flows within the first 5 years of the LTEMP period.
- (7) With respect to sustained low flows for benthic invertebrate production, a commenter requested a comprehensive review of these flows by a scientific panel during the first several years of the LTEMP period to determine their cost, efficacy, and impact on native and nonnative fish.

<u>Response</u>: DOI has consulted with the U.S. Fish and Wildlife Service and developed conservation measures and alternative-specific experimental actions to avoid, minimize, and mitigate adverse impacts on endangered species and critical habitat potentially affected by the proposed action. The Biological Assessment is found in Appendix O in the final EIS. [519929] DOI has been consulting with FWS throughout the 5-year EIS process. The LTEMP EIS used the best available science and peer-reviewed modeling in its analysis of the effects of the proposed action and alternatives on endangered species and critical habitat.

(1) In developing the Biological Assessment for LTEMP, Reclamation convened an ad hoc group of Grand Canyon aquatic biologists from FWS, GCMRC, AZGFD, NPS, and Reclamation to develop the conservation measures, experimental actions, and trigger levels for those actions to be included in the EIS. GCMRC scientists have been involved since scoping with development of LTEMP alternatives and have provided information, research results, and expert opinions regarding the effects of the proposed action on endangered species and critical habitat throughout the EIS process. Also several Cooperating Agencies, including Tribes and Western Area Power Administration were given a chance to review the Biological Assessment and the proposed conservation measures were presented to the AMWG members at the August 2016 AMWG meeting. The Biological Assessment was updated to incorporate feedback and input resulting from these reviews.

The FWS humpback chub species status assessment that the commenter refers to is in process and will not be completed until well after the publication of the LTEMP FEIS. It is
not required to be complete for the FEIS or the record of decision, and is therefore independent of this NEPA process.

Greater specificity regarding triggers for HFE implementation and humpback chub actions have been added to the EIS (Appendices O and P).

(2) Formal consultation with FWS has resulted in a Biological Assessment and Biological Opinion on the effects of the proposed action on listed species and designated critical habitat, as presented in Appendix O of the EIS. DOI has worked closely with FWS throughout the 5-year EIS process to ensure that the appropriate experiments, dam operations, and non-flow actions were identified in the proposed action or as conservation measures. LTEMP EIS alternatives include experiments that are intended to improve conditions for ESA-listed species and other special status species. The LTEMP EIS team used the best available science and peer-reviewed modeling to determine the potential effects of these alternatives on these species.

Throughout the LTEMP DEIS drafting process, the parties to the DEIS have consistently prioritized the conservation of humpback chub and its habitat. FWS, the expert agency on endangered species issues, has been involved in the DEIS process from its inception. GCMRC scientists, who specialize in studying HBC and its habitat, have also provided valuable input regarding HBC during this process.

- (3) The LTEMP EIS did consider and define "annual implementation considerations" and "off-ramps" for experiments. The implementation framework for Alternative D (Section 2.2.4.3 of the EIS) explicitly calls for the consideration of key resources (including endangered and threatened species and their designated critical habitat) when determining which experiments to conduct in any given year. The framework also establishes off-ramps that would identify when experiments should no longer be conducted if it was established there were potential unacceptable adverse impacts to endangered species and critical habitat. As described in Section 2.2.4.4, DOI will also exercise a formal process of stakeholder engagement to ensure decisions are made with sufficient information regarding the condition and potential effects on important resources. GCMRC has developed a science plan that describes the information that will be gathered during implementation of the LTEMP, and how that information would be used in an adaptive management context in order to promote the conservation of humpback chub and other species of concern.
- (4) The LTEMP EIS did fully consider direct, indirect and cumulative effects of HFEs. Descriptions of the observed effects of HFEs conducted in 2012, 2013, and 2014 have been added to the EIS. The impacts of HFEs on other resources, including endangered species and hydropower, were fully analyzed in the EIS. The EIS acknowledges that experimental testing may provide new information about the effects of extended-duration fall HFEs on endangered species and their habitat. If that experimentation results in significant new information about potentially adverse impacts, outside of what was analyzed in this NEPA process then additional NEPA reviews would be completed as appropriate. These HFEs would only be conducted in years of very large sand input from the Paria River during the fall accounting period and the duration would be adjusted to be compatible with the magnitude of the sand input. Extended-duration fall HFEs would only be allowed in 4 of the

20 years of the LTEMP period and spring HFEs would not be allowed in the same water year to mitigate the cumulative impact of sequential HFEs. In addition, consideration would be given to their effects on key resources including water delivery, the Basin Fund, endangered species, and sediment.

(5) The reintroduction of extirpated species was outside the scope of this EIS and is discussed in Section 1.5.3.

LTEMP EIS alternatives include experiments that are intended to improve conditions for ESA-listed species and other special status species. The LTEMP EIS team used the best available science and peer-reviewed modeling to determine the potential effects of these alternatives on these species. DOI has worked closely with FWS throughout the 5-year EIS process to ensure that the appropriate experiments, dam operations, and non-flow actions were identified as conservation measures. Formal consultation with FWS has resulted in a Biological Assessment and Biological Opinion on the effects of the proposed action on listed species and designated critical habitat, as presented in Appendix O of the EIS.

The original notice of intent to prepare the LTEMP EIS identified the need to determine whether to establish a recovery implementation program for endangered fish species below Glen Canyon Dam. The LTEMP team found that identifying the need to determine whether to establish a recovery implementation program (RIP) for endangered fish species below Glen Canyon Dam did not meet the purpose and need for the action.

- (6) Additional description of the implementation process for low summer flows has been added to the EIS (Section 2.2.4.6) including convening a scientific panel that includes independent experts prior to the first potential use of low summer flows to synthesize the best available scientific information related to low summer flows. Based on input from GCMRC, FWS, and joint-lead experts, we feel the additional restriction of only implementing the first test of low summer flows if humpback chub population status is robust is not needed because any test of low summer flows would consider the possibility for unacceptable adverse impacts to endangered species and critical habitat, and implementation of such a test may be delayed until conditions were suitable and unacceptable adverse impacts were minimized.
- (7) We acknowledge that testing may be needed to determine the efficacy of macroinvertebrate production flows in this ecosystem, but find there is sufficient evidence for testing this tool experimentally. We acknowledge the potential for macroinvertebrate production flows to lead to increases in trout numbers. As described in Section 2.2.4.6, the "effects of the test would be evaluated..." and "...could be discontinued if there were unacceptable effects on other resources." Furthermore, we have consulted with FWS and GCMRC, and they both regard the experiment as low risk to the humpback chub population.

2.3 EIS DOES NOT COMPLY WITH CRSPA

<u>Summary Comment</u>: Several commenters have stated that the DEIS has improperly excluded hydropower enhancing alternatives and that the preferred alternative runs counter to the Grand Canyon Protection Act, the implementation of which is subject to and must be consistent with

Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

Colorado River Storage Project Act (CRSPA). Under Section 7 of CRSPA, hydroelectric power plants are under CRSP A are to be operated "so as to produce the greatest practicable amount of power and energy that can be sold at firm power and energy rates." Alternative B outperforms Alternative D in this regard, while Alternative D provides no experiments to benefit hydropower, which would benefit HBC. Commenters state that the DEIS considers only the water management aspects of CRSPA and disregards the hydropower enhancement aspects and that the GCPA and CRSPA must be considered together, giving full consideration to each and that hydropower enhancement was disregarded because it did not conform to the agencies preconceived notion of where the program was to go. Finally, one commenter stated that the LTEMP is deficient as a management program because ignores the legal responsibility to explore mitigation strategies that would allow the use of the dam for its hydropower generating purposes as originally intended and that mitigation strategies are available that will benefit the trout fishery, reduce mechanical removals, stabilize sediment, and enhance river running while providing hydropower in a load following context. One commenter stated that environmental studies thus far completed and proposed do not support disruptions to continued dam management for water and power with either only negligible or zero species improvements and in the absence of reliably identifiable cause/effect and resulting benefit correlations, they urged prioritization of CRSP hydroelectric power purposes for the Glen Canyon Dam operations.

<u>Response</u>: Contrary to statements in the comment, the LTEMP EIS did evaluate and present the results for an alternative that increased hydropower generation. Alternative B was originally crafted and submitted by CREDA, a power utility stakeholder. It has a greatly increased daily range compared to the No Action Alternative and it includes an experimental implementation of maximum powerplant capacity releases. This alternative showed a number of positive and negative impacts on a variety of resources as were disclosed in the EIS. The preferred alternative showed a better balance between resources to achieve the purpose and need of the proposed action.

DOI formulated the characteristics of the preferred alternative based on initial modeling of a set of 6 alternatives. There were several characteristics that were chosen to specifically reduce cost to the hydropower resource including increasing the fluctuation factor from 7 to 10 in high demand months, adjusting the monthly release pattern to match WAPA's Contract Rate of Delivery, increasing the volume of water in the high demand month of August from 750 to 800 kaf in 8.23 maf years, and eliminating steady flows after HFEs. Many objectives and resource goals were considered in the LTEMP EIS. No one alternative was found which met all goals and objectives. Alternative D performed the second best out of the action alternatives on the hydropower metrics while also meeting the purpose and need and providing the best balance of resource impacts to comply with the Grand Canyon Protection Act and protect, mitigation and improve downstream resources. The preferred alternative was identified as the best balance for meeting the purpose and need of the LTEMP EIS.

Full powerplant capacity operations were considered and eliminated as an alternative as discussed in Section 2.3.10 because fluctuation levels to the degree proposed would have adverse impacts on many downstream resources and would not meet the purpose and need for this EIS.

The DEIS is consistent with both Section 7 of CRSPA and the GCPA. The CRSPA and the GCPA are both statutes that concern operation of Glen Canyon Dam, and the GCPA expressly acknowledges that the Secretary shall implement the GCPA "in a manner fully consistent with and subject to the Colorado River Compact, the Upper Colorado River Basin Compact, the Water Treaty of 1944 with Mexico, the decree of the Supreme Court in Arizona v. California, and the provisions of the Colorado River Storage Project Act of 1956 and the Colorado River Basin Project Act of 1968 that govern allocation, appropriation, development, and exportation of the waters of the Colorado River basin." Accordingly, the Secretary has considered both statutes in this NEPA process analyzing the operation of Glen Canyon Dam. Regarding the "Full-Powerplant Capacity Operations Alternative", as explained in the EIS (see Section 2.3.10), this alternative was rejected because it would not meet the purpose, need, and objectives of the LTEMP including compliance with the GCPA. However Alternative B described in Section 2.3.2 and analyzed in Chapter 4 includes a test of "hydropower improvement flows" that featured wide daily fluctuations (up to 20,000 cfs in some years and months).

2.4 SUPPORT ALTERNATIVE D

<u>Summary Comment</u>: Numerous commenters support Alternative D, the preferred alternative, some noting that Modulated flows as set out in Alternative D will better accomplish the stated objective- "...to protect, mitigate adverse impacts on, and improve the values for which GCNP and GCNRA were established, including, but not limited to natural and cultural resources and visitor use.

<u>Response</u>: Alternative D was identified as the preferred alternative because it met the purpose and need of the proposed action while achieving the best balance between downstream resource protection and hydropower production. It was the second best of the action alternatives in terms of hydropower performance.

2.5 SUPPORT ALTERNATIVE D WITH CAVEATS

<u>Summary Comment</u>: A number of commenters support Alternative D with some changes or caveats:

- (1) Modify the HFE trigger to allow for more frequent spring HFEs that are aimed at benefitting a variety of priority resources besides sediment/sandbars, including aquatic food base, the endangered humpback chub and razorback sucker, other native fishes, the rainbow trout fishery, and riparian vegetation.
- (2) Support and opposition to low summer flows.
- (3) Support testing of sustained low steady flows (macroinvertebrate production flows) designed to increase the production and diversity of the aquatic insects in the Colorado River below Glen Canyon Dam and recommend testing of macroinvertebrate production flows very early in the 20-year implementation process.

- (4) Prefer fewer HFEs in general, as many fishermen and fishing guides believe that repeated/frequent fall HFEs may be adversely impacting the rainbow trout fishery and the aquatic food base.
- (5) Support continued testing of fall HFEs, but noting that extended duration HFEs up to 10 days in length, need to carefully consider potential impacts to the aquatic food base, rainbow trout, and invasive species in both Glen and Marble canyons
- (6) The alternative should employ actions to prevent low dissolved oxygen levels that could be harmful to fish. In general, flows below 8000 cfs should be avoided if the condition of trout or food base is poor, water temperatures are above average, and DO levels are below average.
- (7) Support careful testing of Trout Management Flows (TMFs) when there is an identified need to reduce the number of young of the year rainbow trout in Lee Ferry agreement is reached with AZGFD that the trout population can withstand a TMF experiment that has the potential to eliminate a recruitment year class.
- (1) <u>Response</u>: The implementation framework for Alternative D (Section 2.2.4.3 of the EIS) explicitly calls for the consideration of key resources, including the trout fishery, when determining which experiments to conduct in any given year. The framework also establishes off-ramps that would identify when experiments should no longer be conducted due to unacceptable adverse impacts. As described in Section 2.2.4.4, DOI will also exercise a formal process of stakeholder engagement to ensure decisions are made with sufficient information regarding the condition and potential effects on important resources.

The EIS discloses effects related to the frequency of spring HFEs. Based on modeling and analysis, there are potential positive and negative benefits to downstream resources from spring HFEs. DOI has reviewed the parameters under which spring HFEs are conducted and the way in which the accounting periods are applied. Under the preferred alternative, there would be 4-7 spring HFEs on average over the 20 year period, and there would testing and monitoring of impacts. Under the preferred alternative, spring HFE's are triggered based upon sediment or hydrology triggers. GCMRC was consulted on the potential to modify the accounting periods and the scientists most familiar with the protocol did not believe it warranted a change in accounting periods. The DOI feels that the frequency of spring HFEs in the preferred alternative provides the appropriate balance between potential positive and negative impacts.

(2) The EIS discloses effects related to low summer flows. Positive benefits include potential warming that could help with humpback chub spawning in the mainstem, and negatives include hydropower impacts and potential concerns related to trout recruitment, water quality, recreation and other concerns. The FEIS includes additional language regarding the review of scientific information that would occur prior to use of this tool in the second ten years of the LTEMP experimental period. DOI feels that the preferred alternative provides an appropriate balance between those positive and negative impacts.

- (3) The preferred alternative would allow for the possibility of testing macroinvertebrate production flows early in the LTEMP period. The decision would be based on the recommendations from GCMRC for the conditions of the year and for good experimental design including avoiding confounding results based on the experiments. The process under the preferred alternative includes communication and input from agencies including AZGFD. The impact of these flows on the aquatic food base, trout, and other species are discussed in the EIS.
- (4) The EIS discloses effects related to the frequency of HFEs. The frequency of HFEs varies between alternatives and further modeling and evaluation was performed on long term strategies that has HFEs "turned on" or "turned off" for various modeling runs. Based on evaluation of these modeling results, on analyses of published studies and on input from cooperators and the public, the DOI feels that the preferred alternative provides an appropriate balance between those positive and negative impacts.
- (5) The best available science related to the potential effects of extended duration HFEs on trout and other resources were incorporated into the analysis. The experiments include various "off ramps" in order to respond to potential impacts in the basin, including unacceptable adverse impacts outside of what was analyzed in the EIS to the rainbow trout fishery. In addition, impacts to recreation economics and other resources were included in the EIS. Consideration of unacceptable effects of sequential HFEs, including extended-duration fall HFEs, is explicitly identified in the EIS as a concern to be addressed before implementation. DOI feels that the preferred alternative includes an appropriate level of caution and consideration regarding the use of extended duration HFEs.
- (6) The preferred alternative does not include flows to address low dissolved oxygen levels in the Lees Ferry area. This concern has only been documented as occurring in one year and the recommendations from GCMRC at this time are to monitor this issue to learn more about those potential conditions prior to considering experimental actions. The joint leads recommended to GCMRC to include this consideration of monitoring in their science plan to accompany the LTEMP.
- (7) The use of TMFs has been designed as a tool for managing the trout population with respect to downstream endangered fish concerns but would also be expected to result in a healthier trout fishery with less population oscillation due to density dependence. Regardless of purpose, the trigger for this tool would be a large predicted trout recruitment event. Under these circumstances, TMFs would be expected to have positive benefits for endangered species and the stability and health of the trout fishery. Under the preferred alternative, AZGFD would be involved in annual implementation and planning discussions and Tribes would be consulted regarding the use of this tool. DOI feels that the preferred alternative includes an appropriate level of caution and consideration regarding the use of trout management flows.

2.6 SUPPORT ALTERNATIVE B

Summary Comment: Commenters expressed a preference for Alternative B, stating that Alternative B outperforms all other action alternatives for nearly all resources, and is the only action alternative that addresses the stated hydropower objective of the LTEMP EIS. In addition, DOI should consider eliminating the 8,000 cfs cap on daily fluctuations, eliminate low summer flow experiments, reduce HFE experiments to no more than one every other year, and allow operation at full power plant capacity. In comparison to Alternative D, commenters stated that Alternative B has lower air emissions and performs better than Alternative D across a wide range of resources considered in the LTEMP DEIS - humpback chub, air emissions, archeological resources, recreation value, retail power rates, hydropower value, increase in native vegetation, and water quality among others. In addition, Alternative D proposes a series of treatments which appear to be counterproductive and which would require subsequent treatments that would otherwise be unnecessary, specifically, as noted on p. 4-67, that subsequent HFEs erode the sandbars potentially benefitted by earlier HFEs. Further, Alternative D maintains an artificial intraday fluctuation cap of 8,000 cubic feet per second (cfs), while DOI has failed to show that the same cap under current operations benefits other resources. Finally, the DEIS reveals a bias towards the sediment resource, but is flawed because it considers flows as the only available method for improving the sediment resource.

<u>Response</u>: Alternative B was originally crafted and submitted by CREDA, a power utility stakeholder. It has a greatly increased daily range of water flows compared to the No Action Alternative and it includes an experimental implementation of maximum powerplant capacity releases. This alternative showed a number of positive and negative impacts on a variety of resources as were disclosed in the EIS. The preferred alternative, Alternative D, showed a better balance of effects to resources to achieve the purpose and need of the proposed action. Alternative D performed the second best out of the action alternatives on the hydropower metrics while also meeting the purpose and need and providing the best balance of resource impacts to comply with the Grand Canyon Protection Act and protect, mitigate and improve downstream resources.

As stated in Section 2.4.1, DOI considers sediment augmentation to be outside the scope of the LTEMP EIS because it is currently economically infeasible and would require additional congressional authorizations.

Modeling and NEPA analysis indicates that Alternative B would perform better than other alternatives for hydropower, however the joint leads identified Alternative D as the preferred alternative because it better satisfies the purpose and need of the proposed action and performed better than Alternative B for most other resources.

Note that the wording of the hydropower objective as presented in the EIS is "Maintain or increase Glen Canyon Dam electric energy generation, load following capability, and ramp rate capability, and minimize emissions and costs to the greatest extent practicable, consistent with improvement and long-term sustainability of downstream resources [emphasis added]." With this definition, all of the alternatives considered in the EIS satisfy the hydropower objective; however, the performance varies by alternative. It should be noted that Alternative D performed second best for hydropower out of all the action alternatives.

As stated in Section 4.5.2.3, "The average minimum number of adult humpback chub was highest for Alternatives B, D, and E, slightly lower under Alternatives A and C, and lowest under Alternatives F and G (Figure 4.5-8). These results indicate that although there are small differences among the alternatives with regard to the predicted minimum number of adult humpback chub in the Little Colorado River aggregation, all alternatives would maintain the population above at least 1,000 adults throughout the 20-year LTEMP period." It should be noted that the modeling for Alternative D showed improvement for humpback chub abundance; slightly higher abundance than No Action. Alternative D also includes macroinvertebrate production flow experiments which are expected improve the food base for humpback chub.

As stated in Section 4.15.2, "For both SO₂ and NO_x, steam turbine (coal plant) emissions are slightly lower under Alternatives F and G.... even though these two alternatives generate <2% less Glen Canyon Dam energy than the fluctuating flow alternatives..." such as Alternative B. "Net NO_x emissions related to spot market sales and purchases are lowest (greatest negative value) for the steady flow Alternatives F and G, and highest for the fluctuating flow Alternatives B and A. Net SO₂ spot market emissions are essentially the same across alternatives." However, "Given the very small differences in the estimated emissions after considering all of the factors discussed above and in light of the uncertainty of emissions modeling, it may be concluded that emissions would be similar under all of the alternatives."

As stated in Section 4.16.3, "There are expected to be some differences in the emissions of GHGs among the LTEMP alternatives." All differences were less than 0.1% of the total emissions and less than 0.0005% at the 11-state regional scale and less than 0.0001% at the US scale. The minor differences among alternatives are reported in Section 4.16.3.

The 8,000 cfs maximum daily fluctuation cap that is currently in place has been in place since 1996. The cap is present in both No Action and the preferred alternative. It was put in place in 1996 to address safety, recreation and sediment concerns. The LTEMP EIS analysis has found that the same concerns still apply. The 8,000 cfs maximum daily fluctuation is based on the 1995 Bishop study, which was reviewed by DOI experts and found to still be the best available information and appropriate for this EIS. The Bishop study surveyed both the river guides and the general public regarding preferences and the river guides reported a preference for a maximum of 8,000 cfs daily change for a "tolerable recreation experience" under relatively high average daily flows. See Appendix J, where table J-1 has been updated to show the survey responses at a range of flow levels. The 8,000 cfs maximum daily fluctuation in the preferred alternative is consistent with the over 1,000 comment letters received from river guides and members of conservation groups on the DEIS specifically stating the preference for retaining the 8,000 cfs maximum daily fluctuation in the preferred alternative that is currently in place for No Action.

Under Alternative D, low summer flows are only considered in the second ten years as an experiment and would only be performed more than once if the first test was shown to be successful in terms of effects to the humpback chub population. As disclosed in the DEIS, there are potentially negative impacts of lower summer flows to hydropower, bacteria/pathogens, vegetation, and food base/aquatic invertebrates. In terms of positive benefits, the modeling showed potential modest temperature increases which may be beneficial to humpback chub

populations under certain conditions. When considering individual model traces, variations in downstream temperatures were generally greatest in July (nearly 3°C warmer for low summer flows) and least in September (about 1°C warmer for low summer flows), with August falling in the middle (approximately 2°C warmer for low summer flows). In many meetings and consultations, GCMRC and FWS staff have expressed their expert opinions that LSFs are considered a low risk experiment to endangered fish and a potentially useful tool worthy of experimental testing because of the potential improvement to humpback mainstem spawning.

As determined by the modeling conducted for Alternative B, a reduction in HFE frequency to no more than one every other year would reduce sandbar building with potential negative impacts on camping and riparian species. Implementation of full powerplant capacity operations ("hydropower improvement flows" of Alternative B) would have adverse impacts on a number of downstream resources including aquatic food base, native fish, riparian vegetation, wildlife, cultural resources, and recreation.

Full powerplant capacity operations were considered and eliminated as an alternative as discussed in Section 2.3.10 because fluctuation levels to the degree proposed would have adverse impacts on many downstream resources and would not meet the purpose and need for this EIS. Alternative B was originally crafted and submitted by CREDA, a power utility stakeholder. It has a greatly increased daily range of water flows compared to the No Action alternative and it includes an experimental implementation of maximum powerplant capacity releases. This alternative showed a number of positive and negative impacts on a variety of resources as were disclosed in the DEIS. The preferred alternative showed more positive than negative effects to resources, as compared to Alternative B, to achieve the purpose and need.

Although Alternative D was projected to increase greenhouse gas emissions relative to Alternatives A and B, this difference was 22,908 "metric" tons not "million" tons. This increase represents a very small percentage (0.042%) of the greenhouse gas emissions that would result from the actions considered in Alternative A. The analysis for the EIS determined that under Alternative D, there would be a 152% increase in the sand load index (a measure of sandbar building potential) and a 47% decrease in the sand mass balance index, the lowest decrease of all the action alternatives. No one alternative was found which performed best for all goals and objectives. The preferred alternative was identified as the best balance for meeting the purpose and need of the LTEMP EIS.

2.7 SUPPORT ALTERNATIVE A

<u>Summary Comment</u>: Commenters expressed support for Alternative A, the no-action alternative, noting that Alternative A outperforms Alternative D with respect to a number of resource interests, and that Alternative A and Alternative B are the only two alternatives that meet the goal to "maintain or improve hydropower production." Some of the commenters acknowledged that the Preferred Alternative (Alternative D) has been changed to improve hydropower compared to how it was originally proposed, with more volume in the peak power month of August and more fluctuation in general, and could conditionally support it with the following changes: (i) increased daily fluctuations to 12 times monthly volume (in kaf) in June-August., and 10 times monthly volume (in kaf) in other months; (ii) removal of the 8,000 cfs fluctuation

cap; (iii) inclusion of hydropower improvement flow experiments; (iv) limitation of HFEs to a maximum of one occurrence per calendar year, and (v) exclusion of low summer flow experiments.

<u>Response</u>: DOI formulated the characteristics of the preferred alternative based on initial modeling of a set of 6 alternatives. There were several characteristics that were chosen to specifically reduce cost to the hydropower resource including increasing the fluctuation factor from 7 to 10 in high demand months, adjusting the monthly release pattern to match WAPA's Contract Rate of Delivery, increasing the volume of water in the high demand month of August from 750 to 800 kaf in 8.23 maf years, and eliminating steady flows after HFEs. Many objectives and resource goals were considered in the LTEMP EIS. No one alternative was found which met all goals and objectives. Alternative D performed the second best out of the action alternatives on the hydropower metrics while also meeting the purpose and need and providing the best balance of resource impacts to comply with the Grand Canyon Protection Act and protect, mitigate and improve downstream resources. The preferred alternative was identified as the best balance for meeting the purpose and need of the LTEMP EIS.

The increases in fluctuation described in the comment were all analyzed in the document in association with other alternatives and showed a number of positive and negative impacts to resources that led to those other alternatives not being identified as the preferred alternative. The preferred alternative performed the best with the fluctuation levels it currently has to address the concerns of endangered species, sediment, hydropower and other related resources.

Note that the wording of the hydropower objective as presented in the EIS is "Maintain or increase Glen Canyon Dam electric energy generation, load following capability, and ramp rate capability, and minimize emissions and costs to the greatest extent practicable, consistent with improvement and long-term sustainability of downstream resources [emphasis added]." With this definition, all of the alternatives considered in the EIS satisfy the hydropower objective; however, the performance varies by alternative. It should be noted that Alternative D performed second best for hydropower out of all the action alternatives.

2.8 SUPPORT ALTERNATIVES CONSIDERED BUT REJECTED

<u>Summary Comment</u>: Commenters expressed support for alternatives that were considered but rejected, specifically for the "Fill Lake Mead First, "Run of the River," and Decommission Glen Canyon Dam" alternatives. None of the retained alternatives, the commenters state, would be viable in the event of catastrophic reductions in Colorado River flows due to climate change, would adequately address the impacts of Glen Canyon Dam in driving the decline of Lake Mead, is capable of curtailing and reversing and the harm caused by to Glen Canyon under Lake Powell reservoir, or alleviates the effects of the dam's holding back sediment. DOI's justification for rejecting the alternatives is obsolete in the context of dramatic climate change impacts, while the current infrastructure for providing municipal water supplies through Glen Canyon Dam in unsustainable. Commenters also requested the analysis of a new alternative employing a historically based hydrograph alternative that attempts to restore hydrological functions that would benefit vegetation and possibly beaches. Other commenters stated that the LTEMP

alternatives would not be viable under catastrophic reductions in Colorado River flows due to climate change, nor adequately address the impacts of Glen Canyon Dam in driving the decline of Lake Mead. DOI's rationale for rejecting the proposed The "Fill Mead First Alternative" and "The Run-of-the-River Alternative" would be obsolete under severe climate change. Further, none of the alternatives analyzed the possibility that continued decline of Lake Mead could trigger a "shortage declaration" under the 2007 Interim Guidelines or an even more extreme "Compact Call," under which lower Colorado River basin states would legally compel the upper basin states to deliver their water allocation under the Law of the River.

<u>Response</u>: The LTEMP team developed a set of alternatives that represented the full range of reasonable experimental and management actions; met the purpose, need, and objectives of the proposed action; and were considered within the constraints of existing laws, regulations, and existing decisions and agreements. Other alternatives such as the "Fill Mead First," "Run-of-the River" and "Decommissioning the Dam" proposals were not included in the EIS because they would not meet the purpose, need, or objectives of the proposed action (see Sections 2.3.9 and 2.3.10). These alternatives would not allow compliance with water delivery requirements including the Law of the River and 2007 Interim Guidelines (Reclamation 2007a,b), and would not comply with other federal requirements and regulations, including the GCPA.

The EIS presented an analysis of historical hydrographs to determine how the alternatives would perform under the drier conditions of climate change conditions, and found that, although the resource impacts under climate change would be different from those under historic hydrology, the relative performance of alternatives would be consistent relative to each other. This indicated that Alternative D would continue to be the best of the alternatives under a changed climate.

The "Fill Mead First" proposal was not included in the DEIS because it would not meet the purpose, need, or objectives of the proposed action (see Section 2.3.9). The alternative would not allow compliance with water delivery requirements including the Law of the River and 2007 Interim Guidelines (Reclamation 2007a,b), and would not comply with other federal requirements and regulations, including the GCPA.

Regarding whether the annual volume of water moving from the Upper Colorado River Basin to the Lower Colorado River Basin should be changed for the purpose climate change considerations, annual volume determinations are presently implemented through the LROC as currently implemented through the 2007 Interim Guidelines. Consistent with the GCPA, and the purpose and need for this proposed action, any changes to annual volume determinations are beyond the scope of this NEPA analysis. Accordingly, the Proposed Action in the draft EIS does not require the Federal agencies (NPS and BOR) to either create a plan for providing water to the Colorado River in Grand Canyon during extended drought periods or develop a basin wide plan for the operations of all dams.

In December 2012, Reclamation and agencies representing the seven Colorado River Basin States completed the Colorado River Basin Water Supply and Demand Study (Reclamation 2012e). The purpose of the Study was to define future imbalances in water supply and demand in the Basin through the year 2060, and to develop and analyze options and strategies to resolve those imbalances. The study used several different scenarios for both supply and demand to capture a range in potential future conditions. The LTEMP EIS includes an analysis of meeting water delivery requirements under historic hydrology (Section 4.2.2.1) and climate change scenarios (Section 4.16.2.2). This analysis demonstrated that, although there were differences in the performance of alternatives under historic and climate change scenarios, the same relative performance of alternatives was observed under the two scenarios. The analysis also demonstrated that all alternatives could continue to meet Compact requirements.

2.9 EIS SHOULD HAVE CONSIDERED INFRASTRUCTURE ADDITIONS/MODIFICATIONS

<u>Summary Comment</u>: Commenters disagree with DOI's decision to exclude some forms of infrastructure additions, or modifications, mainly involving temperature control of dam discharges or sediment augmentation below the dam, or with DOI's definition of sediment augmentation as an infrastructure addition or modification. Commenters stated that the DEIS dismissed any augmentation outright as deeming them economically unfeasible, request that DOI provide additional information as to the decision making process used to eliminate sediment augmentation from consideration because it is "infeasible," and note that no cost evaluation or studies were performed to determine whether the installation of temperature control devices would be an appropriate alternative to analyze. Options are available to deliver trapped sediment upstream of the dam downstream, to reposition bed-load sand to beaches and/or enhance supplies from the Paria, including the use of dredges. Dredging and beach building is not new infrastructure, but a possible alternative to HFEs that should be considered.

<u>Response</u>: As stated in Section 2.4.1, DOI considers any infrastructure modifications or additions to be outside the scope of the LTEMP EIS because they are currently economically infeasible and would require additional congressional authorizations (16 U.S.C. § 460I-19). However, the DOI does not rule out future new infrastructure if resource conditions warrant. Any infrastructure addition or modification would require additional time and study. Future potential infrastructure modifications would need to be evaluated in NEPA assessments (EAs or EISs) that fully considered the environmental impacts of construction and operation. These assessments and the construction of the infrastructure would necessarily result in some delay from the time of the LTEMP ROD and actual start of operation of the infrastructure. It could take as many as 10 years or more to evaluate and construct a TCD or sediment augmentation. Sediment augmentation was also determined to be outside the scope of the 1995 EIS. The EIS has demonstrated how the preferred alternative is designed to meet as many of the environmental conditions as possible as mandated by the GCPA.

The analysis for the EIS determined that under Alternative D, there would be a 152% increase in the sand load index (a measure of sandbar building potential) and a 47% decrease in the sand mass balance index, the lowest decrease of all the action alternatives.

Sediment is an important resource in the Grand Canyon and supports aquatic and riparian ecosystems, cultural resources, and recreation. Options for how to mitigate sediment impacts have been intensively studied for more than 20 years by the Grand Canyon Research and Monitoring Center (GCMRC) as well as by many academics and DOI bureau staff. Many options were considered throughout the LTEMP EIS process to address this problem. Options for

sediment augmentation were considered originally but after evaluation were considered economically infeasible at this time. Without augmentation, HFEs are considered the most feasible and potentially beneficial approach to building sandbars with the limited sand supply available downstream of Glen Canyon Dam.

2.10 ALTERNATIVE D IGNORES (OR DOES NOT MEET) HYDROPOWER OBJECTIVE

<u>Summary Comment</u>: Commenters state that the LTEMP EIS preferred alternative does not meet the renewable resource goal of NEPA Section 101 (b) 6 or the hydropower resource goal established by the DOI for the LTEMP EIS, that power production be "maintain[ed] (or/and) improve[d]," as stated in the Purpose and Need Statement (76 FR 129, July 6, 2011. Alternative D, places restrictions on the utilization of Glen Canyon Dam for hydropower production that are in addition to the restrictions that were established as a result of the 1996 Record of Decision (ROD). As a result, the renewable hydropower resource objectives of NEPA and the LTEMP EIS process will not be met if the preferred alternative is selected. In addition, low steady flow in July, August and September would result in a detrimental impact to hydropower production and would not meet the hydropower DFC objective.

<u>Response</u>: Hydropower production was one of the key resources thoroughly evaluated in the EIS. Many objectives and resource goals were considered in the LTEMP EIS. No one alternative was found which met all goals and objectives. Alternative D was identified as the preferred alternative because it met the purpose and need of the proposed action while achieving the best balance between downstream resource protection and hydropower production. It was the second best of the action alternatives in terms of hydropower performance.

DOI formulated the characteristics of the preferred alternative based on initial modeling of a set of 6 alternatives. There were several characteristics that were chosen to specifically reduce cost to the hydropower resource including increasing the fluctuation factor from 7 to 10 in high demand months, adjusting the monthly release pattern to match WAPA's Contract Rate of Delivery, increasing the volume of water in the high demand month of August from 750 to 800 kaf in 8.23 maf years, and eliminating steady flows after HFEs.

Note that the wording of the hydropower objective as presented in the EIS is "Maintain *or* increase Glen Canyon Dam electric energy generation, load following capability, and ramp rate capability, and minimize emissions and costs to the greatest extent practicable, consistent with improvement and long-term sustainability of downstream resources [emphasis added]." With this definition, all of the alternatives considered in the EIS satisfy the hydropower objective; however, the performance varies by alternative. It should be noted that Alternative D performed second best for hydropower out of all the action alternatives.

Several commenters provided letters addressing the role of hydropower in the purpose and need of this NEPA process. The joint leads feel the purpose and need as drafted reflects the appropriate role of hydropower given the authorities surrounding the operation of Glen Canyon Dam. In certain places we have edited the text to clarify the role of GCPA downstream resource priorities and hydropower. Water delivery and hydropower are fundamental principles that show

up as the first objectives of the LTEMP. The joint leads worked hard with the stakeholders to ensure representation of all interests. The GCPA is clear in its goals and the objectives that were outlined for the LTEMP reflect the goals of GCPA and the goals reflected in the "Law of the River". Please see the resource and objective goals outlined in the EIS.

2.11 EIS DID NOT CONSIDER A FULL RANGE OF ALTERNATIVES

<u>Summary Comment</u>: Commenters expressed disappointment that no alternative increased daily hydropower flexibility and expressed a desire to allow operation at full power plant capacity (with mitigation). Commenters disagreed with DOI's rationale for excluding the Full-Powerplant Capacity Operations Alternative - that it "would not meet the purpose, need, and objectives of the LTEMP including compliance with the GCPA."

<u>Response</u>: Contrary to statements in the comment, the LTEMP EIS did evaluate and present the results for an alternative that increased hydropower generation. Alternative B was originally crafted and submitted by CREDA, a power utility stakeholder. It has a greatly increased daily range compared to the No Action Alternative and it includes an experimental implementation of maximum powerplant capacity releases. This alternative showed a number of positive and negative impacts on a variety of resources as were disclosed in the EIS. The preferred alternative showed a better balance between resources to achieve the purpose and need of the proposed action.

As determined by the modeling conducted for Alternative B, a reduction in HFE frequency to no more than one every other year would reduce sandbar building with potential impacts on camping and riparian species. Implementation of full powerplant capacity operations ("hydropower improvement flows" of Alternative B) would have adverse impacts on a number of downstream resources including aquatic food base, native fish, riparian vegetation, wildlife, cultural resources, and recreation.

Regarding the "Full-Powerplant Capacity Operations Alternative", as explained in the EIS (see Section 2.3.10), this alternative was rejected because fluctuation levels to the degree proposed would have adverse impacts on many downstream resources and would not meet the purpose, need, and objectives of the LTEMP including compliance with the GCPA. However Alternative B described in Section 2.3.2 and analyzed in Chapter 4 includes a test of "hydropower improvement flows" that featured wide daily fluctuations (up to 20,000 cfs in some years and months).

Full powerplant capacity operations were considered and eliminated as an alternative as discussed in Section 2.3.10 because fluctuation levels to the degree proposed would have adverse impacts on many downstream resources and would not meet the purpose and need for this EIS. Alternative B was originally crafted and submitted by CREDA, a power utility stakeholder. It has a greatly increased daily range compared to the No Action Alternative and it includes an experimental implementation of maximum powerplant capacity releases. This alternative showed a number of positive and negative impacts on a variety of resources as were disclosed in the DEIS. The preferred alternative showed a better balance between resources to achieve the purpose and need.

The EIS presented an analysis of historical hydrographs to determine how the alternatives would perform under the drier conditions of climate change conditions, and found that, although the resource impacts under climate change would be different from those under historic hydrology, the relative performance of alternatives would be consistent relative to each other. This indicated that Alternative D would continue to be the best of the alternatives under conditions resulting from climate change.

2.12 FLOWS SHOULD NOT DROP BELOW 8,000 CFS

<u>Summary Comment</u>: Commercial rafters note that it is critical to their operations and public safety that the actual, not mean, daily minimum flow never drops below 8000cfs. Minimum flows of 5000 cfs from 7pm-7am for Alternative D indicated in Table 2.1 (page 2-9) are inadequate and dangerous. Navigational safety for boating is compromised when actual flows drop below 8000 cfs due to exposed rocks and increased difficulty in navigating the channel. Further, rafters are concerned with the minimum water level over the entire 24-hour day period; water released overnight will have an effect on many boaters downstream in the subsequent daytime hours. Also, low flow minimums below that of the 8,000 cfs level will restrict the time off-river by slowing the on-river pace and in turn diminish the overall experience for the visitor.

<u>Response</u>: Under the No Action Alternative, current minimum releases from the dam are set at 8,000 cfs from 7 am to 7 pm and 5,000 cfs from 7 pm to 7 am. This flow pattern, averaging a minimum over 24 hours of 6,500 cfs, has been in place since the 1996 Record of Decision was signed. Lower flows travel at a slower rate of speed and flows attenuate as the water moves downstream. The current minimums that have been in place for 20 years would be continued under the preferred alternative. Additionally, monthly volumes in the preferred alternative represent a relatively even pattern, allowing for more predictable and stable flow patterns throughout the year, including the high visitation summer months.

The EIS evaluates impacts on recreation related to navigation, fluctuations, and safety. These evaluations considered the minimum flow during each 24-hr period and the maximum daily fluctuation of each alternative. The preferred alternative is very similar to the No Action Alternative with regard to navigation, fluctuation level, and safety. As for all experiments, the implementation of low summer flows, TMFs, and macroinvertebrate production flows would take into account unacceptable adverse impacts on key resources including recreation. If unacceptable adverse impacts on key resources were anticipated, the minimum flow for both of these experiments could be adjusted.

The EIS acknowledges that low flows would reduce the amount of time boaters could spend off river, and assesses the impacts of each alternative using a "time off river" index. The preferred alternative is expected to result in a very small decrease (<2%) in this index relative to the No Action Alternative.

2.13 8,000 CFS FLUCTUATION LIMIT

<u>Summary Comment</u>: Many commenters expressed a preference to keep daily water release fluctuations capped at 8,000 cfs, stating that this will help preserve natural resources that provide habitat for endangered fish and the re-building of beaches along the Colorado River. High fluctuations, some also note, can diminish or reverse some of the benefits of the HFEs and make it very difficult on river rafters and guides as they manage and plan river trips and manage boats on the shoreline. Other commenters support removing the 8,000 cfs cap, stating that it is unsupported by the scientific literature and restricts the hydropower resource without regard to CRSPA provisions to the contrary, and that DOI has failed to show that it benefits other resources.

<u>Response</u>: Alternative B was originally crafted and submitted by CREDA, a power utility stakeholder. It has a greatly increased daily range compared to the No Action Alternative and it includes an experimental implementation of maximum powerplant capacity releases. This alternative showed a number of positive and negative impacts on a variety of resources as were disclosed in the EIS. The preferred alternative showed a better balance between resources to achieve the purpose and need of the proposed action.

The 8,000 cfs maximum daily fluctuation cap that is currently in place has been in place since 1996. The cap is present in both No Action and the preferred alternative. It was put in place in 1996 to address safety, recreation and sediment concerns. The LTEMP EIS analysis has found that the same concerns still apply. The 8,000 cfs maximum daily fluctuation is based on the 1995 Bishop study, which was reviewed by DOI experts and found to still be the best available information and appropriate for this EIS. The Bishop study surveyed both the river guides and the general public regarding preferences and the river guides reported a preference for a maximum of 8,000 cfs daily change for a "tolerable recreation experience" under relatively high average daily flows. See Appendix J, where table J-1 has been updated to show the survey responses at a range of flow levels The 8,000 cfs maximum daily fluctuation in the preferred alternative is consistent with the over 1,000 comment letters received from river guides and members of conservation groups on the DEIS specifically stating the preference for retaining the 8,000 cfs maximum daily fluctuation in place for No Action.

Implementation of full powerplant capacity operations ("hydropower improvement flows" of Alternative B) would have adverse impacts on a number of downstream resources including aquatic food base, native fish, riparian vegetation, wildlife, cultural resources, and recreation.

The increases in fluctuation described in comments were analyzed in the document in association with various alternatives and showed a number of positive and negative impacts to resources that led to those other alternatives not being identified as the preferred alternative. The preferred alternative performed the best with the fluctuation levels it currently has to address the concerns of endangered species, sediment, hydropower and other related resources.

Modeling also indicated that the higher fluctuation levels under Alternative B or E would result in a number of adverse impacts including impacts on sediment, aquatic food base, vegetation, and recreation.

2.14 EIS DID NOT ADEQUATELY CONSIDER THE EFFECTS OF CLIMATE CHANGE

<u>Summary Comment</u>: Commenters expressed that the DEIS fails to take climate change seriously, fails to analyze the impacts on the water supply system of climate change for the entire Colorado River including the ecology in the Grand Canyon due to the operation of Glen Canyon Dam and Lake Powell, fails to analyze the likelihood of a "Compact Call" on the Colorado River due to the water lost by evaporation and seepage due to the operation of Glen Canyon Dam and Lake Powell, and fails to analyze a full range of alternatives that should be considered due to climate change including the decommissioning of Glen Canyon Dam and the draining of Lake Powell, stating that the criteria used to eliminate these alternatives would be obsolete in the face of catastrophic reductions in Colorado River flows caused by climate change, while none of the current alternatives would be viable under such conditions.

<u>Response</u>: The LTEMP utilized the best available science provided through the peer reviewed Basin Study (Reclamation 2012) regarding climate change projections in the Colorado River. The hydrological traces generated for the Basin Study were utilized as described in Section 4.16.1 and Appendix C.

This EIS used 21 reconstructed historical hydrological traces that were reweighted based on the 112 climate change traces to represent the best current understanding of what might happen under the drier conditions of climate change. Based on this analysis, the weights on the hydrological traces had a small effect on the aggregate performance but did not change the rankings of alternatives. This result suggests the relative performance of the alternatives would be consistent regardless of the uncertainty of the effects of climate change, and that Alternative D would continue to be the best of the alternatives under a changed climate.

The LTEMP team developed a set of alternatives that represented the full range of reasonable experimental and management actions; met the purpose, need, and objectives of the proposed action; and were considered within the constraints of existing laws, regulations, and existing decisions and agreements. Other alternatives such as decommissioning of Glen Canyon Dam, "Fill Mead First" and "Run-of-the River" proposals were not included in the EIS because they would not meet the purpose, need, or objectives of the proposed action (see Sections 2.3.8, 2.3.9 and 2.3.10). These alternatives would not allow compliance with water delivery requirements including the Law of the River and 2007 Interim Guidelines (Reclamation 2007a,b), and would not comply with other federal requirements and regulations, including the GCPA. The performance of these alternatives under climate change during the 20-year LTEMP period was discussed in Section 4.16.2.

Regarding whether the annual volume of water moving from the Upper Colorado River Basin to the Lower Colorado River Basin should be changed for the purpose climate change considerations, annual volume determinations are presently implemented through LROC as currently implemented through the 2007 Interim Guidelines. Consistent with the GCPA, and the purpose and need for this proposed action, any changes to annual volume determinations are beyond the scope of this NEPA analysis. Accordingly, the Proposed Action in the draft EIS does not require the Federal agencies (NPS and BOR) to either create a plan for providing water to the

Colorado River in Grand Canyon during extended drought periods or develop a basin wide plan for the operations of all dams.

The Basin Study (Reclamation 2012) was peer reviewed and is the best available science regarding climate change projections in the Colorado River Study. The hydrological traces generated for the Basin Study were utilized as described in Section 4.16.1.2 and Appendix C. This EIS did not model the 112 climate change hydrologic traces generated as part of the Basin Study due to insufficient data to drive the complex suite of models. We used 21 reconstructed historical hydrological traces that were reweighted based on the 112 climate change traces to represent the best current understanding of what might happen because of climate change. Based on this analysis, the weights on the hydrological traces had a small effect on the aggregate performance but did not change the rankings of alternatives. This result suggests the alternatives are robust to uncertainty about climate change.

By utilizing historic hydrology information and climate change-weighted hydrology information, we evaluated the effects of alternative operations over the 20-year LTEMP period under a widerange of hydrologic and sediment conditions. We used a complete set of integrated models to evaluate effects of these changes in hydrology and sediment on the full range of resources including aquatic resources, terrestrial vegetation and wildlife, endangered species, cultural resources, Tribal resources, recreation, and socioeconomics. We modeled scenarios where the pool elevation dropped below the estimated minimum power pool (3,490 ft) and the results showed that there was no significant difference in the relative performance of the alternatives (Section 4.2). As stated in Section 4.2.2 of the EIS, "For Lake Powell elevations, all alternatives show very similar percentages for elevations that are $\leq 3,490$ ft. The percentage of traces ranges between 0 and 5 and remains relatively constant throughout the 20 year period." In addition, we looked at the cumulative impacts of LTEMP in combination with other past, present, and future projects. We determined that the alternatives were robust to projected climate change-related hydrologies and could operate within the constraints of each alternative over the 20-year LTEMP period.

2.15 GEOGRAPHIC SCOPE OF THE EIS

<u>Summary Comment</u>: One commenter stated that the DEIS fails to clearly distinguish between 1) the scope of the proposed activity, i.e., Colorado River Ecosystem, and 2) the area of impact associated with implementing any activity proposed under LTEMP, and that it is unclear why some resources are discussed beyond the specific geographic limitations of the Colorado River Ecosystem and other resources are not. The DEIS should remain consistent with the geographic scopes intended for the proposed activities and impacts identified throughout its discussions and analyses.

<u>Response</u>: The text of the EIS has been revised to better clarify the geographic scope (See Section 1.5.1). It should be noted that the areas of potential impact vary by resource by necessity to evaluate the impacts. For instance, the effects of dam operations on vegetation or sediment are much more limited geographically than the effects on air quality or hydropower. In the cumulative analysis section (4.17), the region of influence for each resource is defined in the second column of Table 4.17-2.

2.16 MONITORING TO SUPPORT ADAPTIVE MANAGEMENT

<u>Summary Comment</u>: Commenters on this issue requested that the lead agencies conduct longterm resource monitoring to support adaptive management and to measure progress toward achieving goals and desired future condition.

Response: Long-term-monitoring for adaptive management will continue under the LTEMP. GCMRC has developed a Science Plan for the LTEMP that will guide these monitoring activities.

2.17 INCLUDE A SUMMARY OF EXPERIMENTAL AND MANAGEMENT ACTIONS FOR THE RAINBOW TROUT FISHERY

<u>Summary Comment</u>: Commenters on this issue requested that the selected alternative and/or the Record of Decision include a summary of experimental and management actions that are proposed to benefit the Lees Ferry rainbow trout fishery.

<u>Response</u>: The FEIS includes descriptions of all experimental and management actions for the alternatives in Chapter 2. There is not a summary of trout related experiments for all alternatives in one section, however all aquatic resource experiments for Alternative D can be found in Table 2-9 and Section 2.2.4.6.

2.18 GLEN CANYON DAM SAFETY AND SECURITY

<u>Summary Comment</u>: A commenter stated that Reclamation's misrepresentation of the growing risks associated with GCDs operational safety, the security of its water storage and hydropower generation benefits, as well as threats to water quality, not only renders its suite of alternatives and assessments thereof, incomplete, but renders an injustice to the Colorado River society as a whole by denying the public a truthful and up-to-date assessment of the known risks associated with the continued operations of GCD and the implied benefits therein.

<u>Response</u>: It is not clear what safety risks this comment is referring to. Dam safety is a priority of the Bureau of Reclamation and is evaluated comprehensively through Reclamation's Dam Safety program on an -ongoing basis. These on-going evaluations are wholly independent of LTEMP and beyond the scope of this NEPA analysis. The co-lead agencies feel that this NEPA process takes a comprehensive "hard look" at the proposed action and alternatives that is fully consistent with NEPA.

2.19 EXPERIMENTAL UNCERTAINTY

<u>Summary Comment</u>: Commenters stated that the experimental framework included in Alternative D fails to provide any meaningful level of certainty, there is insufficient detail and information regarding experimental triggers, baselines, off-ramps, hypotheses, unacceptable adverse impact, and metrics to determine success, or adaptively managing under LTEMP, and the framework vests sole discretion in the DOI to determine whether a given experiment will be conducted.

Response: Triggers for experiments are clearly articulated. For several fish experiments triggers were developed and defined through the consultation process with FWS with information provided from GCMRC another experts and cooperators. GCMRC also developed a science plan and will be intimately involved in the implementation of LTEMP experiments through the GCDAMP triennial budget process and on-going monitoring, evaluation and reporting. The Colorado River Ecosystem is a dynamic system with complex interactions and changing conditions from year to year. The preferred alternative established a balance between structure and flexibility for experimentation to be able to allow for adaptive management in response to both changing conditions and new information gathered from experimentation. The preferred alternative has a number of annual steps built into the process to coordinate with GCMRC, agencies and stakeholders to consider the conditions of the year and the appropriate experiments given those conditions. Triggers and 'off-ramps' are defined. The triggers for HFEs and LSFs are well defined. Many of the triggers for tools for fish management have intentionally been left to be more flexible in the EIS as those have been defined during the consultation process with the FWS and will be more clearly stated in the FEIS. Greater specificity regarding triggers for HFE implementation and humpback chub actions have been added to the EIS (Appendices O and P).

The implementation framework for Alternative D (Section 2.2.4.3 of the EIS) explicitly calls for the consideration of key resources when determining which experiments to conduct in any given year. The framework also establishes off-ramps that would identify when experiments should no longer be conducted due to unacceptable adverse impacts. As described in Section 2.2.4.4, DOI will also exercise a formal process of stakeholder engagement to ensure decisions are made with sufficient information regarding the condition and potential effects on important resources. GCMRC has developed a science plan that describes the information that will be gathered during implementation of the LTEMP, and how that information regarding triggers for fish actions.

2.20 IMPACTS ON BASIN FUND

<u>Summary Comment</u>: A commenter stated that impacts to the Upper Colorado River Basin Fund must be assessed and considered.

<u>Response</u>: In Section 4.13 and appendix K of this EIS there is extensive analysis of the revenues from Glen Canyon hydropower production contributing to the Basin Fund. There are other factors that affect the balance of the Basin Fund that are outside the scope of the actions analyzed in this EIS, however those would not vary by alternative so they were not analyzed in the LTEMP EIS.

2.21 DUE CONSIDERATION OF HYDROPOWER

<u>Summary Comment</u>: Commenters stated that the Department of Interior must balance numerous important, and often competing, factors in developing the final LTEMP. Commenters urged that due consideration and weight be given to the importance of hydropower in this process.

Response: Many objectives and resource goals were considered in the LTEMP EIS. Hydropower production was one of the key resources thoroughly evaluated in the EIS. No one alternative was found which met all goals and objectives. Alternative D performed the second best out of the action alternatives on the hydropower metrics while also meeting the purpose and need and providing the best balance of resource impacts to comply with the Grand Canyon Protection Act and protect, mitigate adverse impacts to, and improve downstream resources. The preferred alternative was identified as the best balance for meeting the purpose and need of the LTEMP EIS.

2.22 SUPPORT ALTERNATIVE F

<u>Summary Comment</u>: One commenter expressed a preference for Alternative F, seasonally adjusted steady flows, because it would mimic natural conditions, and thus improve tailwater temperature and sediment conditions and riparian ecosystems downstream. Peaking power provided by fluctuating flows could be replaced by utility scale photovoltaic solar power.

<u>Response</u>: Alternative D was chosen as the preferred alternative because it met the purpose and need of the proposed action while achieving the best balance between downstream resource protection and hydropower production. This analysis considered the means and costs of alternate sources of peaking capacity as applicable under the various LTEMP alternatives

3 AQUATIC ECOLOGY

3.1 ALTERNATIVES WILL NOT PROTECT COLORADO RIVER ECOSYSTEM

Summary Comment: Commenters noted that Bureau of Reclamation (BOR) and NPS are setting themselves up for failure by attempting to restore an entire ecosystem based on responses to sediment inputs, temperature, and power grid needs alone instead of basing Glen Canvon Dams hydrograph on historic flow magnitude, frequency, duration, timing, and rate of change. In addition, there is a growing body of evidence on life history adaptations that indicate that several species depend on very localized hydrological conditions for their survival and reproduction. The Colorado River Ecosystem is continually changing, and novel, unanticipated ecosystem developments occur nearly every year. Many environmental and economic uncertainties, both recognized and unknown, complicate future adaptive management of the Dam and the Colorado River Ecosystem. Ensuring flexibility and administrative responsiveness is essential. Also, contingency planning is needed on a regular (annual) basis to deal with surprise events or processes, such as the sudden eruption of non-native green sunfish in Glen Canyon in 2015. Other commenters stated that adaptive management is based on the strategy of not placing the entire ecosystem at risk to answer management questions. Many management questions should be explored through small-scale field experiments or through modeling, rather than subjecting the entire ecosystem to an experiment that might jeopardize component species or processes. Only in cases where whole-ecosystem experiments are known to work, should they be conducted (e.g., testing the impacts of springtime HFEs on fluvial geomorphology and river biota).

<u>Response</u>: The purpose and need for LTEMP clearly align with the Grand Canyon Protection Act, whose goal is "to protect, improve and mitigate adverse impacts to natural and cultural resources and visitor use for which both park units were established". All alternatives, including the preferred alternative (Alternative D), have positive and negative impacts which are disclosed in the EIS analyses. The Colorado River Ecosystem is a dynamic system with complex interactions and changing conditions from year to year. The preferred alternative established a balance between structure and flexibility for experimentation to be able to allow for adaptive management in response to both changing conditions and new information gathered from experimentation. The preferred alternative has a number of annual steps built into the process to coordinate with GCMRC, agencies and stakeholders to consider the conditions of the year and the appropriate experiments given those conditions. The joint leads have concluded that Alternative D has more positives than negatives and provides the best balance to meet the purpose and need for this project and to comply with the Grand Canyon Protection Act and the Law of the River. For a summary of impacts among alternatives, see Table 2-14.

3.2 IMPACTS OF FALL HFEs ON FISH AND AQUATIC FOOD BASE

<u>Summary Comment</u>: A number of commenters believed that repeated/frequent fall HFEs may be adversely impacting the rainbow trout fishery and the aquatic food base. Others stated that HFEs should be either deferred or only conducted in the spring if the condition of trout or the food base in Lees Ferry is poor. Others stated that the effects of sediment-triggered fall HFEs on trout recruitment are uncertain, but fall HFEs are expected to have less effect on trout production than

spring HFEs. Some commenters believed that a connection could be made between three consecutive fall HFEs and an increase in brown trout recruitment. In addition, the Havasupai requested the development of a monitoring site to study the impacts from the backflow of the HFE into Havasu Creek as well as impacts on the endangered fish naturally occurring and introduced at this location.

<u>Response</u>: The effects of HFEs on aquatic resources were fully analyzed in Section 4.5. This section reports that fall HFEs have not been found to have negative impacts to the food base, to trout recruitment or to native or non-native fish generally. There is no evidence of the frequency of HFEs being related to brown trout recruitment. Prior to the implementation of any HFEs, there would consideration of conditions related to humpback chub, trout, and aquatic food base. The HFE implementation process includes meetings with the GCMRC, FWS, AZFGD and other partners to take into consideration annual conditions as well as any new information. If unacceptable adverse impacts are predicted or observed, then experiments may be discontinued or additional NEPA processes may be conducted.

DOI will discuss the idea of a Havasupai monitoring site near Havasu Creek with GCMRC to be considered in the science plan or the monitoring implementation details.

3.3 REDUCE NUMBER OF HFES TO IMPROVE THE AQUATIC FOOD BASE

<u>Summary Comment</u>: Commenters expressed concern regarding the continuation and expansion of HFEs, both in terms of frequency and duration. HFEs scour and remove the aquatic food base resulting in the loss of the main food source for trout and downstream humpback chub. Studies suggest that it takes 1 to 1.5 years for the food base to recover to pre-HFE levels after any HFE event. This indicates that if HFEs are implemented one or more times annually, there will be little opportunity for the food base to recover which will increase stress on the endangered native fish. Therefore, any selected alternative should include no more than one HFE every other year.

<u>Response</u>: Flow effects (including HFEs) on the aquatic food base are discussed in Section 4.5.2.1, with a more detailed discussion in Section F.2.2.1 (Appendix F). Reducing the number of HFEs (e.g., no more than one HFE every other year) is not supported by the analysis presented in the EIS. As discussed in Section F.2.2.1 (Appendix F), repeated HFEs are expected to result in a more productive aquatic food base. Such results do not occur from infrequent HFEs. Nevertheless, the HFE's are experimental in nature and allow for some adaptation over time based on results within the bounds of environmental compliance. If the HFEs cause unacceptable adverse impacts, there is a process defined in the EIS to suspend the HFEs.

3.4 IMPACTS OF MINIMUM FLOWS BEFORE AND AFTER HFEs

<u>Summary Comment</u>: Several commenters support the idea of keeping flows low and stable before and after HFEs. The health of the rainbow trout fishery and the aquatic food base, water temperatures, and dissolved oxygen levels should all be considered when determining the minimum flow that will precede and/or follow a fall high flow experiment (HFE). In general,

flows below 8000 cfs should be avoided if the condition of trout or food base is poor, water temperatures are above average, and dissolved oxygen levels are below average

<u>Response</u>: The issues the commenter raises, the health of the rainbow trout fishery, the aquatic food base, water temperature, and dissolved oxygen levels would be considered as part of the assessment of resources listed in Section 2.2.4.3 that will be considered when determining the flows that would precede or follow an HFE. Though the LTEMP EIS analysis did not find that steady flows after fall HFEs present a risk to the humpback chub population, that experiment has been removed for a number of other reasons including that it would be difficult to measure the results.

3.5 PROVIDE ADDITIONAL INFORMATION ON MOST RECENT FALL HFEs

Summary Comment: Commenters requested the inclusion of results from the recent HFEs (2012, 2013, and 2014) throughout the LTEMP DEIS including the portion of Section of 4.5 which addresses flow effects on benthic invertebrates. Given the impact to the endangered species, hydroelectric capacity and environmental ramifications of purchased power to replace the effects of HFEs; the failure of the DEIS to include and consider all of the HFEs and the cumulative impacts represents a significant analytical flaw. In addition, the DEIS fails to, but must include an analysis of the results of the 2013 and 2014 HFEs on rainbow trout recruitment and humpback chub.

<u>Response</u>: Preliminary results of the more recent HFEs have been added to Sections 3.5 and 4.5 of the EIS.

3.6 SUPPORT CAREFUL TESTING OF TMFs

<u>Summary Comment</u>: A number of commenters support the careful testing trout management flows (TMF), but only when there is an identified need to reduce the number of young of the year rainbow trout in Lees Ferry; while recognizing Tribal opposition to TMFs. A TMF should only be tested if AZGFD is in agreement that the trout population can withstand a potentially very successful TMF experiment that eliminates a recruitment year class. Testing TMFs when young trout are not abundant may adversely affect the Lees Ferry trout fishery, is contrary to the "condition dependent" adaptive management approach, and may not lead to conclusive results on the effectiveness of TMFs as a management tool.

<u>Response</u>: The DOI considers TMFs as an important tool for minimizing risk to downstream endangered fish from trout regardless of the cause of trout population increases or migration. Text has been added to Section 2.2.4.6 to clarify that implementation of TMFs would consider resource condition assessments and resource concerns. Early tests of TMFs would be used to determine the effectiveness of TMFs and a best approach to trout management. TMFs may not be tested when there appears to be the potential for unacceptable adverse impacts to specific resources such as humpback chub. Triggers for the implementation of TMFs would be developed in consultation with the AZGFD and other stakeholders to determine the implementation based on resource conditions in the given year. For Alternative D (the preferred alternative), trout management flows are triggered by the prediction of high recruitment of young-of-year trout regardless of the cause of trout population increases or migration.

Based on the best available information, extensive modeling and extensive dialog with subject matter experts, the joint leads believe there would be concerns associated with testing TMFs when the trout population and trout recruitment is too low as that might not provide the best conditions for testing this tool. The text in the EIS has been edited to reflect this concern. Experimentally, the best conditions for testing TMFs are when high trout recruitment is expected, such as during a high release volume/equalization year or after a spring HFE, however the conditions for both of these are somewhat unpredictable and infrequent. Also DOI recognizes the concern of several Tribes related to conducting TMFs and the concerns of trout fisherman about the status of the trout fishery when TMFs are tested. DOI has designed the process for experiments to include a detailed evaluation of the resource conditions for the year and an extensive coordination process to hear from Tribes, agencies and the AMWG stakeholders about the concerns present at the time of implementing the experiments

3.7 QUAGGA MUSSELS

<u>Summary Comment</u>: A few commenters expressed concern that a potential infestation of quagga mussels below Glen Canyon Dam will have an adverse environmental impact. Because of the range of alternatives were developed before quagga mussel infestation occurred, it will be critical to study how best to mitigate the environmental damage it may cause. It was believed that there is a high probability that quagga mussels will change the food base chain and conflict with the proposed need for low summer flows to improve the aquatic food base. It is unclear what will happen to productivity in the Glen Canyon Reach if organic matter is reduced as a result of quagga mussels becoming established in Lee's Ferry.

<u>Response</u>: Quagga mussels continue to be found in the Colorado River below Glen Canyon Dam. Their distribution is patchy and highly influenced by location-specific flow regimes. While the potential impact from quagga mussels is uncertain, their management is beyond the scope of the EIS. If required, they would be managed through other actions conducted by the NPS. The Kennedy (2007) report cited in the EIS points out that negative ecological impacts appear to be low and that moderate densities of quagga mussels may increase food available to fish, increase the complexity of habitat and stimulate additional benthic production. (A link to the report can be found at http://pubs.usgs.gov/of/2007/1085/.)

3.8 PROVIDE ADDITIONAL/UPDATED INFORMATION ON AQUATIC RESOURCES AND SPECIES

<u>Summary Comment</u>: Several commenters wanted to see additional or updated information on a particular aquatic resource or species. These included wanting the EIS to provide more recent information regarding the status of razorback suckers in Lake Mead and the Lower Grand Canyon. Another commenter wanted the EIS to include the most recent data on brown trout recruitment in the Lee's Ferry section of the Colorado River. One commenter requested that the connection between mayflies, stoneflies and caddisflies and the fish community be more

thoroughly addressed. Another commenter was concerned about predation of larval humpback chub by juvenile and adult humpback chub, as this could be a significant factor in year-to-year recruitment. One commenter noted that to fulfill the requirements of the Endangered Species Act, the best scientific and commercial data available needs to be used.

<u>Response</u>: The assessment of potential impacts from LTEMP activities on aquatic resources and species in the EIS is based on an extensive review of the scientific literature and other available data; as well as input from federal and state agencies (including the FWS and AZGFD) and other interested stakeholders. Recent literature and information on the aquatic resources and species made available since publication of the DEIS has been incorporated into the EIS. In regards to listed fish species, the recently completed Biological Assessment (BA) for the LTEMP is included as Appendix 0 of the EIS. The aquatic ecology sections (Sections 3.5 and 4.5) have been modified to include information from the BA.

3.9 AZGFD MANAGEMENT PLAN

<u>Summary Comment</u>: Commenter wanted information added describing the Arizona Game and Fish Departments Fisheries Management Plan, Colorado River-Lees Ferry 2015-2025 (Rogers 2015). The plan includes implementing actions that may be considered ongoing actions. The AZGFD refined the goals and objectives for the Lees Ferry Fishery in that plan. The purpose of the plan was to set measurable goals and identify several strategies to reach these goals. This differs from the NPS comprehensive fisheries management plan which focused on stocking triggers for Lees Ferry in the event the fishery returned to conditions prior to MLFF where it was primarily supported by stocking. A key concern which is addressed in the AZGFD plan, and not in the NPS comprehensive fish management plan, is the definition and response to a catastrophic loss of the fishery. The AZGFD plan is a proactive approach that aligns with the NPS Comprehensive Fisheries Management Plan that seeks to avoid a catastrophic loss and to ensure that the highly valued recreational trout fishery is sustained or enhanced while minimizing impacts to humpback chub.

<u>Response</u>: The AZGFD management plan and NPS's Comprehensive Fisheries Management Plan are independent of the LTEMP EIS. Currently, there are aspects of the AZGFD plan that are not in agreement with the LTEMP or with the Comprehensive Fisheries Management Plan. These issues include HFEs triggered by dissolved oxygen conditions or trout population levels, and details related to stocking conditions. Because of these differences, at this time, the joint leads cannot include these objectives or reference this plan in its current form because several of the actions in the plan do not meet the purpose and need of the LTEMP or do not agree with the Comprehensive Fisheries Management Plan. The NPS is continuing to work with AZGFD to work toward agreement on these concerns.

4 CULTURAL RESOURCES

4.1 APPLICABILITY AND COMPOSITION OF CULTURAL RESOURCE METRICS

4.1.1 Wind Transport

<u>Summary Comment</u>: Commenters expressed concern with how effectively this metric captures the potential for wind transported sediment to help preserve archaeological sites along the Colorado River. The comments ranged from requests to remove the metric from the analysis to clarifications on how the components of the metric were developed.

<u>Response</u>: The text in the EIS was updated to include the results of several recently published studies on the extent and range that wind transported sediment can be distributed. The text was also modified to indicate that some Tribal members may not consider the reburial of sites by wind transported sediment to be a positive effect.

4.1.2 Time Off River

<u>Summary Comment</u>: Commenters expressed that the time-off-river metric, a measure of the time available to rafters for shore exploration during river trips and a function of river flow levels, is a poor measure of potential impacts on archaeological sites and should not play an important role in the EIS analyses. Commenters noted that while the metric is a logical construct, there is no evidence that more time off river leads to greater visitation because visitors have many other options for exploration and, further, that respectful visitation of Class I sites is actually expected and encouraged.

<u>Response</u>: In the cultural section, 4.8, this metric is presented as a **potential** for increased effects, rather than as a direct link to increased effects, because the monitoring of these sites which NPS has performed over the last 20 years has not shown a strong correlation or a cause and effect relationship. The metric was developed in response to concerns from the Tribes who have noted negative effects at archaeological sites due to visitors and their observations that additional time off river allows for boating parties to spend more time hiking and visiting sites. The value of the metric for cultural resources varied little among the LTEMP alternatives as noted in Section 4.8.

In the recreation analysis in Section 4.11, this metric measures the potential effects of alternate flow regimes on recreation and therefore is the most pertinent means of evaluating the LTEMP alternatives with respect to effects on recreation. In the recreation analysis, more time-off-river was considered a positive for recreation as it was used to "evaluate the opportunity for onshore exploration." The value of the metric varied little among the LTEMP alternatives (Figure 4.10-1), and thus had little value in differentiating them as noted in Section 4.11.2.

4.1.3 Glen Canyon Flow Effect Metric

<u>Comment Summary</u>: Commenters raised concerns with the choice of Ninemile Terrace as a proxy for other historic properties in GCNRA and also questioned the effectiveness of the metric as a predictor of erosion at archaeological sites in GCNRA because of a lack of direct studies on this topic.

<u>Response</u>: Text was added in Appendix H to further clarify why Ninemile Terrace is an effective proxy for other archaeological sites in GCNRA. The components of the metric are based on direct observations of effects from past HFEs on the terrace.

4.2 AREA OF POTENTIAL EFFECTS

Request disclosure on how APE was determined. Do not support the rim to rim application of the APE.

<u>Summary Comment</u>: Several commenters inquired about the Area of Potential Effect (APE) for the Section 106 of the National Historic Preservation Act review for LTEMP. The DEIS contained general language on this topic as consultation was on going when the DEIS was issued. Specifically they were seeking information on how the APE was determined and which historic properties were being included. There was a commenter who disagreed with defining the APE to include all historic properties within the canyon.

Response: Text in Chapters 3 and 4 has been updated to explain the differences between the analysis of potential effects between NEPA and NHPA Section 106. Reclamation and the NPS are developing a new programmatic agreement for the NHPA Section 106 process and to help guide management of cultural resources along the Colorado River. These consultations are on-going. Once the Cultural Programmatic Agreement is complete the information on the APE will be provided to the public.

4.3 LANDMARK STATUS FOR GRAND CANYON AND ITS RESOURCES

<u>Summary Comment</u>: Comments were received suggesting that the cultural resources found within the Grand Canyon warrant consideration as a National Historic Landmark and that nomination of the resources as a National Historic Landmark begin soon.

Response: The NPS is continuing consultations with the Tribes concerning future nominations for listing of resource in the National Register of Historic Places. However, the nomination of the canyon as a National Historic Landmark is beyond the scope of the LTEMP EIS.

5 CUMULATIVE IMPACTS

5.1 ADDITIONAL FORESEEABLE ACTIONS (INCLUDING MANAGEMENT PLANS) TO BE INCLUDED

<u>Summary Comment</u>: Commenters requested that additional foreseeable actions be included in the cumulative impacts analysis. Among them were the MLFF alternative (1996 ROD), releases of contaminated water into Lake Powell (related to mines), the AZGFD fisheries management plan, and the Flaming Gorge Dam and Aspinall Unit EISs.

<u>Response</u>: The cumulative impacts analysis was revised to include the additional foreseeable actions as requested and to account for the effects of these actions on various resources, as warranted.

5.2 REFINEMENT OF DESCRIPTION AND/OR IMPACT ASSESSMENT FOR FORESEEABLE ACTIONS CONSIDERED

<u>Summary Comment</u>: Commenters requested revisions to modify or update the descriptions of foreseeable actions identified; e.g., to flesh out the implications of the 2007 Guidelines to water resources or to update the status of the Clean Power Plan proposed rule. Other requests pertained to defining terms (e.g., the region of influence or the term "future projects"). Commenters also requested that descriptions of power plants (Four Corners and San Juan Generating Station), listed as foreseeable actions, be revised to reflect the current and scheduled modifications at those plants and that the implications of these changes to regional hydropower generation be addressed.

<u>Response</u>: The cumulative impacts analysis was revised to update descriptions of foreseeable actions and to account for the effect of these changes on water resources and hydropower, as warranted. Definitions of terms such as "region of influence" and "future projects" were added in Section 4.17.

6 HYDROPOWER

6.1 SCOPE OF ANALYSIS AND STUDY ELEMENTS

6.1.1 Hydropower Goals and Objectives

<u>Summary Comment</u>: Hydropower goal and objectives are not clear and use conflicting language in the various sections of the report.

<u>Response</u>: Text has been revised to make consistent throughout EIS.

6.1.2 Generation at Hoover Dam

<u>Summary Comment</u>: Analysis of changes in generation at Hoover Dam is beyond the scope of LTEMP because Hoover and Glen Canyon are statutorily separate.

<u>Response</u>: Although Glen and Hoover are statutorily separate and effect different entities, the power systems economic analysis focuses on power production costs and capacity expansion related expenditures for the grid as a whole not on the financial impacts for specific entities. Economic impacts on Lake Mead and Hoover Dam are the result of changes in monthly water release patterns from Lake Powell. These changes to impacts on power economics have been identified and therefore were included in the FEIS.

6.1.3 Cumulative Impacts on Power Systems

Summary Comment: Cumulative impacts on power systems were not adequately addressed.

Response: Text has been added to the EIS to address this concern in Section 4.17.

6.1.4 Analysis of Impacts on Basin Fund

<u>Summary Comment</u>: The impacts on Upper Colorado Basin Fund, cash flow and its ability to fund important research should be studied in more detail.

<u>Response</u>: Section 3.13.1.2 includes a description of the Basin Fund and states "Most of the revenues come from sales of hydroelectric power and transmission service." This EIS did analyze revenues from Glen Canyon hydropower production and their contribution to the Basin Fund. See the extensive hydropower analysis in 4.13 and Appendix K. There are other factors that affect the balance of the Basin Fund; however, they are outside the scope of the actions analyzed in the LTEMP EIS and would not be affected by the LTEMP alternatives. It should also be noted that consideration of unacceptable adverse impacts on the Basin Fund is one of the considerations included in Section 2.2.4.3 as part of the implementation process for experiments that may affect hydropower.

6.1.5 Consideration of Impacts on Basin Fund

<u>Summary Comment</u>: Impact on Upper Colorado Basin Fund should be a secondary consideration when selecting a preferred alternative.

<u>Response</u>: DOI considers the Upper Basin Fund important to the LTEMP given its role in funding environmental activities. Consideration of unacceptable adverse impacts on the Basin Fund is one of the considerations included in Section 2.2.4.3 as part of the implementation process for experiments.

6.1.6 Power System Impacts on Climate Change

<u>Summary Comment</u>: Potential climate change impacts on LTEMP alternatives were not incorporated in the power systems analysis.

<u>Response</u>: Climate change effects on hydropower generation results are presented in Section 4.16.3. The LTEMP climate change analysis was based on the Basin Study (Reclamation 2012), which was peer reviewed and is the best available science regarding climate change projections in the Colorado River Study. The hydrological traces generated for the Basin Study were utilized as described in Section 4.16.1.2 and Appendix C. The Basin Study used 112 climate change hydrologic traces generated as part of the Basin Study, however due to insufficient data to drive the complex suite of models, we used a subset of 21 traces out of the 112 for the LTEMP EIS. These 21 reconstructed historical hydrological traces were reweighted based on the 112 traces to represent the best current understanding of what might happen because of climate change. Based on this analysis, the weights on the hydrological traces had a small effect on the aggregate performance but did not change the rankings of alternatives. This result suggests the alternatives are robust to uncertainty about climate change.

6.1.7 Sensitivity Analysis of Fuel Prices

<u>Summary Comment</u>: A sensitivity analysis using a range of projected utility fuel prices should have been conducted because natural gas prices are currently very low.

<u>Response</u>: Text was added to the EIS text acknowledging the possible effects of the currently low natural gas prices on results.

6.1.8 Clean Powerplant Legislation

Summary Comment: Clean powerplant legislation was not considered in the analysis.

<u>Response</u>: Information about clean powerplant legislation is included in the cumulative impacts Section 4.17, but that legislation is currently in draft form. In February 2016 the Supreme Court ordered EPA to stop enforcement until a lower court rules on a lawsuit. A ruling on the lawsuit will most likely not come until after the 2016 presidential election and enforcement may depend upon the results of the election as well. Because specifics are unknown at this time and it may be

tied up in litigation for many years, it was not appropriate to analyze it further than this in this EIS.

6.2 GENERAL OVERARCHING COMMENTS ON THE POWER SYSTEMS ANALYSIS

6.2.1 Adequacy of Power Systems Analysis

<u>Summary Comment</u>: The analysis is strongly biased against hydropower resources, incomplete, and significantly underestimates costs in all cases using simplified models and metrics for hydropower resource rendering it inadequate.

<u>Response</u>: The commenter has provided no evidence to support these conclusions. Therefore, we have no basis to determine whether or not they are accurate. We do not believe that the analysis is biased against hydropower resources, is incomplete, or underestimates costs. LTEMP used economic measures that are consistent with the ones used in other Upper Colorado River Basin environmental studies. The level of technical detail that was used to support the computation of these measures has in general exceeded the ones used in other studies, some of which only applied the GTMax-lite model. For the LTEMP analysis we also used the AURORA model to explore grid-wide interactions. All major assumptions are both noted and documented and rely on publically available sources such as those published by the U.S. Energy Information Administration (EIA) and the Federal Energy Regulatory Commission. There were several sensitivity analyses included in Appendix K to explore possible additional sources of costs. In response to utility cooperator comments on the DEIS, an additional analysis was added to Appendix K looking at a different mix of powerplant replacement technologies which could result in a higher estimate of capacity replacement costs. The text in Chapter 4 Section 4.13 was also updated to include a summary of those sensitivity analyses.

6.2.2 Ratepayer Analysis Flawed

<u>Summary Comment</u>: The retail ratepayer analysis that was included within the draft LTEMP EIS report is flawed because the agencies that conducted the analysis do not provide retail electric service, do not deal with issues that are directly related to the provision of retail electric service, do not have expertise with respect to retail electric rate making, and did not consult with retail electric service providers such as SRP when conducting the analysis.

<u>Response</u>: Retail rate analyses were primarily conducted by Edward Bodmer who led retail electricity rate and utility financial viability analyses for Western's Electric Power Marketing EIS. He is an experienced financial consultant who develops complex corporate project finance and simulation models. He provides expert testimony on financial issues before regulatory agencies and has taught customized in-house courses for Shell Oil, General Electric, HSBC, Citibank, CIMB, Lindlakers, Saudi Aramco and many other energy and industrial clients around the world. He also worked for a regulatory commission and has testified approximately 40 times on rate making issues both from a financial perspective and a rate allocation perspective. The rate impacts were reviewed by experts in the field who had no problems with the overall approach. Formerly, Bodmer was the Vice President at the First National Bank of Chicago, where he directed created financial modeling techniques used in advisory projects. With respect to the comment that we did not consult with entities such as SRP, this is simply incorrect. We received comments, letters and made presentations where ratemaking entities were present. In addition, we did extensive research on the financial structure of the entities and reviewed financial data in the context of rate setting. In addition, the ratepayer methodology and results report were peer reviewed by three independent reviewers as well as by GCMRC.

6.2.3 Inputs to Capital Costs Analysis

<u>Summary Comment</u>: Capital costs should have been supplied by CREDA members since they actually build new powerplants and entities such as Argonne and Energy and Environmental Economics, Inc. (E3) do not.

<u>Response</u>: LTEMP assumptions regarding the characteristics of all capacity expansion candidates including capital investment costs are noted and documented in K.1.6.2. These costs are based on Energy Information Administration (EIA) data that are used for the EIA Annual Energy Outlook. This section discusses these costs and compares these to other estimates including those used by Western Electricity Coordinating Council (WECC) and those supplied by CREDA for a previous Upper Colorado River Basin EIS. LTEMP hydropower analysts are both well qualified and have extensive experience in applying these costs for economic analyses. We note that a very similar approach that also used EIA data sources was used by E3 for a study conducted for WECC.

As stated in Chapter 4, Section 4.13.1 "The LTEMP hydropower resources impact analysis was largely an economic analysis rather than a financial analysis. A financial analysis focuses on the revenues and costs accrued by a particular entity, including transfer payments, such as power transactions, taxes, and insurance. It also includes payments made by individual entities for previous investments. In contrast, an economic analysis focuses on societal costs and benefits. Transfer payments among entities are excluded because the total net change to society of these transactions is zero; that is, the amount paid by the buying entity equals the amount received by the selling entity. Also excluded from economic costs are past investments, such as those to construct power plants, because these expenditures have already been incurred on society and cannot be recovered. Similar to other power systems EIS analyses performed by Argonne, the economic analysis performed for LTEMP estimates changes to the U.S. economy as the result of altering operating criteria at Glen Canyon Dam. These economic costs include expenditures to build and operate new capacity in the future to replace Glen Canyon Dam Powerplant lost capacity and both fuel and variable operation and maintenance (O&M) costs associated with altering the dispatch of Western Interconnection generating units. A financial analysis was performed for the LTEMP EIS to estimate the wholesale (see Section 4.13.1.2) and retail rate impacts (see Section 4.13.1.3) on individual affected entities (e.g., individual FES utilities and their retail customers)."

6.3 LEGAL ISSUES

6.3.1 HFEs as Part of Long-Term Operations

<u>Summary Comment</u>: Bypass releases associated with HFE's that are conducted on a routine basis as part of a long-term operational plan are inconsistent with the "Law of the River" and that have not been addressed from a legal perspective.

<u>Response</u>: These releases are experimental and are being implemented in compliance with the Law of the River and specifically in compliance with laws related to the operations of Glen Canyon Dam. Experimental releases only change the monthly and hourly releases with annual volumes still complying with the LROC as currently implemented through the 2007 Interim Guidelines.

6.3.2 Irretrievable Loss of Hydropower

<u>Summary Comment</u>: Experiments cause an irretrievable loss of hydropower production, but there is no analysis included which would meet the requirement of 40 CFR 1506.1

<u>Response</u>: To the extent there is a loss of hydropower production in the future as a result of implementing the preferred alternative, this potential loss has been analyzed in the EIS. Analysis has been added to Section 4.17, and a discussion provided in Section 4.20.

6.4 TECHNICAL POWER SYSTEM ISSUES

6.4.1 Discount Rate

<u>Summary Comment</u>: Discount rate used in the analysis is too high and capital costs have not been escalated resulting in an underestimate of negative power impacts

<u>Response</u>: By policy, Reclamation is required to use the Principles and Guidelines (P&G) discount rate which is currently 3.375%. A rate of 1.4% was used in a sensitivity study, the results of which have been documented. Also, Reclamation P&Gs do not allow real capital cost escalation.

6.4.2 Capital Costs

<u>Summary Comment</u>: Capital costs are too low; all cost considerations for constructing a new generating plant were not included in developing capital and O&M costs.

<u>Response</u>: The economic cost to build a new generating unit (LTEMP costs) and the financial outlays that a utility incurs to build a new generating unit (CREDA costs) are fundamentally different. Utility outlays for taxes and insurance are excluded from an economic analysis because

these are transfer payments that are not part of the cost of construction. LTEMP did not include site specific costs that are advocated by CREDA because that level of detail is beyond a typical economic analysis for an EIS. It would be prohibitively time consuming and expensive to determine all possible sites for power plant replacement and all of the costs associated with those sites, and that level of analysis has never been performed for similar EIS documents on the Colorado River. EIA costs are based on typical values that we adjusted for broad regional construction costs considerations such as material and labor. In response to utility cooperator comments on the DEIS, an additional analysis was added to Appendix K looking at a different mix of powerplant replacement technologies which could result in a higher estimate of capacity replacement costs.

6.4.3 Exceedance Level

<u>Summary Comment</u>: There are concerns about exceedance level assumptions used to estimate firm Glen Canyon Dam Powerplant capacity.

<u>Response</u>: An analysis of Western's current marketing commitments shows that the assumption of 90% exceedance level used in the LTEMP EIS is consistent with their marketing planning assumptions. A 90% exceedance level means that there is up to a 10% risk of those commitments not being met in a given year. To address these concerns about future marketing commitment levels, a sensitivity study was performed on 50%, 90%, and 99% exceedance levels. Results are shown in Appendix K.1.10.4 and summarized in Chapter 4.13. These results generally conclude that lost Glen Canyon Dam powerplant capacity under all alternatives except for Alternative B have the highest capacity loss at the 50% exceedance level that progressively decreases through the 99% exceedance level (see Figure K.1-32 in Appendix K). We therefore performed an economic cost sensitivity analysis on these two levels to quantify reasonable cost "bookends."

6.4.4 Ancillary Service Assumptions

Summary Comment: There are concerns about ancillary service assumptions.

<u>Response</u>: Cooperators were concerned that using a lower ancillary service requirement could significantly affect Glen Canyon Dam dispatch, firm capacity levels, and economic evaluations. A simplified sensitivity analysis was therefore performed to gain a general appreciation for the impact of ancillary service assumptions on model outcomes. The analysis was performed on Alternatives A, D, and F under two disparate ancillary service market assumptions. Results should that difference in the combined net present value of capacity and energy was 0.14% or less. Under all cases, this small economic difference was primarily due to the existence of significant levels of "unused" Glen Canyon Dam Powerplant capacity which could be used to fulfill ancillary service obligations. This analysis is documented in Appendix K.1.10.8.

6.4.5 Modeling Assumptions

<u>Summary Comment</u>: There were questions about the modeling assumptions regarding locational marginal prices (LMP) responses to changes operations at Glen Canyon Dam and the Western Electricity Coordinating Council study (WECCi-leaks study).

<u>Response</u>: Text has been edited for clarity and both figures and reference to the WECCi Leaks study have been removed as it was not used as part of the LTEMP analysis.

6.4.6 Small Customer Representation

<u>Summary Comment</u>: The representation of small customers is unrealistic and flawed. Small customers are more limited in their choices of suppliers due to transmission and resources and do not have the financial resources or the load to build new facilities to replace capacity and energy losses.

<u>Response</u>: The representation of small customers is neither unrealistic nor flawed. For the power systems analysis (both expansion and dispatch modeling), we aggregated the small customers together for precisely the reasons outlined in the comment; namely, that small customers could not build their own facilities and are dependent upon other suppliers and power purchase agreements.

6.4.7 Small Customer Loads and Load Growth

<u>Summary Comment</u>: The representation of small customer loads and load growth rates are incorrect.

<u>Response</u>: Small customer load assumptions and simplifications are fully explained and disclosed. Because hourly generation levels were not readily available from publically available sources, we used normalized load profiles from other entities as approximations. Normalized load profiles express loads in terms of a fraction of the peak load such that the peak load is set equal to 1.000 and all other loads are less than 1.000 (i.e., hourly load/peak load). These load shapes were scaled to match estimated historical total load based on information that was collected for the retail rate 40 payer analyses. For example, the UAMPS normalized load profile was used to represent small customers in the west. It was selected because most of the 44 UAMPS members are small municipal systems that serve small to medium cities and towns that have attributes which are similar to many of WAPA's small customers. We judged that this modeling approach produces a reasonable approximation of aggregate small customer loads. In addition, errors occur equally under all alternatives and therefore probably have minimal effects on the comparative results.

6.4.8 System Expansion Plan

<u>Summary Comment</u>: Assumptions regarding a cooperative SLCA/IP system-wide expansion plan are overly simplistic and results in an underestimate of power impacts.
Glen Canyon Dam Long-Term Experimental and Management Plan Final Environmental Impact Statement

Response: Text revisions have been made for clarification. Power impacts have not been underestimated because this level of modeling and simplification is consistent with and improves upon previous EISs conducted for Glen Canvon Dam, such as the 2007 interim guidelines. Due to the complex nature of the grid, all power systems models with scopes that cover a large geographical area over a multiple year time span, such as the one conducted in support of LTEMP, need to make simplifying assumptions in order for the problem to be tractable. For LTEMP we tailored models and methods to meet study objectives; that are to rank alternatives in terms of economic costs and to gain an appreciation for relative costs among alternatives. As detailed in Appendix K, we performed sensitivity analyses on several key modeling assumptions to learn about the effect of these assumptions on model results and to obtain a reasonable estimate of potential outcomes. We also applied methods and models, such as AURORA, that were more detailed than those used for previous EIS's conducted on power facilities in the Colorado River Basin. We observed that the ranking are very robust changing little as a function of modeling assumptions. We also observed the GTMax-lite model that strips away many of the modeling complexities embedded within AURORA lead to very similar alternative rankings. Thus, we made assumptions and implications at the level appropriate for this type of analysis and have fully disclosed those assumptions in the EIS. We also discussed the possible implications of this simplification of model results. The document also includes several examples of joint powerplant/unit ownerships that currently exist in the study region. The power analysis methods, assumptions and results have been peer reviewed by three external experts and the peer review results are available on the LTEMP website.

6.4.9 Transmission Constraints

<u>Summary Comment</u>: Transmission system constraints on power exchanges in the region should have been included in the analysis.

<u>Response</u>: Detailed transmission system constraints on power exchanges in the region were not explicitly modeled because it would have significantly increased the size and complexity of the problem and it was not expected to have a significant impact on comparative results among alternatives. It is also consistent with previous EISs conducted for Glen Canyon Dam that also did not explicitly represent the transmission system. We did however include on- and off-peak costs on AURORA modeled energy flows (i.e., power transactions) among customers based on average historical transmission rates. These costs tend to dampen bulk power transactions among energy trading entities. The end result was that AURORA estimated generation levels are very similar to historical levels by both fuel type and at the utility level. We have disclosed methodologies, assumptions, limitations and results which were peer reviewed. It is noted in the EIS that the WAPA contract rate of delivery (CROD) will be identical under all alternatives. If feasible, customers can utilize the CROD to schedule the delivery of both WAPA energy and other energy transactions up to the contractual limit.

6.4.10 Renewable Portfolio Standards

<u>Summary Comment</u>: SLCA/IP customers such as SRP are not required to comply with State Renewable Portfolio Standards (RPS).

<u>Response</u>: We acknowledge that RPSs are not required of all LTF customers. SRP and Tri-State do have an internal program to set a target for building significant amounts of renewable or sustainable resources.

6.4.11 Costs of HFEs

<u>Summary Comment</u>: Costs to conduct experiments such as HFEs are underestimated because trading partners would take advantage of Western's inflexible situation during experiments.

<u>Response</u>: We acknowledge that the prices that WAPA is offered from its trading partners during HFEs sometime depart from overall market prices. This has financial implications for WAPA. The power systems economic analysis does not consider these financial transactions because there are no system-level economic impacts. The savings gained by entities that purchase at below market prices are exactly offset by financial losses for entities that sell below market price. The net economic impact is therefore zero.

6.4.12 Costs of Experiments

<u>Summary Comment</u>: The costs to conduct experiments are too low and experiments such as low summer flow would have a devastating impact on power.

<u>Response</u>: The analysis shows that a low summer flow experiment would not significantly, if at all, change annual hydropower production during low summer flow years. There would be an economic impact however with a decrease in NPV between \$13.93 million and \$21.01 million. Low summer flows are projected to occur infrequently; under the preferred alternative, low summer flow would be expected to be tested once in the second 10 years of the LTEMP period.

6.4.13 Impacts of Macroinvertebrate Production Flows on Hydropower

<u>Summary Comment</u>: Macroinvertebrate production flows may cause serious negative impacts on power and should be conducted in a way that is least damaging to hydropower.

<u>Response</u>: After the DEIS was published, hydropower assumptions and the modeling of macroinvertebrate production flows were revised based on input from WAPA, the results of which show a positive benefit to hydropower capacity. Revised results are consistent with analyses conducted by WAPA regarding a positive net benefit to power.

6.4.14 Effects at Hoover Dam and Glen Canyon Dam

<u>Summary Comment</u>: Concerns were raised about potential perceptions regarding changes in higher Lake Mead elevations and Hoover powerplant "benefits" offsetting Glen Canyon Dam economic impacts.

<u>Response</u>: To address these concerns the description of Hoover Dam impacts were substantially revised. The FEIS no longer refers to changes in the NPV of Hoover as a "benefit" or as "offsetting" alternative costs at Glen Canyon. We also indicate that the economics of Hoover and Glen Canyon are not directly comparable because the analyses used different levels of modeling details. Instead we indicate that the Hoover analysis was performed primarily to disclose the general direction and relative magnitudes of Hoover impacts.

6.5 RETAIL RATE ANALYSIS

6.5.1 Consideration of Wholesale Rate Input

<u>Summary Comment</u>: Retail electric rates were computed without accounting for wholesale rate charged by Western

<u>Response</u>: Wholesale rates were not used because: (1) the wholesale rate study was completed after the retail rate analysis; (2) through directly using data from the power systems model the direct connection between power system costs and rate impacts could be observed and the process was transparent; (3) the capital cost of constructing new capacity incurred by utility systems (such as Tri-state) was directly estimated rather than assuming Western would carry the burden of replacing the capacity. In addition, modeling was peer reviewed by three independent reviewers as well by GCMRC.

6.6 WHOLESALE RATE ANALYSIS

6.6.1 Assumptions

<u>Summary Comment</u>: Concerns regarding the bookends marketing assumptions selected for the wholesale rate analysis

<u>Response</u>: Wholesale rate bookends were chosen because they were thought to represent a reasonable range for the marketing of SLCA/IP power in the post-2024 period and illustrate that the effect of the SLCA/IP rates from the LTEMP EIS alternatives are, in part, driven by sales of AHP energy and capacity.

6.7 EDITORIAL COMMENTS

Numerous substantive editorial comments were submitted on hydropower sections in the DEIS including Sections 3.13, 4.13, and Appendix K. Where appropriate, the text of the EIS was modified to address the specific comments.

7 NATURAL PROCESSES

7.1 NATURALLY PATTERNED FLOW AND RUN-OF-THE-RIVER FLOW ALTERNATIVES

<u>Summary Comment</u>: One commenter suggested that the LTEMP should improve habitats by attempting to recreate the shape of the historic hydrograph by generally increasing water levels until June followed by a gradual ramp down to a lower level between September and February, with a possible late summer spike to simulate late-summer monsoons. The daily fluctuations with dramatic step-ups and step-downs in flow that now occur almost never occurred before the dam was in place.

<u>Response</u>: "Naturally patterned flow" and "Run-of-the-river" alternatives were considered but dismissed in the draft EIS (see Sections 2.3.2, 2.3.11).

7.2 REMOVE NATURAL PROCESSES FROM THE EIS

Summary Comment: One commenter wanted natural processes deleted from the EIS.

<u>Response</u>: No text change made. The natural processes goal was based on several of the Desired Future Conditions (see Appendix A of the EIS) and on NPS management policy.

7.3 NATURAL RESOURCE PERFORMANCE GOAL

7.3.1 Quantitative Performance Metric

<u>Summary Comment</u>: Several commenters stated that a quantitative performance goal should have been established for natural processes, possibly comparing alternative flows with pre-dam flows.

<u>Response</u>: The best available science and models were used in this EIS. There was not adequate information to develop a quantitative performance metric for pre-dam versus post-dam conditions in a consistent manner to facilitate comparison across alternatives for natural processes.

7.3.2 Goal to Restore Is Not Achievable

<u>Summary Comment</u>: A few commenters wanted the following sentence deleted: "For the LTEMP, the analogous natural processes resource goal is to 'restore, to the extent practicable, ecological patterns and processes within their range of natural variability, including the natural abundance, diversity, and genetic and ecological integrity of the plant and animal species native to those ecosystems'." It is not possible to operate Glen Canyon Dam in a manner that could

fully restore natural processes and their drivers to those that occurred under unregulated conditions

<u>Response</u>: No text change made. The natural processes goal was based on several of the desired future conditions (see Appendix A of the EIS) and on NPS management policy. The EIS recognizes that management of dam operations and any subsequent effects on natural processes may not act to fully restore natural resources in the system. Rather, the EIS compares alternatives with regard to how alternative-specific temperature, flow, and sediment conditions may affect processes that depend on these conditions, and possible subsequent responses in natural resources, including not only maintenance of native biota in their current conditions, but also enhance or improve habitats and thus benefit associated biota.

7.3.3 National Park Service Natural Processes Management Policy and Goals

<u>Summary Comment</u>: One commenter noted that an important objective of management of the Colorado River Ecosystem is the ability to sustain healthy populations of native plants and animals and natural ecological processes. NPS management policies state that (1) "whenever possible, natural processes will be relied upon to maintain native plants and animals and influence natural fluctuations in populations of these species" and (2) "the Service ... will try to maintain all components and processes of naturally evolving park ecosystems, including the natural abundance, diversity, and genetic and ecological integrity of the plant and animal species native to those ecosystems" (NPS 2006b). For the LTEMP, the analogous natural processes resource goal is to "restore, to the extent practicable, ecological patterns and processes within their range of natural variability, including the natural abundance, diversity, and genetic and ecological integrity of the plant and animal species native to those that occurred under unregulated conditions." Since management policies are not dictum; why can they not be changed? The commenter further states that statements such as "rely on natural processes" and "try to maintain" are much different from "restore".

<u>Response</u>: Changing NPS Policy is outside the scope of the EIS. The EIS recognizes that management of dam operations and any subsequent effects on natural processes may not act to fully restore natural resources in the system. Rather, the EIS compares alternatives with regard to how alternative-specific temperature, flow, and sediment conditions may affect processes that depend on these conditions, and possible subsequent responses in natural resources, including not only maintenance of native biota in their current conditions, but also enhance or improve habitats and thus benefit associated biota

7.4 OPPORTUNITY TO EXPAND NON-FLOW MANAGEMENT OPTIONS TO IMPROVE AQUATIC AND RIPARIAN HABITAT

<u>Summary Comment</u>: A commenter noted that Glen Canyon Dam is presently functioning to protect native aquatic and riparian species from upstream and downstream threats. However, the very existence of the Dam is at odds with the National Park Service (NPS) Organic Act and the enabling legislation of Grand Canyon National Park. This means that the management objectives

of the NPS at Grand Canyon do not really include using the Dam to foster aquatic and riparian habitat rehabilitation. Reclamation and the NPS could be doing much more to improve riparian habitat quality, particularly now that much of the non-native tamarisk habitat along the river is declining due to tamarisk leaf beetle introduction. Collaboration of these two agencies in this DEIS and the Record of Decision is an opportunity to greatly expand non-flow management options to improve aquatic and riparian habitat area and quality.

<u>Response</u>: The LTEMP is intended to identify operational and experimental management alternatives specifically for protecting and enhancing aquatic, riparian, and terrestrial habitats and their associated biota, as well as protecting cultural resources and balancing human uses in the system. Alternatives considered in the EIS include flow conditions that may support improvements in riparian and aquatic habitats. The LTEMP also includes non-flow management options, such as mechanical trout removal, that may benefit native aquatic biota in the system. All the action alternatives also include a non-flow vegetation treatment experiment to address riparian vegetation including removal of invasive species and replanting of natives. The agencies have been working together, and expect to continue that collaboration beyond the ROD for protecting and improving natural and cultural resources while providing for cultural, recreational, and commercial uses and activities in the Grand Canyon National Park.

7.5 HFES SHOULD BE INCLUDED IN THE EVALUATION OF EFFECTS TO NATURAL PROCESSES

<u>Summary Comment</u>: A couple of commenters stated that since HFEs have a direct effect to fisheries, plants, animals, and water quality, they should be considered in the effects on natural processes.

<u>Response</u>: HFEs were considered in the effects on all resources in Chapter 3, as well as on Natural Processes.

8 PROCESS AND POLICY

8.1 INCLUDE THPOS IN CONTACT LIST

<u>Summary Comment</u>: Commenters on this issue requested that THPOs be included in the list of groups to be contacted.

<u>Response</u>: Text has been modified as suggested and THPOs have been added to the list of groups to be contacted.

8.2 TRIBES INVOLVEMENT IN ALTERNATIVES DEVELOPMENT

<u>Summary Comment</u>: A commenter on this issue indicated that Tribes should have been invited to participate in alternative development like the Arizona Game and Fish Department was.

<u>Response</u>: Tribes were invited to be Cooperating Agencies in the LTEMP EIS process – see Section 5.1.3 for the process that was used to inviting and communicating with Tribes. Many meetings were held with Tribes throughout the alternative development process, the modeling development and results, the structured decision analysis, and the draft writing and editing to allow for input and comment. Meetings are still continuing with Tribes on the programmatic agreement and on a non-native fish agreement related to the LTEMP. Several traditionally associated Tribes are part of the AMWG process as well as being cooperators, and so were part of the LTEMP process through many meetings and webinars provided through that process.

8.3 SIERRA CLUB'S SCOPING COMMENTS

<u>Comment Summary</u>: The commenter on this issue requested that Sierra Club's scoping comments be added to the official public record and requested that their comments be addressed in the Final Environmental Impact Statement.

<u>Response</u>: We apologize that the Sierra Club scoping letter was inadvertently left out of the initial scoping report that was made available on the LTEMP website. Thank you for calling this to our attention in your comments on the DEIS and we have now added your scoping letter to that report and made the revised version available on our website. Your letter was reviewed and considered in the process of drafting the LTEMP EIS. Below is a summary of the issues you presented in your scoping letter and how we have addressed them in this EIS process.

 Introduction – concern regarding declining conditions since the dam has been in place – 95% sediment loss, nutrient base, dissolved oxygen, temperature reduction, changes to flood regime, extirpated species, increasing non-native plant species.

<u>Response</u>: This list of concerns were considered in the preparation of the EIS. Chapter 3 of the EIS describes changes that have occurred in the Colorado River Ecosystem since construction of the dam.

2a) Legal Authorities - hydropower generation should be removed from the need statement.

<u>Response</u>: The purpose and need statement clearly articulates the responsibilities of the Secretary of the Interior under the Grand Canyon Protection Act (GCPA); the goal of which is to preserve, mitigate adverse impact to and improve the values for which Grand Canyon National Park and Glen Canyon National Recreation Area were established, including, but not limited to natural and cultural resources and visitor use. The purpose and need and alternatives evaluated in the EIS take direction from the GCPA and attempt to develop an alternative that meets the requirements of the GCPA and applicable Federal laws, regulations, and policies. In addition to Section 1802(a) of GCPA, the Secretary is required to follow numerous other Federal laws, including the 1956 Colorado River Storage Project Act. Both Section 1 quoted in the letter, and Section 7, direct the Secretary to generate the greatest practicable amount of power at Glen Canyon Dam. This requirement is also found in Section 1802(b) of the GCPA. Therefore, removal of power generation from the GCPA.

2b) Scientific Framework – Full analysis of downstream resources. GCMRC involvement. Clear experimental framework, thresholds, triggers. Implement recovery of native species, and analyze alternatives that experiment with sediment augmentation, temperature modification and restoration of the natural hydrograph.

Response: The LTEMP EIS provided a full analysis of downstream resources that could be affected by the proposed action and alternatives. GCMRC was involved in the development of alternatives, in developing modeling approaches, in subject matter expert meetings throughout the process, in reviewing and editing the EIS, and in peer review of models and finally in the development of a science plan to guide the implementation of LTEMP once a final preferred alternative and ROD are in place. The reintroduction of extirpated species was outside the scope of the LTEMP as stated in Section 1.5.3. Early in scoping, we did consider some alternative concepts with new infrastructure, but determined that consideration of new infrastructure on the dam such as sediment augmentation and temperature modification were outside of the scope as well as economically infeasible at this time as stated in Section 1.5.3. In early scoping we did evaluate several alternatives that would have resulted in dam releases intended to mimic the natural hydrograph. However, preliminary modeling indicated several of those had unacceptable adverse resource impacts, particularly to sediment. Seasonally Adjusted Steady Flows was the most natural hydrograph in an alternative and was carried all the way through as Alternative G. The preferred alternative does incorporate two types of spring releases (sediment triggered and proactive spring HFEs) which provide peak flows that mimic the natural hydrograph to some degree.

3) Modification of Purpose and Need

Response: Please see the answer to Legal Authorities comment.

4) Impacts to Natural Resources – Sediments and Soils

<u>Response</u>: Sediment conservation was fully considered and analyzed in the LTEMP EIS using the best available science and modeling. See Sections 3.3 and 4.3. All of the action alternatives

included high flows of various types and frequencies. The effect of the flow regimes on natural processes and native species was considered in the analysis based on the best available information.

5) Impacts on Natural Resources – Threatened, Endangered and Sensitive Species

<u>Response</u>: Endangered, threatened and sensitive species were considered in the LTEMP EIS analysis based on the best available science and modeling. See Sections 3.5, 3.7, 4.5, 4.7.

6) Unsuccessful Native Fish Recovery

<u>Response</u>: Temperature and flow effects of the preferred alternative on native fish were evaluated in the LTEMP EIS. The preferred alternative includes a spring high flow, called a proactive spring HFE, which would occur generally at a time that the historical record indicates was when spring flooding in Grand Canyon typically occurred. Though proactive spring HFEs have been designed for sediment conservation purposes, they may provide experimental data through GCMRC monitoring efforts on effects to native fish. Though all of the alternatives are expected to have relatively similar effects on native and endangered fish, the preferred alternative includes several actions to manage nonnative fish and experiments designed to improve aquatic invertebrate populations in order to benefit native fish. The preferred alternative performed slightly better than No Action on several native and endangered fish metrics.

7) Unsuccessful Adaptive Management Program: The unsuccessful "Modified Low Fluctuating Flow Alternative" (MLFF) enacted in the Adaptive Management Program has not achieved the desired results of protecting the Canyon's resources including the beaches, the native fishes, and the cultural sites. New alternatives should be developed by the Grand Canyon Monitoring and Research Center to address scientific hypotheses regarding the best possible alternative for sediment retention (i.e., Wright et al. 2008) and mimicking the natural historic hydrograph.

<u>Response</u>: GCMRC helped in the development of alternatives and a seasonally adjusted steady flow alternative was considered as Alternative G. Several alternatives, including the preferred alternative, performed better than No Action on beach and sandbar building metrics.

8) Cumulative Impacts

<u>Response</u>: Cumulative impacts to vegetation, fish, wildlife and endangered species were analyzed in the LTEMP EIS (see Section 4.17). The EIS incorporates elements of, and will supersede, the HFE Protocol and Non-Native Fish Removal Protocol that have been in place since 2012.

9) Impacts on Natural Resources – Water Quality: Restoration of water quality must be addressed in the EIS taking into account the effects of different release structures and their effects on downstream ecology. The EIS should also consider salinity levels, water temperature, turbidity and suspended sediment, nutrients and dissolved oxygen concentrations. An additional critical factor that must be considered is the impact of drought on the quality of dam releases. For example, when Lake Powell is lower, water releases are warmer than, and oxygen levels differ from, water released from a fuller lake

<u>Response</u>: Impacts to water quality in the Colorado River and reservoirs were considered in the LTEMP EIS. See Sections 3.2 and 4.2. Climate change was analyzed with respect to hydrological effects in Section 4.2. It should be noted that the LTEMP did not affect levels of Lake Powell and Lake Mead at an annual or great time scale; the scope of the LTEMP includes monthly, daily and hourly releases from Glen Canyon Dam, but not annual volumes from Glen Canyon Dam, so the effects to water quality were considered within this scope.

10) Impacts of Reduced Flows Associated with Climate Change: The LTEMP must specifically create a plan for providing water to the Colorado River in Grand Canyon during extended drought periods.

<u>Response</u>: Climate change was considered within the LTEMP EIS. The 2012 Colorado River Basin Study provided data for the EIS, and the modeling included a set of modeling runs to address climate change concerns. This climate change modeling did not alter the pattern of performance of the alternatives and the preferred alternative still performed well when considered in the context of projections regarding future climate change. The scope of the LTEMP includes monthly and hourly releases from Glen Canyon Dam, but not annual volumes from Glen Canyon Dam, so the scope of the analysis was considered with respect to for monthly and hourly changes.

11) Impacts on Natural Resources - Terrestrial Wildlife

Response: Impacts to terrestrial wildlife were considered in the LTEMP EIS in Section 4.7.

12) Impacts on Natural Resources - Vegetation

<u>Response</u>: Impacts to vegetation were considered in the LTEMP EIS in Section 4.6. There was considerable discussion between joint lead staff and GCMRC about different flows to benefit native vegetation species. We discussed the timing and scouring effects of various water flows that could disadvantage nonnative species. Modeling was performed using the best available vegetation model and GIS mapping information and the preferred alternative performed better than the No Action alternative for maintaining native vegetation condition. However, under all alternatives, vegetation condition is expected to decline over time. Because of this, a feature common to all of the action alternative sanalyzed in this EIS is an experimental vegetation treatment project to address nonnative vegetation removal, native vegetation planting and other vegetation issues. This was developed in coordination with GCMRC and after discussions with other partners and Tribes.

13) Commenter Recommendations

Recommendation: Restore essential sediment and nutrients into the main stem of the Colorado River in the Grand Canyon.

<u>Response</u>: Sediment conservation was a resource goal as expressed in Section 1.4. More even monthly volumes and frequent HFEs are components of the preferred alternative to address sediment conservation.

Recommendation: Restore flow regimes to properly transport the sediment and nutrients within Grand Canyon, when and where it belongs. The 1994 Biological Opinion and the EIS referred to these as Seasonally Adjusted Steady Flows (USDOI 1995)

<u>Response</u>: The LTEMP EIS evaluated a seasonally adjusted steady flow option, Alternative G, however it was not identified as the preferred alternative as Alternative D showed a better balance to protect downstream resources.

Recommendation: Restore the seasonally variable water temperature in the main stem of the Colorado River through Grand Canyon. The 1994 Biological Opinion and the EIS referred to this as Selective Withdrawal by means of a Temperature Control Device (USDOI 1995). Look to the Temperature Control Device at Flaming Gorge Dam as a source of information and experimentation to guide planning at Glen Canyon Dam. Answer the question of "whether the potential benefits to the endangered fish of operating a TCD and warming the water outweigh the potential adverse effects from potential increases in nonnative predators, parasites and diseases, or other unintended, systemic interactions in the downstream environment." (USDOI USBOR 2007) It is entirely possible – even probable -- that native fish will perform better in the temperature environment in which they adapted (i.e., Clarkson and Childs 2000).

<u>Response</u>: Flow options for affecting temperature were considered as part of the LTEMP and the preferred alternative includes an experimental option for low summer flows. Temperature control devices were outside the scope of this project.

Recommendation: Implement a restoration and recovery program for the Colorado River corridor in Grand Canyon that includes the recovery of all species known to be native to Grand Canyon prior to the operation of Glen Canyon Dam.

<u>Response</u>: The reintroduction of extirpated species was outside the scope of this EIS and is discussed in Section 1.5.3. LTEMP EIS alternatives include experiments that are intended to improve conditions for ESA-listed species and other special status species. The LTEMP EIS team used the best available science and peer-reviewed modeling to determine the potential effects of these alternatives on these species. DOI has worked closely with FWS throughout the 5-year EIS process to ensure that the appropriate experiments, dam operations, and non-flow actions were identified as conservation measures. Formal consultation with FWS has resulted in a Biological Assessment and Biological Opinion on the effects of the proposed action on listed species and designated critical habitat, as presented in Appendix O of the EIS.

The original notice of intent to prepare the LTEMP EIS identified the need to determine whether to establish a recovery implementation program for endangered fish species below Glen Canyon Dam. The LTEMP team found that identifying the need to determine whether to establish a recovery implementation program (RIP) for endangered fish species below Glen Canyon Dam did not meet the purpose and need for the action.

Recommendation: Implement a non-native eradication program to minimize alien species in the Grand Canyon river corridor with a priority on those that prey on, compete with, or otherwise impair the health of native plants and animals. Non-native fish retention should not be a priority in designing new flow regimes.

<u>Response</u>: The LTEMP EIS included and evaluated a number of tools for address non-native fish and non-native vegetation. Some Tribes, particularly Zuni and Hopi, have concerns regarding non-native fish removal methods, so the preferred alternative contains some constraints and mitigation measures to address those concerns.

Recommendation: Complete the Little Colorado River Management Plan as recommended by the 1994 Biological Opinion (USDOI 1995).

Response: Development of this plan is considered outside the scope of the LTEMP EIS.

Recommendation: Address the dysfunction of the Adaptive Management Program. The AMP should be replaced by an open source and independent body of research and advisory scientists, where the monitoring and research data are consistently and thoroughly peer-reviewed prior to providing any recommendation to the Secretary of Interior. The Grand Canyon Research and Monitoring Center should have a significant role in creating the LTEMP, including choosing testable hypotheses, designing best-case flow scenarios to retain sediments and native species in Glen and Grand Canyons, and crafting monitoring regimes.

<u>Response</u>: We assume the commenter is referring to the Adaptive Management Work Group (AMWG) when they refer to the AMP. The GCMRC was involved throughout the LTEMP process and assisted as subject matter experts in with the modeling, analysis, experimental design and assisted with peer review of aspects of the LTEMP. The AMWG is a FACA committee established through the GCPA and exists to provide input to the Secretary of the Interior. Changes to the organization of the committee should be directed to the Secretary. Changes to the structure and focus of the AMWG are outside the scope of this EIS.

Recommendation: Assess how the river could be managed with shrinking reservoirs and emphasize water conservation in long-term dam and reservoir management plans. A study by the National Research Council has indicated that long-term drought is the likely outcome of climate change in the Southwest (NRC 2007). The Bureau should consider at what point river management – specifically, water and power needs – would be better served by maximizing water storage in Lake Mead rather than dividing it between Mead and Powell reservoirs. In fact, an article released just this week raises tensions about keeping Lake Mead full see http://www.climatecentral.org/blogs/dry-weather-is-drawing-down-lake-mead/, accessed 1/31/12). The Bureau should assess the comparative loss of water from bank storage and evaporation between these two maintaining both reservoirs, maintaining only Lake Mead, and an alternative where Lake Powell is kept low to reduce "losses" to infiltration and evaporation.

<u>Response</u>: The LTEMP EIS evaluated a range of annual hydrologic inflow scenarios. It also evaluated a range of monthly, daily, and hourly releases from the Glen Canyon Dam while complying with the LROC as currently implemented through the 2007 Interim Guidelines. The "Fill Lake Mead First" alternative was dismissed as it was outside the scope of the LTEMP. We

also addressed the impacts of climate change in the EIS, which could result in "shrinking reservoirs" in the future.

Recommendation: An alternative that examines what would be required to generate a flow magnitude equal to the pre-dam annual or bi-annual average peak flow, and/or a flow that exceeds the current annual average flow by the same proportion as a pre-dam peak exceeded pre-dam average annual flow, should be included, so that such a flow could be planned when water is available.

<u>Response</u>: Flows greater than 45,000 cfs were considered but dismissed as part of this process for a number of reasons including operational and safety concerns.

Recommendation: If beach sizes continue to shrink, recreational capacity should be adjusted downward to prevent exacerbating beach erosion, damaging cultural sites, and damaging vegetation.

<u>Response</u>: The preferred alternative is expected to increase beach and sandbar size. Visitation is a separate process and not an issue within the scope of the LTEMP EIS. Visitation is set currently through the 2006 Grand Canyon Colorado River Management Plan.

Recommendation: Specific trigger points to guide management decisions must be outlined in the plan. If the dam is truly to be adaptively managed, then the LTEMP must identify detectable thresholds that, if reached, will cause a change in operations to be executed.

Response: The adaptive management process and triggers are adequately addressed in this NEPA document. Section 4.1 and Appendices C and K of the LTEMP DEIS have detailed information on the extensive modeling that was performed with a variety of long term strategies and referring to uncertainties explicitly. The long term strategies were variations of the alternatives that turned various experiments "off" or "on" to model different combinations and frequencies of experiments under a variety of hydrologic inflow conditions. This extensive modeling was coupled with extensive literature review and subject matter expert input to provide thorough analysis. This is sufficient for a NEPA analysis. Additional details of experiments and monitoring will be developed through the AMP annual work plans. Greater specificity regarding triggers for HFE implementation and humpback chub actions have been added to the EIS (Appendices O and P).

Recommendation: A healthy Colorado River in Glen and Grand Canyons must be the top priority of all alternatives proposed in the LTEMP, in order to meet the requirements of the GCPA. There is no evidence that Glen Canyon Dam can be operated to restore species and beaches to their predam condition. This plan should be devised as a last, best effort to restore the National Park Service lands downstream of Glen Canyon Dam. Planners should think holistically about what is required to achieve restoration goals, and create alternatives that aim for success. The cost of these alternatives may be high, but will represent the full cost of Glen Canyon Dam to the American public. We are subsidizing the services provided by Glen Canyon Dam by giving away parts of the Colorado River ecosystem for free, and we deserve to know what it will cost to replace all of it. Response: The GCPA is central to the purpose and need for this project.

Recommendation: Consider the non-use economic value of the Colorado River to the power and water market area when conducting economic analyses.

<u>Response</u>: A non-use economic study was completed as part of the LTEMP EIS process, however it was completed after the release of DEIS. The results were very comparable to the previous Welsh 1995 study. The results of the new study have been incorporated in the FEIS in Section 4.14.

Recommendation: This is an Experimental and Management Plan. Experimentation must be a component of all alternatives. That is, clearly defined hypotheses, ample monitoring and data analyses, and timely reporting need to be included in every alternative.

<u>Response</u>: Experimentation is included in the LTEMP EIS. Triggers for sediment experiments are clearly articulated. For several fish experiments triggers were developed and defined through the consultation process with FWS with information provided from GCMRC and other experts and cooperators. GCMRC also developed a science plan and will be intimately involved in the implementation of LTEMP experiments through the triennial budget process and on-going monitoring, evaluation and reporting.

Recommendation: The Grand Canyon Monitoring and Research Center should be involved in the development of dam operations alternatives and analyzing the pros and cons of different flow regimes. They should provide significant input into developing testable hypotheses, monitoring regimes, and determining thresholds that will trigger operational changes.

Response: GCMRC was involved throughout the LTEMP EIS process.

Recommendation: Glen Canyon Dam is now approximately 50 years old, and its right spillways have never been tested. We have now experienced climate fluctuations that have come very close to testing both the highest capacity and the lowest capacity of the dam for controlling water flow into the Lower Colorado River. There needs to be a plan for dam failure, either by overtopping or by extremely low lake levels. The safety of the dam, spillways, and adjacent sandstone walls should be examined in this analysis.

<u>Response</u>: Dam safety is a priority of the Bureau of Reclamation and is evaluated comprehensively through Reclamation's Dam Safety program on an -ongoing basis. These ongoing evaluations are wholly independent of LTEMP and beyond the scope of this NEPA analysis.

8.4 INVOLVEMENT OF COOPERATING AGENCIES AND SME TEAMS

<u>Summary Comment</u>: Commenters on this issue expressed concern regarding a lack of involvement by Cooperating Agencies on SME teams or in the analysis process, or requested more involvement in ongoing and future research and management of CRE resources.

Response: For the LTEMP EIS process there was a high level of involvement and interaction between the joint lead agencies, the Tribes, the cooperators and the stakeholders of the Adaptive Management Working Group (AMWG). There were also several opportunities for public input above and beyond what is required by NEPA. For over four years regular monthly calls were held with Tribes and cooperators, totaling over 50 meetings. Western Area Power Administration (WAPA) was involved in many additional meetings for alternative development and hydropower subject matter expert questions. Well over 150 meetings were held between the joint lead agencies, Tribes and stakeholders during the course of the development of the draft EIS. All cooperators, Tribes and AMWG stakeholders were provided the opportunity to comment on the purpose and need, objectives, resource goals, modeling metrics, modeling methods, and preliminary modeling results. All cooperators and Tribes were afforded review and comment on a preliminary draft of the EIS as well as the public draft EIS. The joint leads also held additional meetings with Tribes, held many consultation meetings with Tribal councils and provided Tribes the opportunity to submit text to fully reflect their perspectives on all resources within the EIS. Implementation details of experiments and monitoring tailored to the specific resource conditions of the year will be developed through the AMP and GCMRC annual work plans with communication and consultation with a number of partners and stakeholders.

8.5 LTEMP EFFECTS ON RESERVOIR OPERATIONS

<u>Summary Comment</u>: A commenter on this issue requested assurance that management actions and experiments proposed in the DEIS will not operate to modify determinations made under the Interim Shortage Guidelines for both Lake Powell and Lake Mead or to trigger a mid-year review of reservoir operations that would otherwise not occur.

<u>Response</u>: The scope of the LTEMP EIS was to evaluate a range of alternatives with different monthly, daily, and hourly releases from the Glen Canyon Dam. All alternatives are consistent with the LROC as currently implemented through the 2007 Interim Guidelines for annual release volumes. See Sections 3.2 and 4.2 for additional detail.

8.6 LIST RELEVANT LAWS AND MANAGEMENT DOCUMENTS

<u>Summary Comment</u>: A commenter on this issue stated that it is unclear as to how the extensive list of laws identified in the DEIS is specifically relevant to the proposed actions. The FEIS would benefit from identifying only those laws and management documents that are directly related and pertinent to the LTEMP process, or explaining why additional laws are included.

<u>Response</u>: Minor changes have been made. The text was revised to focus on the most relevant laws. There is a statement that these lists are not exhaustive.

8.7 APPLICATION OF STRUCTURED DECISION ANALYSIS (SDA)

<u>Summary Comment</u>: Some commenters questioned and did not support the use of SDA to help determine operations at Glen Canyon Dam, did not participate in the voluntary SDA process for

LTEMP, and rejected any reliance on SDA as the basis for operating Glen Canyon Dam to implement the LTEMP for the next 20 years. Some commenters stated that the DEIS relies on a flawed SDA and SDA should not be included in the DEIS as a basis for selecting a Preferred Alternative.

A commenter believed that the SDA overly relied on modelling outputs to claim objectivity and transparency when actual results depended heavily on the assumptions made in creating the models, inputs selected for running the models, and the manner in which the modeling analyses were made and results reported. The commenter believed that the DEIS relied on modeling outputs to justify a scientific and objective result without acknowledging the significant subjectivity that goes into creating and running the models in the first place.

A commenter believed that the DEIS is legally deficient because it relies on an inadequate and incomplete analysis of hydropower developed for the SDA process.

A commenter stated that the DEIS is Legally Inadequate because it Relies on Flawed Structured Decision Analysis. Specific to the hydropower Performance Metric, because of the varying degrees to which swings and weights are established for the various performance metrics, the hydropower metric is automatically given less weight than other metrics. The swing weighting is set up in such a way that sediment is weighted far more importantly than hydropower.

<u>Response</u>: In an effort to provide multiple opportunities for interested stakeholders to provide input in the LTEMP process, the National Park Service (NPS) and the Bureau of Reclamation (BOR) decided to incorporate facilitated structured decision analysis (SDA) into the LTEMP EIS process. The use of SDA in the LTEMP process was not required by NEPA, nor did it replace the NEPA impact analysis. Participation in the SDA process was a voluntary opportunity for stakeholder input. Text was modified to note that not all stakeholders were in support of the process. See Sections 1.7, 2.1 and Appendix C which explain that SDA one of the many tools utilized in this EIS. Text was revised to clarify the scope and authority of resource goals. As stated in Section 1.7, "While structured decision analysis informed the analysis of the joint leads, it was not the only method by which a preferred alternative was identified. The identification of a preferred alternative was based on the full DEIS analysis and considerations relating to qualitative and quantitative evaluations of impacts. Public comment, socioeconomic considerations, AMWG stakeholder input, and other factors were all considered in this decision." All models have limitations and this EIS includes disclosure of those limitations, as well assumptions. The models and the SDA analysis have also been peer-reviewed.

Section 4.1 and Appendices C and K of the LTEMP DEIS have detailed information on the extensive modeling that was performed with a variety of long term strategies. The long term strategies were variations of the alternatives that turned various experiments off or on to model different combinations and frequencies of experiments under a variety of hydrologic inflow conditions. This extensive modeling was coupled with extensive literature review and subject matter expert input to provide thorough analysis. This is sufficient for a NEPA analysis. EPA gave this document an LO-1 rating, the highest possible.

The swing weighting results were not used to quantitatively evaluate the impacts to hydropower in Chapter 4. Any analysis of swing weighting was independently used in Appendix C and was not the only criteria used for identifying a preferred alternative.

8.8 DOI DISCLAIMER

<u>Summary Comment</u>: A commenter recommended that the FEIS and ROD include appropriate disclaimer language to allow the various interested stakeholders to refrain from disputing or contesting general legal characterizations in the DEIS.

Response: DOI will consider the proposed disclaimer language for inclusion for the ROD.

8.9 DOI RESPONSE TO COMMENTS

<u>Summary Comment</u>: A number of commenters requested acknowledgement and response to their comments, that their comments be incorporated into the EIS, that revisions be shown as tracked changes. Commenters stated that they reserve the right to supplement their comments, and requested that their comment letter, material cited, and DOI response be incorporated into the LTEMP administrative record.

<u>Response</u>: Comments were solicited, collected, and incorporated into the FEIS in accordance with CEQ NEPA regulations and DOI policy and guidance.

8.10 COMMENT PERIOD DEADLINE

<u>Summary Comment</u>: A number of commenters stated that DOI had made several statements as to when comments needed to be mailed and received, causing confusion.

<u>Response</u>: All comments uploaded to the NPS PEPC site or received at Argonne with a postmark of May 9, 2016 were considered. The Federal Register notice states: "Reclamation and the NPS will accept' comments that are received or postmarked by May 9, 2016."

8.11 LANGUAGE REGARDING LROC

<u>Summary Comment</u>: A commenter stated that the LTEMP EIS should not include subjective language regarding implementation of LROC or Interim Guidelines.

<u>Response</u>: Nothing in this document affects future decisions regarding the LROC. We have reviewed the text and modified as appropriate to make that clear.

8.12 DESCRIPTION OF HFE PROTOCOL AND NONNATIVE FISH CONTROL PROTOCOL FOR THE PREFERRED ALTERNATIVE

<u>Summary Comment</u>: Commenters suggested that the specific protocols for implementation of HFEs and nonnative fish control under the preferred alternative should be specified in the EIS rather than simply referring to the existing protocols.

<u>Response</u>: Protocols to be followed for implementation of HFEs and nonnative fish control are presented in the EIS in Section 2.2.4.6, Appendix O (Biological Assessment), and Appendix P (HFE protocol for the preferred alternative). Appendix O includes an attachment that describes the triggers to be applied to various experimental aquatic resource actions and the basis of those triggers.

8.13 CONSERVATION MEASURES TO BE APPLIED UNDER ALTERNATIVE D

<u>Summary Comment</u>: One commenter suggested that the conservation measures to be applied under Alternative D should be identified and described.

<u>Response</u>: Conservation measures to be applied under Alternative D are described in Section 2.2.4.6 and Appendix O (Biological Assessment) of the EIS.

9 RECREATION

9.1 NOTE ECONOMIC BENEFITS OF THE RECREATION INDUSTRY

<u>Summary Comment</u>: Commenters stressed the importance of accounting for the benefits of river recreation to the local economy and enumerated the various direct and indirect economic activities associated with recreation, including rafting trip fares paid by visitors, franchise fees paid by rafting companies, support of rafting services, outfitters and businesses in the area, and spending on car rentals, hotel stays and restaurants. One commenter suggested that trout fishing may not be as important as it is perceived to be.

<u>Response</u>: These issues were all considered in the recreation economics sections of the LTEMP EIS; see Section 4.14 and Appendix L of the EIS. The EIS analysis fully considered the direct and indirect economic effects of recreation use in the project area.

9.2 RECREATION RESOURCE GOALS INCOMPLETE

<u>Summary Comment</u>: Commenters expressed that recreation in the Grand Canyon has been understudied in general as compared to other resources and thus that recreation resource goals and impacts on recreation are not sufficiently articulated in the LTEMP EIS.

<u>Response</u>: Recreation is one of the aspects of the Grand Canyon Protection Act which is an inherent part of the purpose and need for the LTEMP EIS and Section 1.4 includes goals and objectives for recreation. Although there are a limited number of recreation studies, the LTEMP team utilized best available studies for this EIS and utilized modeling where appropriate to address many aspects of recreation. See Section 4.10.

9.3 ADOPT THE GOALS AND OBJECTIVES OF THE AZGFD FISHERIES MANAGEMENT PLAN

<u>Summary Comment</u>: Commenters requested that the goals and objectives of the Arizona Game and Fish Department be incorporated into the LTEMP EIS and that the Arizona Game and Fish Department Fisheries Management Plan- Colorado River-Lees Ferry (Rogers 2015) be cited along with the Park Service's Comprehensive Fisheries Management Plan in the discussion of the recreational fishery.

<u>Response</u>: The AZGFD management plan and NPS's Comprehensive Fisheries Management Plan are independent of the LTEMP EIS. Because of differences in aspects of the two plans, at this time, the joint leads cannot include the objectives of the AZGFD plan or reference this plan in its current form. See also response to comment summary 3.9.

9.4 ADOPTION OF MLFF FLOWS CORRESPONDED WITH DEGRADATION OF THE LEES FERRY FISHERY DROP IN VISITATIONS

<u>Summary Comment</u>: A commenter observed that angler satisfaction and use was very high in the 1970s and 1980s, but declined following the advent of MLFF and the implementation of HFEs.

<u>Response</u>: DOI has not identified any studies that attribute the reduction in angler use days or satisfaction to the advent of MLFF or the implementation of HFEs. No change to the EIS was made in response to this comment.

10 TRIBAL RESOURCES¹

10.1 OPPOSITION TO MECHANICAL REMOVAL OF TROUT AND TROUT MANAGEMENT FLOWS

<u>Summary Comment</u>: Comments in this category generally oppose mechanical removal or trout management flows due to the adverse effects these actions would have on the cultural values ascribed to the Canyons and Rivers by the Hopi, Zuni and other Tribes. Of particular note are the adverse psychological and sociological effects to the larger Zuni community during mechanical removal and trout management flow actions, the taking of life without justification, and the seemingly opposing management objectives for trout. Additional comments support the Tribal perspective on these matters.

<u>Response</u>: DOI considers mechanical removal as an important tool for minimizing risk to downstream endangered fish from trout regardless of the cause of trout population increases or migration. DOI acknowledges the potential negative effects from all alternatives (and common to all actions) that include the taking of life in the Canyon through direct removal and/or trout management flows. We recognize the Hopi and Zuni perspective on mechanical removal and trout management flows and are committed to working with Tribal leadership on ways to address the ongoing concerns and ensure appropriate mitigation within the river corridor. Under the LTEMP EIS preferred alternative TMFs would only be used when triggered by situations where high numbers of young of the year are predicted. In years when TMFs or mechanical removal may be appropriate, the joint-leads would discuss the trout conditions of the year with AZGFD, GCMRC, and concerned Tribes prior to implementing to develop appropriate options to lessen the potential negative impacts from the proposed actions. In response to specific Zuni comments regarding Zuni perspectives on the adverse effects to their community, we have retained language in 4.9.3 and added language to a number of sections, most specifically in Section 4.9.1.3.

10.2 LACK OF TRIBAL REPRESENTATION IN DEVELOPMENT OF EIS

<u>Summary Comment</u>: Comments in the category generally question the representation of Tribes in the development of the EIS including whether or not Tribal members were part of the SME teams, how many Tribes were contacted in regards to the LTEMP EIS, and how many and what types of Tribal consultation meetings were held.

<u>Response</u>: NPS and Reclamation are committed to working with Tribal communities in order to better understand their concerns in regard to the LTEMP process. Initially, 43 Tribes, bands, and organizations were formally invited to enter into government-to-government consultation on the LTEMP EIS. Five chose to participate as Cooperating Agencies. Cooperating Agencies were invited to monthly conference calls to review progress and reach agreement on major issues

¹ Comments and responses related to Tribal concerns are also presented in Section 11, Socioeconomics and Environmental Justice.

during the preparation of the LTEMP DEIS. In addition, a number of face-to-face meetings as well as webinars and conference calls were held. Detailed information on the consultation process can be found in Chapter 5 and Appendix N. Under most resource sections there are also Tribal perspectives section where the concerns of Tribes are explained in their own words.

10.3 METRIC SHOULD NOT BE USED TO ANALYZE IMPACTS TO CULTURAL AND TRIBAL RESOURCES

<u>Summary Comment</u>: Comments under this category suggest that metrics used to analyze the impacts to cultural and Tribal resources were unwarranted. Specifically, comments suggested that there is little data to support the wind-blown sediment hypothesis and Sand Load Index in relation to availability of sediment for wind deposition and that Time Off-River is not a viable metric to evaluate visitor impacts to archaeological sites.

<u>Response</u>: We have used the GCDAMP Desired Future Conditions (DFCs) as a basis and as a guide for the resource goals and performance criteria. The Cooperating Agencies, Tribes and AMWG stakeholders were given several opportunities to comment on these for the LTEMP EIS and a number of changes were made based on those comments. We also sent the AMWG members and Cooperating Agencies a draft document of the resource goals and objectives on March 27, 2013 that further illustrated how the DFCs were used in developing each resource goal and objective.

The metric values for the Sand Load Index represent the potential for sand to be transported to cultural sites rather than the actual transport that would occur or the level of protection that transport may provide to cultural sites. The EIS acknowledges that the extent to which wind-deposited sediment can stabilize and protect archaeological sites is limited by the local geomorphology and vegetation. The text in Section 4.8.2 and Appendix H has been updated to reflect the most recently published studies regarding wind-blown sediment.

Tribes regularly monitor the condition of culturally important sites and resources within the Canyons. Many Tribes have noted intentional and unintentional damage to sites from visitors, including trailing, trampling, removal of vegetation, disturbance of artifacts, vandalism, and disruption of the sacred context through inappropriate behavior (See Sections 3.9 and 4.9). The time-of-river metric was included to respond this concern. The results showed that there was very little difference among the alternatives for time-off-river.

10.4 CULTURAL LANDSCAPES AND TRADITIONAL CULTURAL PROPERTIES NOT THOROUGHLY DISCUSSED

<u>Summary Comment</u>: Comments in this category suggest that Traditional Cultural Properties, have not been thoroughly analyzed and that the larger cultural landscape of the LTEMP study area, including the Grand Canyon as a Historic National Landmark was not acknowledged.

<u>Response</u>: The LTEMP EIS references forthcoming National Register of Historic Places nominations for the Hopi, Zuni, Navajo, and Hualapai TCPs in Section 3.9. The designation of

the Grand Canyon as a Historic National Landmark is outside the scope of this EIS; however, DOI will continue dialog with the Hualapai and other Tribes throughout the PA process and within the AMWG regarding the importance of cultural resources.

10.5 TRIBAL LANDS AND RESOURCES ON TRIBAL LANDS NOT TAKEN INTO CONSIDERATION

<u>Summary Comment</u>: The comment under this category suggests that the LTEMP EIS did not identify cultural resources on Hualapai and Navajo lands in the Introduction.

<u>Response</u>: The description of lands in this section uses terms based on geography of the canyons rather than on land ownership. Therefore, we feel the wording is appropriate as is; however, we did add text to reflect the perspective of various Tribal cultures regarding the interconnected nature of the cultural landscape.

10.6 IMPACTS TO HUALAPAI TOURISM OPERATIONS NOT THOROUGHLY ANALYZED

<u>Summary Comment</u>: Comments in this category were specifically from the Hualapai Tribe and indicated a concern with the lack of identification of potential impacts to Hualapai Tourism operations near Quartermaster Canyon as well as other Hualapai tourism operations.

<u>Response</u>: Potential impacts to Hualapai tourisms operations near Quartermaster Canyon and other Hualapai operations were consider in Sections 4.10. Based on the analysis conducted for the EIS, sedimentation near the Quartermaster Canyon would not vary significantly among the alternatives. The sedimentation this far down is primarily driven by other factors including the reservoir level at Lake Mead and annual release volumes which are outside the scope of the LTEMP EIS. Additional text was added regarding the details of Hualapai tourism operations in Sections 3.10 and 4.10 of the EIS. Text has been added under each alternative to clarify the impacts to Hualapai tourism operations.

10.7 TRIBAL PERSPECTIVES ON COMPONENTS OF THE COLORADO RIVER ECOSYSTEM

<u>Summary Comment</u>: Several commenters provided comments or text to add to Chapters 3 and 4 that expresses Tribal perspectives on components of the Colorado River Ecosystem such as fish, vegetation, and wildlife. The Zuni expressed concern related to the effects of management actions downstream of Glen Canyon Dam, specifically those directed towards fish, on the well-being of Tribal members.

<u>Response</u>: In most cases, the information provided by Tribes was incorporated as provided with only minor editorial modification as needed. Changes were made in Chapters 1, 2, 3, and 4.

10.8 EDITORIAL COMMENTS

Numerous substantive editorial comments were submitted by the Tribes on sections in the DEIS, including portions of Chapters 1, 2, 3, and 4, related to Tribal resources and concerns. Many of these comments were related to descriptions of resources or lands important to the Tribes, Tribal perspectives and views of specific resources in the Canyons or the Canyons as a whole and proper representation of Tribal views on aspects of the alternatives, especially actions that could affect aquatic ecology, vegetation, or wildlife within the Colorado River Ecosystem. In most cases, substantive editorial suggestions provided by Tribes were incorporated as provided with only minor editorial modification as needed.

11 SEDIMENT

11.1 CONCERN OVER EROSION OF SANDBARS AND BEACHES FOLLOWING HIGH FLOWS (I.E., HFEs)

<u>Summary Comment</u>: Commenters expressed concern that, while HFEs may provide benefits to certain beaches, they also create a net loss of sand and erosion to others.

<u>Response</u>: In Sections 3.3 and 4.3 there is discussion of the complexity of beach and sandbar response to HFEs. The geomorphology and other factors dictate that different HFEs build some beaches, but erode others and this is clearly stated but the average results are also considered and the modeling focuses on the average results. Based on the modeling and the extensive literature on sediment in the Grand Canyon, the preferred alternative is expected to perform well for sandbar and beach building and for many other downstream resources to comply with the GCPA. If there were unacceptable adverse impacts caused by HFEs, there is a process defined in the EIS through which HFEs would be suspended

11.2 SPRING HFES—UNCERTAINTY AND CONCERN OVER IMPACTS; MORE TESTING IS NEEDED

<u>Summary Comment</u>: Commenters emphasized that the uncertainties related to spring HFEs, resource trade-offs, and timing. They also noted that spring HFEs need to be experimentally tested and evaluated before adoption becomes an accepted management practice, but the current language surrounding the conduct of spring HFEs (as described in the Glen Canyon Dam High Flow Experiment Protocol Environmental Assessment) severely limits the potential for experimental testing of HFEs during the springtime period.

<u>Response</u>: DOI acknowledge that there are uncertainties regarding the potential impacts of spring HFEs and has reviewed the parameters under which springtime HFE's are conducted and the way in which the accounting periods are applied. Under the preferred alternative, there would be 4-7 spring HFEs on average over the 20 year period, which allows for testing and monitoring of impacts. Under the preferred alternative, spring HFE's are triggered based upon sediment triggers. GCMRC was consulted on the potential to modify the accounting periods and the scientists most familiar with the protocol did not believe it warranted a change in accounting periods.

11.3 EQUALIZATION FLOWS—CONCERN OVER INCREASED EROSION AND HIGHER RATE OF SEDIMENT TRANSPORT

<u>Summary Comment</u>: Commenters noted that equalization flows transport an enormous amount of sediment out of the system. As a result, they should be studied carefully and an effort should be made to change management practices to a proactive approach in order to minimize and mitigate the negative effects of equalization flows.

<u>Response</u>: The scope of this EIS does not include modification of annual volumes. Therefore alternatives that change equalization flows and other annual volumes were not included in this EIS. However, the EIS includes the full range of potential future releases, including equalization flows, to ensure a robust analysis of impacts to the alternatives. The alternatives analyzed in the LTEMP EIS looked at all practical approaches to conserving sediment via changes to monthly, daily, and hourly releases from the Glen Canyon Dam. Recognizing the science of sediment transport and flow, the range of alternatives include experimental proactive spring HFEs to mobilize and deposit or "park" sediment higher on the beaches in anticipation of annual releases greater than 10 maf. Further experimentation or approaches that may be devised in the future could be brought forward through the adaptive management program or GCDAMP triennial budget process and may or may not require additional NEPA compliance.

11.4 SEDIMENT SHOULD BE CONSIDERED A "SECONDARY" RESOURCE

<u>Summary Comment</u>: Commenters expressed concern that the characterization of sediment as the primary objective of the LTEMP EIS appears to be inconsistent with the AMWG-approved and Secretary-recognized identification of sediment as a "secondary" resource to be used to support other DFC resources (i.e., subset of the larger Colorado Ecosystem DFC).

<u>Response</u>: We have used the GCDAMP Desired Future Conditions (DFCs) as a basis and as a guide for the resource goals and performance criteria. The Cooperating Agencies, Tribes and AMWG stakeholders were given several opportunities to comment on these for the LTEMP EIS and we made a number of changes based on those comments. We also sent the AMWG members and Cooperating Agencies a draft document of the resource goals and objectives on March 27, 2013 that further illustrated how the DFCs were used in developing each resource goal and objective. The resource goals and objectives are based on the DFCs and consistent with them but relate more directly to the purpose and need of the LTEMP EIS. Recognizing the science of sediment transport and flow, several alternatives included proactive spring HFEs to mobilize and deposit sediment in anticipation of annual releases greater than 10 maf. Sediment conservation was a resource goal as expressed in Section 1.4.

11.5 CONCERN OVER SEDIMENT ACCUMULATION IN LAKE POWELL AND BEHIND THE DAM

<u>Summary Comment</u>: Commenters expressed concern related to the long-term accumulation of sediment into Lake Powell and its displacement of water storage, effect on flood control capacities, and potential impairment of downstream water delivery. Some commenters also requested a new and comprehensive sediment survey of Lake Powell, including a sediment removal plan.

<u>Response</u>: Long-term of accumulation of sediment into Lake Powell is an uncontrollable function of hydrology above Lake Powell and is also outside the scope of this EIS. The alternatives being considered do not affect the inflow of sediment into Lake Powell.

12 SOCIOECONOMICS AND ENVIRONMENTAL JUSTICE

12.1 REGIONAL ECONOMIC IMPACTS ASSOCIATED WITH BOATING VISITOR EXPENDITURES

<u>Summary Comment</u>: Economic impacts associated with boating visitor expenditures on lodging, food and supplies, equipment sales and rentals, fuel, guide and license fees, etc. form an important part of the local and regional economy.

<u>Response</u>: The economic impacts associated with boating visitor expenditures are assessed in Section 4.14.2.3 of the EIS. The issues listed were considered in this analysis.

12.2 REGIONAL ECONOMIC IMPACTS ASSOCIATED WITH ANGLER VISITOR EXPENDITURES, IN PARTICULAR IMPACTS OF HFEs

<u>Summary Comment</u>: Economic impacts associated with angler visitor expenditures on lodging, food and supplies, equipment sales and rentals, fuel and guide and license fees, etc. form an important part of the local and regional economy.

<u>Response</u>: The economic impacts associated with visitor expenditures related to fishing are assessed in Section 4.14.2.3 of the EIS. The effects of HFEs were considered as well as the issues listed.

12.3 USE VALUES ASSOCIATED WITH ANGLING AND BOATING TRIPS

<u>Summary Comment</u>: Economic impacts associated with angler visitor expenditures on lodging, food and supplies, equipment sales and rentals, fuel and guide and license fees, etc., during HFEs would form an important part of the local and regional economy.

<u>Response</u>: Under current NPS regulations the number of angling boat trips would not change under any of the alternatives given the excess demand for boating permits, with no consequent impact on license revenues, or on employment and income in the region. In addition, access restrictions during HFE events under any given alternative may mean a slight decline in angler visits, slightly reducing license revenues, and employment and income in the region. These impacts are discussed in Section 4.14.2.3 of the EIS.

12.4 NON-USE VALUES ASSOCIATED WITH FISHING RESOURCES

<u>Summary Comment</u>: In addition to the economic impacts, use values associated with angling and boating trips, and non-use values associated with fishing resources are a significant part of the economic value of river resources.

<u>Response</u>: Use values associated with angling in the Upper Grand Canyon, and non-use values associated with fishing resources are assessed in Section 4.14.2.2 of the EIS.

12.5 ECONOMIC IMPACTS ON TRIBES

<u>Summary Comment</u>: Economic impacts of each alternative in individual Tribes should be considered in the EIS

<u>Response</u>: The economic impacts to Tribes were considered in the EIS in the recreation (4.10), Tribal Resources (4.9) and socioeconomics (4.14) sections. There were limitations based on the available data regarding hydropower impacts that precluded separating out impacts further by individual Tribe. The geographic location of minority and low-income populations in an 11-county region is described in Section 3.14.1.5 of the EIS. The environmental justice analysis described in Section 4.14.2.5 includes economic impacts on Tribal groups in the 11-county region as a whole. The impact of HFEs on Tribes is addressed in Section 4.14.2.4 of the EIS.

12.6 IMPACT ON WATER PROVISION

<u>Summary Comment</u>: Impacts of each alternative on water provision should be considered in the EIS.

<u>Response</u>: The scope of the LTEMP EIS was to evaluate a range of alternatives with different monthly, daily, and hourly releases from the Glen Canyon Dam. All alternatives are consistent with the LROC as currently implemented through the 2007 Interim Guidelines for annual release volumes. See Sections 3.2 and 4.2 for additional detail. Because the LTEMP EIS does not affect annual volumes, it will not affect water delivery, allocation or provision. Under the LTEMP variations of the reservoir elevations within the year are small and temporary and based on modeling would not affect potential water diversions from Lake Powell or Lake Mead.

12.7 REGIONAL ELECTRIC POWER ECONOMIC IMPACTS

<u>Summary Comment</u>: Regional economic impact of changes in retail rates and electric power capacity expansion on minority and low-income populations should be considered.

<u>Response</u>: The geographic location of minority and low-income populations in an 11-county region is described in Section 3.14.1.5 of the EIS. The environmental justice analysis described in Section 4.14.2.5 includes economic impacts on Tribal groups in the 11-county region as a whole. The impact of HFEs on Tribes is addressed in Section 4.14.2.4 of the EIS.

12.8 ECONOMIC IMPACTS OF ACTIVITIES OF CONFLUENCE PARTNERS OUT OF SCOPE OF EIS

<u>Summary Comment</u>: Economic impacts of the activities of Confluence Partners (the construction of a tram, walkway, restaurant and amphitheater) should be considered in the EIS.

<u>Response</u>: A full and comprehensive analysis of the impacts potential economic development activities by Confluence Partners was outside the scope of the EIS, however effects of the Grand

Canyon Escalade project based on available project information were considered within the cumulative impacts section (4.17).

12.9 DEPTH OF LITERATURE REVIEW ON USE VALUATION IN APPENDIX I

<u>Comment Summary</u>: The literature review on use valuation in provided in Appendix I does not provide a comprehensive overview of all the relevant literature on use values.

<u>Response</u>: The Appendix referred to (Socioeconomic Technical Information and Analysis) is now Appendix L. This appendix does not summarize use value literature, it provides a summary of the benefit transfer methods and models used to estimate recreation use values in the EIS. References were added in Chapter 3 (Section 3.14) and in appendix L to the Loomis 2014 study, which was a literature review to evaluate the potential for a passive use value for hydropower. More information on the background literature, data, and assumptions in applying the benefits transfer method to the analysis undertaken for the EIS can be found in Reclamation (2014).

12.10 MECHANICAL REMOVAL AND TROUT MANAGEMENT FLOWS

<u>Summary Comment</u>: The killing of fish through mechanical removal and trout management flows are objectionable to the Zuni sensibilities.

<u>Response</u>: Section 4.14.2.5 of the EIS recognizes that the Zuni have established a lasting familial relationship with aquatic life in the Colorado River, and that the taking of life through the mechanical removal of trout or TMFs is considered to be and adverse impact and an offensive, to the Zuni. Based on input and consultations with the Pueblo of Zuni, text has been included considering mitigations such as beneficial use of fish that are removed and Reclamation consulted with the FWS under Section 7 to make some adjustments to the preferred alternative. Based on those adjustments and 'tiers' that were added in the biological assessment (see Appendix O), mechanical removal is now a second 'tier' action, as other actions would be tried first as a 'first tier'. This may reduce the frequency of mechanical removal. Differences in the frequency of the mechanical removal of trout between alternatives are also addressed in this section of the EIS. Consultations continue with Tribes on potential mitigations through an agreement in process regarding non-native fish management.

13 WATER

13.1 CONSIDERATION OF CLIMATE CHANGE EFFECTS RELATED TO WATER SUPPLY REDUCTIONS, EVAPORATION, AND OPERATIONAL LEVELS OF LAKE POWELL

<u>Summary Comment</u>: Commenters emphasized the importance and requested the consideration of climate change effects (e.g., increased temperatures, droughts, etc.) related to water supply reductions, evaporation, seepage, and operational levels of Lake Powell.

<u>Response</u>: This EIS used 21 reconstructed historical hydrological traces that were reweighted based on the 112 traces to represent the best current understanding of what might happen because of climate change. Based on this analysis, the weights on the hydrological traces had a small effect on the aggregate performance but did not change the rankings of alternatives. This result suggests the relative performance of the alternatives would be consistent regardless of the uncertainty of the effects of climate change.

The Basin Study (Reclamation 2012) was peer reviewed and is the best available science regarding climate change projections in the Colorado River Study. The hydrological traces generated for the Basin Study were utilized as described in Section 4.16.1.2 and Appendix C. See also response to comment summary 2.14.

13.2 CONSIDERATION OF CLIMATE CHANGE IMPACTS ON WATER AVAILABILITY, INCREASED DEMAND, AND GREATER DEPLETION

<u>Summary Comment</u>: Commenters emphasized the importance and requested the consideration of climate change impacts (e.g., increased temperatures, droughts, etc.) on water availability, increased water demand (both diversion and consumptive), and greater depletion.

<u>Response</u>: Water depletion assumptions used for the LTEMP EIS represent the best available and documented data provided by and agreed to by the Basin States for use in CRSS modeling.

13.3 INFLUENCE OF TRIBUTARIES AND LAKE POWELL ON MAIN STEM AND LAKE MEAD SALINITY LEVELS

<u>Summary Comment</u>: Commenters inquired about the comparative salinity values and resulting influence of tributaries and Lake Powell on main stem and Lake Mead salinity levels.

<u>Response</u>: As addressed in the EIS in Sections 3.2 and 4.2, some tributaries, such as the Little Colorado River, are significant sources of salinity for the mainstem Colorado River. But, this is diluted as it enters the mainstem and moves downstream. There are also a number of smaller spring-fed tributaries that originate within the Grand Canyon reach, which tend to have very different physicochemical properties than the mainstem; however, their mean flows are so low that their contribution to water quality during base flow is not significant. For Lake Powell, broader patterns are described in the EIS; but, focus was placed on the forebay, since this is the

water that is released downstream. The salinity module of the CRSS RiverWareTM model was used to analyze changes in salinity concentration for Colorado River reaches from Lake Powell to Imperial Dam, which is located downstream of Hoover Dam and Lake Mead. The Salinity Control Act sets numerical criteria for salinity concentrations on the Colorado River. Results and analysis indicate that, regardless of alternative or operating conditions, salinity would not increase over time or exceed control criteria.

13.4 EDITORIAL COMMENTS

Numerous substantive editorial comments were submitted on sections in the DEIS related to water resources, including Sections 3.2, 4.2, and Appendix D. Most of these editorial comments focused on the specific wording of how Glen Canyon Dam would be operated to comply with existing laws and regulations related to water delivery, collectively known as the Law of the River. Where appropriate, the text of the EIS was modified to address the specific comments.

14 WILDLIFE

14.1 AVERTING THE RESPONSIBILITY TO PROTECT NATIVE SPECIES

<u>Summary Comment</u>: A commenter stated that DOI is averting its responsibility to protect native species because all of the alternatives would result in a decrease in native plant community cover. Special status wildlife species will also be impacted (i.e., every alternative causes losses of habitat for the Kanab ambersnail and every alternative threatens to negatively affect at least one other special status wildlife species). Also, all alternatives will cause wetland loss that will affect the northern leopard frog and Yuma clapper rail. Instead of attempting to create an alternative that prevents continued habitat loss, DOI is just attempting to slow the degree of loss. DOI is averting its legal responsibilities to protect endangered species and Grand Canyon National Park, instead of seeking real solutions.

<u>Response</u>: The purpose and need is consistent with the requirements of the Grand Canyon Protection Act and NPS Organic Act because it focuses on protection and preservation of park resources. Alternatives must meet those basic requirements. All alternatives perform similarly when evaluating effects to wildlife species, showing little difference among the alternatives (Tables 4.7-1 and 4.7-2), although the preferred alternative (Alternative D) shows a benefit to wildlife in general compared to the no action alternative (Alternative A), and Alternative D performed the best for vegetation condition overall and resulted in the least wetland loss. The potential benefits of the non-flow vegetation treatment experiment could not be modeled, but were addressed qualitatively in the EIS and would help to further improve vegetation condition. Alternative D was identified as the preferred alternative because it provided the best balance for protecting, mitigating and improve downstream resources.

14.2 HFE IMPACTS ON KANAB AMBERSNAIL

<u>Summary Comment</u>: Commenters stated that HFEs, including extended duration HFEs, could negatively impact the population of the Kanab ambersnail, a Grand Canyon endangered species. Recovery time of the ambersnail population and habitat scoured by HFEs can take 2.5 years and yet this effect was apparently not considered important to the development of Alternative D.

<u>Response</u>: Endangered and threatened species were considered in the LTEMP EIS analysis based on the best available science and modeling, while the FWS has been consulted throughout the preparation of the EIS. The purpose and need of LTEMP activities are consistent with the requirements of the Grand Canyon Protection Act and NPS Organic Act because they focus on the protection and preservation of park resources. Impacts to the Kanab ambersnail are addressed in Section 4.7.2.5, 4.7.3, and the recovery time of 2.5 years was stated in those sections and considered. Additional information is included in Appendix O (biological assessment) of the Final EIS. It should also be noted that new genetic information on the Kanab ambersnail indicates that it is no longer considered a distinct species and the FWS is considering a delisting process for this species.

14.3 OPERATE GLEN CANYON DAM IN A WAY THAT PROTECTS ALL WILDLIFE (INCLUDING LISTED SPECIES)

Summary Comment: A few commenters wanted the life and safety of all wildlife to be considered. They want the Glen Canyon Dam to be operated in a way that preserves wildlife. Wildlife, including listed species, need to be considered when making decisions on the future of the Grand Canyon. The purpose and need of the LTEMP should be to reverse past damage and prevent future damage to the species of Glen and Grand Canyons.

<u>Response</u>: Impacts on wildlife, including special status species, were considered in the LTEMP EIS in Section 4.7. The purpose and need of LTEMP activities are consistent with the requirements of the Grand Canyon Protection Act and NPS Organic Act because they focus on the protection and preservation of park resources. All alternatives must meet those basic requirements.

15 COMMENTERS, AFFILIATION, AND LOCATION WHERE RESPONSES TO COMMENTS ARE FOUND

An index presenting a list of individuals who submitted comments, their affiliated organizations (when provided), and the sections in this appendix where responses to issues raised in their comments is presented in Table Q-3. All individuals who provided a first or last name in their comment submittal to NPS's PEPC database or whose names were otherwise provided on comment letters submitted by other means are included. An affiliation was listed for an individual if it was clear from their submittal that the commenter was representing the organization and was not simply a member of the organization and commenting as an individual. The right-hand column presents the sections in this appendix where responses to substantive issues raised in commenter's letters are located.

Commenter and Affiliation When Provided ^a	Sections Where Responses Are Found
Aaron, Kris	2.4; 2.13
Abate, Jo Ann	N/A ^b
Abbott, Gayle	2.4; 2.13
Abbott, Robert	3.2; 3.6
Abdel-Gawad, Aliaa	2.4; 2.13
Abrahamson, Dennis	N/A
Abshagen, WP	3.2; 3.6; 8.3
Adams, Mark	3.2; 3.6; 8.3
Adams, A.	2.4
Adams, Jon	2.4; 2.13
Adams, Mary	2.4; 2.13
Adams, Robert	N/A
Aderhold, Steven	2.4; 2.13
Adibi, Elise	2.4; 2.13
Aguirre, Elizabeth	N/A
Aguirre, Sonia	N/A
Ahearn, Mary Ann	2.4; 2.13
Ahrens, Christopher	2.4; 2.13
Aiken, Bianca	2.4; 2.13; 4.1.2; 7.3.1;; 11.3
Aiken, Edwin	2.4; 2.13
Aiken, Silas	2.4; 2.13; 4.1.2; 7.3.1; 11.3
Ainsley, Brian	2.4; 2.13
Akom, Denise	N/A
Alarie, Kim	N/A
Albanese, Dawn	2.4; 2.13
Albano, Tahyra	2.4; 2.13
Albert, Harrison	2.4; 2.13
Alcantara, Ivan	N/A
Alderson, George and Frances	2.4; 2.13

TABLE Q-3 Commenter, Affiliation, and Location Where Responses Are Found

TABLE Q-3 (Cont.)

Alexander, Charles	N/A
Alford, Cassie	N/A
Allen, Charles	2.4
Allen, Donna	2.4; 2.13
Allen, III, Doug	2.4; 2.13
Allen, James	3.6
Allen, Johnnie	N/A
Allen, Mary	2.4; 2.13
Alloway, Richard	N/A
Altum, Angelika	N/A
Alvarez, David	2.4
Amaro, Gabriel	2.4
Amato, Julie	2.4
Ancel-Wisner, Annette	2.4; 2.13
Andersen, Brian	2.8
Andersen, Kelton (Utah Associated Municipal Power	2.6; 2.10
Systems)	,
Anderson, Barry	2.4
Anderson. Dan	2.4: 2.13
Anderson, Dianne	2.4: 2.13
Anderson, Judith S	24:2.13
Anderson Karen	2.4
Anderson, Mona	2.4: 2.13
Anderson, Svlvia	24:2.13
Anderson, Trov	32: 3.6: 8.3
Anderson, Wayne	2.4: 2.13
Anderson, William	2.4: 2.13
Andrew, Jeffrey	N/A
Andrews, Richard C.	N/A
Andrews, Rosalind	2.4; 2.13
Angel. Mary	N/A
Anton, Kathleen	2.4
Antuna. Martin	2.4: 2.13
Apolinar, Augustine	N/A
Arambula, Richard	3.2: 3.6: 8.3
Arbolaez, Fidel	24:2.13
Archuleta, William	32: 3.6: 8.3
Arias Elvira	2 4 2 13
Arko Hargrove Barbara	N/A
Armes Hal	N/A
Armstead Vetty	N/A
Armstrong Leslie	2 4 2 13
Arneson Paul	24.213
Arnold Alan	24.213
Arnold Ben	2 4 2 13
Arnold Earl W (Sesne Fly Fishers)	N/A
Artin Thomas	28
Ary David	N/A
Asher Meredith	24
Ashton Richard	2.1
	<i>4.</i> T

TABLE Q-3 (Cont.)

Athavale, Anjali $24, 2.13$ Atkins, BettyeN/AAtkins, ToddN/AAtvater, Chelsea 2.12 Auer, Patricia $2.4, 2.13$ Auer, Patricia $2.4, 2.13$ Aurigemma, Kaye $2.4, 2.13$ Ayris, Jeremiah 2.4 Ayres, Peter $2.4, 2.13$ Babbit, Susan 2.4 Babbit, Susan 2.4 Babbit, Susan 2.4 Babcock, Iteather $2.1, 2.2, 2.8, 2.14, 13.1$ Backin, Martin $2.4, 2.13$ Backin, Amrin $2.4, 2.13$ Backer, Grant 12.2 Bagerman, David $3.2, 3.6, 8.3$ Bagnoli, Dana $2.2, 3.6, 8.3$ Balis, Jann $2.4, 2.13$ Baker, Grant 12.2 Bager, Mary Ann $2.4, 2.13$ Baker, Shayaa $2.4, 2.13$ Baker, John $2.4, 2.13$ <th></th> <th></th>		
Arkins, FettyeN/AAtkins, ToddN/AAtwater, Chelsea2.12Auer, Brad $2.23.6$, 8.3Aurin, Brad $2.23.6$, 8.3Avins, Jeremiah 2.4 , 2.13Ayins, Jeremiah 2.4 Ayres, Peter 2.4 , 2.13B, John 12.4 Babcock, Heather $2.12.2, 2.8, 2.14, 13.1$ Babcock, Heather $2.12, 2.2, 8, 2.14, 13.1$ Backer, Kora N/A Babcock, Heather $2.1, 2.2, 2.8, 2.14, 13.1$ Backer, Grant 12.2 Bagegeman, David $3.2, 3.6, 8.3$ Backer, Grant 12.2 Bageman, David $3.2, 3.6, 8.3$ Bain, Theodore $2.4, 2.13$ Bair, Mary Ann $2.4, 2.13$ Bair, Kirk 2.4 Baker, DonnyN/ABaker, ConnyN/ABaker, Ron $2.4, 2.13$ Bair, Mary John $2.4, 2.13$ Bair, Mary Ann $2.4, 2.13$ Bair, Kirk $2.4, 2.13$ Bair, Mary Ann $2.4, 2.13$ </td <td>Athavale, Anjali</td> <td>2.4; 2.13</td>	Athavale, Anjali	2.4; 2.13
Atkins, ToddN/AAtwater, Chelsea2.12Auer, Patricia 2.4 ; 2.13Auer, Brad 3.2 ; 3.6 ; 8.3 Aurigerma, Kaye 2.4 ; 2.13Avins, Jeremiah 2.4 B, John12.4B, John12.4B, Peter 2.4 Babbit, Susan 2.4 Backlund, Stanley 3.2 ; 3.6 ; 8.3 Bader, Sandra 2.4 ; 2.13 Backer, Grant 12.2 Baggerman, David 3.2 ; 3.6 ; 8.3 Bagnoli, Dana 3.2 ; 3.6 Bahr, WilliamN/ABaiter, Mary Ann 2.4 ; 2.13 Bailey, Shayna 2.4 ; 2.13 Baker, John 3.2 ; 3.6 ; 8.3 Baker, John 3.2 ; 3.6 ; 8.3 Baker, Ponnit, MAMA	Atkins, Bettye	N/A
Avær, Particia2.12Auer, Pard 2.4 ; 2.13Aure, Brad 3.2 ; 3.6; 8.3Auring, Jeremiah 2.4 Ayres, Peter 2.4 ; 2.13B, John 12.4 B, Peter 2.4 Babbock, Heather 2.1 ; 2.2; 2.8; 2.14; 13.1Babcock, Heather 2.1 ; 2.2; 2.8; 2.14; 13.1Backlund, Stanley 3.2 ; 3.6; 8.3Bacon, PatriciaNABacor, PatriciaNABacor, PatriciaNABacker, Grant12.2Bagerman, David 3.2 ; 3.6; 8.3Bagerman, David 3.2 ; 3.6; 8.3Bagerman, David 3.2 ; 3.6; 8.3Bager, Mary Ann 2.4 ; 2.13Baker, Grant12.2Bagerman, David 3.2 ; 3.6; 8.3Bagerman, David 3.2 ; 3.6; 8.3Bagerman, David 3.2 ; 3.6; 8.3Bagerman, David 2.4 ; 2.13Bair, Shayna 2.4 ; 2.13Bair, Sikh 2.4 Baker, DannyN/ABaire, Mary Ann 2.4 ; 2.13Baire, Sanch, Gerrit and Elizabeth 2.4 ; 2.13Baire, John 2.4 ; 2.13Baker, DannyN/ABaker, Patricia 2.4 ; 2.13Baldouri, LefandN/ABalder, Eric 2.8 ; 2.14; 13.1Baldouri, LefandN/ABalder, Eric (Glen Canyon Institute) $2.1; 2: 2.8; 2.14Baldouri, LefandN/ABalland, CynthiaN/ABalland, CynthiaN/ABalland, CynthiaN/ABalland, Cynthia$	Atkins, Todd	N/A
Auer, Patricia $24, 213$ Aure, Brad $32, 36, 8.3$ Aurigemma, Kaye $24, 213$ Avins, Jeremiah 24 Ayres, Peter $24, 213$ B, John 124 B, Peter 24 Babbit, Susan 24 Backcock, Heather $21, 22, 28, 214, 13.1$ Backcock, Staley $32, 36, 8.3$ Backlund, Staley $32, 36, 8.3$ Bach, Staley $32, 36, 8.3$ Bacn, Patricia NA Badeer, Grant 122 Bagerman, David $32, 36, 8.3$ Bagnoli, Dana $32, 36, 8.3$ Bagnoli, Dana $32, 36, 8.3$ Baley, Shayna $24, 2.13$ Baits, Jean $24, 2.13$ Baits, Jean $24, 2.13$ Baits, Jean $24, 2.13$ Baits, Kirk 24 Baker, John $24, 2.13$ Baits, Jean $24, 2.13$ Baits, Jean $24, 2.13$ Baits, Jean $24, 2.13$ Baits, Jean $24, 2.13$ Baits, Joanne & Joseph $24, $	Atwater, Chelsea	2.12
Aune, Brad $32, 3, 6, 8.3$ Aurigemma, Kaye $2.4, 2.13$ Avins, Jeremiah 2.4 Ayres, Peter $2.4, 2.13$ B, John 12.4 B, Peter 2.4 Babrock, Heather $2.1, 2.2, 2.8, 2.14, 13.1$ Backlind, Stanley $32, 3, 6, 8.3$ Backlind, Martin $2.4, 2.13$ Backlind, Martin $2.4, 2.13$ Backrind, Martin $2.4, 2.13$ Backrind, Grant 12.2 Baggerman, David $32, 3, 6, 8.3$ Baggerman, David $32, 3, 6, 8.3$ Baglin, Maran N/A Bader, Sandra $2.4, 2.13$ Backer, Grant 12.2 Baggerman, David $32, 3, 6, 8.3$ Bagn, Theodore $2.4, 2.13$ Bair, Mary Ann $2.4, 2.13$ Baits, Kirk 2.4 Baker, DannyN/ABaker, DannyN/ABaker, ConnN/ABaker, Sinth, Gerritt and Elizabeth $2.4, 2.13$ Baits, Virk 2.4 Baker, Sinth, Gerritt and Elizabeth $2.4, 2.13$ Baldouri, MicheleN/ABaldeur, Fric $2.8, 2.14, 13.1$ Baldeur, Eric $2.8, 2.14, 13.1$ Baldeur, Eric $2.8, 2.14, 13.1$ Baltar, Brin, Brin $2.3, 3.6, 8.3$ Baltar, SantaN/ABaltar, DannyN/ABaltar, DannyN/ABaltar, SantaN/ABaltar, SantaN/ABaltar, SantaN/ABaltar, SantaN/ABaltari, Brin $2.3, 3.6, 8.3$ <	Auer, Patricia	2.4; 2.13
Aurigemma, Kaye24Avins, Jeremiah24Ayres, Peter2.4; 2.13B, John12.4B, Peter2.4Babbitt, Susan2.4Babcock, Heather2.1; 2.2; 2.8; 2.14; 13.1Backlund, Stanley3.2; 3.6; 8.3Baclija, Martin2.4; 2.13Bacock, Forant12.2Bagerman, David3.2; 3.6; 8.3Bagerman, David3.2; 3.6; 8.3Bager, MilliamN/ABaier, Mary Ann2.4; 2.13Bails, Jean2.4; 2.13Bails, Van2.4; 2.13Bails, Van2.4; 2.13Bails, Kirk2.4Baker, DannyN/ABaker, RonN/ABaker, RonN/ABaker, RonN/ABalay, Joanne & Joseph2.4; 2.13Baldy, Joanne & Joseph2.4; 2.13Baldy, Joanne & Joseph2.4; 2.13Baldwin, LelandN/ABaldry, TickeleN/ABaldry, TickeleN/ABaltar, Eric (Glen Canyon Institute)2.1; 2.2; 2.8; 2.14Ballar, Petricia3.2; 3.6; 8.3Ballar, SiniceN/ABallar, JaniceN/ABallar, SiniceN/ABallar, SiniceN/ABallar, SiniceN/ABannon, Robert1.2Barbier, Lynn2.4; 2.13Barbarii Balagas, Ayana </td <td>Aune, Brad</td> <td>3.2; 3.6; 8.3</td>	Aune, Brad	3.2; 3.6; 8.3
Avins, Jeremiah 2.4 Ayres, Peter 2.4 B, John 12.4 B, John 12.4 Babbit, Susan 2.4 Babbecok, Heather $2.1; 2.2; 2.8; 2.14; 13.1$ Backlia, Martin $2.4; 2.13$ Backris, Martin $2.4; 2.13$ Backris, Martin $2.4; 2.13$ Backris, Martin $2.4; 2.13$ Backris, Grant 12.2 Baggerman, David $3.2; 3.6; 8.3$ Bagnoli, Dana $3.2; 3.6; 8.3$ Baitr, Mary Ann $2.4; 2.13$ Baiter, Mary Ann $2.4; 2.13$ Baiter, Mary Ann $2.4; 2.13$ Baker, DannyN/ABaker, DannyN/ABaker, DannyN/ABaker, DannyN/ABaker, Conn N/A Baker, ConnN/ABaker, Panny N/A Baker, RonN/ABalay, Jonne & Joseph $2.4; 2.13$ Baldy, Jonne & Joseph $2.4; 2.13$ Baldyni, LelandN/ABaley, Patricia $2.4; 2.13$ Baldwin, LelandN/ABaley, Patricia $2.4; 2.13$ Ballady, CynthiaN/ABallady, JonnyN/ABallady, JonnyN/ABallady, JonnyN/ABallady, Synan $2.4; 2.13$ Ballady, LelandN/ABallady, Jonno Résolve $2.1; 2.2; 2.14.4$ Ballady, Jonno Résolve $2.1; 2.2; 2.14.4$ Ballady, Jonno Robert $1.2; 2.2; 2.14.4$ Ballady, Synan $3.2; 3.6; 8.3$ Bannon, Robert $1.2;$	Aurigemma, Kaye	2.4; 2.13
Ayres, Peter $24; 2.13$ B, John 12.4 B, Peter 2.4 Babbitt, Susan 2.4 Babcock, Heather $21; 22; 23; 2.14; 13.1$ Backlund, Stanley $32; 36; 8.3$ Baclija, Martin $24; 2.13$ Bacor, PatriciaNABader, Sandra $24; 2.13$ Bagerman, David $32; 36; 8.3$ Baggerman, David $32; 36; 8.3$ Baggorn, Dana $32; 36; 8.3$ Baggorn, Dana $32; 36; 8.3$ Baggorn, Dana $32; 36; 8.3$ Bagnoli, Dana $32; 36; 8.3$ Bagnoli, Dana $32; 36; 8.3$ Bagnoli, Jana $32; 36; 8.3$ Bagnoli, Jean $24; 2.13$ Bails, Jean $24; 2.13$ Baker, John $24; 2.13$ Baker, John $24; 2.13$ Baker, John $24; 2.13$ Baker, RonNABaker, RonNABaley, Joanne & Joseph $24; 2.13$ Balay, Joanne & Joseph $24; 2.13$ Balay, Joanne & Joseph $24; 2.13$ Balker, Fric $21; 2.2; 2.8; 2.14; 13.1$ Balker, Pirici $24; 2.13$ Balker, Pirici $24; 2.13$ Balker, Pirici $24; 2.13$ Balker, Pirici $24; 2.13$ Balker, Sima $32; 36; 8.3$ Banks, Janice<	Avins, Jeremiah	2.4
B, John12.4B, Peter2.4Babbitt, Susan2.4Babbott, Susan2.4Babcock, Heather2.1; 2.2; 2.8; 2.14; 13.1Backlund, Stanley3.2; 3.6; 8.3Baclija, Martin2.4; 2.13Bacon, PatriciaN/ABacker, Grant12.2Baggerman, David3.2; 3.6; 8.3Bagnoli, Dana3.2; 3.6Bair, Mary Ann2.4; 2.13Baire, Mary Ann2.4; 2.13Baire, Mary Ann2.4; 2.13Baire, Mary Ann2.4; 2.13Baik, Kirk2.4Baker, John2.4; 2.13Baker, Sonth, Gerritt and Elizabeth2.4; 2.13Baldwin, LelandN/ABaley, Patricia2.4; 2.13Baldwin, LelandN/ABalken, Eric (Glen Canyon Institute)2.1; 2.2; 2.8; 2.14Ball, WilliamN/ABalter, Diring Balagas, AyanaN/ABaltald, CynthiaN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, Wesley2.4; 2.13Banks, JaniceN/A <td< td=""><td>Ayres, Peter</td><td>2.4; 2.13</td></td<>	Ayres, Peter	2.4; 2.13
B. Peter2.4Babbock, Heather 2.1 ; 2.2 ; 2.8 ; 2.14 ; 13.1 Backlund, Stanley 3.2 ; 3.6 ; 8.3 Baclija, Martin 2.4 ; 2.13 Backry, Stanley 3.2 ; 3.6 ; 8.3 Bador, PatriciaNABader, Sandra 2.4 ; 2.13 Backekr, Grant 12.2 Bagerinan, David 3.2 ; 3.6 ; 8.3 Bagnoli, Dana 3.2 ; 3.6 ; 8.3 Bagnoli, Dana 3.2 ; 3.6 Bahr, WilliamNABaier, Mary Ann 2.4 ; 2.13 Bailey, Shayna 2.4 ; 2.13 Bailey, Shayna 2.4 ; 2.13 Bails, Lean 2.4 ; 2.13 Baker, DamyNABaker, DannyNABaker, John 2.4 ; 2.13 Baker, John 2.4 ; 2.13 Baker, PannyNABaker, Smith, Gerritt and Elizabeth 2.4 ; 2.13 Baldwin, LelandN/ABaldey, Patricia 2.4 ; 2.13 Baldwin, LelandN/ABalken, Eric 2.8 ; 2.14 ; 13.1 Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14Balker, Fric (Glen Canyon Institute)2.1; 2.2; 2.3; 2.14Ball, Evelyn2.4; 2.13Ball, Evelyn2.4; 2.13Ballari, Janagas, AyanaN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABaresich, Dennis2.4; 2.13Baresich, Dennis$	B, John	12.4
Babbit 2.4 Babcock, Heather $2.1; 2.2; 2.8; 2.14; 13.1$ Backlund, Stanley $3.2; 3.6; 8.3$ Backlund, Stanley $3.2; 3.6; 8.3$ Bacon, PatriciaN/ABader, Sandra $2.4; 2.13$ Bagerman, David $3.2; 3.6; 8.3$ Baggerman, David $3.2; 3.6; 8.3$ Baggerman, David $3.2; 3.6; 8.3$ Baggerman, David $3.2; 3.6; 8.3$ Bagnoli, Dana $3.2; 3.6; 8.3$ Bagnoli, Dana $3.2; 3.6; 8.3$ Bagnoli, Dana $3.2; 3.6; 8.3$ Bair, Mary Ann $2.4; 2.13$ Bair, Mary Ann $2.4; 2.13$ Bairs, Mary Ann $2.4; 2.13$ Bails, Jean $2.4; 2.13$ Bails, Jean $2.4; 2.13$ Bails, Jean $2.4; 2.13$ Baker, DannyN/ABaker, John $2.4; 2.13$ Baker, RonN/ABaker, Smith, Gerritt and Elizabeth $2.4; 2.13$ Balay, Joanne & Joseph $2.4; 2.13$ Baldwin, LelandN/ABalfour, MicheleN/ABalter, Eric $2.8; 2.14; 13.1$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Ball, WilliamN/ABaltrip Balagas, AyanaN/ABaltrip Balagas, AyanaN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, Wesley $2.4; 2.13$ Banks, JaniceN/ABanks, Wesley $2.4; 2.13$ Banesich, Dennis $2.4; 2.13$ <t< td=""><td>B, Peter</td><td>2.4</td></t<>	B, Peter	2.4
Babcock, Heather $2.1; 2.2; 2.8; 2.14; 13.1$ Backlind, Stanley $3.2; 3.6; 8.3$ Baclija, Martin $2.4; 2.13$ Bacon, PatriciaN/ABader, Sandra $2.4; 2.13$ Backer, Grant 12.2 Baggerman, David $3.2; 3.6; 8.3$ Bagnoli, Dana $3.2; 3.6; 8.3$ Bahn, Theodore $2.4; 2.13$ Bahr, Theodore $2.4; 2.13$ Baiter, Mary Ann $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Baker, Mary Ann $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Baker, DannyN/ABaker, RonN/ABaker, RonN/ABaker, Smith, Gerritt and Elizabeth $2.4; 2.13$ Baley, Patricia $2.4; 2.13$ Baley, Patricia $2.4; 2.13$ Baley, Patricia $2.4; 2.13$ Balken, Eric $2.8; 2.14; 13.1$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Balt, WilliamN/ABaltarj, Danlagas, AyanaN/ABaltarj, Danlagas, AyanaN/ABaltar, CynthiaN/ABaltar, CynthiaN/ABaltar, CynthiaN/ABantar, Jim $3.2; 3.6; 8.3$ Bantas, JariceN/ABantas, JariceN/ABantas, JariceN/ABantas, JariceN/ABarbarder, Dennis $2.4; 2.13$ Barbarder, DonaldN/ABarbarder, Cong $2.4; 2.13$ Barbarder	Babbitt, Susan	2.4
Backlund, Stanley $3.2; 3.6; 8.3$ Baclija, Martin $2.4; 2.13$ Bacon, PatriciaNABader, Sandra $2.4; 2.13$ Bagernan, David $3.2; 3.6; 8.3$ Baggerman, David $3.2; 3.6; 8.3$ Bagnoli, Dana $3.2; 3.6; 8.3$ Bagnoli, Dana $3.2; 3.6; 8.3$ Bagnoli, Dana $3.2; 3.6; 8.3$ Bahr, WilliamN/ABair, Mary Ann $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Bails, Jean $2.4; 2.13$ Baker, DannyN/ABaker, John $2.4; 2.13$ Baker, RonN/ABaker, Smith, Gerritt and Elizabeth $2.4; 2.13$ Balay, Joanne & Joseph $2.4; 2.13$ Baldwin, LelandN/ABaker, Smith, Gerritt and Elizabeth $2.4; 2.13$ Baldwin, LelandN/ABaley, Patricia $2.4; 2.13$ Balfour, MicheleN/ABalken, Eric $2.8; 2.14; 13.1$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Ball, Evelyn $2.4; 2.13$ Ballard, CynthiaN/ABallard, CynthiaN/ABallard, S, JiniceN/ABanks, JiniceN/ABanks, JiniceN/ABanks, JaniceN/ABarseich, Dennisi $2.4; 2.13$ Barseich, Dennisi $2.4; 2.13$ Barseir, Jonn $2.4; 2.13$ Barseir, Jonn $2.4; 2.13$ Barseir, Jonn $2.4; 2.13$ Barseire, Jonn $2.4; 2.13$ <t< td=""><td>Babcock, Heather</td><td>2.1; 2.2; 2.8; 2.14; 13.1</td></t<>	Babcock, Heather	2.1; 2.2; 2.8; 2.14; 13.1
Baclija, Martin $2.4; 2.13$ Bacon, PatriciaNABader, Sandra $2.4; 2.13$ Baceker, Grant 12.2 Baggerman, David $3.2; 3.6; 8.3$ Bagnoli, Dana $3.2; 3.6$ Bahn, Theodore $2.4; 2.13$ Bahr, WilliamNABaier, Mary Ann $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Baiks, Jean $2.4; 2.13$ Baiks, Jean $2.4; 2.13$ Baker, John $2.4; 2.13$ Baker, Smith, Gerritt and Elizabeth $2.4; 2.13$ Balay, Joanne & Joseph $2.4; 2.13$ Balker, RonNABaley, Patricia $2.4; 2.13$ Balker, Gelon Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.3; 2.14$ Ball, WilliamN/ABallard, CynthiaN/ABaltrip Balagas, AyanaN/ABalnks, Jim $3.2; 3.6; 8.3$ Banks, JanceN/ABanks, JanceN/ABaraser, John $2.4; 2.13$ Baraser, John $2.4; 2.13$ <tr< td=""><td>Backlund, Stanley</td><td>3.2; 3.6; 8.3</td></tr<>	Backlund, Stanley	3.2; 3.6; 8.3
Bacon, PatriciaNABader, Sandra $2.4; 2.13$ Baceker, Grant 12.2 Baggerman, David $32; 3.6; 8.3$ Bagnoli, Dana $32; 3.6; 8.3$ Bahn, Theodore $2.4; 2.13$ Bahr, WilliamNABairer, Mary Ann $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Bails, Jean $2.4; 2.13$ Baiker, DannyNABaker, DannyNABaker, John $2.4; 2.13$ Baker, Smith, Gerritt and Elizabeth $2.4; 2.13$ Baley, Joanne & Joseph $2.4; 2.13$ Baley, Patricia $2.4; 2.13$ Baley, Patricia $2.4; 2.13$ Balker, Eric $2.4; 2.13$ Balker, Eric $2.4; 2.13$ Balker, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Ball, Fvelyn $2.4; 2.13$ Ballard, CynthiaNABalteri, SlimicNABaltrip Balagas, AyanaN/ABaltrip Balagas, AyanaN/ABanks, Jimic $3.2; 3.6; 8.3$ Banks, JamiceNABanks, JamiceNABanks, JamiceNABanks, Jonnik, Jennik $2.4; 2.13$ Banks, JaniceNABanks, Jonnik, Jennik $2.4; 2.13$ Banks, JaniceNABanks, JaniceNABanks, Jonnik $2.4; 2.13$ Banks, Jonnik $2.4; 2.13$ <td< td=""><td>Baclija, Martin</td><td>2.4; 2.13</td></td<>	Baclija, Martin	2.4; 2.13
Bader, Sandra $24; 2.13$ Baecker, Grant 12.2 Baggerman, David $32; 3, 6; 8.3$ Bagnoli, Dana $32; 3, 6$ Bahn, Theodore $24; 2.13$ Bahr, WilliamN/ABaiter, Mary Ann $24; 2.13$ Bails, Jean $24; 2.13$ Baiks, Jean $24; 2.13$ Baker, DannyN/ABaker, John $24; 2.13$ Baker, John $24; 2.13$ Baker, RonN/ABaker, Smith, Gerritt and Elizabeth $24; 2.13$ Baldwin, LelandN/ABaker, Froi $24; 2.13$ Balkour, MicheleN/ABaker, Fric (Glen Canyon Institute) $21; 22; 28; 2.14$ Ball, VilliamN/ABallar, Eric (Glen Canyon Institute) $21; 22; 28; 2.14$ Ballard, CynthiaN/ABalter, Jim $32; 36; 8.3$ Banks, Jim $32; 36; 8.3$ Banks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, Jonna, Ava $24; 2.13$ Banson, Robert 122 Barbeir, Lyonna $24; 2.13$ Barker, DonaldN/ABarker, DonaldN/ABarker, DonaldN/ABarker, DonaldN/ABarker, DonaldN/ABarker, DonaldN/ABarker, DonaldN/ABarker, DonaldN/ABarker, DonaldN/A	Bacon, Patricia	N/A
Baecker, Grant12.2Baggerman, David 3.2 ; 3.6 ; 8.3 Bagnoli, Dana 3.2 ; 3.6 ; 8.3 Bahn, Theodore 2.4 ; 2.13 Bahn, Theodore 2.4 ; 2.13 Bahr, William N/A Baier, Mary Ann 2.4 ; 2.13 Bailey, Shayna 2.4 ; 2.13 Bails, Ican 2.4 ; 2.13 Bails, Kirk 2.4 Baker, Danny N/A Baker, Conny N/A Baker, Ron N/A Baker, Smith, Gerritt and Elizabeth 2.4 ; 2.13 Balay, Joanne & Joseph 2.4 ; 2.13 Baldwin, Leland N/A Baley, Patricia 2.4 ; 2.13 Balker, Eric 2.8 ; 2.14 ; 13.1 Balker, Eric (Glen Canyon Institute) 2.1 ; 2.2 ; 2.8 ; 2.14 Ball, William N/A Ballard, Cynthia N/A Ballard, Cynthia N/A Ballrip Balagas, Ayana N/A Banks, Jim 3.2 ; 3.6 ; 8.3 Banks, Janice N/A Banks, Jim 2.4 ; 2.13 Banks, Mesley 2.4 ; 2.13 Banks, Vesley 2.4 ; 2.13 Banks, Vesley 2.4 ; 2.13 Banks, Wesley 2.4 ; 2.13 Banks, Jimin 2.4 ; 2.13 Barbarder, Dennis 2.4 ; 2.13 Barbarder, Dennis 2.4 ; 2.13 Baresich, Dennis 2.4 ; 2.13 Barbarder, Donald N/A Barbarder, Donald N/A Barbarder, Donald N/A	Bader, Sandra	2.4; 2.13
Baggerman, David $3.2; 3.6; 8.3$ Bagnoli, Dana $3.2; 3.6$ Bahn, Theodore $2.4; 2.13$ Bahr, WilliamN/ABaier, Mary Ann $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Bails, Jean $2.4; 2.13$ Bakr, Kirk 2.4 Baker, John $2.4; 2.13$ Baker, John $2.4; 2.13$ Baker, John $2.4; 2.13$ Baker, John $2.4; 2.13$ Baker, RonN/ABaker, RonN/ABaker, Smith, Gerritt and Elizabeth $2.4; 2.13$ Baldwin, LelandN/ABalken, Eric $2.8; 2.14; 13.1$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Ball, WilliamN/ABallard, CynthiaN/ABallard, CynthiaN/ABallard, CynthiaN/ABarks, JaniceN/ABanks, JaniceN/ABanks, JariceN/ABarks, Wesley $2.4; 2.13$ Barks, Wesley $2.4; 2.13$ Barks, Vesley $2.4; 2.13$ Banks, JariceN/ABanks, JariceN/ABarker, Donald $2.4; 2.13$ Barker, Donald $2.4; 2.13$ Barks, Vesley $2.4; 2.13$ Barker, Donald $3.2; 3.6; 8.3$ Barker, Donald $3.2; 3.6; 8.3$ Barker, Greg $2.8; 2.14$	Baecker, Grant	12.2
Bagnoli, Dana $3.2; 3.6$ Bahn, Theodore $2.4; 2.13$ Bahr, WilliamN/ABaier, Mary Ann $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Bails, Jean $2.4; 2.13$ Bails, Kirk 2.4 Baker, DannyN/ABaker, John $2.4; 2.13$ Baker, RonN/ABaker, RonN/ABaker, Smith, Gerritt and Elizabeth $2.4; 2.13$ Balay, Joanne & Joseph $2.4; 2.13$ Balay, Joanne & Joseph $2.4; 2.13$ Balay, Joanne & Joseph $2.4; 2.13$ Balken, Eric $2.8; 2.14; 13.1$ Balken, Eric $2.8; 2.14; 13.1$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Ball, WilliamN/ABallard, CynthiaN/ABallard, CynthiaN/ABans, Jim $3.2; 3.6; 8.3$ Banks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, Wesley $2.4; 2.13$ Barbieri, Lynn $2.4; 2.13$ Barbieri, Lynn $2.4; 2.13$ Barsich, Dennis $2.4; 2.13$ Barger, John $2.4; 2.13$ Barger, John $2.4; 2.13$ Barker, Greg $2.8; 2.14$	Baggerman, David	3.2; 3.6; 8.3
Bahn, Theodore $2.4; 2.13$ Bahr, WilliamN/ABaier, Mary Ann $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Bails, Jean $2.4; 2.13$ Bails, Kirk 2.4 Baker, DannyN/ABaker, John $2.4; 2.13$ Baker, NonN/ABaker-Smith, Gerritt and Elizabeth $2.4; 2.13$ Balay, Joanne & Joseph $2.4; 2.13$ Baldwin, LelandN/ABalken, Eric $2.4; 2.13$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.14$ Ball, WilliamN/ABallard, CynthiaN/ABallard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks, JaniceN/ABanks, JaniceN/ABanks, Vesley $2.4; 2.13$ Banks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, Dennis $2.4; 2.13$ Barker, Dennis $2.4; 2.13$ Barker, DonaldN/ABarker, DonaldN/ABarker, Greg $2.8; 2.14$	Bagnoli, Dana	3.2; 3.6
Bahr, William N/A Baier, Mary Ann $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Bails, Jean $2.4; 2.13$ Bails, Kirk 2.4 Baker, Danny N/A Baker, John $2.4; 2.13$ Baker, Ron N/A Baker, Smith, Gerritt and Elizabeth $2.4; 2.13$ Balay, Joanne & Joseph $2.4; 2.13$ Baldwin, Leland N/A Baley, Patricia $2.4; 2.13$ Ballour, Michele N/A Balken, Eric $2.8; 2.14; 13.1$ Ball, Evelyn $2.4; 2.13$ Ball, William N/A Ball, William N/A Balard, Cynthia N/A Baltrip Balagas, Ayana N/A Banks, Jim $3.2; 3.6; 8.3$ Banks, Janice N/A Banks, Janice N/A Banks, Janice N/A Banks, Vasely $2.4; 2.13$ Barbeiri, Lynn $2.4; 2.13$ Baresich, Dennis $2.4; 2.13$ Barks, Vas $2.4; 2.13$ Barks, F, Donald N/A Barker, Greg $2.8; 2.14$	Bahn, Theodore	2.4; 2.13
Baier, Mary Ann $2.4; 2.13$ Bailey, Shayna $2.4; 2.13$ Bails, Jean $2.4; 2.13$ Bails, Kirk 2.4 Baker, DannyN/ABaker, Ron $2.4; 2.13$ Baker, Ron $2.4; 2.13$ Baker, Smith, Gerritt and Elizabeth $2.4; 2.13$ Balay, Joanne & Joseph $2.4; 2.13$ Baldwin, LelandN/ABaley, Patricia $2.4; 2.13$ Balkon, Fric $2.8; 2.14; 13.1$ Balkon, Eric $2.8; 2.14; 13.1$ Balkon, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Ball, WilliamN/ABalard, CynthiaN/ABalard, CynthiaN/ABanks, Jim $3.2; 3.6; 8.3$ Banks, JaniceN/ABanks, Vesley $2.4; 2.13$ Banks, Vesley $2.4; 2.13$ Barbieri, Lynn $2.4; 2.13$ Barbieri, Lynn $2.4; 2.13$ Barbard, Ponnis $2.4; 2.13$ Barbard, Ponnis $2.4; 2.13$ Barbard, Ponnis $2.4; 2.13$ Barbard, Ponnis $2.4; 2.13$ Barbard, DonaldN/ABarbard, SonaldN/ABarbard, SonaldN/ABarbard, Roma $2.4; 2.13$ Barbard, DonaldN/ABarbard, SonaldN/A	Bahr, William	N/A
Bailey, Shayna $2.4; 2.13$ Bails, Jean $2.4; 2.13$ Bails, Kirk 2.4 Baker, DannyN/ABaker, John $2.4; 2.13$ Baker, RonN/ABaker-Smith, Gerritt and Elizabeth $2.4; 2.13$ Balay, Joanne & Joseph $2.4; 2.13$ Baldwin, LelandN/ABalken, Eric $2.4; 2.13$ Balken, Eric $2.4; 2.13$ Balken, Eric $2.8; 2.14; 13.1$ Balken, Eric $2.4; 2.13$ Ball, WilliamN/ABall, Evelyn $2.4; 2.13$ Ballard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks, Jim $3.2; 3.6; 8.3$ Banks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, Vesley $2.4; 2.13$ Baresich, Dennis $2.4; 2.13$ Barker, Greg $2.8; 2.14$	Baier, Mary Ann	2.4; 2.13
Bails, Jean 2.4 ; 2.13 Bails, Kirk 2.4 Baker, DannyN/ABaker, John 2.4 ; 2.13 Baker, RonN/ABaker, Smith, Gerritt and Elizabeth 2.4 ; 2.13 Balay, Joanne & Joseph 2.4 ; 2.13 Baldwin, LelandN/ABalker, Fric 2.4 ; 2.13 Balfour, MicheleN/ABalken, Eric 2.8 ; 2.14 ; 13.1 Balken, Eric (Glen Canyon Institute) 2.1 ; 2.2 ; 2.8 ; 2.14 Ball, Evelyn 2.4 ; 2.13 Ballard, CynthiaN/ABallard, CynthiaN/ABanks, Jim 3.2 ; 3.6 ; 8.3 Banks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, Janice 2.4 ; 2.13 Bannon, Robert 12.2 Barbieri, Lynn 2.4 ; 2.13 Barbieri, Lynn 2.4 ; 2.13 Barger, John 2.4 ; 2.13 Barker, OrnaldN/ABarker, Greg 2.8 ; 2.14	Bailey, Shayna	2.4; 2.13
Bails, Kirk 2.4 Baker, DannyN/ABaker, John 2.4 ; 2.13 Baker, RonN/ABaker-Smith, Gerritt and Elizabeth 2.4 ; 2.13 Balay, Joanne & Joseph 2.4 ; 2.13 Baldwin, LelandN/ABaley, Patricia 2.4 ; 2.13 Balfour, MicheleN/ABalken, Eric 2.8 ; 2.14 ; 13.1 Balken, Eric (Glen Canyon Institute) 2.1 ; 2.2 ; 2.8 ; 2.14 Ball, Evelyn 2.4 ; 2.13 Ballard, CynthiaN/ABallard, CynthiaN/ABanks, Jim 3.2 ; 3.6 ; 8.3 Banks, JaniceN/ABanks, JaniceN/ABanks, JaniceN/ABanks, John 2.4 ; 2.13 Bannon, Robert 12.2 Barbieri, Lynn 2.4 ; 2.13 Barseich, Dennis 2.4 ; 2.13 Barager, John 2.4 ; 2.13 Barker, Cong 2.8 ; 2.14	Bails, Jean	2.4; 2.13
Baker, DannyN/ABaker, John $2.4; 2.13$ Baker, RonN/ABaker-Smith, Gerritt and Elizabeth $2.4; 2.13$ Balay, Joanne & Joseph $2.4; 2.13$ Baldwin, LelandN/ABaley, Patricia $2.4; 2.13$ Balfour, MicheleN/ABalken, Eric $2.8; 2.14; 13.1$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Ball, Evelyn $2.4; 2.13$ Ballard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks, Jim $3.2; 3.6; 8.3$ Banks, JaniceN/ABanks, Wesley $2.4; 2.13$ Barbieri, Lynn $2.4; 2.13$ Barbieri, Dennis $2.4; 2.13$ Barbieri, DonaldN/ABarker, OnaldN/ABarker, Greg $2.8; 2.14$	Bails, Kirk	2.4
Baker, John 2.4 ; 2.13 Baker, RonN/ABaker-Smith, Gerritt and Elizabeth 2.4 ; 2.13 Balay, Joanne & Joseph 2.4 ; 2.13 Baldwin, LelandN/ABaley, Patricia 2.4 ; 2.13 Balfour, MicheleN/ABalken, Eric 2.8 ; 2.14 ; 13.1 Balken, Eric (Glen Canyon Institute) 2.1 ; 2.2 ; 2.8 ; 2.14 Ball, Evelyn 2.4 ; 2.13 Ball, WilliamN/ABallard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks, Jim 3.2 ; 3.6 ; 8.3 Banks, JaniceN/ABanks, Wesley 2.4 ; 2.13 Barbieri, Lynn 2.4 ; 2.13 Barsci, Dennis 2.4 ; 2.13 Barsci, Dennis 2.4 ; 2.13 Barana, Ava 2.4 ; 2.13 Barana, Ava 2.4 ; 2.13 Barker, OnaldN/ABarker, Greg 2.8 ; 2.14	Baker, Danny	N/A
Baker, Ron N/A Baker-Smith, Gerritt and Elizabeth $2.4; 2.13$ Balay, Joanne & Joseph $2.4; 2.13$ Baldwin, Leland N/A Baley, Patricia $2.4; 2.13$ Balfour, Michele N/A Balken, Eric $2.8; 2.14; 13.1$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Ball, Evelyn $2.4; 2.13$ Ball, Evelyn $2.4; 2.13$ Ballard, Cynthia N/A Baltrip Balagas, Ayana N/A Banks, Janice N/A Banks, Wesley $2.4; 2.13$ Barnon, Robert 12.2 Barbieri, Lynn $2.4; 2.13$ Barsci, Dennis $2.4; 2.13$ Barger, John $2.4; 2.13$ Barker, Donald N/A Barker, Greg $2.8; 2.14$	Baker, John	2.4; 2.13
Baker-Smith, Gerritt and Elizabeth 2.4 ; 2.13 Balay, Joanne & Joseph 2.4 ; 2.13 Baldwin, LelandN/ABaley, Patricia 2.4 ; 2.13 Balfour, MicheleN/ABalken, Eric 2.8 ; 2.14 ; 13.1 Balken, Eric (Glen Canyon Institute) 2.1 ; 2.2 ; 2.8 ; 2.14 Ball, Evelyn 2.4 ; 2.13 Ball, Evelyn 2.4 ; 2.13 Ballard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks, Jim 3.2 ; 3.6 ; 8.3 Banks, JaniceN/ABanks, Wesley 2.4 ; 2.13 Barbieri, Lynn 2.4 ; 2.13 Barsich, Dennis 2.4 ; 2.13 Barsich, Dennis 2.4 ; 2.13 Barger, John 2.4 ; 2.13 Barker, DonaldN/ABarker, Greg 2.8 ; 2.14	Baker, Ron	N/A
Balay, Joanne & Joseph $2.4; 2.13$ Baldwin, LelandN/ABaley, Patricia $2.4; 2.13$ Balfour, MicheleN/ABalken, Eric $2.8; 2.14; 13.1$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Ball, Evelyn $2.4; 2.13$ Ball, WilliamN/ABallard, CynthiaN/ABanks, Jim $3.2; 3.6; 8.3$ Banks, JaniceN/ABanks, Wesley $2.4; 2.13$ Bannon, Robert 12.2 Barsich, Dennis $2.4; 2.13$ Barger, John $2.4; 2.13$ Bariana, Ava $2.4; 2.13$ Barker, Greg $2.8; 2.14$	Baker-Smith, Gerritt and Elizabeth	2.4; 2.13
Baldwin, LelandN/ABaley, Patricia $2.4; 2.13$ Balfour, MicheleN/ABalken, Eric $2.8; 2.14; 13.1$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Ball, Evelyn $2.4; 2.13$ Ball, WilliamN/ABallard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks, Jim $3.2; 3.6; 8.3$ Banks, Wesley $2.4; 2.13$ Bannon, Robert 12.2 Barbieri, Lynn $2.4; 2.13$ Baresich, Dennis $2.4; 2.13$ Barager, John $2.4; 2.13$ Bariana, Ava $2.4; 2.13$ Barker, DonaldN/A	Balay, Joanne & Joseph	2.4; 2.13
Baley, Patricia $2.4; 2.13$ Balfour, MicheleN/ABalken, Eric $2.8; 2.14; 13.1$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Ball, Evelyn $2.4; 2.13$ Ball, WilliamN/ABallard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks, Jim $3.2; 3.6; 8.3$ Banks, JaniceN/ABanks, Wesley $2.4; 2.13$ Bannon, Robert 12.2 Barbieri, Lynn $2.4; 2.13$ Barger, John $2.4; 2.13$ Bariana, Ava $2.4; 2.13$ Barker, DonaldN/ABarker, Greg $2.8; 2.14$	Baldwin, Leland	N/A
Balfour, MicheleN/ABalken, Eric2.8; 2.14; 13.1Balken, Eric (Glen Canyon Institute)2.1; 2.2; 2.8; 2.14Ball, Evelyn2.4; 2.13Ball, WilliamN/ABallard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks, Jim3.2; 3.6; 8.3Banks, JaniceN/ABanks, Wesley2.4; 2.13Bannon, Robert12.2Barbieri, Lynn2.4; 2.13Barger, John2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/A	Baley, Patricia	2.4; 2.13
Balken, Eric $2.8; 2.14; 13.1$ Balken, Eric (Glen Canyon Institute) $2.1; 2.2; 2.8; 2.14$ Ball, Evelyn $2.4; 2.13$ Ball, WilliamN/ABallard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks , Jim $3.2; 3.6; 8.3$ Banks, JaniceN/ABannon, Robert 12.2 Barbieri, Lynn $2.4; 2.13$ Baresich, Dennis $2.4; 2.13$ Barger, John $2.4; 2.13$ Bariana, Ava $2.4; 2.13$ Barker, DonaldN/ABarker, Greg $2.8; 2.14$	Balfour, Michele	N/A
Balken, Eric (Glen Canyon Institute)2.1; 2.2; 2.8; 2.14Ball, Evelyn2.4; 2.13Ball, WilliamN/ABallard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks , Jim3.2; 3.6; 8.3Banks, JaniceN/ABannon, Robert12.2Barbieri, Lynn2.4; 2.13Baresich, Dennis2.4; 2.13Barger, John2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Balken, Eric	2.8; 2.14; 13.1
Ball, Evelyn2.4; 2.13Ball, WilliamN/ABallard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks , Jim3.2; 3.6; 8.3Banks, JaniceN/ABanks, Wesley2.4; 2.13Bannon, Robert12.2Barbieri, Lynn2.4; 2.13Baresich, Dennis2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Balken, Eric (Glen Canyon Institute)	2.1; 2.2; 2.8; 2.14
Ball, WilliamN/ABallard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks , Jim3.2; 3.6; 8.3Banks, JaniceN/ABanks, Wesley2.4; 2.13Bannon, Robert12.2Barbieri, Lynn2.4; 2.13Baresich, Dennis2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Ball, Evelyn	2.4; 2.13
Ballard, CynthiaN/ABaltrip Balagas, AyanaN/ABanks , Jim3.2; 3.6; 8.3Banks, JaniceN/ABanks, Wesley2.4; 2.13Bannon, Robert12.2Barbieri, Lynn2.4; 2.13Baresich, Dennis2.4; 2.13Barger, John2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Ball, William	N/A
Baltrip Balagas, AyanaN/ABanks, Jim3.2; 3.6; 8.3Banks, JaniceN/ABanks, Wesley2.4; 2.13Bannon, Robert12.2Barbieri, Lynn2.4; 2.13Baresich, Dennis2.4; 2.13Barger, John2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Ballard, Cynthia	N/A
Banks, Jim3.2; 3.6; 8.3Banks, JaniceN/ABanks, Wesley2.4; 2.13Bannon, Robert12.2Barbieri, Lynn2.4; 2.13Baresich, Dennis2.4; 2.13Barger, John2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Baltrip Balagas, Ayana	N/A
Banks, JaniceN/ABanks, Wesley2.4; 2.13Bannon, Robert12.2Barbieri, Lynn2.4; 2.13Baresich, Dennis2.4; 2.13Barger, John2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Banks, Jim	3.2; 3.6; 8.3
Banks, Wesley2.4; 2.13Bannon, Robert12.2Barbieri, Lynn2.4; 2.13Baresich, Dennis2.4; 2.13Barger, John2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Banks, Janice	N/A
Bannon, Robert12.2Barbieri, Lynn2.4; 2.13Baresich, Dennis2.4; 2.13Barger, John2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Banks, Wesley	2.4; 2.13
Barbieri, Lynn2.4; 2.13Baresich, Dennis2.4; 2.13Barger, John2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Bannon, Robert	12.2
Baresich, Dennis2.4; 2.13Barger, John2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Barbieri, Lynn	2.4; 2.13
Barger, John2.4; 2.13Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Baresich, Dennis	2.4; 2.13
Bariana, Ava2.4; 2.13Barker, DonaldN/ABarker, Greg2.8; 2.14	Barger, John	2.4; 2.13
Barker, DonaldN/ABarker, Greg2.8; 2.14	Bariana, Ava	2.4; 2.13
Barker, Greg 2.8; 2.14	Barker, Donald	N/A
	Barker, Greg	2.8; 2.14
Barkley, David	3.2; 3.6; 8.3	
------------------------------------	----------------------------------	
Barkman, Jeff	N/A	
Barlow, Coni	2.4	
Barmann, Adriene	2.4; 2.13	
Barnes, Alexander	2.4; 2.13	
Barnes, Linda Sue	N/A	
Barnett, Lance	N/A	
Barondes, Lynda	N/A	
Barr, Ford	2.4; 2.8; 2.13	
Barradas, Carlos	2.4; 2.13	
Barrett, Dorie	2.4; 2.10	
Barrington, Tim	2.4	
Barry, Karyn	N/A	
Bartkowicz, Richard	2.4; 2.13	
Bartleman, Mark	2.4; 2.13	
Basford, Jerry	3.2; 3.6; 8.3	
Bashore. Thomas	2.4: 2.13	
Bass. Linda	N/A	
Bates, Bryan	2.4: 2.11: 2.13	
Batina, John	N/A	
Battaly Robert	2 4 2 13	
Battle Adriana	N/A	
Bauman Joan-Marie	2 4 2 13	
Baxter Lou	2 4 2 13	
Baver Robert	3 2: 3 6: 8 3	
Beam Stephanie	N/A	
Beatini Tom	2 4. 2 13	
Beattie Jane H	N/A	
Bechtel Paul	2 4: 2 13	
Beck Deborah	N/A	
Beck, Kim	2 4: 2 13	
Becker Elaine	2 4 2 13	
Bedell Diane	N/A	
Begave Russell (The Navaio Nation)	4 2: 7 5: 8 1: 10 2: 10 8	
Beil William	2 4. 2 13	
Belcastro Bernadette	2.4, 2.13	
Beleastro, Frank	2.4, 2.13	
Bell Don	N/A	
Belles Mark	N/A	
Bello D	2 4: 2 13	
Bellows David	N/Δ	
Bender Jerry	3 2 3 6 8 3	
Bender, Kae	2 A	
Bendiven Kirsten	2. π N/Δ	
Benedetto Mona Stenhanie	N/A	
Benford Alan	$2 4 \cdot 2 13$	
Benjamin Frank	Δ , λ , λ	
Bonnett F	2 2 3 6 8 2	
Bennett Fave	N/A	
Bonnett Mary	$2 \Lambda \cdot 2 13$	
Denneu, Iviai y	2.4, 2.13	

Bennett, Robert	2.4; 2.13
Bennion, Joseph	2.4; 2.13
Bennion, Lee	2.13; 4.1.2;; 12.2
Benson, Andy	2.4; 2.13
Benson, William	2.4; 2.13
Benware, BettyAnn	2.4; 2.13
Bergen, Joanne	N/A
Bergeron, Brad	2.4
Berglund, John	3.6
Bergstedt, Charlie	2.4; 2.13
Bernon, Douglas	2.4
Bernstein, Joel	2.4; 2.13
Bernstein, Laura Ann K	N/A
Bertani, Christy	2.4; 2.13
Berteaux, Elizabeth	2.4
Bessette, Wayne	2.4; 2.13
Bhattacharji, Lee	2.4
Bickel, Michael	2.4; 2.13
Biddle. Maxine	2.4: 2.13
Bieritz, David	N/A
Biggs Susan	N/A
Billeaud Theresa	2 4 2 13
Bilodeau-Lanne Michelle	N/A
Bilton Gretchen	2 4. 2 13
Bilwin Gina	2.1, 2.13
Bing Donna	2.1, 2.13
Bircher K Kay	2.1, 2.13
Bird Paul	2.4, 2.13
Bishon Cori	2.4, 2.13
Bishop, Coll Bishop, Norman	2.1, 2.15
Blackwall Marchant Patricia	2. 1 N/A
Black Frank	$2 4 \cdot 2 13$
Diance, Frank Planton, Dobin	2.4, 2.13
Dianton, Koom	$1\sqrt{A}$
Diaustein, John Diaifald Nail	2.4, 2.13
Bielleid, Nell	2.4, 2.13
Bloch, Mark	2.4, 2.13
Blombach, Ann	2.4; 2.15
Blome, Konald	N/A
Bloxsom, Daniel	3.2; 3.6; 8.3
Boardman, Scott	3.2; 3.6
Bobb, Douglas	3.6
Bobnick, Jacqueline	N/A
Bocchino, Jackie	2.4
Boche, Philip	N/A
Bockino, Alida	2.4; 2.13
Bodine, Frank	2.4; 2.13
Boguske, Matthew	N/A
Boisvert, Chantal	2.4; 2.13
Bolen, Dk	2.4
Bollinger, Lesley	2.4

	Bonds, John	2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
	Boone, James	2.4
	Boone, Jim	2.4; 2.13
	Booth, Robert	2.4
	Borie, Edith	2.4; 2.13
	Boris, Donna	2.4; 2.13
	Bornholtz, Gavin	2.4; 2.13
	Borrelli, Silvana	2.4
	Bortoletto, Federico	N/A
	Boston, Lovie	N/A
	Bottrell, Gregory	3.2; 3.6; 8.3
	Boucher, J	2.4; 2.13
	Boumali, Omar	N/A
	Bowden, Joan	2.4; 2.13
	Bowen, Mary Ellen	2.4; 2.13
	Bowers Margarita	14.3
	Bowers, Carla	2.4
	Bowers, Robert	3.2; 3.6
	Bowie, Martin	N/A
	Boyce, Ashley	2.4
	Boyce, Justin	2.4
	Boyd, Kerry	2.4: 2.13
	Boydston, Charlene	2.4; 2.13
	Bover, David	2.4; 2.13
	Boynton, Robin	2.4; 2.13
	Brachman, Phyllis	2.4; 2.13
	Bradley, Kathy	2.4; 2.13
	Bradley, Mark	2.4; 2.13
	Bradley, Shari	N/A
	Bradley, Stacey	2.4; 2.13
	Bradshaw, Barbara	N/A
	Bradshaw, Jacqui	N/A
	Braithwaite, Georgia	2.4; 2.13
	Bramblett, Sharon	2.4; 2.13
	Brandt, Deborah	N/A
	Brandt, Vicky	2.8
	Brault, Robert	N/A
	Breckenridge, Bonnie	2.4; 2.13
	Breckenridge, Claudia	3.2; 3.6
	Breedlove, Elizabeth	2.4; 2.13
	Brehmer, Dianne	N/A
	Breitenstein, Mark	3.2; 3.6; 12.2
	Brenner, Jared	2.4; 2.13
	Bridges, Linda	N/A
	Brigham, Lawrence	2.4; 2.13
	Brinker, Mary Jo	N/A
	Brinkley, Mike	2.4; 2.13
	Brochhagen, Ann	2.4; 2.13
	Brocious, Pamela	2.4
	Brockett, Peter	2.4; 2.13
-		

Brockway, Barbara	2.4; 2.13
Broe, Paula	2.4; 2.13
Bromage, Joan	2.4; 2.13
Brookover, Cicely	3.2; 3.6; 8.3
Brooks, William	3.2; 3.6; 8.3
Brooks, Carrie	2.4; 2.13
Brooks, Dorothy	N/A
Brooks, Prudence	N/A
Brown, James	2.4; 2.13
Brown, Jessica	2.4; 2.13
Brown, Lolly	2.1; 2.2; 2.8; 2.14; 13.1
Browning, Henry	N/A
Bruce, John	2.4
Brunick. Cathy	2.4: 2.13
Brunson, Richard	3.6
Bruton, Babette	2.4: 2.13
Bryan Karol	N/A
Bryant Elizabeth	2 4 2 13
Buck Mary Lou	2 4
Budlong Tom	21
Budziack Thomas	3.6.12.2
Buech Heidi	2 4
Buell Nancy	2.4
Bular Ed	3 2 3 6
Bullis Robert	2 4: 2 13
Bunch Fugene	N/Δ
Bundy Jay	3 2 3 6 8 3
Burback Heidi	2 4: 2 13
Burger Nancy	N/Δ
Burgess Wendy	$2 4 \cdot 2 13$
Burgett Debbie	2.4, 2.13
Burk Dobert	Σ ,
Durk, Robert	1V/A 2 9
Durke, Diyan Durke, Mauraan	2.0
Durke, Mauleen	2.4, 2.13
Durks, Faul Durland Marcia	1N/A
Burraga Edward	2.4, 2.13
Burstoin Mimi	2.4, 2.13
Duisteili, Millill Durt Dorm	2.4, 2.15
Duil, Dally Duashat-las Thomas (Arizona Donartmant of Water	5.2, 5.0, 6.5
Buschatzke, Thomas (Arizona Department of Water	2.4, 3.0, 5.2, 8.3, 12.11, 15.1, 15.3, 15.4
Resources)	2.4.2.12
Busnnell, Martna	2.4; 2.13
Buss, william	2.4; 2.13
Busse, Chris	2.4
Buszmann, Jett	N/A
Butter, Sam	2.4; 2.13
Buttacavoli, Knonda	2.4
Buvala, Ken	3.2; 3.6; 8.3
Byerley, Erica	2.4
Byland, John	N/A

	Durman Vally	2.4
	Byrum Noney	2.4
	C Dob	2.4, 2.13
	C., DOU	5.2, 5.0, 8.5 2.4: 2.12
	Capaliero, Luis	2.4, 2.13
	Cacciapuoli, Anniony	2.4, 2.13
	Calena Jana	N/A
	Caldwall Domala	
	Caldwell, Palliela	2.4 N/A
	California Charles	
	Calle, Charles	2.4
	Callagirone, Michael	3.2; 3.0; 8.3 2 0: 2 (: 8 2
	Camp, ward	3.2; 3.0; 8.3 2.4: 2.12
	Campbell, Allan	2.4; 2.13
	Campbell, Grant (South Florida Audubon Society)	2.4
	Campbell, Jacqueline	N/A
	Campbell, Jerry	2.4; 2.13
	Campbell, Nancy	2.4; 2.13
	Campbell, Susan	N/A
	Cannon, John	2.4; 2.13
	Cano, Martha	2.4; 2.13
	Canter, Linda	N/A
	Cantor, Francine	2.4; 2.13
	Caplin, Marilyn	N/A
	Caputo, Michael	2.4; 2.13
	Carley, Daniel	2.4; 2.13
	Carlile, Carol	N/A
	Carlson, J. Tyler (Mohave Electric Cooperative, Inc.)	2.4; 2.10
	Carlson, Robin	N/A
	Carlson, Sandy	2.4; 2.13
	Carnahan, Michael	12.4
	Carney, Marilyn	2.4; 2.13
	Carolan, Daniel	3.2; 3.6; 8.3
	Carpenter, Steven	2.4; 2.13
	Carrasco Serrano, Grace	2.4; 2.13
	Carrico, John	2.4; 2.13
	Carrillo, Stephen	2.4; 2.13
	Carrington, Martha	2.4; 2.13
	Carroll, Celia (Wilderness Fly Fishers of Santa Monica)	2.5; 3.2; 3.3; 3.4; 3.5; 12.2
	Carter, D.	2.4; 2.13
	Carter, Michelle	2.4; 2.13
	Case, Nicole (Northwest Public Power Association)	2.6; 2.10
	Casey, Nancy	N/A
	Cassens, Clarence	3.2; 3.6
	Castri, Serenella	2.4
	Caswell, Susan	N/A
	Catches, Margaret	2.4
	Catlin, Linda	2.4; 2.13
	Caywood, Wayne	N/A
	Cefola, Elaine	N/A
	Chaffin, Claudia	2.4; 2.13
-		

Chalfen, Karen	2.4
Chalker, Mikki	2.4; 2.13
Chambers, John	2.1; 2.2; 2.8; 2.14; 13.1
Chambless, Rochelle	2.4
Chang, Bill	2.4; 2.13
Chapman, Andrew	2.4
Chappell, Christina	2.4; 2.13
Chardon, Denise	2.4; 2.13
Chase, Linda	2.4; 2.13
Chasse, Rob	2.1; 2.2; 3.6
Chati, Janet	N/A
Chavet, Kent	3.2; 3.6; 8.3
Chelak, Mark	N/A
Chen, Allan	2.4; 2.13
Cherrington, Howard	2.4
Child, Katrina	2.4
Christ, Andrew	2.4; 2.13
Christensen, Kelly	3.6; 12.2
Christie, Tess	N/A
Christman, Margaret	2.4; 2.13
Christman, Mary	N/A
Christo, Jeffrey	N/A
Christopher, Bruce	2.4; 2.13
Christopher, Lucy	2.4; 2.13
Chunko, Mike	3.2; 3.6; 8.3
Chunko, Mike	3.2; 3.6
Chwalisz, Bart	2.4; 2.13
Claggett, Suzanne	2.4; 2.13
Claridge, Jeanne	2.4; 2.13
Clark, Carolyn	2.4; 2.13
Clark, James A. Jr.	2.4; 2.13
Clark, Pam	2.4
Clark, Todd	2.4; 2.13
Clark, Zulma	N/A
Clarke, Tom	2.4; 2.13
Claus, Carol	2.4; 2.13
Clayton, David	N/A
Clement, Susan	2.4
Cleveland, Al	2.5
Cling, Marvin	2.8
Clinton, 86001	N/A
Clough, Cyndi	2.4; 2.13
Coahran, Scott	N/A
Coates, Charles, Sr.	3.2; 3.6; 8.3
Cobb, Robert	2.4; 2.13
Cobb, Sandra	2.4; 2.13
Coble, Melvin	3.6
Coburn, Della	N/A
Coffee, Eileen	N/A
Cohen, Annelisa	N/A

Cohen, Daniel	2.4; 2.13
Cohen, David	N/A
Cohen, Francine	2.4; 2.13
Cohen, Howard	2.4; 2.13
Cohen, Judy	2.4; 2.13
Cohen, Sheara	2.4; 2.13
Cohen, Yoel	2.4; 2.13
Coleman, Timothy	2.4; 2.13
Colingsworth, Julia	N/A
Collins, Carol	2.4; 2.13
Collins, Jennifer	2.4; 2.13
Colten, Lora	2.4
Colton, Joseph	3.2; 3.6; 8.3
Columbia, James	2.4; 2.13
Colver. Frank	2.1: 2.2: 2.8: 2.14: 13.1
Colver. Matthew	2.1: 2.2: 2.8: 2.14: 13.1
Combes Dale	N/A
Comella John	N/A
Comer Patrick	2 4. 2 13
Commarata Anthony	2 4 2 13
Conforti Susan	2.1, 2.13
Conner Kathleen	2.4, 2.13
Conner Lisa	N/Δ
Conred Lori	2 A· 2 13
Conroy Beverly	N/Λ
Conroy, Beverly	$1\sqrt{A}$
Conroy, Fatti	2.4, 2.13
Conroy, James	2.4, 2.13
Colle Darmy	2.4, 2.15 N/A
Cook, Dally Cook, Charlotta	IN/A
Cook, Charlotte	N/A
Cook, Jason	3.2; 3.0
Cook, Joy and Mike	N/A N/A
Cook, Susan	N/A
Cook, Suzanne	2.4; 2.13
Cooley, Marian	2.4; 2.13
Coon, John	2.4; 2.13
Coonfield, Sherril	N/A
Coons, Kathryn	2.4; 2.13
Cooper, Charlene	2.4; 2.13
Cope, Denys	2.4; 2.13
Copeland, Sue	N/A
Copenhaver, Patricia	N/A
Coppotelli, Heide Catherina	2.4; 2.13
Corby, Kathleen	2.4; 2.13
Cordaro, Tom	2.4
Cordero, David and Ann	2.4; 2.13
Corkey, Peter	N/A
Cornelia, Jared	2.4; 2.13
Corriere, Jim	N/A
Corry, Ronit	N/A

Costa, Lynn	2.4; 2.13
Cottingham, Charlie	N/A
Couch, Sandra	N/A
Coulter, D. Mountainhawk	3.2; 3.6; 8.3
Coulter, David	3.2; 3.6; 8.3
Counterman, Michael	12.2
Courtney, Andy	3.2; 3.6
Courtright, Anne	2.4; 2.13
Covell, Sandi	2.4; 2.13
Cowan, Christina	2.4; 2.13
Cowin, Caryn	2.4; 2.13
Cox, Chadwick	2.4; 2.13
Cox, Jerry	2.5
Cracchiolo, Clara	N/A
Craig. Ann	2.4: 2.13
Cramer, Marilyn	2.4: 2.13
Cresseveur Jessica	2 4 2 13
Creswell Sandra	N/A
Croft Samuel	N/A
Crosland Richard	2 4: 2 13
Cross Dave and Rita	2 4: 2 13
Crow Benita	2 4: 2 13
Crowley John	N/A
Cruger Kurt	N/A
Crumpacker Barb	N/A
Crumpacker, Nancy	2 4: 2 13
Cuddy William	2 4: 2 13
Cuff Kermit	2.4, 2.13
Cui Michael	N/A
Culmore Matthew	N/A
Cummings Joan	N/A
Cunningham David	3 2: 3 6: 8 3
Curtis Helen	2 4 2 13
Cutting Ken	12.4
Cwyk Francine	3 2 3 6
Cyzner Steven	2 4
Czingula Christian	N/A
D Mary	2 4: 2 13
Daetz Douglas and Gisela	2.4. 2.13
Dagney Orysia	N/A
Dailey Susan	2 4: 2 13
Dalla John	2.4. 2.13
Dalla, John Dally, Charles	2.7, 2.13
D'Ambra John	Σ.¬, Σ.15 N/Δ
Daniel Kian	N/Δ
D'Anne Denise	N/Δ
Dannels Paul	1 N/ FA 2 7 · 2 6 · 8 2
Dannevia John	5.2, 5.0, 0.5 N/A
Damevig, John Dosher, Don	1N/2N
Dashiri, Duli Dashiril, Jamas	2.4, 2.13
Dasmen, James	2.4, 2.13

Dashiell, Marniyon $24, 2.13$ Dashiell, Marrilyn $24, 2.13$ Davis, Alam $32, 36, 8.3$ Davis, Eleanor-Ann $24, 2.13$ Davis, Glenn $24, 2.13$ Davis, Sileanor-Ann $24, 2.13$ Davis, Jian $24, 2.13$ Davis, Vian N/A Del La Torre, John $24, 2.13$ Dean, NancyN/ADecora, Kerri $2.1, 2.2, 2.8, 2.14, 13.1$ Decell, Kerri $2.1, 2.2, 2.8, 2.14, 13.1$ Deese, DonnaN/ADeGrav, Catherine 2.4 Del Valle, JavierN/ADelaney, Janet $2.4, 2.13$ Delaney, Janet $2.4, 2.13$ Demontigner, SandraN/ADemonigner, SandraN/ADennigner, SandraN/ADevin, JohnN/ADevin, SummerN/ADevin, SummerN/ADevin, Summer		
Dashell, Marilyn $2.4; 2.13$ Davidson, Paul and MaryN/ADavis, Alan $3.2; 3.6; 8.3$ Davis, Eleanor-Ann $2.4; 2.13$ Davis, Lisa $2.4; 2.13$ Daviss, Lisa $2.4; 2.13$ Dawson-Barker, JoelleN/ADe La Torre, John $2.4; 2.13$ Davas, Glenn $2.4; 2.13$ Dawson-Barker, JoelleN/ADe La Torre, John $2.4; 2.13$ Deamoells, VictorN/ADeborah, N/AN/ADeborah, N/AN/ADecese, DonnaN/ADefayelis, VictorN/ADefay, Catherine 2.4 Deill, Richard 2.4 Deill, Richard 2.4 Del Valle, JavierN/ADelaney, PatrickN/ADelaney, Janet $2.4; 2.13$ Deleney, PatrickN/ADelaney, Janet $2.4; 2.13$ Deleney, Rathew $2.4; 2.13$ Demonbrun, CarlN/ADemorest, Cynthia $2.4; 2.13$ Dennobrun, CarlN/ADennobrun, CarlN/ADennobrun, CarlN/ADernoty, JohnN/ADernoty, JohnN/ADernoty, SadraN/ADevin, JohnN/ADevin, JohnN/ADevin, SummerN/ADevin, Summer $2.3; 3.6; 8.3$ Derninger, SandraN/ADimator, NanadaN/ADimator, Summer $2.4; 2.13$ Dimator, Summer $2.4; 2.13$ Dimator, NanadaN/ADimator	Dashiell, Mannyson	2.4; 2.13
Davis AlanN/ADavis, Alan $32, 36, 8.3$ Davis, Eleanor-Ann $24, 2.13$ Davis, Lia $24, 2.13$ Davis, Lia $24, 2.13$ Dawson-Barker, JoelleN/ADe La Torre, John $24, 2.13$ Dawson-Barker, JoelleN/ADehan, NancyN/ADeborah, N/AN/ADecoll, Kerri $2.1, 2.2, 2.8, 2.14, 13.1$ Descraw, Catherine 2.4 Deffar, Catherine 2.4 Della, Richard 2.4 Devalle, Richard 2.4 Della, Richard 2.4 Della, SancyN/ADelaney, PatrickN/ADelaney, PatrickN/ADelaney, Janet $2.4, 2.13$ Delaney, Janet $2.4, 2.13$ Delaney, Janet $2.4, 2.13$ Delaney, Janet $2.4, 2.13$ Demoren, CarlN/ADemoren, CarlN/ADennobrun, CarlN/ADennobrun, CarlN/ADenvin, John $3.2, 3.6, 8.3$ Denninger, SandraN/ADerwin, SummerN/ADeskins, Merrill 2.4 Deskins, Merrill $2.4, 2.13$ Deskins, Merrill $2.4, 2.13$ Deskins, Merrill 2.4 Diamante, NinaN/ADeskins, Merrill 2.4 Diamante, NinaN/ADiamante, NinaN/ADiamante, NinaN/ADiamante, NinaN/ADiamante, Nina $3.2, 3.6, 8.3$ Dillon, John (Grand Canyon River Outfitters	Dashiell, Marilyn	2.4; 2.13
Davis, Alan $32, 36, 8.3$ Davis, Glenn $24, 2.13$ Davis, Glenn $24, 2.13$ Davis, Lisa $24, 2.13$ Davoson-Barker, JoelleN/ADe La Torre, John $24, 2.13$ Dean, NancyN/ADeAngelis, VictorN/ADecoll, Kerri $2.1, 2.2, 2.8, 2.14, 13.1$ Deese, DonnaN/ADeill, Kichord 2.4 Deill, Kichord 2.4 Deill, Kichord 2.4 Deill, Kichord 2.4 Dell Valle, JavierN/ADelaney, PatrickN/ADelaney, Janet $2.4, 2.13$ Deltognoarmanasco, John $2.4, 2.13$ Deltognoarmanasco, John $2.4, 2.13$ Demorstur, Cynthia $2.4, 2.13$ Demorstur, Cynthia $2.4, 2.13$ Demorstur, Cynthia $2.4, 2.13$ Dennobrun, CarlN/ADennobrun, CarlN/ADennobrun, CarlN/ADennobrun, CarlN/ADennobrun, CarlN/ADennobrun, CarlN/ADennobrun, CarlN/ADennobrun, CarlN/ADennobrun, CarlN/ADennobrun, SandraN/ADennobrun, SandraN/ADennobrun, SandraN/ADennobrun, SandraN/ADennobrun, SandraN/ADennobrun, SandraN/ADevin, JohnN/ADevin, SummerN/ADevin, SummerN/ADiamante, Nia $2.2, 36, 8.3$ Dilman, Michael $2.4, 2.13$ <	Davidson, Paul and Mary	N/A
Davis, Eleanor-Ann2.4; 2.13Davis, Lisa $2.4; 2.13$ Davis, Lisa $2.4; 2.13$ Dawson-Barker, JoelleN/ADe La Torre, John $2.4; 2.13$ Dean, NancyN/ADechoraln, N/AN/ADeckoraln, N/AN/ADecoraln, N/AN/ADecoraln, N/AN/ADecoraln, Catherine 2.4 Del Valle, JavierN/ADelarot, P. Attick, Catherine 2.4 Del Valle, JavierN/ADelaney, Janet $2.4; 2.13$ Deleney, Janet $2.4; 2.13$ Deleney, Janet $2.4; 2.13$ Deleney, Janet $2.4; 2.13$ Deleney, Janet $2.4; 2.13$ Demostrun, CarlN/ADemorest, Cynthia $2.4; 2.13$ Deniser, PaulaN/ADemostrun, CarlN/ADemostrun, CarlN/ADemostrun, CarlN/ADernest, SandraN/ADerninger, SandraN/ADervin, JohnN/ADeskins, Merrill $2.4; 2.3; 3.6; 8.3$ Dervin, JohnN/ADeskins, Merrill $2.4; 2.13; 3.6; 3.3; 5.1; 9.3; 12.3; 3.6; 3.3; 5.1; 9.3; 12.3; 3.6; 3.3; 5.1; 9.3; 12.3; 3.6; 3.3; 5.1; 9.3; 12.3; 5.1; 3.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2; 3.6; 8.3; 5.1; 12.1; 12.2;$	Davis , Alan	3.2; 3.6; 8.3
Davis, Glenn $24; 2.13$ Davis, Lisa $24; 2.13$ Dawson-Barker, JoelleNADe La Torre, John $24; 2.13$ Dean, NancyN/ADeAngelis, VictorN/ADecell, Kerri $2.1; 2.2; 2.8; 2.14; 13.1$ Deese, DonnaN/ADefil, Kichard 2.4 Del Valle, JavierN/ADelaney, Janet $2.4; 2.13$ Delaney, Janet $2.4; 2.13$ Delaney, Janet $2.4; 2.13$ Delaney, Janet $2.4; 2.13$ Deloney, Janet $2.4; 2.13$ Demors, Gymman $2.4; 2.13$ Demors, Gymman $2.4; 2.13$ Demors, Cynthia $2.4; 2.13$ Demors, Cynthia $2.4; 2.13$ Denses, PaulaN/ADenniger, SandraN/ADenriser, SundraN/ADenringer, SandraN/ADervin, JohnN/ADevine, Timothy $3.2; 3.6; 8.3$ Devine, Timothy $3.2; 3.6; 8.3$ Devine, SummerN/ADevine, Timothy $3.2; 3.6; 8.3$ Devine, Caranea and Fish Department) $3.2; 3.6; 8.3$ Difulin, JohnN/ADietriech, Kevin $3.2; 3.6; 8.3$ Dillon, John (Grand Canyon River Outfitters $2.4; 2.13$ Dimanate, NinaN/ADillon, John (Grand Canyon River Outfitters $2.4; 2.13$ Dimanet, Strad 2.4 Dimock, Brad 2.4 <	Davis, Eleanor-Ann	2.4; 2.13
Davis, Lisa $2.4; 2.13$ Dawson-Barker, JoelleN/ADe La Torre, John $2.4; 2.13$ Dean, NancyN/ADeAngelis, VictorN/ADeborah, N/AN/ADecese, DonnaN/ADeGraw, Catherine 2.4 Del Valle, JavierN/ADel Valle, JavierN/ADelaney, PatrickN/ADelehant, Raymond $2.4; 2.13$ Delehant, Raymond $2.4; 2.13$ Delehant, Raymond $2.4; 2.13$ Delehant, Raymond $2.4; 2.13$ Delehant, Raymond $2.4; 2.13$ Demorbun, CarlN/ADennortun, CarlN/ADennortun, CarlN/ADennedy, Frank, Daniel $3.2; 3.6; 8.3$ Denninger, SandraN/ADervin, JohnN/ADesroe, PaulaN/ADerokes, DonN/ADevins, SummerN/ADestore, John $3.2; 3.6; 8.3$ Dervin, JohnN/ADestore, JohnN/ADestore, DonN/ADestore, Jawa $3.2; 3.6; 3.3; 5.1; 9.3; 12.3$ Diamante, NinaN/ADiamante, Nina $3.2; 3.6; 8.3$ Dillon, John Grand Canyon River Outfitters $3.2; 3.6; 8.3$ Dillon, John Grand Canyon River Outfitters $3.2; 3.6; 8.3$ Dillon, John Grand Canyon River Outfitters $2.4; 2.13$ Dillon, John Grand Canyon River Outfitters $2.4; 2.13$ Dillon, John Grand Canyon River Outfitters $2.4; 2.13$ Dillon, John Grand Canyon River Outfitters $2.4; 2.1$	Davis, Glenn	2.4; 2.13
Dawson-Barker, JoelleN/ADe La Torre, John2.4; 2.13Dean, NancyN/ADehnapelis, VictorN/ADeborah, N/AN/ADecell, Kerri2.1; 2.2; 2.8; 2.14; 13.1Deese, DonnaN/ADeGraw, Catherine2.4Delh, Richard2.4Dellar, SchartN/ADelarey, PatrickN/ADelaney, PatrickN/ADelaney, Janet2.4; 2.13Delhan, Raymond2.4; 2.13Delebant, Raymond2.4; 2.13Delebant, Raymond2.4; 2.13Demorbrun, CarlN/ADemorst, Cynthia2.4; 2.13Dennisten, PaulaN/ADennisten, PaulaN/ADennisten, PaulaN/ADennisten, PaulaN/ADennisten, PaulaN/ADennisten, PaulaN/ADennisten, PaulaN/ADenninger, SadraN/ADesroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devine, NinaN/ADesroches, DonN/ADevine, StartN/ADevine, NinaN/ADiamante, NinaN/ADiamante, NinaN/ADietriech, Kevin3.2; 3.6; 8.3Dillon, John (Grand Canyon River Outfitters2.13; 2.5; 12.1; 12.2Association)2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 2.5; 12.1; 12.2Association2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Associat	Davis, Lisa	2.4; 2.13
De La Torre, John $2.4; 2.13$ Dean, NancyN/ADeAngelis, VictorN/ADeborah, N/AN/ADecese, DonnaN/ADeGraw, Catherine $2.1; 2.2; 2.8; 2.14; 13.1$ Deese, DonnaN/ADeGraw, Catherine 2.4 Del Valle, JavierN/ADel Valle, JavierN/ADelaney, Janet $2.4; 2.13$ Delehant, Raymond $2.4; 2.13$ Delehant, Raymond $2.4; 2.13$ Delognarmanasco, John $2.4; 2.13$ Demonbrun, CarlN/ADemost, Cynthia $2.4; 2.13$ Dennsen, CarlN/ADenney, Frank, Daniel $3.2; 3.6; 8.3$ Denninger, SandraN/ADerwing, Kerrill 2.4 Destons, Merrill 2.4 Destons, Merrill 2.4 Destons, Merrill 2.4 Destons, Merrill 2.4 Devine, Timothy $3.2; 3.6; 8.3$ Derwine, SimmerN/ADevine, SummerN/ADevine, SummerN/ADibliBlasi, Dawn 2.4 Dickinson, AmadaN/ADietriech, Kevin $3.2; 3.6; 8.3$ Dillin, John (Grand Canyon River Outfitters $2.4; 2.13$ Diamante, Kirad 2.4 Dillon, John (Grand Canyon River Outfitters $2.4; 2.13$ Dillon, John (Grand Canyon River Outfit	Dawson-Barker, Joelle	N/A
Dean, NancyN/ADeAngelis, VictorN/ADeborah, N/AN/ADecorah, N/AN/ADecell, Kerri $2.1; 2.2; 2.8; 2.14; 13.1$ Deese, DonnaN/ADeGraw, Catherine 2.4 Deihl, Richard 2.4 Delaney, Janet $2.4; 2.13$ Delaney, Janet $2.4; 2.13$ Delbaney, Janet $2.4; 2.13$ Delbaney, Janet $2.4; 2.13$ Delbaney, Janet $2.4; 2.13$ Delbaney, Janet $2.4; 2.13$ Denomprumanasco, John $2.4; 2.13$ Demorst, Cynthia $2.4; 2.13$ Demorst, Cynthia $2.4; 2.13$ Dennohrun, CarlN/ADenney, Frank, Daniel $3.2; 3.6; 8.3$ Denney, Frank, Daniel $3.2; 3.6; 8.3$ Denniger, SandraN/ADeskins, Merrill 2.4 Deskons, JamesN/ADeskins, Merrill 2.4 Derlor, JohnN/ADevley, James (Arizona Game and Fish Department) $3.2; 3.6; 3.9; 5.1; 9.3; 12.3$ Diamante, NinaN/ADiblajai, Dawn 2.4 Dickinson, AmadaN/ADietriech, Kevin $3.2; 3.6; 8.3$ Dillon, John (Grand Canyon River Outfitters $2.13; 8.5; 12.1; 12.2$ Association) $-3.4; 2.13$ Dillon, John (Grand Canyon River Outfitters $2.13; 8.5; 12.1; 12.2$ Association) $-3.4; 2.13$ Dillon, John (Grand Canyon River Outfitters $2.4; 2.13$ Dimock, Brad 2.4 Dimock, Brad 2.4 Dinmock, Donal	De La Torre, John	2.4; 2.13
DeAngelis, VictorN/ADecorl, N/AN/ADecorl, Kerri2.1; 2.2; 2.8; 2.14; 13.1Deese, DonnaN/ADefind, Richard2.4Delihl, Richard2.4Del Valle, JavierN/ADelaney, PatrickN/ADelaney, PatrickN/ADelognoarmanasco, John2.4; 2.13Deltognoarmanasco, John2.4; 2.13Demobrun, CarlN/ADemostru, CarlN/ADenney, Fank, Daniel3.2; 3.6; 8.3Denninger, SandraN/ADervin, JohnN/ADervin, JohnN/ADervin, JohnN/ADevine, Timothy3.2; 3.6; 8.3Devine, Timothy3.2; 3.6; 8.3Devine, Timothy3.2; 3.6; 8.3Dervine, Jimothy3.2; 3.6; 8.3Dervine, Timothy3.2; 3.6; 8.3Devine, Timothy3.2; 3.6; 8.3Dillanante, NinaN/ADickinson, AmandaN/ADickinson, AmandaN/ADickinson, AmandaN/ADickinson, AmandaN/ADickinson, AmandaN/ADickinson, AmandaN/ADirbure, Kevin3.2; 3.6; 8.3Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)TaliDillon, Nichael2.4; 2.13Dillon, Nichael2.4; 2.13Dillon, Nichael2.4; 2.13Dillon, Nichael2.4; 2.13Dillon, S. Brad2.4Dimmock, DonaldN/ADinmock, DonaldN/	Dean, Nancy	N/A
Deborah, N/AN/ADeceel, Kerri $2.1; 2.2; 2.8; 2.14; 13.1$ Deese, DonnaN/ADeGraw, Catherine 2.4 Delh, Richard 2.4 Del Valle, JavierN/ADelaney, PatrickN/ADelaney, Janet $2.4; 2.13$ Delhogmannaneso, John $2.4; 2.13$ Demonbrun, CarlN/ADenorst, Cynthia $2.4; 2.13$ Denis, SandraN/ADenorst, Cynthia $2.4; 2.13$ Dennobrun, CarlN/ADenniger, SandraN/ADenniger, SandraN/ADenviny, Frank, Daniel $3.2; 3.6; 8.3$ Dervin, JohnN/ADevrin, JohnN/ADevine, Timothy $3.2; 3.6; 3.9; 5.1; 9.3; 12.3$ Diamante, NinaN/ADiamante, NinaN/ADietvin, MandaN/ADevlin, SummerN/ADevlin, SummerN/ADietrine, Kevin $3.2; 3.6; 3.9; 5.1; 9.3; 12.3$ Diamante, NinaN/ADietriech, Kevin $3.2; 3.6; 8.3$ Dietriech, Kevin $3.2; 3.6; 8.3$ Dietriech, Kevin $3.2; 3.6; 8.3$ Diillon, John (Grand Canyon River Outfitters $2.13; 8.5; 12.1; 12.2$ Association) $-24; 2.13$ Dillon, John (Grand Canyon River Outfitters $2.4; 2.13$ Dimock, DonaldN/ADim	DeAngelis, Victor	N/A
Decell, Kerri2.1; 2.2; 2.8; 2.14; 13.1Decese, DonnaN/ADeGraw, Catherine2.4Deihl, Richard2.4Delaney, PatrickN/ADelaney, Janet2.4; 2.13Delhant, Raymond2.4; 2.13Delhant, Raymond2.4; 2.13Delmosy, Janet2.4; 2.13Delongoarmanasco, John2.4; 2.13Demonbrun, CarlN/ADemorest, Cynthia2.4; 2.13Densen, PaulaN/ADennedy, Frank, Daniel3.2; 3.6; 8.3Denninger, SandraN/ADervin, JohnN/ADeskins, Merrill2.4Devins, SummerN/ADevins, SummerN/ADevins, SummerN/ADiffanda, Dawn2.4; 2.13Dervin, JohnN/ADeskins, Merrill2.4Devins, SummerN/ADevins, SummerN/ADevins, MinaN/ADiffanan, Michael2.4; 2.13Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)Juillon, John (Grand Canyon River OutfittersDillina, Kibrad2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)Juillon, John (Grand Canyon River OutfittersDillina, Nichael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)Juinya, 8.14, 2.13Dillon, Stawrence2.4; 2.13Dinmock, DonaldN/A<	Deborah, N/A	N/A
Deese, DonnaN/ADeGraw, Catherine2.4Deihl, Richard2.4Del Valle, JavierN/ADelaney, PatrickN/ADelaney, Janet2.4; 2.13Deltognoarmanasco, John2.4; 2.13Deltognoarmanasco, John2.4; 2.13Demobrun, CarlN/ADemost, Cynthia2.4; 2.13Denost, CarlN/ADennobrun, CarlN/ADennissen, PaulaN/ADennedy, Frank, Daniel3.2; 3.6; 8.3Denniger, SandraN/ADesvin, JohnN/ADesvins, Merrill2.4Desvins, Marthew2.4Devins, JohnN/ADesroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devine, SummerN/ADevine, SummerN/ADiflasti, Dawn2.4Dickinson, AmandaN/ADillasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)Jillon, John (Grand Canyon River OutfittersDillon, John Strad2.4Dimock, Brad2.4Dimock, DonaldN/ADingman, S. Lawrence2.4; 2.13	Decell, Kerri	2.1; 2.2; 2.8; 2.14; 13.1
DeGraw, Catherine2.4Deilh, Richard2.4Del Valle, JavierN/ADelaney, PatrickN/ADelaney, Janet2.4; 2.13Delehant, Raymond2.4; 2.13Deltognoarmanasco, John2.4; 2.13Demonbrun, CarlN/ADemorest, Cynthia2.4; 2.13Denorest, Cynthia2.4; 2.13Denorest, Cynthia2.4; 2.13Dennedy, Frank, Daniel3.2; 3.6; 8.3Denninger, SandraN/ADervin, JohnN/ADessen, PaulaN/ADervin, JohnN/ADervin, JohnN/ADeskins, Merrill2.4Deskins, Merrill2.4Deskins, Merrill2.4Deskins, Merrill2.4Deskins, Merrill2.4Devine, Timothy3.2; 3.6; 8.3Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADiblasi, Dawn2.4Dickinson, AmandaN/ADillen, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingnan, S. Lawrence2.4; 2.13	Deese, Donna	N/A
Deihl, Richard2.4Del Valle, JavierN/ADelaney, PatrickN/ADelaney, Janet2.4; 2.13Delehant, Raymond2.4; 2.13Deltognoarmanasco, John2.4; 2.13DeMars, Matthew2.4; 2.13Demorburn, CarlN/ADemorest, Cynthia2.4; 2.13Denses, PaulaN/ADennoedy, Frank, Daniel3.2; 3.6; 8.3Denniger, SandraN/ADervin, JohnN/ADervin, JohnN/ADervin, JohnN/ADervin, JohnN/ADevine, Timothy3.2; 3.6; 8.3Devine, Carizona Game and Fish Department)3.2; 3.6; 8.3Diamate, NinaN/ADistlines, Kevin3.2; 3.6; 8.3Dinmante, NinaN/ADiblasi, Dawn2.4Dicktrisch, Kevin3.2; 3.6; 8.3Dillinan, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)Jills, 8.5; 12.1; 12.2MassociationJills, 8.5; 12.1; 12.2Dimock, Brad2.4Dimock, Brad2.4Dimock, DonaldN/ADingman, S. Lawrence2.4; 2.13	DeGraw, Catherine	2.4
Del Valle, JavierN/ADelaney, PatrickN/ADelaney, Janet2.4; 2.13Deltognoarmanasco, John2.4; 2.13Deltognoarmanasco, John2.4; 2.13Demonbrun, CarlN/ADemorst, Cynthia2.4; 2.13Denis, Laurie2.4; 2.13Dennsken, PaulaN/ADennedy, Frank, Daniel3.2; 3.6; 8.3Denninger, SandraN/ADervin, JohnN/ADeskins, Merrill2.4Devine, Timothy3.2; 3.6; 8.3Devine, SummerN/ADeskins, Merrill2.4Destoches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devline, SummerN/ADevline, Timothy3.2; 3.6; 8.3Devline, Timothy3.2; 3.6; 8.3Dillasi, Dawn2.4Diblasi, Dawn2.4Dietriech, Kevin3.2; 3.6; 8.3Dillnan, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)Jills, RobertDills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Deihl, Richard	2.4
Delaney, Patrick N/A Delaney, Janet $2.4; 2.13$ Delehant, Raymond $2.4; 2.13$ Deltognoarmanasco, John $2.4; 2.13$ Demorst, Ratthew $2.4; 2.13$ Demorst, Cynthia $2.4; 2.13$ Denorest, Cynthia $2.4; 2.13$ Denis, Laurie $2.4; 2.13$ Denssen, Paula N/A Dennedy, Frank, Daniel $3.2; 3.6; 8.3$ Denninger, Sandra N/A Dervin, John N/A Desches, Don N/A Devine, Timothy $3.2; 3.6; 8.3$ Devlin, Summer N/A Devin, Summer N/A Devin, John N/A Destins, Merrill 2.4 Dickinson, Amanda N/A Dietriech, Kevin $3.2; 3.6; 8.3$ Dillasi, Dawn 2.4 Dickinson, Amanda N/A Dietriech, Kevin $3.2; 3.6; 8.3$ Dillman, Michael $2.4; 2.13$ Dillon, John (Grand Canyon River Outfitters $2.1; 8.5; 12.1; 12.2$ Association) $2.4; 2.13$ Dimock, Brad 2.4 Dimock, Brad 2.4 Dimock, Donald N/A Dingleberry, Pat 2.4 Dingleberry, Pat 2.4 Dingman, S. Lawrence $2.4; 2.13$	Del Valle, Javier	N/A
Delaney, Janet $2.4; 2.13$ Delehant, Raymond $2.4; 2.13$ Deltognoarmanasco, John $2.4; 2.13$ DeMars, Matthew $2.4; 2.13$ Demonbrun, CarlN/ADemorest, Cynthia $2.4; 2.13$ Denis, Laurie $2.4; 2.13$ Denis, Laurie $2.4; 2.13$ Dennedy, Frank, Daniel $3.2; 3.6; 8.3$ Denninger, SandraN/ADervin, JohnN/ADesches, DonN/ADevine, Timothy $3.2; 3.6; 8.3$ Devining, SummerN/AdeVos, James (Arizona Game and Fish Department) $3.2; 3.6; 3.9; 5.1; 9.3; 12.3$ Diamante, NinaN/ADietriech, Kevin $3.2; 3.6; 8.3$ Diillman, Michael $2.4; 2.13$ Dillon, John (Grand Canyon River Outfitters $2.4; 2.13$ Dillon, John (Grand Canyon River Outfitters $2.4; 2.13$ Dimock, Brad 2.4 Dimock, DonaldN/ADiedering, Stavence $2.4; 2.13$	Delaney, Patrick	N/A
Delehant, Raymond2.4; 2.13Deltognoarmanasco, John2.4; 2.13Demohrun, CarlN/ADemorest, Cynthia2.4; 2.13Denis, Laurie2.4; 2.13Denissen, PaulaN/ADennedy, Frank, Daniel3.2; 3.6; 8.3Denninger, SandraN/ADervin, JohnN/ADestones, DonN/ADestins, Metrill2.4Destones, Continue3.2; 3.6; 8.3Denninger, SandraN/ADervin, JohnN/ADeskins, Metrill2.4Destones, DonN/ADevine, Timothy3.2; 3.6; 8.3Devlin, SummerN/ADiblasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4Dills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S, Lawrence2.4; 2.13	Delaney, Janet	2.4; 2.13
Deltognoarmanasco, John2.4; 2.13DeMars, Matthew2.4; 2.13Demonbrun, CarlN/ADemorest, Cynthia2.4; 2.13Denises, Cynthia2.4; 2.13Denissen, PaulaN/ADennedy, Frank, Daniel3.2; 3.6; 8.3Denninger, SandraN/ADervin, JohnN/ADeskins, Merrill2.4Desroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devlin, SummerN/ADevline, Timothy3.2; 3.6; 8.3Devlin, SummerN/ADislasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillan, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4; 2.13Dimock, Brad2.4Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S, Lawrence2.4; 2.13	Delehant, Raymond	2.4; 2.13
DeMars, Matthew2.4; 2.13Demonbrun, CarlN/ADemorest, Cynthia2.4; 2.13Denis, Laurie2.4; 2.13Denissen, PaulaN/ADennedy, Frank, Daniel3.2; 3.6; 8.3Denniger, SandraN/ADervin, JohnN/ADeskins, Merrill2.4Desroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADickinson, AmandaN/ADietriech, Kevin2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4; 2.13Dillos, Rad2.4Dimock, Brad2.4Dingleberry, Pat2.4Dingleberry, Pat2.4Dingleberry, Pat2.4; 2.13	Deltognoarmanasco, John	2.4; 2.13
Demonbrun, CarlN/ADemorest, Cynthia2.4; 2.13Denis, Laurie2.4; 2.13Denissen, PaulaN/ADennedy, Frank, Daniel3.2; 3.6; 8.3Denninger, SandraN/ADervin, JohnN/ADeskins, Merrill2.4Desroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADiskinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)Jills, RobertDills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	DeMars, Matthew	2.4; 2.13
Demorest, Čynthia2.4; 2.13Denis, Laurie2.4; 2.13Denissen, PaulaN/ADennedy, Frank, Daniel3.2; 3.6; 8.3Denninger, SandraN/ADervin, JohnN/ADeskins, Merrill2.4Desroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADistriech, Kevin2.4Dickinson, AmandaN/ADillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dimann, S. Lawrence2.4; 2.13	Demonbrun, Carl	N/A
Denis, Laurie2.4; 2.13Denissen, PaulaN/ADennedy, Frank, Daniel3.2; 3.6; 8.3Denninger, SandraN/ADervin, JohnN/ADeskins, Merrill2.4Desroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADiBlasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4Dills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S, Lawrence2.4; 2.13	Demorest, Cynthia	2.4; 2.13
Denissen, PaulaN/ADennedy, Frank, Daniel3.2; 3.6; 8.3Denninger, SandraN/ADervin, JohnN/ADeskins, Merrill2.4Desroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADiblasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4Dimock, Brad2.4Dimock, Brad2.4Dimock, Prad2.4Dimock, Prad2.4Dimock, Prad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S, Lawrence2.4; 2.13	Denis, Laurie	2.4; 2.13
Dennedy, Frank, Daniel3.2; 3.6; 8.3Denninger, SandraN/ADervin, JohnN/ADeskins, Merrill2.4Desroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADiBlasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.4; 2.13Dimock, Brad2.4Dimock, Brad2.4Dimock, Prad2.4Dimock, Parad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S, Lawrence2.4; 2.13	Denissen, Paula	N/A
Denninger, SandraN/ADervin, JohnN/ADeskins, Merrill2.4Desroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADiBlasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Dennedy, Frank, Daniel	3.2; 3.6; 8.3
Dervin, JohnN/ADeskins, Merrill2.4Desroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADiBlasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4Dinock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Denninger, Sandra	N/A
Deskins, Merrill2.4Desroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADiBlasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)Dills, RobertDills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S, Lawrence2.4; 2.13	Dervin, John	N/A
Desroches, DonN/ADevine, Timothy3.2; 3.6; 8.3Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADiBlasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S, Lawrence2.4; 2.13	Deskins, Merrill	2.4
Devine, Timothy3.2; 3.6; 8.3Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADiBlasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Desroches, Don	N/A
Devlin, SummerN/AdeVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADiBlasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Devine, Timothy	3.2; 3.6; 8.3
deVos, James (Arizona Game and Fish Department)3.2; 3.6; 3.9; 5.1; 9.3; 12.3Diamante, NinaN/ADiBlasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4; 2.13Dills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Devlin, Summer	N/A
Diamante, NinaN/ADiBlasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4; 2.13Dills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	deVos, James (Arizona Game and Fish Department)	3.2; 3.6; 3.9; 5.1; 9.3; 12.3
DiBlasi, Dawn2.4Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4; 2.13Dills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Diamante, Nina	N/A
Dickinson, AmandaN/ADietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4; 2.13Dills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	DiBlasi, Dawn	2.4
Dietriech, Kevin3.2; 3.6; 8.3Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4; 2.13Dills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Dickinson, Amanda	N/A
Dillman, Michael2.4; 2.13Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4; 2.13Dills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Dietriech, Kevin	3.2; 3.6; 8.3
Dillon, John (Grand Canyon River Outfitters2.13; 8.5; 12.1; 12.2Association)2.4; 2.13Dills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Dillman, Michael	2.4; 2.13
Association)2.4; 2.13Dills, Robert2.4Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Dillon, John (Grand Canyon River Outfitters	2.13; 8.5; 12.1; 12.2
Dills, Robert2.4; 2.13Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Association)	
Dimock, Brad2.4Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Dills, Robert	2.4; 2.13
Dimock, DonaldN/ADingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Dimock, Brad	2.4
Dingleberry, Pat2.4Dingman, S. Lawrence2.4; 2.13	Dimock, Donald	N/A
Dingman, S. Lawrence 2.4; 2.13	Dingleberry, Pat	2.4
	Dingman, S. Lawrence	2.4; 2.13
DiRenzo, Jennifer 2.4; 2.13	DiRenzo, Jennifer	2.4; 2.13
Dishion, Catherine N/A	Dishion, Catherine	N/A
Diss, Marybeth 2.4	Diss, Marybeth	2.4
DiVall, Nelson N/A	DiVall, Nelson	N/A

Dixon, Joyce	2.4; 2.13
Doherty, Jeanne	2.4
Dombroski, Ed	N/A
Dominiak, Adam	2.4
Donnelly, Stephen	2.4; 2.13
D'Onofrio, Adam	2.4; 2.13
Donovan, Elaine	2.4; 2.13
Dorfman, Ellen	2.4
Dorn, John	2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
Dorr, Kelly	N/A
Dougherty, Janet	2.4; 2.13
Downes, Susan	2.4; 2.13
Downing, Larry	N/A
Dowson, Eleanor	2.4; 2.13
Drahos, Ronald	2.4; 2.13
Drake, Mercy	N/A
Drake, Tracy	2.4; 2.13
Dresben, Fred	3.2; 3.6; 8.3
Drever, Daniel	N/A
Driessen, Lynn	2.4; 2.13
Driskell, Shelley	2.4: 2.13
Drummond, William (Mid-West Electric Consumers	2.1
Association)	
Duckworth Michael	12.4
Duffus. Kathleen	N/A
Duion Phyllis	2 4 2 13
Dunaetz Ron	N/A
Dunbrack Jan	N/A
Duncan Kimberly	2 4. 2 13
Duncan Pat	2.4
Duncan Renee	2.4
Dunkley Julianne	2.1 2 4. 2 13
Dunn Elmo	2 4 2 13
Dunn Timothy	2 4 2 13
Dunn W Ronald (Strawberry Anglers Association)	3 2: 3 6: 8 3
Dunwell Nancy	2 4 · 2 13
Durrer Mary	N/A
Durrum Kathy	N/A
Dutschke Stenhen	2 <u>4</u> · 2 13
Dzhonova Iveta	2.1, 2.13
Fames Mary	2.4, 2.13
Early Lance	2.4, 2.13
Earnshaw Shinann	2.4, 2.15
Faton Kathleen	2.1
Eaton Lorena	2.7, 2.15 7 A
Earon, Eorona Fhert Frik	2. 7 2 <i>A</i>
Eduri, Elik Eduria Lisca	2. 4 N/A
Eddy Lee Gladys	$2 \Lambda \cdot 2 13$
Edwanson Michele	2.4, 2.13 N/A
Edmunda Alicio	
Eumunds, Alicia	1N/A

Egger, Kathleen	2.4; 2.13
Eglin, Jennifer	2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
Eichel, Richard	2.4; 2.13
Eisenberg, Andrea	N/A
Eive, Gloria	2.4; 2.13
Elkins, E	2.4; 2.13
Ellenbecker, Marvin	N/A
Elliot, Nan	2.4; 2.13
Elliott, Lynn	2.4; 2.13
Elliott, R	2.4
Ellis, Julie	N/A
Elson, Adam	2.4; 2.13
Embry, Judith	2.4; 2.13
Emerson, Chelsea	2.4; 2.13
Emery, Susan	2.4; 2.13
Enger, Carolyn	N/A
Engler, Pamela	2.4; 2.13
English, Jeff	12.4
Entress, Greg	3.6; 12.4
Epstein, Harry	N/A
Epstein, Kelly	2.4; 2.13
Erb, Cheryl	N/A
Erbs, Lori	2.4; 2.13
Ernst, Cathie	2.4
Escobar, Victor	2.4; 2.13
Espinoza, Debra	2.4; 2.13
Estel, Karen	2.4
Estrada, Jennifer	2.4; 2.13
Evans, Bronwen	2.4; 2.13
Ewert, Henry	N/A
Ewing, Peter	2.4; 2.13
Ewoldsen, Brooke	N/A
F, Annette	14.3
Fabian, Bill	2.4; 2.13
Faegre, Dirk	2.4
Fahey, Keith	2.4
Fairchild-Ehm, Audrey	N/A
Faires, APRIL	N/A
Falcone, Janet	2.4; 2.13
Falcone, Pamela	N/A
Falconer, Russ	3.2; 3.6; 8.3
Fallon, Laura	2.12
Fantano, Theodore	2.4
Farber, Carol	2.4; 2.13
Farmer, Bonnie	2.4; 2.13
Fary, Jim	N/A
Fasano, Marianne	2.4; 2.13
Fear, Marge	N/A
Fearey, Donna	2.4; 2.13
Fehr, Richard	2.4

Fei, Eddijohn	2.4
Fenley, Bette-Burr	N/A
Ferguson, Charlene	N/A
Ferguson, David	3.2; 3.6; 8.3
Ferguson, Roy	3.2; 3.6; 8.3
Ferman, Pam	N/A
Fermin, Christina	N/A
Fernandez, Daniel	2.4; 2.13
Fickling, Karl	2.4; 2.13
Fidler, Vicki	N/A
Fiederer, Conrad	N/A
Fiedler, David	2.4; 2.13
Field, Patrick	2.4
Fielder, Linda	2.4
Fields, Mark J	2.4
Finch, Bonnie	N/A
Fine, Cindy	2.4; 2.13
Fink, Christine	N/A
Fink, Harry	2.4; 2.13
Finkelstein, David	2.4; 2.13
Finkelstein, Sheldon	2.4; 2.13
Finley, Diane	N/A
Finley, Patricia	2.4; 2.13
Finn, Peter	2.4; 2.13
Fisch, Greg	2.4; 2.13
Fischer, Elaine	2.4; 2.13
Fischer, Fred	2.4; 2.13
Fish, Richard	2.4; 2.13
Fisher, Judith	2.4
Fisher, Michael	N/A
Fishleder, Sam	N/A
Fitzpatrick, John	2.4; 2.13
Fladager, Susan	2.4; 2.13
Flanagan, Marianne	2.4; 2.13
Flanigan, Kevin (New Mexico Interstate Stream	5.2; 6.4.12; 8.11; 13.2, 13.4
Commission)	
Fleetwood, Patricia	N/A
Fleischer, Tim	2.4; 2.13
Fleming, Nancy	2.4
Fletcher, Carol	2.4; 2.13
Fletcher, Ken	3.2; 3.6; 8.3
Flickinger, R Charles	2.4; 2.13
Flores, Regina	2.4; 2.13
Flory, John	2.1; 2.2; 2.8; 2.14; 13.1
Floyd, Joel	2.4; 2.13
Fluker, Richard	2.4
Fogarty, Geraldin	N/A
Foley, Doris	N/A
Follansbee, Meghan	N/A
Fonken, Miryam	2.4

Forbes, Doreen2.4Ford, Michael2.4; 2.13Forlie, Kai2.8Forschner, Jillian2.4; 2.13Forwand, ArleneN/AFosburgh, Eric2.4; 2.13Fost, Rebecca2.4Foster, Dawn2.4; 2.13Fouche, David2.4; 2.13Fox, CinnamonN/AFox, Lyndsey2.1; 2.2; 2.8; 2.14; 13.1Fraikor, Fred3.2; 3.6; 8.3Franchi, Irena2.4; 2.13Franchi, Irena2.4; 2.13Franckowiak, PaulN/AFranco, Merrill2.4; 2.13
Ford, Michael2.4; 2.13Forlie, Kai2.8Forschner, Jillian2.4; 2.13Forwand, ArleneN/AFosburgh, Eric2.4; 2.13Fost, Rebecca2.4Foster, Dawn2.4; 2.13Fouche, David2.4; 2.13Fox, CinnamonN/AFox, Lyndsey2.1; 2.2; 2.8; 2.14; 13.1Fraikor, Fred3.2; 3.6; 8.3Franchi, Irena2.4; 2.13Francis, StuartN/AFranckowiak, PaulN/AFranco, Merrill2.4; 2.13
Forlie, Kai2.8Forschner, Jillian2.4; 2.13Forwand, ArleneN/AFosburgh, Eric2.4; 2.13Fost, Rebecca2.4Foster, Dawn2.4; 2.13Fouche, David2.4; 2.13Fox, CinnamonN/AFox, Lyndsey2.1; 2.2; 2.8; 2.14; 13.1Fraikor, Fred3.2; 3.6; 8.3Franchi, Irena2.4; 2.13Francki, IrenaN/AFrancki, PaulN/AFranckowiak, PaulN/AFranco, Merrill2.4; 2.13
Forschner, Jillian2.4; 2.13Forwand, ArleneN/AFosburgh, Eric2.4; 2.13Fost, Rebecca2.4Foster, Dawn2.4; 2.13Fouche, David2.4; 2.13Fox, CinnamonN/AFox, Lyndsey2.1; 2.2; 2.8; 2.14; 13.1Fraikor, Fred3.2; 3.6; 8.3Franchi, Irena2.4; 2.13Francki, StuartN/AFranckowiak, PaulN/AFranco, Merrill2.4; 2.13
Forwand, ArleneN/AFosburgh, Eric2.4; 2.13Fost, Rebecca2.4Foster, Dawn2.4; 2.13Fouche, David2.4; 2.13Fox, CinnamonN/AFox, Lyndsey2.1; 2.2; 2.8; 2.14; 13.1Fraikor, Fred3.2; 3.6; 8.3Franchi, Irena2.4; 2.13Francki, StuartN/AFranckowiak, PaulN/AFranco, Merrill2.4; 2.13
Fosburgh, Eric 2.4; 2.13 Fost, Rebecca 2.4 Foster, Dawn 2.4; 2.13 Fouche, David 2.4; 2.13 Fox, Cinnamon N/A Fox, Lyndsey 2.1; 2.2; 2.8; 2.14; 13.1 Fraikor, Fred 3.2; 3.6; 8.3 Franchi, Irena 2.4; 2.13 Francis, Stuart N/A Franckowiak, Paul N/A Franco, Merrill 2.4; 2.13
Fost, Rebecca 2.4 Foster, Dawn 2.4; 2.13 Fouche, David 2.4; 2.13 Fox, Cinnamon N/A Fox, Lyndsey 2.1; 2.2; 2.8; 2.14; 13.1 Fraikor, Fred 3.2; 3.6; 8.3 Franchi, Irena 2.4; 2.13 Francis, Stuart N/A Franckowiak, Paul N/A Franco, Merrill 2.4; 2.13
Foster, Dawn 2.4; 2.13 Fouche, David 2.4; 2.13 Fox, Cinnamon N/A Fox, Lyndsey 2.1; 2.2; 2.8; 2.14; 13.1 Fraikor, Fred 3.2; 3.6; 8.3 Franchi, Irena 2.4; 2.13 Francis, Stuart N/A Franckowiak, Paul N/A Franco, Merrill 2.4; 2.13
Fouche, David 2.4; 2.13 Fox, Cinnamon N/A Fox, Lyndsey 2.1; 2.2; 2.8; 2.14; 13.1 Fraikor, Fred 3.2; 3.6; 8.3 Franchi, Irena 2.4; 2.13 Francis, Stuart N/A Franckowiak, Paul N/A Franco, Merrill 2.4; 2.13
Fox, Cinnamon N/A Fox, Lyndsey 2.1; 2.2; 2.8; 2.14; 13.1 Fraikor, Fred 3.2; 3.6; 8.3 Franchi, Irena 2.4; 2.13 Francis, Stuart N/A Franckowiak, Paul N/A Franco, Merrill 2.4; 2.13
Fox, Lyndsey 2.1; 2.2; 2.8; 2.14; 13.1 Fraikor, Fred 3.2; 3.6; 8.3 Franchi, Irena 2.4; 2.13 Francis, Stuart N/A Franckowiak, Paul N/A Franco, Merrill 2.4; 2.13
Fraikor, Fred3.2; 3.6; 8.3Franchi, Irena2.4; 2.13Francis, StuartN/AFranckowiak, PaulN/AFranco, Merrill2.4; 2.13
Franchi, Irena2.4; 2.13Francis, StuartN/AFranckowiak, PaulN/AFranco, Merrill2.4; 2.13
Francis, StuartN/AFranckowiak, PaulN/AFranco, Merrill2.4; 2.13
Franckowiak, PaulN/AFranco, Merrill2.4; 2.13
Franco, Merrill 2.4; 2.13
Frank, Henry 2.4
Frank, Mitzi 2.4
Frankel, Myrna 2.4; 2.13
Frantz-Crafton, Candy N/A
Franzmann, Paul 2.4; 2.13
Fraser, Bonnie 2.4; 2.13
Fraser, Evelyn 2.4; 2.13
Frazee, Janis 2.4; 2.13
Fredricks, Judith 2.4; 2.13
Frerichs, Joan N/A
Frey, Brenda N/A
Frey, Lawrence 2.4; 2.13
Friedel, Diane 2.4; 2.13
Friedman, Thomas N/A
Friehauf, Mike 3.2; 3.6; 8.3
Friel, Bernard 2.4
Friel, Bernard P. 2.4
Fromowitz, Allen and Carol N/A
Fuessel, Chere N/A
Fugate, John 3.2; 3.6; 8.3
Fugina, Vincent 3.2; 3.6; 8.3
Fuller, George 2.4; 2.13
Fuller, Roy 2.4; 2.13
Fulmer, James 2.4
Furno, Saragh N/A
Fusi, David 3.2; 3.6; 8.3
G, C. 2.4; 2.13
G, Derek 2.2; 3.6; 12.2
Gabriel, Robert 2.4; 2.13
Gaines, Marsha 2.4; 2.13
Galindo, Thomas 3.2; 3.6; 8.3
Gallo, Gina N/A
Gambriel, John 2.4; 2.13
Gannon, Kristine N/A

Garber, Barbara	2.4; 2.13
Garcia, Jeffery	2.4; 2.13
Gardner, Chris	2.4
Garey, JG	2.4; 2.13
Garnas , D	3.2; 3.6; 8.3
Garratt, Liz	2.4; 2.13
Garrity, Coleen	2.4; 2.13
Garton, Katie	2.4; 2.13
Garvey, Patrick	2.4
Gates, Joanne	N/A
Gauthier, Dale	3.6
Gay, Larry	2.4; 2.13
Geis, Emery	3.2; 3.6; 8.3
Gelbart, Susannah	N/A
Gelles, Kat	2.4; 2.13
Gensler, Donna	N/A
George, Thomas (State of Colorado, Colorado Water	2.2: 2.5: 2.15: 2.20: 6.1.2: 6.1.3: 6.4.12: 6.4.13: 8.6:
Conservation Board)	8.7: 8.8: 8.12: 10.8: 12.6: 13.4
George, Thomas (Arizona, California, Colorado,	2.2: 2.5: 2.19: 2.20: 3.6: 5.1
Nevada New Mexico Utah Wyoming & Upper	
Colorado River Commission)	
Gerak Edward (Buckeye Water Conservation &	2 2 2 9 3 5 5 2 11 1
Drainage District)	,,,,,,
Gerber Steve	12.2
Gerstenfeld Judith	3 2: 3 6
Ghirardelli Susan	2 4
Giantomasi David	N/A
Giasson Bertrand	N/A
Gibbon Jocelyn	2 4. 2 14
Gibson Jody	2.7, 2.17
Giere Linda	2.4, 2.13
Gilbert Don	Σ.4, Σ.15 Ν/Λ
Cill Stophonic	
Cillica Dan	
Cillicon Michael	N/A 2 4: 2 12
Cilliland Charles	2.4, 2.15
Cilmore Mure	2.4
Gilmore, Myra	2.4, 2.13
Ginn, Kenneth	2.4, 2.15
Girard, Brian	N/A
Girvin, Darryin	2.4; 2.13
Gist, Sally Citlin Alizzar (Sizzar Chab. Crand Concern Chapter)	2.4
Gitlin, Alicyn (Sierra Club - Grand Canyon Chapter)	2.8; 2.14; 3.1; 3.4; 3.8; 7.1; 8.3; 13.1; 14.1; 14.2
Glascock, Katherine	N/A
Glass, Perri	N/A
Glasser, Mark	N/A
Gleaton, Dianne	N/A
Glenn, Alice Ann	2.4; 2.13
Glessing, Kathryn	N/A
Gliva, Stephen	2.4; 2.13
Gloeckler, Bill (Grand Canyon Whitewater)	12.1; 12.2

Gloyd, Jan	N/A
Giynn, Jonn Gadich Manzie	2.4; 2.13
Godich, Marcia	2.4; 2.13
Goebel, Fred	2.4; 2.13
Goebel, Lawrence	
Goertz, Harry	3.2; 3.6; 8.3
Goetschius, Lascinda	N/A
Goforth, Kathleen (U.S. Environmental Protection	N/A
Agency, Region IX)	
Golden, Gene	N/A
Golden, Kathleen	2.4
Goldstein, Arthur	N/A
Goldstein, Dale	2.4; 2.13
Gonzales Nielsen, Rosemarie	N/A
Gonzalez, Joy	N/A
Gonzalez, William G	2.4; 2.13
Goodale, Margaret	2.4; 2.13
Goodrich, D'Arcy	2.4; 2.13
Goodwin, Elizabeth	2.4; 2.13
Goodwin, Margaret	N/A
Goodwin, Tom	3.2; 3.3; 3.6
Gorrin, Eugene	2.4; 2.13
Gorsetman, Mark	2.4; 2.13
Gottlieb, Olga S.	2.4; 2.13
Gould, Michael	N/A
Gould, Jo Anne	N/A
Gover, Pat and Gary	2.4; 2.13
Goyette, Marc	N/A
Goyette, Margo	2.4
Grabar, Christine	N/A
Grace, George	2.4; 2.13
Grace, Harry	2.4; 2.13
Grace, Howard	N/A
Graffagnino, Mary Ann and Frank	2.4; 2.13
Graham, James	2.4; 2.13
Graham, James	3.2; 3.6
Graham, Jim	2.4; 2.13
Grames, Patricia	2.4; 2.13
Granakis, George	2.4
Granlund, Fred	2.4; 2.13
Grasso, Jen	N/A
Gray, Heather	2.4; 2.13
Gray, Richard	3.2; 3.6
Gray, Tony	N/A
Green, Jesse	2.4; 2.13
Greene, Minna	N/A
Greenfield, Cariln	2.4; 2.13
Greenhalgh, Diana	2.4; 2.13
Greenhill, Barry	N/A
Greenlee, Philip (International Federation of Fly Fishers)	2.5; 3.2; 3.3; 3.4; 3.6; 12.2

Gregory, Paul	N/A
Gregory, Probyn	3.2; 3.6; 8.3
Greif, Kevin	2.4
Grenard, Mark Hayduke	2.8
Grenci, Ann	N/A
Gresko, Andrew	N/A
Grierson, Phillip	2.4; 2.13
Griffin, Byron	2.4; 2.13
Griffin, Leah	2.4; 2.13
Griffiths Vega, Joan	2.4; 2.13
Grisez, Bernard	2.4; 2.13
Griswold, Dave	2.4; 2.13
Griswold, Tracy	2.4; 2.13
Gritsch, Maria	2.4; 2.13
Grom, Jeff	N/A
Groshardt, Joanne	N/A
Gross, Linda	N/A
Gross. Todd	N/A
Grove, Earl	2.4: 2.13
Grove. Stephen	2.4: 2.13
Grover, Justin	3.2: 3.6: 8.3
Grover, Justin	2.4: 2.13
Grube, Mary Alvce	2.5: 2.17: 3.2: 3.3: 3.4: 3.6: 8.3: 12.2
Grzanna Maureen	N/A
Guard Mary	2 4 2 13
Guckian Michael	2 1 2 2 2 8 2 14 13 1
Guenther Lavaune	N/A
Guile Roger	3.6
Guinnun David	2 4: 2 13
Guion William	2 4 2 13
Gullam Paul	2 4 2 13
Gunning John	2 4 2 13
Gunter Karlene	2 4 2 13
Gunther Ken	2.1, 2.13
Gutierrez Laura	2.1, 2.13
Gwinn Carol	N/A
H Dennis	2 4: 2 13
Haag James	2.4. 2.13
Haarr I ars	N/A
Haber Steven	2 4: 2 13
Hackel Helena	N/A
Hackett Marcia	2 4
Hackman Jim	2.7
Hackman, Jim Hackney, Stenhen	2.1, 2.2, 2.0, 2.14, 15.1
Haertal Malissa	2.4, 2.13
Hadrich, Michola Hafar Sarah	2.7, 2.13
Hahn Dahart	2.4, 2.13 N/A
Hahn, KOUCH Hahn, Theodore	1N/A
Hahua Donno	
nanus, Donna Lloinea, Thomas	IN/A
naines, 1 nomas	2.4, 2.13

Haley, Patty	N/A
Hall, Beth	2.4
Hall, Diana	2.4; 2.13
Hall, James	2.4; 2.13
Hall, Janice	N/A
Hall, Sue	N/A
Hallett, Mark	2.4; 2.13
Halloran, Michael	N/A
Halpain. Dale	N/A
Hamill John (Trout Unlimited International Federation	2 5 3 2 3 3 3 4 3 6 5 1 8 3 9 3 12 2 12 3
of Fly Fishers)	,,, _,
Hamilton Lynn (Grand Canyon River Guides Inc.)	2 13 4 1 2 7 3 1 9 2 10 3 11 3
Hamilton Roy	N/A
Hammer Brian	N/A
Hammer, Dorothy	N/A
Hammarated Charles	$1\sqrt{A}$
Hammerstau, Charles	5.2, 5.0, 6.5
Hampton, Hugn	N/A 2.2: 2.6
Hand, David	3.2, 3.0
Handelsman, Robert	2.4; 2.13
Hanlon, K	2.4; 2.13
Hansen, Greg	3.2; 3.6; 8.3
Hansen, Jan	2.4; 2.13
Hanson, Bruce & Michelle	2.4; 2.13
Hanson, David	N/A
Hanson, Norman	2.4
Hanson, Norman	N/A
Harden, Ronald	2.4; 2.13
Harding, Steve	3.2; 3.6
Hardt, Vincent	2.4; 2.13
Hardy, Wallace	3.2; 3.6
Harker, William	2.1; 2.2; 2.8; 2.14; 13.1
Harland, Donald	2.4; 2.13
Harmer, Jill	2.4
Harmon, Amy	2.4; 2.13
Harmon, Chris	3.2: 3.6: 8.3
Harmon, Gail	2.4: 2.13
Harned Kristin	2 4 2 11 2 13
Harper Barbara	2 4 · 2 13
Harper, Dennis	3 2 3 6 8 3
Harper, Behecca	2 4 2 13
Harper, Rebecca Harper, Robin	2.4.2.13
Harrington Warren	N/Λ
Harris Condice	$1\sqrt{2}$
Harris Dahra	2.4, 2.13
Harris, Debla	2.4, 2.15 N/A
Hamis, Gall	IN/A
Hamis, James	2.4, 2.13
Harris, Susan	2.4; 2.13
Harrison, David	2.4; 2.13
Harrison, Marie	2.4; 2.13
Hartman, Jenifer	2.4; 2.13

Harrey, Richard 24, 2.13 Harvey, Richard 32, 36, 8.3 Haskell, Michael 24, 2.13 Hask, James NA Hatch, Steve (Hatch River Expeditions) 121, 12, 2 Hathaway, Susan 24, 2, 13 Hauenstein, Cathleen 24, 2, 13 Hauenstein, Cathleen 24, 2, 13 Haver, Richard 24, 2, 13 Haver, Christine 24, 2, 13 Hayes, Christine 24 Hayns, Dennis 24, 2, 13 Hays, Helen Logan 24 Hayowrth, Steven 24, 2, 13 Haynorth, Steven 24, 2, 13 Hadu, Douglas 32, 36, 8, 3 Heap, Finen Logan 24 Haynorth, Steven 24, 2, 13 Head, Kris 24, 2, 13 Head, Douglas 32, 36, 8, 3 Head, Douglas 32, 36, 8, 3 Headper, Find 32, 36, 8, 3 Headper, Find 32, 36, 8, 3		
Harvey, Richard $32; 36; 8.3$ Haskell, Michael $24; 2.13$ Hastip, James N/A Hath, Steve (Hatch River Expeditions) $12, 1, 12.2$ Hathaway, Susan $24; 2.13$ Hauens, David $24; 2.13$ Hauenstein, Cathleen $24; 2.13$ Havek, Robert $24; 2.13$ Havkins, Philip $32; 3.6$ Haycok, Robert $24; 2.13$ Haycok, Robert $24; 2.13$ Hayes, Christine 24 Hayes, Tim N/A Hayes, Stristine $24; 2.13$ Hayes, Kits $24; 2.13$ Hayes, Steven $24; 2.13$ Hayes, Steven $24; 2.13$ Hael, Douglas $32; 3.6; 8.3$ Head, Prancesca 24 Heady, Douglas $32; 3.6; 8.3$ Head, Douglas $32; 3.6; 8.3$ Headper, Joann $24; 2.13$ Headbayes, Bill $32; 3.6; 8.3$ Headpers, David $24; 2.13$ Headpers, David $24; 2.13$ Hedges, Bill $32; 3.6; 8.3$ Hedges, Bill $32; 3.6; 8$	Hartman, Todd	2.4; 2.13
Haskell, Michael $2, 4; 2, 13$ Hastip, James NA Hatch, Steve (Hatch River Expeditions) $12, 1; 12.2$ Haumstein, Cathleen $2, 4; 2, 13$ Haugen, David $2, 4; 2, 13$ Haverfield, Heather $2, 4; 2, 13$ Hayoock, Robert $2, 4; 2, 13$ Hayos, Christine $2, 4$ Hayos, Danna $2, 4; 2, 13$ Hays, Blein Logan $2, 4; 2, 13$ Hayon, Jonana $2, 4; 2, 13$ Hayon, Kits $2, 4; 2, 13$ Hazen, Alona $2, 4; 2, 13$ Head, Kris $2, 2, 36$ Head, Spain $3, 2; 3, 6, 8.3$ Hedgeck, Charles $2, 4; 2, 13$ Hedgeck, Andra $2, 4; 2, 13$ </td <td>Harvey, Richard</td> <td>3.2; 3.6; 8.3</td>	Harvey, Richard	3.2; 3.6; 8.3
Hash, JamesNAHatch, Steve (Hatch River Expeditions) $12.1; 12.2$ Hathaway, Susan $2.4; 2.13$ Hauenstein, Cathleen $2.4; 2.13$ Hauenstein, Cathleen $2.4; 2.13$ Haverifeld, Heather $2.4; 2.13$ Hawkins, Phillip $3.2; 3.6$ Haycock, Robert $2.4; 2.13$ Haycock, Robert $2.4; 2.13$ Haycock, Robert $2.4; 2.13$ Hayes, Christine 2.4 Hayes, Christine 2.4 Hayes, Christine $2.4; 2.13$ Hays, Benian $2.4; 2.13$ Hays, Belen Logan $2.4; 2.13$ Hays, Helen Logan $2.4; 2.13$ Hazen, Alona $2.4; 2.13$ Head, Jouglas $3.2; 3.6; 8.3$ Head, Prancesca $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Nona $2.4; 2.13$ Head, Susan $2.4; 2.13$ Head, Susan $2.4; 2.13$ Heder, Francesca $2.4; 2.13$ Heder, SarbaraN/AHediger, JoAnnN/AHeffer, Joshua $2.4; 2.13$ Hediger, JoAnnN/AHeider, Andra $2.4; 2.13$ Heine, Kathleen $2.4; 2.13$ Heiner, Kathleen $2.4; 2.13$ Henra, Nay 2.4 Henra, DorotheaN/AHerra	Haskell, Michael	2.4; 2.13
Hath, Steve (Hatch River Expeditions) 12,1,12.2 Halmaway, Susan 24,2.13 Hauenstein, Cathleen 24,2.13 Haverfield, Heather 24,2.13 Hawkins, Phillip 32,3.6 Haycock, Robert 24,2.13 Haycock, John 3.6,8.3 Hayes, Christine 2.4 Hays, Strin N/A Hays, Dennis 2.4,2.13 Hays, Helen Logan 2.4 Hayorth, Steven 2.4,2.13 Hayo, Nona 2.4,2.13 Hadaway, Steven 2.4,2.13 Hayo, Helen Logan 2.4 Hayorth, Steven 2.4,2.13 Head, Nis 2.4,2.13 Head, Kris 2.4,2.13 Head, Nona 2.4,2.13 Head, Nona 2.4,2.13 Head, Nona 2.4,2.13 Head, Nona 2.4,2.13 Heade, Naman, Susan 2.4,2.13 Hedgeock, Charles 2.4,2.13 Hedgeock, Charles 2.4,2.13 Hedge, JoAnn N/A Heigeer, JoAnn N/A Heide, Andra 2.4,2.13	Haslip, James	N/A
Hathaway, Susan 24, 2.13 Hauenstein, Cathleen 24, 2.13 Hauerfield, Heather 24, 2.13 Haverifield, Heather 24, 2.13 Haverifield, Heather 24, 2.13 Haycock, Robert 24, 2.13 Hayes, Tim NA Haymans, Deana 24, 2.13 Hays, Dennis 24, 2.13 Hays, Helen Logan 24 Haywoth, Steven 24, 2.13 Headk, Kris 24, 2.13 Headk, Kris 24, 2.13 Headk, Charles 24, 2.13 Headk, Steven 24, 2.13 Hedgeock, Charles 24, 2.13 </td <td>Hatch, Steve (Hatch River Expeditions)</td> <td>12.1; 12.2</td>	Hatch, Steve (Hatch River Expeditions)	12.1; 12.2
Hauenstein, Cathleen $2.4; 2.13$ Haugen, David $2.4; 2.13$ Haverfield, Heather $2.4; 2.13$ Hawkins, Phillip $3.2; 3.6$ Haycock, Robert $2.4; 2.13$ Haydock, John $3.6; 8.3$ Hayes, Christine 2.4 Hayses, Christine 2.4 Hayses, TimN/AHaymans, Deanna $2.4; 2.13$ Hays, Bennis $2.4; 2.13$ Hays, Helen Logan $2.4; 2.13$ Hayen, Nichen Logan $2.4; 2.13$ Hayen, Nichen Logan $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Douglas $3.2; 3.6; 8.3$ Hedgeock, Charles $2.4; 2.13$ Hedgeock, Charles $2.4; 2.13$ Hedgeock, Charles $2.4; 2.13$ Hedgeost, BarbaraN/AHeide, Andra $2.4; 2.13$ Hein, Janet $2.4; 2.13$ Heink, Miden $2.4; 2.13$ Heink, Kick $3.2; 3.6; 8.3$ Heink, Kick $3.2; 3.6; 8.3$ Heink, Kick $3.2; 3.6; 8.3$ Heink, Sherie 2.4 Henninger, Maryann $2.4; 2.13$ Henrik, Kick $3.2; 3.6; 8.3$ Henrik, Michael $2.4; 2.13$ Hernar, David $3.2; 3.6; 8.3$ Hernar, David $3.2; 3.6; 8.3$ Hernar, David $3.2; 3.6; 8.3$ Hernar, Maryann $2.4; 2.13$ Hernar, Maryann $2.4; 2.13$ Hernare, Nilliam $2.4;$	Hathaway, Susan	2.4; 2.13
Haugen, David 24; 2:13 Haverfield, Heather 24; 2:13 Hawkins, Phillip 3; 2; 3.6 Haycock, Robert 24; 2:13 Haycock, Robert 24; 2:13 Haycock, John 3; 6; 8.3 Hayes, Christine 24 Hayes, Tim N/A Hayes, Tim N/A Haynans, Deanna 24; 2:13 Hays, Dennis 24; 2:13 Hays, Helen Logan 24 Hayen, Alona 24; 2:13 Head, Nis 24; 2:13 Head, Douglas 3; 2; 3; 6; 8.3 Head, Douglas 3; 2; 3; 6; 8.3 Head, Douglas 3; 2; 3; 6; 8.3 Heathman, Susan 24; 2:13 Hedgeock, Charles 24; 2:13 Hedgeock, Charles 24; 2:13 Hedgedus, Barbara N/A Hefforn, Joshua 24; 2:13 Heigedus, Barbara N/A Heinder, Janes 24; 2:13 Heinder, Kick 3; 2; 3; 6; 8.3 Heinder, Kathleen 24; 2:13 Heinder, Kick 3; 2; 3; 6; 8.3 Heinty, Bridgett <t< td=""><td>Hauenstein, Cathleen</td><td>2.4; 2.13</td></t<>	Hauenstein, Cathleen	2.4; 2.13
Haverfield, Heather $24; 2.13$ Hawkins, Phillip $32; 3.6$ Haycock, Robert $24; 2.13$ Haydock, John $3.6; 8.3$ Hayes, Christine 2.4 Hayes, Christine 2.4 Hayes, TimN/AHaymans, Deanna $2.4; 2.13$ Hays, Helen Logan $2.4; 2.13$ Hays, Helen Logan $2.4; 2.13$ Hazen, Alona $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, J. Douglas $3.2; 3.6; 8.3$ Head, Douglas $3.2; 3.6; 8.3$ Head, Douglas $3.2; 3.6; 8.3$ Hedgoes, Bill $3.2; 3.6; 8.3$ Hedges, Bill $3.2; 3.6; 8.3$ Hedges, Bill $3.2; 3.6; 8.3$ Hedges, BarbaraN/AHeffron, Joshua $2.4; 2.13$ Heide, Andra $2.4; 2.13$ Heide, Andra $2.4; 2.13$ Heinde, Janet $2.4; 2.13$ Heinde, Janet $2.4; 2.13$ Heinde, Kathleen $2.4; 2.13$ Heinde, Rick $3.2; 3.6; 8.3$ Heinde, Rick $3.2; 3.6; 8.3$ Heinde, Rick $2.4; 2.13$ Heinde, Rick $2.4; 2.13$ Heinde, Rick $3.2; 3.6; 8.3$ Helmick, Rick $3.2; 3.6; 8.3$ Helmick, Rick $3.2; 3.6; 8.3$ Helmick, Rick $3.2; 3.6; 8.3$ Heinde, Rick $3.2; 3.6; 8.3$	Haugen, David	2.4; 2.13
Hawkins, Phillip $3.2; 3.6$ Haycock, Robert $2.4; 2.13$ Haycock, John $3.6; 8.3$ Hayes, Christine 2.4 Hayes, Tim N/A Hayes, Tim N/A Hays, Dennis $2.4; 2.13$ Hays, Dennis $2.4; 2.13$ Hays, Helen Logan $2.4; 2.13$ Hazen, Alona $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Douglas $3.2; 3.6; 8.3$ Heap, Francesca 2.4 Heathman, Susan $2.4; 2.13$ Hedgeock, Charles $2.4; 2.13$ Hedge, Sill $3.2; 3.6; 8.3$ Hedger, JoAnnN/AHeider, JoAnnN/AHeider, JohanN/AHeide, Andra $2.4; 2.13$ Heinel, Janet $2.4; 2.13$ Heinel, Janet $2.4; 2.13$ Heinel, Janet $2.4; 2.13$ Heinel, Janet $2.4; 2.13$ Heinel, Joavn $2.4; 2.13$ Heinel, Joavn $2.4; 2.13$ Heinel, Joavn $2.4; 2.13$ Heinel, Rick $3.2; 3.6; 8.3$ Helmerk, Rick $3.2; 3.6; 8.3$ Helmerk, Rick $3.2; 3.6; 8.3$ Hendry, Dawn 2.4 Hengesbaugh, Matt $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Hermandez, Patricia $2.4; 2.13$ Hermandez, Patricia $2.4; 2.13$ Hermandez, Patricia $2.4; 2.13$ Hermandez, Patricia $2.4; 2.13$ <t< td=""><td>Haverfield, Heather</td><td>2.4; 2.13</td></t<>	Haverfield, Heather	2.4; 2.13
Haycock, Robert $24; 2.13$ Haycock, John $36; 8.3$ Hayes, Christine 24 Hayes, Tim NA Haymans, Deanna $24; 2.13$ Hays, Bonnis $24; 2.13$ Hays, Helen Logan 24 Hayworth, Steven $24; 2.13$ Haaca, Alona $24; 2.13$ Head, Kris $24; 2.13$ Head, Douglas $32; 36; 8.3$ Heap, Francesca 24 Headinman, Susan $24; 2.13$ Hedgeock, Charles $24; 2.13$ Hedges, Bill $32; 36; 8.3$ Hedges, Bill $32; 3, 6; 8.3$ Hedges, Bill $32; 3, 6; 8.3$ Hedges, Andra $24; 2.13$ Heidges, JoAnn NA Heffron, Joshua $24; 2.13$ Heide, Andra $24; 2.13$ Heing, Janet $24; 2.13$ Heing, Kathleen $24; 2.13$ Heing, Kathleen $24; 2.13$ Heing, Kathleen $24; 2.13$ Heing, Rick $32; 3, 6; 8.3$ Helmick, Rick $32; 3, 6; 8.3$ Helmerk,	Hawkins, Phillip	3.2; 3.6
Haydock, John $3.6; 8.3$ Hayes, Christine 2.4 Hayes, Christine 2.4 Hayes, Christine 2.4 Haymans, Deanna $2.4; 2.13$ Hays, Belte Logan 2.4 Hayworth, Steven $2.4; 2.13$ Hazen, Alona $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Douglas $3.2; 3.6; 8.3$ Heap, Francesca 2.4 Heathman, Susan $2.4; 2.13$ Hede, David $3.2; 3.6; 8.3$ Hedgeck, Charles $2.4; 2.13$ Hedger, JoAnnN/AHefgeock, Charles $2.4; 2.13$ Hedges, Bill $3.2; 3.6; 8.3$ Hedges, BarbaraN/AHeideger, JoAnnN/AHeideger, JoAnnN/AHeideger, JoAnnN/AHeideger, JoAnnN/AHeideger, Janes $2.4; 2.13$ Heinel, Janet $2.4; 2.13$ Heinel, Andra $2.4; 2.13$ Heinel, State $2.4; 2.13$ Heinel, Janet $2.4; 2.13$ Heinel, Janet $2.4; 2.13$ Heinel, Guenter $2.4; 2.13$ Helmick, Rick $3.2; 3.6; 8.3$ Helmick, Rick $3.2; 3.6; 8.3$ Helminel, David $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henrandez, Patricia $2.4; 2.13$ Hernandez, Patricia $2.4; 2.13$ Hernandez, Patricia $2.4; 2.13$ Herran, Lois $2.4; 2.13$ Herran, DorotheaN/A<	Haycock, Robert	2.4; 2.13
Hayes, Christine2.4Hayes, TimN/AHaymans, Deanna2.4; 2.13Hays, Dennis2.4; 2.13Hays, Helen Logan2.4Haxen, Alona2.4; 2.13Hazen, Alona2.4; 2.13Head, Kris2.4; 2.13Head, Kris2.4; 2.13Head, Kris2.4; 2.13Head, Kris2.4; 2.13Head, Kris2.4; 2.13Head, Kris3.2; 3.6; 8.3Heap, Francesca2.4Hethman, Susan2.4; 2.13Hedges, Bill3.2; 3.6; 8.3Hedges, Bill3.2; 3.6; 8.3Hedges, Ball3.2; 3.6; 8.3Hedges, BarbaraN/AHeider, Joshua2.4; 2.13Heine, Janet2.4; 2.13Heine, Janet2.4; 2.13Heine, Janet2.4; 2.13Heine, Janet2.4; 2.13Heine, Kathleen2.4; 2.13Heiner, Kathleen2.4; 2.13Helmer, Kathleen2.4; 2.13Helmer, Kathleen2.4; 2.13Helmer, Kathleen2.4; 2.13Helmick, Rick3.2; 3.6; 8.3Helmer, Willam2.4; 2.13Henninger, Maryann2.4; 2.13Henry, David3.2; 3.6; 8.3Henry, David3.2; 3.6; 8.3Henry, David2.4; 2.13Herran, DorotheaN/AHerrandez, Patricia2.4; 2.13Herrandez, Patricia2.4; 2.13Herrand, ChrisN/AHerrand, ChrisN/AHerran, ChrisN/AHerrandez, Rehard3.	Haydock, John	3.6; 8.3
Hayes, TimN/AHaymans, Deanna 2.4 ; 2.13 Hays, Dennis 2.4 ; 2.13 Hays, Helen Logan 2.4 Hayworth, Steven 2.4 ; 2.13 Hazen, Alona 2.4 ; 2.13 Head, Kris 2.4 ; 2.13 Head, Kris 2.4 ; 2.13 Head, Nouglas 3.2 ; 3.6 ; 8.3 Heap, Francesca 2.4 Hedgeock, Charles 2.4 ; 2.13 Hedgey, Bill 3.2 ; 3.6 ; 8.3 Heiger, JoAnnN/AHeifton, Joshua 2.4 ; 2.13 Heidek, Andra 2.4 ; 2.13 Heine, Janet 2.4 ; 2.13 Heine, Kathleen 2.4 ; 2.13 Heiner, Kathleen 2.4 ; 2.13 Helmer, Kathleen 2.4 ; 2.13 Helmer, Kathleen 2.4 ; 2.13 Henrey, Dawn 2.4 Hengesbaugh, Matt 2.4 ; 2.13 Henry, David 3.2 ; 3.6 ; 8.3 Henry, David 3.2 ; 3.6 ; 8.3 Henry, Chara N/A Herrandez, Patricia 2.4 ; 2.13 Herrera, Lois 2.4 ; 2.13 <	Hayes, Christine	2.4
Haymans, Deanna $2.4; 2.13$ Hays, Dennis $2.4; 2.13$ Hays, Helen Logan 2.4 Hayworth, Steven $2.4; 2.13$ Hazen, Alona $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Douglas $3.2; 3.6; 8.3$ Heap, Francesca 2.4 Heathman, Susan $2.4; 2.13$ Hedges, K. Charles $2.4; 2.13$ Hedges, BarbaraN/AHeffron, Joshua $2.4; 2.13$ Hegdus, BarbaraN/AHeine, Janet $2.4; 2.13$ Heine, Janet $2.4; 2.13$ Heine, Guenter $2.4; 2.13$ Heine, Shafder $2.4; 2.13$ Heine, Shafder $2.4; 2.13$ Heine, Kick $3.2; 3.6; 8.3$ Hehrer, Kathleen $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Hendry, Dawn 2.4 Hendry, Dawn 2.4 Henrer, Maryann $2.4; 2.13$ Henrer, Maryann $2.4; 2.13$ Henry, Amy 2.4 Henry, Amy 2.4 Henrandez, Patricia $2.4; 2.13$ Herrera, Desiree $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Desiree $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera	Hayes, Tim	N/A
Hays, Dennis $2.4; 2.13$ Hays, Helen Logan 2.4 Hayworth, Steven $2.4; 2.13$ Hazen, Alona $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Jouglas $3.2; 3.6; 8.3$ Heap, Francesca 2.4 Heathman, Susan $2.4; 2.13$ Hedgock, Charles $2.4; 2.13$ Hedgock, Charles $2.4; 2.13$ Hedgock, Charles $2.4; 2.13$ Hedges, Bill $3.2; 3.6; 8.3$ Hediger, JoAnnN/AHeffron, Joshua $2.4; 2.13$ Heide, Andra $2.4; 2.13$ Heide, Andra $2.4; 2.13$ Heilmann, James $2.4; 2.13$ Heiny, Bridgett $2.4; 2.13$ Heiny, Sirdgett $2.4; 2.13$ Helmick, Rick $3.2; 3.6; 8.3$ Helstien, Sherie 2.4 Henrick, Rick $3.2; 3.6; 8.3$ Helstien, Sherie 2.4 Henringer, Maryann $2.4; 2.13$ Henringer, Maryann $2.4; 2.13$ Henringer, Maryann $2.4; 2.13$ Henringer, Maryann $2.4; 2.13$ Henringer, Charles 2.4 Henringer, Maryann $2.4; 2.13$ Herrera, Dosirbea N/A Herrera, Desirbe $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera	Haymans, Deanna	2.4; 2.13
Hays, Helen Logan 2.4 Hayworth, Steven $2.4; 2.13$ Hazen, Alona $2.4; 2.13$ Head, Kris $2.4; 2.13$ Heald, Douglas $3.2; 3.6; 8.3$ Heap, Francesca 2.4 Heathman, Susan $2.4; 2.13$ Hedgook, Charles $2.4; 2.13$ Hedges, Bill $3.2; 3.6; 8.3$ Hedges, Bill $3.2; 3.6; 8.3$ Hedges, Bill $3.2; 3.6; 8.3$ Hedges, BarbaraN/AHeffron, Joshua $2.4; 2.13$ Heidge, Andra $2.4; 2.13$ Heine, Janet $2.4; 2.13$ Heine, Janet $2.4; 2.13$ Heine, Kathleen $2.4; 2.13$ Heinek, Sick $3.2; 3.6; 8.3$ Helmick, Rick $3.2; 3.6; 8.3$ Helmick, Rick $3.2; 3.6; 8.3$ Helmick, Rick $3.2; 3.6; 8.3$ Hengesbaugh, Matt $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henran, DorotheaN/AHerran, DorotheaN/AHerran, Desiree $2.4; 2.13$ Herran, Desiree $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herreran, Chris N/A Herran, Chris N/A Herrera, Chris N/A Herran, Chris N/A Herreran, Chris N/A	Hays, Dennis	2.4; 2.13
Hayworth, Steven $2.4; 2.13$ Hazen, Alona $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Kris $2.4; 2.13$ Head, Douglas $32; 3.6; 8.3$ Heap, Francesca 2.4 Heathman, Susan $2.4; 2.13$ Hedgock, Charles $2.4; 2.13$ Hedgock, Charles $2.4; 2.13$ Hedges, Bill $32; 3.6; 8.3$ Hediger, JoAnnN/AHeffron, Joshua $2.4; 2.13$ Hegedus, BarbaraN/AHeide, Andra $2.4; 2.13$ Heinman, James $2.4; 2.13$ Heinly, Bridgett $2.4; 2.13$ Heinly, Sridgett $2.4; 2.13$ Heinly, Rick $32; 3.6; 8.3$ Helmer, Kathleen $2.4; 2.13$ Heinre, Sherie 2.4 Henninger, Maryann $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henran, DorotheaN/AHerran, DorotheaN/AHerran, DosineaN/AHerran, Desiree $2.4; 2.13$ Herran, Lois $2.4; 2.13$ Herran, Chris N/A Herran, Chris N/A	Hays, Helen Logan	2.4
Hazen, Alona $2.4; 2.13$ Head, Kris $2.4; 2.13$ Heald, Douglas $3.2; 3.6; 8.3$ Heap, Francesca 2.4 Heathman, Susan $2.4; 2.13$ Hedex, David $3.2; 3.6$ Hedgeock, Charles $2.4; 2.13$ Hedge, Sill $3.2; 3.6; 8.3$ Hediger, JoAnnN/AHeffron, Joshua $2.4; 2.13$ Hegedus, BarbaraN/AHeide, Andra $2.4; 2.13$ Heide, Andra $2.4; 2.13$ Heinel, Janet $2.4; 2.13$ Heinel, Singett $2.4; 2.13$ Heinik, Kick $3.2; 3.6; 8.3$ Helmick, Rick $3.2; 3.6; 8.3$ Helmick, Rick $3.2; 3.6; 8.3$ Hengesbaugh, Matt $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henran, DorotheaN/AHernza, Desiree $2.4; 2.13$ Herrara, Chris N/A Herrara, Chras N/A <td>Hayworth, Steven</td> <td>2.4; 2.13</td>	Hayworth, Steven	2.4; 2.13
Head, Kris $2.4; 2.13$ Head, Douglas $3.2; 3.6; 8.3$ Heap, Francesca 2.4 Heathman, Susan $4.4; 2.13$ Heck, David $3.2; 3.6$ Hedgeock, Charles $2.4; 2.13$ Hedges, Bill $3.2; 3.6; 8.3$ Hediger, JoAnnN/AHeffron, Joshua $2.4; 2.13$ Hegedus, BarbaraN/AHeinen, Janes $2.4; 2.13$ Heine, Andra $2.4; 2.13$ Heine, Janet $2.4; 2.13$ Heine, Kathleen $2.4; 2.13$ Heinik, Rick $3.2; 3.6; 8.3$ Helmick, Rick $3.2; 3.6; 8.3$ Helmick, Rick $3.2; 3.6; 8.3$ Helming, Matt $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henran, DorotheaN/AHerran, DorotheaN/AHerran, DorotheaN/AHerran, Chris $2.4; 2.13$ Herran, Chris N/A	Hazen, Alona	2.4; 2.13
Heald, Douglas $3.2; 3.6; 8.3$ Heap, Francesca 2.4 Heathman, Susan $2.4; 2.13$ Heck, David $3.2; 3.6$ Hedgock, Charles $2.4; 2.13$ Hedges, Bill $3.2; 3.6; 8.3$ Hediger, JoAnnN/AHeffron, Joshua $2.4; 2.13$ Hegdus, BarbaraN/AHeide, Andra $2.4; 2.13$ Heide, Andra $2.4; 2.13$ Heinmann, James $2.4; 2.13$ Heinn, James $2.4; 2.13$ Heinr, Kathleen $2.4; 2.13$ Heinr, Kathleen $2.4; 2.13$ Heinr, Kathleen $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Hendry, Dawn 2.4 Hendry, Dawn 2.4 Henzel, William $2.4; 2.13$ Henry, Amy 2.4 Henzel, William $2.4; 2.13$ Herran, DorotheaN/AHerrera, Desiree $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Desiree $2.4; 2.13$ Herrera, Desiree $2.4; 2.13$ Herrera, Desiree $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Kichael $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Kichael $2.4; 2.13$ Herrera, Kichael $2.4; 2.13$ Herrera, Kichael 3.6	Head, Kris	2.4; 2.13
Heap, Francesca 2.4 Heathman, Susan 2.4 ; 2.13 Heck, David 3.2 ; 3.6 Hedgeock, Charles 2.4 ; 2.13 Hedges, Bill 3.2 ; 3.6 ; 8.3 Hediger, JoAnnN/AHeffron, Joshua 2.4 ; 2.13 Hegedus, BarbaraN/AHeide, Andra 2.4 ; 2.13 Heine, Janet 2.4 ; 2.13 Heine, Janet 2.4 ; 2.13 Heine, Janet 2.4 ; 2.13 Heine, Kathleen 2.4 ; 2.13 Heinic, Kick 3.2 ; 3.6 ; 8.3 Helstien, Sherie 2.4 Hendry, Dawn 2.4 Hendry, Dawn 2.4 Henry, David 3.2 ; 3.6 ; 8.3 Henry, May 2.4 Henry, Chris 2.4 ; 2.13 Herrra, DorotheaN/AHerrra, Lois 2.4 ; 2.13 Herrra, Lois 2.4 ; 2.13 Herrra, Lois 2.4 ; 2.13 Herrrat, Lois 2.4 ; 2.13 Herrrat, Lois 2.4 ; 2.13 Herrato, DorotheaN/AHerrato, Districe 2.4 ; 2.13 Herrato, Districe	Heald, Douglas	3.2; 3.6; 8.3
Heathman, Susan $2.4; 2.13$ Heck, David $3.2; 3.6$ Hedgeock, Charles $2.4; 2.13$ Hedges, Bill $3.2; 3.6; 8.3$ Hedger, JoAnnN/AHeffron, Joshua $2.4; 2.13$ Hegedus, BarbaraN/AHeinde, Andra $2.4; 2.13$ Heinde, Janet $2.4; 2.13$ Heinle, Janet $2.4; 2.13$ Heinle, Janet $2.4; 2.13$ Heinle, Janet $2.4; 2.13$ Heinle, Kathleen $2.4; 2.13$ Helmick, Rick $3.2; 3.6; 8.3$ Helstien, Sherie 2.4 Hendry, Dawn 2.4 Henngesbaugh, Matt $2.4; 2.13$ Henry, Amy 2.4 Henzel, William $2.4; 2.13$ Herran, Dorothea N/A Herran, Dorothea N/A Herrara, Desiree $2.4; 2.13$ Herrara, Lois $2.4; 2.13$ Herrara, Lois $2.4; 2.13$ Herrara, Disiree $2.4; 2.13$ Herrara, Lois $2.4; 2.13$ Herrara, Chris N/A Herrara, Chris N/A Herrara, Chris N/A Herrara, Chris N/A	Heap, Francesca	2.4
Heck, David 3.2 ; 3.6 Hedgcock, Charles 2.4 ; 2.13 Hedges, Bill 3.2 ; 3.6 ; 8.3 Hediger, JoAnnN/AHeffron, Joshua 2.4 ; 2.13 Hegdus, BarbaraN/AHeide, Andra 2.4 ; 2.13 Heilmann, James 2.4 ; 2.13 Heinle, Janet 2.4 ; 2.13 Heinre, Guenter 2.4 ; 2.13 Heimer, Kathleen 2.4 ; 2.13 Helmer, Kick 3.2 ; 3.6 ; 8.3 Helmer, Kick 3.2 ; 3.6 ; 8.3 Helmery, Dawn 2.4 Henninger, Maryann 2.4 ; 2.13 Henry, Amy 2.4 Henry, Amy 2.4 Henry, Amy 2.4 Herran, DorotheaN/AHerrera, Lois 2.4 ; 2.13 Herrera, Lois 2.4 ; 2.13 Herrera, Lois 2.4 ; 2.13 Herrera, Lois 2.4 ; 2.13 Herreran, Chris N/A Herreran, Chris N/A	Heathman, Susan	2.4; 2.13
Hedgeock, Charles 2.4 ; 2.13 Hedges, Bill 3.2 ; 3.6 ; 8.3 Hediger, JoAnnN/AHeffron, Joshua 2.4 ; 2.13 Hegedus, BarbaraN/AHeide, Andra 2.4 ; 2.13 Heinann, James 2.4 ; 2.13 Heinle, Janet 2.4 ; 2.13 Heinle, Janet 2.4 ; 2.13 Heinle, Janet 2.4 ; 2.13 Heinle, Kathleen 2.4 ; 2.13 Helmer, Kathleen 2.4 ; 2.13 Helmick, Rick 3.2 ; 3.6 ; 8.3 Helmick, Rick 3.2 ; 3.6 ; 8.3 Henninger, Maryann 2.4 ; 2.13 Henny, David 3.2 ; 3.6 ; 8.3 Henry, David 3.2 ; 3.6 ; 8.3 Henry, Amy 2.4 Hernz, Amy 2.4 Herran, DorotheaN/AHerreran, Dosinee 2.4 ; 2.13 Herreran, Lois 2.4 ; 2.13 Herreran, Chris 2.4 ; 2.13 Herrick, Michael 2.4 ; 2.13 Herreran, Chris N/A	Heck, David	3.2; 3.6
Hedges, Bill $3.2; 3.6; 8.3$ Hediger, JoAnnN/AHeffron, Joshua $2.4; 2.13$ Hegedus, BarbaraN/AHeide, Andra $2.4; 2.13$ Heilmann, James $2.4; 2.13$ Heinle, Janet $2.4; 2.13$ Heinly, Bridgett $2.4; 2.13$ Heinr, Guenter $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Helmick, Rick $3.2; 3.6; 8.3$ Helsten, Sherie 2.4 Hendry, Dawn 2.4 Henninger, Maryann $2.4; 2.13$ Henry, Amy 2.4 Henry, Amy 2.4 Herran, DorotheaN/AHerrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Chris N/A Herrz, Richard 3.6	Hedgcock, Charles	2.4; 2.13
Hediger, JoAnnN/AHeffron, Joshua $2.4; 2.13$ Hegedus, BarbaraN/AHeide, Andra $2.4; 2.13$ Heilmann, James $2.4; 2.13$ Heinle, Janet $2.4; 2.13$ Heinly, Bridgett $2.4; 2.13$ Heinz, Guenter $2.4; 2.13$ Helmick, Rick $3.2; 3.6; 8.3$ Helstien, Sherie 2.4 Hendry, Dawn 2.4 Hengesbaugh, Matt $2.4; 2.13$ Henry, David $3.2; 3.6; 8.3$ Henzel, William $2.4; 2.13$ Herman, DorotheaN/AHerrandez, Patricia $2.4; 2.13$ Herrandez, Patricia $2.4; 2.13$ Herrand, Chris N/A Herrand, Chris N/A	Hedges, Bill	3.2; 3.6; 8.3
Heffon, Joshua $2.4; 2.13$ Hegedus, BarbaraN/AHeide, Andra $2.4; 2.13$ Heilmann, James $2.4; 2.13$ Heinle, Janet $2.4; 2.13$ Heinle, Janet $2.4; 2.13$ Heinly, Bridgett $2.4; 2.13$ Heinz, Guenter $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Helmick, Rick $3.2; 3.6; 8.3$ Helstien, Sherie 2.4 Hendry, Dawn 2.4 Hengesbaugh, Matt $2.4; 2.13$ Henry, David $3.2; 3.6; 8.3$ Henzel, William $2.4; 2.13$ Herman, DorotheaN/AHerrandez, Patricia $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrick, Michael $2.4; 2.13$ Herrick, Michael $2.4; 2.13$ Herrick, Kickael 3.6	Hediger, JoAnn	N/A
Hegedus, BarbaraN/AHeide, Andra $2.4; 2.13$ Heilmann, James $2.4; 2.13$ Heinle, Janet $2.4; 2.13$ Heinly, Bridgett $2.4; 2.13$ Heinly, Bridgett $2.4; 2.13$ Heinz, Guenter $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Helmick, Rick $3.2; 3.6; 8.3$ Helstien, Sherie 2.4 Hendry, Dawn 2.4 Hengesbaugh, Matt $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henry, David $3.2; 3.6; 8.3$ Henry, Amy 2.4 Henzel, William $2.4; 2.13$ Herman, DorotheaN/AHermandez, Patricia $2.4; 2.13$ Herrera, Desiree $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrick, Michael $2.4; 2.13$ Herrick, Michael $2.4; 2.13$ Herrick, Richard 3.6	Heffron, Joshua	2.4; 2.13
Heide, Andra $2.4; 2.13$ Heilmann, James $2.4; 2.13$ Heinle, Janet $2.4; 2.13$ Heinly, Bridgett $2.4; 2.13$ Heinly, Bridgett $2.4; 2.13$ Heinz, Guenter $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Helmick, Rick $3.2; 3.6; 8.3$ Helstien, Sherie 2.4 Hengesbaugh, Matt $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henry, David $3.2; 3.6; 8.3$ Henzel, William $2.4; 2.13$ Herman, DorotheaN/AHerrera, Desiree $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Chris N/A Herrandez, Patricia $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois 3.6	Hegedus, Barbara	N/A
Heilmann, James $2.4; 2.13$ Heinle, Janet $2.4; 2.13$ Heinly, Bridgett $2.4; 2.13$ Heinz, Guenter $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Helmick, Rick $3.2; 3.6; 8.3$ Helstien, Sherie 2.4 Hendry, Dawn 2.4 Henninger, Maryann $2.4; 2.13$ Henry, David $3.2; 3.6; 8.3$ Henry, David $3.2; 3.6; 8.3$ Henry, David $2.4; 2.13$ Hernzel, William $2.4; 2.13$ Herrandez, Patricia $2.4; 2.13$ Herrera, Dorothea N/A Herrera, Lois $2.4; 2.13$ Herrick, Michael $2.4; 2.13$ Herrman, Chris N/A	Heide, Andra	2.4; 2.13
Heinle, Janet 2.4; 2.13 Heinly, Bridgett 2.4; 2.13 Heinz, Guenter 2.4; 2.13 Helmer, Kathleen 2.4; 2.13 Helmick, Rick 3.2; 3.6; 8.3 Helstien, Sherie 2.4 Hendry, Dawn 2.4 Hengesbaugh, Matt 2.4; 2.13 Henninger, Maryann 2.4; 2.13 Henry, David 3.2; 3.6; 8.3 Henry, David 3.2; 3.6; 8.3 Henry, David 3.2; 3.6; 8.3 Henry, Amy 2.4 Henzel, William 2.4; 2.13 Herman, Dorothea N/A Herrera, Desiree 2.4; 2.13 Herrera, Lois 2.4; 2.13 Herrera, Lois 2.4; 2.13 Herrick, Michael 3.6	Heilmann, James	2.4; 2.13
Heinly, Bridgett $2.4; 2.13$ Heinz, Guenter $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Helmick, Rick $3.2; 3.6; 8.3$ Helstien, Sherie 2.4 Hendry, Dawn 2.4 Hengesbaugh, Matt $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henry, David $3.2; 3.6; 8.3$ Henry, David $3.2; 3.6; 8.3$ Henry, Mary 2.4 Henzel, William $2.4; 2.13$ Herman, DorotheaN/AHerrera, Desiree $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrick, Michael $2.4; 2.13$ Herrman, ChrisN/AHerrz, Richard 3.6	Heinle, Janet	2.4; 2.13
Heinz, Guenter $2.4; 2.13$ Helmer, Kathleen $2.4; 2.13$ Helmick, Rick $3.2; 3.6; 8.3$ Helstien, Sherie 2.4 Hendry, Dawn 2.4 Hengesbaugh, Matt $2.4; 2.13$ Henninger, Maryann $2.4; 2.13$ Henry, David $3.2; 3.6; 8.3$ Henry, Amy 2.4 Henzel, William $2.4; 2.13$ Herman, DorotheaN/AHerrera, Desiree $2.4; 2.13$ Herrera, Lois $2.4; 2.13$ Herrick, Michael $2.4; 2.13$ Herrizk, Richard 3.6	Heinly, Bridgett	2.4; 2.13
Helmer, Kathleen2.4; 2.13Helmick, Rick3.2; 3.6; 8.3Helstien, Sherie2.4Hendry, Dawn2.4Hengesbaugh, Matt2.4; 2.13Henninger, Maryann2.4; 2.13Henry , David3.2; 3.6; 8.3Henry , David2.4Henzel, William2.4; 2.13Herman, DorotheaN/AHerrera, Desiree2.4; 2.13Herrera, Lois2.4; 2.13Herrera, Lois2.4; 2.13Herrera, ChrisN/AHerran, ChrisN/AHertz, Richard3.6	Heinz, Guenter	2.4; 2.13
Helmick, Rick3.2; 3.6; 8.3Helstien, Sherie2.4Hendry, Dawn2.4Hengesbaugh, Matt2.4; 2.13Henninger, Maryann2.4; 2.13Henry , David3.2; 3.6; 8.3Henry, Amy2.4Henzel, William2.4; 2.13Herman, DorotheaN/AHerrera, Desiree2.4; 2.13Herrera, Lois2.4; 2.13Herrera, Lois2.4; 2.13Herrera, ChrisN/AHerran, ChrisN/AHertz, Richard3.6	Helmer, Kathleen	2.4; 2.13
Helstien, Sherie2.4Hendry, Dawn2.4Hengesbaugh, Matt2.4; 2.13Henninger, Maryann2.4; 2.13Henry , David3.2; 3.6; 8.3Henry, Amy2.4Henzel, William2.4; 2.13Herman, DorotheaN/AHerrera, Desiree2.4; 2.13Herrera, Lois2.4; 2.13Herrick, Michael2.4; 2.13Herrman, ChrisN/AHerrman, Chris3.6	Helmick, Rick	3.2; 3.6; 8.3
Hendry, Dawn2.4Hengesbaugh, Matt2.4; 2.13Henninger, Maryann2.4; 2.13Henry, David3.2; 3.6; 8.3Henry, Amy2.4Henzel, William2.4; 2.13Herman, DorotheaN/AHerrera, Desiree2.4; 2.13Herrera, Lois2.4; 2.13Herrick, Michael2.4; 2.13Herrman, ChrisN/AHertz, Richard3.6	Helstien, Sherie	2.4
Hengesbaugh, Matt2.4; 2.13Henninger, Maryann2.4; 2.13Henry, David3.2; 3.6; 8.3Henry, Amy2.4Henzel, William2.4; 2.13Herman, DorotheaN/AHernandez, Patricia2.4; 2.13Herrera, Desiree2.4; 2.13Herrera, Lois2.4; 2.13Herrick, Michael2.4; 2.13Herrman, ChrisN/AHertz, Richard3.6	Hendry, Dawn	2.4
Henninger, Maryann2.4; 2.13Henry, David3.2; 3.6; 8.3Henry, Amy2.4Henzel, William2.4; 2.13Herman, DorotheaN/AHernandez, Patricia2.4; 2.13Herrera, Desiree2.4; 2.13Herrera, Lois2.4; 2.13Herrick, Michael2.4; 2.13Herrman, ChrisN/AHertz, Richard3.6	Hengesbaugh, Matt	2.4; 2.13
Henry , David3.2; 3.6; 8.3Henry, Amy2.4Henzel, William2.4; 2.13Herman, DorotheaN/AHernandez, Patricia2.4; 2.13Herrera, Desiree2.4; 2.13Herrera, Lois2.4; 2.13Herrick, Michael2.4; 2.13Herrman, ChrisN/AHertz, Richard3.6	Henninger, Maryann	2.4; 2.13
Henry, Amy2.4Henzel, William2.4; 2.13Herman, DorotheaN/AHernandez, Patricia2.4; 2.13Herrera, Desiree2.4; 2.13Herrera, Lois2.4; 2.13Herrick, Michael2.4; 2.13Herrman, ChrisN/AHertz, Richard3.6	Henry, David	3.2; 3.6; 8.3
Henzel, William2.4; 2.13Herman, DorotheaN/AHernandez, Patricia2.4; 2.13Herrera, Desiree2.4; 2.13Herrera, Lois2.4; 2.13Herrick, Michael2.4; 2.13Herrman, ChrisN/AHertz, Richard3.6	Henry, Amy	2.4
Herman, DorotheaN/AHernandez, Patricia2.4; 2.13Herrera, Desiree2.4; 2.13Herrera, Lois2.4; 2.13Herrick, Michael2.4; 2.13Herrman, ChrisN/AHertz, Richard3.6	Henzel, William	2.4; 2.13
Hernandez, Patricia2.4; 2.13Herrera, Desiree2.4; 2.13Herrera, Lois2.4; 2.13Herrick, Michael2.4; 2.13Herrman, ChrisN/AHertz, Richard3.6	Herman, Dorothea	N/A
Herrera, Desiree2.4; 2.13Herrera, Lois2.4; 2.13Herrick, Michael2.4; 2.13Herrman, ChrisN/AHertz, Richard3.6	Hernandez, Patricia	2.4; 2.13
Herrera, Lois2.4; 2.13Herrick, Michael2.4; 2.13Herrman, ChrisN/AHertz, Richard3.6	Herrera, Desiree	2.4; 2.13
Herrick, Michael2.4; 2.13Herrman, ChrisN/AHertz, Richard3.6	Herrera, Lois	2.4; 2.13
Herrman, ChrisN/AHertz, Richard3.6	Herrick, Michael	2.4; 2.13
Hertz, Richard 3.6	Herrman, Chris	N/A
	Hertz, Richard	3.6

Herzog, Robert	N/A
Hessler, Charles	2.4; 2.13
Hessler, Charles	2.4; 2.13
Hewelt, Karen	N/A
Heyde, Christiane	N/A
Heyneman, Amy	2.4; 2.13
Heywood, Austin	N/A
Hibbs, Howard	3.2; 3.6
Hicks, Robert	2.4; 2.13
Hiestand, Nancy	2.4; 2.13
Hill, Gerry	3.2; 3.6; 8.3
Hill, Marilyn	2.4; 2.13
Hill, Mark	N/A
Hillard, Dale	2.4: 2.13
Hilliard, Donald	2.4: 2.13
Hills Richard	2 14
Hilton Bill	2 4 2 13
Hinderberger Sacha	2.4
Hines Marianne	N/A
Hinkle Connie	2 4. 2 13
Hinkley David	2.4
Hinton Terry	N/A
Hirschenhein Randy	2 4: 2 13
Hirschman Mark	2.1, 2.13
Hiscox Tom	3 2: 3 6: 8 3
Hissom Terry	N/A
Hlat Mike	2 4: 2 13
Hoagland Linda	2.4, 2.13
Hobbs Joan	N/A
Hobbs Pat	2 4: 2 13
Hodie Jake	2.4, 2.13
Hoenigman Vince	2.4, 2.13
Hoev Roseanne	2.4, 2.15
Hofer Ricardo	2. 1 2 <i>A</i> · 2 13
Hoffman Chrissy	2.4, 2.13
Hoffman Jane	2.4, 2.13
Hoffman Janice	2.4, 2.15 N/A
Hoffman Lincoln	$2 \wedge 2 13$
Hoffman Marc	2.4, 2.13
Hofheing Daul	2.4, 2.13
Home Mary	2.4, 2.15
Hogan Bandalnh	2.4
Holbart, Cumthia	2.4
Holbo, Chadwick	2.4, 2.13 2 2: 2 6: 9 2
Holbrook Lon	5.2, 5.0, 6.5 2 4: 2 12
Holbrook, JOH	2.4, 2.13
Holden Com	3.2, 3.0 2.4, 2.12
Holder, Gary	2.4, 2.13
Holland, Brett	2.4; 2.13
Holland, J.	2.4; 2.13
Hollington, Jason	N/A

Hollinrake, Mark	2.4; 2.13
Holman, Clarisse	N/A
Holmes, David	2.4; 2.13
Holt, Arve	2.4; 2.13
Holthouse, Dave	2.4; 2.13
Holtzman, Dorothy	2.4
Holtzman, Julie	2.4
Honanie, Herman (The Hopi Tribe)	2.9; 4.1.1; 8.1; 10.1; 10.7, 10.8
Honigsblum, Alexander	2.4; 2.13
Honore, Stephanie	2.4; 2.13
Hope, Phillip	2.4; 2.13
Hopkins, Larry	2.4
Hoppe, Lindy	N/A
Horn, Stephanie	2.4; 2.13
Horowitz, Laura	2.4; 2.13
Horton, Deanna	N/A
Horton, Derek	N/A
Horzepa, Judith	N/A
Hosea, David	2.4
Hottenstein, Tara	2.4; 2.13
Hougaard, David	3.2; 3.6; 8.3
Houston, Meghan	2.4; 2.13
Hovorka, Ah	N/A
Howard, Bryan	N/A
Howard, Lex	2.8
Howard, Nancy	2.4; 2.13
Howell, Dorothy (Payson Fly Casters Club)	2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2; 12.4
Howland, Cynthia	2.4; 2.13
Howse, Peter	2.1; 2.2; 2.8; 2.14; 13.1
Hubbard, James	2.4; 2.13
Hubbell, Linda	2.4; 2.13
Hubbs, Dawn (Hualapai Nation)	4.2; 4.3; 8.2; 10.4; 10.5; 10.6; 10.8; 12.5
Hudgins, Jerry	2.4; 2.13
Huerta, John	2.4
Huffman, Shelia	N/A
Hughes, Barbara	N/A
Hughes, Laurel	2.4; 2.13
Hughes, Lisa	2.4; 2.13
Huie, Jonathan	N/A
Hulsey, Tamara	2.4; 2.13
Hunnicutt, Joan	N/A
Hunt, Timothy	3.2; 3.6; 8.3
Hunter, Stanton	N/A
Huntoon, Kristin	2.4; 2.13
Hurabiell, John	3.2; 3.6; 8.3
Hurley, Pat	3.2; 3.6; 8.3
Hurliman, Heidi	2.4
Hus, Richard	3.2; 3.6; 8.3
Hutchinson, Bryce	N/A
Hutchinson, Terrance	N/A

Huycke Mccall, Ruth	N/A
Hyams, Mark	2.8
Hyde, David	2.8; 2.14; 13.1
Ibenthal, Brian	3.2; 3.6; 8.3
Ihle, John	2.4; 2.13
Iltis, Michael	2.4; 2.13
Imar, Delaney	2.4; 2.13
Imlay, Marc and Alice	2.4; 2.13
ImMasche, Sonia	2.4
Ingemi, Lynn	N/A
Ingham, Lula Kay (Katie)	2.4
Ingraham, Carolyn	2.4; 2.13
Ionina, Kate	2.4; 2.13
Iovino, Teresa	2.4; 2.13
Iseri, Martin	N/A
Iverson. Steve	2.4: 3.13
Jackman, Rob	2.5: 2.17: 3.2: 3.3: 3.4: 3.6: 8.3: 12.2
Jackson Harold	N/A
Jackson James	2 4 · 2 13
Jackson, Sue	2 4 2 13
Jackson, Warren	2 4 2 13
Jacob I vnn	N/A
Jacobel Richard	2 4: 2 13
Jacobi Allen	N/Δ
Jacobson Susan	$2 1 \cdot 2 2 \cdot 2 1 4$
Jamal Kate	2.1, 2.2, 2.14 2 A · 2 13
Jamas, Laclia (Colorado Piver Energy Distributors	2.7, 2.15 1 1 · 1 2 · 1 2 · 1 4 · 1 5 · 2 1 · 2 2 · 2 6 · 2 7 · 2 10 · 2 1 4 ·
Association)	1.1, 1.2, 1.3, 1.4, 1.3, 2.1, 2.2, 2.0, 2.7, 2.10, 2.14, 2.10, 2.20, 2.2, 2.8, 4.1, 1, 4.1, 2, 4.1, 2, 5, 2, 6, 1, 2, 6, 2, 1, 2, 2, 2, 2, 2, 4, 1, 1, 4, 1, 2, 4, 1, 2, 5, 2, 6, 1, 2, 6, 2, 1, 2, 2, 2, 2, 2, 2, 2, 4, 1, 1, 4, 1, 2, 4, 1, 2, 5, 2, 6, 1, 2, 6, 2, 1, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2, 2,
Association)	2.19, 2.20, 5.5, 5.6, 4.1.1, 4.1.2, 4.1.5, 5.2, 0.1.2, 0.2.1,
	$6 4 11 \cdot 6 4 14 \cdot 6 5 1 \cdot 6 6 1 \cdot 6 7 \cdot 7 2 \cdot 7 3 2 \cdot 8 8 \cdot 8 0 \cdot 6 4 \cdot 7$
	0.4.11, 0.4.14, 0.5.1, 0.0.1, 0.7, 7.2, 7.5.2, 0.0, 0.9, 0.4, 10.2, 12.0
James Nadina	9.4, 10.5, 12.9
Janics, Naune	2.4 2.8·2.12·2.14·4.12·7.2.1·11.2
Jansen, Gaula	2.0, 2.15, 2.14, 4.1.2, 7.5.1, 11.5
Jardina Guy	2.4, 2.15
Jaruine, Ouy	5.2, 5.0, 6.5
Janvis, Michael	N/A
Jasper, Alali Jastromh Virginia	2.4, 2.15 N/A
Jastionio, viiginia	IN/A N/A
Jeavons, John	N/A
Jennes, Stephen	5.2; 5.0; 8.5 1 2: 1 2: 5 1: 6 12: 6 22: 9 4
Jeka, Lynn (western Area Power Administration)	1.2; 1.3; 3.1; 0.13; 0.23; 8.4
Jennings, Joanna	N/A
Jennings, Scott	2.4; 2.13
Jennings, Sid	N/A
Jensen, Margaret	2.4
Jensen, Sterling	3.2; 3.0; 8.3
Jobe, Laura	2.4
Jobling, Catherine	2.4
Johannsen, Mary	2.4; 2.13
Johnsen, David	2.4; 2.13

Johnsen, Arnold NA Johnson, Arnold NA Johnson, Brett 32, 3.6 Johnson, Druce 2.4 Johnson, Dressa Rae 2.4, 2.13 Johnson, Deborah N/A Johnson, G. Chessa Rae 2.4, 2.13 Johnson, Douglas 2.4 Johnson, Douglas 2.4 Johnson, Nouglas 2.4 Johnson, Karen N/A Johnson, Karen N/A Johnson, Karen 2.4; 2.13 Johnson, Viginia N/A Jones, Andrea 2.4; 2.13 Jones, Andrea 2.4; 2.13 Jones, Andrea 2.4; 2.13 Jones, S. Lauel 2.4 Jones, Tina N/A Jonkosky, Charles N/A Jonkosky, Charles N/A Jordan, S. 2.4 Jordan, S		
Johnson, Arnold N/A Johnson, Brett $32, 3.6$ Johnson, Dessa Rae $24, 2.13$ Johnson, Deborah N/A Johnson, Douglas 24 Johnson, G.G. $24, 2.13$ Johnson, G.G. $24, 2.13$ Johnson, Karen N/A Johnson, Karen $24, 2.13$ Johnson, Kich $32, 3.6, 8.3$ Johnson, Nich $24, 2.13$ Johnson, Nich $24, 2.13$ Johnson, Nich $32, 3.6, 8.3$ Johnson, Vicki $24, 2.13$ Johnson, Vicki $24, 2.13$ Johnson, Viginia N/A Johnson, Viginia N/A Johnson, Viginia N/A Johnson, Viginia N/A Johnson, Kich $32, 3.6, 8.3$ Johnson, Viseki $24, 2.13$ Johnson, Viski $24, 2.13$ Johnson, Vishi $24, 2.13$ Justice, Cheryl $24, 2.13$ Justice, Cheryl $24, 2.13$ Justice, Cheryl $24, 2.13$ Justice, Cheryl $22, 36, 8.3$ Kalman, Janet $24, 2.13$ Kalscheuer, Maria N/A Kang, Inda N/A Kang,	Johnsen, Harold	2.4; 2.13
Johnson, Brett $3, 2, 3.6$ Johnson, Deborah N/A Johnson, Douglas 2.4 Johnson, Douglas 2.4 Johnson, G.G. 2.4 ; 2.13 Johnson, G.G. 2.4 ; 2.13 Johnson, Martha 2.4 ; 2.13 Johnson, Karen N/A Johnson, Nartha 2.4 ; 2.13 Johnson, Rush 2.4 ; 2.13 Johnson, Nuskell 2.4 ; 2.13 Johnson, Vicki 2.4 ; 2.13 Johnson, Viginia N/A Johnson, Viginia N/A Johnson, Viginia N/A Johnson, Filip 2.4 ; 2.13 Jones, Andrea 2.4 ; 2.13 Jones, Andrea 2.4 ; 2.13 Jones, Cauel 2.4 Jones, Terri 2.4 Jones, Terri 2.4 Jones, Tran N/A Jonkosky, Carsandra N/A Jonkosky, Charles N/A Jonkosky, Charles N/A Jonkosky, Charles N/A Jordan, Catherine 2.4 Jordan, Catherine 2.4 Jordan, Kris 3.2 ; 3.6, 8.3 Jordan, Yashoda N/A Jordan, Yashoda N/A Jordan, Yashoda N/A Jordan, Yashoda N/A June, Ana 2.1 ; 2.2; 2.8; 2.14; 13.1 Jusaume, Carol 2.4 ; 2.13 Justice, Cheryl 2.4 ; 2.13 Kaiser, Terry 3.2 ; 3.6, 8.3 Kaitman, Sue 2.4 ; 2.13 Kairman, Sue 2.4 ; 2.13	Johnson, Arnold	N/A
Johnson, Bruce 24 Johnson, Deborah N/A Johnson, Deborah N/A Johnson, G.G. 24 Johnson, G.G. 24,2.13 Johnson, G.G. 24,2.13 Johnson, Karen N/A Johnson, Karen N/A Johnson, Karen 24,2.13 Johnson, Martha 24,2.13 Johnson, Nusell 24,2.13 Johnson, Vicki 24,2.13 Jones, Andra 24,2.13 Jones, Jauel 24 Jones, Jauel 24 Jones, Sina N/A Jonksky, Casandra N/A Jonkosky, Charles N/A Jordan, S. 24 Jordan, S. 24 Jordan, S. 24 Jordan, S. 24	Johnson, Brett	3.2; 3.6
Johnson, Chessa Rae 24, 2.13 Johnson, Debotah N/A Johnson, G.G. 2.4, 2.13 Johnson, G.G. 2.4, 2.13 Johnson, Karen N/A Johnson, Martha 2.4, 2.13 Johnson, Rich 3.2, 3.6, 8.3 Johnson, Rich 3.2, 3.6, 8.3 Johnson, Nich 2.4, 2.13 Johnson, Nich 3.2, 3.6, 8.3 Johnson, Viginia N/A Johnson, Viginia N/A Johnson, Viginia N/A Johnson, Viginia N/A Johnson, Philip 2.4, 2.13 Jones, Martin 3.2, 3.6, 8.3 Jones, Tina N/A Jones, Tina N/A Jones, Tina N/A Jonkosky, Cassandra N/A Jonkosky, Charles N/A Jordan, Cherine 2.4 Jordan, Cherine 2.4, 2.13 Jordan, Sa 2.4 Jordan, Sa 2.4 Jordan, Sa 2.4 Jordan, Sa 2.4 Jor	Johnson, Bruce	2.4
Johnson, DeborahN/AJohnson, G.G.2.4Johnson, G.G.2.4; 2.13Johnson, KarenN/AJohnson, KarenN/AJohnson, Karen2.4; 2.13Johnson, Rich3.2; 3.6; 8.3Johnson, Russell2.4; 2.13Johnson, Nussell2.4; 2.13Johnson, Vicki2.4; 2.13Johnson, ViginiaN/AJohnson, ViginiaN/AJohnson, ViginiaN/AJohnson, ViginiaN/AJohnson, ViginiaN/AJones, Andrea2.4; 2.13Jones, Andrea2.4; 2.13Jones, Ferri2.4Jones, Terri2.4Jones, SandraN/AJonksky, CasandraN/AJonksky, CharlesN/AJonksky, CharlesN/AJordan, Catherine2.4Jordan, S.2.4Jordan, S.2.4 <td>Johnson, Chessa Rae</td> <td>2.4; 2.13</td>	Johnson, Chessa Rae	2.4; 2.13
Johnson, G.G. 2.4 Johnson, G.G. 2.4; 2.13 Johnson, Jay N/A Johnson, Karen N/A Johnson, Martha 2.4; 2.13 Johnson, Martha 2.4; 2.13 Johnson, Kisch 3.2; 3.6; 8.3 Johnson, Viski 2.4; 2.13 Johnson, Viski 2.4; 2.13 Johnson, Pilip 2.4; 2.13 Johnson, Philip 2.4; 2.13 Johnson, Philip 2.4; 2.13 Jones, Martin 3.2; 3.6; 8.3 Jones, Sandrea 2.4; 2.13 Jones, Trina N/A Jones, Trina N/A Joneksy, Cassandra N/A Jonkosky, Cassandra N/A Jonkosky, Casandra N/A Jordan, Catherine 2.4 Jordan, Catherine 2.4 Jordan, S. 2.4 Jordan, Yashoda N/A Jordan, S. 2.4 Jordan, S. 2.4 Jordan, S. 2.4 Jordan, S. 2.4 Jordan, S.	Johnson, Deborah	N/A
Johnson, G.G. $2.4; 2.13$ Johnson, KarenN/AJohnson, KarenN/AJohnson, KarenN/AJohnson, KarenN/AJohnson, Rich $3.2; 36; 8.3$ Johnson, Russell $2.4; 2.13$ Johnson, Vicki $2.4; 2.13$ Jones, Martin $3.2; 36; 8.3$ Jones, Andrea $2.4; 2.13$ Jones, Ferri 2.4 Jones, Terri 2.4 Jones, SandraN/AJokosky, CharlesN/AJordan, Catherine 2.4 Jordan, Catherine 2.4 Jordan, Catherine 2.4 Jordan, S $2.4; 2.13$ Jordan, YashodaN/AJordan, S $2.4; 2.13$ Judy, PaulN/AJordan, S $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AKacaser, LindaN/AKadar, Zach $2.4; 2.13$ Kathani, Janet $2.4; 2.13$ Kathani, Sue $2.4; 2.13$	Johnson, Douglas	2.4
Johnson, JayN/AJohnson, KarenN/AJohnson, Martha $2.4; 2.13$ Johnson, Rusell $3.2; 3.6; 8.3$ Johnson, Rusell $2.4; 2.13$ Johnson, ToddN/AJohnson, Vicki $2.4; 2.13$ Johnson, ViginiaN/AJohnson, ViginiaN/AJohnson, ViginiaN/AJohnson, ViginiaN/AJohnson, ViginiaN/AJones, Martin $3.2; 3.6; 8.3$ Jones, Andrea $2.4; 2.13$ Jones, TriaN/AJones, TriaN/AJones, TriaN/AJones, TriaN/AJones, TriaN/AJonkosky, CasandraN/AJonkosky, CasandraN/AJordan, Catherine 2.4 Jordan, S. 2.4 Jorgensen, John $2.4; 2.13$ Jurdan, YashodaN/AJurgensen, John $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AKaser, LindaN/AKaser, Terry $3.2; 3.6; 8.3$ Kaitser, Terry	Johnson, G.G.	2.4; 2.13
Johnson, KarenN/AJohnson, Ratch $2.4; 2.13$ Johnson, Rich $3.2; 3.6; 8.3$ Johnson, Russell $2.4; 2.13$ Johnson, Vicki $2.4; 2.13$ Johnson, ViginiaN/AJohnson, ViginiaN/AJohnson, ViginiaN/AJohnson, Viginia $2.4; 2.13$ Jones, Martin $3.2; 3.6; 8.3$ Jones, Andrea $2.4; 2.13$ Jones, Ferri 2.4 Jones, Terri 2.4 Jonksy, CassandraN/AJonksy, CassandraN/AJonksy, CassandraN/AJordan, Catherine 2.4 Jordan, Sandra $2.4; 2.13$ Jordan, Catherine 2.4 Jordan, Kris $3.2; 3.6; 8.3$ Jordan, Sandra $2.4; 2.13$ Jordan, Kris $3.2; 3.6; 8.3$ Jordan, Sandra $2.4; 2.13$ Jordan, Sandra $2.4; 2.13$ Jordan, Sandra $2.4; 2.13$ Judy, PaulN/AJue, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AKadar, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kaiman, Janet $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kaiman, Janet $2.4; 2.13$ Karans, GaryNAKarans, GaryNAKarans, GaryNAKarans,	Johnson, Jay	N/A
Johnson, Martha $2.4; 2.13$ Johnson, Russell $2.4; 2.13$ Johnson, Russell $2.4; 2.13$ Johnson, ViginiaN/AJohnson, ViginiaN/AJohnson, ViginiaN/AJohnson, ViginiaN/AJohnson, ViginiaN/AJones, Martin $3.2; 3.6; 8.3$ Jones, Andrea $2.4; 2.13$ Jones, Andrea $2.4; 2.13$ Jones, Ferri 2.4 Jones, TinaN/AJonksky, CasandraN/AJonksky, CasandraN/AJonksky, CasandraN/AJonksky, CharlesN/AJordan, Catherine 2.4 Jordan, Catherine 2.4 Jordan, YashodaN/AJordan, YashodaN/AJorgensen, John $2.4; 2.13$ Judy, PaulN/AJune, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK, J.N/AKadar, Zach $2.4; 2.13$ Katiman, Janet $2.4; 2.13$ Kalser, Terry $3.2; 3.6; 8.3$ Kalman, Janet $2.4; 2.13$ Kalser, LindaN/AKarana, Sue $2.4; 2.13$ Karans, Sue $2.4; 2.13$	Johnson, Karen	N/A
Johnson, Rich 3.2; 3.6; 8.3 Johnson, Todd 2.4; 2.13 Johnson, Todd N/A Johnson, Vicki 2.4; 2.13 Johnson, Viginia N/A Johnson, Philip 2.4; 2.13 Jones, Andrea 2.4; 2.13 Jones, Andrea 2.4; 2.13 Jones, Lauel 2.4 Jones, Terri 2.4 Jones, Tina N/A Jokosky, Charles N/A Jordan, Kris 3.2; 3.6; 8.3 Jordan, Kris 3.2; 3.6; 8.3 Jordan, Kris 3.2; 3.6; 8.3 Jordan, S. 2.4 Jordan, S. 2.4 Jordan, S. 2.4 Jorgensen, Bob N/A Jorgensen, John 2.4; 2.13 Jussaume, Carol 2.4; 2.13 Justice, Cheryl 2.4; 2.13 Kadar, Zach 2.4; 2.13 Kaiser, Terry 3.2; 3.6; 8.3 Kaiha	Johnson, Martha	2.4; 2.13
Johnson, Russell $2.4; 2.13$ Johnson, Vicki $2.4; 2.13$ Johnson, ViginiaN/AJohnson, ViginiaNAJohnson, Philip $2.4; 2.13$ Jones, Martin $3.2; 3.6; 8.3$ Jones, Lauel $2.4; 2.13$ Jones, Ferri 2.4 Jones, TimaN/AJokosky, CassandraN/AJonkosky, CasandraN/AJonkosky, CharlesN/AJordan, Catherine 2.4 Jordan, Catherine 2.4 Jordan, Kris $3.2; 3.6; 8.3$ Jordan, S. 2.4 Judy, PaulN/AJussaume, Carol $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AKadar, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kalara, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kalman, Janet $2.4; 2.13$ Kaiser, LindaN/AKaranjawala, Eric & Armin $2.4; 2.13$ Karans, GaryN/AKarans, Sue $2.4; 2.13$ Karse, David $3.2; 3.6; 8.3$ <td>Johnson, Rich</td> <td>3.2; 3.6; 8.3</td>	Johnson, Rich	3.2; 3.6; 8.3
Johnson, Yicki N/A Johnson, Vicki 2.4; 2.13 Johnson, Viginia N/A Johnston, Philip 2.4; 2.13 Jones, Martin 3.2; 3.6; 8.3 Jones, Andrea 2.4; 2.13 Jones, Andrea 2.4; 2.13 Jones, Terri 2.4 Jones, Tina N/A Jonkosky, Cassandra N/A Jonkosky, Charles N/A Jonkosky, Charles N/A Jondan, Catherine 2.4 Jordan, Catherine 2.4 Jordan, Kris 3.2; 3.6; 8.3 Jordan, S. 2.4 Jordan, Yashoda N/A Jorgensen, John 2.4; 2.13 Jordan, Yashoda N/A Jorgensen, John 2.4; 2.13 Judy, Paul N/A Jusce, Cheryl 2.4; 2.13 Justice, Robert (Electrical District #7) N/A K, Matt N/A Kadar, Zach 2.4; 2.13 Kaiser, Terry 3.2; 3.6; 8.3 Kalscheuer, Maria N/A <td>Johnson, Russell</td> <td>2.4; 2.13</td>	Johnson, Russell	2.4; 2.13
Johnson, Vicki $24; 2.13$ Johnson, ViginiaN/AJohnston, Philip $24; 2.13$ Jones, Martin $32; 3.6; 8.3$ Jones, Andrea $2.4; 2.13$ Jones, Izuel 24 Jones, Tira 2.4 Jones, TinaN/AJonksky, CassandraN/AJonksky, CharlesN/AJordan, Catherine 2.4 Jordan, Kris $32; 3.6; 8.3$ Jordan, S. 2.4 Jordan, S. 2.4 Jorgensen, John $2.4; 2.13$ Jussaume, Carol $2.4; 2.13$ Jussaume, Carol $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK, MattN/AKadar, Zach $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AKadar, Zach $2.4; 2.13$ Kadar, Zach $2.4; 2.13$ Kadar, Zach $2.4; 2.13$ Kadar, Zach $2.4; 2.13$ Kadar, Zach $2.4; 2.13$ Katarajawala, Eric & Armin $2.4; 2.13$ Karna, Janet $2.4; 2.13$ Karna, Janet $2.4; 2.13$ Karna, Janet $2.4; 2.13$ Karna, Janet $2.4; 2.13$ Karna, Sue $2.4; 2.13$ Karna, Sue $2.4; 2.13$ Karna, Sue $2.4; 2.13$	Johnson, Todd	N/A
Johnson, ViginiaNAJohnston, Philip $2.4; 2.13$ Jones, Martin $3.2; 3.6; 8.3$ Jones, Andrea $2.4; 2.13$ Jones, Lauel 2.4 Jones, Terri 2.4 Jones, TinaNAJonkosky, CassandraNAJonkosky, CharlesN/AJordan, Catherine 2.4 Jordan, Kris $3.2; 3.6; 8.3$ Jordan, S. 2.4 Jordan, S. 2.4 Jordan, S. 2.4 Jorgensen, BobN/AJorgensen, John $2.4; 2.13$ Juure, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK, MattN/AKaser, LindaN/AKaser, LindaN/AKaser, LindaN/AKanda, Janet $2.4; 2.13$ Kana, Leah $2.4; 2.13$ Kaser, LindaN/AKaser, LindaN/AKaranjawala, Eric & Armin $2.4; 2.13$ Karanjawala, Eric & Armin $2.4; 2.13$ Karnajawala, Eric & Armin $2.4; 2.13$ Karna, Sue $2.4; 2.13$	Johnson, Vicki	2.4; 2.13
Johnston, Philip $2.4; 2.13$ Jones, Matria $3.2; 3.6; 8.3$ Jones, Andrea $2.4; 2.13$ Jones, Lauel 2.4 Jones, Terri 2.4 Jones, TimaN/AJonkosky, CasandraN/AJokosky, CharlesN/AJordan, Catherine $2.4; 2.13$ Jordan, Catherine $2.4; 2.13$ Jordan, Kris $3.2; 3.6; 8.3$ Jordan, Kris $3.2; 3.6; 8.3$ Jordan, S. 2.4 Jordan, S. 2.4 Jordan, YashodaN/AJorgensen, John $2.4; 2.13$ Judy, PaulN/AJusseaume, Carol $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK, MattN/AKadar, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kadar, Zach $2.4; 2.13$ Kadar, Zach $2.4; 2.13$ Kadar, Zach $2.4; 2.13$ Kadar, Zach $2.4; 2.13$ Karanjawala, Eric & Armin $2.4; 2.13$ Karanjawala, Eric & Armin $2.4; 2.13$ Karanjawala, Eric & Armin $2.4; 2.13$ Karna, Sue $2.4; 2.13$	Johnson, Viginia	N/A
Jones, Martin $3.2; 3.6; 8.3$ Jones, Andrea $2.4; 2.13$ Jones, Lauel 2.4 Jones, Terri 2.4 Jones, TinaN/AJonkosky, CassandraN/AJonkosky, CharlesN/AJoos, Sandra $2.4; 2.13$ Jordan, Catherine 2.4 Jordan, Catherine 2.4 Jordan, S. $2.4; 2.13$ Jordan, S. 2.4 Jordan, S. 2.4 Jordan, S. 2.4 Jorgensen, BobN/AJorgensen, John $2.4; 2.13$ Judy, PaulN/AJune, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Jussaume, Carol $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK, MattN/AKadar, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kaiser, LindaN/AKadar, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Karanjawala, Eric & Armin $2.4; 2.13$ Karnajawala, Eric & Armin $2.4; 2.13$ Karnajawala, Eric & Armin $2.4; 2.13$ Karra, Sheena $2.4; 2.13$ Karra, Sheena $2.4; 2.13$	Johnston, Philip	2.4; 2.13
Jones, Andrea $2.4; 2.13$ Jones, Lauel 2.4 Jones, Terri 2.4 Jones, TinaNAJonkosky, CassandraN/AJonkosky, CharlesN/AJordan, Catherine $2.4; 2.13$ Jordan, Catherine 2.4 Jordan, Kris $3.2; 3.6; 8.3$ Jordan, S. 2.4 Jordan, YashodaN/AJorgensen, BobN/AJorgensen, John $2.4; 2.13$ Jussaume, Carol $2.4; 2.13$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK,J.N/AKadar, Zach $2.4; 2.13$ Katar, Janet $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kalma, Janet $2.4; 2.13$ Kalma, Janet $2.4; 2.13$ Kanan, Janet $2.4; 2.13$ Karanjawala, Eric & Armin $2.4; 2.13$ Karranjawala, Eric & Armin $2.4; 2.13$ Karranjawala, Eric & Armin $2.4; 2.13$ Karran, Sue $2.4; 2.13$	Jones, Martin	3.2; 3.6; 8.3
Jones, Lauel 2.4 Jones, Terri 2.4 Jones, Tina N/A Jonkosky, Cassandra N/A Jonkosky, Charles N/A Jordan, Support $2.4; 2.13$ Jordan, Kris $3.2; 3.6; 8.3$ Jordan, Kris $3.2; 3.6; 8.3$ Jordan, S. 2.4 Jordan, S. 2.4 Jorgensen, Bob N/A Jorgensen, John $2.4; 2.13$ Judy, Paul N/A June, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Jussaume, Carol $2.4; 2.13$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7) N/A K Adar, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kaiser, Terry $3.2; 3.6; 8.3$ Kalscheuer, Maria N/A Kane, Leah $2.4; 2.13$ Karanjawala, Eric & Armin $2.4; 2.13$ Karnajawala, Eric & Armin <t< td=""><td>Jones, Andrea</td><td>2.4; 2.13</td></t<>	Jones, Andrea	2.4; 2.13
Jones, Terri 2.4 Jones, TinaN/AJonkosky, CassandraN/AJonkosky, CharlesN/AJoos, Sandra 2.4 ; 2.13 Jordan, Catherine 2.4 Jordan, Kris 3.2 ; 3.6 ; 8.3 Jordan, S. 2.4 Jordan, S. 2.4 Jordan, YashodaN/AJorgensen, BobN/AJorgensen, John 2.4 ; 2.13 Judy, PaulN/AJues, Ana 2.1 ; 2.2 ; 2.8 ; 2.14 ; 13.1 Jussaume, Carol 2.4 ; 2.13 Justice, Robert (Electrical District #7)N/AK, MattN/AKadar, Zach 2.4 ; 2.13 Kaiser, Terry 3.2 ; 3.6 ; 8.3 Kalman, Janet 2.4 ; 2.13 Kalscheuer, MariaN/AKaranjawala, Eric & Armin 2.4 ; 2.13 Karns, GaryN/AKarrs, David 3.2 ; 3.6 ; 8.3 Kartman, Sue 2.4 ; 2.13	Jones, Lauel	2.4
Jones, Tina N/A Jonkosky, Casandra N/A Jonkosky, Charles N/A Joos, Sandra $2.4; 2.13$ Jordan, Catherine 2.4 Jordan, Kris $3.2; 3.6; 8.3$ Jordan, S. 2.4 Jordan, S. 2.4 Jordan, Yashoda N/A Jorgensen, Bob N/A Jorgensen, John $2.4; 2.13$ Judy, Paul N/A June, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7) N/A K, Matt N/A Kacser, Linda N/A Kadar, Zach $2.4; 2.13$ Kalscheur, Maria N/A Kane, Leah $2.4; 2.13$ Kane, Linda N/A Karanjawala, Eric & Armin $2.4; 2.13$ Karns, Gary N/A Karrs, David $3.2; 3.6; 8.3$ Karns, Sue $2.4; 2.13$	Jones, Terri	2.4
Jonkosky, Cassandra N/A Jonkosky, Charles N/A Joos, Sandra $2.4; 2.13$ Jordan, Catherine 2.4 Jordan, Kris $3.2; 3.6; 8.3$ Jordan, S. 2.4 Jordan, S. 2.4 Jordan, S. 2.4 Jorgensen, Bob N/A Jorgensen, John $2.4; 2.13$ Judy, Paul N/A June, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7) N/A K, Matt N/A Kacser, Linda N/A Kalc, Zach $2.4; 2.13$ Kalscheuer, Maria N/A Kane, Leah $2.4; 2.13$ Kane, Linda N/A Karanjawala, Eric & Armin $2.4; 2.13$ Karns, Gary N/A Karns, David $2.4; 2.13$ Karns, David $2.4; 2.13$ Karns, Sue $2.4; 2.13$	Jones, Tina	N/A
Jonkosky, CharlesN/AJoos, Sandra 2.4 ; 2.13 Jordan, Catherine 2.4 Jordan, Kriis 3.2 ; 3.6 ; 8.3 Jordan, S. 2.4 Jordan, YashodaN/AJorgensen, BobN/AJorgensen, John 2.4 ; 2.13 Judy, PaulN/AJune, Ana 2.1 ; 2.2 ; 2.8 ; 2.14 ; 13.1 Jussaume, Carol 2.4 ; 2.13 Justice, Cheryl 2.4 ; 2.13 Justice, Robert (Electrical District #7)N/AK, MattN/AKadar, Zach 2.4 ; 2.13 Kaiser, Terry 3.2 ; 3.6 ; 8.3 Kalman, Janet 2.4 ; 2.13 Kane, LeahN/AKaranjawala, Eric & Armin 2.4 ; 2.13 Karrs, GaryN/AKarrs, David 3.2 ; 3.6 ; 8.3 Karrs, David 3.2 ; 3.6 ; 8.3 Karrs, David 2.4 ; 2.13	Jonkosky, Cassandra	N/A
Joos, Sandra $2.4; 2.13$ Jordan, Catherine 2.4 Jordan, Kris $3.2; 3.6; 8.3$ Jordan, S. 2.4 Jordan, YashodaN/AJorgensen, BobN/AJorgensen, John $2.4; 2.13$ Judy, PaulN/AJune, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Jussaume, Carol $2.4; 2.13$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK, MattN/AKadar, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kalman, Janet $2.4; 2.13$ Kalscheuer, MariaN/AKaranjawala, Eric & Armin $2.4; 2.13$ Karrs, GaryN/AKarrs, David $3.2; 3.6; 8.3$ Karrs, David $3.2; 3.6; 8.3$ Karrs, David $2.4; 2.13$	Jonkosky, Charles	N/A
Jordan, Catherine 2.4 Jordan, Kris $3.2; 3.6; 8.3$ Jordan, S. 2.4 Jordan, YashodaN/AJorgensen, BobN/AJorgensen, John $2.4; 2.13$ Judy, PaulN/AJune, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Jussaume, Carol $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK, MattN/AK, MattN/AKacser, LindaN/AKadar, Zach $2.4; 2.13$ Kalkeneur, MariaN/AKanan, Janet $2.4; 2.13$ Kane, Leah $2.4; 2.13$ Karanjawala, Eric & Armin $2.4; 2.13$ Karrs, GaryN/AKarrs, David $3.2; 3.6; 8.3$ Karrs, David $3.2; 3.6; 8.3$ Karrs, David $2.4; 2.13$	Joos, Sandra	2.4; 2.13
Jordan, Kris $3.2; 3.6; 8.3$ Jordan, S. 2.4 Jordan, YashodaN/AJorgensen, BobN/AJorgensen, John $2.4; 2.13$ Judy, PaulN/AJune, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Jussaume, Carol $2.4; 2.13$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK, MattN/AKaseer, LindaN/AKaiser, Terry $3.2; 3.6; 8.3$ Kaiser, Terry $2.4; 2.13$ Kalscheuer, MariaN/AKane, Leah $2.4; 2.13$ Karanjawala, Eric & Armin $2.4; 2.13$ Karns, GaryN/AKarrs, David $2.4; 2.13$ Karrs, David $2.2; 3.6; 8.3$ Karman, Sue $2.4; 2.13$	Jordan, Catherine	2.4
Jordan, S. 2.4 Jordan, YashodaN/AJorgensen, BobN/AJorgensen, John 2.4 ; 2.13 Judy, PaulN/AJune, Ana 2.1 ; 2.2 ; 2.8 ; 2.14 ; 13.1 Jussaume, Carol 2.4 ; 2.13 Justice, Cheryl 2.4 ; 2.13 Justice, Robert (Electrical District #7)N/AK, MattN/AK,J.N/AKacser, LindaN/AKaiser, Terry 3.2 ; 3.6 ; 8.3 Kalman, Janet 2.4 ; 2.13 Kascheuer, MariaN/AKane, Leah 2.4 ; 2.13 Karanjawala, Eric & Armin 2.4 ; 2.13 Karrs, GaryN/AKarrs, David 3.2 ; 3.6 ; 8.3 Karrs, David 3.2 ; 3.6 ; 8.3 Karrs, David 3.2 ; 3.6 ; 8.3 Karrs, David 3.2 ; 3.6 ; 8.3	Jordan, Kris	3.2; 3.6; 8.3
Jordan, YashodaN/AJorgensen, BobN/AJorgensen, John 2.4 ; 2.13 Judy, PaulN/AJune, Ana 2.1 ; 2.2 ; 2.8 ; 2.14 ; 13.1 Jussaume, Carol 2.4 ; 2.13 Justice, Cheryl 2.4 ; 2.13 Justice, Robert (Electrical District #7)N/AK, MattN/AK,J.N/AKacser, LindaN/AKadar, Zach 2.4 ; 2.13 Kalman, Janet 2.4 ; 2.13 Kalscheuer, MariaN/AKare, Leah 2.4 ; 2.13 Karanjawala, Eric & Armin 2.4 ; 2.13 Karrs, GaryN/AKars, David 3.2 ; 3.6 ; 8.3 Karman, Sue 2.4 ; 2.13	Jordan, S.	2.4
Jorgensen, BobN/AJorgensen, John 2.4 ; 2.13 Judy, PaulN/AJune, Ana 2.1 ; 2.2 ; 2.8 ; 2.14 ; 13.1 Jussaume, Carol 2.4 ; 2.13 Justice, Cheryl 2.4 ; 2.13 Justice, Robert (Electrical District #7)N/AK, MattN/AK,J.N/AKacser, LindaN/AKadar, Zach 2.4 ; 2.13 Kaiser, Terry 3.2 ; 3.6 ; 8.3 Kalman, Janet 2.4 ; 2.13 Kane, Leah 2.4 ; 2.13 Karanjawala, Eric & Armin 2.4 ; 2.13 Karns, GaryN/AKarrs, David 3.2 ; 3.6 ; 8.3 Karrs, David 3.2 ; 3.6 ; 8.3 Karman, Sue 2.4 ; 2.13	Jordan, Yashoda	N/A
Jorgensen, John $2.4; 2.13$ Judy, PaulN/AJune, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Jussaume, Carol $2.4; 2.13$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK, MattN/AK,J.N/AKacser, LindaN/AKadar, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kalman, Janet $2.4; 2.13$ Kascheuer, MariaN/AKaranjawala, Eric & Armin $2.4; 2.13$ Karrs, GaryN/AKarrs, David $3.2; 3.6; 8.3$ Kartman, Sue $2.4; 2.13$	Jorgensen, Bob	N/A
Judy, PaulN/AJune, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Jussaume, Carol $2.4; 2.13$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK, MattN/AK,J.N/AKacser, LindaN/AKadar, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kalman, Janet $2.4; 2.13$ Kalscheuer, MariaN/AKane, Leah $2.4; 2.13$ Karanjawala, Eric & Armin $2.4; 2.13$ Karrs, GaryN/AKarrs, David $3.2; 3.6; 8.3$ Kartman, Sue $2.4; 2.13$	Jorgensen, John	2.4; 2.13
June, Ana $2.1; 2.2; 2.8; 2.14; 13.1$ Jussaume, Carol $2.4; 2.13$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK, MattN/AK,J.N/AKacser, LindaN/AKadar, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kalman, Janet $2.4; 2.13$ Kase, Leah $2.4; 2.13$ Kane, Leah $2.4; 2.13$ Karns, GaryN/AKarr, Sheena $2.4; 2.13$ Karrs, David $3.2; 3.6; 8.3$ Kartman, Sue $2.4; 2.13$	Judy, Paul	N/A
Jussaume, Carol $2.4; 2.13$ Justice, Cheryl $2.4; 2.13$ Justice, Robert (Electrical District #7)N/AK, MattN/AK,J.N/AKacser, LindaN/AKadar, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kalman, Janet $2.4; 2.13$ Kase, Leah $2.4; 2.13$ Kane, Leah $2.4; 2.13$ Karen, LindaN/AKaren, SaryN/AKarr, Sheena $2.4; 2.13$ Karrs, David $3.2; 3.6; 8.3$ Kartman, Sue $2.4; 2.13$	June, Ana	2.1; 2.2; 2.8; 2.14; 13.1
Justice, Cheryl2.4; 2.13Justice, Robert (Electrical District #7)N/AK, MattN/AK,J.N/AKacser, LindaN/AKadar, Zach2.4; 2.13Kaiser, Terry3.2; 3.6; 8.3Kalman, Janet2.4; 2.13Kalscheuer, MariaN/AKane, Leah2.4; 2.13Karanjawala, Eric & Armin2.4; 2.13Karrs, GaryN/AKarrs, David3.2; 3.6; 8.3Kartman, Sue2.4; 2.13	Jussaume, Carol	2.4; 2.13
Justice, Robert (Electrical District #7)N/AK, MattN/AK,J.N/AKacser, LindaN/AKadar, Zach2.4; 2.13Kaiser, Terry3.2; 3.6; 8.3Kalman, Janet2.4; 2.13Kalscheuer, MariaN/AKane, Leah2.4; 2.13Karanjawala, Eric & Armin2.4; 2.13Karns, GaryN/AKarr, Sheena2.4; 2.13Karrs, David3.2; 3.6; 8.3Kartman, Sue2.4; 2.13	Justice, Cheryl	2.4; 2.13
K, Matt N/A K,J. N/A Kacser, Linda N/A Kadar, Zach $2.4; 2.13$ Kaiser, Terry $3.2; 3.6; 8.3$ Kalman, Janet $2.4; 2.13$ Kalscheuer, Maria N/A Kane, Leah $2.4; 2.13$ Kane, Linda N/A Karanjawala, Eric & Armin $2.4; 2.13$ Karrs, Gary N/A Karr, Sheena $2.4; 2.13$ Karrs, David $3.2; 3.6; 8.3$ Kartman, Sue $2.4; 2.13$	Justice, Robert (Electrical District #7)	N/A
K,J. N/A Kacser, Linda N/A Kadar, Zach 2.4 ; 2.13 Kaiser, Terry 3.2 ; 3.6 ; 8.3 Kalman, Janet 2.4 ; 2.13 Kalscheuer, Maria N/A Kane, Leah 2.4 ; 2.13 Kare, Linda N/A Karanjawala, Eric & Armin 2.4 ; 2.13 Karns, Gary N/A Karr, Sheena 2.4 ; 2.13 Karrs, David 3.2 ; 3.6 ; 8.3 Kartman, Sue 2.4 ; 2.13	K, Matt	N/A
Kacser, Linda N/A Kadar, Zach 2.4; 2.13 Kaiser, Terry 3.2; 3.6; 8.3 Kalman, Janet 2.4; 2.13 Kalscheuer, Maria N/A Kane, Leah 2.4; 2.13 Karanjawala, Eric & Armin 2.4; 2.13 Karns, Gary N/A Karr, Sheena 2.4; 2.13 Karrs, David 3.2; 3.6; 8.3 Kartman, Sue 2.4; 2.13	K,J.	N/A
Kadar, Zach 2.4; 2.13 Kaiser, Terry 3.2; 3.6; 8.3 Kalman, Janet 2.4; 2.13 Kalscheuer, Maria N/A Kane, Leah 2.4; 2.13 Karanjawala, Eric & Armin 2.4; 2.13 Karns, Gary N/A Karr, Sheena 2.4; 2.13 Karrs, David 3.2; 3.6; 8.3 Kartman, Sue 2.4; 2.13	Kacser, Linda	N/A
Kaiser, Terry 3.2; 3.6; 8.3 Kalman, Janet 2.4; 2.13 Kalscheuer, Maria N/A Kane, Leah 2.4; 2.13 Kane, Linda N/A Karanjawala, Eric & Armin 2.4; 2.13 Karns, Gary N/A Karr, Sheena 2.4; 2.13 Karrs, David 3.2; 3.6; 8.3 Kartman, Sue 2.4; 2.13	Kadar, Zach	2.4; 2.13
Kalman, Janet2.4; 2.13Kalscheuer, MariaN/AKane, Leah2.4; 2.13Kane, LindaN/AKaranjawala, Eric & Armin2.4; 2.13Karns, GaryN/AKarr, Sheena2.4; 2.13Karrs, David3.2; 3.6; 8.3Kartman, Sue2.4; 2.13	Kaiser, Terry	3.2; 3.6; 8.3
Kalscheuer, MariaN/AKane, Leah2.4; 2.13Kane, LindaN/AKaranjawala, Eric & Armin2.4; 2.13Karns, GaryN/AKarr, Sheena2.4; 2.13Karrs, David3.2; 3.6; 8.3Kartman, Sue2.4; 2.13	Kalman, Janet	2.4; 2.13
Kane, Leah2.4; 2.13Kane, LindaN/AKaranjawala, Eric & Armin2.4; 2.13Karns, GaryN/AKarr, Sheena2.4; 2.13Karrs, David3.2; 3.6; 8.3Kartman, Sue2.4; 2.13	Kalscheuer, Maria	N/A
Kane, LindaN/AKaranjawala, Eric & Armin2.4; 2.13Karns, GaryN/AKarr, Sheena2.4; 2.13Karrs, David3.2; 3.6; 8.3Kartman, Sue2.4; 2.13	Kane, Leah	2.4; 2.13
Karanjawala, Eric & Armin2.4; 2.13Karns, GaryN/AKarr, Sheena2.4; 2.13Karrs, David3.2; 3.6; 8.3Kartman, Sue2.4; 2.13	Kane, Linda	N/A
Karns, GaryN/AKarr, Sheena2.4; 2.13Karrs, David3.2; 3.6; 8.3Kartman, Sue2.4; 2.13	Karanjawala, Eric & Armin	2.4; 2.13
Karr, Sheena2.4; 2.13Karrs, David3.2; 3.6; 8.3Kartman, Sue2.4; 2.13	Karns, Gary	N/A
Karrs, David 3.2; 3.6; 8.3 Kartman, Sue 2.4; 2.13	Karr, Sheena	2.4; 2.13
Kartman, Sue 2.4; 2.13	Karrs, David	3.2; 3.6; 8.3
	Kartman, Sue	2.4; 2.13

Kashinski, David	2.4; 2.13
Kastel, Diane	2.4; 2.13
Katten, DC	N/A
Katz, Ron	2.4; 2.13
Kaufman, Andrea	2.4; 2.13
Kaufman, Barry	2.4; 2.13
Kaufman, Joan	2.4
Kaufman, Melanie	N/A
Kava, John	3.2; 3.6
Kawa, Sandra	2.4; 2.13
Kawszan, Karen	2.4; 2.13
Kaye, Jackie	2.4; 2.13
Kearns, Peter	N/A
Keck, Robert	2.1; 2.2; 2.8; 2.14; 13.1
Keegan, Barbara	2.4; 2.13
Keegan, Helen	N/A
Keenan, Thomas	N/A
Keeton, Hank	2.4; 2.13
Keim, Lisa	2.4: 2.13
Keiser, Robert	2.4
Keith, Kristin	2.4: 2.13
Kellermann, Thomasin	N/A
Kelley. Dorinda	N/A
Kelly. Diane	2.4
Kelly, Donell	N/A
Kelly, Steve	3.6
Kelly, Wayne	2.4; 2.13
Kelm, Kathleen	2.4; 2.13
Kemper, Erik	N/A
Kendrick, Cindy	2.4; 2.13
Kendy, Arthur	2.4; 2.13
Kent, Ellen	N/A
Kentfield, Maren	N/A
Kepner, Sethsue	2.4; 2.13
Kerivan, Andrea	N/A
Kerman, Paul	2.4; 2.13
Kern, Alicia	2.4; 2.13
Kerwell, Cherrie	2.4; 2.13
Kerwin, Patrick	N/A
Kessler, Susan	2.4; 2.13
Kettelhut, H.	N/A
Key, Laurence	2.4; 2.13
Keymolent, Claudia	2.4; 2.13
Khalsa, Mha Atma S.	2.4; 2.13
Kibler, JK	2.4
Kieffer, Ramsay	2.4
Kiel, Edward	3.2; 3.6
Kilpatrick, Leslie	2.4; 2.13
Kincaid, Ted	2.4; 2.13
Kindel, Karen	2.4

King, KathyN/AKing, KathyN/AKing, RuthN/AKing, RuthN/AKingsley, Robert 2.4 ; 2.13Kinkad, TimothyN/AKirk, Nancy Jo 2.4 ; 2.13Kirkad, TimothyN/AKirk, Nancy Jo 2.4 ; 2.13Kirke, Richard 2.4 ; 2.13Kitte, Richard 2.4 ; 2.13Kitts, Richard 2.4 ; 2.13Kitts, Richard 2.4 ; 2.13Kitts, Richard 2.4 ; 2.13Kitts, Richard 2.4 ; 2.13Kits, Richard 2.4 ; 2.13Kine, Mark 2.4 ; 2.13Kine, Regan $32, 3.6; 8.3$ Klinkhamer, Trevor12.2Klipfel II, George 2.4 ; 2.13Knickerbocker, AIN/AKniekerbocker, AIN/AKnoel, Carolyn 2.4 ; 2.13Knoth, Dean 2.4 ; 2.13Koch, James 2.4 ; 2.13Koch, James 2.4 ; 2.13Koch, SusanneN/AKomisarof, JeffN/AKorese, Rat 2.4 ; 2.13Koristo, Ellen 2.4 ; 2.13Koressel, Karl 2.4 ; 2.13Krath, Tessa		
King, KellyN/AKing, RuthN/AKing, StephenN/AKing, StephenN/AKinked, TimothyN/AKinked, TimothyN/AKirk, Nancy Jo2.4, 2.13Kirkharits, Mary2.4Kirsh, JulieN/AKirk, Richard2.4, 2.13Kirtk, Mick, Mary2.4Kirsh, JulieN/AKirte, Part2.4, 2.13Kirts, MicheleN/AKire, David2.8, 2.1, 41, 1.5Klass, David2.4, 2.13Klein, Mark2.4, 2.13Klein, Mark2.4, 2.13Klink, Mark,2.4, 2.13Klink, Mark,2.4, 2.13Klink, Mark,2.4, 2.13Klinkhamer, Trevor1.2.2Klipfel II, George2.4, 2.13Kniekerbocker, AlN/AKniekerbocker, AlN/AKniekerbocker, AlN/AKniekerbocker, AlN/AKohl, James2.4, 2.13Koch, James2.4, 2.13Korder, Susan2.4, 2.13Korder, Susan2.4, 2.13Korder, Susan2.4, 2.13Korder, Susan2.4, 2.13Korder, Susan2.4, 2.13Kortepa, Michael2.4, 2.13Krame	King, Kathy	N/A
King, RuthN/AKing, StephenN/AKing, StephenN/AKing, StephenN/AKink, Naroy Jo2.4; 2.13Kirk, Naroy Jo2.4; 2.13Kirk, Naroy Jo2.4; 2.13Kirk, Naroy Jo2.4; 2.13Kirta, JulieN/AKita, Kichela2.4; 2.13Kitt, KicheleN/AKitts, MicheleN/AKitts, MicheleN/AKitts, MicheleN/AKitts, MicheleN/AKitts, MicheleN/AKitts, Michele2.4; 2.13Kitts, Mark2.4; 2.13Kitts, Mark2.4; 2.13Kitts, Mark2.4; 2.13Kitine, Regan3.2; 3.6; 8.3Kline, Regan3.2; 3.6; 8.3Kline/Regan2.4; 2.13Knickerbocker, AlN/AKnickerbocker, AlN/AKnort, Carl2.4; 2.13Knort, Carl2.4; 2.13Koch, Janes2.4; 2.13Koch, Janes2.4; 2.13Koch, SusameN/AKoeser, LelandN/AKoeser, LelandN/AKortex, JeffN/AKortex, Terge2.4; 2.13Kortex, Terge2.4; 2.13Kraner, GreggN/AKortexer, Susan2.4; 2.13Kraner, Karen2.42.42.13Kraner, Keren2.42.42.13Kraner, Fileen2.4; 2.13Kraner, Grant2.42.42.13Kraner, Grant2.4<	King, Kelly	N/A
King, StephenN/AKinggley, Robert $2.4; 2.13$ Kinkgat, TimothyN/AKirk, Nancy Jo $2.4; 2.13$ Kirkpatrick, Mary 2.4 Kirkpatrick, Mary 2.4 Kirkpatrick, Mary $2.4; 2.13$ Kitte, Richard $2.4; 2.13$ Kitte, Pat $2.4; 2.13$ Kitts, MicheleN/AKizer, David $2.8; 2.14; 11.5$ Klass, David $2.4; 2.13$ Kits, NicheleN/AKite, Regan $3.2; 3.6; 8.3$ Klink, Mark $2.4; 2.13$ Klink, Regan $3.2; 3.6; 8.3$ Klinkhamer, Trevor 12.2 Klipfell, George $2.4; 2.13$ Knickerbocker, AlN/AKnord, Carl $2.4; 2.13$ Knord, Carl $2.4; 2.13$ Koch, Jaann $2.4; 2.13$ Koch, Jaann $2.4; 2.13$ Koch, Joann $2.4; 2.13$ Koch, Joann $2.4; 2.13$ Koch, SusanneN/AKoesser, LelandN/AKoesser, Karen $2.4; 2.13$ Kortor, Ciflen $2.4; 2.13$ Kortor, GreggN/AKonzen, Susan $2.4; 2.13$ Kortore, Karen $2.4; 2.13$ Kortore, Susan $2.4; 2.13$ Kortore, Karen $2.4; 2.13$ Kramer, Fileen $2.4; 2.13$ Kramer, Fileen $2.4; 2.1$	King, Ruth	N/A
Kingsley, Robert $2, 2, 13$ Kinkead, Timothy NA Kirk, Narcy Jo $2, 4; 2, 13$ Kirsh, Julie NA Kirsh, Julie NA Kitsh, Kancy Jo $2, 4; 2, 13$ Kirsh, Julie NA Kits, Kichard $2, 4; 2, 13$ Kitts, Michele NA Kitz, Kichele NA Kitz, Gavid $2, 4; 2, 13$ Kits, Spavid $2, 4; 2, 13$ Kias, David $2, 4; 2, 13$ Kine, Regan $3; 2, 3, 6; 8, 3$ Klinkhamer, Trevor 122 Klipfell, George $2, 4; 2, 13$ Knickerbocker, Al N/A Koner, Carl $2, 4; 2, 13$ Knuth, Dean $2, 4; 2, 13$ Koch, Janes $2, 4; 2, 13$ Koch, Janan $2, 4; 2, 13$ Koch, Susanne N/A Kolesser, Karl $2, 4; 2, 13$ Koiviso, Ellen $2, 4; 2, 13$ Kobelssar, Gregg <	King, Stephen	N/A
Kinkead, Timothy N/A Kirk, Nancy Jo 2.4 ; 2.13 Kirkpatrick, Mary 2.4 Kirkpatrick, Mary 2.4 Kirkpatrick, Mary 2.4 ; 2.13 Kitte, Richard 2.4 ; 2.13 Kitts, Nichele N/A Kitzer, David 2.8 ; 2.14; 11.5 Klass, David 2.4 ; 2.13 Klink, Mark 2.4 ; 2.13 Klink, Mark 2.4 ; 2.13 Klink, Mark 2.4 ; 2.13 Knick, Betty 2.4 ; 2.13 Knicks, Betty 2.4 ; 2.13 Knort, Carl 2.4 ; 2.13 Knort, Carl 2.4 ; 2.13 Koch, Joann 2.4 ; 2.13 Kortsen, Susan 2.4 ; 2.13 Kortsen, Susan 2.4 ; 2.13	Kingsley, Robert	2.4; 2.13
Kirk, Nancy Jo 24; 2.13 Kirkpatrick, Mary 2.4 Kirsh, Julie NA Kite, Richard 2.4; 2.13 Kitte, Richard 2.4; 2.13 Kitte, Pat 2.4; 2.13 Kitts, Michele NA Kitz, David 2.8; 2.14; 11.5 Klass, David 2.4; 2.13 Kline, Regan 3.2; 3.6; 8.3 Klinkhamer, Trevor 12.2 Klipfel II, George 2.4; 2.13 Knickerbocker, Al NA Knickerbocker, Al NA Knoll, Carolyn 2.4; 2.13 Kond, Dean 2.4; 2.13 Koch, James 2.4; 2.13 Koch, James 2.4; 2.13 Koch, James 2.4; 2.13 Koch, Joann 2.4; 2.13 Koch, Joann 2.4; 2.13 Kobeser, Leland N/A Koesseser, Leland N/A Koesse	Kinkead, Timothy	N/A
Kirkpatrick, Mary 2.4 Kirsh, Julie N/A Kite, Richard 2.4; 2.13 Kitte, Pat 2.4; 2.13 Kitte, Pat 2.4; 2.13 Kitts, Michele N/A Kits, David 2.8; 2.14; 11.5 Klass, David 2.4; 2.13 Klass, David 2.4; 2.13 Klass, David 2.4; 2.13 Klass, David 2.4; 2.13 Kline, Regan 3.2; 3.6; 8.3 Kline, Regan 3.2; 3.6; 8.3 Klinkhamer, Tevor 12.2 Klipfel II, George 2.4; 2.13 Knickerbocker, Al N/A Kniess, Bethy 2.4; 2.13 Knight, Robert N/A Knort, Carl 2.4; 2.13 Knuth, Dean 2.4; 2.13 Koch, James 2.4; 2.13 Koch, Susanne N/A Koeser, Leland N/A Koeser, Leland N/A Korezen, Susan 2.4; 2.13 Korten, Susann 2.4; 2.13 Korten, Susann 2.4; 2.13 Koeser, Leland N/A Koeser, Leland N/A Korezen, Susan 2.4; 2.13 Korten, Karen 2.4 Kosee, Dawn 2.4; 2.13 Ko	Kirk, Nancy Jo	2.4; 2.13
Kirsh, Julie N/A Kite, Richard 2.4; 2.13 Kitte, Pat 2.4; 2.13 Kitts, Michele N/A Kizer, David 2.8; 2.14; 11.5 Klass, David 2.4; 2.13 Klass, David 2.4; 2.13 Klass, David 2.4; 2.13 Klass, David 2.4; 2.13 Kline, Regan 3.2; 3.6; 8.3 Klinkhamer, Trevor 12.2 Klinfell, George 2.4; 2.13 Knickerbocker, Al N/A Kneickerbocker, Al N/A Knickerbocker, Al N/A Knorr, Carl 2.4; 2.13 Knorr, Carl 2.4; 2.13 Knorr, Carl 2.4; 2.13 Koch, James 2.4; 2.13 Koch, James 2.4; 2.13 Koch, James 2.4; 2.13 Koch, Susanne N/A Koesser, Leland N/A Koesser, Karl 2.4; 2.13 Koivisto, Ellen 2.4; 2.13 Korexon, Susan 2.4; 2.13 Korexon, Jaeff N/A Korexon, Susan 2.4; 2.13 Kraft,	Kirkpatrick, Mary	2.4
Kite, Richard 24, 2.13 Kittb, Pat 2.4, 2.13 Kitts, Michele NA Kizer, David 2.8, 2.14, 11.5 Klass, David 2.4, 2.13 Klass, David 2.4, 2.13 Klein, Mark 2.4, 2.13 Kline, Regan 3.2, 3.6, 8.3 Klinkhamer, Trevor 12.2 Klipfel II, George 2.4, 2.13 Knickerbocker, Al N/A Knickerbocker, Al N/A Knickprocent, Carl 2.4, 2.13 Knorl, Carl 2.4, 2.13 Koch, James 2.4, 2.13 Koch, Susanne N/A Koesser, Leland N/A Komszen, Susan 2.4, 2.13 Koristo, Ellen 2.4, 2.13 Kortex, Karen 2.4 Kosec	Kirsh, Julie	N/A
Kittle, Pat $2.4; 2.13$ Kitts, Michele NA Kitzer, David $2.8; 2.14; 11.5$ Klass, David $2.4; 2.13$ Klass, David $2.4; 2.13$ Klass, David $2.4; 2.13$ Klass, David $2.4; 2.13$ Kline, Regan $3.2; 3.6; 8.3$ Klinkhamer, Trevor 12.2 Klipfell II, George $2.4; 2.13$ Knickerbocker, Al N/A Knickerbocker, Al N/A Knickorbocker, Al N/A Knoll, Carolyn $2.4; 2.13$ Knoll, Carolyn $2.4; 2.13$ Knoth, Dean $2.4; 2.13$ Koch, James $2.4; 2.13$ Koch, Joann $2.4; 2.13$ Koch, Joann $2.4; 2.13$ Kobeser, Leland N/A Koeser, Leland N/A Koeser, Jeland N/A Komisarof, Jeff NA Komera, Susan $2.4; 2.13$ Kortora, Wichele $2.4; 2.13$ Kortora, Michel $2.4; 2.13$ Kortora, Michele $2.4; 2.13$ Kraft, Tessa <	Kite, Richard	2.4; 2.13
Kitts, MicheleNAKizer, David $2,8; 2.14; 11.5$ Klass, David $2,4; 2.13$ Klass, David $2,4; 2.13$ Klass, David $2,4; 2.13$ Kleine, Regan $3,2; 3,6; 8.3$ Klinkhamer, Trevor 12.2 Klipfel II, George $2,4; 2.13$ Knickerbocker, AlNAKnickerbocker, AlNAKnight, RobertN/AKnoll, Carolyn $2,4; 2.13$ Kord, James $2,4; 2.13$ Koch, Jaan $2,4; 2.13$ Kouth, Dean $2,4; 2.13$ Koch, James $2,4; 2.13$ Koch, James $2,4; 2.13$ Koch, Jaan $2,4; 2.13$ Koch, Jaan $2,4; 2.13$ Koch, Jaan $2,4; 2.13$ Koch, Jaan $2,4; 2.13$ Koch, Joann $2,4; 2.13$ Koch, SusanneN/AKoesser, LelandN/AKoesser, LelandN/AKomisarof, JeffN/AKomzen, Susan $2,4; 2.13$ Koterba, Michael $2,4; 2.13$ Kraft, Tessa $2,4; 2.13$ Kraft, Tessa $2,4; 2.13$ Kraft, Kevin $3,2; 3,6$ Kramer, Gavin $2,4; 2.13$ Kra	Kittle, Pat	2.4; 2.13
Kizer, David 2.8; 2.14; 11.5 Klass, David 2.4; 2.13 Klein, Mark 2.4; 2.13 Klein, Mark 2.4; 2.13 Klink, Mark 2.4; 2.13 Klinkhamer, Trevor 12.2 Klipfel II, George 2.4; 2.13 Knickerbocker, Al NA Knickerbocker, Al NA Knorr, Carl 2.4; 2.13 Knorr, Carl 2.4; 2.13 Koch, James 2.4; 2.13 Koch, Jaanes 2.4; 2.13 Koch, Susanne NA Koesser, Leland N/A Koesser, Leland N/A Koresser, Gregg N/A Komsarof, Jeff N/A Korese, Nichael 2.4; 2.13 Koreto, Michael 2.4; 2.13 Koreto, Michael 2.4; 2.13 Koreto, Susan 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Kramer, Grant 2.4 Kramer, Grant<	Kitts, Michele	N/A
Klass, David 2.4 ; 2.13 Klass, David 2.4 ; 2.13 Klein, Mark 2.4 ; 2.13 Kline, Regan 32 ; 3.6 ; 8.3 Klinkhamer, Trevor 12.2 Klipfel II, George 2.4 ; 2.13 Knickerbocker, Al NA Knickerbocker, Al N/A Knickerbocker, Al N/A Knoll, Carolyn 2.4 ; 2.13 Knorr, Carl 2.4 ; 2.13 Kouth, Dean 2.4 ; 2.13 Koch, James 2.4 ; 2.13 Koch, James 2.4 ; 2.13 Koch, Jaanes 2.4 ; 2.13 Koch, Susanne N/A Koeser, Leland N/A Koeser, Leland N/A Koisson, Gregg N/A Konzen, Susan 2.4 ; 2.13 Koiterba, Michael 2.4 ; 2.13 Korec, Karen 2.4 Kase, Dawn 2.4 ; 2.13 Krait, Tessa 2.4 ; 2.13 Kramer, Eileen 2.4 ; 2.13 Kramer, Gavin 2.4 Kramer, Gavin 2.4 Kramer, Gavin	Kizer, David	2.8; 2.14; 11.5
Klass, David $2.4; 2.13$ Klein, Mark $2.4; 2.13$ Kline, Regan $3.2; 3.6; 8.3$ Klinkhamer, Trevor 12.2 Klipfel II, George $2.4; 2.13$ Knickerbocker, Al NA Knickerbocker, Al NA Knickerbocker, Al NA Knort, Carl $2.4; 2.13$ Kouth, Dean $2.4; 2.13$ Kouth, Dean $2.4; 2.13$ Koch, James $2.4; 2.13$ Koch, Jaanes $2.4; 2.13$ Koch, Jaanes $2.4; 2.13$ Koch, Susanne NA Koesser, Leland NA Koesser, Karl $2.4; 2.13$ Koivisto, Ellen $2.4; 2.13$ Koivisto, Ellen $2.4; 2.13$ Korez, Karen 2.4 Kosec, Dawn $2.4; 2.13$ Korez, Karen 2.4 Kosec, Dawn $2.4; 2.13$ Koterba, Michael $2.4; 2.13$ Kraft, Tessa $2.4; 2.13$ Kramer, Grant 2.4 Kramer, Grant 2.4 Kramere, Grant 2.4 <td>Klass, David</td> <td>2.4; 2.13</td>	Klass, David	2.4; 2.13
Klein, Mark $2.4; 2.13$ Klinkamer, Trevor 12.2 Klipfel II, George $2.4; 2.13$ Knickerbocker, Al N/A Kniss, Betty $2.4; 2.13$ Knickerbocker, Al N/A Kniss, Betty $2.4; 2.13$ Knoll, Carolyn $2.4; 2.13$ Knor, Carl $2.4; 2.13$ Kort, Carl $2.4; 2.13$ Koch, James $2.4; 2.13$ Koch, James $2.4; 2.13$ Koch, Jaanes $2.4; 2.13$ Koch, Joann $2.4; 2.13$ Koch, Susanne N/A Koesser, Leland N/A Koesser, Leland N/A Komisarof, Jeff N/A Komisarof, Jeff N/A Konzen, Susan $2.4; 2.13$ Koterba, Michael $2.4; 2.13$ Kraft, Tessa $2.4; 2.13$ Kraft, Tessa $2.4; 2.13$ Kraft, Tessa $2.4; 2.13$ Kramer, Gavin $2.4; 2.13$ Kramer, Grant 2.4 Kramer, Grant $2.4; 2.13$ Krause, Doug $2.4; 2.13$	Klass, David	2.4; 2.13
Kline, Regan 3.2; 3.6; 8.3 Klinkhamer, Trevor 12.2 Klipfel II, George 2.4; 2.13 Knickerbocker, Al N/A Kniess, Betty 2.4; 2.13 Knight, Robert N/A Knorr, Carl 2.4; 2.13 Knuth, Dean 2.4; 2.13 Koch, James 2.4; 2.13 Koch, James 2.4; 2.13 Koch, James 2.4; 2.13 Koch, James 2.4; 2.13 Koch, Jaanes 2.4; 2.13 Koch, Susanne N/A Koesser, Leland N/A Koesser, Leland N/A Koivisto, Ellen 2.4; 2.13 Kolessar, Gregg N/A Konsarof, Jeff N/A Koree, Karen 2.4 Karen 2.4 Krabel 2.4; 2.13 Koterba, Michael 2.4; 2.13 Krath, Tessa 2.4; 2.13 Krath, Tessa 2.4; 2.13 Krath, Tessa 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Gavin 2.4	Klein, Mark	2.4; 2.13
Klinkhamer, Trevor12.2Klinkhamer, Trevor2.4; 2.13Knickerbocker, AlN/AKnickerbocker, AlN/AKnight, RobertN/AKnorr, Carl2.4; 2.13Knuth, Dean2.4; 2.13Koch, James2.4; 2.13Koch, SusanneN/AKoesser, LelandN/AKoesser, Karl2.4; 2.13Koltsto, Ellen2.4; 2.13Korex, Karen2.4Korex, Karen2.4Koses, Dawn2.4; 2.13Koretha, Michael2.4; 2.13Krait, Tessa2.4; 2.13Krait, Kevin3.2; 3.6Kramer, Gavin2.4; 2.13Kramer, Gavin2.4; 2.13Kramer, Gavin2.4; 2.13Kramer, Gavin2.4; 2.13Kramer, Gavin2.4; 2.13Krause, William2.4; 2.13Krause, Roger3.2; 3.6; 8.3Krause, Roger3.2; 3.6; 8.3Krause, Roger3.2; 3.6; 8.3Krause, William2.4; 2.13Krause, Roger3.2; 3.6; 8.3Krause, Roger3.2; 3.	Kline, Regan	3.2; 3.6; 8.3
Klipfel II, George $2.4; 2.13$ Knickerbocker, AlN/AKniess, Betty $2.4; 2.13$ Knight, RobertN/AKnoll, Carolyn $2.4; 2.13$ Knouth, Dean $2.4; 2.13$ Koch, James $2.4; 2.13$ Koch, SusanneN/AKoesser, LelandN/AKoesser, Karl $2.4; 2.13$ Koivisto, Ellen $2.4; 2.13$ Koren, Susan $2.4; 2.13$ Korec, Karen 2.4 Korec, Karen 2.4 Korec, Karen $2.4; 2.13$ Koterba, Michael $2.4; 2.13$ Kraft, Tessa $2.4; 2.13$ Krant, Tessa $2.4; 2.13$ Kramer, Gavin $2.4; 2.13$ Kramer, Gavin $2.4; 2.13$ Kramer, Gavin $2.4; 2.13$ Kramer, Gavin $2.4; 2.13$ Krause, Doug $2.4; 2.13$ Krause, Roger $3.2; 3.6; 8.3$ Krause, William $2.4; 2.13$ Krause, William $2.4; 2.13$ Krause, Roger $3.2; 3.6; 8.3$ Krause, William $2.4; 2.13$	Klinkhamer, Trevor	12.2
Knickerbocker, Al N/A Kniess, Betty 2.4; 2.13 Knight, Robert N/A Knoll, Carolyn 2.4; 2.13 Knorr, Carl 2.4; 2.13 Knoth, Carolyn 2.4; 2.13 Knoth, Carolyn 2.4; 2.13 Knotr, Carl 2.4; 2.13 Koch, James 2.4; 2.13 Koch, Joann 2.4; 2.13 Koch, Joann 2.4; 2.13 Kock, Susanne N/A Koesser, Leland N/A Koesser, Leland N/A Koesser, Gregg N/A Kolessar, Gregg N/A Komisarof, Jeff N/A Koreze, Karen 2.4 Koterba, Michael 2.4; 2.13 Koterba, Michael 2.4; 2.13 Krath, Tessa 2.4; 2.13 Kramer, Karen 2.4 Kramer, Gavin 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Kramer, Gavin 2.4; 2.13 <t< td=""><td>Klipfel II, George</td><td>2.4; 2.13</td></t<>	Klipfel II, George	2.4; 2.13
Kniess, Betty $2.4; 2.13$ Knight, Robert N/A Knort, Carl $2.4; 2.13$ Knuth, Dean $2.4; 2.13$ Koch, James $2.4; 2.13$ Koch, James $2.4; 2.13$ Koch, James $2.4; 2.13$ Koch, Joann $2.4; 2.13$ Koch, Joann $2.4; 2.13$ Kock, Susane N/A Koesser, Leland N/A Koessel, Karl $2.4; 2.13$ Koivisto, Ellen $2.4; 2.13$ Koivisto, Ellen $2.4; 2.13$ Koissar, Gregg N/A Konzen, Susan $2.4; 2.13$ Korec, Karen 2.4 Kosec, Dawn $2.4; 2.13$ Koterba, Michael $2.4; 2.13$ Kraft, Tessa $2.4; 2.13$ Kramer, Fileen $2.4; 2.13$ Kramer, Gavin $2.4; 2.13$ Kramer, Grant $2.4;$	Knickerbocker, Al	N/A
Knight, Robert N/A Knoll, Carolyn $2.4; 2.13$ Knorr, Carl $2.4; 2.13$ Knuth, Dean $2.4; 2.13$ Koch, James $2.4; 2.13$ Koch, James $2.4; 2.13$ Koch, Joann $2.4; 2.13$ Koch, Susanne N/A Koeser, Leland N/A Koessel, Karl $2.4; 2.13$ Koivisto, Ellen $2.4; 2.13$ Koivisto, Ellen $2.4; 2.13$ Korex, Gregg N/A Konzen, Susan $2.4; 2.13$ Korec, Karen 2.4 Kosec, Dawn $2.4; 2.13$ Koterba, Michael $2.4; 2.13$ Kratt, Tessa $2.4; 2.13$ Kratt, Kevin $3.2; 3.6$ Kramer, Eileen $2.4; 2.13$ Kramer, Gavin $2.4; 2.13$ Krause, Doug $2.4; 2.13$ Krause, Roger $3.2; 3.6; 8.3$ Krause, Roger $3.2; 3.6; 8.3;$ Krause, Roger $3.2; 3.6; 8.3;$ Krause, Killiam $2.4; 2.13;$ Krell, Elinore $2.4; 2.13;$	Kniess, Betty	2.4; 2.13
Knoll, Carolyn 2.4; 2.13 Knorr, Carl 2.4; 2.13 Knorth, Dean 2.4; 2.13 Koch, James 2.4; 2.13 Koch, James 2.4; 2.13 Koch, Joann 2.4; 2.13 Koch, Susanne N/A Koeser, Leland N/A Koesser, Karl 2.4; 2.13 Koivisto, Ellen 2.4; 2.13 Konzen, Susan 2.4; 2.13 Korec, Karen 2.4 Koses, Leland N/A Konsisarof, Jeff N/A Konzen, Susan 2.4; 2.13 Korec, Karen 2.4 Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Krait, Kevin 3.2; 3.6 Kramer, Karen 2.4 Kramer, Gavin 2.4; 2.13 Krause, Doug 2.4; 2.13 Krause, N/A	Knight, Robert	N/A
Knorr, Carl 2.4; 2.13 Knuth, Dean 2.4; 2.13 Koch, James 2.4; 2.13 Koch, Joann 2.4; 2.13 Koch, Joann 2.4; 2.13 Koch, Susanne N/A Koeser, Leland N/A Koessel, Karl 2.4; 2.13 Koivisto, Ellen 2.4; 2.13 Koivisto, Ellen 2.4; 2.13 Korser, Gregg N/A Konzen, Susan 2.4; 2.13 Korec, Karen 2.4 Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Kratik, Kevin 3.2; 3.6 Kramer, Karen 2.4 Kramer, Gavin 2.4; 2.13 Krause, Moger 3.2; 3.6; 8.3 Krause, Roger 3.2; 3.6; 8.3 Krause, Roger 3.2; 3.6; 8.3 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13	Knoll, Carolyn	2.4; 2.13
Knuth, Dean 2.4; 2.13 Koch, James 2.4; 2.13 Koch, James 2.4; 2.13 Koch, Susanne N/A Koeser, Leland N/A Koessel, Karl 2.4; 2.13 Koivisto, Ellen 2.4; 2.13 Kolessar, Gregg N/A Komisarof, Jeff N/A Korzen, Susan 2.4; 2.13 Korzen, Susan 2.4; 2.13 Korec, Karen 2.4 Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Kramarz, Karen 2.4 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Krasnoff, Joshua N/A Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, Roger 3.2; 3.6; 8.3 Krause, Roger 3.2; 3.6; 8.3 Krehbiel, Robb 2.4; 2.13	Knorr, Carl	2.4; 2.13
Koch, James 2.4; 2.13 Koch, Joann 2.4; 2.13 Koch, Susanne N/A Koeser, Leland N/A Koeser, Leland N/A Koivisto, Ellen 2.4; 2.13 Kolessar, Gregg N/A Komisarof, Jeff N/A Konzen, Susan 2.4; 2.13 Korec, Karen 2.4 Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Kramer, Karen 2.4 Kramer, Gavin 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Krasnoff, Joshua N/A Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krause, William 2.4; 2.13 Krause, William 2.4; 2.13 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13	Knuth, Dean	2.4; 2.13
Koch, Joann 2.4; 2.13 Koch, Susanne N/A Koesser, Leland N/A Koessel, Karl 2.4; 2.13 Koivisto, Ellen 2.4; 2.13 Kolessar, Gregg N/A Komisarof, Jeff N/A Korec, Karen 2.4 Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kratt, Tessa 2.4; 2.13 Kratt, Kevin 3.2; 3.6 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Kreuse, William 2.4; 2.13 Krelheil, Robb 2.4; 2.13	Koch, James	2.4; 2.13
Koch, Susanne N/A Koeser, Leland N/A Koessel, Karl 2.4; 2.13 Koivisto, Ellen 2.4; 2.13 Kolessar, Gregg N/A Komisarof, Jeff N/A Konzen, Susan 2.4; 2.13 Korec, Karen 2.4 Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Krant, Kevin 3.2; 3.6 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Jaura 2.4; 2.13 Krasnoff, Joshua N/A Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krause, William 2.4; 2.13 Krenbiel, Robb 2.4; 2.13	Koch, Joann	2.4; 2.13
Koeser, LelandN/AKoesel, Karl $2.4; 2.13$ Koivisto, Ellen $2.4; 2.13$ Kolessar, GreggN/AKomisarof, JeffN/AKonzen, Susan $2.4; 2.13$ Korec, Karen 2.4 Kosec, Dawn $2.4; 2.13$ Kotreha, Michael $2.4; 2.13$ Kraft, Tessa $2.4; 2.13$ Kramarz, Karen 2.4 Kramer, Eileen $2.4; 2.13$ Kramer, Gavin $2.4; 2.13$ Kramer, Gavin $2.4; 2.13$ Kramer, JoshuaN/AKrause, Doug $2.4; 2.13$ Krause, Roger $3.2; 3.6; 8.3$ Krause, Roger $3.2; 3.6; 8.3$ Krause, William $2.4; 2.13$ Krause, Robb $2.4; 2.13$	Koch, Susanne	N/A
Koessel, Karl 2.4; 2.13 Koivisto, Ellen 2.4; 2.13 Kolessar, Gregg N/A Komisarof, Jeff N/A Konzen, Susan 2.4; 2.13 Korec, Karen 2.4 Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Krath, Tessa 2.4; 2.13 Krath, Tessa 2.4; 2.13 Krath, Kevin 3.2; 3.6 Kramarz, Karen 2.4 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Kramer, Grant 2.4 Krasnoff, Joshua N/A Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, Roger 3.2; 3.6; 8.3 Krause, Robb 2.4; 2.13 Krehbiel, Robb 2.4; 2.13	Koeser, Leland	N/A
Koivisto, Ellen 2.4; 2.13 Kolessar, Gregg N/A Komisarof, Jeff N/A Konzen, Susan 2.4; 2.13 Korec, Karen 2.4 Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Kranarz, Karen 2.4 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Krasoff, Joshua N/A Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krause, Robb 2.4; 2.13	Koessel, Karl	2.4; 2.13
Kolessar, Gregg N/A Komisarof, Jeff N/A Konzen, Susan 2.4; 2.13 Korec, Karen 2.4 Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Krain, Kevin 3.2; 3.6 Kramarz, Karen 2.4 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Koivisto, Ellen	2.4; 2.13
Komisarof, Jeff N/A Konzen, Susan 2.4; 2.13 Korec, Karen 2.4 Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Krai, Kevin 3.2; 3.6 Kramarz, Karen 2.4 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Kramer, Joshua N/A Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13	Kolessar, Gregg	N/A
Konzen, Susan 2.4; 2.13 Korec, Karen 2.4 Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Kramer, Karen 2.4 Kramer, Karen 2.4 Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Kramer, Grant 2.4 Kramer, Jaura 2.4; 2.13 Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Komisarof, Jeff	N/A
Korec, Karen 2.4 Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Krai, Kevin 3.2; 3.6 Kramarz, Karen 2.4 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Kramer, Jaura 2.4; 2.13 Krasnoff, Joshua N/A Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Konzen, Susan	2.4; 2.13
Kosec, Dawn 2.4; 2.13 Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Krai, Kevin 3.2; 3.6 Kramarz, Karen 2.4 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Kramer, Grant 2.4 Kramer, Laura 2.4; 2.13 Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Korec, Karen	2.4
Koterba, Michael 2.4; 2.13 Kraft, Tessa 2.4; 2.13 Krai, Kevin 3.2; 3.6 Kramarz, Karen 2.4 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Kramer, Laura 2.4; 2.13 Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Kreabel, Robb 2.4; 2.13	Kosec, Dawn	2.4; 2.13
Kraft, Tessa 2.4; 2.13 Krai, Kevin 3.2; 3.6 Kramarz, Karen 2.4 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Kramer, Laura 2.4; 2.13 Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Koterba, Michael	2.4; 2.13
Krai, Kevin 3.2; 3.6 Kramarz, Karen 2.4 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Kramer, Laura 2.4; 2.13 Krasnoff, Joshua N/A Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Kraft, Tessa	2.4; 2.13
Kramarz, Karen 2.4 Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Kramer, Laura 2.4; 2.13 Krasnoff, Joshua N/A Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Krai, Kevin	3.2; 3.6
Kramer, Eileen 2.4; 2.13 Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Kramer, Laura 2.4; 2.13 Krasnoff, Joshua N/A Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Kramarz, Karen	2.4
Kramer, Gavin 2.4; 2.13 Kramer, Grant 2.4 Kramer, Laura 2.4; 2.13 Krasnoff, Joshua N/A Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Kramer, Eileen	2.4; 2.13
Kramer, Grant2.4Kramer, Laura2.4; 2.13Krasnoff, JoshuaN/AKrause, Doug2.4; 2.13Krause, Roger3.2; 3.6; 8.3Krause, William2.4; 2.13Krehbiel, Robb2.4; 2.13Krell, Elinore2.4; 2.13	Kramer, Gavin	2.4; 2.13
Kramer, Laura2.4; 2.13Krasnoff, JoshuaN/AKrause, Doug2.4; 2.13Krause, Roger3.2; 3.6; 8.3Krause, William2.4; 2.13Krehbiel, Robb2.4; 2.13Krell, Elinore2.4; 2.13	Kramer, Grant	2.4
Krasnoff, Joshua N/A Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Kramer, Laura	2.4; 2.13
Krause, Doug 2.4; 2.13 Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Krasnoff, Joshua	N/A
Krause, Roger 3.2; 3.6; 8.3 Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Krause, Doug	2.4; 2.13
Krause, William 2.4; 2.13 Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Krause, Roger	3.2; 3.6; 8.3
Krehbiel, Robb 2.4; 2.13 Krell, Elinore 2.4; 2.13	Krause, William	2.4; 2.13
Krell, Elinore 2.4; 2.13	Krehbiel, Robb	2.4; 2.13
	Krell, Elinore	2.4; 2.13

	0.4.0.10
Krieger, Gena	2.4; 2.13
Kring, Juli	2.4; 2.13
Krone, Jeannene	2.4; 2.13
Krueger, David	2.4; 2.13
Krumpos, Steven	3.2; 3.6; 8.3
Krupa, Dave	N/A
Kubik, Cathy	N/A
Kuciej, Walter	2.4; 2.13
Kunn, Kerry	2.4; 2.13
Kulka, Wally	3.2; 3.6; 8.3
Kuppler, Curtis	2.4; 2.13
Kusick, Paul	N/A
Kust, Melina	2.8
Kuter, Ann	N/A
Kuykendall, Gregory	3.2; 3.6
La Falce, Stephen	3.2; 3.6; 8.3
La Falce, Stephen (Trout Unlimited - Grand Canyon	2.5; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
Chapter)	
La Falce, Stephen (Trout Unlimited - Arizona Council)	2.5; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
Labasco, Christopher	2.4; 2.13
LaBranche, Laura	3.2; 3.6; 8.3
Lack, Phil	2.4
Lackey, Mercedes	3.6
LaCognata, Dale	2.4
Lacy, Sharon	2.4; 2.13
Ladd, Larry& Karen	2.4; 2.13
Ladimer, Martin	2.4; 2.13
Lafleur, Steven	2.4; 2.13
LaFour, Liz	N/A
Lago, Don	N/A
Lagonegro, Sean	2.4
Laieski, Caleb	2.4; 2.13
Lair, Jennifer	2.4
Lam, Ofelia	2.4; 2.13
LaMaack, Larry (Wyoming Municipal Power Agency)	2.6; 2.11; 2.13
Lamb, Patricia	2.4
Lambert, Howard	N/A
Lamiquiz, David	N/A
Lamons, Kristina	2.4; 2.13
Land, Martha	2.4; 2.13
Landry, Michael	3.2; 3.6; 8.3
Landry, Michael	3.2; 3.6; 8.3
Lane, Debra	2.4; 2.13
Lane, Troy	2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
Lanfranchi, LJ	2.4; 2.13
Lange, Elena	N/A
Lange, Marlena	2.4; 2.13
Lanka, Mike	2.4; 2.13
Lantry, Gavin	3.2; 3.6; 8.3
Lanus, Howard	N/A

Lanzetta, Dante	2.4; 2.13
LaPlaca, Lisa	2.4; 2.13
LaRoche, David	2.4
Larsen, Lance	3.2; 3.6; 8.3
Lash, Cal	2.8
Lashaway, Lisa	2.4
Lasley, Barbara	2.4; 2.13
Latona, Jaron	2.4; 2.13
Latta, George	2.1; 2.2; 2.8; 2.14; 13.1
Lausmann, Vance	2.4; 2.13
LaVonne, Nadine	N/A
Lawler, Lynn	N/A
Lawrence, Christopher	N/A
Lawrence, Pat	2.4
Lawson, Barbara	N/A
Layfield, Elizabeth	2.4; 2.13
Lazor, John	2.4; 2.13
Lea, Susan	2.4
Leal-McBride, Odilia	N/A
Learmann, Prisca	2.4; 2.13
LeClair, Peg	2.4; 2.13
Ledger, Patrick (Arizona Electric Power Cooperative,	2.1
Inc.)	
Lee, Madeleine	N/A
Lee, Miree	2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
Lee, Robert	2.4; 2.13
Lee, Virginia	2.4; 2.13
Leeuw, Lyn	N/A
Lehmann, Andre	2.1; 2.2; 2.8; 2.14; 13.1
Leichter, Lew	3.2; 3.6; 8.3
Leifker, M.	3.2; 3.6; 8.3
Leland, Lora	2.4; 2.13
Lemkuil, Rita	N/A
Lenchner, Nicholas	2.4; 2.13
Lenz, Andrew	2.4; 2.13
Leo, Carlos	2.4
Lesser, Rob	2.4; 2.13
Leve, Kristin	2.4; 2.13
Levesque, Paul	2.4; 2.13
Levin, Jon	2.4; 2.13
Levine, Lynn	N/A
Levitus, Walt	3.2; 3.6; 8.3
Levy, Linda	2.4
Lewis, O	2.4; 2.13
Lewis, Rita	2.4; 2.13
Lewis, Sherman	2.4; 2.13
Lewis, Tom	N/A
Liaudat, Claudia	N/A
Lichtenberger, Wayne	3.2; 3.6; 8.3
Liebeskind, Al	2.4

Lieder, Cecilia	N/A
Liesche, Kenneth	2.4; 2.13
Liggio, Eleanor	2.4; 2.13
Lilling, Glenda	2.4; 2.13
Lindsay, Brenda	N/A
Lindsay, Leslie	2.4; 2.13
Linna, Patricia	N/A
Lipsey, Joseph	2.4; 2.13
Lipsky, carol	2.4
Lish, Jeannine	2.4; 2.13
Liske, Patricia Ann	2.4
Liss, Mary	2.4; 2.13
Little, Dennis, Jr.	N/A
Little. Rov	3.2: 3.6: 8.3
Littlefield. Joan	N/A
Litwin, Linda	N/A
Livingston Deborah	2.4.2.13
Llewellyn Suzanne	N/A
Lo Jen	N/A
LoCicero Robert	2.4
Lockett Jennifer	2 4. 2 13
Locklear Clyde	2 1 2 2 2 8 2 14 13 1
Logan Newton	3 7. 3 6. 8 3
Lojo Rosemary	N/A
Lolli Mark	2 4. 2 13
Long Dave	2.π, 2.15 N/A
Long, Ernest	$1 \sqrt{2}$
Long, Ludith	2.1, 2.2, 2.0, 2.14, 13.1 N/A
Long, Juditi	N/A 2 4
Looliis, Kea Allii Loolii Ed (The Wildlife Trust)	2.4
Loosi, Ed (The Whante Trust)	2.4
Loper, Digite	2.4, 2.15
Lotentz, Marcel	N/A 2.4.2.12
Lotak, Justin	2.4, 2.13
Love, Mary	2.4; 2.13
Lowans, Jennifer	2.4; 2.13
Lowery, Joanne	2.4; 2.13
Lucas, Mary	2.4; 2.13
Luddon, Barbara	2.4; 2.13
Lueck, Donna	2.4; 2.13
Lundgren, Scott	2.4; 2.13
Lupenko, Andy	2.4; 2.13
Lupori, Stacy	2.4; 2.13
Lusby-Denham, Anne	2.4
Lusche, Jim	3.2; 3.6; 8.3
Luttich, Stu	2.4; 2.13
Luzier, Maresa	N/A
Lynch, Doris	2.4
Lynch, Frances	2.4; 2.13
Lynch, John	N/A

Lynch, Robert (Irrigation and Electrical Districts Association of Arizona)	2.3; 2.6; 2.10; 2.11; 2.21; 8.10; 11.1; 11.2; 11.4; 12.10
Lyon R Terry	N/Δ
Lyons Kathi	2 4. 2 13
M Henry	2.4, 2.13
MacAlnine Barbara	2.4, 2.13
Macan Edward	2.4, 2.13
MacArthur, June	2.4, 2.15 N/A
MacDonald John	2 A
Maciol Maria	2.4
Machen Leslie	2.4, 2.15
MacKay, Leshe MacKanzia, Michalla	2.4, 2.15
Machin Tom (Coconing Sportsmon)	2.4, 2.15
MacLood Dianna	2.4
MacLeou, Dialilla MacDaith Dannia	2.4, 2.13
MacRahlin, Dolline MacRahlia, Tadd	2.4, 2.15
Maddon Kovin	5.0 2 2 : 2 <i>C</i>
Magallan Katia	5.2, 5.0 N/A
Magaa W I	N/A 2 2: 2 6: 9 2
Magee, W.L.	5.2, 5.0, 6.5 N/A
Mallott, Darbara	N/A
Mallow Jerry	$1 \times A$
Malone Michael	2.1, 2.2, 2.6, 2.14, 13.1
Maloney Liam	2.4, 2.15 Ν/Δ
Malven Tania	2 A
Manderscheid B	13.1
Mang I	$2 4 \cdot 2 13$
Mangrum Carl	2.4, 2.15
Mangum Laurie (St. George Energy Services)	2 4. 2 10
Mangus, Tracey	2.4, 2.10
Manka Joann	2.4, 2.15
Mannering Natalie	2.7
Mannesa Howard	2.4, 2.13
Mansfield Tim	12.4
Marano Gina	12.4
Marasco, Summer	2.4, 2.13
Marchello, Linda	2.4, 2.15
Marchini, David and Julia (First United Methodist	2.7
Church)	2.7
Marcus Marilyn	2 4. 2 13
Margulis Elise	2.τ, 2.15 N/Δ
Mark Robert	2 4. 2 13
Markham William	2.τ, 2.15 N/Δ
Marks IB	N/A
Marquis Sharon	N/A
Marrs Cynthia	2 4. 2 13
Marsden Michael	3 2. 3, 6. 8 3
Marsett Robert	N/A
Marsh David	3 2 3 6 8 3
Marsh Sherry	N/A
	* v * *

Marshall, Kobert N/A Marshall, Kobert N/A Martin, Chase 2.4 Martin, Gerry N/A Martin, Joanna N/A Martin, Joanna N/A Martin, Joanna N/A Martin, Joanna N/A Martinez, Drittany 2.4; 2.13 Martinez, Joittany 2.4; 2.13 Martinez, Jone N/A Martinez, Jone N/A Martinez, Presence 2.4; 2.13 Martinez, Ora Marek (The Navajo Nation) 4.2; 7.6; 8.1; 10.2; 10.7; 10.8 Martinez, Ora Marek (The Navajo Nation) 4.2; 7.6; 8.1; 10.2; 10.7; 10.8 Martinez, Ora Marek (The Navajo Nation) 4.2; 7.6; 8.1; 10.2; 10.7; 10.8 Martinez, Ora Marek (The Navajo Nation) 4.2; 7.13 Martinez, Gride N/A Martinez, Gride N/A Marskall, Kola 2.4; 2.13 Martinez, Grody N/A Massey, Carolyn 2.4; 2.13 Matski, Farny 2.4; 2.13 Matski, Janny 2.4; 2.13 Matski, Marty 2.12 </th <th></th> <th></th>		
Marshall, RobertN/AMartin, Chase2.4Martin, Jorew2.4, 2.13Martin, GerryN/AMartin, JoannaN/AMartin, JoannaN/AMartin, Michele2.2, 3.6, 8.3Martin, Michele2.4, 2.13Martinez, Brittany2.4, 2.13Martinez, Ora Marek (The Navajo Nation)4.2, 7.6, 8.1; 10.2; 10.7; 10.8Martinez, Cra Marek (The Navajo Nation)4.2, 7.6, 8.1; 10.2; 10.7; 10.8Martinez, Robecca2.4, 2.13Martinez, Rebecca2.4, 2.13Martinez, Kickie2.4, 2.13Martinez, RobiceN/AMaschke, NicoleN/AMassey, Carolyn2.4, 2.13Martin, Francis2.4, 2.13Matelski, Lauren2.4, 2.13Matelski, Lauren2.4, 2.13Mattis, Jeanne2.4, 2.13Mattis, Jeanne2.4, 2.13Mattis, Jardan2.4, 2.13Mattis, Jardan2.4, 2.13Mattis, Jeanne2.4, 2.13Mattis, Jeanne2.4, 2.13Mattis, Jeanne2.4, 2.13Mattis, Jeanne2.4, 2.13Mattis, Jeanne2.4, 2.13Mattis, HenryN/AMattis, Barbara2.4, 2.13Mattis, Brubara2.4, 2.13Mattis, HenryN/AMattis, Brubara2.4, 2.13Mattis, Brubara2.4, 2.13Mattis, Brubara2.4, 2.13Mattis, Brubara2.4, 2.13Mattis, Brubara2.4, 2.13Mattis, Brubara2.4, 2.13Mattist	Marshall, Linda	2.4
Martin, Chase2.4Martin, Torew 2.4 , 2.13Martin, GerryN/AMartin, JoannaN/AMartin, JoannaN/AMartin, JoannaN/AMartin, Johna 2.2 , 3.6, 8.3Martin, Michele 2.4 , 2.13Martinez, JanieN/AMartinez, JanieN/AMartinez, JanieN/AMartinez, Grabarek (The Navajo Nation) 4.2 , 7.6, 8.1; 10.2; 10.7; 10.8Martinez, Rebecca 2.4 , 2.13Martinez, Rebecca 2.4 , 2.13Marvick, Vicki 2.4 , 2.13Marvin, CindyN/AMaseda-Gile, Sheila 2.4 , 2.13Maseda-Gile, Sheila 2.4 , 2.13Mateika, Harry 2.4 , 2.13Mateika, Harry 2.4 , 2.13Mateika, Harry 2.4 , 2.13Mateika, Harry 2.4 , 2.13Mateika, Barbara 2.4 , 2.13Mateika, Barbara 2.4 , 2.13Mateika, Barbara 2.4 , 2.13Mattis, BernyN/AMatusk, Barbara 2.4 , 2.13Matusk, Barbara 2.4 , 2.13Matusk, Barbara 2.4 , 2.13Matusk, Barbara 2.4 , 2.13Matusk, Harry 2.4 Matusk, Barbara 2.4 , 2.13Matusk, Barbara 2.4 , 2.13Matusk, Barbara 2.4 , 2.13Matusk, Barbara 2.4 , 2.13Matusk, Harbara 2.4 , 2.13McAlister, Robert 2.4 Markand, Macanteiter 2.4 , 2.13McCality, Lesse Louis HenryN/AM	Marshall, Robert	N/A
Martin, Drew $2.4; 2.13$ Martin, GerryNAMartin, JoannaNAMartin, JoennaNAMartin, Michele $2.2; 36; 8.3$ Martin, JoennaNAMartinez, Brittany $2.4; 2.13$ Martinez, JanieNAMartinez, JanieNAMartinez, Gra Marek (The Navajo Nation) $4.2; 7.6; 8.1; 10.2; 10.7; 10.8$ Martinez, Cara Marek (The Navajo Nation) $4.2; 7.6; 8.1; 10.2; 10.7; 10.8$ Martinez, Rebecca $2.4; 2.13$ Marvick, Vicki $2.4; 2.13$ Marvick, Vicki $2.4; 2.13$ Marvin, CindyNAMaschke, NicoleN/AMaseda-Gille, Sheila $2.4; 2.13$ Mastri, Francis $2.4; 2.13$ Matsir, Francis $2.4; 2.13$ Matelski, Lauren $2.4; 2.13$ Mathis, I canne $2.4; 2.13$ Mathis, I canne $2.4; 2.13$ Mathis, Marty 2.12 Matusi, Marty 2.12 Matusi, Barbara $2.4; 2.13$ Mattis, HenryN/AMatusik, Barbara $2.4; 2.13$ Matusik, Barbara $2.4; 2.13$ Mayand, LindaN/AMazuca, R $2.4; 2.13$ Machalister, Kevin $2.4; 2.13$ MeAlion, Hose $2.4; 2.13$ MeAlister, Kevin $2.4; 2.13$ McAlister, Kevin $2.4; 2.13$ McAlister, Kevin $2.4; 2.13$ McCallow, John (State of Colorado) $2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4McConnel, SuzanneN/A$	Martin, Chase	2.4
Martin, GerryN/AMartin, JoenaN/AMartin, Joel $32, 3.6, 8.3$ Martine, Janie $2.4, 2.13$ Martinez, JanieN/AMartinez, JanieN/AMartinez, Cro Marek (The Navajo Nation) $42, 7.6, 8.1, 10.2, 10.7, 10.8$ Martinez, Cro Marek (The Navajo Nation) $42, 7.6, 8.1, 10.2, 10.7, 10.8$ Martinez, Cro Marek (The Navajo Nation) $42, 7.6, 8.1, 10.2, 10.7, 10.8$ Martinez, Cro Marek (The Navajo Nation) $42, 7.6, 8.1, 10.2, 10.7, 10.8$ Martinez, Cro Marek (The Navajo Nation) $42, 7.6, 8.1, 10.2, 10.7, 10.8$ Martinez, Cro Marek (The Navajo Nation) $42, 7.6, 8.1, 10.2, 10.7, 10.8$ Martin, Denise $2.4, 2.13$ Martin, CindyN/AMaseda-Gille, Sheila $2.4, 2.13$ Maseda-Gille, Sheila $2.4, 2.13$ Mastai, Francis $2.4, 2.13$ Matelski, Lauren $2.4, 2.13$ Matelski, Leanne $2.4, 2.13$ Matsui, Vicky $2.4, 2.13$ Matsui, Sarbara $2.4, 2.13$ Matusi, Barbara $2.4, 2.13$ Matusi, Barbara $2.4, 2.13$ Matusi, Barbara $2.4, 2.13$ Mayarad, ItindaN/AMazzuca, R 2.4 McAlister, Revin $2.4, 2.13$ McCarthy, Jose (Sub Henry)N/AMcCarthy, Jose (Sub Henry)N/AMcCarthy, LindaN/AMcCarthy, Jose (Sub Henry)N/AMcCarthy, Jose (Sub Henry)N/AMcCarthy, Jose (Sub Henry)N/AMcCarthy, Jose (Sub Henry)N/AMcC	Martin, Drew	2.4; 2.13
Martin, Joanna N/A Martin, Joel 32; 36; 8:3 Martin, Michele 2.4; 2:13 Martinez, Brittany 2.4; 2:13 Martinez, Ravek (The Navajo Nation) 42; 7:6; 8:1; 10.2; 10.7; 10.8 Martinez, Rebecca 2.4; 2:13 Martinez, Rebecca 2.4; 2:13 Marvin, Cindy N/A Marsek, Nicole N/A Masseh, Carolyn 2.4; 2:13 Marvin, Cindy N/A Massey, Carolyn 2.4; 2:13 Matsin, Jaconson 2.4; 2:13 Matsin, Jaconson 2.4; 2:13 Massey, Carolyn 2.4; 2:13 Matsin, Laren 2.4; 2:13 Matsin, Jaconson 2.4; 2:13 Matsin, Jaconson 2.4; 2:13 Matsin, Marty 2.12 Matsin, Vicky 2.4; 2:13 Mattis, Henry N/A Matusik, Barbara 2.4; 2:13 Matusik, Barbara 2.4; 2:13 Matasik, Barbara 2.4; 2:13 Matasik, Barbara 2.4; 2:13 MacAdoo, Hosea N/A	Martin, Gerry	N/A
Martin, Joel $32, 36, 8.3$ Martin, Joichel $24, 2.13$ Martinez, Brittany $24, 2.13$ Martinez, Janie N/A Martinez, Cro Marck (The Navajo Nation) $42, 76, 81, 10.2; 10.7; 10.8$ Martinez, Cro Marck (The Navajo Nation) $42, 76, 81, 10.2; 10.7; 10.8$ Martin, Denise $24, 2.13$ Martinez, Cro Marck (The Navajo Nation) $42, 76, 81, 10.2; 10.7; 10.8$ Martine, Vicki $24, 2.13$ Martine, Nicole N/A Mascha-Cille, Sheila $24, 2.13$ Mastri, Francis $24, 2.13$ Mateika, Lauren $24, 2.13$ Mateika, Lauren $24, 2.13$ Mathis, Marty 2.12 Matsi, Marty 2.12 Matsi, Marty $2.4, 2.13$ Mathis, Leane $2.4, 2.13$ Mathis, Marty 2.12 Matsi, Barbara $2.4, 2.13$ Matis, Barbara $2.4, 2.13$ Matusi, Barbara $2.4, 2.13$ Matusi, Barbara $2.4, 2.13$ Matusi, Barbara $2.4, 2.13$ Matusi, Barbara $2.4, 2.13$ Maruca,	Martin, Joanna	N/A
Martine, Michele 24, 2.13 Martinez, Brittany 24, 2.13 Martinez, Janie N/A Martinez, Cra Marek (The Navajo Nation) 42, 7.6, 8.1; 10.2; 10.7; 10.8 Martinez, Rebecca 24, 2.13 Martinez, Rebecca 24, 2.13 Martinez, Rebecca 24, 2.13 Martinez, Ora Marek (The Navajo Nation) 42, 2.13 Martinez, Rebecca 24, 2.13 Martinez, Ora Marek (The Navajo Nation) 42, 2.13 Martinez, Rebecca 24, 2.13 Martinez, Gridy N/A Maseder, Citoly N/A Maseder, Citols, Steila 24, 2.13 Mateisk, Harry 24, 2.13 Mateisk, Lauren 24, 2.13 Mateisk, Lauren 24, 2.13 Mateisk, Barbara 24, 2.13 Mattis, Berny N/A Matsik, Barbara 24, 2.13 Matusik, Barbara 24, 2.13 Mayanard, Bruce 24 Mayanard, Bruce 24 Mayanard, Stree 24 Mayanard, Stree 24 Mayanard, Stree 24 Mayanard, Stree	Martin, Joel	3.2; 3.6; 8.3
Martinez, Brittany 24; 2.13 Martinez, Janie NA Martinez, Ora Marek (The Navajo Nation) 4.2; 7.6; 8.1; 10.2; 10.7; 10.8 Martini, Denise 2.4; 2.13 Marvick, Vicki 2.4; 2.13 Marvick, Vicki 2.4; 2.13 Marvin, Cindy NA Maseke, Nicole NA Maseda-Gille, Sheila 2.4; 2.13 Mastri, Francis 2.4; 2.13 Mattist, Francis 2.4; 2.13 Mattist, Learne 2.4; 2.13 Mattist, Learne 2.4; 2.13 Mattist, Learne 2.4; 2.13 Mattist, Bearba 2.4; 2.13 Matustis, Barbara 2.4; 2.13 Matustis, Barbara 2.4; 2.13 Matustis, Bearbara 2.4; 2.13 Maynard, Linda NA McAlister, Robert 2.4 McAlister, Robert 2.4 McAlister, Robert 2.4 McAlister, Robert 2.4 <td>Martin, Michele</td> <td>2.4; 2.13</td>	Martin, Michele	2.4; 2.13
Martinez, Janie N/A Martinez, Ora Marek (The Navajo Nation) 42; 7.6; 8.1; 10.2; 10.7; 10.8 Martinez, Rebecca 2.4; 2.13 Martinez, Ora Marek (The Navajo Nation) 2.4; 2.13 Marvin, Cindy NA Maseda-Gille, Sheila 2.4; 2.13 Maseda-Gille, Sheila 2.4; 2.13 Masey, Carolyn 2.4; 2.13 Mateika, Harry 2.4; 2.13 Mateika, Harry 2.4; 2.13 Mateika, Harry 2.4; 2.13 Mateika, Harry 2.4; 2.13 Mateiski, Lauren 2.4; 2.13 Mateiski, Lauren 2.4; 2.13 Mateiski, Lauren 2.4; 2.13 Mateiski, Barbara 2.4; 2.13 Matusi, Barbara 2.4; 2.13 Matusi, Barbara 2.4; 2.13 Maynard, Bruce 2.4 Maynard, Linda N/A Mazzuca, R 2.4 McAlister, Robert 2.4 McAllister, Robert 2.4 McAllister, Robert 2.4 McAllister, Robert 2.4 McClow, John (State	Martinez, Brittany	2.4; 2.13
Martinez, Ora Marek (The Navajo Nation) 42; 7.6; 8.1; 10.2; 10.7; 10.8 Martini, Denise 2.4; 2.13 Marvin, Cindy NA Masche, Nicole N/A Massey, Carolyn 2.4; 2.13 Mastri, Francis 2.4; 2.13 Mastri, Francis 2.4; 2.13 Mastri, Francis 2.4; 2.13 Matelski, Lauren 2.4; 2.13 Mattisi, Lauren 2.4; 2.13 Mattisi, Vicky 2.4; 2.13 Mattisi, Vicky 2.4; 2.13 Mattisi, Lauren 2.4; 2.13 Mattisi, Vicky 2.4; 2.13 Mattisi, Narty 2.12 Mattisi, Weiky 2.4; 2.13 Mattisi, Barbara 2.4; 2.13 Mattisi, Barbara 2.4; 2.13 Matusik, Barbara 2.4; 2.13 Maynard, Erude 2.4 Maynard, Linda N/A Mazuca, R 2.4 McAloo, Hosea N/A McAllister, Robert 2.4 McAllister, Robert 2.4 McAllister, Robert 2.4 McCluw, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6	Martinez, Janie	N/A
Martinez, Rebecca $2.4; 2.13$ Martini, Denise $2.4; 2.13$ Marvick, Vicki $2.4; 2.13$ Marvick, Vicki $2.4; 2.13$ Marsche, NicoleN/AMascha-Gille, Sheila $2.4; 2.13$ Mastri, Francis $2.4; 2.13$ Mattin, Francis $2.4; 2.13$ Mattin, Francis $2.4; 2.13$ Mattin, Lauren $2.4; 2.13$ Mattin, Lauren $2.4; 2.13$ Mattin, Lauren $2.4; 2.13$ Mattin, John 2.12 Matsin, Narty 2.12 Matsin, Marty 2.12 Matsin, Barbara $2.4; 2.13$ Mattisk, Barbara $2.4; 2.13$ Maynard, Bruce 2.4 Maynard, LindaN/AMazzuca, R 2.4 MacAdoo, HoseaN/AMcAdoo, HoseaN/AMcAllister, Robert $2.4; 2.13$ McAllister, Robert $2.4; 2.13$ McClow, John (State of Colorado) $2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8:7; 8.8; 8.12; 10.8; 12.6; 13.4McClung, JudyN/AMcCornell, David3.2; 3.6; 8.3McCornell, David3.2; 3.6; 8.3McCornell, David2.4; 2.13McCornell, David3.2; 3.6; 8.3McCornell, David3.2; 3.6; 8.3McCornell, David3.2; 3.6; 8.3McCornell, David3.2; 3.6; 8.3McCornell, David3.6McCulloch, Arch2.4McCulloch, SamuelN/AMcCulloch, SamuelN/A$	Martinez, Ora Marek (The Navajo Nation)	4.2; 7.6; 8.1; 10.2; 10.7; 10.8
Martini, Denise 2.4; 2.13 Marvin, Cindy NA Maschke, Nicole NA Massey, Carolyn 2.4; 2.13 Mastri, Francis 2.4; 2.13 Mastri, Francis 2.4; 2.13 Matelski, Lauren 2.4; 2.13 Mattik, Lauren 2.4; 2.13 Mattik, Lauren 2.4; 2.13 Mattik, Lauren 2.4; 2.13 Mattik, Uaren 2.4; 2.13 Mattik, Uaren 2.4; 2.13 Mattik, Uaren 2.4; 2.13 Mattik, Uaren 2.4; 2.13 Mattik, Barbara 2.4; 2.13 Matusik, Barbara 2.4; 2.13 Matusik, Barbara 2.4; 2.13 Matusik, Barbara 2.4; 2.13 Mazuca, R 2.4 Mazuca, R 2.4 McAdoo, Hosea NA McAlister, Robert 2.4 McAlister, Robert 2.4 McAlpine, Emily 2.4 McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8.8.12; 10.8; 12.6; 13.4 McCloung, Judy NA McCornnel, David 3.2; 3.6; 8.3 </td <td>Martinez, Rebecca</td> <td>2.4; 2.13</td>	Martinez, Rebecca	2.4; 2.13
Marvick, Vicki 2.4; 2.13 Marvin, Cindy N/A Maschke, Nicole N/A Maseda-Gille, Sheila 2.4; 2.13 Mastri, Francis 2.4; 2.13 Mastri, Francis 2.4; 2.13 Matsik, Harry 2.4; 2.13 Matelski, Lauren 2.4; 2.13 Mathis, Leanne 2.4; 2.13 Mathis, Marty 2.12 Matsui, Vicky 2.4; 2.13 Matusi, Marty 2.12 Matusi, Barbara 2.4; 2.13 Matusi, Barbara 2.4; 2.13 Matusi, Barbara 2.4; 2.13 Maynard, Bruce 2.4 Maynard, Bruce 2.4 Maynard, Bruce 2.4 Mazca, R 2.4 McAloo, Hosea N/A McAlister, Revin 2.4; 2.13 McAlister, Robert 2.4 McAlpine, Emily 2.4 McAlpine, Emily 2.4; 2.13 McCarty, Jesse Louis Henry N/A McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12;	Martini, Denise	2.4; 2.13
Marvin, Cindy N/A Masche, Nicole N/A Mascda-Gille, Sheila 2.4; 2.13 Massey, Carolyn 2.4; 2.13 Mastri, Francis 2.4; 2.13 Matejka, Harry 2.4; 2.13 Matejka, Harry 2.4; 2.13 Mateljka, Lauren 2.4; 2.13 Mathis, Leanne 2.4; 2.13 Mattis, Marty 2.12 Matsui, Vicky 2.4; 2.13 Mattis, Marty 2.12 Mattis, Henry N/A Matusik, Barbara 2.4; 2.13 Maynard, Bruce 2.4 Maynard, Linda N/A Mazzuca, R 2.4 McAdoo, Hosea N/A McAlister, Robert 2.4 McAlpine, Emily 2.4 McAlpine, Emily 2.4 McClarty, Jesse Louis Henry N/A McClardy, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McCornell, Suzanne N/A McCornell, Suzanne N/	Marvick, Vicki	2.4; 2.13
Maschke, Nicole N/A Maseeda-Gille, Sheila 2.4; 2.13 Massey, Carolyn 2.4; 2.13 Mastri, Francis 2.4; 2.13 Matelski, Lauren 2.4; 2.13 Mathis, Leanne 2.4; 2.13 Mathis, Leanne 2.4; 2.13 Mathis, Leanne 2.4; 2.13 Mathis, Leanne 2.4; 2.13 Matsou, Vicky 2.4; 2.13 Mattes, Dale 2.4; 2.13 Mattes, Dale 2.4; 2.13 Matusi, Renry N/A Matusik, Barbara 2.4; 2.13 Maynard, Bruce 2.4 Maynard, Linda NA Mazzuca, R 2.4 McAloo, Hosea N/A McAlister, Roud 2.4; 2.13 McAlister, Robert 2.4 McAlister, Robert 2.4 McAlister, Robert 2.4 McCarty, Jesse Louis Henry N/A McCung, Judy N/A McClung, Judy N/A McCornell, David 3.2; 3.6; 8.3 McCornell, David 3.2; 3	Marvin, Cindy	N/A
Maseda-Gille, Sheila 24; 2.13 Mastri, Francis 24; 2.13 Matti, Francis 24; 2.13 Matejka, Harry 24; 2.13 Matelski, Lauren 24; 2.13 Mathis, Leanne 24; 2.13 Mattis, Marty 2.12 Matsui, Vicky 24; 2.13 Mattis, Marty 2.12 Matsui, Vicky 24; 2.13 Mattis, Barbara 2.4; 2.13 Matusi, Barbara 2.4; 2.13 Maynard, Bruce 2.4 Maynard, Linda N/A Mazuca, R 2.4 McAlister, Kevin 2.4; 2.13 McAllister, Robert 2.4 McAllister, Robert 2.4 McAllister, Robert 2.4 McAllister, Robert 2.4 McAllister, Nacy 2.4; 2.13 McCarty, Jesse Louis Henry N/A McClung, Judy N/A McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McCornell, David 3.2; 3.6; 8.3 McCornell, Suzanne <td< td=""><td>Maschke, Nicole</td><td>N/A</td></td<>	Maschke, Nicole	N/A
Massey, Carolyn 2.4; 2.13 Mastri, Francis 2.4; 2.13 Mateika, Harry 2.4; 2.13 Mateika, Lauren 2.4; 2.13 Mathis, Leanne 2.4; 2.13 Mathis, Leanne 2.4; 2.13 Mathis, Karty 2.12 Mathis, Vicky 2.4; 2.13 Mattis, Vicky 2.4; 2.13 Mattis, Henry N/A Matusik, Barbara 2.4; 2.13 Mazuca, R 2.4 McAdoo, Hosea N/A McAlister, Robert 2.4; 2.13 McAllister, Bobert 2.4 McSinde, Nancy 2.4; 2.13 McCarthy, Shirley 2.4; 2.13 McCarthy, Shirley 2.4; 2.13 McCluw, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8, 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McConnell	Maseda-Gille, Sheila	2.4; 2.13
Mastri, Francis 2.4; 2.13 Mateijka, Harry 2.4; 2.13 Mateijka, Lauren 2.4; 2.13 Mathis, Leanne 2.4; 2.13 Mathis, Leanne 2.4; 2.13 Mathis, Leanne 2.4; 2.13 Mathis, Marty 2.12 Matsui, Vicky 2.4; 2.13 Mattis, Henry N/A Matusik, Barbara 2.4; 2.13 Maynard, Bruce 2.4 Maynard, Bruce 2.4 Maynard, Bruce 2.4 Mazuca, R 2.4 McAdoo, Hosea N/A McAlister, Robert 2.4 McAlister, Robert 2.4 McAlpine, Emily 2.4 McAlpine, Sarey 2.4; 2.13 McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McConnell, David 3.2; 3.6; 8.3 McConnell, David 3.2; 3.6; 8.3 McConnell, David 3.2; 3.6; 8.3 McConnell, David 2.4; 2.13 McConnell, Suzanne N/A McCulloch, Arch 2.4 </td <td>Massey, Carolyn</td> <td>2.4; 2.13</td>	Massey, Carolyn	2.4; 2.13
Matejka, Harry $2.4; 2.13$ Matelski, Lauren $2.4; 2.13$ Mathis, Leanne $2.4; 2.13$ Mathis, Marty 2.12 Matsui, Vicky $2.4; 2.13$ Mattis, Marty $2.4; 2.13$ Mattis, HenryN/AMatusik, Barbara $2.4; 2.13$ Maynard, Bruce 2.4 Mazuca, R 2.4 McAdoo, HoseaN/AMcAdoo, HoseaN/AMcAdoo, HoseaN/AMcAlister, Robert $2.4; 2.13$ McAllister, Robert $2.4; 2.13$ McAlpine, Emily $2.4; 2.13$ McCarthy, Shirley $2.4; 2.13$ McCluor, John (State of Colorado) $2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4McClung, JudyN/AMcCornnic, LoriN/AMcCornnic, Molly2.4; 2.13McCulloch, Artch2.4; 2.13McCulloch, Artch2.4; 2.13McClung, JudyN/AMcCornnic, LoriN/AMcCornnic, Molly2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4McClundi, Mully2.4; 2.22McCornnic, Molly2.4; 2.13McCornnic, Molly2.4; 2.13McCornnic, Molly2.4; 2.13McCulloch, Artch2.4McCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, Samuel$	Mastri, Francis	2.4; 2.13
Matelski, Lauren 2.4; 2.13 Mathis, Leanne 2.4; 2.13 Mathis, Vicky 2.12 Matsui, Vicky 2.4; 2.13 Mattis, Henry N/A Matusik, Barbara 2.4; 2.13 Matusik, Barbara 2.4; 2.13 Matusik, Barbara 2.4; 2.13 Matusik, Barbara 2.4; 2.13 Maynard, Bruce 2.4 Maynard, Linda N/A Mazzuca, R 2.4 McAdoo, Hosea N/A McAllister, Revin 2.4; 2.13 McAllister, Robert 2.4 McAllister, Robert 2.4 McAllister, Robert 2.4 McAllister, Robert 2.4 McCline, Emily 2.4; 2.13 McCarty, Jesse Louis Henry N/A McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McConnell, David 3.2; 3.6; 8.3 McConnell, Suzanne N/A McCormic, Lori N/A McCormick, Molly 2.4; 2.13 McCulloch, Arch 2	Matejka, Harry	2.4; 2.13
Mathis, Leanne $2.4; 2.13$ Mathis, Marty 2.12 Matsui, Vicky $2.4; 2.13$ Mattes, Dale $2.4; 2.13$ Mattis, HenryN/AMatusik, Barbara $2.4; 2.13$ Maynard, Bruce 2.4 Maynard, LindaN/AMazuca, R 2.4 McAdoo, HoseaN/AMcAlister, Robert $2.4; 2.13$ McAllister, Robert $2.4; 2.13$ McAlister, Robert $2.4; 2.13$ McAlister, Robert $2.4; 2.13$ McCartly, Jesse Louis Henry $2.4; 2.13$ McCluow, John (State of Colorado) $2.2; 2.5; 2.15; 2.20; 6.1.2; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4$ McConnell, David $3.2; 3.6; 8.3$ McConnell, SuzanneN/AMcCormick, Molly $2.4; 2.13$ McCormick, Molly $2.4; 2.22$ McCreary, Jan $2.4; 2.13$ McCluloch, Arch $2.4; 2.13$ McCluloch, SamuelN/AMcCluloch, SamuelN/AMcCluloch, SamuelN/AMcCunlel, BarbaraN/A	Matelski, Lauren	2.4; 2.13
Mathis, Marty 2.12 Matsui, Vicky $2.4; 2.13$ Mattes, Dale $2.4; 2.13$ Mattis, HenryN/AMatusik, Barbara $2.4; 2.13$ Maynard, Bruce 2.4 Maynard, Diruce 2.4 Maynard, LindaN/AMazzuca, R 2.4 McAdoo, HoseaN/AMcAldoo, HoseaN/AMcAloo, Hosea $2.4; 2.13$ McAloo, Hosea $2.4; 2.13$ McAlister, Robert 2.4 McAlister, Robert 2.4 McAlister, Robert 2.4 McArij, Shirley $2.4; 2.13$ Mccarthy, Shirley $2.4; 2.13$ McCarty, Jesse Louis HenryN/AMcClow, John (State of Colorado) $2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4McConnell, David3.2; 3.6; 8.3McConnell, SuzanneN/AMcCormic, LoriN/AMcCormick, Molly2.4; 2.13McCulloch, Arch2.4McCulloch, SamuelN/AMcCulloch, SamuelN/A$	Mathis, Leanne	2.4; 2.13
Matsui, Vicky $2.4; 2.13$ Mattes, Dale $2.4; 2.13$ Mattis, HenryN/AMatusik, Barbara $2.4; 2.13$ Maynard, Bruce 2.4 Maynard, LindaN/AMazzuca, R 2.4 McAdoo, HoseaN/AMcAlister, Kevin $2.4; 2.13$ McAlister, Kevin $2.4; 2.13$ McAlister, Robert 2.4 McAllister, Robert 2.4 McAlpine, Emily 2.4 McBride, Nancy $2.4; 2.13$ McCarty, Jesse Louis HenryN/AMcClung, JudyN/AMcCornell, David $3.2; 3.6; 8.3$ McCormick, Molly $2.4; 2.13$ McCormick, Molly $2.4; 2.13$ McCormick, Molly $2.4; 2.13$ McClung, JudyN/AMcClung, JudyN/AMcCormick, Molly $2.4; 2.22$ McCormick, Molly $2.4; 2.22$ McClund, SamuelN/AMcCulloch, Arnch 2.4 McCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, BarbaraN/A	Mathis, Marty	2.12
Mattes, Dale 2.4; 2.13 Mattis, Henry N/A Matusik, Barbara 2.4; 2.13 Maynard, Bruce 2.4 Maynard, Linda N/A Mazzuca, R 2.4 McAdoo, Hosea N/A McAlister, Kevin 2.4; 2.13 McAlister, Kevin 2.4; 2.13 McAlister, Robert 2.4 McAlpine, Emily 2.4 McAlpine, Emily 2.4 McCarty, Jesse Louis Henry N/A McClung, Judy N/A McConnell, David 3.2; 3.6; 8.3 McCornell, David 3.2; 3.6; 8.3 McCornic, Lori N/A McCerary, Jan 2.4; 2.13 McCornick, Molly 2.4; 2.13 McCornick, Molly 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McConnell, David 3.2; 3.6; 8.3 McConnell, Suzane N/A McCornick, Molly 2.4; 2.22 McCreary, Jan 2.4; 2.13 McCulloch, Arch 2.4 McCulloch, Samuel N/A	Matsui, Vicky	2.4; 2.13
Mattis, Henry N/A Matusik, Barbara 2.4 ; 2.13 Maynard, Bruce 2.4 Maynard, Diruce 2.4 Maynard, Linda N/A Mazzuca, R 2.4 McAdoo, Hosea N/A McAlister, Kevin 2.4 ; 2.13 McAllister, Bud 2.4 ; 2.13 McAllister, Robert 2.4 McAlpine, Emily 2.4 McBride, Nancy 2.4 ; 2.13 McCarthy, Shirley 2.4 ; 2.13 McCarthy, Shirley 2.4 ; 2.13 McCarthy, State of Colorado) 2.2 ; 2.5 ; 2.15 ; 2.20 ; $6.1.2$; $6.1.3$; $6.4.12$; $6.4.13$; 8.6 ; 8.7 ; 8.8 ; 8.12 ; 10.8 ; 12.6 ; 13.4 McClung, JudyN/AMcConnell, David 3.2 ; 3.6 ; 8.3 McCornic, LoriN/AMcCreary, Jan 2.4 ; 2.13 McCreary, Jan 2.4 ; 2.22 McCreary, Jan 2.4 ; 2.13 McCulloch, Arch 2.4 ; 2.13 McCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, BarbaraN/A	Mattes, Dale	2.4; 2.13
Matusik, Barbara 2.4; 2.13 Maynard, Bruce 2.4 Maynard, Linda N/A Mazzuca, R 2.4 McAdoo, Hosea N/A McAlister, Kevin 2.4; 2.13 McAlister, Kevin 2.4; 2.13 McAlister, Bud 2.4; 2.13 McAlister, Robert 2.4 McAlpine, Emily 2.4 McBride, Nancy 2.4; 2.13 Mccarthy, Shirley 2.4; 2.13 McCarty, Jesse Louis Henry N/A McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McConnell, David 3.2; 3.6; 8.3 McCormic, Lori N/A McCormick, Molly 2.4; 2.13 McCreary, Jan 2.4; 2.13 McCulloch, Arch 2.4 McCulloch, Arch 2.4 McCulloch, Samuel N/A McCulloch, Samuel N/A McCulloch, Samuel N/A McCulloch, Samuel N/A McCulloch, Barbara N/A	Mattis, Henry	N/A
Maynard, Bruce2.4Maynard, LindaN/AMazzuca, R2.4McAdoo, HoseaN/AMcAlister, Kevin2.4; 2.13McAlister, Robert2.4McAlister, Robert2.4McAlpine, Emily2.4McBride, Nancy2.4; 2.13McCarthy, Shirley2.4; 2.13McClung, JudyN/AMcClung, JudyN/AMcCornell, David3.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4McClung, JudyN/AMcCornell, David3.2; 3.6; 8.3McCornic, LoriN/AMcCreary, Jan2.4; 2.13McCulloch, Arch2.4; 2.13McCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMccurdy, Robert3.6McDaniel, BarbaraN/A	Matusik, Barbara	2.4; 2.13
Maynard, LindaN/AMazzuca, R2.4McAdoo, HoseaN/AMcAlister, Kevin2.4; 2.13McAlister, Rud2.4; 2.13McAllister, Robert2.4McAlpine, Emily2.4McBride, Nancy2.4; 2.13McCarthy, Shirley2.4; 2.13McCarthy, Shirley2.4; 2.13McCarty, Jesse Louis HenryN/AMcClow, John (State of Colorado)2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4McConnell, David3.2; 3.6; 8.3McCornic, LoriN/AMcCornick, Molly2.4; 2.13McCulloch, Arch2.4; 2.13McCulloch, Arch2.4; 2.13McCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, BarbaraN/A	Maynard, Bruce	2.4
Mazzuca, R 2.4 McAdoo, HoseaN/AMcAlister, Kevin 2.4 ; 2.13 McAllister, Bud 2.4 ; 2.13 McAllister, Robert 2.4 McAlpine, Emily 2.4 McBride, Nancy 2.4 ; 2.13 McCarthy, Shirley 2.4 ; 2.13 McCarthy, Shirley 2.4 ; 2.13 McClow, John (State of Colorado) 2.2 ; 2.5 ; 2.15 ; 2.20 ; $6.1.2$; $6.1.3$; $6.4.12$; $6.4.13$; 8.6 ; 8.7 ; 8.8 ; 8.12 ; 10.8 ; 12.6 ; 13.4 McClung, JudyN/AMcConnell, David 3.2 ; 3.6 ; 8.3 McCormic, LoriN/AMcCormick, Molly 2.4 ; 2.13 McCurloch, Arch 2.4 McCulloch, Arch 2.4 McCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, SamuelN/AMcCulloch, BarbaraN/A	Maynard, Linda	N/A
McAdoo, Hosea N/A McAlister, Kevin 2.4; 2.13 McAllister, Bud 2.4; 2.13 McAllister, Robert 2.4 McAlpine, Emily 2.4 McBride, Nancy 2.4; 2.13 McCarthy, Shirley 2.4; 2.13 McCarthy, Shirley 2.4; 2.13 McCarty, Jesse Louis Henry N/A McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McConnell, David 3.2; 3.6; 8.3 McCormic, Lori N/A McCormick, Molly 2.4; 2.13 McCormick, Molly 2.4; 2.22 McCreary, Jan 2.4; 2.13 McCulloch, Arch 2.4 McCulloch, Samuel N/A McCulloch, Samuel N/A McCulloch, Samuel N/A	Mazzuca, R	2.4
McAlister, Kevin 2.4; 2.13 McAllister, Bud 2.4; 2.13 McAllister, Robert 2.4 McAlpine, Emily 2.4 McBride, Nancy 2.4; 2.13 McCarthy, Shirley 2.4; 2.13 McCarthy, Shirley 2.4; 2.13 McCarthy, Shirley 2.4; 2.13 McCarthy, Shirley 2.4; 2.13 McCarthy, Jesse Louis Henry N/A McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 N/A McConnell, David 3.2; 3.6; 8.3 McCormic, Lori N/A McCormick, Molly 2.4; 2.22 McCreary, Jan 2.4; 2.13 McCulloch, Arch 2.4 McCulloch, Samuel N/A McCurdy, Robert 3.6 McDaniel, Barbara N/A	McAdoo, Hosea	N/A
McAllister, Bud 2.4; 2.13 McAllister, Robert 2.4 McAlpine, Emily 2.4 McBride, Nancy 2.4; 2.13 Mccarthy, Shirley 2.4; 2.13 McCarty, Jesse Louis Henry N/A McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McConnell, David 3.2; 3.6; 8.3 McCormic, Lori N/A McCreary, Jan 2.4; 2.13 McCulloch, Arch 2.4 McCulloch, Samuel N/A McCulloch, Samuel N/A McCurdy, Robert 3.6 McDaniel, Barbara N/A	McAlister, Kevin	2.4; 2.13
McAllister, Robert 2.4 McAlpine, Emily 2.4 McBride, Nancy 2.4 ; 2.13 Mccarthy, Shirley 2.4 ; 2.13 McCarty, Jesse Louis Henry N/A McClow, John (State of Colorado) 2.2 ; 2.5 ; 2.15 ; 2.20 ; $6.1.2$; $6.1.3$; $6.4.12$; $6.4.13$; 8.6 ; 8.7 ; 8.8 ; 8.12 ; 10.8 ; 12.6 ; 13.4 McClung, Judy N/A McConnell, David 3.2 ; 3.6 ; 8.3 McConnell, Suzanne N/A McCormic, Lori N/A McCormick, Molly 2.4 ; 2.22 McCreary, Jan 2.4 ; 2.13 McCulloch, Arch 2.4 McCulloch, Samuel N/A Mccurdy, Robert 3.6 McDaniel, Barbara N/A	McAllister, Bud	2.4; 2.13
McAlpine, Emily 2.4 McBride, Nancy 2.4; 2.13 Mccarthy, Shirley 2.4; 2.13 McCarty, Jesse Louis Henry N/A McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McConnell, David 3.2; 3.6; 8.3 McCornicl, Lori N/A McCreary, Jan 2.4; 2.13 McCulloch, Arch 2.4; 2.13 McCulloch, Samuel N/A McCulloch, Samuel N/A McCulloch, Samuel N/A McCulloch, Barbara N/A	McAllister, Robert	2.4
McBride, Nancy 2.4; 2.13 Mccarthy, Shirley 2.4; 2.13 McCarty, Jesse Louis Henry N/A McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McConnell, David 3.2; 3.6; 8.3 McConnell, Suzanne N/A McCormic, Lori N/A McCreary, Jan 2.4; 2.13 McCulloch, Arch 2.4 McCulloch, Samuel N/A Mccurdy, Robert 3.6 McDaniel, Barbara N/A	McAlpine, Emily	2.4
Mccarthy, Shirley 2.4; 2.13 McCarty, Jesse Louis Henry N/A McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McConnell, David 3.2; 3.6; 8.3 McConnell, Suzanne N/A McCormic, Lori N/A McCreary, Jan 2.4; 2.22 McCulloch, Arch 2.4 McCulloch, Samuel N/A Mccurdy, Robert 3.6 McDaniel, Barbara N/A	McBride, Nancy	2.4; 2.13
McCarty, Jesse Louis Henry N/A McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; 8.7; 8.8; 8.12; 10.8; 12.6; 13.4 McClung, Judy N/A McConnell, David 3.2; 3.6; 8.3 McCornic, Lori N/A McCormick, Molly 2.4; 2.22 McCulloch, Arch 2.4 McCulloch, Samuel N/A McCurdy, Robert 3.6 McDaniel, Barbara N/A	Mccarthy, Shirley	2.4; 2.13
McClow, John (State of Colorado) 2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6; McClung, Judy N/A McConnell, David 3.2; 3.6; 8.3 McCornic, Lori N/A McCreary, Jan 2.4; 2.22 McCulloch, Arch 2.4 McCulloch, Samuel N/A McCurdy, Robert 3.6 McDaniel, Barbara N/A	McCarty, Jesse Louis Henry	N/A
8.7; 8.8; 8.12; 10.8; 12.6; 13.4McClung, JudyN/AMcConnell, David3.2; 3.6; 8.3McConnell, SuzanneN/AMcCormic, LoriN/AMcCormick, Molly2.4; 2.22McCreary, Jan2.4; 2.13McCulloch, Arch2.4McCulloch, SamuelN/AMccurdy, Robert3.6McDaniel, BarbaraN/A	McClow, John (State of Colorado)	2.2; 2.5; 2.15; 2.20; 6.1.2; 6.1.3; 6.4.12; 6.4.13; 8.6;
McClung, JudyN/AMcConnell, David3.2; 3.6; 8.3McConnell, SuzanneN/AMcCormic, LoriN/AMcCormick, Molly2.4; 2.22McCreary, Jan2.4; 2.13McCulloch, Arch2.4McCulloch, SamuelN/AMccurdy, Robert3.6McDaniel, BarbaraN/A		8.7; 8.8; 8.12; 10.8; 12.6; 13.4
McConnell, David3.2; 3.6; 8.3McConnell, SuzanneN/AMcCormic, LoriN/AMcCormick, Molly2.4; 2.22McCreary, Jan2.4; 2.13McCulloch, Arch2.4McCulloch, SamuelN/AMccurdy, Robert3.6McDaniel, BarbaraN/A	McClung, Judy	N/A
McConnell, SuzanneN/AMcCormic, LoriN/AMcCormick, Molly2.4; 2.22McCreary, Jan2.4; 2.13McCulloch, Arch2.4McCulloch, SamuelN/AMccurdy, Robert3.6McDaniel, BarbaraN/A	McConnell, David	3.2; 3.6; 8.3
McCormic, LoriN/AMcCormick, Molly2.4; 2.22McCreary, Jan2.4; 2.13McCulloch, Arch2.4McCulloch, SamuelN/AMccurdy, Robert3.6McDaniel, BarbaraN/A	McConnell, Suzanne	N/A
McCormick, Molly2.4; 2.22McCreary, Jan2.4; 2.13McCulloch , Arch2.4McCulloch, SamuelN/AMccurdy, Robert3.6McDaniel, BarbaraN/A	McCormic, Lori	N/A
McCreary, Jan2.4; 2.13McCulloch, Arch2.4McCulloch, SamuelN/AMccurdy, Robert3.6McDaniel, BarbaraN/A	McCormick, Molly	2.4; 2.22
McCulloch, Arch2.4McCulloch, SamuelN/AMccurdy, Robert3.6McDaniel, BarbaraN/A	McCreary, Jan	2.4; 2.13
McCulloch, SamuelN/AMccurdy, Robert3.6McDaniel, BarbaraN/A	McCulloch , Arch	2.4
Mccurdy, Robert3.6McDaniel, BarbaraN/A	McCulloch, Samuel	N/A
McDaniel, Barbara N/A	Mccurdy, Robert	3.6
	McDaniel, Barbara	N/A

McDaniel, Jared	2.8
McDonald, Barbara	2.4; 2.13
McDonald, Brian	3.6
McDonald, Carolyn	2.4; 2.13
McDonald, Kim	N/A
McDonald, Stacey	2.4
McFall, Cynthia	N/A
McFarland, Brian	2.4; 2.13
McFee, Matt	N/A
McGaughey, Mary	N/A
McGilligan, Micky	N/A
McGillivary, M	N/A
McGlone, Colleen	2.4; 2.13
McKee, Wendy	2.4; 2.13
McKibben, Michael	3.2; 3.6; 8.3
McKinnie, Robert	N/A
Mclaughlin, Dagmar	2.8
McLaughlin, Michael	3.2; 3.6; 8.3
McLean, Patricia	N/A
McLeod, Roderick	N/A
McMahon, Lally	2.1; 2.2; 2.8; 2.14; 13.1
McManus, Veronica	2.4
McMillan, Sharon	2.4
McMullen, Colleen	2.4; 2.13
McNeil, Kerry	2.4; 2.13
McPherson, Tom	N/A
McQueen, Harry	2.4
McVie, Christina	2.4; 2.13
McWilliams, Glen	2.4; 2.13
Mecke, Mike	2.4
Medina, Rachel	2.4; 2.13
Medlock, Richard	2.4
Mee, Diane	N/A
Mee, William (Agua Fria Village Association)	2.4; 2.13
Meehan, Ellie	2.4
Meeker, Helen	2.4; 2.13
Meeker, Tobias	2.8
Meeks, Mark	2.4
Meham, Owen	N/A
Mehle, Anthony	2.4; 2.13
Meinerding, Tony	N/A
Melby, Jane	N/A
Mellica, Jason	2.8
Menasco, Mika	N/A
Mendel, Chris	2.4; 2.13
Mendes, Ruth	2.4
Mendieta, Vince	2.4; 2.13
Mendousa, Tony	2.4; 2.13
Menyuk, Paula	N/A
Mercante, Ron	2.4

Word Prece NA Merki, Stefan NA Merki, Stefan 2.5; 12.2 Mernitz, Richard 3.2; 3.6 Merrigan, Anita N/A Merrigan, Anita N/A Merrigen, Anita N/A Merrigen, Anita N/A Merrigen, Anita N/A Metrier, Kichard 2.4; 2.13 Metelica, Nikita 2.4; 2.13 Mettelica, Nikita 2.4; 2.13 Metter, Michael 3.2; 3.6; 8.3 Meute, Naney 2.4; 2.13 Meyer, Douglas 2.4; 2.13 Meyer, Roger N/A Meyer, Roger N/A Mignella, Amy (Arizona Tribal Energy Association) 2.3; 2.7; 2.21; 12.5 Miles, Robert 2.3; 3.6; 8.3 Miller, Fid N/A Miller, Fid 2.4 Miller, Nachoer 2.4; 2.13 Miller, Nachoer 2.4; 2.13 Miller, Robert 3.2; 3.6; 8.3 Miller, Joe (Gila Tout Chapter 530 Trout Unlimited) 2.5; 2.17; 3.2; 3.3; 4; 3.6; 8.3; 12.2 Miller, Nacholeen </th <th>Mercieca, Charles (International Assoc. of Educators for</th> <th>N/A</th>	Mercieca, Charles (International Assoc. of Educators for	N/A
Methy, MichaelNAMerk, MichaelNAMerk, KichaelNAMerniz, Richard $32, 3.6$ Merrick, DianeNAMerring, AnitaNAMerring, Jim 2.22 Messer, Gretchen $2.4, 2.13$ Metelica, Nikita $2.4, 2.13$ Mettitt, BNAMeute, Nincy $2.4, 2.13$ Mettitt, BNAMeute, Nancy $2.4, 2.13$ Meyer, RiaNAMeute, Nancy $2.4, 2.13$ Meyer, RogerNAMignella, Amy (Arizona Tribal Energy Association) $2.3, 2.7, 2.21; 12.5$ Miles, Robert $2.4, 2.13$ Miller, D Rex 2.4 Miller, D Rex 2.4 Miller, Rathelen $3.2, 3.6, 8.3$ Miller, Rathelen $3.2, 3.6, 8.3$ Miller, Robert $2.4, 2.13$ Miller, Rathelen $3.2, 3.6, 8.3$ Miller, Rathelen $3.2, 3.6, 8.3$ Miller, Robert $2.4, 2.13$ Miller, Rathelen $3.2, 3.6, 8.3$ Miller, Fad $2.4, 2.13$ Miller, Pamela $2.4, 2.13$ Miller, Travis $2.4, 2.13$ Miller, Travis $2.4, 2.13$ Miller, Travis $2.4, 2.13$ Millen, Douglas (Salt River Project) $2.1, 2.7, 2.8, 2.10, 2.11, 2.13, 4.1.1, 5.2, 6.1.8, 6.2.2, 6.2, 6.4.7, 6.4.10, 6.4.11, 6.6.1, 6.7, 10.2, 10.3, 11.4Milling, John2.4, 2.13Miller, Minch, Cris2.4, 2.13Miller, Minchuk, GeneNAMinchuk, GeneNAMinchuk, GeneNA<$	World Peace)	NT/A
Metri, Stefan 2.5; 12.2 Merrikz, Richard 3.2; 3.6 Merrikz, Richard 3.2; 3.6 Merrikz, Richard N/A Merrikz, Richard N/A Merrikz, Richard N/A Merrikz, Richard N/A Metrik, Jim 2.22 Messer, Gretchen 2.4; 2.13 Mettik, B N/A Metzler, Michael 3.2; 3.6; 8.3 Meute, Nancy 2.4; 2.13 Meyer, Roger N/A Meyer, Roger N/A Meyers, Beth N/A Miles, Robert 2.4; 2.13 Miles, Robert 2.4; 2.13 Miles, Robert 2.4; 2.13 Miller, Joe (Gil Torut Chapter 530 Trout Unlimited) 2.3; 2.7; 2.21; 12.5 Miller, Joe (Gil Torut Chapter 530 Trout Unlimited) 2.3; 6; 8.3 Miller, Nark 32; 3.6; 8.3 Miller, Kathleen 2.2; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2 Miller, Robert 2.4; 2.13 Miller, Robert 2.4; 2.13 Miller, Robert 2.4; 2.13 Miller, Robert 2.4; 2.13 Miller, Robert	Merle Michael	N/A
Metrick, Ischard2.5, 12.2Merrick, DianeN/AMerrick, DianeN/AMerrick, DianeN/AMerrill, Jim2.22Messer, Gretchen2.4, 2.13Metelica, Nikita2.4, 2.13Metelica, Nikita2.4, 2.13Metur, BN/AMetur, Nancy2.4, 2.13Metur, Nancy2.4, 2.13Metur, Nancy2.4, 2.13Meyer, Douglas2.4, 2.13Meyer, Douglas2.4, 2.13Meyer, RogerN/AMyre, RogerN/AMiler, D Rex2.4Miler, D Rex2.4Miller, Joe (Gida Trout Chapter 530 Trout Unlimited)2.5, 2.17, 3.2, 3.3, 3.4, 3.6, 8.3, 12.2Miller, Kathleen3.2, 3.6, 8.3Miller, Fad2.4, 2.13Miller, Robert2.4, 2.13Miller, Robert2.4, 2.13Miller, Joe (Gida Trout Chapter 530 Trout Unlimited)2.5, 2.17, 3.2, 3.3, 3.4, 3.6, 8.3, 12.2Miller, Mark3.2, 3.6, 8.3Miller, Pamela2.4, 2.13Miller, Pamela2.4, 2.13Miller, Travis2.4, 2.13Miller, Travis2.4, 2.13Miller, Travis2.4, 2.13Milling, Douglas (Salt River Project)2.1, 2.7, 2.8, 2.10, 2.11, 2.13, 4.1.1, 5.2, 6.1.8, 6.2.2, 6.2, 6.4.5, 6.4.7, 6.4.10, 6.4.11, 6.6.1, 6.7, 10.2, 10.3, 11.4Milling, Douglas (Salt River Project)2.1, 2.7, 2.8, 2.10Miller, Chris2.4, 2.13Minchuk, GeneN/AMinchuk, GeneN/AMinstov, NinaN/AMinstov	Merk, Michael	N/A
Metrick, Diane 3.2 ; 3.6 Merrick, DianeN/AMerrick, DianeN/AMerrick, DianeN/AMester, Gretchen 2.4 ; 2.13 Mettil, Jim 2.4 ; 2.13 Mettil, BN/AMetzler, Michael 3.2 ; 3.6 ; 8.3 Meute, RitaN/AMetzler, Michael 3.2 ; 3.6 ; 8.3 Meute, Narcy 2.4 ; 2.13 Meyer, Douglas 2.4 ; 2.13 Meyer, RogerN/AMeyer, RogerN/AMignella, Amy (Arizona Tribal Energy Association) 2.3 ; 2.7 ; 2.21 ; 12.5 Miles, Robert 2.4 ; 2.13 Miller, De K 2.4 Miller, De Rex 2.4 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) 3.2 ; 3.6 ; 8.3 Miller, Mark 3.2 ; 3.6 ; 8.3 Miller, Mark 3.2 ; 3.6 ; 8.3 Miller, Susan 2.4 ; 2.13 Miller, Susan 2.4 ; 2.13 Miller, Susan 2.4 ; 2.13 Milligan, Douglas (Salt River Project) $2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.3; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4Millins, Nikolaos2.4; 2.13Minerk, GroenN/AMinerk, Carolyn2.4; 2.13Minerk, Carolyn2.4; 2.13Minerk, Carolyn2.4; 2.13Minerk, Carolyn2.4; 2.13Minerk, Carolyn2.4; 2.13Minerk, Carolyn2.4; 2.13Minkh, Daniel2.4; 2.13Minkhell, Nichael2.4; 2.13Mitchell$	Merrita Dishard	2.5; 12.2
Merriga, Anita N/A Merriga, Anita N/A Merriga, Anita N/A Messer, Gretchen 2.4; 2.13 Metelica, Nikita 2.4; 2.13 Mettir, B N/A Metzler, Michael 3.2; 3.6; 8.3 Meuer, Rita N/A Meyer, Douglas 2.4; 2.13 Meyer, Douglas 2.4; 2.13 Meyer, Roger N/A Myeyer, Roger N/A Miler, Robert 2.4; 2.13 Miles, Robert 2.4; 2.13 Miles, Robert 2.4; 2.13 Miler, D Rex 2.4 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) 3.2; 3.6; 8.3 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) 2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) 3.2; 3.6; 8.3 Miller, Nark 3.2; 3.6; 8.3 Miller, Nark 3.2; 3.6; 8.3 Miller, John 3.2; 3.6; 8.3 Miller, John 3.2; 3.6; 8.3 Milley, John 3.2; 3.6; 8.3 Milley, John 3.2; 3.6; 8.	Mernitz, Richard	3.2; 3.6
Merring, AnitaN/AMerring, Jim2.22Messer, Gretchen2.4; 2.13Mettitt, BN/AMettitt, BN/AMetzler, Michael3.2; 3.6; 8.3Meuer, RitaN/AMeuer, RitaN/AMeyer, Douglas2.4; 2.13Meyer, RogerN/AMeyers, BethN/AMignella, Amy (Arizona Tribal Energy Association)2.3; 2.7; 2.21; 12.5Miles, Robert2.4; 2.13Miles, Robert2.4; 2.13Miler, Dack (Arizona Tribal Energy Association)2.3; 2.7; 2.21; 12.5Miles, Robert2.4; 2.13Miler, Joe (Gita Trout Chapter 530 Trout Unlimited)2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2Miller, Fad2.4; 2.13Miller, Mark3.2; 3.6; 8.3Miller, Mark3.2; 3.6; 8.3Miller, Mark3.2; 3.6; 8.3Miller, Susan2.4; 2.13Miller, Susan2.4; 2.13Miller, Susan2.4; 2.13Milligran, Douglas (Salt River Project)2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4Millis, Eric (Utah Division of Water Resources)8.9Minonas, Nikolaos2.4; 2.13Minsky, NinaN/AMinsky, NinaN/AMitchell, Crystal2.4; 2.13Minkuk, GeneN/AMinkuk, Daniel2.4; 2.13Minkuk, Daniel2.4; 2.13Minkuk, Itchael2.4; 2.13Minkuk, NinaN/AMitchell, Michael2.4; 2.13<	Merrick, Diane	N/A
Metril, Jim 2.22 Messer, Gretchen 2.4; 2.13 Metelica, Nikita 2.4; 2.13 Mettly, Michael 3.2; 3.6; 8.3 Meure, Rita NA Meure, Rita NA Meyer, Douglas 2.4; 2.13 Meyer, Roger NA Meyer, Roger NA Mignella, Amy (Arizona Tribal Energy Association) 2.3; 2.7; 2.21; 12.5 Miles, Robert 2.3; 3.6; 8.3 Miller, D Rex 2.4 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) 2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2 Miller, Kathleen 3.2; 3.6; 8.3 Miller, Robert 2.4; 2.13 Miller, Robert 2.4; 2.13 Miller, Robert 2.4; 2.13 Miller, Stathleen 3.2; 3.6; 8.3 Miller, Robert 2.4; 2.13 Miller, Grolyn 3.2; 3.6; 8.3 <td< td=""><td>Merrigan, Anita</td><td>N/A</td></td<>	Merrigan, Anita	N/A
Messer, Gretchen 2.4, 2.13 Metelica, Nikita 2.4, 2.13 Mettit, B N/A Meuer, Rita N/A Meuer, Kita N/A Meuer, Nancy 2.4, 2.13 Meyer, Roger N/A Meyer, Roger N/A Meyer, Roger N/A Mignella, Amy (Arizona Tribal Energy Association) 2.3, 2.7, 2.21; 12.5 Miles, Robert 2.4, 2.13 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) 2.5, 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) 3.2; 3.6; 8.3 Miller, Mark 3.2; 3.6; 8.3 Miller, Pamela 2.4; 2.13 Miller, Susan 2.4; 2.13 Miller, Susan 2.4; 2.13 Miller, Travis 2.4; 2.13 Miller, Sic (Utah Division of Water Resources) 8.9 Minohuk, Gene N/A Minert, Carolyn 2.4; 2.13 Minst, Mich, Chris 2.4; 2.13 Minshy, Nina N/A Mitchell, Michael 2.4; 2.13 Minshy, Nina N/A Minchuk, Gene N/A	Merrill, Jim	2.22
Metelica, Nikita $24; 2.13$ Mettit, BNAMetzler, Michael $32; 3.6; 8.3$ Meuer, RitaNAMeuer, RitaNAMeuer, Roncy $24; 2.13$ Meyer, Douglas $24; 2.13$ Meyer, RogerNAMyeyer, BethNAMiles, Robert $23; 2.7; 2.21; 12.5$ Miles, Robert $32; 3.6; 8.3$ Miller, D Rex 2.4 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) $2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2$ Miller, Kathleen $32; 3.6; 8.3$ Miller, Robert $2.4; 2.13$ Miller, Susan $2.4; 2.13$ Miller, Robert $2.4; 2.13$ Miller, Robert $2.4; 2.13$ Miller, Susan $2.2; 3.6; 8.3$ Miller, Travis $2.4; 2.13$ Miller, Jouglas (Salt River Project) $2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 64.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4Milles, Eric (Utah Division of Water Resources)8.9Milonas, Nikolaos2.4; 2.13Minchuk, GeneNAMink, Daniel2.4; 2.13Mink, Daniel2.4; 2.13Mitchell, Ruby2.4; 2.13$	Messer, Gretchen	2.4; 2.13
Mettlit, BNAMetzler, Michael $32; 3, 6; 8, 3$ Meuer, RitaN/AMeute, Nancy $2.4; 2.13$ Meyer, Douglas $2.4; 2.13$ Meyer, RogerN/AMeyers, BethN/AMignella, Amy (Arizona Tribal Energy Association) $2.3; 2.7; 2.21; 12.5$ Miles, Robert $2.4; 2.13$ Miles, Robert $2.4; 2.13$ Miller, D Rex 2.4 Miller, D Rex 2.4 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) $2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2$ Miller, Mark $3.2; 3.6; 8.3$ Miller, Pamela $2.4; 2.13$ Miller, Robert $2.4; 2.13$ Miller, Susan $2.4; 2.13$ Miller, Travis $2.4; 2.13$ Miller, Travis $2.4; 2.13$ Miller, Sich Galt River Project) $2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.23; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4Millis, Eric (Utah Division of Water Resources)8.9Minonas, Nikolaos2.4; 2.13Minert, Carolyn2.4; 2.13Mink, GeneNAMinert, Carolyn2.4; 2.13Mink, Michell2.4; 2.13Mink, Daniel2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Nichael2.4; 2.13Mitchell, Crystal2.4Mitchell, Nichael2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Crystal2.4; 2.13Mitchell, Ruby2.4; 2.13$	Metelica, Nikita	2.4; 2.13
Metuer, Rita 3.2; 3.6; 8.3 Meuter, Rita NA Meuter, Nancy 2.4; 2.13 Meyer, Douglas 2.4; 2.13 Meyer, Roger N/A Mignella, Amy (Arizona Tribal Energy Association) 2.3; 2.7; 2.21; 12.5 Miles, Robert 2.4; 2.13 Miles, Robert 2.4; 2.13 Miler, D Rex 2.4 Miller, D Rex 2.4 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) 2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2 Miller, Kathleen 3.2; 3.6; 8.3 Miller, Kathleen 3.2; 3.6; 8.3 Miller, Robert 2.4; 2.13 Miller, Robert 2.4; 2.13 Miller, Kathleen 3.2; 3.6; 8.3 Miller, Robert 2.4; 2.13 Miller, Robert 2.4; 2.13 Miller, Robert 2.4; 2.13 Miller, Susan 2.4; 2.13 Miller, Susan 2.4; 2.13 Miller, John 3.2; 3.6; 4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4 Millis, Eric (Utah Division of Water Resources) 8.9 Milonas, Nikolaos 2.4; 2.13 Minchuk, Gene N/A	Mettitt, B	N/A
Meuter, RinNAMeuter, Nancy 2.4 ; 2.13 Meyer, Douglas 2.4 ; 2.13 Meyer, RogerNAMeyers, BethN/AMignella, Amy (Arizona Tribal Energy Association) 2.3 ; 2.7 ; 2.21 ; 12.5 Miles, Robert 3.2 ; 3.6 Miler, D Rex 2.4 Miller, D Rex 2.4 Miller, I, De (Gilla Trout Chapter 530 Trout Unlimited) 2.5 ; 2.17 ; 3.2 ; 3.3 ; 3.4 ; 3.6 ; 8.3 ; 12.2 Miller, Nark 3.2 ; 3.6 ; 8.3 Miller, Nark 3.2 ; 3.6 ; 8.3 Miller, Robert 2.4 ; 2.13 Miller, Robert 2.4 ; 2.13 Miller, Robert 2.4 ; 2.13 Miller, Susan 2.4 ; 2.13 Miller, Susan 2.4 ; 2.13 Miller, Susan 2.4 ; 2.13 Miller, John 3.2 ; 3.6 ; 8.3 Milligan, Douglas (Salt River Project) 2.1 ; 2.7 ; 2.8 ; 2.10 ; 2.11 ; 2.13 ; $4.1.1$; 5.2 ; $6.1.8$; $6.2.2$; $6.2.3$; $6.4.5$; $6.4.7$; $6.4.10$; $6.6.1$; 6.7 ; 10.2 ; 10.3 ; 11.4 Millis, Eric (Utah Division of Water Resources) 8.9 Minchak, GeneN/AMinert, Carolyn 2.4 ; 2.13 Mink, Daniel 2.4 ; 2.13 Minsky, NinaN/AMitchell, Michael 2.4 ; 2.13 Mitchell, Michael <t< td=""><td>Metzler, Michael</td><td>3.2; 3.6; 8.3</td></t<>	Metzler, Michael	3.2; 3.6; 8.3
Meute, Nancy 2.4; 2.13 Meyer, Boger NA Meyers, Beth NA Mignella, Amy (Arizona Tribal Energy Association) 2.3; 2.7; 2.21; 12.5 Miles, Robert 2.4; 2.13 Miller, De Rex 2.4 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) 2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2 Miller, Mark 3.2; 3.6; 8.3 Miller, Robert 2.4 Miller, Robert 2.4; 2.13 Miller, Intervis 2.4; 2.13 Miller, John 3.2; 3.6; 8.3 Miller, State 2.4; 2.13 Miller, State 8.9 Milonas, Nikolaos 2.4; 2.13 Minchuk, Gene N/A Minert, Carolyn 2.4; 2.13 Minkopl, Chris <td>Meuer, Rita</td> <td>N/A</td>	Meuer, Rita	N/A
Meyer, Douglas2.4; 2.13Meyer, RogerN/AMignella, Amy (Arizona Tribal Energy Association) $2.3; 2.7; 2.21; 12.5$ Miles, Robert $2.4; 2.13$ Miler, D Rex $2.4; 2.13$ Miller, D Rex 2.4 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) $2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2$ Miller, Kathleen $3.2; 3.6; 8.3$ Miller, Nark $3.2; 3.6; 8.3$ Miller, Robert $2.4; 2.13$ Miller, Susan $2.4; 2.13$ Miller, Tarvis $2.4; 2.13$ Miller, John $3.2; 3.6; 8.3$ Miller, Susan $2.4; 2.13$ Miller, John $3.2; 3.6; 8.3$ Miller, Cutha Division of Water Resources) 8.9 Milonas, Nikolaos $2.4; 2.13$ Minchuk, GeneN/AMinchuk, Carolyn $2.4; 2.13$ Minchuk, Carolyn $2.4; 2.13$ Mink, Daniel $2.4; 2.13$ Mink, Daniel $2.4; 2.13$ Mitchell, Crystal 2.4 Mitchell, Michael $2.4; 2.13$ Mitchell, Michael $2.4; 2.13$ Mitchell, Wuby $2.4; 2.13$ Mitchell, Wuby $2.4; 2.13$ Mitchell, Wuby $2.4; 2.13$ Mitchell, Wuby $2.4; 2.13$ Mitchell, Michael $2.4; 2.13$ Mitchell, Michael $2.4; 2.13$ Mitchell, Monroe 2.5 Modarelli, David $2.4; 2.13$ Mitchell, Monroe 2.5 Modarelli, David $2.4; 2.13$ Mitchell, Moriesota Rural Electric Association) 2.10	Meute, Nancy	2.4; 2.13
Meyer, RogerN/AMeyers, BethN/AMignella, Amy (Arizona Tribal Energy Association) $2.3; 2.7; 2.21; 12.5$ Miles, Robert $2.4; 2.13$ Miles, Robert $3.2; 3.6$ Miller, D Rex 2.4 Miller, Ed 2.4 Miller, Kathleen $3.2; 3.6; 8.3$ Miller, Mark $3.2; 3.6; 8.3$ Miller, Robert $2.4; 2.13$ Miller, Rathleen $3.2; 3.6; 8.3$ Miller, Rathleen $2.4; 2.13$ Miller, Susan $2.4; 2.13$ Miller, Travis $2.4; 2.13$ Miller, Jougas (Salt River Project) $2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4Millis, Eric (Utah Division of Water Resources)8.9Milonas, Nikolaos2.4; 2.13Minchuk, GeneN/AMinchuk, GeneN/AMinter, Carolyn2.4; 2.13Mitchell, Crystal2.4; 2.13Mitchell, Michael2.4; 2.13Mitchell, Nichael2.4; 2.13Mitchell, Yolanda2.4; 2.13Mitchell, Nichael2.4; 2.13Mitchell, Noroe2.5Mock. Randall (Randy)2.4; 2.13$	Meyer, Douglas	2.4; 2.13
Meyers, Beth N/A Mignella, Amy (Arizona Tribal Energy Association) $2.3; 2.7; 2.21; 12.5$ Miles, Robert $2.4; 2.13$ Miles, Robert $3.2; 3.6$ Miller, D Rex 2.4 Miller, D Rex 2.4 Miller, Led 2.4 Miller, Kathleen $3.2; 3.6; 8.3$ Miller, Mark $3.2; 3.6; 8.3$ Miller, Robert $2.4; 2.13$ Miller, Robert $2.4; 2.13$ Miller, Susan $2.4; 2.13$ Miller, Travis $2.4; 2.13$ Miller, John $3.2; 3.6; 8.3$ Miller, Susan $2.4; 2.13$ Miller, Group as (Salt River Project) $2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4 Miller, Gene N/A Minert, Carolyn 2.4; 2.13 Minch, Chris 2.4; 2.13 Minsky, Nina N/A Minsky, Nina N/A $	Meyer, Roger	N/A
Mignella, Amy (Arizona Tribal Energy Association) $2.3; 2.7; 2.21; 12.5$ Miles, Robert $2.4; 2.13$ Miles, Robert $3.2; 3.6$ Miller, D Rex 2.4 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) $2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2$ Miller, Kathleen $2.4; 2.13$ Miller, Kathleen $2.4; 2.13$ Miller, Robert $2.4; 2.13$ Miller, Robert $2.4; 2.13$ Miller, Robert $2.4; 2.13$ Miller, Susan $2.4; 2.13$ Miller, Jounglas (Salt River Project) $2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4 Millis, Eric (Utah Division of Water Resources) 8.9 Minchuk, Gene N/A Minert, Carolyn 2.4; 2.13 Mink, Daniel 2.4; 2.13 Mink, Daniel 2.4; 2.13 Mink, Daniel 2.4; 2.13 Minkell, Crystal 2.4 Mitchell, Nichael 2.4; 2.13 Minkuk, Qane 2.13 Minkuk, Qane 2.4; 2.13 Minkuk, Qane 2.4; 2.13 Minkuk, Qane $	Meyers, Beth	N/A
Miles, Robert $2.4; 2.13$ Miles, Robert $3.2; 3.6$ Miller, D Rex 2.4 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) $2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2$ Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) $3.2; 3.6; 8.3$ Miller, Kathleen $3.2; 3.6; 8.3$ Miller, Ramela $2.4; 2.13$ Miller, Robert $2.4; 2.13$ Miller, Susan $2.4; 2.13$ Miller, Susan $2.4; 2.13$ Miller, Jouglas (Salt River Project) $2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4Millis, Eric (Utah Division of Water Resources)8.9Milonas, Nikolaos2.4; 2.13Minert, Carolyn2.4; 2.13Minch, GeneN/AMinchel, Chris2.4; 2.13Mink, Daniel2.4; 2.13Mink, Daniel2.4; 2.13Mink, Daniel2.4; 2.13Minkell, Chris2.4; 2.13Minkell, Nichael2.4; 2.13Minkell, Ruby2.4; 2.13Mitchell, Nichael2.4; 2.13Mitchell, Nuby2.4; 2.13Mitchell, Nuby2.1; 2.13; 4.1; 5.2; 5.5; 5.5; 5.5; 5.5; 5.5; 5.5; 5.5$	Mignella, Amy (Arizona Tribal Energy Association)	2.3; 2.7; 2.21; 12.5
Miles, Robert 3.2; 3.6 Miller, D Rex 2.4 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) 2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) 3.2; 3.6; 8.3 Miller, Kathleen 3.2; 3.6; 8.3 Miller, Mark 3.2; 3.6; 8.3 Miller, Robert 2.4; 2.13 Miller, Susan 2.4; 2.13 Miller, Susan 2.4; 2.13 Miller, Travis 2.4; 2.13 Milley, John 3.2; 3.6; 8.3 Milley, John 3.2; 3.6; 8.3 Milligan, Douglas (Salt River Project) 2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4 Millis, Eric (Utah Division of Water Resources) 8.9 Milonas, Nikolaos 2.4; 2.13 Minchuk, Gene N/A Minet, Carolyn 2.4; 2.13 Mink, Daniel 2.4; 2.13 Minky, Nina N/A Mitchell, Nichael 2.4; 2.13 Mitchell, Ruby 2.4; 2.13 Mitchell, Nuchael 2.4; 2.13 Mitchell, Nuchael 2.4; 2.13 Mitchell, Notae 2.5	Miles, Robert	2.4; 2.13
Miller, D Rex 2.4 Miller, Ed 2.4 Miller, Ge (Gila Trout Chapter 530 Trout Unlimited) 3.2; 3.6; 8.3 Miller, Kathleen 3.2; 3.6; 8.3 Miller, Mark 3.2; 3.6; 8.3 Miller, Pamela 2.4; 2.13 Miller, Robert 2.4; 2.13 Miller, Travis 2.4; 2.13 Miller, Susan 2.4; 2.13 Miller, John 3.2; 3.6; 8.3 Milley, John 3.2; 3.6; 8.3 Milley, John 3.2; 3.6; 8.3 Milley, John 3.2; 3.6; 8.3 Milligan, Douglas (Salt River Project) 2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4 Millis, Eric (Utah Division of Water Resources) 8.9 Milonas, Nikolaos 2.4; 2.13 Minchuk, Gene N/A Minch, Chris 2.4; 2.13 Minky, Nina N/A Mitchell, Crystal 2.4 Mitchell, Yolanda 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Mitchell, Yolanda 2.4; 2.13 Mitchell, Yolanda 2.4; 2.13 Mizel, Monroe 2.5 <	Miles, Robert	3.2; 3.6
Miller, Ed 2.4 Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) 2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2 Miller, Kathleen 3.2; 3.6; 8.3 Miller, Mark 3.2; 3.6; 8.3 Miller, Pamela 2.4; 2.13 Miller, Robert 2.4; 2.13 Miller, Susan 2.4; 2.13 Miller, Travis 2.4; 2.13 Milley, John 3.2; 3.6; 8.3 Milligan, Douglas (Salt River Project) 2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4 Millis, Eric (Utah Division of Water Resources) 8.9 Milonas, Nikolaos 2.4; 2.13 Minchuk, Gene N/A Minetr, Carolyn 2.4; 2.13 Mink, Daniel 2.4; 2.13 Minky, Nina N/A Mitchell, Ruby 2.4; 2.13 Minktplanel 2.4; 2.13 Minktplanel 2.4; 2.13 Mirkell, Yolanda 2.4; 2.13 Mirkell, Nornee	Miller, D Rex	2.4
Miller, Joe (Gila Trout Chapter 530 Trout Unlimited) $2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2$ Miller, Kathleen $3.2; 3.6; 8.3$ Miller, Mark $3.2; 3.6; 8.3$ Miller, Pamela $2.4; 2.13$ Miller, Robert $2.4; 2.13$ Miller, Travis $2.4; 2.13$ Miller, Travis $2.4; 2.13$ Miller, John $3.2; 3.6; 8.3$ Miller, John $3.2; 3.6; 8.3$ Miller, Travis $2.4; 2.13$ Miller, John $3.2; 3.6; 8.3$ Miller, Susan $2.4; 2.13$ Miller, Travis $2.4; 2.13$ Miller, Douglas (Salt River Project) $2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 62.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4Millinas, Nikolaos2.4; 2.13Minnehuk, GeneN/AMinchuk, GeneN/AMink, Daniel2.4; 2.13Minky, NinaN/AMitchell, Crystal2.4Mitchell, Crystal2.4Mitchell, Ruby2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Ruby2.4; 2.13Mitzel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Moze, Dartick (Minnesota Rural Electric Association)2.10$	Miller, Ed	2.4
Miller, Kathleen 3.2; 3.6; 8.3 Miller, Mark 3.2; 3.6; 8.3 Miller, Pamela 2.4; 2.13 Miller, Susan 2.4; 2.13 Miller, Travis 2.4; 2.13 Miller, Travis 2.4; 2.13 Miller, John 3.2; 3.6; 8.3 Miller, Susan 2.4; 2.13 Miller, Travis 2.4; 2.13 Milley, John 3.2; 3.6; 8.3 Milligan, Douglas (Salt River Project) 2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4 Millinas, Nikolaos 2.4; 2.13 Milonas, Nikolaos 2.4; 2.13 Minchuk, Gene N/A Minert, Carolyn 2.4; 2.13 Mink, Daniel 2.4; 2.13 Mink, Nina N/A Mitchell, Crystal 2.4 Mitchell, Michael 2.4; 2.13 Mitchell, Musel 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Mitzel, Monroe 2.5 Mock. Randall (Randy)	Miller, Joe (Gila Trout Chapter 530 Trout Unlimited)	2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
Miller, Mark 3.2; 3.6; 8.3 Miller, Pamela 2.4; 2.13 Miller, Robert 2.4; 2.13 Miller, Susan 2.4; 2.13 Miller, Travis 2.4; 2.13 Milley, John 3.2; 3.6; 8.3 Milley, John 3.2; 3.6; 8.3 Milley, John 3.2; 3.6; 8.3 Milligan, Douglas (Salt River Project) 2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; (5.23; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4 Millis, Eric (Utah Division of Water Resources) 8.9 Milonas, Nikolaos 2.4; 2.13 Minchuk, Gene N/A Minert, Carolyn 2.4; 2.13 Mink, Daniel 2.4; 2.13 Mink, Sy, Nina N/A Mitchell, Crystal 2.4 Mitchell, Michael 2.4; 2.13 Mitchell, Nichael 2.4; 2.13 Mitchell, Yolanda 2.4; 2.13 Mitchell, Yolanda 2.4; 2.13 Mitchell, Noroe 2.5 Mock. Randall (Randy) 2.4; 2.13 Modarelli, David 2.4; 2.13 Modarelli, David 2.4; 2.13	Miller, Kathleen	3.2; 3.6; 8.3
Miller, Pamela 2.4; 2.13 Miller, Robert 2.4; 2.13 Miller, Susan 2.4; 2.13 Miller, Travis 2.4; 2.13 Milley, John 3.2; 3.6; 8.3 Milligan, Douglas (Salt River Project) 2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; (6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4 Millis, Eric (Utah Division of Water Resources) 8.9 Milonas, Nikolaos 2.4; 2.13 Minchuk, Gene N/A Minert, Carolyn 2.4; 2.13 Mink, Daniel 2.4; 2.13 Mink, Sy, Nina N/A Mitchell, Crystal 2.4 Mitchell, Ruby 2.4; 2.13 Mitchell, Nichael 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Mitchell, Nonroe 2.5 Mock. Randall (Randy) 2.4; 2.13 Modarelli, David 2.4; 2.13 Modarelli, David 2.4; 2.13	Miller, Mark	3.2; 3.6; 8.3
Miller, Robert 2.4; 2.13 Miller, Susan 2.4; 2.13 Miller, Travis 2.4; 2.13 Milley, John 3.2; 3.6; 8.3 Milligan, Douglas (Salt River Project) 2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4 Millis, Eric (Utah Division of Water Resources) 8.9 Milonas, Nikolaos 2.4; 2.13 Minchuk, Gene N/A Minert, Carolyn 2.4; 2.13 Mink, Daniel 2.4; 2.13 Mink, Daniel 2.4; 2.13 Mitchell, Crystal 2.4 Mitchell, Michael 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Mitchell, Michael 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Mo	Miller, Pamela	2.4; 2.13
Miller, Susan 2.4; 2.13 Miller, Travis 2.4; 2.13 Milley, John 3.2; 3.6; 8.3 Milligan, Douglas (Salt River Project) 2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4 Millis, Eric (Utah Division of Water Resources) 8.9 Milonas, Nikolaos 2.4; 2.13 Minchuk, Gene N/A Minch, Chris 2.4; 2.13 Mink, Daniel 2.4; 2.13 Mink, Daniel 2.4; 2.13 Mitchell, Crystal 2.4 Mitchell, Michael 2.4; 2.13 Mitchell, Nuby 2.4; 2.13 Modarelli, David 2.4; 2.13 Modare	Miller, Robert	2.4; 2.13
Miller, Travis 2.4; 2.13 Milley, John 3.2; 3.6; 8.3 Milligan, Douglas (Salt River Project) 2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4 Millis, Eric (Utah Division of Water Resources) 8.9 Milonas, Nikolaos 2.4; 2.13 Minchuk, Gene N/A Minert, Carolyn 2.4; 2.13 Mink, Daniel 2.4; 2.13 Mink, Daniel 2.4; 2.13 Mitchell, Crystal 2.4 Mitchell, Michael 2.4; 2.13 Mitchell, Nuchael 2.4; 2.13 Mitchell, Noroe 2.5 Mock. Randall (Randy) 2.4; 2.13 Modarelli, David 2.4; 2.13	Miller, Susan	2.4; 2.13
Milley, John $3.2; 3.6; 8.3$ Milligan, Douglas (Salt River Project) $2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4Millis, Eric (Utah Division of Water Resources)8.9Milonas, Nikolaos2.4; 2.13Minchuk, GeneN/AMinert, Carolyn2.4; 2.13Mink, Daniel2.4; 2.13Mink, Daniel2.4; 2.13Mitchell, Crystal2.4Mitchell, Michael2.4; 2.13Mitchell, Michael2.4; 2.13Mitchell, Noroe2.5Mock. Randall (Randy)2.4; 2.13More, Darrick (Minnesota Rural Electric Association)2.4; 2.13$	Miller, Travis	2.4; 2.13
Milligan, Douglas (Salt River Project) $2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2; 6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4Millis, Eric (Utah Division of Water Resources)8.9Milonas, Nikolaos2.4; 2.13Minchuk, GeneN/AMinert, Carolyn2.4; 2.13Mink, Daniel2.4; 2.13Mink, Daniel2.4; 2.13Mink, Daniel2.4; 2.13Minkehell, Crystal2.4Mitchell, Michael2.4; 2.13Mitchell, Nichael2.4; 2.13Mitchell, Noroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10$	Milley, John	3.2; 3.6; 8.3
6.2.3; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3; 11.4Millis, Eric (Utah Division of Water Resources)8.9Milonas, Nikolaos2.4; 2.13Minchuk, GeneN/AMinert, Carolyn2.4; 2.13Mink, Daniel2.4; 2.13Minsky, NinaN/AMitchell, Crystal2.4Mitchell, Michael2.4; 2.13Mitchell, Nuby2.4; 2.13Mitchell, Nuby2.4; 2.13Mitchell, Nuby2.4; 2.13Mitchell, Nuby2.4; 2.13Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Modarelli, David2.10	Milligan, Douglas (Salt River Project)	2.1; 2.7; 2.8; 2.10; 2.11; 2.13; 4.1.1; 5.2; 6.1.8; 6.2.2;
11.4Millis, Eric (Utah Division of Water Resources)8.9Milonas, Nikolaos2.4; 2.13Minchuk, GeneN/AMinert, Carolyn2.4; 2.13Minich, Chris2.4; 2.13Mink, Daniel2.4; 2.13Minsky, NinaN/AMitchell, Crystal2.4Mitchell, Michael2.4; 2.13Mitchell, Nuby2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10		6.2.3; 6.4.5; 6.4.7; 6.4.10; 6.4.11; 6.6.1; 6.7; 10.2; 10.3;
Millis, Eric (Utah Division of Water Resources)8.9Milonas, Nikolaos2.4; 2.13Minchuk, GeneN/AMinert, Carolyn2.4; 2.13Minich, Chris2.4; 2.13Mink, Daniel2.4; 2.13Minsky, NinaN/AMitchell, Crystal2.4Mitchell, Michael2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10		11.4
Milonas, Nikolaos2.4; 2.13Minchuk, GeneN/AMinert, Carolyn2.4; 2.13Minich, Chris2.4; 2.13Mink, Daniel2.4; 2.13Minsky, NinaN/AMitchell, Crystal2.4Mitchell, Michael2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Millis, Eric (Utah Division of Water Resources)	8.9
Minchuk, GeneN/AMinert, Carolyn2.4; 2.13Minich, Chris2.4; 2.13Mink, Daniel2.4; 2.13Minsky, NinaN/AMitchell, Crystal2.4Mitchell, Michael2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Milonas, Nikolaos	2.4; 2.13
Minert, Carolyn2.4; 2.13Minich, Chris2.4; 2.13Mink, Daniel2.4; 2.13Minsky, NinaN/AMitchell, Crystal2.4Mitchell, Michael2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Minchuk, Gene	N/A
Minich, Chris2.4; 2.13Mink, Daniel2.4; 2.13Minsky, NinaN/AMitchell, Crystal2.4Mitchell, Michael2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Minert, Carolyn	2.4; 2.13
Mink, Daniel2.4; 2.13Minsky, NinaN/AMitchell, Crystal2.4Mitchell, Michael2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Minich, Chris	2.4; 2.13
Minsky, NinaN/AMitchell, Crystal2.4Mitchell, Michael2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Mink, Daniel	2.4; 2.13
Mitchell, Crystal2.4Mitchell, Michael2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Minsky, Nina	N/A
Mitchell, Michael2.4; 2.13Mitchell, Ruby2.4; 2.13Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Mitchell, Crystal	2.4
Mitchell, Ruby2.4; 2.13Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Mitchell, Michael	2.4: 2.13
Mitchell, Yolanda2.4; 2.13Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Mitchell, Ruby	2.4; 2.13
Mizel, Monroe2.5Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Mitchell, Yolanda	2.4: 2.13
Mock. Randall (Randy)2.4; 2.13Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Mizel, Monroe	2.5
Modarelli, David2.4; 2.13Moe, Darrick (Minnesota Rural Electric Association)2.10	Mock, Randall (Randy)	2.4: 2.13
Moe, Darrick (Minnesota Rural Electric Association) 2.10	Modarelli, David	2.4: 2.13
	Moe. Darrick (Minnesota Rural Electric Association)	2.10

Moffatt, James	N/A
Moffatt, James	N/A
Moffit, Ian	3.2; 3.6; 8.3
Moir, Dolie	N/A
Molling, Corrine	N/A
Monaghan, Dina	N/A
Monfredini, Janet	N/A
Monie, Sherry	2.4; 2.13
Monroe, James R	2.4: 2.13
Monson. Michael	2.8: 2.14
Monson, Todd	N/A
Montelongo, Monica	N/A
Montoro Ernest	N/A
Moody Richard	2 4. 2 13
Moon Peggy	2.1, 2.13
Moore Ariel	2.4, 2.13
Moore Chris	2.5, 2.15
Moore Chris	2.5, 5.0, 12.2
Moore, Chills	2.4, 2.15
Moore, Dieg	5.2, 5.0 N/A
Moore, Lucy	IN/A
Moore, Maic	N/A
Moore, Matt	2.4
Moore, Nancy	2.4; 2.13
Moore, Robert	N/A
Moore, Thomas	N/A
Moorehead, Elisabeth	2.4; 2.13
Mooz, William	2.4
Morales, Susan	N/A
Moran, Patricia	N/A
Morebellob, Sam	3.2; 3.6; 8.3
Moreira, Rui	2.4; 2.13
Morel, N/A	2.4
Morello, Phyl	2.4
Morgan, Edward	2.4; 2.13
Morgan, Julie	N/A
Morgan, Michelle	N/A
Morgan, Sarah	2.4; 2.13
Morin, Inez	2.4; 2.13
Moritz, Robert	N/A
Morris. John	3.2: 3.6: 8.3
Morris. Lynn	N/A
Morris Peter	3 2 3 6
Morris William	2 4 2 13
Morrison N	2 4: 2 13
Morrow James	3 2: 3 6: 8 3
Mosea-Clark Vivianne	N/A
Mosel Sharon	N/A
Moser Rich	N/A
Moshier Katharine	N/A
Mothley Drugille	
wouncy, Diucilia	1N/ <i>F</i> A

Motz, Mary	2.4; 2.13
Moulton, Jamie	2.8
Mourant, Wanda	N/A
Moyer, Steve (Trout Unlimited)	2.5; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
Mudrick, Stephen	2.4; 2.13
Mueller, Ray	2.8; 2.14
Mugglestone, Lindsay	2.4; 2.13
Mulholland, Jane	N/A
Mullen, Charles	2.4
Munar, Dwayne	2.4; 2.13
Munday, Sherrie	2.4
Mundine, Jennifer	N/A
Munsell, Steven	2.4
Murchison, Mary	2.4
Murdoch Sarah	N/A
Murphy Brian	3.6
Murphy Dacia	24
Murphy Daniel	2.4. 2.13
Murphy John	3 7. 3 6. 8 3
Muss Leff	2 4. 2 13
Muss, Juli	2.4, 2.13
Myers, Stanhan	2.4, 2.15 N/A
Myers, Stephen Myers Davis, Alan	
Myers-Davis, Alan	N/A 2 (
Myl, David	3.0
Mylrea, Mary Ellen	2.4; 2.13
NA ^c (Arizona Raft Adventures LLC, Arizona)	2.5; 12.4
NA, Aaron	2.4; 2.13
NA, Adam	2.4; 2.13
NA, Cameron	3.6
NA, Charleen	2.4; 2.13
NA, Darina	N/A
NA, Karen	N/A
NA, Kristine	2.4; 2.13
NA, Martha	2.4; 2.13
NA, Michelle	2.4; 2.13
NA, NA (Arizona River Runners)	2.12; 12.1
NA, Nick	N/A
NA, Paul	N/A
NA, Peter	N/A
NA, Robert	2.4
NA, Ronald	N/A
NA. Shannin	2.4: 2.13
NA Shawn	N/A
NA Tom	24
NA Victor	2 4: 12 2
NA Wendy	N/A
NA NA (Western River Expeditions)	2 4: 3 7: 9 1: 12 1: 12 2
NA NA (Petition)	3 2. 3 6. 8 3
Naciri Nour	2 4. 2 13
Naegele Alice	2.τ, 2.13 N/Δ
Ivaugul, Allut	1N/ /

_

Nagel, Clinton	2.4; 2.13
Nahill, Brad	2.4; 2.13
Naidich, Sandra	N/A
Naples, Jean	2.4; 2.13
Nappi, Glenn	2.4; 2.13
Nazzaro, Alan	2.4; 2.13
Neal, Stacy	2.4
Nearing, Sue	N/A
Nebergall, Bradford (Tri-State Generation and	2.1; 2.2; 2.3; 2.7; 2.9; 2.10; 2.19; 3.2; 4.1.3; 4.2; 5.2;
Transmission Association. Inc.)	6.4.6: 6.4.8: 6.4.9: 7.3.3.: 8.9: 10.1: 14.2
Nedeau, E. James	2.4: 2.13
Neff Laura	2.4.2.13
Nelson B	2 4 2 13
Nelson, B.	2 4: 2 13
Nelson David	2.4, 2.13
Nelson David	2.7, 2.13
Nelson, Den	2.7, 2.15
Nelson, Jon	2.5
Nelson, Jaieu	2.5
Nelson, Justin	5.2, 5.0, 8.5
Neuson, Köpert	5.2, 5.0, 8.5
Nemko, Roy	N/A
Neste, George	2.4; 2.13
Neste, Lisa	2.4; 2.13
Netti , Steve	3.2; 3.6; 8.3
Neumann, Nancy	2.4; 2.13
Newman, Eric	N/A
Newman, Ricki	2.4; 2.13
Nichols, Carmen	N/A
Nichols, Nathaniel	2.4; 2.13
Nicholson, Jane	2.4
Nicolai, Nicola	2.4; 2.13
Nielsen, Leonor	N/A
Niemiec, Michael	2.4; 2.13
Nieters, Lenore	2.4; 2.13
Nimkin, David (National Parks Conservation	2.4; 2.13; 6.1.5; 7.3; 10.1; 10.3
Association & Grand Canyon Trust)	
Nitido, Vincent	2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
Noble, Bruce	N/A
Nochimson Martha	2.4.2.13
Nolan Antoinette	2 4 2 13
Noll Fred	2.4
Noon Gail	N/A
Norhury Christopher	2 4. 2 13
Nord Randall	2.1, 2.13
Nordberg Hella	2.1, 2.13
Norden Michael	2. 7 , 2. 1 <i>3</i> 2 <i>A</i>
Norman John	2. - 2.4
Norris Dick Ir	2. 2.4· 2.12
Novale Nanay	2. 4 , 2.13 2.4, 2.12
Novak, INANCY	2.4, 2.13
inovak, Trina	2.4; 2.13

Novkov, Russell	N/A
Nowack, Laura	N/A
Nowell, Jerry	3.2; 3.6
Nowicki, Susan	N/A
Noyes, Alan	2.8
Nudi, Flovd	2.4; 2.13
Nunez. Stephanie	N/A
Nunn. Tom	3.2: 3.6: 8.3
Nussbaum Rhoda	2 4 2 13
Nutini Michael	2 4 2 13
Nylen Fric	2.1, 2.13
O Ryan	N/A
Oakes Dorothy	2 4. 2 13
Oakley Jane	2.π, 2.15 N/Δ
Oates Judy	N/A
Oates, Judy	N/A
Odecs, Tilli Oha Daggy	N/A 2.4: 2.12
Obart Learnard	2.4, 2.13
Obert, Leonard	N/A
Obr, Brooks	2.4; 2.13
O'Brien, IEE	3.2; 3.0; 8.3
O'Brien, Laura	2.4; 2.13
O'Connell, John	2.4
OConnor, Dan	2.4
O'Connor, Susan	2.4; 2.13
ODear, Elizabeth	2.4
Odom, Cassandra	N/A
O'Donnell, Richard	N/A
Ogella, Edith	N/A
Ogrosky, Wendell	3.2; 3.6; 8.3
Ohara, Stanley	3.2; 3.6; 8.3
Ohlendorf, Carol	2.4; 2.13
OKeefe, Alice	2.4; 2.13
Olander, Alan	2.4; 2.13
Olch, Jonathan	3.2; 3.6; 8.3
Oldfield, Barbara	N/A
Olivares, Yvonne	2.4
Olsen, Donna	2.4
Olson, Bruce	N/A
Olson, Jason	3.2; 3.6; 8.3
Olson, Lynn	2.4; 2.8; 2.13
Ommen, Elizabeth	N/A
Orcholski, Gerald	2.4; 2.13
O'Reilly, Patricia	2.4
Orlinski, Patricia (West Valley Neighborhoods	2.4
Coalition)	
Orr Jenne	2.4
Osada Susan	2 4 2 13
Osborn Carole	2.4.2.13
Osborne Kenneth	2.4.2.13
OSteen Karen	2.1, 2.13
	2.T, 2.1J

Ostler, Don (Upper Colorado River Commission)	2.19; 4.1.1; 6.3.1; 6.4.12; 6.4.13; 6.4.14; 6.5.1; 8.8; 10.3		
Ostopoff Christine	2.4		
Ostrander William Ir	2.4 2.13		
Ostwinkle, Stephen	3.6		
Ouellette Tracy	2 4 2 13		
Outland Jennifer	2 13		
Overton Steve	2.4.2.13		
Owen Cheryl	2.4		
Owens Hilery	2.4		
Owens Mary	2.4 2.13		
Ozias Iulie	2 4 2 13		
Pace Ann	N/A		
Packer Patti	2 4 2 13		
Padula P S	2.8		
Paige Richard	N/A		
Paisley Lorna	N/A		
Pakaln Laura	2 4. 2 13		
Paley Leon	2.4. 2.13		
Palmer Carol	N/A		
Pannaman Stanley	24		
Panteah Val (Pueblo of Zuni)	2.4		
Panza Mike	N/A		
Panesch Peter	2 4		
Pappas Robin	N/A		
Paradise Brian	2 4		
Parker Delores	2.1		
Parker, Johones Parker, Jin Adams	2 4 2 13		
Parker Larry	3 2 3 6		
Parker Mark	3 2 3 6		
Parker Patricia	2 4		
Parkins Ianet	2.4 2.13		
Parks Antonio Mary Anne	2.4		
Parus Christine	N/A		
Paschel Richard	2 4		
Pasqua John	N/A		
Patten Robin	2 4 2 13		
Patton Therese	2 4 2 13		
Pauls Terry	2 4 2 13		
Pawlowski Vincent (Association for the Tree of Life)	2 4 2 13		
Paxton G	2 4 2 13		
Pearlman Karen	2.4. 2.13		
Pecha Richard	N/A		
Pehrson Danny	N/A		
Pelleg Josh	N/A		
Pellegrino Colby (Southern Nevada Water Authority)	4 1 3 5 2 8 8 8 12 10 1 13 3 13 4		
Peltzer Bryan	3 6. 12 4		
Pence Ioanne	N/A		
Pender Jacqueline	2 4 2 13		
Percy Patrick	N/A		
Periz, Thomas 24 Perez, Thomas 24, 2.13 Perkins, Guy 24, 2.13 Perkins, Marie 24, 2.13 Pertry, Sul NA Person, Wayne 24, 2.13 Peteinarki, Maria NA Peters, Robert NA Peters, Robert NA Peters, Robert NA Peterse, Robert NA Peterse, Robert NA Peterse, Joine 24, 2.13 Petrose, Jon NA Petigrew, Jill N/A Petigrew, Jill N/A Phares, David NA Phares, David NA Phares, David N/A Phares, David N/A Philips, Charles 32, 36, 8.3 Philips, Stomas 2.4, 2.13 Philips, Charles 2.4, 2.13 Procker, Seth 2.4, 2.13 Picker, Seth 2.4, 2.13 Picker, Seth 2.4, 2.13 Pierson, James 2.4, 2.13 Pierson, James 2.4, 2.13 Pierson, James 2.4, 2.13			
--	---	------------------------	---------------
Perec, Thomas $2.4; 2.13$ Perkins, Marie $2.4; 2.13$ Perkins, Marie $2.4; 2.13$ Person, Wayne $2.4; 2.13$ Peters, RobertN/APeters, RobertN/APeters, RobertN/APeters, RobertN/APeters, RobertN/APeters, RobertN/APeters, SobertN/APeters, SobertN/APeters, SobertN/APeters, JohnN/APeters, Diane $2.4; 2.13$ Petroni, JohnN/APhug, MariaN/APhares, DavidN/APhares, Charles $3.2; 3.6; 8.3$ Phillips, George $2.4; 2.13$ Phillips, George $2.4; 2.13$ Phillips, Thomas $2.4; 2.13$ Picchetti, Gloria $2.4; 2.13$ Picker, Seth $2.4; 2.13$ Picker, Seth $2.4; 2.13$ Picker, Seth $2.4; 2.13$ Piercer, Anden $2.4; 2.13$ Pierce, Arden $2.4; 2.13$ Pierce, Arden $2.4; 2.13$ Pierce, Arden $2.4; 2.13$ Pierson, James $2.4; 2.13$ Pierso		Perdios, Dan	2.4
Perkins, Guy 24, 2.13 Perry, Saul N/A Person, Wayne 24, 2.13 Perry, Saul N/A Peters, Kathleen 24, 2.13 Peters, Kathleen 24, 2.13 Peters, Robert N/A Peters, Robert N/A Peters, Robert N/A Peters, Jon N/A Pettosen, Jon N/A Pettosen, Jon N/A Pettosen, Jon N/A Pettosen, Jon N/A Philips, Gorge 24, 2.13 Philips, George 24, 2.13 Philips, Gorge 24, 2.13 Philips, Gorge 24, 2.13 Philips, Calapai, Jean 24, 2.13 Piccheri, Goria 24, 2.13 Piccheri, Seth 24, 2.13 Picker, Seth 24, 2.13 Picker, Seth 24, 2.13 Pierce, Arden 24,		Perez, Thomas	2.4; 2.13
Perkins, Marie24, 2.13Person, Wayne2.4, 2.13Petron, Wayne2.4, 2.13Peters, RobertNAPeters, RobertNAPeters, RobertNAPeters, RobertNAPeters, RobertNAPeters, JohnNAPetroni, JohnNAPetrigrew, JillNAPhug, MariaNAPhug, Maria2.4, 2.13Phug, Storege2.4, 2.13Philips, Charles2.4, 2.13Picker, Seth2.4, 2.13Picker, FayaNAPinda, FayeNAPinda, FayeNAPinda, FayeNAPiadoria, Stephen3.6 <td></td> <td>Perkins, Guy</td> <td>2.4; 2.13</td>		Perkins, Guy	2.4; 2.13
Perry, SaulNAPerson, Wayne2.4; 2.13Peters, Kathleen2.4; 2.13Peters, Kathleen2.4; 2.13Peters, RobertN/APetersen, Diane2.4; 2.13Petroni, JohnN/APetroni, JohnN/APetroni, JonnN/APetroni, JonnN/APetroni, JonnN/APetroni, JonnN/APetroni, JonnN/APetroni, JonnN/APetroni, JonnN/APhilips, Coorge2.4; 2.13Philips, George2.4; 2.13Philips, Coorge2.4; 2.13Philips, Coorge2.4; 2.13Philips, Clapia, Jean2.4; 2.13Piccinai, LaureenN/APicker, Seth2.4; 2.13Picker, Seth2.4; 2.13Picker, Seth2.4; 2.13Pierizek, Annette2.4; 2.13Pierizek, Annette2.4; 2.13Piere, Philip3.2; 3.6; 8.3Piercee, Arden2.4; 2.13Pier, Philip3.2; 3.6; 8.3Pierce, Arden2.4; 2.13Pierker, Syan1.1Pierker, Sayn1.2.1Pike, Ryan2.4Pindo, Juliann2.4Pindo, Juliann2.4Piske, Barbara2.4Pistas, HaroldN/APoderson, MyraN/APoderson, MyraN/APoland, DianneN/APollack, Michael2.4; 2.13Pollack, Michael2.4; 2.13Pollack, Michael2.4; 2.13P		Perkins, Marie	2.4; 2.13
Person, Wayne 2.4; 2.13 Peters, Rathleen 2.4; 2.13 Peters, Robert N/A Peters, Robert N/A Petersen, Diane 2.4; 2.13 Petroni, John N/A Petrosni, John N/A Philips, Aria N/A Phars, David N/A Phars, David N/A Phars, David N/A Phars, Charles 32; 3.6; 8.3 Phillips, Fatrick N/A Phillips, Fatrick N/A Phillips, Calapai, Jean 2.4; 2.13 Pricket, Gloria 2.4; 2.13 Pricket, Gloria 2.4; 2.13 Pricket, Gloria 2.4; 2.13 Pricket, Annette 2.4; 2.13 Pricket, Annette 2.4; 2.13 Prier, Philip 32; 3.6; 8.3 Prier, Paya N/A Prineda, Fave N/A		Perry, Saul	N/A
Peteriaraki, MariaN/APeters, Kathleen2.4; 2.13Peters, RobertN/APetersen, Diane2.4; 2.13Petroni, JohnN/APetrijorev, JillN/APetrijorev, JillN/APhares, DavidN/APhares, DavidN/APhilips, Cloarles2.4; 2.13Philips, Thomas2.4; 2.13Piccheri, Gloria2.4; 2.13Piccheri, Gloria2.4; 2.13Piccheri, Seth2.4; 2.13Picker, Seth2.4; 2.13Piecravi, LaureenN/APiere, Anden2.4; 2.13Piere, Anden2.4; 2.13Piere, Arden2.4; 2.13Piere, Arden2.4; 2.13Piere, Arden2.4; 2.13Piere, Ryan12.1Pillar, InaN/APinda, FayeN/APinda, Robert2.4Piske, Barbara2.42.42.13Piekar, Stephen3.6Pitagora, Robert3.6Pitagora, RobertN/APoland, DianneN/APoland, DianneN/APollack, Michael2.4; 2.13Pollack, Michael2.4; 2.13Pollack, Michael2.4; 2.13Pollack,		Person, Wayne	2.4; 2.13
Peters, Kathleen2.4; 2.13Petersen, Diane2.4; 2.13Petroni, JohnN/APettigrew, JillN/APutigrew, JillN/APhares, DavidN/APhares, DavidN/APhares, DavidN/APhares, DavidN/APhares, DavidN/APhares, Charles3.2; 3.6; 8.3Phillips, FatrickN/APhillips, FatrickN/APhillips, FatrickN/APhillips, FatrickN/APhillips, Calapai, Jean2.4; 2.13Piccheti, Gloria2.4; 2.13Piccheti, Gloria2.4; 2.13Piccheti, Gloria2.4; 2.13Pickert-Harner, MollyN/APiere, Seth2.4; 2.13Piere, Arden2.4; 2.13Piere, Arden2.4; 2.13Piere, Arden2.4; 2.13Piere, Arden2.4; 2.13Piere, Arden2.4; 2.13Piere, Ryan12.1Pillar, InaN/APinda, FayeN/APinda, Faye3.6Pithip3.6Pithon, Juliann2.4Pistorius, Stephen3.6Pitagora, Robert2.11; 2.13Piatora, Robert2.4; 2.13Pocius, F. JayN/APodad, DianneN/APoland, DianneN/APolack, Michael2.4; 2.13Polack, Michael2.4; 2.13Polack, RobertN/APolock, RobertN/APolock, RobertN/APolock, Carole <t< td=""><td></td><td>Peteinaraki, Maria</td><td>N/A</td></t<>		Peteinaraki, Maria	N/A
Peters, RobertN/APetersen, Diane2.4; 2.13Petroni, JohnN/APettigrew, JillN/APhity, MariaN/APhars, Charles3.2; 3.6; 8.3Phillips, George2.4; 2.13Phillips, George2.4; 2.13Phillips, Tomas2.4; 2.13Phillips, Calapai, Jean2.4; 2.13Phillips, Thomas2.4; 2.13Picciani, LaureenN/APicker, Seth2.4; 2.13Picker, Seth2.4; 2.13Picker, Seth2.4; 2.13Pientack, Annette2.4; 2.13Pierez, Arden2.4; 2.13Pierez, Arden2.4; 2.13Pieres, Annette2.4; 2.13Pieres, Annette2.4; 2.13Pieres, Annette2.4; 2.13Pieres, Ryan12.1Pierson, James2.4; 2.13Pierson, James2.4; 2.13Pierket, Barbara2.42.42.13Pinda, FayeN/APineda, FayeN/APineda, FayeN/APineda, FayeN/APineda, FayeN/APineda, FayeN/APindo, Julian2.4Pistorius, Stephen3.6Pitagora, Robert2.4; 2.13Poitagora, RobertN/APoderson, MyraN/APoderson, MyraN/APoland, DianneN/APolack, Michael2.4; 2.13Polack, Michael2.4; 2.13Polack, RichardN/APolek, Carole2.4; 2.13 <t< td=""><td></td><td>Peters, Kathleen</td><td>2.4; 2.13</td></t<>		Peters, Kathleen	2.4; 2.13
Petersen, Diane $2.4; 2.13$ Petrioni, JohnN/APettipone, JonN/APettigrew, JillN/APflug, MariaN/APhares, DavidN/APhares, DavidN/APharris, Charles $3.2; 3.6; 8.3$ Phillips, George $2.4; 2.13$ Phillips, Thomas $2.4; 2.13$ Phillips, Cloria $2.4; 2.13$ Picheti, Gloria $2.4; 2.13$ Piccheti, Gloria $2.4; 2.13$ Picker, Seth $2.4; 2.13$ Picker, Seth $2.4; 2.13$ Picker, Seth $2.4; 2.13$ Picker, Seth $2.4; 2.13$ Pier, Philip $3.2; 3.6; 8.3$ Piere, Arden $2.4; 2.13$ Piere, Raya 1.1 Pillar, Ina N/A Pincham, Debra 2.4 Pinto, Juliann 2.4 Piske, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Plat, David $2.4; 2.13$ Podorson, Myra N/A Poderson, Myra N/A Poderson, Myra N/A Polerka, Thomas $2.4; 2.13$ Polerka, Thomas $2.4; 2.13$ Poleka, Thomas $2.4; 2.13$ Poleka, Thomas $2.4; 2.13$ Poleka, Thomas $2.4; 2.13$ Poleka, Thomas $2.4; 2.13$ Pole		Peters, Robert	N/A
Petroni, JohnN/APettijone, JonN/APettijerw, JillN/APflug, MariaN/APhares, DavidN/APharris, Charles3,2; 3,6; 8,3Phillips, George2,4; 2,13Phillips, Thomas2,4; 2,13Phillips, Thomas2,4; 2,13Phillips, Thomas2,4; 2,13Picciani, LaureenN/APiccani, LaureenN/APicker, Seth2,4; 2,13Pierker, Annette2,4; 2,13Pierker, Seth2,4; 2,13Pierker, Seth2,4; 2,13Pierker, Seth2,4; 2,13Pierker, Seth2,4; 2,13Pierker, Seth2,4; 2,13Pierker, Annette2,4; 2,13Pierker, Annette2,4; 2,13Pierker, Ryan12,1Pilar, InaN/APinda, FayeN/APinker, Brabrana2,4Piske, Barbara2,4Pistorius, Stephen3,6Piagora, Robert2,11; 2,13Platas, HaroldN/APolad, DianneN/APolad, Nora2,4; 2,13Polefka, Thomas2,4; 2,13Pollack, RobertN/APo		Petersen, Diane	2.4; 2.13
Pettibone, JonN/APettigrew, JillN/APhares, DavidN/APhares, DavidN/APharrs, Charles 32 ; 3.6 ; 8.3 Phillips, George 2.4 ; 2.13 Phillips, George 2.4 ; 2.13 Phillips, Tornas 2.4 ; 2.13 Phillips, Calapai, Jean 2.4 ; 2.13 Picciani, LaureenN/APicer, Seth 2.4 ; 2.13 Picker, Seth 2.4 ; 2.13 Picer, Seth 2.4 ; 2.13 Picer, Annette 2.4 ; 2.13 Pier, Philip 3.2 ; 3.6 ; 8.3 Pierce, Arden 2.4 ; 2.13 Pierson, James 2.4 Pinda, FayeN/APineda, FayeN/APindag, FayeN/APindag, FayeN/APindagra, Robert 2.11 ; 2.13 Pistorius, Stephen 3.6 Piatgora, Robert $2.11; 2.13$ Polat, David 2.4 ; 2.13 Poderson, Myra N/A Polefka, Thomas 2.4 ; 2.13 Polack, Michael 2.4 ; 2.13 Polack, Michael 2.4 ; 2.13 Polack, Robert N/A Polefka, Thomas 2.4 ; 2.13 Polack, Richard N/A Poleck, Carole 2.4 ; 2.13 Polack, Richard N/A <td></td> <td>Petroni, John</td> <td>N/A</td>		Petroni, John	N/A
Pettigrew, JillN/APhug, MariaN/APhares, DavidN/APharris, Charles $3.2; 3.6; 8.3$ Phillips, George $2.4; 2.13$ Phillips, PatrickN/APhillips, Thomas $2.4; 2.13$ Phillips, Thomas $2.4; 2.13$ Piccheti, Gloria $2.4; 2.13$ Piccheti, Gloria $2.4; 2.13$ Piccheti, Gloria $2.4; 2.13$ Piccheti, Gloria $2.4; 2.13$ Piccheti, Ret $2.4; 2.13$ Pickett-Harner, MollyN/APienzek, Annette $2.4; 2.13$ Pierce, Arden $2.4; 2.13$ Pierce, Arden $2.4; 2.13$ Pierce, Arden $2.4; 2.13$ Pierce, Arden $2.4; 2.13$ Pierson, James $2.4; 2.13$ Pineda, FayeN/APinda, FayeN/APinto, Juliann 2.4 Pisko, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Platas, HaroldN/APoderson, MyraN/APoderson, MyraN/APolefka, Thomas $2.4; 2.13$ Poleka, FayeN/APoleka, FayaN/APoleka, FayaN/APoleka, FayaN/APoleka, FayaN/APoleka, FayaN/APoleka, Kora $2.4; 2.13$ Poleka, Kora $2.4; 2.13$ <		Pettibone, Jon	N/A
Pflug, MariaN/APhares, DavidN/APharris, Charles $32, 36, 8.3$ Phillips, George $24, 2.13$ Phillips, Thomas $24, 2.13$ Phillips, Thomas $24, 2.13$ Phillips, Calapai, Jean $24, 2.13$ Picchetti, Gloria $24, 2.13$ Picchetti, Gloria $24, 2.13$ Picker, Seth $24, 2.13$ Picker, Seth $24, 2.13$ Pier, Philips $32, 36, 8.3$ Piere, Arden $24, 2.13$ Pier, Philip $32, 36, 8.3$ Piere, Arden $24, 2.13$ Pier, Arden $24, 2.13$ Pier, Philip $32, 36, 8.3$ Piere, Arden $24, 2.13$ Pike, Ryan 12.1 Pillar, InaN/APineda, FayeN/APinkham, Debra 2.4 Piske, Barbara 2.4 Piske, Barbara 2.4 Piske, Barbara 2.4 Pistagora, Robert $2.11; 2.13$ Plastas, HaroldN/APodorson, MyraN/APoderka, Thomas $2.4; 2.13$ Poland, DianneN/APolerka, Thomas $2.4; 2.13$ Polack, Michael $2.4; 2.13$ Polack, Michael $2.4; 2.13$ Poleka, Thomas $2.4; 2.13$ Poleka, RobertN/APons, Scott $2.4; 2.13$ Poole, RichardN/APoole, Richard<		Pettigrew, Jill	N/A
Phares, DavidN/APharris, Charles $3.2; 3.6; 8.3$ Phillips, George $2.4; 2.13$ Phillips, George $2.4; 2.13$ Phillips, Thomas $2.4; 2.13$ Phillips-Calapai, Jean $2.4; 2.13$ Picchetti, Gloria $2.4; 2.13$ Picchetti, Gloria $2.4; 2.13$ Picchetti, Gloria $2.4; 2.13$ Picketr-Harner, MollyN/APieker, Seth $2.4; 2.13$ Pierce, Arden $2.4; 2.13$ Pierce, Arden $2.4; 2.13$ Pierson, James $2.4; 2.13$ Pierson, James $2.4; 2.13$ Pikek, Pyan12.1Piheda, FayeN/APineda, FayeN/APineda, FayeN/APinkam, Debra 2.4 Piske, Barbara 2.4 Piske, Barbara 2.4 Piske, Barbara 2.4 Piske, Barbara 2.4 Pisto, Julian 2.4 Pisto, Julian 2.4 Pisto, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Platas, HaroldN/APoder, G. $2.4; 2.13$ Polerka, Thomas $2.4; 2.13$ Polerka, Thomas $2.4; 2.13$ Polerka, Thomas $2.4; 2.13$ Polerka, Thomas $2.4; 2.13$ Polek, Kora $2.4; 2.13$ Polek, Kora $2.4; 2.13$ Polek, Kora $2.4; 2.13$ Polek, Kora $2.4; 2.13$ Poleka, Kora $2.4; 2.13$ Poleka, Kora $2.4; 2.13$ Polock, RobertN/APoo		Pflug, Maria	N/A
Pharris , Charles 3.2 ; 3.6 ; 8.3 Phillips, George 2.4 ; 2.13 Phillips, Thomas 2.4 ; 2.13 Phillips, Thomas 2.4 ; 2.13 Picchenti, Gloria 2.4 ; 2.13 Picchenti, Gloria 2.4 ; 2.13 Picchenti, LaureenN/APicker, Seth 2.4 ; 2.13 Pientaek, Annette 2.4 ; 2.13 Pier, Philip 3.2 ; 3.6 ; 8.3 Piere, Arden 2.4 ; 2.13 Pierson, James 2.4 ; 2.13 Pierson, James 2.4 ; 2.13 Pinkakn, Debra 2.1 Pinkakn, Debra 2.4 Pinto, Juliann 2.4 Pistorius, Stephen 3.6 Pitagora, Robert 2.11 ; 2.13 Plastas, HaroldN/APoduson, MyraN/APodorson, MyraN/APoleda, TayaN/APoland, Dianne 2.4 ; 2.13 Poland, DianneN/APoland, DianneN/APolenda, Nichael 2.4 ; 2.13 Poland, DianneN/APolenda, DianneN/APolenda, DianneN/APolenda, DianneN/APololock, RobertN/APolock, RobertN/APolock, RobertN/APolock, RobertN/APolock, RobertN/APolock, RobertN/APolock, RobertN/APolock, RobertN/APooler, Carole 2.4 ; 2.13 Polock, RobertN/APooler, Carole 2.4 ; 2.13 Pooler,		Phares, David	N/A
Phillips, George 2.4; 2.13 Phillips, Patrick N/A Phillips, Calapai, Jean 2.4; 2.13 Picchetti, Gloria 2.4; 2.13 Picchetti, Gloria 2.4; 2.13 Picchetti, Gloria 2.4; 2.13 Picchetti, Gloria 2.4; 2.13 Picker, Seth 2.4; 2.13 Picket-Harner, Molly N/A Piene, Annette 2.4; 2.13 Piere, Arden 2.4; 2.13 Piere, Arden 2.4; 2.13 Pieres, Faye N/A Pinda, Faye N/A Pinto, Juliann 2.4 Pistorius, Stephen 3.6 Pitagora, Robert 2.11; 2.13 Platt, David 2.4; 2.13 Podiad, Dianne N/A Poler, Tay N/A Polefka, Thomas		Pharris, Charles	3.2; 3.6; 8.3
Phillips, PatrickN/APhillips, Thomas $2.4; 2.13$ Phillips-Calapai, Jean $2.4; 2.13$ Picchetti, Gloria $2.4; 2.13$ Picchetti, Gloria $2.4; 2.13$ Picciani, LaureenN/APickett-Hamer, MollyN/APieniazek, Annette $2.4; 2.13$ Piere, Philip $3.2; 3.6; 8.3$ Pierce, Arden $2.4; 2.13$ Pierson, James $2.4; 2.13$ Piket, Ryan 12.1 Pillar, InaN/APinda, FayeN/APinken, Ryan 2.1 Pilke, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Platt, David $2.4; 2.13$ Pocius, F. JayN/APodorson, MyraN/APolefka, Thomas $2.4; 2.13$ Polek, Nora $2.4; 2.13$ Polek, Nora $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Poleka, Thomas $2.4; 2.13$ Poleka, Richael $2.4; 2.13$ Poleka, Richael $2.4; 2.13$ Poleka, Richael $2.4; 2.13$ Poleck, RobertN/APones, Scott $2.4; 2.13$ Pooler, RichardN/APooler, RichardN/APooler, RichardN/APooler, RichardN/A <td></td> <td>Phillips, George</td> <td>2.4; 2.13</td>		Phillips, George	2.4; 2.13
Phillips, Thomas $2.4; 2.13$ Phillips-Calapai, Jean $2.4; 2.13$ Picchati, Gloria $2.4; 2.13$ Picchati, LaureenN/APicker, Seth $2.4; 2.13$ Picktet-Harner, MollyN/APieniazek, Annette $2.4; 2.13$ Pier, Philip $3.2; 3.6; 8.3$ Pierce, Arden $2.4; 2.13$ Pirker, Ryan 12.1 Pineda, FayeN/APineda, FayeN/APinda, FayeN/APinto, Juliann 2.4 Piske, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Plat, David $2.4; 2.13$ Pocius, F. JayN/APolefsa, Thomas $2.4; 2.13$ Polefsa, Thomas $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollack, RobertN/APolefsa, Thomas $2.4; 2.13$ Pollack, RichardN/APoleck, RobertN/APones, Scott $2.4; 2.13$ Pollack, RichardN/APones, Carole $2.4; 2.13$ Poleck, Carole $2.4; 2.13$ Poleck, Carole $2.4; 2.13$		Phillips, Patrick	N/A
Phillips-Calapai, Jean $2.4; 2.13$ Picchetti, Gloria $2.4; 2.13$ Picchetti, Gloria $2.4; 2.13$ Picket, Seth $2.4; 2.13$ Picket, Seth $2.4; 2.13$ Picket-Harner, MollyN/APieniazek, Annette $2.4; 2.13$ Pierce, Arden $2.4; 2.13$ Pierce, Arden $2.4; 2.13$ Pike, Ryan12.1Pillar, InaN/APinda, FayeN/APinda, FayeN/APinto, Juliann 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Plats, HaroldN/APolatc, G. $2.4; 2.13$ Poletka, Thomas $2.4; 2.13$ Poletka, Thomas $2.4; 2.13$ Poletka, Thomas $2.4; 2.13$ Poletka, Thomas $2.4; 2.13$ Poletka, Michael $2.4; 2.13$ Poleka, Thomas $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollock, RobertN/APoleka, Thomas $2.4; 2.13$ Pollock, RobertN/APoleka, Thomas $2.4; 2.13$ Pollock, RobertN/APonele, RichardN/APonele, RichardN/APonele, Carole $2.4; 2.13$ Polole, RichardN/APonele, RichardN/APonele, Carole $2.4; 2.13$ Ponele, RichardN/APonele, RichardN/APonele, PriscillaN/A		Phillips, Thomas	2.4; 2.13
Picchetti, Gloria $2.4; 2.13$ Picciani, LaureenN/APicket, Seth $2.4; 2.13$ Picket-Harner, MollyN/APieniazek, Annette $2.4; 2.13$ Pier, Philip $3.2; 3.6; 8.3$ Pierce, Arden $2.4; 2.13$ Pierson, James $2.4; 2.13$ Piket, Panes $2.4; 2.13$ Piker, Ryan 12.1 Pillar, InaN/APincha, FayeN/APinto, Juliann 2.4 Piske, Barbara 2.4 Piske, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Platt, David $2.4; 2.13$ Pocius, F. JayN/APoderson, MyraN/APolefka, Thomas $2.4; 2.13$ Poland, DianneN/APolefka, Thomas $2.4; 2.13$ Pollack, Nichael $2.4; 2.13$ Pollack, RobertN/APolefka, Thomas $2.4; 2.13$ Pollack, Richael $2.4; 2.13$ Pooler, Carole $2.4; 2.13$ Pooler, Carole $2.4; 2.13$ Pooler, RichardN/APooler, PriscillaN/A		Phillips-Calapai, Jean	2.4; 2.13
Picciani, LaureenN/APicker, Seth $2.4; 2.13$ Pickter, Seth $2.4; 2.13$ Picktet-Harner, MollyN/APienizzek, Annette $2.4; 2.13$ Pier, Philip $3.2; 3.6; 8.3$ Pierce, Arden $2.4; 2.13$ Pierson, James $2.4; 2.13$ Pierson, James $2.4; 2.13$ Pierke, Ryan12.1Pinda, FayeN/APinda, FayeN/APinkham, Debra 2.4 Piske, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Plastas, HaroldN/APocius, F. JayN/APoderson, MyraN/APolefka, Thomas $2.4; 2.13$ Poland, Dianne $2.4; 2.13$ Polefka, Thomas $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollock, RobertN/APoelefka, Thomas $2.4; 2.13$ Pollock, RobertN/APonelefka, Thomas $2.4; 2.13$ Pollock, RobertN/APonelefka, Thomas $2.4; 2.13$ Pollock, RobertN/APonelef, Carole $2.4; 2.13$ Pollock, RobertN/APonelef, Carole $2.4; 2.13$ Pollock, RobertN/APonelef, Carole $2.4; 2.13$ Ponelef, Carole $2.4; 2.13$		Picchetti, Gloria	2.4; 2.13
Picker, Seth $2.4; 2.13$ Pickett-Harner, MollyN/APieniazek, Annette $2.4; 2.13$ Pier, Philip $3.2; 3.6; 8.3$ Pierce, Arden $2.4; 2.13$ Pierce, Arden $2.4; 2.13$ Pierson, James $2.4; 2.13$ Pike, Ryan12.1Pillar, InaN/APinda, FayeN/APinke, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Plastas, HaroldN/APodus, F. JayN/APodus, F. JayN/APolefka, Thomas $2.4; 2.13$ Polefka, Thomas $2.4; 2.13$ Polefka, Thomas $2.4; 2.13$ Polock, Robert N/A Polefka, Thomas $2.4; 2.13$ Poland, Dianne N/A Polefka, Thomas $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollack, Robert N/A Poeler, Carole $2.4; 2.13$ Polack, Richard N/A Ponoler, Carole $2.4; 2.13$ Polack, Polepe, Priscilla N/A		Picciani, Laureen	N/A
Pickett-Harner, MollyN/APieniazek, Annette $2.4; 2.13$ Pier, Philip $3.2; 3.6; 8.3$ Pierce, Arden $2.4; 2.13$ Pierson, James $2.4; 2.13$ Pike, Ryan 12.1 Pillar, InaN/APineda, FayeN/APinto, Juliann 2.4 Piske, Barbara 2.4 Piske, Barbara 2.4 Piske, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Platt, David $2.4; 2.13$ Pocius, F. JayN/APoderson, MyraN/APolefka, Thomas $2.4; 2.13$ Poleka, Thomas $2.4; 2.13$ Poleka, Michael $2.4; 2.13$ Poleka, Kora $2.4; 2.13$ Poleka, Thomas $2.4; 2.13$ Pollock, Robert N/A Poelne, Scott $2.4; 2.13$ Pollock, Robert N/A Poelper, Carole $2.4; 2.13$ Poolper, Carole $2.4; 2.13$ Poolper, Carole $2.4; 2.13$ Poolper, Priscilla N/A		Picker, Seth	2.4; 2.13
Pieniazek, Annette $2.4; 2.13$ Pier, Philip $3.2; 3.6; 8.3$ Pierce, Arden $2.4; 2.13$ Pierson, James $2.4; 2.13$ Pike, Ryan 12.1 Pillar, InaN/APineda, FayeN/APinto, Juliann 2.4 Piske, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Platas, HaroldN/APlatt, David $2.4; 2.13$ Pocius, F. JayN/APoderson, MyraN/APollad, DianneN/APollad, Dianne $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollack, RobertN/APolerka, Thomas $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollack, Crole $2.4; 2.13$ Pollack, Corele $2.4; 2.13$ Polerka, Thomas $2.4; 2.13$ Pollack, Corele $2.4; 2.13$ Polerka, Pone, Carole $2.4; 2.13$ Polerka Pone, Carole $2.4; 2.13$ Polerka Pone, PriscillaN/A		Pickett-Harner, Molly	N/A
Pier, Philip $3.2; 3.6; 8.3$ Pierce, Arden $2.4; 2.13$ Pierson, James $2.4; 2.13$ Pike, Ryan 12.1 Pillar, InaN/APineda, FayeN/APinkham, Debra 2.4 Pinto, Juliann 2.4 Piske, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Plastas, HaroldN/APlatta, David $2.4; 2.13$ Pocius, F. JayN/APodorson, MyraN/APolefka, Thomas $2.4; 2.13$ Polladk, Michael $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollock, RobertN/APoner, Carole $2.4; 2.13$ Pole, RichardN/APooler, Carole $2.4; 2.13$ Pooler, Carole $2.4; 2.13$ Pooler, Carole $2.4; 2.13$ Pooler, Carole $2.4; 2.13$		Pieniazek, Annette	2.4; 2.13
Pierce, Arden $2.4; 2.13$ Pierson, James $2.4; 2.13$ Pierson, James $2.4; 2.13$ Pike, Ryan 12.1 Pillar, InaN/APineda, FayeN/APinkbam, Debra 2.4 Pinto, Juliann 2.4 Piske, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Plastas, HaroldN/APocius, F. JayN/APoderson, MyraN/APogel, G. $2.4; 2.13$ Poland, DianneN/APolefka, Thomas $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollack, Mora $2.4; 2.13$ Pollock, RobertN/APone, Scott $2.4; 2.13$ Poleck, RohertN/APoel, RichardN/APoel, RichardN/APoel, RichardN/APooler, Carole $2.4; 2.13$ Pooler, Carole $2.4; 2.13$ Pooler, Carole $2.4; 2.13$		Pier, Philip	3.2; 3.6; 8.3
Pierson, James $2.4; 2.13$ Pike, Ryan 12.1 Pillar, InaN/APineda, FayeN/APinkham, Debra 2.4 Pinto, Juliann 2.4 Piske, Barbara 2.4 Piske, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert $2.11; 2.13$ Plastas, HaroldN/APlatt, David $2.4; 2.13$ Pocius, F. JayN/APogel, G. $2.4; 2.13$ Polack, Thomas $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollack, Michael $2.4; 2.13$ Pollack, RobertN/APonolack, RobertN/APole, RichardN/APooler, Carole $2.4; 2.13$ Pooler, PriscillaN/A		Pierce, Arden	2.4; 2.13
Pike, Ryan 12.1 Pillar, Ina N/A Pineda, Faye N/A Pineda, Faye N/A Pinkham, Debra 2.4 Pinto, Juliann 2.4 Piste, Barbara 2.4 Pistorius, Stephen 3.6 Pitagora, Robert 2.11; 2.13 Plastas, Harold N/A Platt, David 2.4; 2.13 Pocius, F. Jay N/A Pogel, G. 2.4; 2.13 Polarda, Dianne N/A Polefka, Thomas 2.4; 2.13 Pollack, Michael 2.4; 2.13 Pollack, Michael 2.4; 2.13 Pollack, Michael 2.4; 2.13 Poloeg, Richard N/A Pooler, Carole 2.4; 2.13 Pooler, Carole 2.4; 2.13 Pooler, Carole 2.4; 2.13		Pierson, James	2.4; 2.13
Pillar, InaN/APineda, FayeN/APinkham, Debra2.4Pinto, Juliann2.4Piske, Barbara2.4Pistorius, Stephen3.6Pitagora, Robert2.11; 2.13Plastas, HaroldN/APlatt, David2.4; 2.13Pocius, F. JayN/APogel, G.2.4; 2.13Polarda, Thomas2.4; 2.13Polarda, Thomas2.4; 2.13Polarda, Thomas2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APons, Scott2.4; 2.13Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Pike, Ryan	12.1
Pineda, FayeN/APinkham, Debra2.4Pinto, Juliann2.4Piske, Barbara2.4Pistorius, Stephen3.6Pitagora, Robert2.11; 2.13Plastas, HaroldN/APlatt, David2.4; 2.13Pocius, F. JayN/APodorson, MyraN/APolel, G.2.4; 2.13Polarka, Thomas2.4; 2.13Polada, DianneN/APolefka, Thomas2.4; 2.13Pollock, RobertN/APons, Scott2.4; 2.13Poloc, RichardN/APooler, Carole2.4; 2.13Poler, Carole2.4; 2.13Pope, PriscillaN/A		Pillar, Ina	N/A
Pinkham, Debra2.4Pinto, Juliann2.4Piske, Barbara2.4Pistorius, Stephen3.6Pitagora, Robert2.11; 2.13Plastas, HaroldN/APlatt, David2.4; 2.13Pocius, F. JayN/APodorson, MyraN/APogel, G.2.4; 2.13Poland, DianneN/APolefka, Thomas2.4; 2.13Pollack, Michael2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APoole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Pineda, Faye	N/A
Pinto, Juliann2.4Piske, Barbara2.4Pistorius, Stephen3.6Pitagora, Robert2.11; 2.13Plastas, HaroldN/APlatt, David2.4; 2.13Pocius, F. JayN/APodorson, MyraN/APogel, G.2.4; 2.13Poland, DianneN/APolefka, Thomas2.4; 2.13Polk, Nora2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APoole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Pinkham, Debra	2.4
Piske, Barbara2.4Pistorius, Stephen3.6Pitagora, Robert2.11; 2.13Plastas, HaroldN/APlatt, David2.4; 2.13Pocius, F. JayN/APodorson, MyraN/APogel, G.2.4; 2.13Poland, DianneN/APolefka, Thomas2.4; 2.13Polk, Nora2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APons, Scott2.4; 2.13Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Pinto, Juliann	2.4
Pistorius, Stephen3.6Pitagora, Robert2.11; 2.13Plastas, HaroldN/APlatt, David2.4; 2.13Pocius, F. JayN/APodorson, MyraN/APogel, G.2.4; 2.13Poland, DianneN/APolefka, Thomas2.4; 2.13Polk, Nora2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APons, Scott2.4; 2.13Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Piske, Barbara	2.4
Pitagora, Robert2.11; 2.13Plastas, HaroldN/APlatt, David2.4; 2.13Pocius, F. JayN/APodorson, MyraN/APogel, G.2.4; 2.13Poland, DianneN/APolefka, Thomas2.4; 2.13Polk, Nora2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APoole, RichardN/APooler, Carole2.4; 2.13Pooler, Carole2.4; 2.13Pope, PriscillaN/A		Pistorius, Stephen	3.6
Plastas, HaroldN/APlatt, David2.4; 2.13Pocius, F. JayN/APodorson, MyraN/APogel, G.2.4; 2.13Poland, DianneN/APolefka, Thomas2.4; 2.13Polk, Nora2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APoole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Pitagora, Robert	2.11; 2.13
Platt, David2.4; 2.13Pocius, F. JayN/APodorson, MyraN/APogel, G.2.4; 2.13Poland, DianneN/APolefka, Thomas2.4; 2.13Polk, Nora2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APoole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Plastas, Harold	N/A
Pocius, F. JayN/APodorson, MyraN/APogel, G.2.4; 2.13Poland, DianneN/APolefka, Thomas2.4; 2.13Polk, Nora2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APons, Scott2.4; 2.13Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Platt, David	2.4; 2.13
Podorson, MyraN/APogel, G.2.4; 2.13Poland, DianneN/APolefka, Thomas2.4; 2.13Polk, Nora2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APons, Scott2.4; 2.13Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Pocius, F. Jay	N/A
Pogel, G.2.4; 2.13Poland, DianneN/APolefka, Thomas2.4; 2.13Polk, Nora2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APons, Scott2.4; 2.13Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Podorson, Myra	N/A
Poland, DianneN/APolefka, Thomas2.4; 2.13Polk, Nora2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APons, Scott2.4; 2.13Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Pogel, G.	2.4; 2.13
Polefka, Thomas2.4; 2.13Polk, Nora2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APons, Scott2.4; 2.13Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Poland, Dianne	N/A
Polk, Nora2.4; 2.13Pollack, Michael2.4; 2.13Pollock, RobertN/APons, Scott2.4; 2.13Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Polefka, Thomas	2.4; 2.13
Pollack, Michael2.4; 2.13Pollock, RobertN/APons, Scott2.4; 2.13Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Polk, Nora	2.4; 2.13
Pollock, RobertN/APons, Scott2.4; 2.13Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Pollack, Michael	2.4; 2.13
Pons, Scott2.4; 2.13Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Pollock, Robert	N/A
Poole, RichardN/APooler, Carole2.4; 2.13Pope, PriscillaN/A		Pons, Scott	2.4; 2.13
Pooler, Carole2.4; 2.13Pope, PriscillaN/A		Poole, Richard	N/A
Pope, Priscilla N/A		Pooler, Carole	2.4; 2.13
	-	Pope, Priscilla	N/A

Pope, Susan	2.4; 2.13
Porter, Betsey	2.4; 2.13
Porter, Brian	2.4; 2.13
Porter, K.T.	2.4; 2.13
Porter, Sharon	N/A
Portney, Thomas	2.4; 2.13
Portnoy, Dennis	2.1; 2.2; 2.8; 2.14; 13.1
Potochnik, Andre	2.4; 2.11; 2.13; 2.16
Potter, Dave	2.4
Pottinger, Hans	N/A
Potucek, Kimberly	2.4: 2.13
Povill. Jon	2.4
Powell. Lance	2.4: 2.13
Powers Marlene	N/A
Powers Sheila	2 4 2 13
Prezant Jennifer	N/A
Price Charles	3 2: 3 6
Price Norman	N/A
Prochowski Richard	N/Δ
Proffitt Terry and Karen	$2 \wedge 2 13$
Proteon Mary	2.4, 2.13
Provence D	2.4, 2.15
Purcell Dougles	2.4
Purington II. Kon	2.4, 2.15 N/A
Putiligion II, Ken	IN/A
Pynn, Doug	N/A
Quammen, Parker	2.4; 2.13
Querner, Kathleen	2.4
R, Lynn	N/A
Raftery, Rita	N/A
Rahn, Gerald	N/A
Raitt, Jacob	N/A
Ralston, Aron	2.4; 2.13
Ramirez, Carina	N/A
Ramirez, Hank	2.4; 2.13
Ramo, Carol	2.4; 2.13
Ramos, Joann	2.4; 2.13
Randall, Dorene	N/A
Rangel, Louise	2.4; 2.13
Ranz, Lauren	2.4; 2.13
Raper, Connie	2.4; 2.13
Rappaport, Alexandra	2.4; 2.13
Rasmussen, Nancy	2.4; 2.13
Ratcliff, Steve	3.2; 3.6; 8.3
Ratcliff, Philip	2.4; 2.13
Ray, Jack	2.4; 2.13
Ray, Linda	2.4; 2.13
Ray, Mike	2.4; 2.13
Raychaudhuri, Sumana	2.4; 2.13
Re, Sa	2.4
Reames, Robin	2.4

Redstone, Ray	N/A
Redstrom, John	N/A
Reed, Dirk	N/A
Reed, Jennifer	12.8
Reeder, Ben	2.4; 2.11; 2.13
Reeder, William	N/A
Rees, Michael	2.4; 2.13
Reichel, Tom	N/A
Reid, Nancy	2.4; 2.13; 12.2
Reid, W.S.	3.2; 3.6; 8.3
Reilly, Holly	2.4
Reimer, Dana	12.2
Reinking, Tom	3.2; 3.6; 8.3
Reischl. Terri	N/A
Reisman, Emil	N/A
Reiter Jane	2.4
Reiter Steve (Old Pueblo Chapter of Trout Unlimited)	2 5 2 17 3 2 3 3 3 4 3 6 8 3 12 2
Renard Robert	3 2: 3 6
Rennacker Ann	24
Renton Kristen	2.1
Rever John	3.6
Revilla Oscar	2 4: 2 13
Revford Bridgett	N/A
Reynolds Ronda	2 <i>A</i> · 2 13
Reynolds, Ronda Rhodes, Janet	2.4, 2.13
Rhodes, Babert	2.4, 2.15
Rice Marybeth	2.4
Rice, Marybelli Pich Chris	2.4, 2.15
Rich, Chills Richardson, Cail	$2.4 \cdot 2.12$
Richardson James	2.4, 2.15
Richardson Matt	5.0 2 2 2 6 9 2 12 2
Richardson, Virginia	5.2, 5.0, 6.5, 12.2
Richardson, Virginia Dishay Charles	N/A
Richey, Charles Dislotta Corolum	2.4, 2.15
Ricketts, Carolyn Diddle, Carolyn	2.4, 2.15 N/A
Riddle, Carolyn Didanaur, Dad	N/A
Ridenour, Rod	2.4, 2.15
Rieck, Michael Diffle Law (Sente Derhans Elevenhore)	N/A
RITTle, Lew (Santa Barbara Flyfisners)	2.4 N/A
Riger, Richard	N/A
Riggs, Richard	2.4; 2.13
Rigney, J.	2.4; 2.13
Rimbeaux, B.	2.4
Ringgaard, Line	2.4; 2.13
Riopelle, Robert	2.4; 2.13
Kipp, Kudolph	2.4; 2.13
Kisley, Paul	N/A
Rist, Wally (Grand Canyon Private Boaters	2.4
Associations)	27/4
Rittenhouse, Nancy	N/A
Rivera, Sergio	2.4; 2.13

Roader, Jim 32: 36 Robertson, Gracie 24: 2.13 Roberts, Keith 32: 36, 8.3 Roberts, Keith 32: 36, 8.3 Roberts, Sally 24: 2.13 Roberts-Bharra, Suni N/A Roberts-Shepherd, Ruth 24: 2.13 Roberts-Shepherd, Ruth 24: 2.13 Roberts-Shepherd, Ruth 24: 2.13 Robinson, Craig N/A Robinson, Craig N/A Robinson, Joaneta 24: 2.13 Robinson, Don 24: 2.13 Robinson, Joyce 2.4 Robinson, Richard N/A Robinson, Richard N/A Robinson, Richard 24: 2.13 Roden, Karen N/A Roberty, Marge N/A Roberty, Marge N/A Roberty, Marge N/A Roberty, Braham 2.4 Roberty, Braham 2.4 Roberty, Braham 2.4 Roberty, Braham 2.4 Roll, Carol 2.4: 2.13 Romanowski, Scott 2.4: 2.13 Romanowski, Scott 2.4: 2.13		
Roberson, Gracie 2.4, 2.13 Roberts, Duane N/A Roberts, Keith 3.2, 3.6, 8.3 Roberts, Sally 2.4, 2.13 Roberts-Moneir, Nancy 2.4, 2.13 Roberts-Moneir, Nancy 2.4, 2.13 Roberts-Moneir, Nancy 2.4, 2.13 Robinson, Aaron 2.4, 2.13 Robinson, Craig N/A Robinson, Dameta 2.4, 2.13 Robinson, Dameta 2.4, 2.13 Robinson, Dameta 2.4, 2.13 Robinson, Joure 2.4 Robinson, Likard N/A Robinson, Richard N/A Robinson, Richard N/A Robenson, Richard N/A Robenson, Richard N/A Robenson, Richard N/A Robenson, Richard N/A Roebuck, Marge N/A Roebuck, Marge N/A Roebuck, Marge N/A Roebuck, Marge N/A Role, Atranham 2.4 Rolf, Carol 2.4, 2.13 Romain, Bella 2	Roache, Jim	3.2; 3.6
Roberts, Duane N/A Roberts, Sully $24; 2.13$ Roberts, Sally $24; 2.13$ Roberts-Ibarra, Suni N/A Roberts-Shepherd, Ruth $24; 2.13$ Robinson, Aaron $24; 2.13$ Robinson, Craig N/A Robinson, Dameta $24; 2.13$ Robinson, Don $24; 2.13$ Robinson, Jone $24; 2.13$ Robinson, Jon $24; 2.13$ Robinson, Jon $24; 2.13$ Robinson, Jon $24; 2.13$ Robinson, Joyce 24 Robinson, Richard N/A Robinson, Richard $24; 2.13$ Roden, Karen N/A Roberty, Drug $24; 2.13$ Roden, Karen N/A Roberty, Drug $24; 2.13$ Roden, Karen N/A Roberty, Drug $24; 2.13$ Roberty, Scott $24; 2.13$ Romano, David $32; 36; 8.3$ Romes, Mark $24; 2.13$ Romany, Bella $24; 2.13$ Romany, Bella $24; 2.13$ Romany, Mark $24; 2.13$ Romany, Mark $24; 2.13$ Romany, Berlen 24 Rosenblatt, Jon N/A Rosenblatt, Jon N/A <	Roberson, Gracie	2.4; 2.13
Robers, Keith $32; 36; 8.3$ Roberts, Sally $24; 2.13$ Roberts-Moneir, Nancy $24; 2.13$ Roberts-Moneir, Nancy $24; 2.13$ Robinson, Aaron $24; 2.13$ Robinson, Aaron $24; 2.13$ Robinson, CraigN/ARobinson, Dameta $24; 2.13$ Robinson, Dameta $24; 2.13$ Robinson, Dameta $24; 2.13$ Robinson, Dameta $24; 2.13$ Robinson, RichardN/ARobinson, RichardN/ARobinson, RichardN/ARoberts, KarenN/ARoberts, KarenN/ARoberts, Stellar, RameN/ARoberts, Stellar, RameN/ARoberts, Candace $24; 2.13$ Roden, KarenN/ARoberts, John $32; 36; 8.3$ Roder, KarenN/ARoberts, John $32; 36; 8.3$ Rollo, Pat $24; 2.13$ Romanowski, Scott $24; 2.13$ Roman, David $32; 36; 8.3$ Rood, Edson $32; 36; 8.3$ Rood, Edson $32; 36; 8.3$ Rood, Jamin 24 Rosenblatt, Jon $32; 36; 8.3$ Rood, Edson $32; 36; 8.3$ <t< td=""><td>Roberts, Duane</td><td>N/A</td></t<>	Roberts, Duane	N/A
Roberts, Sally $2,4,2.13$ Roberts-Shepherd, Ruth $2,4,2.13$ Roberts-Shepherd, Ruth $2,4,2.13$ Robinson, Aaron $2,4,2.13$ Robinson, CraigN/ARobinson, Don $2,4,2.13$ Robinson, Don $2,4,2.13$ Robinson, Don $2,4,2.13$ Robinson, Joyce $2,4$ Robinson, RichardN/ARobinson, Richard $2,4,2.13$ Robinson, Richard $2,4,2.13$ Roberts, Sally $2,4,2.13$ Roberts, Calace $2,4,2.13$ Roberts, Sally $2,4,2.13$ Roberts, Sally $2,4,2.13$ Roberts, Sally $2,4,2.13$ Roberts, Sally $2,4,2.13$ Roberts, LifthN/ARoberts, LifthN/ARoberts, LifthN/ARoberts, Carol $2,2,3,6,8.3$ Role, Abraham $2,4,2.13$ Roman, Bella $2,4,2.13$ Roman, Bella $2,4,2.13$ Romanowski, Scott $2,4,2.13$ Roman, Mark $2,4,2.13$ Roman, Mark $2,4,2.13$ Roman, Mark $2,2,3.6,8.3$ Robert, JoninN/ARosenblatt, JonN/ARosenblatt, JonN/ARosenblatt, JonN/ARosenblatt, JonN/ARosenblatt, JonN/ARosenblatt, JonN/ARosenblatt, Jon $2,4,2.13$	Roberts, Keith	3.2; 3.6; 8.3
Roberts-Ibarra, Suni N/A Roberts-Moneir, Nancy 2.4; 2.13 Roberts-Shepherd, Ruth 2.4; 2.13 Robinson, Craig N/A Robinson, Dameta 2.4; 2.13 Robinson, Dameta 2.4; 2.13 Robinson, Dameta 2.4; 2.13 Robinson, Dameta 2.4; 2.13 Robinson, Ellen R. N/A Robinson, Richard N/A Robinson, Richard 2.4 Robinson, Richard 2.4; 2.13 Rocha, Cancadace 2.4; 2.13 Rocha, Karen N/A Roberty, Drew 2.4; 2.13 Roden, Karen N/A Roberty, Drew 2.4; 2.13 Roenerburg, Drew 2.4; 2.13 Roenerburg, Drew 2.4; 2.13 Rohner, John 3.2; 3.6; 8.3 Role, Carol 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Romero, Monika 2.2; 3.6; 8.3 Rood, Fdson 3.2; 3.6; 8.3	Roberts, Sally	2.4; 2.13
Roberts-Moneir, Nancy $2.4; 2.13$ Roberts-Shepherd, Ruth $2.4; 2.13$ Robinson, CraigN/ARobinson, CraigN/ARobinson, Don $2.4; 2.13; 12.3$ Robinson, Don $2.4; 2.13; 12.3$ Robinson, Joyce 2.4 Robinson, Iaura $2.4; 2.13; 12.3$ Robinson, Iaura $2.4; 2.13; 12.3$ Robinson, Iaura $2.4; 2.13; 12.3$ Robinson, RichardN/ARobent, Candace $2.4; 2.13; 12.3$ Rocha, Candace $2.4; 2.13; 12.$	Roberts-Ibarra, Suni	N/A
Roberts-Shepherd, Ruth 2.4; 2.13 Robinson, Aaron 2.4; 2.13 Robinson, Craig N/A Robinson, Dameta 2.4; 2.13 Robinson, Don 2.4; 2.13; 12.3 Robinson, Joyce 2.4 Robinson, Iavra 2.4 Robinson, Richard N/A Robinson, Richard N/A Robinson, Richard N/A Rocha, Candace 2.4; 2.13 Rocha, Karen N/A Robeuck, Marge N/A Robenco, Karen N/A Robenck, Marge N/A Robenes, John 3.2; 3.6; 8.3 Rolf, Carol 2.4; 2.13 Romano, David 3.2; 3.6; 8.3 Romano, Bella 2.4; 2.13 Romes, Monika 2.4; 2.13 Romes, Monika 2.4; 2.13 Romero, Monika 2.4; 2.13	Roberts-Moneir, Nancy	2.4; 2.13
Robinson, Aaron $2.4; 2.13$ Robinson, Dameta $2.4; 2.13; 12.3$ Robinson, Don $2.4; 2.13; 12.3$ Robinson, Don $2.4; 2.13; 12.3$ Robinson, Joyce 2.4 Robinson, Joyce 2.4 Robinson, Richard $2.4; 2.13$ Robenson, Richard $2.4; 2.13$ Romanowski, Sout $2.4; 2.13$ Romesburg, Denise $2.4; 2.13$ Romesburg, Denise $2.4; 2.13$ Romesburg, Pat $3.2; 3.6; 8.3$ Rosenblatt, Jon N/A Rosenblatt, Jon N/A Rosenblatt, Jon $3.2; 3.6; 8.3$ Rosenstatt, William N/A Rosenstatt, William N/A Rosenstatt, William N/A Rosenstatt, William N/A Rosenstatt, William $2.4; 2.13$ Rosenstatt, William $2.$	Roberts-Shepherd, Ruth	2.4; 2.13
Robinson, Craig N/A Robinson, Dameta 2.4; 2.13 Robinson, Don 2.4; 2.13; 12.3 Robinson, Don 2.4; 2.13; 12.3 Robinson, Filen R. N/A Robinson, Richard N/A Robinson, Richard N/A Robinson, Richard N/A Robed, Karen N/A Roebuck, Marge N/A Roebuck, Marge N/A Robenson, John 2.4; 2.13 Roden, Karen N/A Robency, Drew 2.4; 2.13 Roden, Karen N/A Robency, Liltih N/A Robency, Laben 2.4; 2.13 Romano, David 3.2; 3.6; 8.3 Romano, David 3.2; 3.6; 8.3 Romanowski, Scott 2.4; 2.13 Roming, Mark 2.4; 2.13 Romesburg, Denise 2.4; 2.13 Romesburg, Denise 2.4; 2.13	Robinson, Aaron	2.4; 2.13
Robinson, Dameta 2.4; 2.13 Robinson, Ellen R. N/A Robinson, Ellen R. N/A Robinson, Richard 2.4 Robinson, Richard 2.4 Robinson, Richard 2.4; 2.13 Roden, Karen N/A Robenson, Richard 2.4; 2.13 Roden, Karen N/A Roebuck, Marge N/A Roden, Karen N/A Roeneburg, Drew 2.4; 2.13 Rogers, Liluth N/A Rohen, Carol 2.4; 2.13 Rogers, Liluth N/A Rohen, Arana 2.4 Rolf, Carol 2.4; 2.13 Roman, Bella 2.4; 2.13 Romano, David 3.2; 3.6; 8.3 Romanowski, Scott 2.4; 2.13 Romesburg, Denise 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Roman, Thomas 2.4 Romeou, Monika 2.4; 2.13 Roman, Thomas 2.4	Robinson, Craig	N/A
Robinson, Don $2.4; 2.13; 12.3$ Robinson, Ellen R.N/ARobinson, Joyce 2.4 Robinson, Jaura 2.4 Robinson, RichardN/ARocha, Candace $2.4; 2.13$ Rocha, Candace $2.4; 2.13$ Rocha, Candace $2.4; 2.13$ Rocha, KarenN/ARoegner, Debby $2.4; 2.13$ Romeburg, Drew $2.4; 2.13$ Romer, John $3.2; 3.6; 8.3$ Roll, Abraham 2.4 Roll, Abraham 2.4 Roll, Pat $2.4; 2.13$ Romanowski, Scott $2.4; 2.13$ Romano, David $3.2; 3.6; 8.3$ Romanowski, Scott $2.4; 2.13$ Rosenfold, Jamie	Robinson, Dameta	2.4; 2.13
Robinson, Joyce N/A Robinson, Joyce 2.4 Robinson, Laura 2.4 Robinson, Richard 2.4; 2.13 Rocha, Candace 2.4; 2.13 Roden, Karen N/A Robenson, Richard 2.4; 2.13 Roden, Karen N/A Roegner, Debby 2.4; 2.13 Rogers, Lilith N/A Rohnamer, John 3.2; 3.6; 8.3 Role, Abraham 2.4 Rolf, Carol 2.4; 2.13 Romano, David 3.2; 3.6; 8.3 Romano, Bella 2.4; 2.13 Romano, Bella 2.4; 2.13 Romano, Bayid 3.2; 3.6; 8.3 Romano, David 3.2; 3.6; 8.3 Romanowski, Scott 2.4; 2.13 Romesburg, Denise 2.4; 2.13 Romesburg, Denise 2.4; 2.13 Romano, Mark 2.4; 2.13 Romano, Thomas 2.4 Ronneburg, Pat 3.2; 3.6; 8.3 Rosenblatt, Jon N/A Rosenblood, Jamie 2.4 Rosenblood, Jamie	Robinson, Don	2.4; 2.13; 12.3
Robinson, Joyce 2.4 Robinson, Laura 2.4 Robinson, Richard N/A Robinson, Richard 2.4; 2.13 Roden, Candace 2.4; 2.13 Roden, Karen N/A Roegner, Debby 2.4; 2.13 Roerneburg, Drew 2.4; 2.13 Rogers, Lilith N/A Romeburg, Drew 2.4; 2.13 Rogers, Lilith N/A Rohmer, John 3.2; 3.6; 8.3 Rolf, Carol 2.4; 2.13 Romano, David 3.2; 3.6; 8.3 Romanowski, Scott 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Romero, Monika 2.4; 2.13 Romero, Monika 2.4; 2.13 Romesburg, Denise 2.4; 2.13 Romesburg, Denise 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Romano, Thomas 2.4 Rode, Edson 3.2; 3.6; 8.3 Rosenblatt, Jon N/A Rosenblaum, Stephen 2.4 Rosenblaum, St	Robinson, Ellen R.	N/A
Robinson, Laura2.4Robinson, RichardNARobinson, Richard2.4; 2.13Rocha, Candace2.4; 2.13Rocha, Candace2.4; 2.13Roden, KarenN/ARoebuck, MargeN/ARoemeburg, Debby2.4; 2.13Rogers, LilithN/ARobinson, Richard2.4Rogers, LilithN/ARohner, John3.2; 3.6; 8.3Role, Abraham2.4Rolf, Carol2.4; 2.13Romano, David2.4; 2.13Romano, St, Scott2.4; 2.13Romanowski, Scott2.4; 2.13Romesburg, Denise2.4; 2.13Romanowski, Scott2.4; 2.13Romano, David3.2; 3.6; 8.3Romero, Monika2.4; 2.13Romano, David3.2; 3.6; 8.3Romet, John3.2; 3.6; 8.3Romet, John3.2; 3.6; 8.3Rometor, John3.2; 3.6; 8.3Rometor, John3.2; 3.6; 8.3Rometor, Patl3.2; 3.6; 8.3Rood, Edson3.2; 3.6; 8.3Rosenblatt, JonN/ARosenblod, Jamie2.4Rosenblod, Jamie2.4Rosenstaft, WilliamN/ARosenstaft, WilliamN/ARosenstaft, WilliamN/ARosenstaft, WilliamN/ARosenstaft, William2.4; 2.13Rosenstaft, William2.4; 2.13Rosenstaft, WilliamN/ARosenstaft, WilliamN/ARosenstaft, William2.4; 2.13Rosenstaft, William2.4; 2.13 <td>Robinson, Jovce</td> <td>2.4</td>	Robinson, Jovce	2.4
Robinson, Richard N/A Robinson, Richard 2.4; 2.13 Rocha, Candace 2.4; 2.13 Roden, Karen N/A Roebuck, Marge N/A Roepuck, Marge N/A Roegner, Debby 2.4; 2.13 Rogens, Lilith N/A Rogers, Lilith N/A Rohmer, John 3.2; 36; 8.3 Role, Abraham 2.4 Rolf, Carol 2.4; 2.13 Roman, David 2.2; 3.6; 8.3 Romano, David 3.2; 3.6; 8.3 Romanowski, Scott 2.4; 2.13 Romero, Monika 2.4; 2.13 Romero, Monika 2.4; 2.13 Romero, Monika 2.4; 2.13 Romero, Monika 2.4; 2.13 Romes, Jacott 2.4; 2.13 Romig, Mark 2.4; 2.13 Ronan, Thomas 2.4 Rosenblat, Jon 3.2; 3.6; 8.3 Rosenbloto, Jamie 2.4 Rosenbloto, Jamie 2.4 Rosenfeld, Ryan 3.2; 3.6; 8.3 Rosenfeld, Ryan 3.2	Robinson, Laura	2.4
Robinson, Richard 2.4; 2.13 Rocha, Candace 2.4; 2.13 Roden, Karen N/A Roebuck, Marge N/A Roepuck, Marge N/A Romen, Debby 2.4; 2.13 Romano, David 3.2; 3.6; 8.3 Romano, David 3.2; 3.6; 8.3 Romanowski, Scott 2.4; 2.13 Romero, Monika 2.4; 2.13	Robinson, Richard	N/A
Rocha, Candace 2.4; 2.13 Roden, Karen N/A Roebuck, Marge N/A Roeper, Debby 2.4; 2.13 Roenneburg, Drew 2.4; 2.13 Rohmer, John 3.2; 3.6; 8.3 Role, Abraham 2.4 Rolf, Carol 2.4; 2.13 Roman, Bella 2.4; 2.13 Romano, David 3.2; 3.6; 8.3 Romanowski, Scott 2.4; 2.13 Rome, Monika 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Romanowski, Scott 2.4; 2.13 Rometo, Monika 2.4; 2.13 Rometo, Monika 2.4; 2.13 Rometo, Monika 2.4; 2.13 Rometo, Mark 2.4; 2.13 Rometo, Mark 2.4; 2.13 Rometo, Mark 2.4; 2.13 Romanowski, Scott 2.3; 3.6; 8.3 Rood, Edson 3.2; 3.6; 8.3 Rosen, Paul N/A Rosenbloud, Jamie 2.4 Rosenfolod, Jamie 2.4 Rosenfold, William N/A	Robinson Richard	2 4: 2 13
Roden, KarenNARoden, KarenNARoebuck, MargeN/ARoepne, Debby 2.4 ; 2.13 Rogenr, Debby 2.4 ; 2.13 Rogers, LilithN/ARohmer, John 3.2 ; 3.6 ; 8.3 Role, Abraham 2.4 Rolf, Carol 2.4 ; 2.13 Roman, Bella 2.4 ; 2.13 Romano, David 3.2 ; 3.6 ; 8.3 Romero, Monika 2.4 ; 2.13 Romesburg, Denise 2.4 ; 2.13 Romano, Mark 2.4 ; 2.13 Romano, Mark 2.4 ; 2.13 Romesburg, Denise 2.4 ; 2.13 Romano, Mark 2.4 ; 2.13 Romano, Mark 2.4 ; 2.13 Romesburg, Pati 3.2 ; 3.6 ; 8.3 Rosen, PaulN/ARosenblatt, JonN/ARosenblum, Stephen 2.4 Rosenblum, Step	Rocha Candace	2 4: 2 13
NA NA Roebuck, Marge N/A Roegner, Debby 24; 2.13 Roenneburg, Drew 24; 2.13 Roepers, Lilith N/A Rohmer, John 3.2; 3.6; 8.3 Role, Abraham 24 Rollo, Pat 24; 2.13 Romano, David 2.4; 2.13 Romano, David 3.2; 3.6; 8.3 Romano, David 3.2; 3.6; 8.3 Romesburg, Denise 2.4; 2.13 Romesburg, Denise 2.4; 2.13 Romesburg, Denise 2.4; 2.13 Romesburg, Denise 2.4; 2.13 Ronneburg, Pat 3.2; 3.6; 8.3 Ronneburg, Pat 3.2; 3.6; 8.3 Rosenblat, Jon N/A Rosenblut, Jon N/A Rosenblut, Jon N/A Rosenblum, Stephen 2.4 Rosenblum, Stephen 2.4 Rosenblum, Stephen 2.4; 2.13 Rosenblum, Stephen 2.4; 2.13 Rosenblum, Stephen 2.4; 2.13 Rosenblum, Stephen 2.4; 2.13 Roses, Jean	Roden Karen	N/A
Roegner, Debby 2.4; 2.13 Roegner, Debby 2.4; 2.13 Rogers, Lilith N/A Rohmer, John 3.2; 3.6; 8.3 Role, Abraham 2.4 Rolf, Carol 2.4; 2.13 Romain, Bella 2.4; 2.13 Romano, David 3.2; 3.6; 8.3 Romano, David 3.2; 3.6; 8.3 Romanowski, Scott 2.4; 2.13 Romero, Monika 2.4; 2.13 Romey, Denise 2.4; 2.13 Romey, Denise 2.4; 2.13 Romey, Denise 2.4; 2.13 Romesburg, Denise 2.4; 2.13 Romeburg, Pat 3.2; 3.6; 8.3 Rood, Edson 3.2; 3.6; 8.3 Rosenblud, Jamie 2.4 Rosenblod, Jamie 2.4 Rosenblod, Jamie 2.4 Rosenblud, Stephen 2.4 Rosenstadt, William N/A Roses, Adam 2.4; 2.13 Roses, S. Belliot 2.4; 2.13 Roses, Jean 2.4; 2.13 Roses, Jean 2.4; 2.13 Roser, Ellen 2.4; 2.13 Roser, Ellen 2.4; 2.13	Roebuck Marge	N/A
Roenneburg, Drew 2.4 ; 2.13Rogers, LilithN/ARohmer, John 3.2 ; 3.6 ; 8.3 Role, Abraham 2.4 Rolf, Carol 2.4 ; 2.13 Rollo, Pat 2.4 ; 2.13 Romano, David 3.2 ; 3.6 ; 8.3 Romanowski, Scott 2.4 ; 2.13 Romesburg, Denise 2.4 ; 2.13 Romesburg, Denise 2.4 ; 2.13 Romesburg, Denise 2.4 ; 2.13 Romesburg, Path 3.2 ; 3.6 ; 8.3 Romesburg, Path 3.2 ; 3.6 ; 8.3 Rosenblurg, Path 3.2 ; 3.6 ; 8.3 Rosenblurg, Path 3.2 ; 3.6 ; 8.3 Rosenblurd, JonnN/ARosenblurd, JonnN/ARosenblurd, Stephen 2.4 Rosenblurd, Stephen 2.4 Rosenstadt, WilliamN/ARosenstadt, William 2.4 ; 2.13 Roser, Ellen 2.4 ; 2.13 Rothschild, Louis 2.4 Rothschild, Louis 2.4	Roegner Debby	2 4 2 13
Rogers, Lilith N/A Rohmer, John 3.2 ; 3.6 ; 8.3 Role, Abraham 2.4 Rolf, Carol 2.4 ; 2.13 Rollo, Pat 2.4 ; 2.13 Romano, David 3.2 ; 3.6 ; 8.3 Romanowski, Scott 2.4 ; 2.13 Romero, Monika 2.4 ; 2.13 Romeso, Monika 2.4 ; 2.13 Romeso, Monika 2.4 ; 2.13 Romeso, Jose 2.4 ; 2.13 Romeso, Monika 2.4 ; 2.13 Romeso, Monika 2.4 ; 2.13 Romeso, Monika 2.4 ; 2.13 Romeso, Jose 2.4 ; 2.13 Romeso, Jame 2.4 Rosenblug, Pat 3.2 ; 3.6 ; 8.3 Rosen, PaulN/ARosenblut, JonN/ARosenblut, Stephen 2.4 Rosenstadt, WilliamN/ARosenstadt, WilliamN/ARosenstadt, William 2.4 ; 2.13 Roses, B. Elliot 2.4 ; 2.13 Rosser, Ellen 2.4 ; 2.13 Rotschild, Louis 2.4 Rothschild, Louis 2.4	Roenneburg. Drew	2.4: 2.13
Rolmer, John $3.2; 3.6; 8.3$ Role, Abraham 2.4 Rolf, Carol $2.4; 2.13$ Rollo, Pat $2.4; 2.13$ Romain, Bella $2.4; 2.13$ Romano, David $3.2; 3.6; 8.3$ Romanowski, Scott $2.4; 2.13$ Romero, Monika $2.4; 2.13$ Romesburg, Denise $2.4; 2.13$ Romanowski, Scott $2.4; 2.13$ Romesburg, Denise $2.4; 2.13$ Romesburg, Denise $2.4; 2.13$ Ronesburg, Denise $2.4; 2.13$ Ronesburg, Pati $3.2; 3.6; 8.3$ Rosen, PaulN/ARosenblod, Jamie 2.4 Rosenblod, Jamie 2.4 Rosenstadt, WilliamN/ARosest, Adam $2.4; 2.13$ Roses, Jean $2.4; 2.13$ Roses, Jean $2.4; 2.13$ Roses, Jean $2.4; 2.13$ Roses, Jean $2.4; 2.13$ Rosest, Ellen $2.4; 2.13$ Rosest, Jean $2.4; 2.13$ Rostschild, Louis 2.4 Rothschild, Louis 2.4 Rothschild, Louis 2.4	Rogers, Lilith	N/A
Role, Abraham2.4Rolf, Carol 2.4 ; 2.13Rollo, Pat 2.4 ; 2.13Romain, Bella 2.4 ; 2.13Romano, David 3.2 ; 3.6 ; 8.3 Romanowski, Scott 2.4 ; 2.13 Romero, Monika 2.4 ; 2.13 Romesburg, Denise 2.4 ; 2.13 Romesburg, Denise 2.4 ; 2.13 Ronnan, Thomas 2.4 Roneburg, Pat 3.2 ; 3.6 ; 8.3 Rosen, PaulN/ARosenblod, Jamie 2.4 Rosenblod, Jamie 2.4 Rosenfeld, Ryan 3.2 ; 3.6 ; 8.3 Roses, Adam 2.4 Roses, S. B. Elliot 2.4 ; 2.13 Ross, Jean 2.4 ; 2.13 Ross, Jean 2.4 ; 2.13 Ross, Jean 2.4 ; 2.13 Rosser, Ellen 2.4 ; 2.13 Rosser, Ellen 2.4 ; 2.13 Rotschild, Louis 2.4 Rothschild, Louis 2.4 Rothschild, Quinn 2.5	Rohmer, John	3.2: 3.6: 8.3
Rolf, Carol $2.4; 2.13$ Rollo, Pat $2.4; 2.13$ Romain, Bella $2.4; 2.13$ Romano, David $3.2; 3.6; 8.3$ Romanowski, Scott $2.4; 2.13$ Romero, Monika $2.4; 2.13$ Romero, Monika $2.4; 2.13$ Romero, Monika $2.4; 2.13$ Romesburg, Denise $2.4; 2.13$ Romesburg, Denise $2.4; 2.13$ Roman, Thomas 2.4 Rooneburg, Pat $3.2; 3.6; 8.3$ Rood, Edson $3.2; 3.6; 8.3$ Rosen, Paul N/A Rosenblatt, Jon N/A Rosenbloud, Jamie 2.4 Rosenbloud, Jamie 2.4 Rosenblum, Stephen 2.4 Rosenstadt, William N/A Roses, Adam $2.4; 2.13$ Ross, Jean $2.4; 2.13$ Rosser, Ellen $2.4; 2.13$ Rosser, Ellen $2.4; 2.13$ Rotser, Huu $2.4; 2.13$ Rotser, Len $2.4; 2.13$ Rotser, Lu $2.4; 2.13$ Rotser, Ellen $2.4; 2.13$ Rothschil	Role. Abraham	2.4
Rollo, Pat 2.4 ; 2.13 Romain, Bella 2.4 ; 2.13 Romano, David 3.2 ; 3.6 ; 8.3 Romanowski, Scott 2.4 ; 2.13 Romero, Monika 2.4 ; 2.13 Romesburg, Denise 2.4 ; 2.13 Roma, Thomas 2.4 ;Ronneburg, Pat 3.2 ; 3.6 ; 8.3 Roosenlug, Yat 3.2 ; 3.6 ; 8.3 Roosenlug, Jamie 2.4 Rosenblud, Jamie 2.4 Rosenblud, Jamie 2.4 Rosenstell, Ryan 3.2 ; 3.6 ; 8.3 Rosenstell, Ryan 3.2 ; 3.6 ; 8.3 Roses, B. Elliot 2.4 ; 2.13 Ross, Jean 2.4 ; 2.13 Rotschild, Lu 2.4 ; 2.13 Rothschild, Louis 2.4 Rothschild, Louis 2.4	Rolf. Carol	2.4: 2.13
Romain, Bella 2.4 ; 2.13 Romano, David 3.2 ; 3.6 ; 8.3 Romanowski, Scott 2.4 ; 2.13 Romero, Monika 2.4 ; 2.13 Romesburg, Denise 2.4 ; 2.13 Romig, Mark 2.4 ; 2.13 Ronan, Thomas 2.4 Ronneburg, Pat 3.2 ; 3.6 ; 8.3 Rosen, PaulN/ARosenblatt, JonN/ARosenblud, Jamie 2.4 Rosenblud, Stephen 2.4 Rosenstadt, WilliamN/ARoses, Adam 2.4 ; 2.13 Ross, B. Elliot 2.4 ; 2.13 Ross, Jean 2.4 ; 2.13 Rosser, Ellen 2.4 ; 2.13 Rotschild, Lu 2.4 ; 2.13 Rotschild, Louis 2.4 ; 2.13 Rothschild, Louis 2.4 ; 2.13 Rothschild, Quinn 2.5	Rollo, Pat	2.4; 2.13
Romano, David 3.2 ; 3.6 ; 8.3 Romanowski, Scott 2.4 ; 2.13 Romero, Monika 2.4 ; 2.13 Romesburg, Denise 2.4 ; 2.13 Romig, Mark 2.4 ; 2.13 Ronnan, Thomas 2.4 Ronneburg, Pat 3.2 ; 3.6 ; 8.3 Rood, Edson 3.2 ; 3.6 ; 8.3 Rosen, PaulN/ARosenblatt, JonN/ARosenblod, Jamie 2.4 Rosensteld, Ryan 3.2 ; 3.6 ; 8.3 Rosenstadt, WilliamN/ARoske, Adam 2.4 ; 2.13 Ross, B. Elliot 2.4 ; 2.13 Ross, Jean 2.4 ; 2.13 Rosser, Ellen 2.4 ; 2.13 Rotermund, Kristy 2.4 ; 2.13 Roth, Lu 2.4 ; 2.13 Rothschild, Louis 2.4 Rotschild, Quinn 2.5	Romain, Bella	2.4; 2.13
Romanowski, Scott 2.4 ; 2.13 Romero, Monika 2.4 ; 2.13 Romesburg, Denise 2.4 ; 2.13 Romig, Mark 2.4 ; 2.13 Ronan, Thomas 2.4 Ronneburg, Pat 3.2 ; 3.6 ; 8.3 Rood, Edson 3.2 ; 3.6 ; 8.3 Rosen, PaulN/ARosenblatt, JonN/ARosenbloud, Jamie 2.4 Rosenblum, Stephen 2.4 Rosenstadt, WilliamN/ARosenstadt, WilliamN/ARoses, Adam 2.4 ; 2.13 Ross, Jean 2.4 ; 2.13 Ross, Jean 2.4 ; 2.13 Rotermund, Kristy 2.4 ; 2.13 Rothschild, Louis 2.4 Rothschild, Louis 2.4 Rothschild, Quinn 2.5	Romano, David	3.2; 3.6; 8.3
Romero, Monika $2.4; 2.13$ Romesburg, Denise $2.4; 2.13$ Romig, Mark $2.4; 2.13$ Ronan, Thomas 2.4 Ronneburg, Pat $3.2; 3.6; 8.3$ Rood, Edson $3.2; 3.6; 8.3$ Rosen, PaulN/ARosenblatt, JonN/ARosenbloud, Jamie 2.4 Rosenblum, Stephen 2.4 Rosenfeld, Ryan $3.2; 3.6; 8.3$ Rosenstatt, WilliamN/ARoses, B. Elliot $2.4; 2.13$ Ross, B. Elliot $2.4; 2.13$ Rosser, Ellen $2.4; 2.13$ Roser, Lien $2.4; 2.13$ Rotermund, Kristy $2.4; 2.13$ Roth, Lu $2.4; 2.13$ Rothschild, Louis 2.4 Rothschild, Quinn 2.5	Romanowski, Scott	2.4; 2.13
Romesburg, Denise 2.4; 2.13 Romig, Mark 2.4; 2.13 Ronan, Thomas 2.4 Ronneburg, Pat 3.2; 3.6; 8.3 Rosen, Paul N/A Rosenblatt, Jon N/A Rosenblood, Jamie 2.4 Rosenbloum, Stephen 2.4 Rosenstadt, William N/A Roses, B. Elliot 2.4; 2.13 Ross, B. Elliot 2.4; 2.13 Rosser, Ellen 2.4; 2.13 Rotermund, Kristy 2.4; 2.13 Roth, Lu 2.4; 2.13 Rothschild, Louis 2.4; 2.13 Rothschild, Quinn 2.5	Romero, Monika	2.4; 2.13
Romig, Mark 2.4; 2.13 Ronan, Thomas 2.4 Ronneburg, Pat 3.2; 3.6; 8.3 Rood, Edson 3.2; 3.6; 8.3 Rosen, Paul N/A Rosenblatt, Jon N/A Rosenblod, Jamie 2.4 Rosenblod, Jamie 2.4 Rosenblum, Stephen 2.4 Rosenstadt, William N/A Roske, Adam 2.4; 2.13 Ross, B. Elliot 2.4; 2.13 Rosser, Ellen 2.4; 2.13 Roser, Ellen 2.4; 2.13 Rotermund, Kristy 2.4; 2.13 Roth, Lu 2.4; 2.13 Roth, Lu 2.4; 2.13 Rothschild, Louis 2.4 Rothschild, Louis 2.4	Romesburg, Denise	2.4; 2.13
Ronan, Thomas 2.4 Ronneburg, Pat $3.2; 3.6; 8.3$ Rood, Edson $3.2; 3.6; 8.3$ Rosen, PaulN/ARosenblatt, JonN/ARosenblood, Jamie 2.4 Rosenblum, Stephen 2.4 Rosenfeld, Ryan $3.2; 3.6; 8.3$ Rosenstadt, WilliamN/ARoske, Adam $2.4; 2.13$ Ross, B. Elliot $2.4; 2.13$ Ross, Jean $2.4; 2.13$ Rosser, Ellen $2.4; 2.13$ Rotermund, Kristy $2.4; 2.13$ Roth, Lu $2.4; 2.13$ Roth, Lu $2.4; 2.13$ Rothschild, Louis 2.4 Rothschild, Louis 2.4 Rothschild, Quinn 2.5	Romig, Mark	2.4; 2.13
Ronneburg, Pat $3.2; 3.6; 8.3$ Rood, Edson $3.2; 3.6; 8.3$ Rosen, PaulN/ARosenblatt, JonN/ARosenblood, Jamie 2.4 Rosenblum, Stephen 2.4 Rosenfeld, Ryan $3.2; 3.6; 8.3$ Rosenstadt, WilliamN/ARoske, Adam $2.4; 2.13$ Ross, B. Elliot $2.4; 2.13$ Ross, Jean $2.4; 2.13$ Rosser, Ellen $2.4; 2.13$ Rotermund, Kristy $2.4; 2.13$ Roth, Lu $2.4; 2.13$ Rothschild, Louis 2.4 Rothschild, Quinn 2.5	Ronan, Thomas	2.4
Rood, Edson3.2; 3.6; 8.3Rosen, PaulN/ARosenblatt, JonN/ARosenblood, Jamie2.4Rosenblum, Stephen2.4Rosenfeld, Ryan3.2; 3.6; 8.3Rosenstadt, WilliamN/ARoske, Adam2.4; 2.13Ross, B. Elliot2.4; 2.13Ross, Jean2.4; 2.13Rosser, Ellen2.4; 2.13Rotermund, Kristy2.4; 2.13Roth, Lu2.4; 2.13Rothschild, Louis2.4Rothschild, Quinn2.5	Ronneburg, Pat	3.2; 3.6; 8.3
Rosen, PaulN/ARosenblatt, JonN/ARosenblod, Jamie2.4Rosenblum, Stephen2.4Rosenfeld, Ryan3.2; 3.6; 8.3Rosenstadt, WilliamN/ARoske, Adam2.4; 2.13Ross, B. Elliot2.4; 2.13Ross, Jean2.4; 2.13Rosser, Ellen2.4; 2.13Rotermund, Kristy2.4; 2.13Roth, Lu2.4; 2.13Rothschild, Louis2.4Rothschild, Quinn2.5	Rood, Edson	3.2; 3.6; 8.3
Rosenblatt, JonN/ARosenblood, Jamie2.4Rosenblum, Stephen2.4Rosenfeld, Ryan3.2; 3.6; 8.3Rosenstadt, WilliamN/ARoske, Adam2.4; 2.13Ross, B. Elliot2.4; 2.13Ross, Jean2.4; 2.13Rosser, Ellen2.4; 2.13Rothschild, Louis2.4; 2.13Rothschild, Louis2.4; 2.13Rothschild, Quinn2.5	Rosen, Paul	N/A
Rosenblood, Jamie2.4Rosenblum, Stephen2.4Rosenfeld, Ryan3.2; 3.6; 8.3Rosenstadt, WilliamN/ARoske, Adam2.4; 2.13Ross, B. Elliot2.4; 2.13Ross, Jean2.4; 2.13Rosser, Ellen2.4; 2.13Rotermund, Kristy2.4; 2.13Roth, Lu2.4; 2.13Rothschild, Louis2.4Rothschild, Quinn2.5	Rosenblatt, Jon	N/A
Rosenblum, Stephen2.4Rosenfeld, Ryan3.2; 3.6; 8.3Rosenstadt, WilliamN/ARoske, Adam2.4; 2.13Ross, B. Elliot2.4; 2.13Ross, Jean2.4; 2.13Rosser, Ellen2.4; 2.13Rotermund, Kristy2.4; 2.13Roth, Lu2.4; 2.13Rothschild, Louis2.4Rothschild, Quinn2.5	Rosenblood, Jamie	2.4
Rosenfeld, Ryan3.2; 3.6; 8.3Rosenstadt, WilliamN/ARoske, Adam2.4; 2.13Ross, B. Elliot2.4; 2.13Ross, Jean2.4; 2.13Rosser, Ellen2.4; 2.13Rotermund, Kristy2.4; 2.13Roth, Lu2.4; 2.13Rothschild, Louis2.4Rothschild, Quinn2.5	Rosenblum, Stephen	2.4
Rosenstadt, WilliamN/ARoske, Adam2.4; 2.13Ross, B. Elliot2.4; 2.13Ross, Jean2.4; 2.13Rosser, Ellen2.4; 2.13Rotermund, Kristy2.4; 2.13Roth, Lu2.4; 2.13Rothschild, Louis2.4Rothschild, Quinn2.5	Rosenfeld, Ryan	3.2; 3.6; 8.3
Roske, Adam 2.4; 2.13 Ross, B. Elliot 2.4; 2.13 Ross, Jean 2.4; 2.13 Rosser, Ellen 2.4; 2.13 Rotermund, Kristy 2.4; 2.13 Roth, Lu 2.4; 2.13 Rothschild, Louis 2.4; 2.13 Rothschild, Louis 2.4 Rothschild, Quinn 2.5	Rosenstadt, William	N/A
Ross, B. Elliot 2.4; 2.13 Ross, Jean 2.4; 2.13 Rosser, Ellen 2.4; 2.13 Rotermund, Kristy 2.4; 2.13 Roth, Lu 2.4; 2.13 Rothschild, Louis 2.4; 2.13 Rothschild, Louis 2.4 Rothschild, Quinn 2.5	Roske, Adam	2.4; 2.13
Ross, Jean 2.4; 2.13 Rosser, Ellen 2.4; 2.13 Rotermund, Kristy 2.4; 2.13 Roth, Lu 2.4; 2.13 Rothschild, Louis 2.4 Rothschild, Quinn 2.5	Ross, B. Elliot	2.4; 2.13
Rosser, Ellen2.4; 2.13Rotermund, Kristy2.4; 2.13Roth, Lu2.4; 2.13Rothschild, Louis2.4Rothschild, Quinn2.5	Ross, Jean	2.4; 2.13
Rotermund, Kristy2.4; 2.13Roth, Lu2.4; 2.13Rothschild, Louis2.4Rothschild, Quinn2.5	Rosser, Ellen	2.4; 2.13
Roth, Lu2.4; 2.13Rothschild, Louis2.4Rothschild, Quinn2.5	Rotermund, Kristy	2.4; 2.13
Rothschild, Louis2.4Rothschild, Quinn2.5	Roth, Lu	2.4; 2.13
Rothschild, Quinn 2.5	Rothschild, Louis	2.4
	Rothschild, Quinn	2.5

Rothschild, Teal	N/A
Rowe, Spencer	2.1; 2.2; 2.8; 2.14; 13.1
Rowe, William	N/A
Rowell, Patricia	2.4
Rowitz, Roger	2.4; 2.13
Royer, Allen	2.4; 2.13
Rubenstein, Harvey	2.4; 2.13
Rubino, Karen	N/A
Rubino, Vincent	2.4; 2.13
Ruhl, John	2.4; 2.13
Ruiz, Mabel	N/A
Ruiz, O.	2.4; 2.13
Rutkowski, Robert	2.1; 2.2; 2.4; 2.8; 2.13; 2.14; 13.1
Ryaby, James	3.2; 3.6; 8.3
Ryan, George	2.4; 2.13
Ryan, Jack	2.5
Ryan, Joanne	2.4; 2.13
Ryan, Patricia	2.4
Ryerson, William	2.4; 2.13
Sacherer, Janice	2.4; 2.13
Sadowskas, Bruce	2.4; 2.13
Sailer, Randy	2.4
Salazar, Gladys	2.4; 2.13
Salomone, Michael	N/A
Samelson, Audrey	2.4; 2.13
Samoylo, Charles	N/A
Samp, Cecelia	2.4
Sanchez, Ralph	2.4; 2.13
Sanders, Raymond	3.2; 3.6; 8.3
Sanders, David	2.4
Sangster, Carol	2.4; 2.13
Santi, Giri	2.4; 2.13
Sargent, Jackie (Platte River Power Authority)	2.1; 2.3; 2.6; 2.9; 2.10; 2.13; 2.19; 8.9; 8.10
Sarracino, Brendon	3.2; 3.6; 8.3
Sarraille, Marijeannee	2.4; 2.13
Sass, Sherry	2.4; 2.13
Sauerman, Jacqueline	2.4; 2.13
Sawall, Erwin F.	2.8
Sawyer, Mary	2.4; 2.13
Saxon, Diana	2.4; 2.13
Sayler, Becky	2.4; 2.13
Scar, Dick	N/A
Scarpace, Leonard	3.2; 3.6; 8.3
Scarpelli, Kenneth	2.4; 2.13
Scarry, Patrick	2.4; 2.13
Scavezze, Barb	2.4; 2.8; 2.13
Schade, Corey	2.4; 2.13
Schafer, Maggie	N/A
Schafer, Peter	2.4; 2.13
Schaming, Carol	2.4; 2.13

Scharin, Lisa	N/A
Scherer, W.	2.4; 2.13
Schetzer, Kathryn	2.4; 2.13
Schindele, Paulette	2.4; 2.13
Schindler, Maury	2.4; 2.13
Schlein, Elizabeth	N/A
Schlesinger, Sybil	2.4; 2.13
Schmid, Genevieve	2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
Schmidt, Ted	N/A
Schmidt, Wayne	2.4
Schmitt, Eileen	N/A
Schmitt, Jeff	2.5; 2.17; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
Schmittauer, John	N/A
Schneebeli, Christiane	N/A
Schneider Edward	2 4 2 13
Schneider L	2 4 2 13
Schneller Douglas	2 4 2 13
Schoech D	2 4 2 13
Scholl Barbara	N/A
Scholl Susan	N/A
Schreier Bryna	N/A
Schultz Deter	2 4
Schultz, William	2. 7 2 <i>A</i> · 2 13
Schunz, William	2.4, 2.13
Schupp, Norma	2.4, 2.13
Schwartz, Aligera	2.4, 2.15
Schwartz, Dehart	2.4
Schwartzborg Lorg	2.4, 2.15
Seilluffe Ice	2.4
Scinuito, joe	2.4, 2.15
Scopes, Leslie	11.5
Scott, Genel	2.4, 2.15
Scott, Carol	N/A
Scott, Marl	2.4, 2.13
Scott, Mark	N/A 2.4: 2.12
Scott, Kaeann	2.4; 2.13
Scranton, Chris	2.4; 2.13
Seals, Suzanne	2.4; 2.13
Sealy, Berenice	2.4; 2.13
Searles, Dave	2.4; 2.13
Season, Ron	2.4; 2.13
Sebastian, Scott	N/A
Sedlock, Evan	3.2; 3.6; 8.3
Sees, Heather	3.2; 3.6; 8.3
Seidner, Dan	12.4
Selbin, Susan	2.4; 2.13
Selig, Stephanie	12.3
Sell, David	2.1; 2.2; 2.8; 2.14; 13.1
Seltzer, Elizabeth	N/A
Seltzer, Rob	2.4; 2.13
Sennett, Frank	N/A

Sensing, Wanda	2.4; 2.13
Serafino, Stephen	N/A
Sergio, Janette	2.4; 2.13
Severson, Dona	2.4
Sewald, Michelle	2.4; 2.13
Seymour, Linda	2.4; 2.13
Shaffer, Susan	2.4; 2.13
Shapiro, Jane	N/A
Sharfman, William	2.4; 2.13
Sharpton, Debra (Southwest Council International	N/A
Federation of Flyfishers and the Sierra Pacific	
Flyfishers)	
Shaw, Lori	N/A
Shear, Julie	2.4; 2.13
Sheets-Johnstone, Maxine	N/A
Sheetz, Jamie	3.2; 3.6; 8.3
Sheetz, Jennifer	2.4
Sheffer, Jeanne	2.4; 2.13
Shelley, Martha	2.4; 2.13
Sheridan, Michelle	2.4; 2.13
Sherman, David	2.4; 2.13
Sherman, Marcia	N/A
Sherman, Rick	2.4; 2.13
Sherwood, Kate	2.4
Shields, Michele	2.4: 2.13
Shiffrin, Joyce	N/A
Shirley, Cameron	3.2: 3.6: 8.3
Shitama, Celeste	2.4; 2.13
Shogren, Martha	N/A
Short, John	2.4
Shouse, Corbin	2.4; 2.13
Shultz, Betty Jane	N/A
Shushan, Cheryl	N/A
Sicklesteel, Cory	8.3
Sidwell, Sarah	2.8
Siegfried, Daniel	3.2; 3.6; 8.3
Siegwald, Joan	2.4; 2.13
Sifuentes, D.G.	2.4; 2.13
Sikand Vikram	2.4: 2.13
Silver. Ron	2.4: 2.13
Silver, Victoria	N/A
Simle, Anna	2.4: 2.13
Simmerman, Scott	N/A
Simmons, Adrienne	2.4: 2.13
Simmons, Chuck	2.4
Simons Anita	2 4 2 13
Singleton Jon	2 4: 2 13
Sinnott Larkin	N/A
Sinden Fric	N/A
Skaggs Brian	2 4
Jru665, D11011	4. T

Change des Deurses	2.4.2.12
Skarada, Darcy	2.4, 2.15
Skiendzielewski, Danielle	
Skinner, Russell	N/A
Skipworth, Carl	2.4; 2.13
Skirbunt-Kozabo, William	2.4; 2.13
Skow, Loren	N/A
Skrzypczak, Lida	
Skuce, David	3.2; 3.6; 8.3
Skup, Debra	2.4; 2.13
Slater, Dan	3.2; 3.6
Slaughter, Angela (Colorado River Commission of	4.1.3; 5.2; 6.4.14; 6.5.1; 6.7; 8.12; 13.3; 13.4
Nevada)	0.4.0.10
Sloat, Dale	2.4; 2.13
Sloat, Jan	2.4; 2.13
Slote, Karen	2.4; 2.13
Sluis, Janet	3.2; 3.6; 8.3
Smarr, Janet	2.4; 2.13
Smart, David	3.2; 3.6; 8.3
Smereck, Amy	2.4
Smetana, Carol	2.4; 2.13
Smith, Bryan	3.2; 3.6; 8.3
Smith, April	2.4; 2.13
Smith, Bradley	2.4; 2.13
Smith, Dea	N/A
Smith, Donald	3.2; 3.6; 8.3
Smith, Fendrick	2.4; 2.13
Smith, G. Austin	2.4; 2.13
Smith, Jean	2.4; 2.13
Smith, Judith	2.4
Smith, Latimer	2.12
Smith, Lib	N/A
Smith, Mack	2.4; 2.13
Smith, Mary Jordan	N/A
Smith, Mollie	N/A
Smith, Moreland	2.4; 2.13
Smith, Neill	N/A
Smith, Phyllis	N/A
Smith, Roger	2.4; 2.13
Smith, Sean	8.3
Smock, Amanda	2.4
Smoot, Leslie	2.4; 2.13
Smudin, Carole	N/A
Sneath, Barbara	2.4; 2.13
Snider, Jay	2.4; 2.13
Snow, N/A	N/A
Snyder, Cindy	2.4; 2.13
Snyder, Joanne	N/A
Snyder, Paul, Jr	3.2; 3.6; 8.3
Snyder, Robert	2.4; 2.13
Sobanski, Sandra	2.4; 2.13

Sodes, Michael $2.4; 2.13$ Solomon, DavidN/ASolomon, DavidN/ASolomon, DavidN/ASolomon, David $2.6; 2.7; 2.10; 2.12; 2.21; 8.1$ Somers, Jeff $2.4; 2.13$ Sorensen, Colleen $2.4; 2.13$ Sorensen, Collen $2.4; 2.13$ Sorensen, Collen $2.4; 2.13$ Sorensen, Collen $2.4; 2.13$ Soraz, P. 2.8 Spangle, Steven (U.S. Fish and Wildlife Service) $2.2; 2.9; 3.2; 3.6; 3.7; 3.8; 8.12; 8.13; 11.2$ Sparger, Janet $2.4; 2.13$ Sparks, Alan 3.6 Sparker, Mark $2.4; 2.13$ Speer, Scott $2.4; 2.13$ Speer, Mark $3.2; 3.6; 8.3$ Spenet, Frett $2.4; 2.13$ Speer, Mark $3.2; 3.6; 8.3$ Spenet, Frett $2.4; 2.13$ Spillane, Rita $2.2; 3.6; 8.3$ Spillane, Rita $2.4; 2.13$ Standig, Joseph $1.2; 2.3; 2.3; 8; 8.3; 3.5; 3.5; 3.5; 3.5; 3.5; 3.5; 3.5; 3$		
Solomon, DavidN/ASolomon, DavidN/ASolomon, DavidN/ASolomon, DavidN/ASourers, Leff $2.6; 2.7; 2.10; 2.12; 2.21; 8.1$ Sorekrasz, David $3.2; 3.6$ Sorrum, Robert $2.4; 2.13$ Souza, P. 2.8 Spangle, Steven (U.S. Fish and Wildlife Service) $2.2; 2.9; 3.2; 3.6; 3.7; 3.8; 8.12; 8.13; 11.2$ Sparger, Janet 2.8 Sparks, Alan 3.6 Sparuw, Deb $2.4; 2.13$ Spauding, D.N/ASpeakman, DavidN/ASpeakman, DavidN/ASpeakman, DavidN/ASpeakman, DavidN/ASpeakman, DavidN/ASpeakman, DavidN/ASpeakman, DavidN/ASpeakman, DavidN/ASpeakman, DavidN/ASpeek, Karen $2.4; 2.13$ Spiegel, Karen $2.4; 2.13$ Spiegel, Karen $2.4; 2.13$ Spiidlen, Rita 2.4 Spiraki, Nike $3.2; 3.6; 8.3$ Spere, Liter 2.4 Spiraki, Steven $2.4; 2.13$ Spilane, Rita 2.4 Spiraki, JennN/ASpondi, Nancy 2.4 Spiraki, JennN/ASpiraki, JennN/ASpiraki, JennN/ASpiraki, Jenn $2.4; 2.13$ Stadnik, George $2.4; 2.13$ Stadnik, George $2.4; 2.13$ Stankey, RonN/AStankey, Ron $3.2; 3.6; 8.3$ Stankey, Roh $3.2; 3.6; 8.3$ Stee	Sodos, Michael	2.4; 2.13
Solomon, DavidNASolomon, Dillip (Deseret Power Electric Cooperative) $2.6; 2.7; 2.10; 2.12; 2.21; 8.1$ Solomon, Mary $2.4; 2.13$ Sorensen, Colleen $2.4; 2.13$ Sorensen, Colleen $2.4; 2.13$ Soranz, Robert $2.4; 2.13$ Souza, P. $2.4; 2.13$ Sparger, Janet $2.4; 2.13$ Sparger, Janet $2.8; 2.9; 3.2; 3.6; 3.7; 3.8; 8.12; 8.13; 11.2$ Sparger, Janet $2.8; 2.9; 3.2; 3.6; 3.7; 3.8; 8.12; 8.13; 11.2$ Sparger, Janet $2.4; 2.13$ Sparder, Janet $2.4; 2.13$ Spaulding, D.N/ASpecies, Scott $2.4; 2.13$ Species, Scott $2.4; 2.13$ Speigel, Karen $2.4; 2.13$ Spilker, Mike $3.2; 3.6; 8.3$ Spilker, Mike $3.2; 3.6; 8.3$ Spilker, Mike $2.4; 2.13$ Speignellane, Rita 2.4 Spinolli, Nancy 2.4 Sprains, John $2.4; 2.13$ St. Angelo, Richard 2.4 Stanley, Joseph 12.2 Stan, TailaN/AStanley, Edwin RN/AStanley, LauraN/AStanley, Sorgh $2.4; 2.13$ Starley, Gaylord (Canyoneers, Inc.) $5.2; 12.4$ <td>Solomon, David</td> <td>N/A</td>	Solomon, David	N/A
Solomon, Phillip (Descret Power Electric Cooperative) $2, 6, 27, 2.10, 2.12, 2.21, 8.1$ Solum, Mary $2.4, 2.13$ Sorners, Jeff $2.4, 2.13$ Sorners, Colleen $2.4, 2.13$ Sorners, David $3.2, 3.6$ Sorum, Robert $2.4, 2.13$ Souza, P. 2.8 Spangle, Steven (U.S. Fish and Wildlife Service) $2.2, 2.9, 3.2, 3.6, 3.7, 3.8, 8.12, 8.13, 11.2$ Sparger, Janet 2.8 Sparks, Alan 3.6 Sparrow, Deb $2.4, 2.13$ Sparks, Alan 3.6 Spearkman, DavidN/ASpeer, Mark $3.2, 3.6, 8.3$ Speer, Mark $3.2, 3.6, 8.3$ Spiegel, Karen $2.4, 2.13$ Spiegel, Karen $2.4, 2.13$ Spiegel, Karen $2.4, 2.13$ Spillen, Mike $3.2, 3.6, 8.3$ Spillane, Rita 2.4 Spillong, Naccy 2.4 Spillong, Naccy 2.4 Spillong, Richard $2.4, 2.13$ Spillong, Richard $2.4, 2.13$ Spillong, Richard $2.4, 2.13$ Stambaugh, Joseph 1.22 Stambaugh, Joseph 1.22 Stambaugh, Joseph $2.4, 2.13$ Starley, Roin $3.6, 8.3$ Starley, Rolard (Canyoneers, Inc.) $5.2, 12.4,$	Solomon, David	N/A
Solum, Mary 2.4; 2.13 Sorrensen, Colleen 2.4 Sorokwasz, David 3.2; 3.6 Sorokwasz, David 3.2; 3.6 Soran, Robert 2.4 Soragle, Steven (U.S. Fish and Wildlife Service) 2.2; 2.9; 3.2; 3.6; 3.7; 3.8; 8.12; 8.13; 11.2 Sparger, Janet 2.8 Sparger, Janet 2.4 Sparger, Mark 3.2; 3.6; 8.3 Speaker, Mark 3.2; 3.6; 8.3 Speaker, Mark 3.2; 3.6; 8.3 Spillen, Twike 3.2; 3.6; 8.3 Spillen, Rita <td>Solomon, Phillip (Deseret Power Electric Cooperative)</td> <td>2.6; 2.7; 2.10; 2.12; 2.21; 8.1</td>	Solomon, Phillip (Deseret Power Electric Cooperative)	2.6; 2.7; 2.10; 2.12; 2.21; 8.1
Somers, Jeff 24, 2.13 Sorensen, Colleen 24 Sorokwaz, David 32, 3.6 Sorum, Robert 24, 2.13 Souza, P. 28 Spangle, Steven (U.S. Fish and Wildlife Service) 22, 2.9, 3.2, 3.6, 3.7, 3.8, 8.12, 8.13, 11.2 Sparger, Janet 2.8 Sparks, Alan 3.6 Sparks, Alan 3.6 Spaulding, D N/A Speakman, David N/A Specter, Scott 2.4, 2.13 Spente, Liter 2.4 Spengler, Brett 2.4, 2.13 Spileger, Karen 2.4, 2.13 Spileger, Karen 2.4, 2.13 Spileger, Karen 2.4, 2.13 Spileger, Raten 2.4, 2.13 Spileger, Raten 2.4, 2.13 Spileger, Brett 2.4 Spileger, Brett 2.4 Spileger, Brett 2.4 Spileger, Bret 2.4 Spileger, Baren 2.4, 2.13 Spileger, Baren 2.4, 2.13 Spileger, Baren 2.4, 2.13 Spirakis, Jenn N/A Spirakis, Jenn <t< td=""><td>Solum, Mary</td><td>2.4; 2.13</td></t<>	Solum, Mary	2.4; 2.13
Sorensen, Colleen2.4Sorokwasz, David $3.2; 3.6$ Sorum, Robert $2.4; 2.13$ Soura, P. 2.8 Spangle, Steven (U.S. Fish and Wildlife Service) $2.2; 2.9; 3.2; 3.6; 3.7; 3.8; 8.12; 8.13; 11.2$ Sparker, Janet 2.8 Sparker, Janet 2.8 Sparker, John 3.6 Sparker, JohnN/ASpeakman, DavidN/ASpecies, Scott $2.4; 2.13$ Species, Brett $2.4; 2.13$ Spiller, Brett $2.4; 2.13$ Spiller, Mike $3.2; 3.6; 8.3$ Spillane, Rita 2.4 Spiralin, Nancy 2.4 Spirakin, JennN/ASpokony, Irving 2.4 Spirakin, John $2.4; 2.13$ St. Angelo, Richard 2.4 Standbuy, Joseph 12.2 Stan, TalilaN/AStanley, RonN/AStanley, Ron $2.4; 2.13$ Starley, Gaylord (Canyoneers, Inc.) $5.2; 12.4$ Sterker, Gaylord (Canyoneers, Inc.) $5.2; 12.4$ Sterker, John $2.4; 2.13$ Sterker, Gaylord (Canyoneers, Inc.) <td< td=""><td>Somers, Jeff</td><td>2.4; 2.13</td></td<>	Somers, Jeff	2.4; 2.13
Sorokwasz, David $3.2; 3.6$ Sorum, Robert $2.4; 2.13$ Souza, P. 2.8 Spangle, Steven (U.S. Fish and Wildlife Service) $2.2; 2.9; 3.2; 3.6; 3.7; 3.8; 8.12; 8.13; 11.2$ Sparger, Janet 2.8 Sparks, Alan 3.6 Spartow, Deb $2.4; 2.13$ Spaulding, D.N/ASpeakman, DavidN/ASpecer, Mark $3.2; 3.6; 8.3$ Spence, Liter $2.4; 2.13$ Spiegel, Karen $2.4; 2.13$ Spiker, Mike $3.2; 3.6; 8.3$ Spiegel, Karen $2.4; 2.13$ Spiker, Mike $3.2; 3.6; 8.3$ Spillane, Rita 2.4 Spiker, Mike $3.2; 3.6; 8.3$ Spiker, Mike $3.2; 3.6; 8.3$ Spillane, Rita 2.4 Spiker, Mike $3.2; 3.6; 8.3$ Spillane, Rita 2.4 Spiker, Nike $3.2; 3.6; 8.3$ Spillane, Rita 2.4 Spikour, Delaine (Bristlecone Alliance)N/ASpinelli, Nancy 2.4 Stangelo, Richard 2.4 Stangle, Joseph $1.2.2$ Stan, TalilaN/AStanley, RonN/AStanley, Ron 2.4 Starley, Ron 2.4 Starley, Ron 2.4 Starley, Ron 2.4 Starley, Gaylord (Canyone Explorations Expeditions) $9.1; 9.2; 12.1; 12.4$ Starley, Gaylord (Canyone Explorations Expeditions) $9.1; 9.2; 12.1; 12.4$ Sterley, Laura $3.2; 3.6; 8.3$ Steeley, Ion $3.2; 3.6; 8.3$ Steeley, Ion $3.2; 3.6; $	Sorensen, Colleen	2.4
Sorum, Robert $2.4; 2.13$ Souza, P. 2.8 Spangle, Steven (U.S. Fish and Wildlife Service) $2.2; 2.9; 3.2; 3.6; 3.7; 3.8; 8.12; 8.13; 11.2$ Sparger, Janet 2.8 Sparks, Alan 3.6 Sparks, Alan 3.6 Spaulding, D.NASpeakman, DavidNASpeckers, Scott $2.4; 2.13$ Speeker, Mark $3.2; 3.6; 8.3$ Spence, Liter $2.4; 2.13$ Speeker, Mark $3.2; 3.6; 8.3$ Spence, Liter $2.4; 2.13$ Speigel, Karen $2.4; 2.13$ Spiegel, Karen $2.4; 2.13$ Spiller, Mike $3.2; 3.6; 8.3$ Spillane, Rita 2.4 Spillsen, Wike $3.2; 3.6; 8.3$ Spillane, Rita 2.4 Spillsen, Nike $3.2; 3.6; 8.3$ Spillane, Rita 2.4 Spinelli, Nancy 2.4 Spikony, Irving 2.4 Spakins, John $2.4; 2.13$ Stambaugh, Joseph 12.2 Stam, TalilaNAStanley, RonNAStanley, Ron 2.4 Starley, Gayord (Canyoneers, Inc.) $5.2; 12.4$ Starley, Gayord (Canyoneers, Inc.) $5.2; 12.4$ Steele, William 3.6 Steele, William 3.6 Steeley, Canyon Explorations Expeditions) $9.1; 9.2; 12.1; 12.4$ Steeley, Canyon Copert, Canyon Explorations Expeditions) $9.1; 9.2; 2.1; 12.4$ Sterley, Canyon Explorations Expeditions) $9.1; 9.2; 3.6; 8.3$ Steeley, William 3.6 Steeley, William $3.6; $	Sorokwasz, David	3.2; 3.6
Souza, P. 2.8 Spangel, Steven (U.S. Fish and Wildlife Service) 22; 2.9; 3.2; 3.6; 3.7; 3.8; 8.12; 8.13; 11.2 Sparks, Alan 3.6 Spartow, Deb 2.4; 2.13 Sparks, Alan 3.6 Spartow, Deb 2.4; 2.13 Speaking, D. N/A Species, Scott 2.4; 2.13 Spere, Fart 2.4; 2.13 Spere, Kark 3.2; 3.6; 8.3 Spence, Liter 2.4 Spiegle, Karen 2.4; 2.13 Spilker, Mike 3.2; 3.6; 8.3 Spillane, Rita 2.4 Spilasbury, Delaine (Bristlecone Alliance) N/A Spokony, Irving 2.4 Spadink, George 2.4; 2.13 Standpel, Richard 2.4 Standpel, Roeph 12.2 Stan, Tailla N/A Standpel, Ron N/A Standpel, Karen 2.4 Standpel,	Sorum, Robert	2.4; 2.13
Spangle, Steven (U.S. Fish and Wildlife Service) 2.2; 2.9; 3.2; 3.6; 3.7; 3.8; 8.12; 8.13; 11.2 Sparks, Akan 3.6 Sparrow, Deb 2.4; 2.13 Spaulding, D. NA Species, Scott 2.4; 2.13 Spece, Mark 3.2; 3.6; 8.3 Spece, Mark 3.2; 3.6; 8.3 Spence, Liter 2.4 Spengel, Brett 2.4; 2.13 Spiegel, Karen 2.4; 2.13 Spiegel, Karen 2.4; 2.13 Spiegel, Karen 2.4; 2.13 Spiller, Mike 3.2; 3.6; 8.3 Spillane, Rita 2.4 Spinelli, Naney 2.4 Spinelli, Naney 2.4 Spiralins, John 2.4; 2.13 St. Angelo, Richard 2.4 Stambaugh, Joseph 2.2 Stan, Talila N/A Standley, Ron N/A Standley, Ron N/A Standley, Ron 2.4 Staudrey, Galyond (Canyoneers, Inc.) 5.2; 12.4 Staudrey, Galyond (Canyoneers, Inc.) 5.2; 12.4 Staueley, Jaura (Canyon Ex	Souza, P.	2.8
Sparger, Janet2.8Sparks, Alan3.6Sparrow, Deb2.4; 2.13Spaulding, D.N/ASpeekman, DavidNASpecies, Scott2.4; 2.13Spece, Mark32; 3.6; 8.3Spence, Liter2.4Spengler, Brett2.4; 2.13Spiles, Rita32; 3.6; 8.3Spilener, Rita2.4Spilsbury, Delaine (Bristlecone Alliance)N/ASpins, Ison2.4Spirakis, JennN/ASpokony, Irving2.4Spakony, Irving2.4Stanbaugh, Joseph12.2Stanbaugh, Joseph12.2Stanbaugh, Joseph12.2Stanley, RonN/AStanley, RonN/AStanley, Ron2.4Stanley, RonN/AStanley, Ron2.4Stanley, RonN/AStanley, Ron2.4Stanley, Ron2.4Stanley, Ron2.4Stanley, Ron2.4Starley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Steinhard, Carol2.4Steinhard, Carol2.4Steinhard, Carol2.4Steinhard, Rodney3.2; 3.6; 8.3Steinhard, Rodney3.2; 3.6; 8.3Stanley, Born3.2; 3.6; 8.3Starley, Calara (Canyoneers, Inc.)5.2; 12.4Sterenstra, Philip12.4 <t< td=""><td>Spangle, Steven (U.S. Fish and Wildlife Service)</td><td>2.2; 2.9; 3.2; 3.6; 3.7; 3.8; 8.12; 8.13; 11.2</td></t<>	Spangle, Steven (U.S. Fish and Wildlife Service)	2.2; 2.9; 3.2; 3.6; 3.7; 3.8; 8.12; 8.13; 11.2
Sparks, Alan 3.6 Spartow, Deb $2.4; 2.13$ Spaulding, D.N/ASpeakman, DavidN/ASpecer, Mark $3.2; 3.6; 8.3$ Specr, Mark $3.2; 3.6; 8.3$ Specr, Liter $2.4; 2.13$ Spiegel, Karen $2.4; 2.13$ Spiegel, Karen $2.4; 2.13$ Spilker, Mike $3.2; 3.6; 8.3$ Spilker, Mike $3.2; 3.6; 8.3$ Spilkbury, Delaine (Bristlecone Alliance)N/ASpindlin, Nancy 2.4 Spiralii, Nancy 2.4 Sprakis, JennN/ASpokony, Irving 2.4 Sprakis, JennN/ASpokony, Irving 2.4 Sprakis, JennN/ASpokony, Irving 2.4 Stanbaugh, Joseph 12.2 Stan, TalilaN/AStandley, RonNAStanley, Rom 2.4 Stanley, Rom 2.4 Stark, Rochny $2.4; 2.13$ Stauley, Rom $3.2; 3.6; 8.3$ Steele, Jon $3.2; 3.6; 8.3$ Steele, William 3.6 Steenstar, Philip 12.4 Steining, KarenN/ASteinher, Carol $2.4; 2.13$ Steinher, Carol $2.4; 2.13$ <	Sparger, Janet	2.8
Sparrow, Deb $2.4; 2.13$ Spaulding, D.N/ASpeakman, DavidN/ASpecies, Scott $2.4; 2.13$ Specer, Mark $3.2; 3.6; 8.3$ Spence, Liter 2.4 Spengler, Brett $2.4; 2.13$ Spiegel, Karen $2.4; 2.13$ Spillare, Rita 2.4 Spillare, Rita 2.4 Spirakis, JennN/ASpirakis, JennN/ASporagins, John $2.4; 2.13$ Stambaugh, Joseph 2.4 Stambaugh, Joseph 2.4 Stambaugh, Joseph 2.4 Stambaugh, Joseph 12.2 Stan, TailiaN/AStanley, RonN/AStanley, RonN/AStanley, Ron $3.2; 3.6; 8.3$ Stanley, Ron 12.2 Stan, TailiaN/AStanley, Ron N/A Stark, Rodney $2.4; 2.13$ Starker, Ron $3.2; 3.6; 8.3$ Steele, Jon $3.2; 3.6; 8.3$ Steele, Jon $3.2; 3.6; 8.3$ Steele, William 3.6 Steenstra, Philip 12.4 Stefanic, KarenN/ASteinhart, Carol $2.4; 2.13$ Steinhart, Ca	Sparks, Alan	3.6
Spaulding, D.N/ASpeakman, DavidN/ASpecies, Scott 2.4 ; 2.13Specre, Kark 3.2 ; 3.6 ; 8.3 Spence, Liter 2.4 Spengler, Brett 2.4 ; 2.13 Spilker, Mike 3.2 ; 3.6 ; 8.3 Spilker, Mike 3.2 ; 3.6 ; 8.3 Spillane, Rita 2.4 Spingline, Rita 2.4 Spingline, Rita 2.4 Spirgline, Rita 2.4 Spirgline, Rita 2.4 Spirgline, Rita 2.4 Spirgline, Rita 2.4 Spirgling, John 2.4 ; 2.13 St. Angelo, Richard 2.4 Spragins, John 2.4 ; 2.13 Stadnik, George 2.4 ; 2.13 Stambaugh, Joseph 12.2 Stan, TalilaN/AStanley, RonN/AStanley, Edwin RN/AStark, Rodney 2.4 ; 2.13 Staufer, Ron 2.4 Stark, Rodney 2.4 ; 2.13 Staufer, Ron 2.4 Staufer, Ron 2.4 Stauley, Laura N/A Stauley, Sayong Explorations Expeditions) $9.1; 9.2; 12.1; 12.4$ Steele, Jon $3.2; 3.6; 8.3$ Steele, William 3.6 Steenstar, Philip 12.4 Steinhar, Carol $2.4; 2.13$ <tr< td=""><td>Sparrow. Deb</td><td>2.4: 2.13</td></tr<>	Sparrow. Deb	2.4: 2.13
Speakman, David N/A Species, Scott 2.4; 2.13 Speer, Mark 3.2; 3.6; 8.3 Spence, Liter 2.4 Spengler, Brett 2.4; 2.13 Spilker, Mike 3.2; 3.6; 8.3 Spinklis, Jean N/A Spragins, John 2.4 Spragins, John 2.4; 2.13 St. Angelo, Richard 2.4 Stambaugh, Joseph 12.2 Stan, Talila N/A Stanley, Ron N/A Stanley, Ron N/A Start, Rodney 2.4; 2.13 Start, Rodney 2.4; 2.13 Starter, Ron 2.4 Starter, Ron N/A Starter, Ron 2.4 Staveley, Gaylord (Canyon Explorations Expeditions) 9.1; 9.2; 12.1; 12.4 Steele, Jon 3.2; 3.6; 8.3 Ste	Spaulding, D.	N/A
Species, Scott 2.4; 2.13 Spece, Mark 32; 3.6; 8.3 Spence, Liter 2.4 Spiggel, Karen 2.4; 2.13 Spilder, Mike 32; 3.6; 8.3 Spillane, Rita 2.4 Spillane, Rita 2.4 Spinslury, Delaine (Bristlecone Alliance) N/A Spinslury, Delaine (Bristlecone Alliance) N/A Spinslury, Delaine (Bristlecone Alliance) N/A Spokony, Irving 2.4 Spragins, John 2.4; 2.13 Stanki, George 2.4; 2.13 Stanki, George 2.4; 2.13 Stanki, George 2.4; 2.13 Standley, Ron N/A Standley, Ron N/A Stanley, Korn 2.4 Staples, Laura N/A Starbey, Norm 2.4 Starbey, Gaylord (Canyon Explorations Expeditions) 91; 92; 12.1; 12.4 Staveley, Gaylord (Canyon Explorations Expeditions) 91; 92; 12.1; 12.4 Steele, Jon 3.6 Steenstra, Philip 12.4 Steinherg, Chip 3.2; 3.6; 8.3 Steinherg, Chip 3.2; 3.6; 8.3	Speakman David	N/A
Speer, Mark $3.2, 3.6, 8.3$ Spence, Liter 2.4 Spengler, Brett $2.4, 2.13$ Spilker, Mike $3.2, 3.6, 8.3$ Spilker, Mike $3.2, 3.6, 8.3$ Spillane, Rita 2.4 Spirakis, JennN/ASpokony, Irving 2.4 Spragins, John $2.4, 2.13$ Standey, Rosephan 2.4 Standaug, Richard 2.4 Standaug, Rosephan $2.4, 2.13$ Standaug, Joseph 12.2 Stan, TalilaN/AStandley, RonN/AStanley, Edwin RN/AStarley, SuranN/AStarley, Gaylord (Canyoneers, Inc.) $5.2, 12.4$ Staveley, Gaylord (Canyoneers, Inc.) $5.2, 12.4$ Steele, Jon $3.2, 3.6, 8.3$ Steele, William 3.6 Steenstra, Philip 12.4 Steinker, KarenN/ASteinker, KarenN/ASteinker, Ching $2.4, 2.13$ Steinker, Karen $3.2, 3.6, 8.3$ Steele, Jon $3.2, 3.6, 8.3$ Steele, Jon $3.2, 3.6, 8.3$ Steinker, KarenN/ASteinker, Carol $2.4, 2.13$ Steinker, Carol $2.4, 2.13$ Steinker, Carol $2.4, 2.13$ Steinker, Carol $2.4, 2.13$ Steinker, Sobert $2.4, 2.13$ Steinker, Targe, Ching $3.2, 3.6, 8.3$ Steinker, Targe, Targe, Ching $3.2, 3.6, 8.3$ Steinker, Targe, Ching $3.2, 3.6, 8.3$ Steinker, Targe, Ching $2.4, 2.13$ Steinker, Robert $2.$	Species Scott	2.4.2.13
Spence, Liter 2.4 Spence, Liter 2.4 Spence, Liter 2.4 Spence, Liter 2.4 Spilace, Rita 2.4 Spilabury, Delaine (Bristlecone Alliance)N/ASpiralis, JennN/ASpragins, John 2.4 Spragins, John 2.4 Stadnik, George 2.4 Stanbaugh, Joseph12.2Stan, TaillaN/AStank, George 2.4 Stanke, Kern 2.4 Stanke, Koney 2.4 Stanke, Koney 2.4 Stanke, Koney 2.4 Starke, Rodney 2.4 Staveley, Gaylord (Canyoneers, Inc.) 5.2 ; 12.4Steveley, Gaylord (Canyoneers, Inc.) 5.2 ; 12.4Steveley, Gaylord (Canyoneers, Inc.) 5.2 ; 12.4Steele, Jon 3.6 Steenstra, Philip 12.4 Stefanic, KarenN/AStein, Herbert 2.4 ; 2.13Steinhart, Carol 2.4 ; 2.13Steinhart, Carol 2.4 ; 2.13Stephens, Robert 2.4 ; 2.13Stephens, Robert 2.4 ; 2.13Stephens, Robert 2.4 ; 2.13Stephens, Torol 2.4 ; 2.13Stephens, Torol 2.4 ; 2.13Stephens, Robert 2.4 ; 2.13Stephens, Torol 2.4 ; 2.13	Speeres, Seete	3 2: 3 6: 8 3
Spengler, Brett 2.4 ; 2.13 Spiegel, Karen 2.4 ; 2.13 Spilder, Mike 3.2 ; 3.6 ; 8.3 Spillane, Rita 2.4 Spiraline, Rita 2.4 Spiralin, Nancy 2.4 Spirakis, JennN/ASpokony, Irving 2.4 Spragins, John 2.4 ; 2.13 St. Angelo, Richard 2.4 Stambaugh, Joseph 12.2 Standpauly, Joseph 12.2 Standpauly, Joseph 12.2 Standley, RonN/AStanley, Edwin RN/AStanley, Sorth 2.4 Starkey, Edwin RN/AStarkey, Gaylord (Canyoneers, Inc.) $5.2; 12.4$ Staveley, Gaylord (Canyoneers, Inc.) $5.2; 12.4$ Steele, William 3.6 Steenstra, Philip 12.4 Stein, Tehrbert $2.4; 2.13$ Stein, Herbert $2.4; 2.13$ Stein, Herbert $2.4; 2.13$ Steinhart, Carol $2.4; 2.13$ Steiphens, Robert $2.4; 2.13$ Stephens, Robert $2.4; 2.13$ Stephens, Robert $2.4; 2.13$ Sterphons, Linda $2.4; 2.13$	Speed, Main	24
Spiegel, Karen2.4, 2.13Spiiker, Mike3.2; 3.6; 8.3Spilker, Mike3.2; 3.6; 8.3Spillane, Rita2.4Spinslibury, Delaine (Bristlecone Alliance)N/ASpinelli, Nancy2.4Spirakis, JennN/ASpokony, Irving2.4Spragins, John2.4; 2.13St. Angelo, Richard2.4Stambaugh, Joseph12.2Stan, TalilaN/AStanbley, RonN/AStanley, Edwin RN/AStarley, Edwin RN/AStarley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Gaylord (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Stein, Philip12.4Steinster, Philip12.4Steinster, Philip12.4Steinster, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Sterping, TysonN/A	Spengler Brett	2.1
Spiker, Mike2.4, 2.13Spilker, Mike3.2; 3.6; 8.3Spillane, Rita2.4Spilsbury, Delaine (Bristlecone Alliance)N/ASpinelli, Nancy2.4Spirakis, JennN/ASpokony, Irving2.4Spragins, John2.4; 2.13St. Angelo, Richard2.4Stadnik, George2.4; 2.13Stanhaugh, Joseph12.2Stan, TalilaN/AStandley, RonN/AStanley, Edwin RN/AStarbley, Edwin RN/AStarbley, Edwin RN/AStarbley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Gaylord (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, Jon3.6Steenstra, Philip12.4Steinhart, Carol2.4; 2.13Steinhart, Carol2.4; 2.13Stering, TysonN/A	Spiegel Karen	2.4, 2.13
Spillar, Rita2.4Spillar, Rita2.4Spinelli, Nancy2.4Spinelli, Nancy2.4Spirakis, JennN/ASpokony, Irving2.4Spragins, John2.4; 2.13St. Angelo, Richard2.4Stambaugh, Joseph12.2Stan, TalilaN/AStanley, Edwin RN/AStanley, Edwin RN/AStarffer, Ron2.4Starffer, Ron2.4Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Steinhart, Carol2.4; 2.13Steinhart, Carol2.4; 2.13Stering, TysonN/A	Spieger, Kuren	3 2: 3 6: 8 3
Spinkar, Nu2.4Spiskur, Delaine (Bristlecone Alliance)N/ASpinkis, JennN/ASpokony, Irving2.4Spragins, John2.4; 2.13St. Angelo, Richard2.4Stadnik, George2.4; 2.13Stambaugh, Joseph12.2Stan, TalilaN/AStanley, RonN/AStanley, Korm2.4Staples, LauraN/AStarley, Norm2.4Starley, Soldroff (Canyoneers, Inc.)5.2; 12.4Staveley, Gaylord (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, William3.6Steenstra, Philip12.4Steinhart, Carol2.4; 2.13Steinhart, Carol2.4; 2.13Steinhart, Carol2.4; 2.13Steinhart, Carol2.4, 2.13Steinhart, Carol2.4; 2.13Steinhart, Carol2.4; 2.13Steinhart, Carol2.4; 2.13Steinhart, Carol2.4; 2.13Steinhart, Carol2.4; 2.13Sterhare, TysonN/A	Spillane Rita	2.2, 5.0, 8.5 2.4
Spinold y, Delanc (Distriction Annalec)N/ASpinelli, Nancy2.4Spirakis, JennN/ASpokony, Irving2.4Spragins, John2.4; 2.13St. Angelo, Richard2.4Stadnik, George2.4; 2.13Stambaugh, Joseph12.2Stan, TalilaN/AStanley, RonN/AStanley, Edwin RN/AStanley, Korm2.4Stapes, LauraN/AStarley, Norm2.4Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steentra, Philip12.4Steinberg, Chip3.2; 3.6; 8.3Steinberg, Chip3.2; 3.6; 8.3Steinbart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Sterling, TysonN/A	Spillance, Nita Snilshury, Delaine (Bristlecone Alliance)	Z:T N/A
Spirati, Natry 2.4 Spirakis, JennN/ASpokony, Irving 2.4 Spragins, John 2.4 ; 2.13 St. Angelo, Richard 2.4 Stadnik, George 2.4 ; 2.13 Stambaugh, Joseph 12.2 Stan, TalilaN/AStanley, RonN/AStanley, Edwin RN/AStaples, LauraN/AStark, Rodney 2.4 ; 2.13 Staveley, Gaylord (Canyoneers, Inc.) 2.2 Staveley, Gaylord (Canyon Explorations Expeditions) 9.1 ; 9.2 ; 12.1 ; 12.4 Steele, Jon 3.2 ; 3.6 ; 8.3 Steele, William 3.6 Steenstra, Philip 12.4 Steinhert, Carol 2.4 ; 2.13 Steinhert, Carol 2.4 ; 2.13 Steinhert, Carol 2.4 ; 2.13 Steinhart, Carol 2.4 ; 2.13 Stephens, Robert 2.4 ; 2.13 Sterling, Tyson N/A	Spinsbury, Delaine (Bristiceone Amanee)	2 A
Spiracks, Jehn N/A Spokony, Irving2.4Spragins, John2.4; 2.13St. Angelo, Richard2.4Stadnik, George2.4; 2.13Stambaugh, Joseph12.2Stan, Talila N/A Standley, Ron N/A Stanley, Edwin R N/A Staples, Laura N/A Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Gaylord (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steelet, William3.6Steinberg, Chip3.2; 3.6; 8.3Steinberg, Robert2.4; 2.13Steinbarg, Robert2.4; 2.13Steinperg, Robert2.4; 2.13Sterling, Tyson N/A	Spinelli, Nancy Spirakis Jenn	2.4 N/A
Spokoly, Iving2.4Spragins, John2.4; 2.13St. Angelo, Richard2.4Stadnik, George2.4; 2.13Stambaugh, Joseph12.2Stan, TalilaN/AStanley, RonN/AStanley, Edwin RN/AStanley, Edwin RN/AStapes, LauraN/AStark, Rodney2.4; 2.13Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Laura (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Villiam3.6Steenstra, Philip12.4Steinhart, Carol2.4; 2.13Steinhart, Carol2.4; 2.13Steinhart, Carol2.4; 2.13Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Stephens, Robert2.4; 2.13Sterling, TysonN/A	Spilakis, Jeilin Spokony, Jeying	N/A 2.4
Spragnis, John2.4, 2.13St. Angelo, Richard2.4Stadnik, George2.4; 2.13Stambaugh, Joseph12.2Stan, TalilaN/AStandley, RonN/AStanley, Edwin RN/AStanley, Norm2.4Staples, LauraN/AStark, Rodney2.4; 2.13Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Gaylord (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stefanic, KarenN/ASteinhert, Carol2.4; 2.13Steinhert, Carol2.4; 2.13Steinhert, Carol2.4; 2.13Stephens, Robert2.4; 2.13Stephens, Robert2.4; 2.13Stephens, Robert2.4; 2.13Sterling, TysonN/A	Spokony, nying	2.4
St. Angelo, Klenard2.4Stadnik, George2.4; 2.13Stambaugh, Joseph12.2Stan, TalilaN/AStandley, RonN/AStanley, Edwin RN/AStanley, Korm2.4Staples, LauraN/AStark, Rodney2.4; 2.13Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Gaylord (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stein, Herbert2.4; 2.13Steinberg, Chip3.2; 3.6; 8.3Steinberg, Chip3.2; 3.6; 8.3Steinberg, Chip2.2; 1.3Stephens, Robert2.4; 2.13Stephens, Robert2.4; 2.13Sterling, TysonN/A	Spragins, John	2.4, 2.15
Stadnik, George2.4, 2.13Stambaugh, Joseph12.2Stan, TalilaN/AStandley, RonN/AStanley, Edwin RN/AStanley, Norm2.4Staples, LauraN/AStark, Rodney2.4; 2.13Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Gaylord (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stein, Herbert2.4; 2.13Steinherg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	St. Angelo, Richard	2.4
Stambaugh, Joseph12.2Stan, TalilaN/AStandley, RonN/AStanley, Edwin RN/AStanley, Norm2.4Staples, LauraN/AStark, Rodney2.4; 2.13Stauffer, Ron2.4Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Laura (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Steinherg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Stephens, Robert2.4; 2.13Sterling, TysonN/A	Staanik, George	2.4; 2.13
Stan, TalliaIN/AStandley, RonN/AStanley, Edwin RN/AStanley, Norm2.4Staples, LauraN/AStark, Rodney2.4; 2.13Stauffer, Ron2.4Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Laura (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stefanic, KarenN/ASteinherg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Stephens, Robert2.4; 2.13Sterling, TysonN/A	Stambaugn, Joseph	12.2
Standiey, RonIN/AStanley, Edwin RN/AStanley, Norm2.4Staples, LauraN/AStark, Rodney2.4; 2.13Stauffer, Ron2.4Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Laura (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stefanic, KarenN/ASteinherg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Stan, Talila	N/A
Stanley, Edwin RIN/AStanley, Norm2.4Staples, LauraN/AStark, Rodney2.4; 2.13Stauffer, Ron2.4Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Laura (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stefanic, KarenN/ASteinherg, Chip3.2; 3.6; 8.3Steinherg, Chip3.2; 3.6; 8.3Stephens, Robert2.4; 2.13Stephens, Robert2.4; 2.13Stephan, Linda2.4; 2.13Sterling, TysonN/A	Standley, Kon	N/A
Stanley, Norm2.4Staples, LauraN/AStark, Rodney2.4; 2.13Stauffer, Ron2.4Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Laura (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stefanic, KarenN/AStein, Herbert2.4; 2.13Steinherg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Stanley, Edwin R	N/A
Staples, LauraN/AStark, Rodney2.4; 2.13Stauffer, Ron2.4Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Laura (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stefanic, KarenN/AStein, Herbert2.4; 2.13Steinberg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Stanley, Norm	2.4
Stark, Rodney2.4; 2.13Stauffer, Ron2.4Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Laura (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stefanic, KarenN/ASteinherg, Chip3.2; 3.6; 8.3Stephens, Robert2.4; 2.13Stephens, Robert2.4; 2.13Sterling, TysonN/A	Staples, Laura	N/A
Stauffer, Ron2.4Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Laura (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stefanic, KarenN/AStein, Herbert2.4; 2.13Steinberg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Stark, Rodney	2.4; 2.13
Staveley, Gaylord (Canyoneers, Inc.)5.2; 12.4Staveley, Laura (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stefanic, KarenN/AStein, Herbert2.4; 2.13Steinberg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Stauffer, Ron	2.4
Staveley, Laura (Canyon Explorations Expeditions)9.1; 9.2; 12.1; 12.4Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stefanic, KarenN/AStein, Herbert2.4; 2.13Steinberg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Staveley, Gaylord (Canyoneers, Inc.)	5.2; 12.4
Steele, Jon3.2; 3.6; 8.3Steele, William3.6Steenstra, Philip12.4Stefanic, KarenN/AStein, Herbert2.4; 2.13Steinberg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Staveley, Laura (Canyon Explorations Expeditions)	9.1; 9.2; 12.1; 12.4
Steele, William3.6Steenstra, Philip12.4Stefanic, KarenN/AStein, Herbert2.4; 2.13Steinberg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Steele, Jon	3.2; 3.6; 8.3
Steenstra, Philip12.4Stefanic, KarenN/AStein, Herbert2.4; 2.13Steinberg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Steele, William	3.6
Stefanic, KarenN/AStein, Herbert2.4; 2.13Steinberg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Steenstra, Philip	12.4
Stein, Herbert 2.4; 2.13 Steinberg, Chip 3.2; 3.6; 8.3 Steinhart, Carol 2.4; 2.13 Stephens, Robert 2.4; 2.13 Steppan, Linda 2.4; 2.13 Sterling, Tyson N/A	Stefanic, Karen	N/A
Steinberg, Chip3.2; 3.6; 8.3Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Stein, Herbert	2.4; 2.13
Steinhart, Carol2.4; 2.13Stephens, Robert2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Steinberg, Chip	3.2; 3.6; 8.3
Stephens, Robert2.4; 2.13Steppan, Linda2.4; 2.13Sterling, TysonN/A	Steinhart, Carol	2.4; 2.13
Steppan, Linda2.4; 2.13Sterling, TysonN/A	Stephens, Robert	2.4; 2.13
Sterling, Tyson N/A	Steppan, Linda	2.4; 2.13
	Sterling, Tyson	N/A

Sternberg, Karin	2.4; 2.13
Stevens, Carol	2.4
Stevens, Gavi	N/A
Stevens, Lawrence (Grand Canyon Wildlands Council)	2.2; 2.5; 3.1; 7.4; 11.2
Stevens, Nike	N/A
Stevenson, Joey	N/A
Stewart, Mark	2.4; 2.13
Stewart, Sarah	2.4; 2.13
Stickel, Ann	N/A
Stieglitz, Joseph	2.4; 2.13
Stiles, Sarah	2.4; 2.13
Stillwell, Robert	3.2; 3.6; 8.3
Stoker, Cayley	N/A
Stoltenberg, John and Martha	N/A
Stone, Harry	2.4; 2.13
Stoner, Cynthia	2.4; 2.13
Stoner, Jon	2.12; 2.22
Stong, Stuart	N/A
Story, Don	2.4; 2.13
Strange, Marisa	N/A
Stratton, Jim	2.4; 2.13
Strauss, John	2.4
Streeter, Matthew	3.2; 3.6; 8.3
Strogen, Jim	2.9
Strong, Timothy	2.4
Struthers, Sue	N/A
Stuart, Michael	2.4; 2.13
Stucker, Melinda	2.4
Stump, David	2.4; 2.13
Sturm, Madeline	2.4; 2.13
Suarez, Joe	2.4; 2.13
Suda, Maryska	N/A
Sullivan, Gail	2.4; 2.13
Sullivan, Teresa	N/A
Summerville, Deborah	2.4; 2.13
Suniga, Michael	3.2; 3.6; 8.3
Sutkowski, John	2.4; 2.13
Sutton, Susan	2.4
Swan, Curtis	2.4; 2.13
Swanepoel, Karen	2.4; 2.13
Swanson, Cynthia	2.4; 2.13
Sweeney, John	N/A
Sweet, Harold	2.4
Sweet, Justin	2.4; 2.13
Sweetling, William	2.4; 2.13
Swift, Josh	3.2; 3.6; 8.3
Syapin, Peter	N/A
Sykes, Tom	N/A
Szof, Cheryl	2.4; 2.13
Szymanowski, Paul	2.4; 2.13

lacker, Barbara	2.4
Tajdari, Amy	2.4; 2.13
Talbert, Aaron	3.2; 3.6; 8.3
Laibot, J	2.4; 2.13
Talbot, I nomas	2.4; 2.13
Talleagle, David	N/A
I anner, Mary	2.4
Tao, Carol	2.4; 2.13
I arallo, Mary	2.4
laube, N/A	N/A
laylor, Rocky	3.2; 3.6; 8.3
Taylor, Donald	N/A
Teed, Halcyon	2.4
Tendler, Marlene	2.4; 2.13
Terry, Thomas	2.4; 2.13
Tetarenko, Pamela	2.4; 2.13
Thelander, Donna	2.4; 2.13
Theriault, Laurence	N/A
Thiele, Jim	2.4; 2.13
Thomas, Ron	3.2; 3.6; 8.3
Thomas, Amy (American Public Power Association)	2.4; 2.10
Thomas, George	N/A
Thomas, Ron	N/A
Thomason, Michael	2.1; 2.2; 2.8; 2.14; 13.1
Thompson, Don	N/A
Thompson, Thurston	2.4; 2.13
Thomsen, Gary	2.4; 2.13
Thorn, Debbie	2.4; 2.13
Thornbloom, Gary	2.4
Thrailkill, James	2.4
Tibbets, Linda	N/A
Tibbitts, Connie	N/A
Tibbitts, Connie	2.4; 2.13
Tigerlily, Eliot	N/A
Tipton, James	3.6
Tobin, John (Pasadena Casting Club)	2.16; 3.6
Toler, Jean	N/A
Tomaselli, Susan	N/A
Tomlin, Curtis	2.4; 2.13
Tomlinson, Lynne	2.4
Tonks. Mary	2.4: 2.13
Torchenot, Ferold	N/A
Torgerson Eric	2.4
Torres Joseph	2 4 2 13
Townsend Darlene (Phoenix Institute for Human	2 4 2 13
Development)	,
Trafecanty Chris	3 2 3 6 8 3
Travis Donna Mae	N/A
Traweek James	N/A
Trebbe Divie	24
110000, 21/110	· · · · · · · · · · · · · · · · · · ·

Trevillian, Linda	N/A
Trice, Tina	2.4; 2.13
Trinkle, Heidi	2.4; 2.13
Trout, Larry	2.4; 2.13
Trujillo, Tanya (Colorado River Board of California)	2.2; 2.4; 3.5; 3.8; 5.1; 5.2; 6.1.3; 6.7; 8.12
Tuch, Christopher	2.4; 2.13
Tudor, Geoffrey	2.4
Tuke, Carla	2.1; 2.2; 2.8; 2.14; 13.1
Turley, Steven	2.1; 2.2; 2.8; 2.14; 13.1
Turner, James	3.2; 3.6; 8.3
Turner, Dawn	2.4; 2.13
Turner, Doris	2.4; 2.13
Turner, Julie	2.4; 2.13
Turner, Kathleen	2.4; 2.13
Turner, Phyllis	2.4; 2.13
Turner, Tamara	2.4
Tuttle, Jeffrey	2.4
Twickler, Carrie	N/A
Tyrrell. Patrick (Wyoming State Engineer's Office)	2.5; 2.19; 8.7, 8.12
Tyson, Kathleen	N/A
Upp. Heidi	N/A
Uribe. Esmeralda	2.4: 2.13
Utley, Chad	3.2: 3.6: 8.3
Utter. Mark	3.2: 3.6: 8.3
Utzig. Albert	2.4: 2.13
Uvetanaka Steven	N/A
Vacek Radko	2.4
Vachula William	2 4 2 13
Vail Mark	2 4 2 13
Valdez Richard	N/A
Valentino David	2 4 2 13
Van Buren, Kenneth	N/A
Van Kampen Art	N/A
Van Kolken Robert	2 4 2 13
Van Leekwiick Natalie	2.4, 2.13
Van Pelt Jason	2.4
Van Tassell Bruce	N/A
Van Velson Nathan	2 4. 2 13
van Wijk Melissa	2.4, 2.13
Vanderhill Margo	N/Δ
Vanderl aan Harold	2 4. 2 13
VanWinkle, Jean Marie	2.4, 2.13
Variantic, Jean Marie	2.4, 2.13
Variantisa, Oleg	2.4, 2.13
Valloli, Dee Vattor Shorry	2.4, 2.15
Vauel, Shelly Vaaahiatti Darathaa	2.4, 2.15
Veltkamp Dabert	2.4, 2.13
Vermeulen Mary	2.τ, 2.13 N/Δ
Vermeulen Mary	
Vernen Cereld	1N/A
venion, Geraid	2.4, 2.13

Varnan Margarat	2 4: 2 13
Vernor, Vance	2.4, 2.15
Vigerueia John	5.2, 5.0, 6.5 2 <i>A</i> · 2 12
Viale Margaret (Hevegunei Trihe)	2.4, 2.15
Villen Jaffrer	3.2, 10.7, 10.8
Villalahan Ladia	5.2, 5.0
Villalobos, Lydia	N/A
villanova , Carolyn	N/A
Vincent, Joseph	N/A
Voelker, Paul	3.6
Vogel, Steven	2.4; 2.13
Vogelsong, Patrick	2.4; 2.13
von Eberstein, Lesli	2.4; 2.13
Von Tobel, Jefffrey	3.2; 3.6
Von tobel, William	N/A
Vorachek, Mary	2.4; 2.13
Vrymoed, John	2.1; 2.11
Waddell, Christine	N/A
Wagner, Robert	2.4
Wagner, Vickie	2.4
Wagoner, Douglas	2.4; 2.13
Waite, Paul	3.2; 3.6; 8.3
Walchak, Shelley	2.4; 2.13
Walker, Jim (Zane Grey Trout Unlimited)	2.5; 3.2; 3.3; 3.4; 3.6; 8.3; 12.2
Wallace, John (Grand Canyon State Electric Cooperative	2.6; 2.11
Association, Inc.)	,
Wallis, Tracy	2.4; 2.13
Walls, Mary	2.4
Walsh. Gary	N/A
Walsh, Justin	2.4
Walsh Tom	N/A
Walters Robyn	2.4.2.13
Walters Sandra	2.4
Walton John	2 4. 2 13
Walz Kenneth	N/A
Ward Everett	$2 1 \cdot 2 2 \cdot 2 8 \cdot 2 14 \cdot 13 1$
Ward Ken	2.1, 2.2, 2.0, 2.14, 15.1
Ward Marvin	2.1, 2.13
Warner Charles	2.π, 2.15 N/A
Warniel, Charles	24
Washburn Thomas	2.4
Washington Mortin	2.4 N/A
Washeev Swaar	N/A 2.4.2.12
Waskey, Susan	2.4; 2.13
wasser, Nancy	2.4; 2.13
Watanabe, Wayne	3.2; 3.6; 8.3
Watland, Bob	N/A
watt, Celeste	2.4
watters, Whitney	2.4; 2.13
Watts II, Howard	3.2; 3.6; 8.3
Weales, Ron	2.1; 2.2; 2.8
Weaver, Larry	3.2; 3.6; 8.3

Weber, Carol	2.4; 2.13
Wechter, Michael	N/A
Weeks, Ken	2.4
Weihrauch, Gary	3.2; 3.6; 8.3
Weinberger, Deborah	N/A
Weiner, Jeffrey	3.2; 3.6; 8.3
Weinstein, Elyette	2.4; 2.13
Weisheit, John (Living Rivers & Colorado Riverkeeper)	2.8; 2.9; 2.14; 2.18; 5.1; 6.1.1; 6.1.4; 6.1.6; 6.1.7; 11.5
Weiss, Leslie	2.4
Weiss, Stuart	2.4
Weisz, Russell	2.4; 2.13
Wellin, Paul	N/A
Wells, J.	2.4; 2.13
Wells, Michael	2.4
Welty, Walter	3.2; 3.6; 8.3
Wemer, Kim	2.4; 2.13
Wen, William	3.2; 3.6
Wendt, George (O.A.R.S.)	2.12
Wenzel. Ouentin and Jacquelyn	N/A
Werner, Katherine	2.4: 2.13
Westfall, Brian	N/A
Westfall, Sara	N/A
Wheadon Rick	N/A
Wheeler Dorothy	N/A
Wheeless Erin	2 4 2 13
Whitaker Howard	2.4, 2.13
White Bryson	N/A
White Diane	N/A
White Fugene	N/A
White Howard	2 4. 2 13
White Michael	2.τ, 2.15 N/Δ
White Scott	$2 4 \cdot 2 13$
Whitehouse Charlie	2.4, 2.15
Whitehouse, Chame	5.2, 5.0, 6.5 2 <i>A</i> · 2 12
Whiteside Cethorine	2.4, 2.15
Whiteshe, Caller Whiteshe Glen	2.4, 2.15 12.4
Whitmen Biok	2.4
Whitmire Donne	2.4, 2.15 N/A
Whitmere, Thomas	N/A N/A
White examples	N/A
Whiteer James	2.4, 2.13
Whittigh Vand	2.4, 2.13
Whitten Frile	2.4; 2.13
Whitton, Erika	2.4; 2.13
Whorley, K	N/A
wichman, Michael	2.4; 2.13
W1CK, JOd1	2.4; 2.13
WICKS, Alicia	N/A
Widen, Alyson	2.4
Wieduwilt, Trudi	2.4; 2.13
Wieland, Charles	2.4; 2.13

Wiener, William 2.4 Wightman, Kxvin 2.4 (-2.13) Wilkins, Marcelle N/A Wilkins, Rose 2.4 (-2.13) Wilkins, Rose 2.4 (-2.13) Wilkins, Rose 2.4 (-2.13) Williams, Catherine 2.4 (-2.13) Williams, Judd N/A Williams, Mara N/A Williams, Pati 2.4 (-2.13) Williams, Pati 2.4 (-2.13) Williams, Oldy N/A Williams, Ciyde 2.4 (-2.13) Williams, Ciyde 2.4 (-2.13) Williams, Ciyde 2.4 (-2.13) Williams, Ciyde 2.4 (-2.13) Williams, Angela 2.4 2.13 Williams, Mara Wilson, Ken 2.2 (-3.6, 8.3) Wilson, Kin 2.4 (-2.13) Wilson, Mark 2.4 Wilson, Mark 2.4 <th></th> <th></th>		
Wightman, Kevin 2.4; 2.13 Wilkerson, Colleen 2.4 Wilkins, Marcelle N/A Wilkins, Sose 2.4; 2.13 Wilkins, Sose 2.4; 2.13 Williams, Catherine 2.4; 2.13 Williams, Lee N/A Williams, Mariy 2.4; 2.13 Williams, Pati N/A Williams, Pati 2.4; 2.13 Williams, Oray N/A Williams, Oray N/A Williams, Clyde 2.4; 2.13 Williams, Nargela 2.4 Williams, Nagela 2.4 Wilson, Angela 2.4 Wilson, King 2.4; 2.13 Wilson, King 2.4; 2.13<	Wiener, William	2.4
Wilding, Michael N/A Wilkerson, Colleen 2.4 Wilkins, Rose 2.4, 2.13 Williams, Catherine 2.4, 2.13 Williams, Judd N/A Williams, Lee N/A Williams, Mara N/A Williams, Mitch 2.4 Williams, Patt 2.4, 2.13 Williams, Okitch 2.4, 2.13 Williams, Okito 2.4, 2.13 Williamson, Gay N/A Williamson, Gay N/A Williamson, Kiyoshi N/A Willson, Angela 2.4, 2.13 Wilson, Angela 2.4, 2.13 Wilson, Ken 2.2, 3.6, 8.3 Wilson, Michael 3.2, 3.6, 8.3 Wilson, Michael 2.2, 1.3 Wilson, Michael 2.4, 2.13 Wilson, Maron 2.4, 2.13 Wilson, Michael 2.2, 3.6, 8.3	Wightman, Kevin	2.4; 2.13
Wilkerson, Colleen 2.4 Wilkins, Rose 2.4; 2.13 Wilkins, Rose 2.4; 2.13 Williams, Catherine 2.4; 2.13 Williams, Lee N/A Williams, Lee N/A Williams, Marilyn 2.4; 2.13 Williams, Marilyn 2.4; 2.13 Williams, Marilyn 2.4; 2.13 Williams, Marilyn 2.4; 2.13 Williams, Patt 2.4; 2.13 Williams, Patt 2.4; 2.13 Williams, Oay N/A Williams, Oay N/A Williams, Oily N/A Williams, Narelea 2.4; 2.13 Williams, Narelea 2.4; 2.13 Willow, Narelea 2.2; 3.6; 8.3 Wilson, Michael 2.2; 3.6; 8.3 Wilson, Michael 2.4; 2.13	Wilding, Michael	N/A
Wilkins, MarcelleN/AWilkins, Soas 2.4 ; 2.13Williams, Catherine 2.4 ; 2.13Williams, LeeN/AWilliams, MaraN/AWilliams, MaraN/AWilliams, MaraN/AWilliams, MaraN/AWilliams, Mitch 2.4 ; 2.13Williams, Mitch 2.4 ; 2.13Williams, Pati 2.4 ; 2.13Williams, Pati 2.4 ; 2.13Williams, Williams, OayN/AWilliams, OayN/AWilliams, Oiliams, Oiliams	Wilkerson, Colleen	2.4
Wilkins, Rose 2.4; 2.13 Williams, Catherine 2.4; 2.13 Williams, Indd N/A Williams, Mara 2.4; 2.13 Williams, Mara 2.4; 2.13 Williams, Pati 2.4; 2.13 Williams, Pati 2.4; 2.13 Williams, Patiy 2.4; 2.13 Williams, Oray N/A Williams, Oray N/A Williams, Gay N/A Williams, Clyde 2.4; 2.13 Williams, David 2.4; 2.13 Wilson, Clyde 2.4; 2.13 Wilson, Mara 2.4; 2.13 Wilson, Mima 2.4; 2.13 Wilson, Michael 3.2; 3.6; 8.3 Wilson, Michael 2.2; 3.6; 8.3 Wilson, Michael 2.4; 2.13 Winster, Lindsay 2.12 <td>Wilkins, Marcelle</td> <td>N/A</td>	Wilkins, Marcelle	N/A
Williams, Lee 2.4; 2.13 Williams, Judd N/A Williams, Mara N/A Williams, Maray 2.4; 2.13 Williams, Maray 2.4; 2.13 Williams, Maray 2.4; 2.13 Williams, Patti 2.4; 2.13 Williams, Patti 2.4; 2.13 Williams, Patti 2.4; 2.13 Williams, Nitch 2.4; 2.13 Williams, Nulliam 2.4; 2.13 Williams, Niton 2.4; 2.13 Williams, Clyde 2.4; 2.13 Williams, Clyde 2.4; 2.13 Williams, Nicoshi N/A Willson, Clyde 2.4; 2.13 Wilson, Angela 2.4 Vilson, David 2.4; 2.13 Wilson, Ken 2.2; 3.6; 8.3 Wilson, Mayne 2.4; 2.13 Wilson, Michael 3.2; 3.6; 8.3 Wilson, Michael 2.4; 2.13 Wine, Jordam 2.4; 2.13 Withorp, Lauren	Wilkins, Rose	2.4; 2.13
Williams, Lee N/A Williams, Mara N/A Williams, Mara N/A Williams, Mariyn 2.4; 2.13 Williams, Mitch 2 Williams, Patt N/A Williams, Patt 2.4; 2.13 Williams, Patt 2.4; 2.13 Williams, Patty 2.4; 2.13 Williamson, Gay N/A Williamson, Kiyoshi N/A Williamson, Kiyoshi N/A Willson, Clyde 2.4; 2.13 Wilson, Angela 2.4 Wilson, Mara 2.4; 2.13 Wilson, Mara 2.4; 2.13 Wilson, Mara 2.4; 2.13 Wilson, Mim 2.4; 2.13 Wilson, Mim 2.4; 2.13 Wilson, Miran 2.4; 2.13 Wilson, Miran 2.4; 2.13 Wilson, Maran 2.4; 2.13 Wilson, Maran 2.4; 2.13 Wilson, Maran 2.4; 2.13 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wirthy, Mark 2.4 <	Williams, Catherine	2.4; 2.13
Williams, LeeN/AWilliams, Marilyn $2.4; 2.13$ Williams, Marilyn $2.4; 2.13$ Williams, Pati $2.4; 2.13$ Williams, Pati $2.4; 2.13$ Williams, Pati $2.4; 2.13$ Williams, Pati $2.4; 2.13$ Williams, Marilyn $2.4; 2.13$ Williams, Williams, Williams, Williamson, GayN/AWilliamson, GayN/AWilliamson, KiyoshiN/AWillson, Clyde $2.4; 2.13$ Wilson, Angela 2.4 Wilson, Angela $2.4; 2.13$ Wilson, Ken $32; 3.6; 8.3$ Wilson, Kim $2.4; 2.13$ Wilson, Kim $2.4; 2.13$ Wilson, Michael $32; 3.6; 8.3$ Wilson, Michael $2.2; 3.6; 8.3$ Wilson, Sharon $2.4; 2.13$ Winstead, Annie 2.4 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wisbor, JudyN/AWitorp, Lauren $2.2; 3.6; 8.3$ Witorp, Terry $32; 3.6; 8.3$ Witorp, Terry $32; 3.6; 8.3$ Wolf, Martin $32; 3.6; 8.3$ Wolf, KarenN/AWolf, KarenN/AWolf, KarenN/AWolf, KarenN/AWolf, KarenN/AWolf, Sara $2.4; 2.13$ Wood, Sara $2.4; 2.13$ Wood, Sara $2.4; 2.13$ Woodyard, Ellis, Jr. $2.4; 2.13$ Woodward, Ellis, Jr. $2.4; 2.13$ </td <td>Williams, Judd</td> <td>N/A</td>	Williams, Judd	N/A
Williams, MaraN/AWilliams, Marilyn 2.4 ; 2.13 Williams, Mitch 2.4 Williams, Pati 2.4 ; 2.13 Williams, Pati 2.4 ; 2.13 Williams, Patiy 2.4 ; 2.13 Williams, Nation 2.4 ; 2.13 Williamson, GayN/AWilliamson, GayN/AWilliamson, Clyde 2.4 ; 2.13 Williamson, KiposhiN/AWillon, Clyde 2.4 ; 2.13 Wilson, Angela 2.4 Wilson, David 2.4 ; 2.13 Wilson, Marce 2.4 ; 2.13 Wilson, Marce 2.2 ; 2.13 Wilson, Kim 2.2 ; 2.6 ; 8.3 Wilson, Mina 2.2 ; 2.13 Wilson, Mina 2.2 ; 2.13 Wilson, Mina 2.4 ; 2.13 Wilson, Miran 2.4 ; 2.13 Wilson, Maron 2.4 ; 2.13 Winstead, Annie 2.4 Wiretr, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, JudyN/AWittorp, Terry 3.2 ; 3.6 ; 8.3 Wockner, Gary (Save the Colorado) 1.6 ; 2.8 ; 2.14 ; 5.2 Wockner, Gary (Save the Colorado) 1.6 ; 2.5 ; 2.14 ; 1.5 Wolf, KarenN/AWolf, Martin 3.2 ; 3.6 ; 8.3 Wolf, KarenN/AWolf, Sara 2.4 ; 2.13 Wood, JonN/AWood, Sara 2.4 ; 2.13 Wood, Sara 2.4 ; 2.13 Woodward, Elis, Jr 2.4 ; 2.13	Williams, Lee	N/A
Williams, Marilyn 2.4; 2.13 Williams, Pat N/A Williams, Pati 2.4; 2.13 Williams, Pati 2.4; 2.13 Williams, Paty 2.4; 2.13 Williams, William 2.4; 2.13 Williams, Williams, Cay N/A Williamson, Gay N/A Williamson, Kiyoshi N/A Willon, Clyde 2.4; 2.13 Wilson, Clyde 2.4; 2.13 Wilson, Clyde 2.4; 2.13 Wilson, Angela 2.4 Wilson, David 2.4; 2.13 Wilson, Ken 3.2; 3.6; 8.3 Wilson, Kim 2.4; 2.13 Wilson, Michael 3.2; 3.6; 8.3 Wilson, Michael 2.2; 3.6; 8.3 Wilson, Jordann 2.4; 2.13 Winstead, Annie 2.4 Winter, Lindsay 2.12 Wirth, Mark 2.4 Vitorp, Terry 3.2; 3.6; 8.3 Wochext, Gary (Save the Colorado) <td< td=""><td>Williams, Mara</td><td>N/A</td></td<>	Williams, Mara	N/A
Williams, Nitch 2.4 Williams, Patti N/A Williams, Patti 2.4 ; 2.13 Williams, Patti 2.4 ; 2.13 Williams, Naty 2.4 ; 2.13 Williams, Gay N/A Williamson, Kiyoshi N/A Williamson, Clyde 2.4 ; 2.13 Willow, Olyde 2.4 ; 2.13 Willow, Angela 2.4 Wilson, Angela 2.4 Wilson, Ken $3.2; 3.6; 8.3$ Wilson, Kim $2.4; 2.13$ Wilson, Kin $2.4; 2.13$ Wilson, Kin $2.4; 2.13$ Wilson, Kin $2.4; 2.13$ Wilson, Kin $2.4; 2.13$ Wilson, Michael $3.2; 3.6; 8.3$ Wilson, Sharon $2.4; 2.13$ Winste, Jordann $2.4; 2.13$ Winste, Jordann $2.4; 2.13$ Winter, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, JudyN/AWitcher, BeverlyN/AWittorp, Lauren $3.2; 3.6; 8.3$ Wittorp, Lauren $3.2; 3.6; 8.3$ Wockner, Garg (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Wockner, Garg (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Wockner, Garg (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Wockner, Garg (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Wockner, Garg (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Wockner, Garg (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Wockner, Garg (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Wockner, Garg (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Woolf, Martin	Williams, Marilyn	2.4; 2.13
Williams, PatN/AWilliams, Pati $2.4; 2.13$ Williams, Setti $2.4; 2.13$ Williams, William $2.4; 2.13$ Williamson, GayN/AWilliamson, KiyoshiN/AWillon, Clyde $2.4; 2.13$ Wilson, Clyde $2.4; 2.13$ Wilson, David $2.4; 2.13$ Wilson, Angela 2.4 Wilson, Ken $3.2; 3.6; 8.3$ Wilson, Kin $2.4; 2.13$ Wilson, Ken $3.2; 3.6; 8.3$ Wilson, Mwayne $2.4; 2.13$ Wilson, Michael $3.2; 3.6; 8.3$ Wilson, Sharon $2.4; 2.13$ Wilson, Sharon $2.4; 2.13$ Winstead, Annie 2.4 Wirk, Jordann $2.4; 2.13$ Wirke, Jordann $2.4; 2.13$ Wirke, Jordann $2.4; 2.13$ Wirke, Jordann $2.4; 2.13$ Wirke, Jordann $2.4; 2.13$ Wirth, Mark 2.4 Virthy, Mark $2.4; 2.13$ Wirthy, Jarken $2.2; 3.6; 8.3$ Witchner, BeverlyN/AWitchner, BeverlyN/AWitchner, Gary (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Woekner, Gary (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Woekner, Gary (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Woekner, Gary (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Woekner, Gary (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Woekner, Gary (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Woekner, Gary (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Woekner, Gary (Save the Colorado) $1.6; $	Williams, Mitch	2.4
Williams, Patti $2.4; 2.13$ Williams, Williams, Williams, Williams, Williams, Williams, Gay $2.4; 2.13$ Williamson, Gay N/A Williamson, Gyoshi N/A Willson, Clyde $2.4; 2.13$ Willoms, Clyde $2.4; 2.13$ Willson, Clyde $2.4; 2.13$ Wilson, Angela 2.4 Wilson, Angela $2.4; 2.13$ Wilson, Ken $2.2; 3.6; 8.3$ Wilson, Ken $2.2; 3.6; 8.3$ Wilson, Michael $3.2; 3.6; 8.3$ Wilson, Michael $3.2; 3.6; 8.3$ Wilson, Sharon $2.4; 2.13$ Wine, Jordann $2.4; 2.13$ Winter, Lindsay 2.12 Wirth, Mark 2.4 Wise, Barbara $2.4; 2.13$ Wittorp, Lauren $3.2; 3.6; 8.3$ Wittorp, Farry $3.2; 3.6; 8.3$ Wolenter, Richard $2.4; 2.13$ Wolenter, Richard $2.4; 2.13$ Wolenter, Richard $2.4; 2.13$ Wolenter, Richard $2.2; 3.6; 8.3$ Wolenter, Richard $2.4; 2.13$ Wolf, Karen N/A	Williams, Pat	N/A
Williams, Patty 2.4 ; 2.13 Williams, William 2.4 ; 2.13 Williamson, Gay N/A Williamson, Kiyoshi N/A Williams, O. Kiyoshi N/A Willon, Clyde 2.4 ; 2.13 Wilson, Angela 2.4 Wilson, Angela 2.4 Wilson, Ken 3.2 ; 3.6 ; 8.3 Wilson, Kim 2.4 ; 2.13 Wilson, Kim 2.4 ; 2.13 Wilson, Kim 2.4 ; 2.13 Wilson, Michael 3.2 ; 3.6 ; 8.3 Wilson, Michael 3.2 ; 3.6 ; 8.3 Wilson, Jordann 2.4 ; 2.13 Wine, Jordann 2.4 ; 2.13 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wiseboro, Judy N/A Witchner, Beverly N/A Witchner, Beverly N/A Witchner, Beverly N/A Witchner, Gary (Save the Colorado) 1.6 ; 2.8 ; 2.14 ; 5.2 Woekner, Gary (Save the Colorado) 1.6 ; 2.8 ; 2.14 ; 5.2 Wolet, Rarban 2.4 ; 2.13 Wolf, Karen N/A Wolf	Williams, Patti	2.4; 2.13
Williams, William $2.4; 2.13$ Williamson, GayN/AWilliamson, KiyoshiN/AWillson, Clyde $2.4; 2.13$ Wilson, Clyde $2.4; 2.13$ Wilson, Angela 2.4 Wilson, Nagela 2.4 Wilson, Ken $3.2; 3.6; 8.3$ Wilson, Kim $2.4; 2.13$ Wilson, Mayne $2.4; 2.13$ Wilson, Michael $3.2; 3.6; 8.3$ Wilson, Sharon $2.4; 2.13$ Winstead, Annice 2.4 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wise, Barbara $2.4; 2.13$ Wittorp, Terry $3.2; 3.6; 8.3$ Wittorp, Terry $3.2; 3.6; 8.3$ Wittorp, Terry $3.2; 3.6; 8.3$ Woelk, Barbara $2.4; 2.13$ Woelk, Barbara $2.4; 2.13$ Woolf, KarenN/AWolf, KarenN/AWolf, KarenN/AWolf, Martin $3.2; 3.6; 8.3$ Wolf, Martin $3.2; 3.6; 8.3$ Woolf, Martin $3.2; 3.6; 8.3$ Woolf, KarenN/AWoolf, KarenN/AWool, JonN/AWood, JonN/AWood, JonN/AWood, Sara $2.4; 2.13$ Woodaul, Greg $2.4; 2.13$ Woodaul, Greg $2.4; 2.13$	Williams, Patty	2.4: 2.13
Williamson, GayN/AWilliamson, KiyoshiN/AWillson, Clyde 2.4 ; 2.13 Wilson, Clyde 2.4 ; 2.13 Wilson, Angela 2.4 Wilson, David 2.4 ; 2.13 Wilson, Kim 2.4 ; 2.13 Wilson, Kim 2.4 ; 2.13 Wilson, Mina 2.2 ; 3.6 ; 8.3 Wilson, Kim 2.4 ; 2.13 Wilson, Michael 3.2 ; 3.6 ; 8.3 Wilson, Sharon 2.4 ; 2.13 Winstead, Annie 2.4 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wise, Barbara 2.4 ; 2.13 Withorp, Terry 3.2 ; 3.6 ; 8.3 Workner, Gary (Save the Colorado) 1.6 ; 2.8 ; 2.14 ; 5.2 Woodkner, Gary (Save the Colorado) 1.6 ; 2.8 ; 2.14 ; 5.2 Wolf, KarenN/AWolf, KarenN/AWolf, KarenN/AWolf, KarenN/AWood, Sara 2.4 ; 2.13 Woodall, Greg 2.4 ; 2.13 Woodall, Greg 2.4 ; 2.13 Woodall, Greg 2.4 ; 2.13	Williams, William	2.4: 2.13
Willamson, Kiyoshi N/A Willamson, Kiyoshi N/A Wilson, Clyde 2.4; 2.13 Wilson, Angela 2.4 Wilson, David 2.4; 2.13 Wilson, Magela 2.4 Wilson, Men 3.2; 3.6; 8.3 Wilson, Kim 2.4; 2.13 Wilson, MWayne 2.4; 2.13 Wilson, Michael 3.2; 3.6; 8.3 Wilson, Sharon 2.4; 2.13 Winson, Sharon 2.4; 2.13 Winstead, Annie 2.4 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, Judy N/A Wiseboro, Judy N/A Witorp, Terry 3.2; 3.6; 8.3 Wittorp, Terry 3.2; 3.6; 8.3 Wittorp, Terry 3.2; 3.6; 8.3 Wocker, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolf, Karen N/A Wolf, Karen N/A Wolf, Karen N/A Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6	Williamson Gay	N/A
Willson, Clyde 2.4; 2.13 Willson, Angela 2.4 Wilson, Angela 2.4 Wilson, David 2.4; 2.13 Wilson, Ken 3.2; 3.6; 8.3 Wilson, Kim 2.4; 2.13 Wilson, Kim 2.4; 2.13 Wilson, Kim 2.4; 2.13 Wilson, Kichael 3.2; 3.6; 8.3 Wilson, Sharon 2.4; 2.13 Wine, Jordann 2.4; 2.13 Wine, Jordann 2.4; 2.13 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wise, Barbara 2.4; 2.13 Witchner, Beverly N/A Wittorp, Lauren 3.2; 3.6; 8.3 Wittorp, Lauren 3.2; 3.6; 8.3 Wockner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woekler, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolf, Martin 3.2; 3.6; 8.3 Wolf, Karen N/A Wolf, Karen N/A Wolg, Steve 3.6 Wood, Jon N/A Wood, Sara 2.4; 2.13 Wood, Greg <	Williamson, Kiyoshi	N/A
Mink, Ope $2.4, 2.13$ Wilson, Angela 2.4 Wilson, David $2.4; 2.13$ Wilson, Ken $3.2; 3.6; 8.3$ Wilson, Kim $2.4; 2.13$ Wilson, Min $2.4; 2.13$ Wilson, Michael $3.2; 3.6; 8.3$ Wilson, Sharon $2.4; 2.13$ Winstead, Annie $2.4; 2.13$ Winter, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, JudyN/AWise, Barbara $2.4; 2.13$ Wittorp, Lauren $3.2; 3.6; 8.3$ Wittorp, Lauren $3.2; 3.6; 8.3$ Wockner, Gary (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Woelk, Barbara $2.4; 2.13$ Wolf, KarenN/AWolf, KarenN/AWolf, KarenN/AWolf, Sara $2.4; 2.13; 12.3$ Wood, JonN/AWood, JonN/AWood, Sara $2.4; 2.13$ Woodall, Greg $2.4; 2.13$ Woodaul, Greg $2.4; 2.13$ Woodaul, Greg $2.4; 2.13$	Willson Clyde	2 4. 2 13
Nip, Daged 2.4 Wilson, Angela 2.4 Wilson, Ken 3.2; 3.6; 8.3 Wilson, Kim 2.4; 2.13 Wilson, M Wayne 2.4; 2.13 Wilson, Michael 3.2; 3.6; 8.3 Wilson, Michael 3.2; 3.6; 8.3 Wilson, Sharon 2.4; 2.13 Winstead, Annie 2.4 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wise, Barbara 2.4; 2.13 Witchner, Beverly N/A Witchp, Lauren 3.2; 3.6; 8.3 Wittorp, Lauren 3.2; 3.6; 8.3 Woolk, Barbara 2.4; 2.13 Woolk, Barbara 2.4; 2.13 Wooletter, Richard 2.4; 2.13 Wooletter, Richard 2.4; 2.13 Wolf, Martin 3.2; 3.6; 8.3 Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wood, Jon N/A Wood, Sara 2.4; 2.13 Woodudil, Greg 2.4; 2.13	Wiln Ludger	N/A
Nilson, Jugut 2.4 Wilson, David 2.4 ; 2.13 Wilson, Ken 3.2 ; 3.6 ; 8.3 Wilson, Kim 2.4 ; 2.13 Wilson, Michael 3.2 ; 3.6 ; 8.3 Wilson, Michael 3.2 ; 3.6 ; 8.3 Wilson, Sharon 2.4 ; 2.13 Wine, Jordann 2.4 ; 2.13 Winetr, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, JudyN/AWitsboro, JudyN/AWitcher, BeverlyN/AWittorp, Terry 3.2 ; 3.6 ; 8.3 Wittorp, Lauren 3.2 ; 3.6 ; 8.3 Wockner, Gary (Save the Colorado) 1.6 ; 2.8 ; 2.14 ; 5.2 Woelk, Barbara 2.4 ; 2.13 Wolenter, Richard 2.4 ; 2.13 Wolf, Martin 3.2 ; 3.6 ; 8.3 Wolf, Martin 3.2 ; 3.6 ; 8.3 Wolf, Wesley 2.1 ; 2.2 , 2.8 ; 2.14 ; 13.1 Wolongevicz, Patricia 2.4 Wood, JonN/AWood, JonN/AWood, Sara 2.4 ; 2.13 Woodall, Greg 2.4 ; 2.13	Wilson Angela	24
Nilson, Ken $2.2, 3.6, 8.3$ Wilson, Kim $2.4, 2.13$ Wilson, Michael $3.2, 3.6, 8.3$ Wilson, Michael $3.2, 3.6, 8.3$ Wilson, Sharon $2.4, 2.13$ Wine, Jordann $2.4, 2.13$ Winter, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, JudyN/AWise, Barbara $2.4, 2.13$ Wittorp, Lauren $3.2, 3.6, 8.3$ Wittorp, Lauren $3.2, 3.6, 8.3$ Wockner, Gary (Save the Colorado) $1.6, 2.8, 2.14, 5.2$ Woelk, Barbara $2.4, 2.13$ Wolf, Martin $3.2, 3.6, 8.3$ Wolf, Martin $3.2, 3.6, 8.3$ Wolf, Martin $3.2, 3.6, 8.3$ Wolf, Wesley $2.1, 2.2, 2.8, 2.14, 13.1$ Wolongevicz, Patricia 2.4 Wood, JonN/AWood, JonN/AWood, Sara $2.4, 2.13$ Woodall, Greg $2.4, 2.13$ Woodall, Greg $2.4, 2.13$	Wilson David	2.1
Wilson, Kim 2.4; 2.13 Wilson, Michael 3.2; 3.6; 8.3 Wilson, Michael 3.2; 3.6; 8.3 Wilson, Sharon 2.4; 2.13 Winson, Jordann 2.4; 2.13 Winstead, Annie 2.4 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, Judy N/A Wisboro, Judy N/A Witchner, Beverly N/A Wittorp, Lauren 3.2; 3.6; 8.3 Workner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolf, Karen N/A Wolf, Karen N/A Wolf, Karen N/A Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wood, Jon N/A Wood, Sara 2.4; 2.13 Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.13	Wilson Ken	3 2: 3 6: 8 3
Wilson, Mi 2-4, 2-13 Wilson, Michael 3.2; 3.6; 8.3 Wilson, Sharon 2.4; 2.13 Wine, Jordann 2.4; 2.13 Winstead, Annie 2.4 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, Judy N/A Wise, Barbara 2.4; 2.13 Witchner, Beverly N/A Wittorp, Lauren 3.2; 3.6; 8.3 Wittorp, Terry 3.2; 3.6; 8.3 Wockner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolenter, Richard 2.4; 2.13 Wolf, Karen N/A Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.13	Wilson Kim	2 4. 2 13
Wilson, Mi wahr $2.4, 2.13$ Wilson, Michael $3.2, 3.6, 8.3$ Wilson, Sharon $2.4, 2.13$ Wine, Jordann $2.4, 2.13$ Winter, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, JudyN/AWise, Barbara $2.4, 2.13$ Witchner, BeverlyN/AWittorp, Terry $3.2, 3.6, 8.3$ Wockner, Gary (Save the Colorado) $1.6, 2.8, 2.14, 5.2$ Woelk, Barbara $2.4, 2.13$ Wolf, KarenN/AWolf, KarenN/AWolf, Wesley $2.1, 2.2, 2.8, 2.14, 13.1$ Wolf, Wesley $2.1, 2.2, 2.8, 2.14, 13.1$ Wood, JonN/AWood, JonN/AWood, Sara $2.4, 2.13$ Woodall, Greg $2.4, 2.13$ Woodall, Greg $2.4, 2.13$ Woodall, Greg $2.4, 2.13$ Woodall, Greg $2.4, 2.13$	Wilson M Wayne	2.1, 2.13
Wilson, Nichael $5.2, 5.0, 5.3$ Wilson, Sharon $2.4; 2.13$ Wine, Jordann $2.4; 2.13$ Winstead, Annie 2.4 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, JudyN/AWisboro, JudyN/AWitsboro, JudyN/AWittorp, BeverlyN/AWittorp, Lauren $3.2; 3.6; 8.3$ Wockner, Gary (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Woelk, Barbara $2.4; 2.13$ Wolenter, Richard $2.4; 2.13$ Wolf, KarenN/AWolf, Wartin $3.2; 3.6; 8.3$ Wolf, Wesley $2.1; 2.2; 2.8; 2.14; 13.1$ Wolongevicz, Patricia 2.4 Wood, JonN/AWood, JonN/AWood, Sara $2.4; 2.13$ Wood, Sara $2.4; 2.13$ Woodall, Greg $2.4; 2.13$ Woodall, Greg $2.4; 2.13$	Wilson Michael	2.7, 2.15
Wine, Jordann $2.4, 2.13$ Wine, Jordann $2.4, 2.13$ Winter, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, JudyN/AWise, Barbara $2.4, 2.13$ Witchner, BeverlyN/AWittorp, Lauren $3.2; 3.6; 8.3$ Wockner, Gary (Save the Colorado) $1.6; 2.8; 2.14; 5.2$ Woelk, Barbara $2.4; 2.13$ Wolenter, Richard $2.4; 2.13; 12.3$ Wolf, KarenN/AWolf, Wartin $3.2; 3.6; 8.3$ Wolf, Wesley $2.1; 2.2; 2.8; 2.14; 13.1$ Wolongevicz, Patricia 2.4 Wood, JonN/AWood, JonN/AWood, Sara $2.4; 2.13$ Woodall, Greg $2.4; 2.13$ Woodall, Greg $2.4; 2.13$	Wilson Sharon	2.4. 2.12
Wink, Jordani 2.4, 2.13 Winstead, Annie 2.4 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, Judy N/A Wise, Barbara 2.4; 2.13 Witthner, Beverly N/A Wittorp, Lauren 3.2; 3.6; 8.3 Wockner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolenter, Richard 2.4; 2.13; 12.3 Wolf, Karen N/A Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wood, Jon N/A Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.13 Woodall, Greg 2.4; 2.13	Wine Jordann	2.4, 2.13
Winter, Lindsay 2.4 Winter, Lindsay 2.12 Wirth, Mark 2.4 Wisboro, Judy N/A Wise, Barbara 2.4; 2.13 Witchner, Beverly N/A Wittorp, Lauren 3.2; 3.6; 8.3 Wotkner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolenter, Richard 2.4; 2.13; 12.3 Wolf, Karen N/A Wolf, Martin 3.2; 3.6; 8.3 Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Peter 2.4; 2.13 Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.13 Woodward, Ellis, Jr 2.4; 2.13	Wine, Jordann	2.4, 2.15
Winter, Enidsay 2.12 Wirth, Mark 2.4 Wisboro, Judy N/A Wise, Barbara 2.4; 2.13 Witchner, Beverly N/A Wittorp, Lauren 3.2; 3.6; 8.3 Wittorp, Terry 3.2; 3.6; 8.3 Wockner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolenter, Richard 2.4; 2.13; 12.3 Wolf, Karen N/A Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.13 Woodward, Ellis, Jr 2.4; 2.13	Winter Lindson	2.4
Wirth, Mark 2.4 Wisboro, Judy N/A Wise, Barbara 2.4; 2.13 Witchner, Beverly N/A Wittorp, Lauren 3.2; 3.6; 8.3 Wittorp, Terry 3.2; 3.6; 8.3 Wockner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Woletter, Richard 2.4; 2.13; 12.3 Wolf, Karen N/A Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolngevicz, Patricia 2.4 Woog, Steve 3.6 Wood, Jon N/A Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.13 Woodall, Greg 2.4; 2.13	winter, Elitosay	2.12
Wisboro, Judy N/A Wise, Barbara 2.4; 2.13 Witchner, Beverly N/A Wittorp, Lauren 3.2; 3.6; 8.3 Wittorp, Terry 3.2; 3.6; 8.3 Wockner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolenter, Richard 2.4; 2.13; 12.3 Wolf, Karen N/A Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.13 Woodward, Ellis, Jr 2.4; 2.13	Wirth, Mark	2.4
Wise, Barbara 2.4; 2.13 Witchner, Beverly N/A Wittorp, Lauren 3.2; 3.6; 8.3 Wittorp, Terry 3.2; 3.6; 8.3 Wockner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolenter, Richard 2.4; 2.13; 12.3 Wolf, Karen N/A Wolf, Martin 3.2; 3.6; 8.3 Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.13 Woodward, Ellis, Jr 2.4; 2.13	Wisboro, Judy	N/A
Witchner, Beverly N/A Wittorp, Lauren 3.2; 3.6; 8.3 Wittorp, Terry 3.2; 3.6; 8.3 Wockner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolenter, Richard 2.4; 2.13; 12.3 Wolf, Karen N/A Wolf, Martin 3.2; 3.6; 8.3 Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.13 Woodward, Ellis, Jr 2.4; 2.13	Wise, Barbara	2.4; 2.13
Wittorp, Lauren 3.2; 3.6; 8.3 Wittorp, Terry 3.2; 3.6; 8.3 Wockner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolenter, Richard 2.4; 2.13; 12.3 Wolf, Karen N/A Wolf, Martin 3.2; 3.6; 8.3 Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Peter 2.4; 2.13 Woodall, Greg 2.4; 2.13 Woodward, Ellis, Jr 2.4; 2.13	Witchner, Beverly	N/A
Wittorp, Terry 3.2; 3.6; 8.3 Wockner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolenter, Richard 2.4; 2.13; 12.3 Wolf, Karen N/A Wolf, Martin 3.2; 3.6; 8.3 Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Peter 2.4; 2.13 Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.13 Woodward, Ellis, Jr 2.4; 2.13	Wittorp, Lauren	3.2; 3.6; 8.3
Wockner, Gary (Save the Colorado) 1.6; 2.8; 2.14; 5.2 Woelk, Barbara 2.4; 2.13 Wolenter, Richard 2.4; 2.13; 12.3 Wolf, Karen N/A Wolf, Martin 3.2; 3.6; 8.3 Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Peter 2.4; 2.13 Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.8 Woodward, Ellis, Jr 2.4; 2.13	Wittorp, Terry	3.2; 3.6; 8.3
Woelk, Barbara 2.4; 2.13 Wolenter, Richard 2.4; 2.13; 12.3 Wolf, Karen N/A Wolf, Martin 3.2; 3.6; 8.3 Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Peter 2.4; 2.13 Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.8 Woodward, Ellis, Jr 2.4; 2.13	Wockner, Gary (Save the Colorado)	1.6; 2.8; 2.14; 5.2
Wolenter, Richard 2.4; 2.13; 12.3 Wolf, Karen N/A Wolf, Martin 3.2; 3.6; 8.3 Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Peter 2.4; 2.13 Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.8 Woodward, Ellis, Jr 2.4; 2.13	Woelk, Barbara	2.4; 2.13
Wolf, Karen N/A Wolf, Martin 3.2; 3.6; 8.3 Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Peter 2.4; 2.13 Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.8 Woodward, Ellis, Jr 2.4; 2.13	Wolenter, Richard	2.4; 2.13; 12.3
Wolf, Martin 3.2; 3.6; 8.3 Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Peter 2.4; 2.13 Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.8 Woodward, Ellis, Jr 2.4; 2.13	Wolf, Karen	N/A
Wolf, Wesley 2.1; 2.2; 2.8; 2.14; 13.1 Wolongevicz, Patricia 2.4 Wong, Steve 3.6 Wood, Jon N/A Wood, Peter 2.4; 2.13 Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.8 Woodward, Ellis, Jr 2.4; 2.13	Wolf. Martin	3.2: 3.6: 8.3
Wolongevicz, Patricia2.4Wong, Steve3.6Wood, JonN/AWood, Peter2.4; 2.13Woodall, Greg2.4; 2.8Woodward, Ellis, Jr2.4; 2.13	Wolf, Wesley	2.1: 2.2: 2.8: 2.14: 13.1
Wong, Steve3.6Wood, JonN/AWood, Peter2.4; 2.13Wood, Sara2.4; 2.13Woodall, Greg2.4; 2.8Woodward, Ellis, Jr2.4; 2.13	Wolongevicz Patricia	2.4
Wood, JonN/AWood, Peter2.4; 2.13Wood, Sara2.4; 2.13Woodall, Greg2.4; 2.8Woodward, Ellis, Jr2.4; 2.13	Wong Steve	3.6
Wood, Peter 2.4; 2.13 Wood, Sara 2.4; 2.13 Woodall, Greg 2.4; 2.8 Woodward, Ellis, Jr 2.4; 2.13	Wood. Jon	N/A
Wood, Sara2.4; 2.13Woodall, Greg2.4; 2.8Woodward, Ellis, Jr2.4; 2.13	Wood, Peter	2.4: 2.13
Woodall, Greg2.4; 2.8Woodward, Ellis, Jr2.4; 2.13	Wood, Sara	2.4: 2.13
Woodward, Ellis, Jr 2.4; 2.13	Woodall, Greg	2.4: 2.8
	Woodward, Ellis, Jr	2.4: 2.13

W7 1'1 XT	
Woolridge, Nancy	2.4; 2.11; 2.13
Worley, Robert	2.4
Worth, Daniel	3.0
Worthy, Crista	2.8; 2.14; 13.1
Wostenberg, Darren	2.4
Wright III, Trigg	2.4; 2.13
Wright, Steven	2.4; 2.13
Wurtz, Betty	N/A
Wyatt, Cathy	2.4; 2.13
Wyatt, Jennifer	N/A
Wykstra, Kyle	2.13; 4.1.2;7.3.1; 11.3
Wylie, Mary	2.4; 2.13
Wyse, Margo	N/A
Xavier, Marjorie	2.4
Yarger, Andrea	2.4; 2.13
Yasgur, Eleanor	2.4
Yost, Gaylord	2.4; 2.13
Young, Dennis	N/A
Young, Jo Ellen	2.4
Yount, Madeline	N/A
Yount, Robin	3.2; 3.6; 8.3
Yun, Allen	2.4; 2.13
Zagona, Helen	3.6
Zanger, Manfred	N/A
Zastrow, Lila and Hendrickson, Dave	2.4; 2.13
Zei, Wayne	3.2; 3.6; 8.3
Zerr, Laura	2.4; 2.13
Zerzan, Paula	2.4; 2.13
Zeutenhorst, Dennis	2.4; 2.13
Zick, Edward	N/A
Zieber, Thomas	2.4; 2.13
Ziegler, Russell	N/A
Ziegler, russell	N/A
Zielinski, Virginia	2.4; 2.13
Zirasri, Ran	2.4; 2.13
Zornesky, Jerome	2.4; 2.13
Zoupas, Robbin	N/A
Zuckerman, Ben	2.1; 2.2; 2.8; 2.14; 13.1

^a For individuals or organizations that entered comments to the PEPC online database, names and affiliations shown here are as provided in those entries. Occasionally, hard copies of the same comments were mailed to DOI with a different signatory.

^b N/A = Not applicable; only editorial or non-substantive comments were provided.

^c NA -= Not available; first or last name was not available in the comment submittal to PEPC